2023-06.06.q

A resolution authorizing the approval of a service agreement by and between Fairfield County Job & Family Services, Child Protective Services Division and A New Leaf, Inc.

WHEREAS, Fairfield County Job & Family Services, Child Protective Services is requesting the Board of Commissioners approval of a service agreement with A New Leaf, Inc., 221 Tarlton Rd., Circleville, OH 43113 and

WHEREAS, the purpose of the service agreement is to provide Network Placement and Related Services for children who are in the care and custody of the Agency; and

WHEREAS, this agreement shall be effective July 1st, 2023 through June 30th, 2024; and

WHEREAS, a purchase order encumbering the funds for the services was acquired; and

WHEREAS, the Prosecuting Attorney has approved the agreement as to form.

NOW THEREFORE, BE IT RESOLVED BY THE BOARD OF COUNTY COMMISSIONERS, COUNTY OF FAIRFIELD, AND STATE OF OHIO:

Section 1. That the Fairfield County Board of Commissioners hereby approves the attached Network Placement Service Agreement for A New Leaf, Inc.

Prepared by: Brandi Downhour cc: JFS / Fiscal Supervisor

ROUTING FORM FOR CONTRACTS
The undersigned designee of the County affirms that he/she has reviewed the attached contract to ensure that it complies with County's needs and previous negotiations. The undersigned designee further affirms that the County has complied with the competitive selection process, as prescribed by the Ohio Revised Code, by selecting one of the boxes below.
A. Goods and/or Services in excess of \$50,000.00—competitively selected via an Invitation to Bid, pursuant to R.C. 307.86-307.92
B. Goods and/or Services in excess of \$50,000.00—competitively selected via a Request for Proposals, pursuant to R.C. 307.862
C. Dublic Improvement contracts—competitively selected pursuant to R.C. 153.08-153.12
D. Architect/Engineer design services for public improvements—selected through the Request for Qualifications process pursuant to R.C. 153.65-153.72
E. County Road Improvement/Construction-competitively selected pursuant to R.C. 5555.61
F. The subject matter was exempt from competitive selection for the following reason(s):
 Under \$50,000 State Term #: (copy of State Term Contract must be attached) ODOT Term #: (See R.C. 5513.01) Professional Services (See R.C. 307.86) Emergency (Follow procedure under ORC 307.86(A)) Sole Source (attach documentation as to why contract is sole source) Other: (cite to authority or explain why matter is exempt from competitive bidding)
G. Agreement not subject to Sections A-F (explain):
H. Compliance with Fairfield County Board of Commissioners Procurement Guidelines
 No County employee, employee's family member, or employee's business associate has an interest in this contract OR such interest has been disclosed and reviewed by the Prosecutor's Office No Finding for Recovery against Vendor as required under R.C. 9.24 (search via "Certified Search" on http://ffr.ohioauditor.gov/) Obtained 3 quotes for purchases under \$50,000 Purchase Order is included with Agreement
Signed this day of, 20
$T_{2}(1)$

* Please note that this checklist only addresses County and statutory requirements. If a contract is paid for with state and/or federal funds, please consult with the appropriate state and/or federal agency to ensure your department is complying with any additional requirements. By submitting a request for approval, you are certifying you have addressed County, statutory, and grant requirements.*

Fiscal Supervisor

Name and Title

01.2018 COST ANALYSIS:

For foster care placement, network providers have Title IV-E reimbursement ceilings, and Fairfield County rates have typically been below the state negotiated ceilings. Historically, the traditional, daily rate is less than \$225.

The review and evaluation of the separate cost elements and proposed profit would include an evaluation of special considerations and special needs, as there are cases which would be reviewed independently based on extraordinary factors. If the rate was more than \$225 per day, it is expected that there would be extraordinary, case specific needs, knowing what we know about the market in our area.

REPRINT	Carrí L. Brown, phd, mba, cgfm		Purchase Order
	Fairfield County Auditor 210 East Main Street Lancaster, Ohio 43130	Fiscal Year 2023	Page: 1 of 1
			PPEAR ON ALL INVOICES, SHIPPING PAPERS.
B JOB & FAMILY SERVICES	Revisions: 004	Purchase Order #	23000290 - 04
 239 W MAIN STREET L ANCASTER, OH 43130 Phone: 740-652-7889 		Delivery must be made within	n doors of specified destination.
		Expiration Date:	12/15/2023
A NEW LEAF INC PO BOX 615 KINGSTON, OH 45644	S H P T O	JOB & FAMILY SERVICES 239 W MAIN STREET LANCASTER, OH 43130 Phone: 740-652-7889	5

VENDOR PHONE N	UMBER VEN	IDOR FAX NUMBER	REQUISITION NUMBER	DELIVERY REFERENCE
740-420-070	0		349	
DATE ORDERED	VENDOR NUMBER	DATE REQUIRED	FREIGHT METHOD/TERMS	DEPARTMENT/LOCATION
01/01/2023	8246			JOB & FAMILY SERVICES
NOTES				

BOARD AND CARE

The Above Purchase Order Number Must Appear On All Correspondence - Packing Sheets And Bills Of Lading

ITEM #	DESCRIPTION / PART #	QTY	UOM	UNIT PRICE	EXTENDED PRICE
1	MODIFIED: BOARD AND CARE	1.0	EACH	\$66,100.00	\$66,100.00

06/06/2023	Auditor Fairfield County, OH	Purchase Order Total	\$66,100.00
Date: 01/01/2023	Carli L. Brown	Total Credit	\$0.00
		Total Discount	\$0.00
County Treasury or in process o certification now outstanding.	f collection to the credit of the submitted Fund(s) free from any obligation or	Total Freight	\$0.00
It is hereby certified that the amount \$66,100.00 required to meet the contract, agreement, obligation, payment or expenditure, for the above, has been lawfully appropriated, authorized or directed for such purpose and is in the		Total Sales Tax	\$0.00
COUNTY AUDITOR'S CERTIFICA	TE	Total Ext. Price	\$66,100.00

06/06/2023

Vendor Copy

416

An official website of the United States government <u>Here's how you know</u>

SAM,GOV®		
	All Words	
earch	e.g. 1606N020Q02	
Select Domain All Domains		+
Filter By Keyword Search For more information on how to use o Any Words (i)	our keyword search, visit our help guide	
Keyword Search For more information on how to use o	our keyword search, visit our help guide	
Keyword Search For more information on how to use o Any Words (i) All Words (i)	our keyword search, visit our help guide	
Keyword Search For more information on how to use o Any Words (i) All Words (i) Exact Phrase (i)	our keyword search, visit our help guide	
Keyword Search For more information on how to use o Any Words (i) All Words (i) Exact Phrase (i) e.g. 1606N020Q02 "A New Leaf, Inc."		
Keyword Search For more information on how to use o Any Words (i) All Words (i) Exact Phrase (i) e.g. 1606N020Q02		

Inactive



i	No matches found Your search did not return any results. To view Entity Registrations, you must sign in.		
	Sign In		
	Would you like to include inactive records in your search results? Yes Go Back		



Our Website	
Our Partners	
Policies	
Customer Service	



This is a U.S. General Services Administration Federal Government computer system that is **"FOR OFFICIAL USE ONLY."** This system is subject to monitoring. Individuals found performing 418

https://sam.gov/search/?page=1&pageSize=25&sort=-modifiedDate&sfm%5BsimpleSearch%5D%5BkeywordRadio%5D=EXACT&sfm%5BsimpleSear... 2/3

SAM.gov | Search

unauthorized activities are subject to disciplinary action including criminal prosecution.

Ohio Department of Job and Family Services

AGREEMENT FOR TITLE IV-E AGENCIES AND PROVIDERS FOR THE PROVISION OF CHILD PLACEMENT

This Agreement sets forth the terms and conditions between the parties for placement services for children who are in the care and custody of the Agency named below.

This Agreement is between Fairfield County Department of Job and Family Services, a Title IV-E Agency, hereinafter "Agency", whose address is:

Fairfield County Department of Job and Family Services 239 W Main St Lancaster, OH 43130

and

A New Leaf, Inc., hereinafter "Provider", whose address is:

A New Leaf, Inc. 221 Tarlton Rd Circleville, OH 43113

Collectively the "Parties".

Table of Contents

ARTICLE I. Section 1.01 Section 1.02 Section 1.03 ARTICLE II. ARTICLE II. ARTICLE IV. ARTICLE VI. ARTICLE VI. ARTICLE VI. ARTICLE VII. ARTICLE XI. ARTICLE XI. ARTICLE XI. ARTICLE XI. ARTICLE XI. ARTICLE XV. ARTICLE XXI. ARTICLE XXI. ARTICLE XXI. ARTICLE XXI. ARTICLE XXI. ARTICLE XXV. ARTICLE XXV.	SCOPE OF PLACEMENT SERVICES FOR AGREEMENTS COMPETITIVELY PROCURED FOR AGREEMENTS NOT COMPETITIVELY PROCURED EXHIBITS TERM OF AGREEMENT ORDER OF PRECEDENCE DEFINITIONS GOVERNING THIS AGREEMENT PROVIDER RESPONSIBILITIES AGENCY RESPONSIBILITIES INVOICING FOR PLACEMENT SERVICES REIMBURSEMENT FOR PLACEMENT SERVICES TERMINATION; BREACH AND DEFAULT RECORDS RETENTION, CONFIDENTIALITY AND DATA SECURITY REQUIREMENTS PROVIDER ASSURANCES AND CERTIFICATIONS INDEPENDENT CONTRACTOR AUDITS AND OTHER FINANCIAL MATTERS GRIEVANCE/DISPUTE RESOLUTION PROCESS ATTACHMENTS/ADDENDA NOTICE CONSTRUCTION NO ASSURANCES CONFLICT OF INTEREST INSURANCE INDEMNIFICATION AND HOLD HARMLESS SCREENING AND SELECTION PROHIBITION OF CORPORAL & DEGRADING PUNISHMENT FINDINGS FOR RECOVERY PUBLIC RECORDS CHILD SUPPORT ENFORCEMENT DECLARATION OF PROPERTY TAX DELINQUENCY SUBCONTRACTING AND DELEGATION PROPERTY OF AGENCY SEVERABILITY NO ADDITIONAL WAIVER IMPLIED COUNTERPARTS
ARTICLE XXXIII. ATTACHMENTS TO T	

.

RECITALS

WHEREAS, the Agency is responsible under Ohio Revised Code (ORC) Title 51, Chapter <u>5153</u> for the provision of protective services for dependent, neglected, and abused children; and,

WHEREAS, the Agency is authorized under ORC Title 51, Chapter <u>5153.16</u> to provide care and services which it deems to be in the best interest of any child who needs or is likely to need public care and services; and,

WHEREAS, the Provider is an organization duly organized and validly existing and is qualified to do business under the laws in the State of Ohio or in the state where the Provider of services is located and has all requisite legal power and authority to execute this Agreement and to carry out its terms, conditions and provisions; and is licensed, certified or approved to provide services to children and families in accordance with Ohio law or the state where the Provider of services is located.

NOW, THEREFORE, in consideration of the mutual promises and responsibilities set forth herein, the Agency and Provider agree as follows:

Article I. SCOPE OF PLACEMENT SERVICES

In addition to the services described in Exhibit I-Scope of Work, Provider agrees to provide and shall provide the placement and related services specified in each Individual Child Care Agreement (ICCA) for children in the care and custody of the Title IV-E Agency. The ICCA shall be consistent with current federal, state and local laws, rules and regulations applicable to the Provider's license or certified functions and services. If an Agreement and ICCA both exist, the Agreement supersedes.

See Attachment 1 for additional details.

Section 1.01 FOR AGREEMENTS COMPETITIVELY PROCURED

Without limiting the services set forth herein, Provider will provide the Services pursuant to and consistent with the Requests for Proposals (RFP) and the Provider's Proposal submitted in response to the RFP, the Provider agrees to provide and shall provide the placement and related services described in Exhibit I-Scope of Work.

Section 1.02 FOR AGREEMENTS NOT COMPETITIVELY PROCURED

The Provider agrees to provide and shall provide the placement and related services described in the Exhibit I- Scope of Work.

Section 1.03 EXHIBITS

The following exhibits are deemed to be a part of this Agreement as if fully set forth herein:

- A. Exhibit I Scope of Work;
- B. Exhibit II Request for Proposals (if applicable);
- C. Exhibit III Provider's Response to the Request for Proposals (if applicable); and
- D. Exhibit IV Schedule A Rate Information.

Article II. TERM OF AGREEMENT

This Agreement is in effect from **07/01/2023** through **06/30/2024**, unless this Agreement is suspended or terminated pursuant to Article VIII prior to the termination date.

In addition to the initial term described above, this Agreement may be extended, at the option of the Agency and upon written agreement of the Provider, for ______ additional, _____ year terms not to exceed _____ years. Notice of Agency's intention to extend the Agreement shall be provided in writing to Provider no less than 90 calendar days before the expiration of any Agreement term then in effect. (If a previous Request for Proposal [RFP] allows, the Agreement may be extended for a period of time to ensure adequate completion of the Agency's competitive procurement process at the rates existing for the term then in effect.)

07/01/2023 - 06/30/2024 Page 3 of 25

Article III. ORDER OF PRECEDENCE

This Agreement and all Exhibits are intended to supplement and complement each other and shall, where possible, be so interpreted. However, if any provision of this Agreement irreconcilably conflicts with an Exhibit, this Agreement takes precedence over the Exhibit(s).

In the event there is an inconsistency between the Exhibit(s), the inconsistency shall be resolved in the following order:

- A. Exhibit I: Scope of Work; then
- B. Exhibit II: Request for Proposals (if applicable); then
- C. Exhibit III: Provider's Proposals (if applicable); then
- D. Exhibit IV: Title IV-E Schedule A Rate Information.

Article IV. DEFINITIONS GOVERNING THIS AGREEMENT

The following definitions govern this Agreement:

- A. Agreement means this Agreement, attachments and exhibits thereto.
- B. Material Breach shall mean an act or omission that violates or contravenes an obligation required under the Agreement and which, by itself or together with one or more other breaches, has a negative effect on, or thwarts the purpose of the Agreement as stated herein. A Material Breach shall not include an act or omission, which has a trivial or negligible effect on the quality, quantity, or delivery of the goods and services to be provided under the Agreement.
- C. Child(ren) means any person under eighteen years of age or a mentally or physically handicapped person under twenty-one years of age in the Agency's custody and under the care of the Provider for the provision of placement services.
- D. All other definitions to be resolved through Federal Regulations, Ohio Administrative Code (OAC) 5101:2-1-01 and any related cross-references.

Article V. PROVIDER RESPONSIBILITIES

- A. Provider agrees to participate with Agency in the development and implementation of the Case Plan and ICCA including participation in case reviews and / or semi-annual administrative reviews, and the completion of reunification assessments for the children in placement with the Provider. Parties shall make best efforts to share information timely regarding participants and contact information involved with planning efforts related to children and families.
- B. Provider agrees to provide services agreed to in the Case Plan and ICCA (i.e.,transportation of children for routine services, including, but not limited to, court hearings, medical appointments, school therapy, recreational activities, visitations/family visits) unless otherwise negotiated in writing as an attachment to this Agreement. Any disputes involving services or placement will be resolved through mutual-agreement and modification to the ICCA. Provider agrees the Agency is the final authority in the process. The cost of providing these services is to be included in the Agency approved per diem.
- C. Provider agrees to ensure that any and all persons who may act as alternative caregivers or who have contact with the children are suitable for interaction pursuant to all applicable federal, state and local laws and regulations.
- D. Provider agrees that all caregivers must be approved by the Agency.
- E. Provider agrees to submit a progress report as negotiated by the parties for each child. The progress report will be based on the agreed upon services to be delivered to the child and/or family and will include documentation of services provided to the child and/or discharge summary. If Monthly Progress Reports are not received within 90 calendar days following the month of service provision, payment may be withheld at the Agency's discretion.

- 1. Monthly Progress Reports shall be submitted by the 20th of the month following the month of service.
- 2. The Monthly Progress Report will include the following medical related information:
 - a. Service type (i.e. medical, dental, vision, etc.);
 - b. Date(s) of service;
 - c. Reason for visit (i.e. routine, injury, etc.);
 - d. Practitioner name, address and contact number;
 - e. Name of hospital, practice, urgent care, etc.;
 - f. Prescribed medications and dosages;
 - g. Date(s) medication(s) were prescribed or changed; and
 - h. Changes to medications.
- F. Placement changes, emergency or non-emergency, shall occur only with the approval of the Agency. The following information shall be provided to the Agency for all placement changes: Name, address and phone number of the new foster home or other out-of-home care setting, the license/home study of the new care provider within 24 hours, excluding weekends and holidays.
- G. Provider agrees to notify all Agencies who have children placed in the same caregiver's home/group home/CRC when any child residing in the placement is critically injured or dies in that location. Notification will be made to the Agencies' Child Abuse/Neglect Hotline number or assigned Caseworker immediately.
- H. Notification to the Agency of Emergency Critical Incidents shall occur ASAP but no later than one hour of the Incident becoming known. Notification will be made to the Agency via the Agency's Child Abuse/Neglect Hotline or assigned Caseworker or by other established system. Critical incidents are those incidents defined in the Ohio Administrative Code that are applicable to the licensed or certified programs (ODJFS 5101:2-7-14, 5101:2-9-23 ODMHAS 5122-30-16, 5122-26-13, OAC 5123-17-02).

Emergency situations include but are not limited to the following:

- 1. Absent Without Leave (AWOL);
- 2. Child Alleging Physical or Sexual Abuse / Neglect;
- Death of Child;
- 4. Illicit drug/alcohol use; Abuse of medication or toxic substance;
- 5. Sudden injury or illness requiring an unplanned medical treatment or visit to the hospital;
- 6. Perpetrator of Delinquent/Criminal Act (Assault, Dangerous Behaviors, Homicidal Behaviors);
- 7. School Expulsion / Suspension (formal action by school);
- 8. Self-Injury (Suicidal Behaviors, Self-Harm Requiring external Medical Treatment, Hospital or ER);
- 9. Victim of assault, neglect, physical or sexual abuse; and
- 10. The filing of any law enforcement report involving the child.
- I. The Provider also agrees to notify the Agency within Twenty-four (24) hours, of any non-emergency situations. Non-emergency situations include but are not limited to the following:
 - 1. When physical restraint is used/applied; and
 - 2. Medication lapses or errors.

Notification will be made to the Agency via the Agency's Child Abuse Neglect Hotline / assigned Caseworker or by other established notification system.

- J. Documentation of the emergency and non-emergency incidents as identified in "H and I" above shall be provided to the Agency via email, fax or other established notification system within 24 hours excluding weekends and holidays.
- K. The Provider agrees to submit each child's assessment and treatment plans as completed but no later than the 30th day of placement. Provider further agrees to provide treatment planning that will include, but is not limited to, education on or off site, preparation for integration into community-based school or vocational/job skills training, community service activities, independent living skills if age 14 or older, monitoring and supporting community adjustment.
- L. The Provider agrees to participate in joint planning with the Agency regarding modification to case plan services. Provider agrees that while the Provider may have input into the development of the child's case plan services and

the ICCA, any disputes involving services or placement will be resolved through mutual agreement and modification to the ICCA. Provider agrees the Agency is the final authority in the process.

- M. The Provider shall participate in a Placement Preservation meeting if requested by the Agency prior to issuing a notice of removal of a child. A placement Preservation meeting shall be held within seven (7) business days of said request. Unless otherwise mutually agreed upon a minimum of thirty (30) calendar days' notice shall be given if placement preservation is unable to be achieved. A Discharge Plan Summary shall be provided no later than fifteen (15) calendar days after the date of discharge in accordance with the applicable licensed or certified program. (OAC 5101:2-5-17, OAC 5122-30-22, OAC 5122-30-04, OAC 5123:2-3-05).
- N. The Provider shall work in cooperation and collaboration with the Agency to provide information for each child's Lifebook and will fully comply with the provision of <u>OAC 5101:2-42-67</u> as applicable to private Providers. Provider's contribution to the Agency Lifebook for a child shall be for the episode of care with the Provider.
- O. The Provider agrees to provide Independent Living Services as set forth in accordance with OAC 5101:2-42-19 for all children age 14 and above.
- P. When applicable, due to the Provider being part of a managed care agreement as defined in <u>OAC 5101:2-1-01</u>, the Provider agrees to visit with the child face-to-face in the foster home, speak privately with the child and to meet with the caregiver at least monthly in accordance with rule <u>OAC 5101:2-42-65</u> of the Ohio Administrative Code.
- Q. The Provider agrees to maintain its licenses and certifications from any source in good standing. The Provider agrees to report to Agency in writing any change in licensure or certification that negatively impacts such standing immediately if the negative action results in a temporary license, suspension of license or termination of license.
- R. Provider agrees that the reasonable and prudent parent standard training required by SEC. 471. [42 U.S.C. 671] of the Social Security Act and in accordance to <u>OAC 5101:2-5-33</u>, <u>OAC 5101:2-9-02</u> or <u>OAC 5101:2-9-03</u> has been completed.
- S. The Provider shall notify Agency of any changes in its status, such as intent to merge with another business or to close no later than forty-five (45) business days prior to the occurrence.
- T. The Provider agrees that the Agency shall have access to foster parent home studies and re-certifications for foster parents caring for children in placement, subject to confidentiality considerations. The Provider shall submit to Agency a copy of the current foster home license at the time of placement and recertification. Provider also agrees to notify Agency within twenty-four (24) hours of any change in the status of the foster home license.
- U. When there is a rule violation of a caregiver, a copy of the corrective action plan, if applicable, must be submitted to the Agency when the investigation is complete.
- V. The Provider agrees to notify the Agency of scheduling no less than fourteen (14) calendar days prior to all formal meetings (i.e. FTMs, Treatment Team Meetings, IEPs, etc.).
- W. The Provider agrees to adhere to the following Medical/Medication guidelines:
 - 1. To provide over-the-counter medications and/or supplies as part of the per diem of care;
 - 2. To comply with the medical consent process as identified by Agency;
 - 3. Only the Agency can give permission for the administering or change (addition or elimination) of psychotropic medication and its ongoing management; and
 - 4. Provide an initial placement medical screening within 72 hours of child's placement into a placement resource under the Provider's operation and/or oversight.
- X. To arrange for required health care/medical examinations within time frames required by <u>OAC 5101:2-42-66.1</u> and provide reports from the health care providers to the agency within 30 days of occurrence if the appropriate releases of information have been obtained by the Provider.
- Y. The Network Provider agrees to notify the Agency if placement resource is currently under investigation for license violations or misconduct toward children or other third-party investigation.
- Z. The Provider will immediately notify the Agency:

- 1. If the Provider is out of compliance with any licensing authority rules or the placement resource is under investigation for license violations or misconduct toward children. Immediately is defined as within one hour of knowledge of the non-compliance issue.
- 2. Child Abuse/Neglect Hotline or assigned Caseworker of any allegations of abuse or neglect made against the Caregiver within one hour of gaining knowledge of the allegation.
- 3. Of any corrective action and the result of the correction action plan. The Provider will submit a comprehensive written report to the agency within sixty (60) days of the rules violation.
- 4. Within twenty-four (24) hours any time there is an event which would impact the placement resource license.

See Attachment 2 for additional details.

Article VI. AGENCY RESPONSIBILITIES

- A. Agency certifies that it will comply with the Multiethnic Placement Act, 108 STAT. 3518, as amended by Section 1808 of the Small Business Jobs Protection Act of 1996, 110 STAT. 1755, which prohibits any Agency from denying any person the opportunity to become an adoptive or foster parent on the basis of race, color, national origin, or delaying or denying the placement of a child for adoption or into foster care on the basis of race, color, or national origin of the adoptive or foster parent or of the child involved.
- B. The Agency shall provide to the Provider within thirty (30) calendar days of placement or within a reasonable time thereafter as agreed to by the parties, a copy of each child's social history, medical history, and Medicaid card once obtained by the Agency for new cases, or at time of placement for existing cases. Agency shall make best efforts to share information timely regarding participants and contact information involved with planning efforts related to children and families.
- C. The Agency acknowledges that clinical treatment decisions must be recommended by licensed clinical professionals. Agency and Provider acknowledge that disagreement with a treatment decision may be taken through the dispute resolution process contained in Article XIV of this Agreement.
- D. Agency agrees to visit with the child in accordance with rule <u>OAC 5101:2-42-65</u> of the Ohio Administrative Code.
- E. Agency agrees to participate in periodic meetings with each child's treatment team for case treatment plan development, review, and revision. The Agency agrees to participate in the development of the treatment plan of each child placed with the Provider by the Agency.
- F. Agency certifies that it will comply with Every Student Succeeds Act (34 CFR part 200) and will work with local school districts in developing individualized plans to address the transportation needed for a child to remain in the school of origin. Agency agrees to arrange for the transfer of each child's school records to the child's new school upon placement but not later than ten (10) business days. The Agency agrees to work with the Provider for the timely enrollment of the child in the receiving school district. The Agency has the final responsibility to obtain the child's school records and to enroll the child in the receiving school district.
- G. The Agency shall provide an opportunity for the Provider to give input in the development, substantive Addendum or modification of case plans. The Agency agrees to notify the Provider of scheduling no less than seven (7) calendar days prior to of all formal meetings (e.g. SARs, court hearings, family team conferences, etc.).
- H. The Agency shall participate in a Placement Preservation meeting if requested by the Provider prior to issuing a notice of removal of a child. The Agency shall provide a minimum of thirty (30) calendar days' notice for planned removals, to the Provider for each child who is being terminated from placement with the Provider, unless so ordered by a court of competent jurisdiction.
- I. Agency agrees to provide the Provider with an emergency contact on a twenty-four (24) hour, seven (7) day per week basis.
- J. The Agency represents:
 - 1. It has adequate funds to meet its obligations under this Agreement; subject to the availability of funds as referenced in Article VIII (I);
 - 2. It intends to maintain this Agreement for the full period set forth herein and has no reason to believe that it

Contract ID: 19346132

Fairfield County Department of Job and Family Services / A New Leaf, Inc.

07/01/2023 - 06/30/2024 Page 7 of 25 will not have sufficient funds to enable it to make all payments due hereunder during such period; and

- 3. It will make its best effort to obtain the appropriation of any necessary funds during the term of this Agreement.
- K. The Agency will provide information about the child being referred for placement in accordance with <u>OAC</u> <u>5101:2-42-90</u>.Prior to a child's placement in alternative care or respite, <u>OAC 5101:2-42-90 (D)</u> requires the Agency to share with care givers information that could impact the health, safety, or well-being of the child or others in the home.

Article VII. INVOICING FOR PLACEMENT SERVICES

- A. The Provider agrees to submit a monthly invoice following the end of the month in which services were provided. The invoice shall be for services delivered in accordance with Article I of this Agreement and shall include:
 - 1. Provider's name, address, telephone number, fax number, federal tax identification number, Title IV-E Provider number, if applicable and Medicaid Provider number, if applicable.
 - 2. Billing date and the billing period.
 - 3. Name of child, date of birth of child, and the child's Statewide Automated Child Welfare Information System (SACWIS) person I.D. number.
 - 4. Admission date and discharge date, if available.
 - 5. Agreed upon per diem for maintenance and the agreed per diem administration; and
 - 6. Invoicing procedures may also include the per diems associated with the following if applicable and agreeable to the Agency and Provider:
 - a. Case Management; allowable administration cost;
 - b. Transportation, allowable maintenance cost;
 - c. Transportation; allowable administration cost;
 - d. Other Direct Services; allowable maintenance cost;
 - e. Behavioral health care; non-reimbursable cost; and
 - f. Other costs (any other cost the Title IV-E Agency has agreed to participate in); non-allowable/ non-reimbursable cost.
- B. Provider warrants and represents claims made for payment for services provided are for actual services rendered and do not duplicate claims made by Provider to other sources of public funds for the same service.

Article VIII. REIMBURSEMENT FOR PLACEMENT SERVICES

- A. The maximum amount payable pursuant to this contract is **\$500,000.00**.
- B. In accordance with Schedule A of this Agreement, the per diem for maintenance and the per diem for administration will be paid for each day the child was in placement. The first day of placement will be paid regardless of the time the child was placed. The last day of placement will not be paid regardless of the time the child left the placement.
- C. In accordance with Schedule A of this Agreement and in addition to Maintenance and Administration, the Agency may agree to pay a per diem for Case Management, Other Direct Services, Transportation Administration, Transportation Maintenance, Behavioral Health Care and Other. All other services and/or fees to be paid for shall be contained in the Attachments/Exhibits of this Agreement.
- D. To the extent that the Provider maintains a foster care network, the agreed upon per diem for maintenance shall be the amount paid directly to the foster parent. Maintenance includes the provision of food, clothing, shelter, daily supervision, graduation expenses, a child's personal incidentals, and liability insurance with respect to the child, reasonable cost of travel to the child's home for visitation and reasonable cost of travel for the child to remain in the school the child was enrolled in at the time of placement. Payment for private Agency staff transporting a child to a home visit or keeping the child in their home school will be paid in accordance with Schedule A (Transportation Maintenance) of this Agreement.
- E. If the plan as determined by the Agency is to return the child to placement with the Provider, the Agency may

agree to pay for the days that a child is temporarily absent from the direct care of the Provider, as agreed to by the parties in writing.

- F. The service provider is required to utilize Medicaid-approved healthcare providers in the appropriate managed care network for the provision of mental health, dental and/or medical services (hereafter referred to collectively as "medical services") to children in the custody of Agency. The Service Provider will report applicable Medicaid/insurance information to the healthcare providers and instruct healthcare providers to seek payment from Medicaid or any other available third-party payer for medical services rendered to children in agency custody. Agency will not pay for the provision of any medical services to children in agency custody unless the agency Executive Director or authorized designee has provided specific prior written authorization for such medical services and associated costs.
- G. The Agency agrees to pay the Provider for all services agreed to on Schedule A and in the Attachments/Exhibits to this Agreement, where applicable, that have been provided and documented in the child's case file. Agency shall make best efforts to make payment of undisputed charges within thirty (30) business days of receipt.
- H. In the event of a disagreement regarding payment, Agency shall withhold payment only for that portion of the placement with which it disagrees. Agency will use best efforts to notify the Provider of any invoice discrepancies. Agency and Provider will make every effort to resolve payment discrepancies within 60 calendar days. Payment discrepancies brought to the Agency after 60 days will be reviewed on a case by case basis.
- I. This Agreement is conditioned upon the availability of federal, state, or local funds appropriated or allocated for payment for services provided under the terms and conditions of this Agreement. By sole determination of the Agency, if funds are not sufficiently allocated or available for the provision of the services performed by the Provider hereunder, the Agency reserves the right to exercise one of the following alternatives:
 - 1. Reduce the utilization of the services provided under this Agreement, without change to the terms and conditions of the Agreement; or
 - 2. Issue a notice of intent to terminate the Agreement.

The Agency will notify the Provider at the earliest possible time of such decision. No penalty shall accrue to the Agency in the event either of these provisions is exercised. The Agency shall not be obligated or liable for any future payments due or for any damages as a result of termination under this section.

Any denial of payment for service(s) rendered may be appealed in writing and will be part of the dispute resolution process contained in Article XIV.

See Attachment 3 for additional details.

Article IX. TERMINATION; BREACH AND DEFAULT

- A. This Agreement may be terminated for convenience prior to the expiration of the term then in effect by either the Agency or the Provider upon written notification given no less than sixty (60) calendar days in advance by certified mail, return receipt requested, to the last known address of the terminated party shown hereinabove or at such other address as may hereinafter be specified in writing.
- B. If Provider fails to provide the Services as provided in this Agreement for any reason other than Force Majeure, or if Provider otherwise Materially Breaches this Agreement, Agency may consider Provider in default. Agency agrees to give Provider thirty (30) days written notice specifying the nature of the default and its intention to terminate. Provider shall have seven (7) calendar days from receipt of such notice to provide a written plan of action to Agency to cure such default. Agency is required to approve or disapprove such plan within five (5) calendar days of receipt. In the event Provider fails to submit such plan or Agency disapproves such plan, Agency has the option to immediately terminate this Agreement upon written notice to Provider. If Provider fails to cure the default in accordance with an approved plan, then Agency may terminate this Agreement at the end of the thirty (30) day notice period.
- C. Upon of the effective date of the termination, the Provider agrees that it shall cease work on the terminated activities under this Agreement, take all necessary or appropriate steps to limit disbursements and minimize

costs, and furnish a report as of the date of discharge of the last child describing the status of all work under this Agreement, including without limitation, results accomplished, conclusions resulting therefrom, and such other matters as the Agency may require. The Agency agrees to remove all children in placement immediately with the Provider, consistent with the effective termination date. In all instances of termination, the Provider and Agency agree that they shall work in the best interests of children placed with the Provider to secure alternative placements for all children affected by the termination.

- D. In the event of termination, the Provider shall be entitled to reimbursement, upon submission of an invoice, for the agreed upon per diem incurred prior to the effective termination date. The reimbursement will be calculated by the Agency based on the per diem set forth in Article VIII. The Agency shall receive credit for reimbursement already made when determining the amount owed to the Provider. The Agency is not liable for costs incurred by the Provider after the effective termination date of the discharge of the last child.
- E. Notwithstanding the above, Agency may immediately terminate this Agreement upon delivery of a written notice of termination to the Provider under the following circumstances:
 - 1. Improper or inappropriate activities;
 - 2. Loss of required licenses;
 - 3. Actions, inactions or behaviors that may result in harm, injury or neglect of a child;
 - 4. Unethical business practices or procedures; and
 - 5. Any other event that Agency deems harmful to the well-being of a child; or
 - 6. Loss of funding as set forth in Article VIII.
- F. If the Agreement is terminated by Agency due to breach or default of any of the provisions, obligations, or duties embodied contained therein by the Provider, Agency may exercise any administrative, agreement, equitable, or legal remedies available, without limitation. Any extension of the time periods set forth above shall not be construed as a waiver of any rights or remedies the Agency may have under this Agreement.
- G. In the event of termination under this ARTICLE, both the Provider and the placing Agency shall make good faith efforts to minimize adverse effect on children resulting from the termination of the Agreement.

Article X. RECORDS RETENTION, CONFIDENTIALITY AND DATA SECURITY REQUIREMENTS

- A. The Provider agrees that all records, documents, writings or other information, including, but not limited to, financial records, census records, client records and documentation of legal compliance with Ohio Administrative Code rules, produced by the Provider under this Agreement, and all records, documents, writings or other information, including but not limited to financial, census and client used by the Provider in the performance of this Agreement are treated according to the following terms:
 - 1. All records relating to costs, work performed and supporting documentation for invoices submitted to the Agency by the Provider along with copies of all Deliverables, as defined in Article XXIX, submitted to the Agency pursuant to this Agreement will be retained for a minimum of three (3) years after reimbursement for services rendered under this Agreement.
 - 2. If an audit, litigation, or other action is initiated during the time period of the Agreement, the Provider shall retain such records until the action is concluded and all issues resolved or three (3) years have expired, whichever is later.
 - 3. All records referred to in Section A 1) of this Article shall be available for inspection and audit by the Agency or other relevant agents of the State of Ohio (including, but not limited to, the County Prosecutor, the Ohio Department of Job and Family Services (ODJFS), the Auditor of the State of Ohio, the Inspector General of Ohio, or any duly authorized law enforcement officials), and the United States Department of Health and Human Services within a reasonable period of time.
- B. The Provider agrees to keep all financial records in a manner consistent with Generally Accepted Accounting Principles.
- C. The Provider agrees to comply with all federal and state laws applicable to the Agency and the confidentiality of children and families. Provider understands access to the identities of any Agency's child and families shall only be as necessary for the purpose of performing its responsibilities under this Agreement. No identifying

information on child(ren) served will be released for research or other publication without the express written consent of the Agency. Provider agrees that the use or disclosure of information concerning the child for any purpose not directly related to the administration of this Agreement is prohibited. Provider shall ensure all the children's and families' documentation is protected and maintained in a secure and safe manner.

- D. The Provider agrees to comply with all applicable state and federal laws related to the confidentiality and transmission of medical records, including, but not limited to the Health Insurance Portability and Accountability Act of 1996 (HIPAA).
- E. Although information about, and generated under, this Agreement may fall within the public domain, the Provider shall not release information about, or related to, this Agreement to the general public or media verbally, in writing, or by any electronic means without prior approval from the Agency, unless the Provider is required to release requested information by law. Agency reserves the right to announce to the general public and media: award of the Agreement, Agreement terms and conditions, scope of work under the Agreement, Deliverables, as defined in Article XXIX, and results obtained under the Agreement. Except where Agency approval has been granted in advance, the Provider shall not seek to publicize and will not respond to unsolicited media queries requesting: announcement of Agreement award, Agreement terms and conditions, Agreement scope of work, government-furnished documents the Agency may provide to the Provider to fulfill the Agreement scope of work, Deliverables required under the Agreement, results obtained under the Agreement, and impact of Agreement activities.
- F. If contacted by the media about this Agreement, the Provider agrees to notify the Agency in lieu of responding immediately to media queries. Nothing in this section is meant to restrict the Provider from using Agreement information and results to market to specific business prospects.
- G. Client data must be protected and maintained in a secure and safe manner whether located in Provider's facilities, stored in the Cloud, or used on mobile devices outside Provider's facility. Security of Provider's network, data storage, and mobile devices must conform to generally recognized industry standards and best practices. Maintenance of a secure processing environment includes, but is not limited to, network firewall provisioning, intrusion detection, antivirus protection, regular third-party vulnerability assessments, and the timely application of patches, fixes and updates to operating systems and applications.
- H. Provider agrees that it has implemented and shall maintain during the term of this Agreement the highest standard of administrative, technical, and physical safeguards and controls to:
 - 1. Ensure the security and confidentiality of data;
 - 2. Protect against any anticipated security threats or hazards to the security or integrity of data; and
 - 3. Protect against unauthorized access to or use of data. Such measures shall include at a minimum:
 - a. Access controls on information systems, including controls to authenticate and permit access to data only to authorized individuals and controls to prevent Provider employees from providing data to unauthorized individuals who may seek to obtain this information (whether through fraudulent means or otherwise);
 - b. Firewall protection;
 - c. Encryption of electronic data while in transit from Provider networks to external networks;
 - d. Measures to store in a secure fashion all data which shall include multiple levels of authentication;
 - e. Measures to ensure that data shall not be altered or corrupted without the prior written consent of the Agency;
 - f. Measures to protect against destruction, loss or damage of data due to potential environmental hazards, such as fire and water damage.
- I. Immediately upon discovery of a confirmed or suspected breach involving data, Provider will notify Agency no later than twenty-four (24) hours after Provider knows or reasonably suspects a breach has or may have occurred. Provider shall promptly take all appropriate or legally required corrective actions and shall cooperate fully with the Agency in all reasonable and lawful efforts to prevent, mitigate or rectify such data breach. In the event of a suspected breach, Provider shall keep the Agency informed of the progress of its investigation until the uncertainty is resolved.
- J. In the event the Provider does not carry the appropriate cyber security insurance to cover a security breach, the

Provider shall reimburse the Agency for actual costs incurred, including, but not limited to, providing clients affected by a security breach with notice of the breach, and/or complimentary access for credit monitoring services, which the Agency deems necessary to protect such affected client.

K. In the event the Agency discontinues operation, all child records for residential or any other placement settings shall be provided to the custodial agency. If the setting is licensed by ODJFS, licensing records shall be sent to:

ODJFS ATTN: Licensing P.O. Box 183204 Columbus, OH 43218-3204

Article XI. PROVIDER ASSURANCES AND CERTIFICATIONS

- A. As applicable to the Provider's license and/or certification, the Provider certifies compliance with <u>ORC 2151.86</u>, <u>ORC 5103.0328</u>, <u>ORC 5103.0319</u> and applicable OAC Sections as defined in Article XXII of this Agreement concerning criminal record checks, arrests, convictions and guilty pleas relative to foster caregivers, employees, volunteers and interns who are involved in the care for a child. Provider is responsible for any penalties, financial or otherwise, that may accrue because of noncompliance with this provision.
- B. To the extent that the Provider maintains a residential center or group home, the Provider agrees to comply with the provisions of their licensing Agency that relates to the operation, safety and maintenance of residential facilities. Specifically, Provider agrees that no firearm or other projectile weapon and no ammunition for such weapons will be kept on the premises.
- C. Provider certifies compliance with Drug Free Work Place Requirements as outlined in 45 C.F.R. Part 76,Subpart F.
- D. Provider certifies compliance with 45 C.F.R. Part 80, Non-Discrimination under programs receiving Federal assistance through the Department of Health and Human Services effectuation of Title VI of the Civil Rights Act of 1964.
- E. Provider certifies compliance with 45 C.F.R. Part 84, Non-Discrimination on the Basis of Handicap in Programs or Activities Receiving Federal Assistance.
- F. Provider certifies compliance 45 C.F.R. Part 90, Non-Discrimination on the Basis of Age in Programs or Activities Receiving Federal Assistance.
- G. Provider certifies compliance with the American with Disabilities Act, Public Law 101-336.
- H. Provider certifies that it will:
 - 1. Provide a copy of its license(s), certification, accreditation or a letter extending an expiring license, certification, or accreditation from the issuer to the Agency prior to the signing of the Agreement.
 - 2. Maintain its license(s), certification, accreditation and that upon receipt of the renewal of its license, certification, and/or accreditation or upon receipt of a letter extending an expiring license, certification, and/or accreditation from the issuer, a copy of the license, certification and/or accreditation will be provided to the Agency within five (5) business days.
 - 3. Provider shall immediately notify the Agency of any action, modification or issue relating to said licensure, accreditation or certification.
- I. Provider certifies that it will not deny or delay services to eligible persons because of the person's race, color, religion, national origin, gender, orientation, disability, or age.
- J. The Provider shall comply with Executive Order 11246, entitled Equal Employment Opportunity, as amended by Executive Order 11375, and as supplemented in Department of Labor regulation 41 CFR part 60.
- K. Provider further agrees to comply with <u>OAC 5101:9-2-01</u> and <u>OAC 5101:9-2-05(A)(4)</u>, as applicable, which require that assure that persons with limited English proficiency (LEP) can meaningfully access services. To the

extent Provider provides assistance to an LEP Child through the use of an oral or written translator or interpretation services in compliance with this requirement, the LEP Child shall not be required to pay for such assistance.

- L. To the extent applicable, the Provider certifies compliance with all applicable standards, orders, or requirements issued under Section 306 of the Clean Air Act (42 U.S.C. 1857 (h) Section 508 of the Clean Water Act (33 U.S.C. 1368), Executive Order 11738, and Environmental Protection Agency Regulations (40 C.F.R. Part 15).
- M. The Provider certifies compliance, where applicable, with mandatory standards and policies relating to energy efficiency which are contained in the state energy conservation plan issued in compliance with the Energy Policy and Conservation Act (Pub. L. 94-163, 89 Stat. 871).
- N. The Provider certifies that all approvals, licenses, or other qualifications necessary to conduct business in Ohio have been obtained and are current.
- O. Provider shall comply with the Small Business Job Protection Act (Public Law ("P.L.") 104-188), the Multiethnic Placement Act of 1994 (P.L. 103-382), Titles IV-B (42 U.S.C. 620 et seq.) and IV-E (42 U.S.C. 670 et seq.) of the Social Security Act ("the Act"), the Personal Responsibility and Work Opportunity Reconciliation Act of 1996 (P.L. 104-193), Section 471(a) of Title IV-E of the Act (42 U.S.C. 671(a)), and 45 C.F.R. 1356, including all rules, regulations and guidelines issued by federal and state authorities, <u>OAC 5101:9-4-07</u> and <u>OAC 5101:2-47-23.1</u>.

Article XII. INDEPENDENT CONTRACTOR

- A. The Provider and the Agency agree that no employment, joint venture, or partnership has been or will be created between the parties hereto pursuant to the terms and conditions of this Agreement.
- B. The Provider and the Agency agree that the Provider is an independent contractor and assumes all responsibility for any federal, state, municipal, or other tax liabilities along with workers' compensation, unemployment compensation, and insurance premiums which may accrue as a result of compensation received for services or Deliverables rendered hereunder.
- C. The Provider and the Agency agree that no person and/or entities entering into this Agreement, nor any individual employed by any person or entity entering in to this Agreement, are public employees for purposes of contributions to Ohio Public Employees Retirement system by virtue of any work performed or services rendered in accordance with this Agreement.

Article XIII. AUDITS AND OTHER FINANCIAL MATTERS

- A. Provider agrees to submit to Agency a copy of the independent audit it receives in accordance with <u>ORC</u> <u>5103.0323.</u>
- B. Upon request from the Agency, Provider shall submit a copy of the most recent Federal income tax return and related schedules filed with the Internal Revenue Service (IRS).
- C. If Provider participates in the Title IV-E program, Provider agrees to timely file its Title IV-E cost report with all required items as outlined in <u>OAC 5101:2-47-26.2</u> to ODJFS. Provider agrees that in the event a cost report cannot be timely filed, an extension shall be requested prior to the December 31st filing deadline.
- D. If a Provider participates in the Title IV-E program, an Agreed Upon Procedures engagement must be conducted by a certified public accountant for the Provider's cost report in accordance with <u>OAC 5101:2-47-26.2</u>. The procedures are conducted to verify the accuracy of costs used to establish reimbursement ceilings for maintenance and administration costs of child in care. Any overpayments or underpayment of federal funds to the Title IV-E Agency due to adjustments of cost report reimbursement ceiling amounts as a result of an audit, shall be resolved in accordance with <u>ORC 5101.11</u>, <u>ORC 5101.14</u>, and <u>OAC 5101:2-47-01</u>.
- E. Upon request from the Agency, the Provider shall submit a copy of the JFS 02911 and Agreed Upon Procedures.
- F. For financial reporting purposes and for Title IV-E cost reporting purposes, Provider agrees to follow the cost

principles set forth in the following OAC Sections and publications:

- 1. <u>OAC 5101:2-47-11</u>: "Reimbursement for Title IV-E foster care maintenance (FCM) costs for children's residential centers (CRC), group homes, maternity homes, residential parenting facilities, private foster homes, and substance use disorder (SUD) residential facilities".
- <u>OAC 5101:2-47-26.1</u>: "Public child services agencies (PCSA), private child placing agencies (PCPA), private noncustodial agencies (PNA), residential care facilities, substance use disorder (SUD) residential facilities: Title IV-E cost report filing requirements, record retention requirements, and related party disclosure requirements";
- 3. OAC 5101:2-47-26.2: "Cost Report Agreed Upon Procedures Engagement".
- 4. JFS 02911 Single Cost Report Instructions.
- 5. For Private Agencies: 2 CFR part 230, Cost Principles for Non-Profit Organizations.
- 6. For Public Agencies: 2 CFR part 225, Cost Principles for State, Local and Indian Tribal Government.
- 7. 2 CFR part 200.501, Audit Requirements.

Article XIV. GRIEVANCE/DISPUTE RESOLUTION PROCESS

In the event that a dispute arises under the provisions of this Agreement, the parties shall follow the procedures set forth below:

- 1. The party complaining of a dispute shall provide written notice of the nature of the dispute to the other party to this Agreement. A copy of the notice shall be sent to the Director or designee of the Agency and to the Executive Director or designee of the Provider. Within ten (10) business days of receiving the notice of a dispute, the parties involved in the dispute between the Agency and the Provider shall attempt to resolve the dispute.
- 2. If the parties are unable to resolve the dispute in (1 business day), the highest official or designee of the Agency shall make the final determination within twenty (20) business days, which will be non-binding.
- 3. Neither party will be deemed to have waived any other rights or remedies available to them by initiating, participating in or completing this process.

Article XV. ATTACHMENTS/ADDENDA

This Agreement, Attachments, and all Exhibits hereto constitutes the entire Agreement and may be amended only with a written Addendum signed by both parties; however, it is agreed by the parties that any Addenda to laws or regulations cited herein will result in the correlative modification of this Agreement, without the necessity for executing written Addenda. The impact of any applicable law, statute, or regulation not cited herein and enacted after the date of execution of this Agreement by written Addendum signed by both parties and effective as of the date of enactment of the law, statute, or regulation. Any other written Addendum to this Agreement is prospective in nature.

Article XVI. NOTICE

Unless otherwise set forth herein, all notices, requests, demands and other communications pertaining to this Agreement shall be in writing and shall be deemed to have been duly given if delivered or mailed by certified or registered mail, postage pre-paid:

if to Agency, to	Fairfield County Department of Job and Family Services 239 W Main St Lancaster, OH 43130
if to Provider, to	A New Leaf, Inc. 221 Tarlton Rd Circleville, OH 43113

Article XVII. CONSTRUCTION

This Agreement shall be governed, construed, and enforced in accordance with the laws of the State of Ohio. Should

Contract ID: 19346132
Fairfield County Department of Job and Family Services / A New Leaf, Inc.

07/01/2023 - 06/30/2024 Page 14 of 25 any portion of this Agreement be found to be unenforceable by operation of statute or by administrative or judicial decision, the operation of the balance of this Agreement is not affected thereby; provided, however, the absence of the illegal provision does not render the performance of the remainder of the Agreement impossible.

Article XVIII. NO ASSURANCES

- A. Provider acknowledges that, by entering into this Agreement, Agency is not making any guarantees or other assurances as to the extent, if any, that Agency shall utilize Provider's services or purchase its goods. In this same regard, this Agreement in no way precludes, prevents, or restricts Provider from obtaining and working under additional arrangement(s) with other parties, assuming the work in no way impedes Provider's ability to perform the services required under this Agreement. Provider warrants that at the time of entering into this Agreement, it has no interest in nor shall it acquire any interest, direct or indirect, in any Agreement that will impede its ability to provide the goods or perform the services under this Agreement.
- B. This Agreement, Attachments, and all Exhibits embodies the entire agreement of the Parties. There are no promises, terms, conditions or obligations other than those contained herein; and this Agreement shall supersede all previous communications, representations or Agreements, either written or oral, between the parties to this Agreement. Also, this Agreement shall not be modified in any manner except by an instrument, in writing, executed by both the parties.

Article XIX. CONFLICT OF INTEREST

- A. Provider agrees that the Provider, its officers, members and employees currently have no, nor will they acquire any interest, whether personal, professional, direct or indirect, which is incompatible, in conflict with or which would compromise the discharge and fulfillment of Provider's functions, duties and responsibilities hereunder. If the Provider, or any of its officers, members or employees acquire any incompatible, conflicting, or compromising personal or professional interest, the Provider shall immediately disclose, in writing, such interest to the Agency. If any such conflict of interest develops, the Provider agrees that the person with the incompatible, conflicting, or compromising personal or professional interest will not participate in any activities related to this Agreement.
- B. Provider agrees: (1) to refrain from promising or giving to Agency employees anything of value to manifest improper influence upon the employee; (2) to refrain from conflicts of interest; and, (3) to certify that Provider complies with <u>ORC 102.03</u>, <u>ORC 102.04</u>, <u>ORC 2921.42</u>, <u>ORC 2921.43</u>.
- C. The Provider further agrees that there is no financial interest involved on the part of the Agency or the respective county authority(ies) governing the agency. The Provider has no knowledge of any situation which would be a conflict of interest. It is understood that a conflict of interest occurs when an Agency employee or county official will gain financially or receive personal favors as a result of signing or implementation of this agreement. The Provider will report the discovery of any potential conflict of interest to the Agency. Should a conflict of interest be discovered during the term of this agreement, the Agency may exercise any right under the agreement, including termination of the agreement.

Article XX. INSURANCE

The Provider shall purchase and maintain for the term of this Agreement insurance of the types and amounts identified herein. Maintenance of the proper insurance for the duration of the Agreement is a material element of the Agreement.

Provider agrees to procure and maintain for the term of this Agreement the insurance set forth herein. The cost of all insurance shall be borne by Provider. Insurance shall be purchased from a company licensed to provide insurance in Ohio. Insurance is to be placed with an insurer provided an A.M. Best rating of no less than A-. Provider shall purchase the following coverage and minimum limits:

A. Commercial general liability insurance policy with coverage contained in the most current Insurance Services Office Occurrence Form CG 00 01 or equivalent with limits of at least One Million Dollars (\$1,000,000.00) per occurrence and One Million Dollars (\$1,000,000.00) in the aggregate and at least One Hundred Thousand Dollars (\$100,000.00) coverage in legal liability fire damage. Coverage will include:

07/01/2023 - 06/30/2024 Page 15 of 25

- 1. Additional insured endorsement;
- 2. Product liability;
- 3. Blanket contractual liability;
- 4. Broad form property damage;
- 5. Severability of interests;
- 6. Personal injury; and
- 7. Joint venture as named insured (if applicable).

Endorsements for physical abuse claims and for sexual molestation claims must be a minimum of Three Hundred Thousand Dollars (\$300,000.00) per occurrence and Three Hundred Thousand Dollars (\$300,000.00) in the aggregate.

- B. Business auto liability insurance of at least One Million Dollars (\$1,000,000.00) combined single limit, on all owned, non-owned, leased and hired automobiles. If the Agreement contemplates the transportation of the users of County services (such as but not limited to Agency consumers), "Consumers" and Provider provides this service through the use of its employees' privately owned vehicles "POV", then the Provider's Business Auto Liability insurance shall sit excess to the employees "POV" insurance and provide coverage above its employee's "POV" coverage. Provider agrees the business auto liability policy will be endorsed to provide this coverage.
- C. Professional liability (errors and omission) insurance of at least One Million Dollars (\$1,000,000.00) per claim and in the aggregate.
- D. Umbrella and excess liability insurance policy with limits of at least One Million Dollars (\$1,000,000.00) per occurrence and in the aggregate, above the commercial general and business auto primary policies and containing the following coverage:
 - 1. Additional insured endorsement;
 - 2. Pay on behalf of wording;
 - 3. Concurrency of effective dates with primary;
 - 4. Blanket contractual liability;
 - 5. Punitive damages coverage (where not prohibited by law);
 - 6. Aggregates: apply where applicable in primary;
 - 7. Care, custody and control follow form primary; and
 - 8. Drop down feature.

The amounts of insurance required in this section for General Liability, Business Auto Liability and Umbrella/Excess Liability may be satisfied by Provider purchasing coverage for the limits specified or by any combination of underlying and umbrella limits, so long as the total amount of insurance is not less than the limits specified in General Liability, Business Auto Liability and Umbrella/Excess Liability when added together.

- E. Workers' Compensation insurance at the statutory limits required by ORC.
- F. The Provider further agrees with the following provisions:
 - 1. All policies, except workers' compensation and professional liability, will endorse as additional insured the Board of County Commissioners, and Agency and their respective officials, employees, agents, and volunteers, including their Board of Trustees if applicable. The additional insured endorsement shall be on an ACORD or ISO form.
 - 2. The insurance endorsement forms and the certificate of insurance forms will be sent to the Agency Director or Designee. The forms must state the following: "Board of County Commissioners, and Agency and their respective officials, employees, agents, and volunteers are endorsed as additional insured as required by agreement on the commercial general, business auto and umbrella/excess liability policies."
 - 3. Each policy required by this clause shall be endorsed to state that coverage shall not be canceled or materially changed except after thirty (30) calendar days prior written notice given to the Agency Director or Designee.
 - 4. Provider shall furnish the Agency with original certificates and amendatory endorsements effecting coverage required by this clause. All certificates and endorsements are to be received by the Agency

before the Agreement commences. The Agency reserves the right at any time to require complete, certified copies of all required insurance policies, including endorsements affecting the coverage required by these specifications.

- 5. Failure of the Agency to demand such certificate or other evidence of full compliance with these insurance requirements or failure of the Agency to identify a deficiency from evidence provided shall not be construed as a waiver of Provider's obligation to maintain such insurance.
- 6. Provider shall declare any self-insured retention to the Agency pertaining to liability insurance. Provider shall provide a financial guarantee satisfactory to the Agency guaranteeing payment of losses and related investigations, claims administration and defense expenses for any self-insured retention.
- 7. If Provider provides insurance coverage under a "claims-made" basis, Provider shall provide evidence of either of the following for each type of insurance which is provided on a claims-made basis: unlimited extended reporting period coverage, which allows for an unlimited period of time to report claims from incidents that occurred after the policy's retroactive date and before the end of the policy period (tail coverage), or; continuous coverage from the original retroactive date of coverage. The original retroactive date of coverage means original effective date of the first claim-made policy issued for a similar coverage while Provider was under Agreement with the County on behalf of the Agency.
- 8. Provider will require all insurance policies in any way related to the work and secured and maintained by Provider to include endorsements stating each underwriter will waive all rights of recovery, under subrogation or otherwise, against the County and the Agency. Provider will require of subcontractors, by appropriate written agreements, similar waivers each in favor of all parties enumerated in this section.
- 9. Provider, the County, and the Agency agree to fully cooperate, participate, and comply with all reasonable requirements and recommendations of the insurers and insurance brokers issuing or arranging for issuance of the policies required here, in all areas of safety, insurance program administration, claim reporting and investigating and audit procedures.
- 10. Provider's insurance coverage shall be primary insurance with respect to the County, the Agency, their respective officials, employees, agents, and volunteers. Any insurance maintained by the County or the Agency shall be excess of Provider's insurance and shall not contribute to it.
- 11. If any of the work or Services contemplated by this Agreement is subcontractors, Provider will ensure that any subcontractors comply with all insurance requirements contained herein.
- 12. If the Agreement provider is a government entity, insurance requirements will be fulfilled under the County Risk Sharing Authority (CORSA).

Article XXI. INDEMNIFICATION & HOLD HARMLESS

- A. To the fullest extent permitted by, and in compliance with, applicable law, Provider agrees to protect, defend, indemnify and hold harmless the Agency and the Board of County Commissioners, their respective members, officials, employees, agents, and volunteers (the "Indemnified Parties") from and against all damages, liability, losses, claims, suits, actions, administrative proceedings, regulatory proceedings/hearings, judgments and expenses, subrogation (of any party involved in the subject of this Agreement), attorneys' fees, court costs, defense costs or other injury or damage (collectively "Damages"), whether actual, alleged or threatened, resulting from injury or damages of any kind whatsoever to any business, entity or person (including death), or damage to property (including destruction, loss of, loss of use of resulting without injury damage or destruction) of whatsoever nature, arising out of or incident to in any way, the performance of the terms of this Agreement including, without limitation, by Provider, its subcontractor(s), Provider's or its subcontractor(s') employees, agents, assigns, and those designated by Provider to perform the work or services encompassed by the Agreement. Provider agrees to pay all damages, costs and expenses of the Indemnified Parties in defending any action arising out of the aforementioned acts or omissions.
- B. Each Party agrees to be responsible for any personal injury or property damage caused solely by its negligent acts or omissions as determined by a court of competent jurisdiction, or as the parties may otherwise mutually agree in writing.
- C. This Article is not applicable to Agreements between governmental entities.

Article XXII. SCREENING AND SELECTION

A. Criminal Record Check

- 1. Provider warrants and represents it will comply with Article X as it relates to criminal record checks. Provider shall insure that every individual subject to a Bureau of Criminal Investigation (BCI) criminal records check will sign a release of information to allow inspection and audit of the above criminal records transcripts or reports by the Agency or a private vendor hired by the Agency to conduct compliance reviews on their behalf.
- 2. Provider shall not assign any individual to work with or transport children until a BCI report and a criminal record transcript has been obtained.
- 3. Except as provided in Section C below, Provider shall not utilize an employee, foster caregiver or all of the above who has been convicted or plead guilty to any violations contained in <u>ORC 5153.111(B)(1)</u>. <u>ORC 2919.24</u>, and <u>ORC 2151.86</u>, and <u>OAC Chapters 5101:2-5</u>, 5101:2-7, 5101:2-9, 5101:2-48.
- 4. Provider agrees to be financially responsible for any of the following requirements in <u>OAC Chapters</u> <u>5101:2-5, 5101:2-7, 5101:2-9 and 5101:2-48</u> resulting in financial penalty due to lack of compliance with the criminal records checks.
- B. Transportation of Child
 - 1. The caregiver shall ensure the transportation of children in care will be reliable, legal and safe transportation with safety restraints, as appropriate for the child, and must be in compliance with applicable local, state and Federal transportation laws:
 - a. Maintenance of a current valid driver's license and vehicle insurance.
 - b. All children being transported by Provider must follow Ohio's Child Passenger Safety Law as defined in <u>ORC 4511.81.</u>
 - c. No child that is a passenger and is required to have a seat restraint can be transported by said provider until these requirements are met.
 - 2. In addition to the requirements set forth above, Provider shall not permit any individual to transport a Child if:
 - a. The individual has a condition which would affect safe operation of a motor vehicle;
 - b. The individual has six (6) or more points on his/her driver's license; or
 - c. The individual has been convicted of, or pleaded guilty to, a violation of section <u>4511.19</u> (Operating vehicle under the influence of alcohol or drugs OVI or OVUAC) of the Revised Code if the individual previously was convicted of or plead guilty to two or more violations within the three years immediately preceding the current violation.

C. Rehabilitation

- 1. Notwithstanding the above, Provider may make a request to the Agency to utilize an individual if Provider believes the individual has met the rehabilitative standards of <u>OAC 5101:2-07-02(I)</u> as follows:
 - a. If the Provider is seeking rehabilitation for a foster caregiver, a foster care applicant or other resident of the foster caregiver's household, Provider must provide written verification that the rehabilitation standards of <u>OAC 5101:2-7-02</u> have been met.
 - b. If the Provider is seeking rehabilitation for any other individual serving Agency children, Provider must provide written verification from the individual that the rehabilitative conditions in accordance with <u>OAC 5101:2-5-09</u> have been met.
- 2. The Agency shall review the facts presented and may allow the individual to work with, volunteer with or transport Agency children on a case-by-case basis. It is the Agency's sole discretion to permit a rehabilitated individual to work with, volunteer with or transport children.
- D. Verification of Job or Volunteer Application:

Provider shall check and document each applicant's personal and employment references, general work history, relevant experience, and training information. Provider further agrees it will not employ an individual in relation to this Agreement unless it has received satisfactory employment references, work history, relevant experience, and training information.

Article XXIII. PROHIBITION OF CORPORAL & DEGRADING PUNISHMENT

Agency prohibits the use of corporal or degrading punishment against children served by Agency and must comply with requirements in <u>OAC 5101:2-7-09</u>, OAC 5101:2-9-21, and <u>OAC 5101:2-9-22</u>

Article XXIV. FINDINGS FOR RECOVERY

<u>ORC 9.24</u> prohibits public agencies from awarding an Agreement for goods, services, or construction paid for in whole or in part from federal, state and local funds, to an entity against whom a finding for recovery has been issued if the finding is unresolved. By entering into this Agreement, Provider warrants and represents that they do not have an unresolved finding for recovery. Provider shall notify the Agency within ten (10) business days of its notification should the Provider be issued such finding by the Auditor of the State.

Article XXV. PUBLIC RECORDS

This Agreement is a matter of public record under the Ohio public records law. By entering into this Agreement, Provider acknowledges and understands that records maintained by Provider pursuant to this Agreement may also be deemed public records and subject to disclosure under Ohio law. Upon request made pursuant to Ohio law, the Agency shall make available the Agreement and all public records generated as a result of this Agreement.

Article XXVI. CHILD SUPPORT ENFORCEMENT

Provider agrees to cooperate with ODJFS and any Ohio Child Support Enforcement Agency ("CSEA") in ensuring Provider and Provider's employees meet child support obligations established under state or federal law. Further, by executing this Agreement, Provider certifies present and future compliance with any court or valid administrative order for the withholding of support which is issued pursuant to the applicable sections in <u>ORC Chapters 3119, 3121, 3123</u>, and <u>3125</u>.

Article XXVII. DECLARATION OF PROPERTY TAX DELINQUENCY

After award of an Agreement, and prior to the time the Agreement is entered into, the successful Provider shall submit a statement in accordance with <u>ORC 5719.042</u>. Such statement shall affirm under oath that the person with whom the Agreement is to be made was not charged at the time the bid was submitted with any delinquent personal property taxes on the general tax list of personal property of any county in which the taxing district has territory, or that such person was charged with delinquent personal property taxes on any such tax list, in which case the statement shall also set forth the amount of such due and unpaid delinquent taxes any due and unpaid penalties and interest thereon. If the statement indicates that the taxpayer was charged with any such taxes, a copy of the statement shall be transmitted by the fiscal officer to the county treasurer within thirty days of the date it is submitted.

A copy of the statement shall also be incorporated into the Agreement, and no payment shall be made with respect to any contract to which this section applies unless such statement has been so incorporated as a part thereof.

Article XXVIII. SUBCONTRACTING AND DELEGATION

The performance of any duty, responsibility or function which is the obligation of the Provider under this Agreement may be delegated or subcontracted to any agent or subcontractor of Provider if Provider has obtained the prior written consent of the Agency for that delegation subcontract. Provider is responsible for ensuring that the duties, responsibilities or functions so delegated or subcontracted are performed in accordance with the provisions and standards of this Agreement, and the actions and omissions of any such agent or subcontractor shall be deemed to be the actions and omissions of Provider for purposes of this Agreement.

Article XXIX. PROPERTY OF AGENCY

The Deliverable(s) and any item(s) provided or produced pursuant to this Agreement (collectively called "Deliverables") will be considered "works made for hire" within the meaning of copyright laws of the United States of America and the State of Ohio. The Agency is the sole author of the Deliverables and the sole owner of all rights therein. If any portion of

07/01/2023 - 06/30/2024 Page 19 of 25 the Deliverables are deemed not to be a "work made for hire", or if there are any rights in the Deliverables not so conveyed to the Agency, then Provider agrees to, and by executing this Agreement hereby does, assign to the Agency all worldwide rights, title, and interest in and to the Deliverables. The Agency acknowledges that its sole ownership of the Deliverables under this Agreement does not affect Provider's right to use general concepts, algorithms, programming techniques, methodologies, or technology that have been developed by Provider prior to this Agreement or that are generally known and available. Any Deliverable provided or produced by Provider under this Agreement or with funds hereunder, including any documents, data, photographs and negatives, electronic reports/records, or other media, are the property of the Agency, which has an unrestricted right to reproduce, distribute, modify, maintain, and use the Deliverables. Provider shall not obtain copyright, patent, or other proprietary protection for the Deliverables. Provider to use such copyrighted material, unless the copyright owner gives prior written approval for the Agency and Provider to use such copyrighted material. Provider agrees that all Deliverables will be made freely available to the general public unless the Agency determines that, pursuant to state or federal law, such materials are confidential or otherwise exempt from disclosure.

Article XXX. SEVERABILITY

If any term of this Agreement or its application thereof to any person or circumstance shall to any extent be held invalid or unenforceable, the remainder of this Agreement, or the application of such term or provision to persons or circumstances other than those as to which it is held invalid or unenforceable, shall not be affected thereby. Each term and provision of this Agreement shall be valid and enforced to the fullest extent permitted by law.

Article XXXI. NO ADDITIONAL WAIVER IMPLIED

If the Agency or Provider fails to perform any obligations under this Agreement and thereafter such failure is waived by the other party, such waiver shall be limited to the particular matter waived and shall not be deemed to waive any other failure hereunder, nor a waiver of a subsequent breach of the same provision or condition. Waivers shall not be effective unless in writing.

Article XXXII. COUNTERPARTS

This Agreement may be executed as an original document only, or simultaneously in two or more counterparts, each of which shall be deemed an original, and each of these counterparts shall constitute one and the same instrument. It shall not be necessary in making proof of this Contract to produce or account for more than one such counterpart An electronic signature or a scanned or otherwise reproduced signature shall be a binding signature and carry the same legal force as the original.

Article XXXIII. APPLICABLE LAW AND VENUE

This Agreement and any modifications, Attachments, Exhibits, Addenda, or alterations, shall be governed, construed, and enforced under the laws of Ohio. Any legal action brought pursuant to this agreement will be filed in the Ohio courts, and Ohio law as well as Federal law will apply.

ATTACHMENT

Attachment One.

Reason: Article Section: Article I - Scope of Placement Services Detail: Article I SECTIONS 1.02 & 1.03, References to Exhibit I Article I, Item A ('Scope of Placement Services') will serve as Exhibit I.

Attachment Two.

Reason: Article Section: Article V - Provider Responsibilities Detail: Article V ITEM A

Provider is responsible for ensuring transportation services are in place for all case-related activities and routine needs. If extenuating circumstances exist and provider is not able to utilize its own resources to ensure coverage, Provider must notify Agency in advance of the transportation need, and coordinate with Agency to arrange needed transportation. Provider should bring any extraordinary travel needs to the attention of Agency so both parties can ensure proper coverage and explore potential compensation for needed transportation services.

ITEM B

Progress reports, demonstration completion of monthly activities as required by Ohio Administrative Code, will be submitted on a monthly basis, by no later than the 20th calendar day of the following month.

ITEMS D, E, & F

After-Hours/On-Call Process

Provider shall notify Agency of any items identified in Article V, according to the following options: For calls during business hours (Monday through Friday, from 8:00 a.m. to 4:00 p.m., excluding holidays), Provider shall call (740) 652-7854 and inform the operator of the need to urgently speak to casework staff. For calls outside of business hours, Provider shall call (740)808-0009 or (740)808-0982 in order to notify Agency's On-Call staff member of the urgent situation.

Insert new item - ITEM S

Provider will ensure access to Normalcy activities, based upon the developmental, social, and emotional functioning of each child placement.

Insert new item - ITEM T

All Provider staff, and foster caregivers when applicable, should seek and receive prior authorization from Agency for any type of non-routine medical care or medication needs. This includes, but is not limited to: major medical treatment, medical procedures, surgery, implementation of or change in psychotropic medications, and any other medical intervention that carries a high risk of side effects, impairment, or harm. Routine well visits and treatment for typical childhood illnesses will not require such prior authorization.

In the event of an emergency, the child should be taken to the nearest medical facility for prompt treatment. As soon as possible, Agency shall be contacted, according to the process outlined in the addendum to Article V.

Attachment Three.

Reason: Article Section: Article VIII - Reimbursement for Placement Services Detail: Article VIII ITEM B Agency agrees to pay Provider on the basis of a daily per diem (identified in Schedule A of this agreement) for the

Contract ID: 19346132 Fairfield County Department of Job and Family Services / A New Leaf, Inc.

07/01/2023 - 06/30/2024 Page 21 of 25 placement for each child, as identified by each child's current Level of Care (LOC). The LOC will be agreed upon at the time of placement. The Agency and Provider may request a re-evaluation of the child's LOC at any time in order to best meet the child's identified needs.

ITEM D

Agency agrees to provide a one-time initial clothing authorization of up to \$150.00 for children ten and under and up to \$250.00 for children over the age of ten. If the child is under ten and wearing adult sizes, the authorization will be for \$250.00. Any purchases beyond this must be due to extenuating circumstances and approved in writing, in advance of the purchase. Routine clothing needs are considered part of maintenance and will not be paid or reimbursed by Agency.

ITEM E

If a child goes on any form of unpaid leave and is reasonably expected to return to the same placement, Agency may pay for up to seven (7) consecutive nights of leave. In order for Agency to pay for such leave, Provider (including foster parents, if applicable) must agree to remain available for regular services and needed support during such leave. Leave beyond seven (7) days will not be paid unless extenuating circumstances exist, and both Agency and Provider agree to payment terms in advance, in writing.

Regardless of length, Agency will not pay Provider for any leave during which payment is being made to another provider, nor for leave where the child is in a paid or unpaid alternative placement outside of Provider's network due to a lack of placement availability with Provider. Any deviation from this must be agreed to in advance, in writing, by both Agency and Provider.

Item F

Medicaid/Insurance

Upon receipt of formal documentation, FCCPS will submit to Service Provider Medicaid/insurance numbers for children in FCCPS custody, as applicable. The service provider is required to utilize Medicaid-approved healthcare providers in the appropriate managed care network for the provision of mental health,dental and/or medical services (hereafter referred to collectively as "medical services") to children in the custody of FCCPS. The Service Provider will report applicable Medicaid/insurance information to the healthcare providers and instruct healthcare providers to seek payment from Medicaid or any other available third party payor for medical services rendered to children in FCCPS custody. FCCPS will not pay for the provision of any medical services to children in FCCPS Deputy Director or authorized designee has provided specific prior written authorization for such medical services and associated costs.

In situations where the Service Provider does not possess a Medicaid/insurance number or other information required to bill an alternative source for services provided to children in the custody of FCCPS, the Service Provider must take the following actions.

A. The Service Provider will contact the Finance Department at (740)652-7889 for assistance with resolving Medicaid/insurance number issues.

B. Within thirty (30) days if an invoice from a healthcare provider for services rendered to a child in FCCPS custody, the Service Provider should forward the invoice to the FCCPS Finance Department at: Fairfield County Job and Family Services-Protective Services, 239 West Main St., Lancaster, OH 43130 or fax such invoice to the FCCPS Finance Department at (740)-687-7070. Failure to forward this invoice to FCCPS within thirty (30)days will constitute a waiver of any claim against FCCPS for payment of the invoice. If the Service Provider receives additional notices regarding the invoice, the Service Provider must contact the FCCPS Finance Department at (740)-652-7889 to confirm that FCCPS received the initial invoice and to obtain the status of payment arrangements. The Service Provider SHALL NOT pay the invoice and expect or request reimbursement from FCCPS without the prior written approval of FCCPS.

C. If a child who is in custody of FCCPS requires pharmaceutical supplies, Service Provider must obtain the supplies from a pharmacy that accepts Medicaid/insurance payments.

ITEM G

In any instance where payment cannot be made within 30 days, Agency will make every effort to ensure Provider is

Contract ID: 19346132

Fairfield County Department of Job and Family Services / A New Leaf, Inc.

07/01/2023 - 06/30/2024 Page 22 of 25 paid within 45 days, and is made aware in advance if this is not possible.

FCCPS retains the right to recoup funds from the Service Provider upon the determination that third party funds are duplicative (in the aggregate) of FCCPS payments to the Service Provider, or in the event that the Service Provider fails to properly credit any and all such third party payments. Relative to recouping funds, FCCPS may withhold from subsequent reimbursement to the Service Provider an amount equal to any un-credited or duplicate third party payments. For purposes of this paragraph, "third party" includes, but is not limited to , Medicaid and private insurance companies.

The Service Provider shall obtain and provide a written estimate for any non-routine, non-emergency, or out-ofnetwork medical and dental expenses to FCCPS along with the written recommendation of the physician or dentist. The Servide Provider is not permitted to deliver or authorize any health/dental care or treatment services (including, but not limited to, mental health services), without the prior written consent of the FCCPS Deputy Director or authorized designee (see Consent for Medical Treatment letter).

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date of the signature of the parties.

SIGNATURES OF PARTIES:

Provider: A New Leaf, Inc.

Print Name & Title	Signature // /	Date
HEIDI L. GARWOOD EXECUTIVE Dr 76	YE	5.18-23

Agency: Fairfield County Department of Job and Family Services

Print Name & Title	Signature	Date

Additional Signatures

Signature	Date			
	Signature			

Title IV-E Schedule A Rate Information

Title IV-E Schedule A Rate Information Agency: Fairfield County Department of Job and Family Services Provider / ID: A New Leaf, Inc. / 24569

Run Date: 05/18/2023 Contract Period: 07/01/2023 - 06/30/2024

Service Description		Person	Person ID	Maintenance Per Diem	Administration Per Diem	Case Management Per Diem	Transportation / Administration Per Diem	Transporation / Maintenance Per Diem	Other Direct Services Per Diem	Behavioral Healthcare Per Diem	Other Per Diem Cost	Total Per Diern Cost	Cost Begin Date	Cost End Date
Traditional (30306)- Excpt Need	377626			\$48.80	\$37.00	anan inana ina ina da sa sa sa sa sa						\$85.80	07/01/2023	06/30/2024
Traditional (30306)- FFH	96947			\$32.25	\$36.00		annen "Terler for d'annen for en anne d	telenides e denormente		n na mainte de la constante de		\$68.25	07/01/2023	06/30/2024
Traditional (30306)- Spec Need	375627			\$40.30	\$37.00							\$77.30	07/01/2023	06/30/2024

Contract ID: 19346132 Fairfield County Department of Job and Family Services / A New Leaf, Inc. / 24569 07/01/2023 - 06/30/2024 Page 25 of 25

•



A Contract regarding A New Leaf 23-24 between Job and Family Services and

Approved on 5/25/2023 12:38:03 PM by Sarah Fortner, Assistant Deputy Director

South Jota

Sarah Fortner Assistant Deputy Director

Approved on 5/25/2023 1:43:04 PM by Corey Clark, Director of Fairfield County Job & Family Services

GBCh_

Corey Clark, Director Fairfield County Job & Family Services

Certified Search for Unresolved Findings for Recovery

OHIO AUDITOR OF STATE, KEITH FABER

Auditor of State - Unresolved Findings for Recovery Certified Search

I have searched The Auditor of State's unresolved findings for recovery database using the following criteria:

Contractor's Information:

Name: , Organization: A New Leaf, Inc. Date: 5/18/2023 10:46:59 AM

This search produced the following list of **1** possible matches:

Name/Organization	Address
Helping Africans in a New Direction	6084 Busch Blvd. #4

The above list represents possible matches for the search criteria you entered. Please note that pursuant to ORC 9.24, only the person (which includes an organization) actually named in the finding for recovery is prohibited from being awarded a contract.

If the person you are searching for appears on this list, it means that the person has one or more findings for recovery and is prohibited from being awarded a contract described in ORC 9.24, unless one of the exceptions in that section apply.

If the person you are searching for does not appear on this list, an initialed copy of this page can serve as documentation of your compliance with ORC 9.24(E).

Please note that pursuant to ORC 9.24, it is the responsibility of the public office to verify that a person to whom it plans to award a contract does not appear in the Auditor of State's database. The Auditor of State's office is not responsible for inaccurate search results caused by user error or other circumstances beyond the Auditor of State's control.

06/06/2023

Office of Auditor of State 88 East Broad Street Post Office Box 1140 Columbus, OH 43216-1140

> (614) 466-4514 (800) 282-0370

Prosecutor's Approval Page

Resolution No.

A resolution authorizing the approval of a service agreement by and between Fairfield County Job & Family Services, Child Protective Services Division and A New Leaf, Inc.

(Fairfield County Job and Family Services)

Approved as to form on 6/1/2023 11:28:25 AM by Steven Darnell,

Signature Page

Resolution No. 2023-06.06.q

A resolution authorizing the approval of a service agreement by and between Fairfield County Job & Family Services, Child Protective Services Division and A New Leaf, Inc.

(Fairfield County Job and Family Services)

Upon the motion of Commissioner Jeffrey M. Fix, seconded by Commissioner David L. Levacy, this resolution has been Adopted:

Voting:

Steven A. Davis, President	Aye
David L. Levacy, Vice President	Aye
Jeffrey M. Fix	Aye

Board of County Commissioners Fairfield County, Ohio

CERTIFICATE OF CLERK

It is hereby certified that the foregoing is a true and correct transcript of a resolution acted upon by the Board of County Commissioners, Fairfield County, Ohio on the date noted above.

Rochelle Merringen

Rochelle Menningen Board of County Commissioners Fairfield County, Ohio

2023-06.06.r

A resolution authorizing the approval of a service agreement by and between Fairfield County Job & Family Services, Child Protective Services Division and Ohio Teaching Family Association.

WHEREAS, Fairfield County Job & Family Services, Child Protective Services is requesting the Board of Commissioners approval of a service agreement with Ohio Teaching Family Association, 4020 Waterville Swanton Rd 300, Swanton, OH 43558 and

WHEREAS, the purpose of the service agreement is to provide Network Placement and Related Services for children who are in the care and custody of the Agency; and

WHEREAS, this agreement shall be effective April 1st, 2023 through March 31st, 2024; and

WHEREAS, a purchase order encumbering the funds for the services was acquired; and

WHEREAS, the Prosecuting Attorney has approved the agreement as to form.

NOW THEREFORE, BE IT RESOLVED BY THE BOARD OF COUNTY COMMISSIONERS, COUNTY OF FAIRFIELD, AND STATE OF OHIO:

Section 1. That the Fairfield County Board of Commissioners hereby approves the attached Network Placement Service Agreement for Ohio Teaching Family Association.

Prepared by: Brandi Downhour cc: JFS / Fiscal Supervisor

ROUTING FORM FOR CONTRACTS
The undersigned designee of the County affirms that he/she has reviewed the attached contract to ensure that it complies with County's needs and previous negotiations. The undersigned designee further affirms that the County has complied with the competitive selection process, as prescribed by the Ohio Revised Code, by selecting one of the boxes below.
A. Goods and/or Services in excess of \$50,000.00—competitively selected via an Invitation to Bid, pursuant to R.C. 307.86-307.92
B. Goods and/or Services in excess of \$50,000.00—competitively selected via a Request for Proposals, pursuant to R.C. 307.862
C. Dublic Improvement contracts—competitively selected pursuant to R.C. 153.08-153.12
D. Architect/Engineer design services for public improvements—selected through the Request for Qualifications process pursuant to R.C. 153.65-153.72
E. County Road Improvement/Construction-competitively selected pursuant to R.C. 5555.61
F. The subject matter was exempt from competitive selection for the following reason(s):
 Under \$50,000 State Term #: (copy of State Term Contract must be attached) ODOT Term #: (See R.C. 5513.01) Professional Services (See R.C. 307.86) Emergency (Follow procedure under ORC 307.86(A)) Sole Source (attach documentation as to why contract is sole source) Other: (cite to authority or explain why matter is exempt from competitive bidding)
G. Agreement not subject to Sections A-F (explain):
H. Compliance with Fairfield County Board of Commissioners Procurement Guidelines
 No County employee, employee's family member, or employee's business associate has an interest in this contract OR such interest has been disclosed and reviewed by the Prosecutor's Office No Finding for Recovery against Vendor as required under R.C. 9.24 (search via "Certified Search" on http://ffr.ohioauditor.gov/) Obtained 3 quotes for purchases under \$50,000 Purchase Order is included with Agreement
Signed this day of, 20
$T_{2}(1)$

* Please note that this checklist only addresses County and statutory requirements. If a contract is paid for with state and/or federal funds, please consult with the appropriate state and/or federal agency to ensure your department is complying with any additional requirements. By submitting a request for approval, you are certifying you have addressed County, statutory, and grant requirements.*

Fiscal Supervisor

Name and Title

01.2018 COST ANALYSIS:

For foster care placement, network providers have Title IV-E reimbursement ceilings, and Fairfield County rates have typically been below the state negotiated ceilings. Historically, the traditional, daily rate is less than \$225.

The review and evaluation of the separate cost elements and proposed profit would include an evaluation of special considerations and special needs, as there are cases which would be reviewed independently based on extraordinary factors. If the rate was more than \$225 per day, it is expected that there would be extraordinary, case specific needs, knowing what we know about the market in our area.

Ohio Department of Job and Family Services

AGREEMENT FOR TITLE IV-E AGENCIES AND PROVIDERS FOR THE PROVISION OF CHILD PLACEMENT

This Agreement sets forth the terms and conditions between the parties for placement services for children who are in the care and custody of the Agency named below.

This Agreement is between Fairfield County Department of Job and Family Services, a Title IV-E Agency, hereinafter "Agency", whose address is:

Fairfield County Department of Job and Family Services 239 W Main St Lancaster, OH 43130

and

Ohio Teaching Family Association, hereinafter "Provider", whose address is:

Ohio Teaching Family Association 4020 Waterville Swanton Rd 300 Swanton, OH 43558

Collectively the "Parties".

04/01/2023 - 03/31/2024 Page 1 of 27

Table of Contents

ARTICLE I. Section 1.01 Section 1.02 Section 1.03 ARTICLE II. ARTICLE II. ARTICLE IV. ARTICLE V. ARTICLE V. ARTICLE VI. ARTICLE VII. ARTICLE VII. ARTICLE XII. ARTICLE X. ARTICLE X. ARTICLE XII. ARTICLE X. ARTICLE XV. ARTICLE XV. ARTICLE XV. ARTICLE XV. ARTICLE XV. ARTICLE XV. ARTICLE XV. ARTICLE XXI. ARTICLE XXI. ARTICLE XXI. ARTICLE XXII. ARTICLE XXII. ARTICLE XXII. ARTICLE XXII. ARTICLE XXII. ARTICLE XXII. ARTICLE XXV. ARTICLE XXV.	SCOPE OF PLACEMENT SERVICES FOR AGREEMENTS COMPETITIVELY PROCURED FOR AGREEMENTS NOT COMPETITIVELY PROCURED EXHIBITS TERM OF AGREEMENT ORDER OF PRECEDENCE DEFINITIONS GOVERNING THIS AGREEMENT PROVIDER RESPONSIBILITIES INVOICING FOR PLACEMENT SERVICES REIMBURSEMENT FOR PLACEMENT SERVICES REIMBURSEMENT FOR PLACEMENT SERVICES TERMINATION; BREACH AND DEFAULT RECORDS RETENTION, CONFIDENTIALITY AND DATA SECURITY REQUIREMENTS PROVIDER ASSURANCES AND CERTIFICATIONS INDEPENDENT CONTRACTOR AUDITS AND OTHER FINANCIAL MATTERS GRIEVANCE/DISPUTE RESOLUTION PROCESS ATTACHMENTS/ADDENDA NOTICE CONSTRUCTION NO ASSURANCES CONFLICT OF INTEREST INSURANCE INDEMNIFICATION AND HOLD HARMLESS SCREENING AND SELECTION PROHIBITION OF CORPORAL & DEGRADING PUNISHMENT FINDINGS FOR RECOVERY PUBLIC RECORDS CHILD SUPPORT ENFORCEMENT DECLARATION OF PROPERTY TAX DELINQUENCY SUBCONTRACTING AND DELEGATION PROPERTY OF AGENCY SEVERABILITY NO ADDITIONAL WAIVER IMPLIED
ARTICLE XXIX.	PROPERTY OF AGENCY
ARTICLE XXXII.	COUNTERPARTS
ARTICLE XXXIII.	APPLICABLE LAW AND VENUE
ATTACHMENTS TO TH	HIS AGREEMENT

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 2 of 27

RECITALS

WHEREAS, the Agency is responsible under Ohio Revised Code (ORC) Title 51, Chapter <u>5153</u> for the provision of protective services for dependent, neglected, and abused children; and,

WHEREAS, the Agency is authorized under ORC Title 51, Chapter <u>5153.16</u> to provide care and services which it deems to be in the best interest of any child who needs or is likely to need public care and services; and,

WHEREAS, the Provider is an organization duly organized and validly existing and is qualified to do business under the laws in the State of Ohio or in the state where the Provider of services is located and has all requisite legal power and authority to execute this Agreement and to carry out its terms, conditions and provisions; and is licensed, certified or approved to provide services to children and families in accordance with Ohio law or the state where the Provider of services is located.

NOW, THEREFORE, in consideration of the mutual promises and responsibilities set forth herein, the Agency and Provider agree as follows:

Article I. SCOPE OF PLACEMENT SERVICES

In addition to the services described in Exhibit I-Scope of Work, Provider agrees to provide and shall provide the placement and related services specified in each Individual Child Care Agreement (ICCA) for children in the care and custody of the Title IV-E Agency. The ICCA shall be consistent with current federal, state and local laws, rules and regulations applicable to the Provider's license or certified functions and services. If an Agreement and ICCA both exist, the Agreement supersedes.

See Attachment 1 for additional details.

Section 1.01 FOR AGREEMENTS COMPETITIVELY PROCURED

Without limiting the services set forth herein, Provider will provide the Services pursuant to and consistent with the Requests for Proposals (RFP) and the Provider's Proposal submitted in response to the RFP, the Provider agrees to provide and shall provide the placement and related services described in Exhibit I-Scope of Work.

Section 1.02 FOR AGREEMENTS NOT COMPETITIVELY PROCURED

The Provider agrees to provide and shall provide the placement and related services described in the Exhibit I- Scope of Work.

Section 1.03 EXHIBITS

The following exhibits are deemed to be a part of this Agreement as if fully set forth herein:

- A. Exhibit I Scope of Work;
- B. Exhibit II Request for Proposals (if applicable);
- C. Exhibit III Provider's Response to the Request for Proposals (if applicable); and
- D. Exhibit IV Schedule A Rate Information.

Article II. TERM OF AGREEMENT

This Agreement is in effect from 04/01/2023 through 03/31/2024, unless this Agreement is suspended or terminated pursuant to Article VIII prior to the termination date.

In addition to the initial term described above, this Agreement may be extended, at the option of the Agency and upon written agreement of the Provider, for ______ additional, ______ year terms not to exceed ______ years. Notice of Agency's intention to extend the Agreement shall be provided in writing to Provider no less than 90 calendar days before the expiration of any Agreement term then in effect. (If a previous Request for Proposal [RFP] allows, the Agreement may be extended for a period of time to ensure adequate completion of the Agency's competitive procurement process at the rates existing for the term then in effect.)

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 3 of 27

Article III. ORDER OF PRECEDENCE

This Agreement and all Exhibits are intended to supplement and complement each other and shall, where possible, be so interpreted. However, if any provision of this Agreement irreconcilably conflicts with an Exhibit, this Agreement takes precedence over the Exhibit(s).

In the event there is an inconsistency between the Exhibit(s), the inconsistency shall be resolved in the following order:

- A. Exhibit I: Scope of Work; then
- B. Exhibit II: Request for Proposals (if applicable); then
- C. Exhibit III: Provider's Proposals (if applicable); then
- D. Exhibit IV: Title IV-E Schedule A Rate Information.

Article IV. DEFINITIONS GOVERNING THIS AGREEMENT

The following definitions govern this Agreement:

- A. Agreement means this Agreement, attachments and exhibits thereto.
- B. Material Breach shall mean an act or omission that violates or contravenes an obligation required under the Agreement and which, by itself or together with one or more other breaches, has a negative effect on, or thwarts the purpose of the Agreement as stated herein. A Material Breach shall not include an act or omission, which has a trivial or negligible effect on the quality, quantity, or delivery of the goods and services to be provided under the Agreement.
- C. Child(ren) means any person under eighteen years of age or a mentally or physically handicapped person under twenty-one years of age in the Agency's custody and under the care of the Provider for the provision of placement services.
- D. All other definitions to be resolved through Federal Regulations, Ohio Administrative Code (OAC) 5101:2-1-01 and any related cross-references.
- E. Aftercare Support, as defined, in rule 5101:2-1-01 the Administrative Code, is case management activities performed with or on behalf of a child/family, by the Qualified Residential Treatment Program (QRTP) as part of the required discharge plan developed by the permanency team for a minimum of six months from discharge.

Such activities are to include but are not limited to the following:

- 1. Minimum of monthly contact with child and family (Face-to-Face /Telephonic/Skype/etc.)
- 2. Linkage to community services.
- 3. Follow up with community service.
- Documentation of the monthly contacts in the Residential Treatment Information System (RTIS).

When serving multiple children in the save family, the cost for non-Medicaid Aftercare Supports may be billed for only one child at the same time.

Article V. PROVIDER RESPONSIBILITIES

- A. Provider agrees to participate with Agency in the development and implementation of the Case Plan and ICCA including participation in case reviews and / or semi-annual administrative reviews, and the completion of reunification assessments for the children in placement with the Provider. Parties shall make best efforts to share information timely regarding participants and contact information involved with planning efforts related to children and families.
- B. Provider agrees to provide services agreed to in the Case Plan and ICCA (i.e.,transportation of children for routine services, including, but not limited to, court hearings, medical appointments, school therapy, recreational activities, visitations/family visits) unless otherwise negotiated in writing as an attachment to this Agreement. Any disputes involving services or placement will be resolved through mutual-agreement and modification to the ICCA. Provider agrees the Agency is the final authority in the process. The cost of providing these services is to be included in the Agency approved per diem.

04/01/2023 - 03/31/2024 Page 4 of 27

- C. Provider agrees to deliver aftercare support as described in Article IV.
- D. Provider agrees to ensure that any and all persons who may act as alternative caregivers or who have contact with the children are suitable for interaction pursuant to all applicable federal, state and local laws and regulations.
- E. Provider agrees that all caregivers must be approved by the Agency.
- F. Provider agrees to submit a progress report as negotiated by the parties for each child. The progress report will be based on the agreed upon services to be delivered to the child and/or family and will include documentation of services provided to the child and/or discharge summary. If Monthly Progress Reports are not received within 90 calendar days following the month of service provision, payment may be withheld at the Agency's discretion.
 - 1. Monthly Progress Reports shall be submitted by the 20th of the month following the month of service.
 - 2. The Monthly Progress Report will include the following medical related information:
 - a. Service type (i.e. medical, dental, vision, etc.);
 - b. Date(s) of service;
 - c. Reason for visit (i.e. routine, injury, etc.);
 - d. Practitioner name, address and contact number;
 - e. Name of hospital, practice, urgent care, etc.;
 - f. Prescribed medications and dosages;
 - g. Date(s) medication(s) were prescribed or changed; and
 - h. Changes to medications.
- G. Placement changes, emergency or non-emergency, shall occur only with the approval of the Agency. The following information shall be provided to the Agency for all placement changes: Name, address and phone number of the new foster home or other out-of-home care setting, the license/home study of the new care provider within 24 hours, excluding weekends and holidays.
- H. Provider agrees to notify all Agencies who have children placed in the same caregiver's home/group home/CRC when any child residing in the placement is critically injured or dies in that location. Notification will be made to the Agencies' Child Abuse/Neglect Hotline number or assigned Caseworker immediately.
- Notification to the Agency of Emergency Critical Incidents shall occur ASAP but no later than one hour of the Incident becoming known. Notification will be made to the Agency via the Agency's Child Abuse/Neglect Hotline or assigned Caseworker or by other established system. Critical incidents are those incidents defined in the Ohio Administrative Code that are applicable to the licensed or certified programs (ODJFS 5101:2-7-14, 5101:2-9-23 ODMHAS 5122-30-16, 5122-26-13, OAC 5123-17-02).

Emergency situations include but are not limited to the following:

- 1. Absent Without Leave (AWOL):
- 2. Child Alleging Physical or Sexual Abuse / Neglect;
- Death of Child;
- Illicit drug/alcohol use; Abuse of medication or toxic substance;
- Sudden injury or illness requiring an unplanned medical treatment or visit to the hospital;
- 6. Perpetrator of Delinquent/Criminal Act (Assault, Dangerous Behaviors, Homicidal Behaviors);
- 7. School Expulsion / Suspension (formal action by school);
- 8. Self-Injury (Suicidal Behaviors, Self-Harm Requiring external Medical Treatment, Hospital or ER);
- 9. Victim of assault, neglect, physical or sexual abuse; and
- 10. The filing of any law enforcement report involving the child.
- J. The Provider also agrees to notify the Agency within Twenty-four (24) hours, of any non-emergency situations. Non-emergency situations include but are not limited to the following:
 - 1. When physical restraint is used/applied; and
 - 2. Medication lapses or errors.

Notification will be made to the Agency via the Agency's Child Abuse Neglect Hotline / assigned Caseworker or by other established notification system.

Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 5 of 27

- K. Documentation of the emergency and non-emergency incidents as identified in "I and J" above shall be provided to the Agency via email, fax or other established notification system within 24 hours excluding weekends and holidays.
- L. The Provider agrees to submit each child's assessment and treatment plans as completed but no later than the 30th day of placement. Provider further agrees to provide treatment planning that will include, but is not limited to, education on or off site, preparation for integration into community-based school or vocational/job skills training, community service activities, independent living skills if age 14 or older, monitoring and supporting community adjustment.
- M. The Provider agrees to participate in joint planning with the Agency regarding modification to case plan services. Provider agrees that while the Provider may have input into the development of the child's case plan services and the ICCA, any disputes involving services or placement will be resolved through mutual agreement and modification to the ICCA. Provider agrees the Agency is the final authority in the process.
- N. The Provider shall participate in a Placement Preservation meeting if requested by the Agency prior to issuing a notice of removal of a child. A placement Preservation meeting shall be held within seven (7) business days of said request. Unless otherwise mutually agreed upon a minimum of thirty (30) calendar days' notice shall be given if placement preservation is unable to be achieved. A Discharge Plan Summary shall be provided no later than fifteen (15) calendar days after the date of discharge in accordance with the applicable licensed or certified program. (OAC 5101:2-5-17, OAC 5122-30-22, OAC 5122-30-04, OAC 5123:2-3-05).
- O. The Provider shall work in cooperation and collaboration with the Agency to provide information for each child's Lifebook and will fully comply with the provision of <u>OAC 5101:2-42-67</u> as applicable to private Providers. Provider's contribution to the Agency Lifebook for a child shall be for the episode of care with the Provider.
- P. The Provider agrees to provide Independent Living Services as set forth in accordance with <u>OAC 5101:2-42-19</u> for all children age 14 and above.
- Q. When applicable, due to the Provider being part of a managed care agreement as defined in <u>OAC 5101:2-1-01</u>, the Provider agrees to visit with the child face-to-face in the foster home, speak privately with the child and to meet with the caregiver at least monthly in accordance with rule <u>OAC 5101:2-42-65</u> of the Ohio Administrative Code.
- R. The Provider agrees to maintain its licenses and certifications from any source in good standing. The Provider agrees to report to Agency in writing any change in licensure or certification that negatively impacts such standing immediately if the negative action results in a temporary license, suspension of license or termination of license.
- S. Provider agrees that the reasonable and prudent parent standard training required by SEC. 471. [42 U.S.C. 671] of the Social Security Act and in accordance to <u>OAC 5101:2-5-33</u>, <u>OAC 5101:2-9-02</u> or <u>OAC 5101:2-9-03</u> has been completed.
- T. The Provider shall notify Agency of any changes in its status, such as intent to merge with another business or to close no later than forty-five (45) business days prior to the occurrence.
- U. The Provider agrees that the Agency shall have access to foster parent home studies and re-certifications for foster parents caring for children in placement, subject to confidentiality considerations. The Provider shall submit to Agency a copy of the current foster home license at the time of placement and recertification. Provider also agrees to notify Agency within twenty-four (24) hours of any change in the status of the foster home license.
- V. When there is a rule violation of a caregiver, a copy of the corrective action plan, if applicable, must be submitted to the Agency when the investigation is complete.
- W. The Provider agrees to notify the Agency of scheduling no less than fourteen (14) calendar days prior to all formal meetings (i.e. FTMs, Treatment Team Meetings, IEPs, etc.).
- X. The Provider agrees to adhere to the following Medical/Medication guidelines:
 - 1. To provide over-the-counter medications and/or supplies as part of the per diem of care;
 - 2. To comply with the medical consent process as identified by Agency;
 - 3. Only the Agency can give permission for the administering or change (addition or elimination) of

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 6 of 27

06/06/2023

psychotropic medication and its ongoing management; and

- 4. Provide an initial placement medical screening within 72 hours of child's placement into a placement resource under the Provider's operation and/or oversight.
- Y. To arrange for required health care/medical examinations within time frames required by <u>OAC 5101:2-42-66.1</u> and provide reports from the health care providers to the agency within 30 days of occurrence if the appropriate releases of information have been obtained by the Provider.
- Z. The Network Provider agrees to notify the Agency if placement resource is currently under investigation for license violations or misconduct toward children or other third-party investigation.
- AA. The Provider will immediately notify the Agency:
 - 1. If the Provider is out of compliance with any licensing authority rules or the placement resource is under investigation for license violations or misconduct toward children. Immediately is defined as within one hour of knowledge of the non-compliance issue.
 - 2. Child Abuse/Neglect Hotline or assigned Caseworker of any allegations of abuse or neglect made against the Caregiver within one hour of gaining knowledge of the allegation.
 - 3. Of any corrective action and the result of the correction action plan. The Provider will submit a comprehensive written report to the agency within sixty (60) days of the rules violation.
 - 4. Within twenty-four (24) hours any time there is an event which would impact the placement resource license.

See Attachment 2 for additional details.

Article VI. AGENCY RESPONSIBILITIES

- A. Agency certifies that it will comply with the Multiethnic Placement Act, 108 STAT. 3518, as amended by Section 1808 of the Small Business Jobs Protection Act of 1996, 110 STAT. 1755, which prohibits any Agency from denying any person the opportunity to become an adoptive or foster parent on the basis of race, color, national origin, or delaying or denying the placement of a child for adoption or into foster care on the basis of race, color, or national origin of the adoptive or foster parent or of the child involved.
- B. The Agency shall provide to the Provider within thirty (30) calendar days of placement or within a reasonable time thereafter as agreed to by the parties, a copy of each child's social history, medical history, and Medicaid card once obtained by the Agency for new cases, or at time of placement for existing cases. Agency shall make best efforts to share information timely regarding participants and contact information involved with planning efforts related to children and families.
- C. Agency agrees to participate in the development of the treatment plan of each child placed with the Provider. The Agency acknowledges that clinical treatment decisions must be recommended by licensed clinical professionals. Agency and Provider acknowledge that disagreement with a treatment decision may be taken through the dispute resolution process contained in Article XIV of this Agreement.
- D. Agency agrees to visit with the child in accordance with rule OAC 5101:2-42-65 of the Ohio Administrative Code.
- E. Agency agrees to participate in periodic meetings with each child's treatment team for case treatment plan development, review, and revision. The Agency agrees to participate in the development of the treatment plan of each child placed with the Provider by the Agency.
- F. Agency certifies that it will comply with Every Student Succeeds Act (34 CFR part 200) and will work with local school districts in developing individualized plans to address the transportation needed for a child to remain in the school of origin. Agency agrees to arrange for the transfer of each child's school records to the child's new school upon placement but not later than ten (10) business days. The Agency agrees to work with the Provider for the timely enrollment of the child in the receiving school district. The Agency has the final responsibility to obtain the child's school records and to enroll the child in the receiving school district.
- G. The Agency shall provide an opportunity for the Provider to give input in the development, substantive Addendum or modification of case plans. The Agency agrees to notify the Provider of scheduling no less than seven (7)

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 7 of 27 calendar days prior to of all formal meetings (e.g. SARs, court hearings, family team conferences, etc.).

- H. The Agency shall participate in a Placement Preservation meeting if requested by the Provider prior to issuing a notice of removal of a child. The Agency shall provide a minimum of thirty (30) calendar days' notice for planned removals, to the Provider for each child who is being terminated from placement with the Provider, unless so ordered by a court of competent jurisdiction.
- Agency agrees to provide the Provider with an emergency contact on a twenty-four (24) hour, seven (7) day per week basis.
- J. The Agency represents:
 - 1. It has adequate funds to meet its obligations under this Agreement; subject to the availability of funds as referenced in Article VIII (I);
 - 2. It intends to maintain this Agreement for the full period set forth herein and has no reason to believe that it will not have sufficient funds to enable it to make all payments due hereunder during such period; and
 - 3. It will make its best effort to obtain the appropriation of any necessary funds during the term of this Agreement.
- K. The Agency will provide information about the child being referred for placement in accordance with <u>OAC</u> <u>5101:2-42-90</u>.Prior to a child's placement in alternative care or respite, <u>OAC 5101:2-42-90</u> (D) requires the Agency to share with care givers information that could impact the health, safety, or well-being of the child or others in the home.

Article VII. INVOICING FOR PLACEMENT SERVICES

- A. The Provider agrees to submit a monthly invoice following the end of the month in which services were provided. The invoice shall be for services delivered in accordance with Article I of this Agreement and shall include:
 - 1. Provider's name, address, telephone number, fax number, federal tax identification number, Title IV-E Provider number, if applicable and Medicaid Provider number, if applicable.
 - 2. Billing date and the billing period.
 - Name of child, date of birth of child, and the child's Statewide Automated Child Welfare Information System (SACWIS) person I.D. number.
 - 4. Admission date and discharge date, if available.
 - 5. Agreed upon per diem for maintenance and the agreed per diem administration; and
 - 6. Invoicing procedures may also include the per diems associated with the following if applicable and agreeable to the Agency and Provider:
 - a. Case Management; allowable administration cost;
 - b. Transportation, allowable maintenance cost;
 - c. Transportation; allowable administration cost;
 - d. Other Direct Services; allowable maintenance cost;
 - e. Behavioral health care; non-reimbursable cost; and
 - f. Other costs (any other cost the Title IV-E Agency has agreed to participate in); non-allowable/ non-reimbursable cost.
- B. If Provider is an enrolled provider of Medicaid, Provider shall seek reimbursement for aftercare support provided to children through Medicaid. If a child is an open client with the QRTP the following services or activities may be billed to Medicaid as medically necessary. Aftercare support provided that is not available for Medicaid reimbursement shall be billed to the Agency. If Provider is not enrolled on Medicaid, reimbursement for aftercare support provided shall be billed to the Agency. Aftercare support provided to children who are not enrolled on Medicaid shall be invoiced to the Agency less any private insurance / third-party payor reimbursement obtained by Provider. Rates for aftercare support billed to the Agency shall be consistent with the prevailing Medicaid rate for Community Psychiatric Supportive Treatment (CPST) at the most recent version of which may be found at: Manuals and Rates (ohio.gov). If the parties agree to not use the Medicaid rates, an "Agreement for Title IV-E Agencies for the Provision of Non-Placement Services" will need to be created, and the negotiated rates will be displayed on the Schedule B.

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association 04/01/2023 - 03/31/2024 Page 8 of 27 C. Provider warrants and represents claims made for payment for services provided are for actual services rendered and do not duplicate claims made by Provider to other sources of public funds for the same service.

Article VIII. REIMBURSEMENT FOR PLACEMENT SERVICES

- A. The maximum amount payable pursuant to this contract is \$500,000.00.
- B. In accordance with Schedule A of this Agreement, the per diem for maintenance and the per diem for administration will be paid for each day the child was in placement. The first day of placement will be paid regardless of the time the child was placed. The last day of placement will not be paid regardless of the time the child left the placement.
- C. In accordance with Schedule A of this Agreement and in addition to Maintenance and Administration, the Agency may agree to pay a per diem for Case Management, Other Direct Services, Transportation Administration, Transportation Maintenance, Behavioral Health Care and Other. All other services and/or fees to be paid for shall be contained in the Attachments/Exhibits of this Agreement.
- D. To the extent that the Provider maintains a foster care network, the agreed upon per diem for maintenance shall be the amount paid directly to the foster parent. Maintenance includes the provision of food, clothing, shelter, daily supervision, graduation expenses, a child's personal incidentals, and liability insurance with respect to the child, reasonable cost of travel to the child's home for visitation and reasonable cost of travel for the child to remain in the school the child was enrolled in at the time of placement. Payment for private Agency staff transporting a child to a home visit or keeping the child in their home school will be paid in accordance with Schedule A (Transportation Maintenance) of this Agreement.
- E. If the plan as determined by the Agency is to return the child to placement with the Provider, the Agency may agree to pay for the days that a child is temporarily absent from the direct care of the Provider, as agreed to by the parties in writing.
- F. The service provider is required to utilize Medicaid-approved healthcare providers in the appropriate managed care network for the provision of mental health, dental and/or medical services (hereafter referred to collectively as "medical services") to children in the custody of Agency. The Service Provider will report applicable Medicaid/insurance information to the healthcare providers and instruct healthcare providers to seek payment from Medicaid or any other available third-party payer for medical services rendered to children in agency custody. Agency will not pay for the provision of any medical services to children in agency custody unless the agency Executive Director or authorized designee has provided specific prior written authorization for such medical services and associated costs.
- G. The Agency agrees to pay the Provider for all services agreed to on Schedule A and in the Attachments/Exhibits to this Agreement, where applicable, that have been provided and documented in the child's case file. Agency shall make best efforts to make payment of undisputed charges within thirty (30) business days of receipt.
- H. In the event of a disagreement regarding payment, Agency shall withhold payment only for that portion of the placement with which it disagrees. Agency will use best efforts to notify the Provider of any invoice discrepancies. Agency and Provider will make every effort to resolve payment discrepancies within 60 calendar days. Payment discrepancies brought to the Agency after 60 days will be reviewed on a case by case basis.
- I. This Agreement is conditioned upon the availability of federal, state, or local funds appropriated or allocated for payment for services provided under the terms and conditions of this Agreement. By sole determination of the Agency, if funds are not sufficiently allocated or available for the provision of the services performed by the Provider hereunder, the Agency reserves the right to exercise one of the following alternatives:
 - 1. Reduce the utilization of the services provided under this Agreement, without change to the terms and conditions of the Agreement; or
 - 2. Issue a notice of intent to terminate the Agreement.

The Agency will notify the Provider at the earliest possible time of such decision. No penalty shall accrue to the Agency in the event either of these provisions is exercised. The Agency shall not be obligated or liable for any future payments due or for any damages as a result of termination under this section.

04/01/2023 - 03/31/2024 Page 9 of 27 Any denial of payment for service(s) rendered may be appealed in writing and will be part of the dispute resolution process contained in Article XIV.

See Attachment 3 for additional details.

Article IX. TERMINATION; BREACH AND DEFAULT

- A. This Agreement may be terminated for convenience prior to the expiration of the term then in effect by either the Agency or the Provider upon written notification given no less than sixty (60) calendar days in advance by certified mail, return receipt requested, to the last known address of the terminated party shown hereinabove or at such other address as may hereinafter be specified in writing.
- B. If Provider fails to provide the Services as provided in this Agreement for any reason other than Force Majeure, or if Provider otherwise Materially Breaches this Agreement, Agency may consider Provider in default. Agency agrees to give Provider thirty (30) days written notice specifying the nature of the default and its intention to terminate. Provider shall have seven (7) calendar days from receipt of such notice to provide a written plan of action to Agency to cure such default. Agency is required to approve or disapprove such plan within five (5) calendar days of receipt. In the event Provider fails to submit such plan or Agency disapproves such plan, Agency has the option to immediately terminate this Agreement upon written notice to Provider. If Provider fails to cure the default in accordance with an approved plan, then Agency may terminate this Agreement at the end of the thirty (30) day notice period.
- C. Upon of the effective date of the termination, the Provider agrees that it shall cease work on the terminated activities under this Agreement, take all necessary or appropriate steps to limit disbursements and minimize costs, and furnish a report as of the date of discharge of the last child describing the status of all work under this Agreement, including without limitation, results accomplished, conclusions resulting therefrom, and such other matters as the Agency may require. The Agency agrees to remove all children in placement immediately with the Provider, consistent with the effective termination date. In all instances of termination, the Provider and Agency agree that they shall work in the best interests of children placed with the Provider to secure alternative placements for all children affected by the termination.
- D. In the event of termination, the Provider shall be entitled to reimbursement, upon submission of an invoice, for the agreed upon per diem incurred prior to the effective termination date. The reimbursement will be calculated by the Agency based on the per diem set forth in Article VIII. The Agency shall receive credit for reimbursement already made when determining the amount owed to the Provider. The Agency is not liable for costs incurred by the Provider after the effective termination date of the discharge of the last child.
- E. Notwithstanding the above, Agency may immediately terminate this Agreement upon delivery of a written notice of termination to the Provider under the following circumstances:
 - 1. Improper or inappropriate activities;
 - 2. Loss of required licenses;
 - 3. Actions, inactions or behaviors that may result in harm, injury or neglect of a child;
 - 4. Unethical business practices or procedures; and
 - 5. Any other event that Agency deems harmful to the well-being of a child; or
 - 6. Loss of funding as set forth in Article VIII.
- F. If the Agreement is terminated by Agency due to breach or default of any of the provisions, obligations, or duties embodied contained therein by the Provider, Agency may exercise any administrative, agreement, equitable, or legal remedies available, without limitation. Any extension of the time periods set forth above shall not be construed as a waiver of any rights or remedies the Agency may have under this Agreement.
- G. In the event of termination under this ARTICLE, both the Provider and the placing Agency shall make good faith efforts to minimize adverse effect on children resulting from the termination of the Agreement.

Article X. RECORDS RETENTION, CONFIDENTIALITY AND DATA SECURITY REQUIREMENTS

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 10 of 27

- A. The Provider agrees that all records, documents, writings or other information, including, but not limited to, financial records, census records, client records and documentation of legal compliance with Ohio Administrative Code rules, produced by the Provider under this Agreement, and all records, documents, writings or other information, including but not limited to financial, census and client used by the Provider in the performance of this Agreement are treated according to the following terms:
 - All records relating to costs, work performed and supporting documentation for invoices submitted to the Agency by the Provider along with copies of all Deliverables, as defined in Article XXIX, submitted to the Agency pursuant to this Agreement will be retained for a minimum of three (3) years after reimbursement for services rendered under this Agreement.
 - If an audit, litigation, or other action is initiated during the time period of the Agreement, the Provider shall retain such records until the action is concluded and all issues resolved or three (3) years have expired, whichever is later.
 - 3. All records referred to in Section A 1) of this Article shall be available for inspection and audit by the Agency or other relevant agents of the State of Ohio (including, but not limited to, the County Prosecutor, the Ohio Department of Job and Family Services (ODJFS), the Auditor of the State of Ohio, the Inspector General of Ohio, or any duly authorized law enforcement officials), and the United States Department of Health and Human Services within a reasonable period of time.
- B. The Provider agrees to keep all financial records in a manner consistent with Generally Accepted Accounting Principles.
- C. The Provider agrees to comply with all federal and state laws applicable to the Agency and the confidentiality of children and families. Provider understands access to the identities of any Agency's child and families shall only be as necessary for the purpose of performing its responsibilities under this Agreement. No identifying information on child(ren) served will be released for research or other publication without the express written consent of the Agency. Provider agrees that the use or disclosure of information concerning the child for any purpose not directly related to the administration of this Agreement is prohibited. Provider shall ensure all the children's and families' documentation is protected and maintained in a secure and safe manner.
- D. The Provider agrees to comply with all applicable state and federal laws related to the confidentiality and transmission of medical records, including, but not limited to the Health Insurance Portability and Accountability Act of 1996 (HIPAA).
- E. Although information about, and generated under, this Agreement may fall within the public domain, the Provider shall not release information about, or related to, this Agreement to the general public or media verbally, in writing, or by any electronic means without prior approval from the Agency, unless the Provider is required to release requested information by law. Agency reserves the right to announce to the general public and media: award of the Agreement, Agreement terms and conditions, scope of work under the Agreement, Deliverables, as defined in Article XXIX, and results obtained under the Agreement. Except where Agency approval has been granted in advance, the Provider shall not seek to publicize and will not respond to unsolicited media queries requesting: announcement of Agreement award, Agreement terms and conditions, Agreement scope of work, government-furnished documents the Agency may provide to the Provider to fulfill the Agreement scope of work, Deliverables required under the Agreement, results obtained under the Agreement, and impact of Agreement activities.
- F. If contacted by the media about this Agreement, the Provider agrees to notify the Agency in lieu of responding immediately to media queries. Nothing in this section is meant to restrict the Provider from using Agreement information and results to market to specific business prospects.
- G. Client data must be protected and maintained in a secure and safe manner whether located in Provider's facilities, stored in the Cloud, or used on mobile devices outside Provider's facility. Security of Provider's network, data storage, and mobile devices must conform to generally recognized industry standards and best practices. Maintenance of a secure processing environment includes, but is not limited to, network firewall provisioning, intrusion detection, antivirus protection, regular third-party vulnerability assessments, and the timely application of patches, fixes and updates to operating systems and applications.
- H. Provider agrees that it has implemented and shall maintain during the term of this Agreement the highest standard of administrative, technical, and physical safeguards and controls to:

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 11 of 27

- 1. Ensure the security and confidentiality of data;
- 2. Protect against any anticipated security threats or hazards to the security or integrity of data; and
- Protect against unauthorized access to or use of data. Such measures shall include at a minimum:
 - a. Access controls on information systems, including controls to authenticate and permit access to data only to authorized individuals and controls to prevent Provider employees from providing data to unauthorized individuals who may seek to obtain this information (whether through fraudulent means or otherwise);
 - b. Firewall protection;
 - c. Encryption of electronic data while in transit from Provider networks to external networks;
 - d. Measures to store in a secure fashion all data which shall include multiple levels of authentication;
 - e. Measures to ensure that data shall not be altered or corrupted without the prior written consent of the Agency;
 - f. Measures to protect against destruction, loss or damage of data due to potential environmental hazards, such as fire and water damage.
- I. Immediately upon discovery of a confirmed or suspected breach involving data, Provider will notify Agency no later than twenty-four (24) hours after Provider knows or reasonably suspects a breach has or may have occurred. Provider shall promptly take all appropriate or legally required corrective actions and shall cooperate fully with the Agency in all reasonable and lawful efforts to prevent, mitigate or rectify such data breach. In the event of a suspected breach, Provider shall keep the Agency informed of the progress of its investigation until the uncertainty is resolved.
- J. In the event the Provider does not carry the appropriate cyber security insurance to cover a security breach, the Provider shall reimburse the Agency for actual costs incurred, including, but not limited to, providing clients affected by a security breach with notice of the breach, and/or complimentary access for credit monitoring services, which the Agency deems necessary to protect such affected client.
- K. In the event the Agency discontinues operation, all child records for residential or any other placement settings shall be provided to the custodial agency. If the setting is licensed by ODJFS, licensing records shall be sent to:

ODJFS ATTN: Licensing P.O. Box 183204 Columbus, OH 43218-3204

Article XI. PROVIDER ASSURANCES AND CERTIFICATIONS

- A. As applicable to the Provider's license and/or certification, the Provider certifies compliance with <u>ORC 2151.86</u>, <u>ORC 5103.0328</u>, <u>ORC 5103.0319</u> and applicable OAC Sections as defined in Article XXII of this Agreement concerning criminal record checks, arrests, convictions and guilty pleas relative to foster caregivers, employees, volunteers and interns who are involved in the care for a child. Provider is responsible for any penalties, financial or otherwise, that may accrue because of noncompliance with this provision.
- B. To the extent that the Provider maintains a residential center or group home, the Provider agrees to comply with the provisions of their licensing Agency that relates to the operation, safety and maintenance of residential facilities. Specifically, Provider agrees that no firearm or other projectile weapon and no ammunition for such weapons will be kept on the premises.
- C. Provider certifies compliance with Drug Free Work Place Requirements as outlined in 45 C.F.R. Part 76,Subpart F.
- D. Provider certifies compliance with 45 C.F.R. Part 80, Non-Discrimination under programs receiving Federal assistance through the Department of Health and Human Services effectuation of Title VI of the Civil Rights Act of 1964.
- E. Provider certifies compliance with 45 C.F.R. Part 84, Non-Discrimination on the Basis of Handicap in Programs or

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 12 of 27 Activities Receiving Federal Assistance.

- F. Provider certifies compliance 45 C.F.R. Part 90, Non-Discrimination on the Basis of Age in Programs or Activities Receiving Federal Assistance.
- G. Provider certifies compliance with the American with Disabilities Act, Public Law 101-336.
- H. Provider certifies that it will:
 - 1. Provide a copy of its license(s), certification, accreditation or a letter extending an expiring license, certification, or accreditation from the issuer to the Agency prior to the signing of the Agreement.
 - 2. Maintain its license(s), certification, accreditation and that upon receipt of the renewal of its license, certification, and/or accreditation or upon receipt of a letter extending an expiring license, certification, and/or accreditation from the issuer, a copy of the license, certification and/or accreditation will be provided to the Agency within five (5) business days.
 - 3. Provider shall immediately notify the Agency of any action, modification or issue relating to said licensure, accreditation or certification.
- I. Provider certifies that it will not deny or delay services to eligible persons because of the person's race, color, religion, national origin, gender, orientation, disability, or age.
- J. The Provider shall comply with Executive Order 11246, entitled Equal Employment Opportunity, as amended by Executive Order 11375, and as supplemented in Department of Labor regulation 41 CFR part 60.
- K. Provider further agrees to comply with <u>OAC 5101:9-2-01</u> and <u>OAC 5101:9-2-05(A)(4)</u>, as applicable, which require that assure that persons with limited English proficiency (LEP) can meaningfully access services. To the extent Provider provides assistance to an LEP Child through the use of an oral or written translator or interpretation services in compliance with this requirement, the LEP Child shall not be required to pay for such assistance.
- L. To the extent applicable, the Provider certifies compliance with all applicable standards, orders, or requirements issued under Section 306 of the Clean Air Act (42 U.S.C. 1857 (h) Section 508 of the Clean Water Act (33 U.S.C. 1368), Executive Order 11738, and Environmental Protection Agency Regulations (40 C.F.R. Part 15).
- M. The Provider certifies compliance, where applicable, with mandatory standards and policies relating to energy efficiency which are contained in the state energy conservation plan issued in compliance with the Energy Policy and Conservation Act (Pub. L. 94-163, 89 Stat. 871).
- N. The Provider certifies that all approvals, licenses, or other qualifications necessary to conduct business in Ohio have been obtained and are current.
- O. Provider shall comply with the Small Business Job Protection Act (Public Law ("P.L.") 104-188), the Multiethnic Placement Act of 1994 (P.L. 103-382), Titles IV-B (42 U.S.C. 620 et seq.) and IV-E (42 U.S.C. 670 et seq.) of the Social Security Act ("the Act"), the Personal Responsibility and Work Opportunity Reconciliation Act of 1996 (P.L. 104-193), Section 471(a) of Title IV-E of the Act (42 U.S.C. 671(a)), and 45 C.F.R. 1356, including all rules, regulations and guidelines issued by federal and state authorities, <u>OAC 5101:9-4-07</u> and <u>OAC 5101:2-47-23.1</u>.

Article XII. INDEPENDENT CONTRACTOR

- A. The Provider and the Agency agree that no employment, joint venture, or partnership has been or will be created between the parties hereto pursuant to the terms and conditions of this Agreement.
- B. The Provider and the Agency agree that the Provider is an independent contractor and assumes all responsibility for any federal, state, municipal, or other tax liabilities along with workers' compensation, unemployment compensation, and insurance premiums which may accrue as a result of compensation received for services or Deliverables rendered hereunder.
- C. The Provider and the Agency agree that no person and/or entities entering into this Agreement, nor any individual employed by any person or entity entering in to this Agreement, are public employees for purposes of contributions to Ohio Public Employees Retirement system by virtue of any work performed or services rendered

04/01/2023 - 03/31/2024 Page 13 of 27 in accordance with this Agreement.

Article XIII. AUDITS AND OTHER FINANCIAL MATTERS

- A. Provider agrees to submit to Agency a copy of the independent audit it receives in accordance with <u>ORC</u> <u>5103.0323.</u>
- B. Upon request from the Agency, Provider shall submit a copy of the most recent Federal income tax return and related schedules filed with the Internal Revenue Service (IRS).
- C. If Provider participates in the Title IV-E program, Provider agrees to timely file its Title IV-E cost report with all required items as outlined in <u>OAC 5101:2-47-26.2</u> to ODJFS. Provider agrees that in the event a cost report cannot be timely filed, an extension shall be requested prior to the December 31st filing deadline.
- D. If a Provider participates in the Title IV-E program, an Agreed Upon Procedures engagement must be conducted by a certified public accountant for the Provider's cost report in accordance with <u>OAC 5101:2-47-26.2.</u> The procedures are conducted to verify the accuracy of costs used to establish reimbursement ceilings for maintenance and administration costs of child in care. Any overpayments or underpayment of federal funds to the Title IV-E Agency due to adjustments of cost report reimbursement ceiling amounts as a result of an audit, shall be resolved in accordance with <u>ORC 5101.11</u>, <u>ORC 5101.14</u>, and <u>OAC 5101:2-47-01</u>.
- E. Upon request from the Agency, the Provider shall submit a copy of the JFS 02911 and Agreed Upon Procedures.
- F. For financial reporting purposes and for Title IV-E cost reporting purposes, Provider agrees to follow the cost principles set forth in the following OAC Sections and publications:
 - <u>OAC 5101:2-47-11</u>: "Reimbursement for Title IV-E foster care maintenance (FCM) costs for children's residential centers (CRC), group homes, maternity homes, residential parenting facilities, private foster homes, and substance use disorder (SUD) residential facilities".
 - <u>OAC 5101:2-47-26.1</u>: "Public child services agencies (PCSA), private child placing agencies (PCPA), private noncustodial agencies (PNA), residential care facilities, substance use disorder (SUD) residential facilities: Title IV-E cost report filing requirements, record retention requirements, and related party disclosure requirements";
 - 3. OAC 5101:2-47-26.2: "Cost Report Agreed Upon Procedures Engagement".
 - 4. JFS 02911 Single Cost Report Instructions.
 - 5. For Private Agencies: 2 CFR part 230, Cost Principles for Non-Profit Organizations.
 - 6. For Public Agencies: 2 CFR part 225, Cost Principles for State, Local and Indian Tribal Government.
 - 7. 2 CFR part 200.501, Audit Requirements.

Article XIV. GRIEVANCE/DISPUTE RESOLUTION PROCESS

In the event that a dispute arises under the provisions of this Agreement, the parties shall follow the procedures set forth below:

- 1. The party complaining of a dispute shall provide written notice of the nature of the dispute to the other party to this Agreement. A copy of the notice shall be sent to the Director or designee of the Agency and to the Executive Director or designee of the Provider. Within ten (10) business days of receiving the notice of a dispute, the parties involved in the dispute between the Agency and the Provider shall attempt to resolve the dispute.
- 2. If the parties are unable to resolve the dispute in (1 business day), the highest official or designee of the Agency shall make the final determination within twenty (20) business days, which will be non-binding.
- 3. Neither party will be deemed to have waived any other rights or remedies available to them by initiating, participating in or completing this process.

Article XV. ATTACHMENTS/ADDENDA

This Agreement, Attachments, and all Exhibits hereto constitutes the entire Agreement and may be amended only with a written Addendum signed by both parties; however, it is agreed by the parties that any Addenda to laws or regulations cited herein will result in the correlative modification of this Agreement, without the necessity for executing written

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association 04/01/2023 - 03/31/2024 Page 14 of 27 Addenda. The impact of any applicable law, statute, or regulation not cited herein and enacted after the date of execution of this Agreement will be incorporated into this Agreement by written Addendum signed by both parties and effective as of the date of enactment of the law, statute, or regulation. Any other written Addendum to this Agreement is prospective in nature.

Article XVI. NOTICE

Unless otherwise set forth herein, all notices, requests, demands and other communications pertaining to this Agreement shall be in writing and shall be deemed to have been duly given if delivered or mailed by certified or registered mail, postage pre-paid:

if to Agency, to	Fairfield County Department of Job and Family Services 239 W Main St Lancaster, OH 43130
if to Provider, to	Ohio Teaching Family Association 4020 Waterville Swanton Rd 300 Swanton, OH 43558

Article XVII. CONSTRUCTION

This Agreement shall be governed, construed, and enforced in accordance with the laws of the State of Ohio. Should any portion of this Agreement be found to be unenforceable by operation of statute or by administrative or judicial decision, the operation of the balance of this Agreement is not affected thereby; provided, however, the absence of the illegal provision does not render the performance of the remainder of the Agreement impossible.

Article XVIII. NO ASSURANCES

- A. Provider acknowledges that, by entering into this Agreement, Agency is not making any guarantees or other assurances as to the extent, if any, that Agency shall utilize Provider's services or purchase its goods. In this same regard, this Agreement in no way precludes, prevents, or restricts Provider from obtaining and working under additional arrangement(s) with other parties, assuming the work in no way impedes Provider's ability to perform the services required under this Agreement. Provider warrants that at the time of entering into this Agreement, it has no interest in nor shall it acquire any interest, direct or indirect, in any Agreement that will impede its ability to provide the goods or perform the services under this Agreement.
- B. This Agreement, Attachments, and all Exhibits embodies the entire agreement of the Parties. There are no promises, terms, conditions or obligations other than those contained herein; and this Agreement shall supersede all previous communications, representations or Agreements, either written or oral, between the parties to this Agreement. Also, this Agreement shall not be modified in any manner except by an instrument, in writing, executed by both the parties.

Article XIX. CONFLICT OF INTEREST

- A. Provider agrees that the Provider, its officers, members and employees currently have no, nor will they acquire any interest, whether personal, professional, direct or indirect, which is incompatible, in conflict with or which would compromise the discharge and fulfillment of Provider's functions, duties and responsibilities hereunder. If the Provider, or any of its officers, members or employees acquire any incompatible, conflicting, or compromising personal or professional interest, the Provider shall immediately disclose, in writing, such interest to the Agency. If any such conflict of interest develops, the Provider agrees that the person with the incompatible, conflicting, or compromising personal or professional interest will not participate in any activities related to this Agreement.
- B. Provider agrees: (1) to refrain from promising or giving to Agency employees anything of value to manifest improper influence upon the employee; (2) to refrain from conflicts of interest; and, (3) to certify that Provider complies with <u>ORC 102.03</u>, <u>ORC 102.04</u>, <u>ORC 2921.42</u>, <u>ORC 2921.43</u>.

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 15 of 27 C. The Provider further agrees that there is no financial interest involved on the part of the Agency or the respective county authority(ies) governing the agency. The Provider has no knowledge of any situation which would be a conflict of interest. It is understood that a conflict of interest occurs when an Agency employee or county official will gain financially or receive personal favors as a result of signing or implementation of this agreement. The Provider will report the discovery of any potential conflict of interest to the Agency. Should a conflict of interest be discovered during the term of this agreement, the Agency may exercise any right under the agreement, including termination of the agreement.

Article XX. INSURANCE

The Provider shall purchase and maintain for the term of this Agreement insurance of the types and amounts identified herein. Maintenance of the proper insurance for the duration of the Agreement is a material element of the Agreement.

Provider agrees to procure and maintain for the term of this Agreement the insurance set forth herein. The cost of all insurance shall be borne by Provider. Insurance shall be purchased from a company licensed to provide insurance in Ohio. Insurance is to be placed with an insurer provided an A.M. Best rating of no less than A-. Provider shall purchase the following coverage and minimum limits:

- A. Commercial general liability insurance policy with coverage contained in the most current Insurance Services Office Occurrence Form CG 00 01 or equivalent with limits of at least One Million Dollars (\$1,000,000.00) per occurrence and One Million Dollars (\$1,000,000.00) in the aggregate and at least One Hundred Thousand Dollars (\$100,000.00) coverage in legal liability fire damage. Coverage will include:
 - 1. Additional insured endorsement;
 - 2. Product liability;
 - 3. Blanket contractual liability;
 - 4. Broad form property damage;
 - 5. Severability of interests;
 - 6. Personal injury; and
 - 7. Joint venture as named insured (if applicable).

Endorsements for physical abuse claims and for sexual molestation claims must be a minimum of Three Hundred Thousand Dollars (\$300,000.00) per occurrence and Three Hundred Thousand Dollars (\$300,000.00) in the aggregate.

- B. Business auto liability insurance of at least One Million Dollars (\$1,000,000.00) combined single limit, on all owned, non-owned, leased and hired automobiles. If the Agreement contemplates the transportation of the users of County services (such as but not limited to Agency consumers), "Consumers" and Provider provides this service through the use of its employees' privately owned vehicles "POV", then the Provider's Business Auto Liability insurance shall sit excess to the employees "POV" insurance and provide coverage above its employee's "POV" coverage. Provider agrees the business auto liability policy will be endorsed to provide this coverage.
- C. Professional liability (errors and omission) insurance of at least One Million Dollars (\$1,000,000.00) per claim and in the aggregate.
- D. Umbrella and excess liability insurance policy with limits of at least One Million Dollars (S1,000,000.00) per occurrence and in the aggregate, above the commercial general and business auto primary policies and containing the following coverage:
 - 1. Additional insured endorsement;
 - 2. Pay on behalf of wording;
 - 3. Concurrency of effective dates with primary;
 - Blanket contractual liability;
 - Punitive damages coverage (where not prohibited by law);
 - 6. Aggregates: apply where applicable in primary;
 - 7. Care, custody and control follow form primary; and
 - Drop down feature.

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 16 of 27 The amounts of insurance required in this section for General Liability, Business Auto Liability and Umbrella/Excess Liability may be satisfied by Provider purchasing coverage for the limits specified or by any combination of underlying and umbrella limits, so long as the total amount of insurance is not less than the limits specified in General Liability, Business Auto Liability and Umbrella/Excess Liability when added together.

- E. Workers' Compensation insurance at the statutory limits required by ORC.
- F. The Provider further agrees with the following provisions:
 - 1. All policies, except workers' compensation and professional liability, will endorse as additional insured the Board of County Commissioners, and Agency and their respective officials, employees, agents, and volunteers, including their Board of Trustees if applicable. The additional insured endorsement shall be on an ACORD or ISO form.
 - 2. The insurance endorsement forms and the certificate of insurance forms will be sent to the Agency Director or Designee. The forms must state the following: "Board of County Commissioners, and Agency and their respective officials, employees, agents, and volunteers are endorsed as additional insured as required by agreement on the commercial general, business auto and umbrella/excess liability policies."
 - 3. Each policy required by this clause shall be endorsed to state that coverage shall not be canceled or materially changed except after thirty (30) calendar days prior written notice given to the Agency Director or Designee.
 - 4. Provider shall furnish the Agency with original certificates and amendatory endorsements effecting coverage required by this clause. All certificates and endorsements are to be received by the Agency before the Agreement commences. The Agency reserves the right at any time to require complete, certified copies of all required insurance policies, including endorsements affecting the coverage required by these specifications.
 - 5. Failure of the Agency to demand such certificate or other evidence of full compliance with these insurance requirements or failure of the Agency to identify a deficiency from evidence provided shall not be construed as a waiver of Provider's obligation to maintain such insurance.
 - 6. Provider shall declare any self-insured retention to the Agency pertaining to liability insurance. Provider shall provide a financial guarantee satisfactory to the Agency guaranteeing payment of losses and related investigations, claims administration and defense expenses for any self-insured retention.
 - 7. If Provider provides insurance coverage under a "claims-made" basis, Provider shall provide evidence of either of the following for each type of insurance which is provided on a claims-made basis: unlimited extended reporting period coverage, which allows for an unlimited period of time to report claims from incidents that occurred after the policy's retroactive date and before the end of the policy period (tail coverage), or; continuous coverage from the original retroactive date of coverage. The original retroactive date of coverage means original effective date of the first claim-made policy issued for a similar coverage while Provider was under Agreement with the County on behalf of the Agency.
 - 8. Provider will require all insurance policies in any way related to the work and secured and maintained by Provider to include endorsements stating each underwriter will waive all rights of recovery, under subrogation or otherwise, against the County and the Agency. Provider will require of subcontractors, by appropriate written agreements, similar waivers each in favor of all parties enumerated in this section.
 - 9. Provider, the County, and the Agency agree to fully cooperate, participate, and comply with all reasonable requirements and recommendations of the insurers and insurance brokers issuing or arranging for issuance of the policies required here, in all areas of safety, insurance program administration, claim reporting and investigating and audit procedures.
 - 10. Provider's insurance coverage shall be primary insurance with respect to the County, the Agency, their respective officials, employees, agents, and volunteers. Any insurance maintained by the County or the Agency shall be excess of Provider's insurance and shall not contribute to it.
 - 11. If any of the work or Services contemplated by this Agreement is subcontractors, Provider will ensure that any subcontractors comply with all insurance requirements contained herein.
 - 12. If the Agreement provider is a government entity, insurance requirements will be fulfilled under the County Risk Sharing Authority (CORSA).

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 17 of 27

Article XXI. INDEMNIFICATION & HOLD HARMLESS

- A. To the fullest extent permitted by, and in compliance with, applicable law, Provider agrees to protect, defend, indemnify and hold harmless the Agency and the Board of County Commissioners, their respective members, officials, employees, agents, and volunteers (the "Indemnified Parties") from and against all damages, liability, losses, claims, suits, actions, administrative proceedings, regulatory proceedings/hearings, judgments and expenses, subrogation (of any party involved in the subject of this Agreement), attorneys' fees, court costs, defense costs or other injury or damage (collectively "Damages"), whether actual, alleged or threatened, resulting from injury or damages of any kind whatsoever to any business, entity or person (including death), or damage to property (including destruction, loss of, loss of use of resulting without injury damage or destruction) of whatsoever nature, arising out of or incident to in any way, the performance of the terms of this Agreement including, without limitation, by Provider, its subcontractor(s), Provider's or its subcontractor(s') employees, agents, assigns, and those designated by Provider to perform the work or services encompassed by the Agreement. Provider agrees to pay all damages, costs and expenses of the Indemnified Parties in defending any action arising out of the aforementioned acts or omissions.
- B. Each Party agrees to be responsible for any personal injury or property damage caused solely by its negligent acts or omissions as determined by a court of competent jurisdiction, or as the parties may otherwise mutually agree in writing.
- C. This Article is not applicable to Agreements between governmental entities.

Article XXII. SCREENING AND SELECTION

- A. Criminal Record Check
 - Provider warrants and represents it will comply with Article X as it relates to criminal record checks. Provider shall insure that every individual subject to a Bureau of Criminal Investigation (BCI) criminal records check will sign a release of information to allow inspection and audit of the above criminal records transcripts or reports by the Agency or a private vendor hired by the Agency to conduct compliance reviews on their behalf.
 - 2. Provider shall not assign any individual to work with or transport children until a BCI report and a criminal record transcript has been obtained.
 - 3. Except as provided in Section C below, Provider shall not utilize an employee, foster caregiver or all of the above who has been convicted or plead guilty to any violations contained in <u>ORC 5153.111(B)(1)</u>. <u>ORC 2919.24, and ORC 2151.86, and <u>OAC Chapters 5101:2-5, 5101:2-7, 5101:2-9, 5101:2-48.</u></u>
 - Provider agrees to be financially responsible for any of the following requirements in <u>OAC Chapters</u> <u>5101:2-5, 5101:2-7, 5101:2-9 and 5101:2-48</u> resulting in financial penalty due to lack of compliance with the criminal records checks.
- B. Transportation of Child
 - 1. The caregiver shall ensure the transportation of children in care will be reliable, legal and safe transportation with safety restraints, as appropriate for the child, and must be in compliance with applicable local, state and Federal transportation laws:
 - a. Maintenance of a current valid driver's license and vehicle insurance.
 - b. All children being transported by Provider must follow Ohio's Child Passenger Safety Law as defined in <u>ORC 4511.81.</u>
 - c. No child that is a passenger and is required to have a seat restraint can be transported by said provider until these requirements are met.
 - 2. In addition to the requirements set forth above, Provider shall not permit any individual to transport a Child if:
 - a. The individual has a condition which would affect safe operation of a motor vehicle;
 - b. The individual has six (6) or more points on his/her driver's license; or
 - c. The individual has been convicted of, or pleaded guilty to, a violation of section 4511.19 (Operating

Contract ID: 19340932 04 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 18 of 27 vehicle under the influence of alcohol or drugs – OVI or OVUAC) of the Revised Code if the individual previously was convicted of or plead guilty to two or more violations within the three years immediately preceding the current violation.

C. Rehabilitation

- 1. Notwithstanding the above, Provider may make a request to the Agency to utilize an individual if Provider believes the individual has met the rehabilitative standards of <u>OAC 5101:2-07-02(I)</u> as follows:
 - a. If the Provider is seeking rehabilitation for a foster caregiver, a foster care applicant or other resident of the foster caregiver's household, Provider must provide written verification that the rehabilitation standards of <u>OAC 5101:2-7-02</u> have been met.
 - b. If the Provider is seeking rehabilitation for any other individual serving Agency children, Provider must provide written verification from the individual that the rehabilitative conditions in accordance with <u>OAC 5101:2-5-09</u> have been met.
- The Agency shall review the facts presented and may allow the individual to work with, volunteer with or transport Agency children on a case-by-case basis. It is the Agency's sole discretion to permit a rehabilitated individual to work with, volunteer with or transport children.
- D. Verification of Job or Volunteer Application:

Provider shall check and document each applicant's personal and employment references, general work history, relevant experience, and training information. Provider further agrees it will not employ an individual in relation to this Agreement unless it has received satisfactory employment references, work history, relevant experience, and training information.

Article XXIII. PROHIBITION OF CORPORAL & DEGRADING PUNISHMENT

Agency prohibits the use of corporal or degrading punishment against children served by Agency and must comply with requirements in <u>OAC 5101:2-7-09</u>, OAC 5101:2-9-21, and <u>OAC 5101:2-9-22</u>

Article XXIV. FINDINGS FOR RECOVERY

<u>ORC 9.24</u> prohibits public agencies from awarding an Agreement for goods, services, or construction paid for in whole or in part from federal, state and local funds, to an entity against whom a finding for recovery has been issued if the finding is unresolved. By entering into this Agreement, Provider warrants and represents that they do not have an unresolved finding for recovery. Provider shall notify the Agency within ten (10) business days of its notification should the Provider be issued such finding by the Auditor of the State.

Article XXV. PUBLIC RECORDS

This Agreement is a matter of public record under the Ohio public records law. By entering into this Agreement, Provider acknowledges and understands that records maintained by Provider pursuant to this Agreement may also be deemed public records and subject to disclosure under Ohio law. Upon request made pursuant to Ohio law, the Agency shall make available the Agreement and all public records generated as a result of this Agreement.

Article XXVI. CHILD SUPPORT ENFORCEMENT

Provider agrees to cooperate with ODJFS and any Ohio Child Support Enforcement Agency ("CSEA") in ensuring Provider and Provider's employees meet child support obligations established under state or federal law. Further, by executing this Agreement, Provider certifies present and future compliance with any court or valid administrative order for the withholding of support which is issued pursuant to the applicable sections in <u>ORC Chapters 3119, 3121, 3123</u>, and <u>3125</u>.

Article XXVII. DECLARATION OF PROPERTY TAX DELINQUENCY

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association 04/01/2023 - 03/31/2024 Page 19 of 27 After award of an Agreement, and prior to the time the Agreement is entered into, the successful Provider shall submit a statement in accordance with <u>ORC 5719.042</u>. Such statement shall affirm under oath that the person with whom the Agreement is to be made was not charged at the time the bid was submitted with any delinquent personal property taxes on the general tax list of personal property of any county in which the taxing district has territory, or that such person was charged with delinquent personal property taxes on any such tax list, in which case the statement shall also set forth the amount of such due and unpaid delinquent taxes any due and unpaid penalties and interest thereon. If the statement indicates that the taxpayer was charged with any such taxes, a copy of the statement shall be transmitted by the fiscal officer to the county treasurer within thirty days of the date it is submitted.

A copy of the statement shall also be incorporated into the Agreement, and no payment shall be made with respect to any contract to which this section applies unless such statement has been so incorporated as a part thereof.

Article XXVIII. SUBCONTRACTING AND DELEGATION

The performance of any duty, responsibility or function which is the obligation of the Provider under this Agreement may be delegated or subcontracted to any agent or subcontractor of Provider if Provider has obtained the prior written consent of the Agency for that delegation subcontract. Provider is responsible for ensuring that the duties, responsibilities or functions so delegated or subcontracted are performed in accordance with the provisions and standards of this Agreement, and the actions and omissions of any such agent or subcontractor shall be deemed to be the actions and omissions of Provider for purposes of this Agreement.

Article XXIX. PROPERTY OF AGENCY

The Deliverable(s) and any item(s) provided or produced pursuant to this Agreement (collectively called "Deliverables") will be considered "works made for hire" within the meaning of copyright laws of the United States of America and the State of Ohio. The Agency is the sole author of the Deliverables and the sole owner of all rights therein. If any portion of the Deliverables are deemed not to be a "work made for hire", or if there are any rights in the Deliverables not so conveyed to the Agency, then Provider agrees to, and by executing this Agreement hereby does, assign to the Agency all worldwide rights, title, and interest in and to the Deliverables. The Agency acknowledges that its sole ownership of the Deliverables under this Agreement does not affect Provider's right to use general concepts, algorithms, programming techniques, methodologies, or technology that have been developed by Provider prior to this Agreement or that are generally known and available. Any Deliverable provided or produced by Provider under this Agreement or with funds hereunder, including any documents, data, photographs and negatives, electronic reports/records, or other media, are the property of the Agency, which has an unrestricted right to reproduce, distribute, modify, maintain, and use the Deliverables. Provider shall not obtain copyright, patent, or other proprietary protection for the Deliverables. Provider shall not include in any Deliverable any copyrighted material, unless the copyright owner gives prior written approval for the Agency and Provider to use such copyrighted material. Provider agrees that all Deliverables will be made freely available to the general public unless the Agency determines that, pursuant to state or federal law, such materials are confidential or otherwise exempt from disclosure.

Article XXX. SEVERABILITY

If any term of this Agreement or its application thereof to any person or circumstance shall to any extent be held invalid or unenforceable, the remainder of this Agreement, or the application of such term or provision to persons or circumstances other than those as to which it is held invalid or unenforceable, shall not be affected thereby. Each term and provision of this Agreement shall be valid and enforced to the fullest extent permitted by law.

Article XXXI. NO ADDITIONAL WAIVER IMPLIED

If the Agency or Provider fails to perform any obligations under this Agreement and thereafter such failure is waived by the other party, such waiver shall be limited to the particular matter waived and shall not be deemed to waive any other failure hereunder, nor a waiver of a subsequent breach of the same provision or condition. Waivers shall not be effective unless in writing.

Article XXXII. COUNTERPARTS

This Agreement may be executed as an original document only, or simultaneously in two or more counterparts, each of

Contract ID: 19340932 04/01/2023 - 03/31/2024 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association Page 20 of 27 which shall be deemed an original, and each of these counterparts shall constitute one and the same instrument. It shall not be necessary in making proof of this Contract to produce or account for more than one such counterpart. An electronic signature or a scanned or otherwise reproduced signature shall be a binding signature and carry the same legal force as the original.

Article XXXIII. APPLICABLE LAW AND VENUE

This Agreement and any modifications, Attachments, Exhibits, Addenda, or alterations, shall be governed, construed, and enforced under the laws of Ohio. Any legal action brought pursuant to this agreement will be filed in the Ohio courts, and Ohio law as well as Federal law will apply.

ATTACHMENT

Attachment One.

Reason: Article Section: Article I - Scope of Placement Services Detail: Article I SECTIONS 1.02 & 1.03, References to Exhibit I Article I, Item A ('Scope of Placement Services') will serve as Exhibit I.

Attachment Two.

Reason: Article Section: Article V - Provider Responsibilities Detail: Article V ITEM A

TEM A Provider is responsible for

Provider is responsible for ensuring transportation services are in place for all case-related activities and routine needs. If extenuating circumstances exist and provider is not able to utilize its own resources to ensure coverage, Provider must notify Agency in advance of the transportation need, and coordinate with Agency to arrange needed transportation. Provider should bring any extraordinary travel needs to the attention of Agency so both parties can ensure proper coverage and explore potential compensation for needed transportation services.

ITEM B

Progress reports, demonstration completion of monthly activities as required by Ohio Administrative Code, will be submitted on a monthly basis, by no later than the 20th calendar day of the following month.

ITEMS D, E, & F

After-Hours/On-Call Process

Provider shall notify Agency of any items identified in Article V, according to the following options:

For calls during business hours (Monday through Friday, from 8:00 a.m. to 4:00 p.m., excluding holidays), Provider shall call (740) 652-7854 and inform the operator of the need to urgently speak to casework staff.

For calls outside of business hours, Provider shall call (740)808-0009 or (740)808-0982 in order to notify Agency's On-Call staff member of the urgent situation.

Insert new item - ITEM S

Provider will ensure access to Normalcy activities, based upon the developmental, social, and emotional functioning of each child placement.

Insert new item - ITEM T

All Provider staff, and foster caregivers when applicable, should seek and receive prior authorization from Agency for any type of non-routine medical care or medication needs. This includes, but is not limited to: major medical treatment, medical procedures, surgery, implementation of or change in psychotropic medications, and any other medical intervention that carries a high risk of side effects, impairment, or harm. Routine well visits and treatment for typical childhood illnesses will not require such prior authorization.

In the event of an emergency, the child should be taken to the nearest medical facility for prompt treatment. As soon as possible, Agency shall be contacted, according to the process outlined

Attachment Three.

Reason: Article

Section: Article VIII - Reimbursement for Placement Services Detail: Article VIII

ITEM B

Agency agrees to pay Provider on the basis of a daily per diem (identified in Schedule A of this agreement) for the placement for each child, as identified by each child's current Level of Care (LOC). The LOC will be agreed upon at

Contract ID: 19340932

Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024

Page 22 of 27

the time of placement. The Agency and Provider may request a re-evaluation of the child's LOC at any time in order to best meet the child's identified needs.

ITEM D

Agency agrees to provide a one-time initial clothing authorization of up to \$150.00 for children ten and under and up to \$250.00 for children over the age of ten. If the child is under ten and wearing adult sizes, the authorization will be for \$250.00. Any purchases beyond this must be due to extenuating circumstances and approved in writing, in advance of the purchase. Routine clothing needs are considered part of maintenance and will not be paid or reimbursed by Agency.

ITEM E

If a child goes on any form of unpaid leave and is reasonably expected to return to the same placement, Agency may pay for up to seven (7) consecutive nights of leave. In order for Agency to pay for such leave, Provider (including foster parents, if applicable) must agree to remain available for regular services and needed support during such leave. Leave beyond seven (7) days will not be paid unless extenuating circumstances exist, and both Agency and Provider agree to payment terms in advance, in writing.

Regardless of length, Agency will not pay Provider for any leave during which payment is being made to another provider, nor for leave where the child is in a paid or unpaid alternative placement outside of Provider's network due to a lack of placement availability with Provider. Any deviation from this must be agreed to in advance, in writing, by both Agency and Provider.

Item F

Medicaid/Insurance

Upon receipt of formal documentation, FCCPS will submit to Service Provider Medicaid/insurance numbers for children in FCCPS custody, as applicable. The service provider is required to utilize Medicaid-approved healthcare providers in the appropriate managed care network for the provision of mental health,dental and/or medical services (hereafter referred to collectively as "medical services") to children in the custody of FCCPS. The Service Provider will report applicable Medicaid/insurance information to the healthcare providers and instruct healthcare providers to seek payment from Medicaid or any other available third party payor for medical services rendered to children in FCCPS custody. FCCPS will not pay for the provision of any medical services to children in FCCPS custody unless the FCCPS Deputy Director or authorized designee has provided specific prior written authorization for such medical services and associated costs.

In situations where the Service Provider does not possess a Medicaid/insurance number or other information required to bill an alternative source for services provided to children in the custody of FCCPS, the Service Provider must take the following actions.

A. The Service Provider will contact the Finance Department at (740)652-7889 for assistance with resolving Medicaid/insurance number issues.

B. Within thirty (30) days if an invoice from a healthcare provider for services rendered to a child in FCCPS custody, the Service Provider should forward the invoice to the FCCPS Finance Department at: Fairfield County Job and Family Services-Protective Services, 239 West Main St., Lancaster, OH 43130 or fax such invoice to the FCCPS Finance Department at (740)-687-7070. Failure to forward this invoice to FCCPS within thirty (30)days will constitute a waiver of any claim against FCCPS for payment of the invoice. If the Service Provider receives additional notices regarding the invoice, the Service Provider must contact the FCCPS Finance Department at (740)-652-7889 to confirm that FCCPS received the initial invoice and to obtain the status of payment arrangements. The Service Provider SHALL NOT pay the invoice and expect or request reimbursement from FCCPS without the prior written approval of FCCPS.

C. If a child who is in custody of FCCPS requires pharmaceutical supplies, Service Provider must obtain the supplies from a pharmacy that accepts Medicaid/insurance payments.

ITEM G

In any instance where payment cannot be made within 30 days, Agency will make every effort to ensure Provider is paid within 45 days, and is made aware in advance if this is not possible.

Contract ID: 19340932

Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 23 of 27 FCCPS retains the right to recoup funds from the Service Provider upon the determination that third party funds are duplicative (in the aggregate) of FCCPS payments to the Service Provider, or in the event that the Service Provider fails to properly credit any and all such third party payments. Relative to recouping funds, FCCPS may withhold from subsequent reimbursement to the Service Provider an amount equal to any un-credited or duplicate third party payments. For purposes of this paragraph, "third party" includes, but is not limited to , Medicaid and private insurance companies.

The Service Provider shall obtain and provide a written estimate for any non-routine, non-emergency, or out-ofnetwork medical and dental expenses to FCCPS along with the written recommendation of the physician or dentist. The Servide Provider is not permitted to deliver or authorize any health/dental care or treatment services (including, but not limited to, mental health services), without the prior written consent of the FCCPS Deputy Director or authorized designee (see Consent for Medical Treatment letter).

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date of the signature of the parties.

SIGNATURES OF PARTIES:

Provider: Ohio Teaching Family Association

Print Name & Title	Signature	Date
Jonathan Arnold Business Manager	Justan	4-24-23
<u> </u>		

Agency: Fairfield County Department of Job and Family Services

Print Name & Title	Signature	Date

Additional Signatures

 Print Name & Title	Signature	Date

Contract ID: 19340932 Fairfield County Department of Job and Family Services / Ohio Teaching Family Association

04/01/2023 - 03/31/2024 Page 25 of 27

Title IV-E Schedule A Rate Information

Title IV-E Schedule A Rate Information Agency: Fairfield County Department of Job and Family Services Provider / ID: Ohio Teaching Family Association / 24419

Run Date: 04/21/2023 Contract Period: 04/01/2023 - 03/31/2024

Service Description	Service ID	Person	Person ID	Maintenance Per Diem	Administration Per Diem	Case Management Per Diem	Transportation / Administration Per Diem	Transporation / Maintenance Per Diem	Other Direct Services	Behavioral Healthcare Per Diem	Other Per Diem	Total Per Diem	Cost Begin Date	Cost End Date
									Per Diem		Cost	Cost		
Doug & Sylvia Lee GROUP Home (20396)	107845			\$316.00	\$24.00					\$10.00		\$350.00	04/01/2023	03/31/2024
Doug & Sylvia Lee GROUP Home (20396)	107845			\$326.00	\$24.00				×			\$350.00	04/01/2023	03/31/2024
Farris GROUP Home(20444)	107847			\$316.00	\$24.00					\$10.00		\$350.00	04/01/2023	03/31/2024
Farris GROUP Home(20444)	107847			\$326.00	\$24.00							\$350.00	04/01/2023	03/31/2024
Hartman GROUP Home(20525)	107848			\$345.00	\$24.00					\$10.00		\$379.00	04/01/2023	03/31/2024
Hartman GROUP	107848			\$345.00	\$24.00							\$369.00	04/01/2023	03/31/2024
Home(20525)														
Independent Living	65153			\$504.00	\$25.00							\$529.00	04/01/2023	03/31/2024
Jackson Group Home (20815)	4541664			\$316.00	\$24.00					\$10.00		\$350.00	04/01/2023	03/31/2024
Jackson Group Home (20815)	4541664			\$326.00	\$24.00							\$350.00	04/01/2023	03/31/2024

Title IV-E Schedule A Rate Information

Title IV-E Schedule A Rate Information Agency: Fairfield County Department of Job and Family Services Provider / ID: Ohio Teaching Family Association / 24419

Run Date: 04/21/2023 Contract Period: 04/01/2023 - 03/31/2024

Service Description	Service ID	Person	Person ID	Maintenance Per Diem	Administration Per Diem	Case Management Per Diem	Transportation / Administration Per Diem	Transporation / Maintenance Per Diem	Other Direct Services	Behavioral Healthcare Per Diem	Other Per Diem	Total Per Diem	Cost Begin Date	Cost End Date
									Per Diem		Cost	Cost		
Porter Group Home (20853)	65151			\$316.00	\$24.00					\$10.0D		\$350.00	04/01/2023	03/31/2024
Porter Group Home (20853)	65151			\$326.00	\$24.00							\$350.00	04/01/2023	03/31/2024
Waller Group Home (20796)	3565663			\$316.00	\$24.00					\$10 .00		\$350.00	04/01/2023	03/31/2024
Waller Group Home (20796)	3565663			\$326.00	\$24.00							\$350.00	04/01/2023	03/31/2024
Waller Group Home (20796)	3565663			\$405.00	\$25.00							\$430.00	04/01/2023	03/31/2024



A Contract regarding Ohio Teaching Family Association between Job and Family Services and

Approved on 5/30/2023 8:01:18 AM by Sarah Fortner, Assistant Deputy Director

South Jota

Sarah Fortner Assistant Deputy Director

Approved on 5/30/2023 11:15:22 AM by Corey Clark, Director of Fairfield County Job & Family Services

GBCh_

Corey Clark, Director Fairfield County Job & Family Services

REPRINT		Carrí L. Brown, phd, mba, cgfm	Purchase Order				
		Fairfield County Auditor	Fiscal Year 2023	Page: 1 of 1			
		210 East Main Street Lancaster, Ohio 43130	THIS NUMBER MUST APPEAR ON ALL INVOICES, PACKAGES AND SHIPPING PAPERS.				
В	JOB & FAMILY SERVICES	Revisions: 003	Purchase Order #	23000422 - 03			
 L L	239 W MAIN STREET LANCASTER, OH 43130 Phone: 740-652-7889		Delivery must be made within doors of specified destination.				
т О			Expiration Date	: 12/15/2023			
VENDOR	THE OHIO TEACHING FAMILY ASSOCIATION PO BOX 300 SWANTON, OH 43558	S H P T O	JOB & FAMILY SERVICE 239 W MAIN STREET LANCASTER, OH 43130 Phone: 740-652-7889	S			

VENDOR PHONE NUMBER		DOR FAX NUMBER	REQUISITION NUMBER	DELIVERY REFERENCE				
419-825-3440	0		469					
DATE ORDERED VENDOR NUMBE		R DATE REQUIRED FREIGHT METHOD/TERMS		DEPARTMENT/LOCATION				
01/01/2023	14262			JOB & FAMILY SERVICES				
NOTES								

NETWORK BOARD AND CARE

The Above Purchase Order Number Must Appear On All Correspondence - Packing Sheets And Bills Of Lading

ITEM #	DESCRIPTION / PART #	QTY	UOM	UNIT PRICE	EXTENDED PRICE
1	MODIFIED: NETWORK BOARD AND CARE	1.0	EACH	\$65,000.00	\$65,000.00

06/06/2023	Auditor Fairfield County, OH	Purchase Order Total	\$65,000.00 481
Date: 01/01/2023	Carrie L. Brown	Total Credit	\$0.00
		Total Discount	\$0.00
County Treasury or in process of collection to the credit of the submitted Fund(s) free from any obligation certification now outstanding.		Total Freight	\$0.00
It is hereby certified that the amount \$65,000.00 required to meet the contract, agreement, obligation, payment or expenditure, for the above, has been lawfully appropriated, authorized or directed for such purpose and is in the		Total Sales Tax	\$0.00
COUNTY AUDITOR'S CERTIFICATE		Total Ext. Price	\$65,000.00

06/06/2023

Vendor Copy

it our help guide		+
		+
it our help guide		+
it our help guide		
it our help guide		
×	<	
•		
		×

Inactive	
	Reset 🔿
i	No matches found Nour search did not return any results. To view Entity Registrations, you must sign in. Sign In Would you like to include inactive records in your search results? Yes



Our Website	
Our Partners	
Policies	
Customer Service	



General Services Administration

This is a U.S. General Services Administration Federal Government computer system that is **"FOR OFFICIAL USE ONLY."** This system is subject to monitoring. Individuals found performing 483

unauthorized activities are subject to disciplinary action including criminal prosecution.

Certified Search for Unresolved Findings for Recovery



Auditor of State - Unresolved Findings for Recovery Certified Search

I have searched The Auditor of State's unresolved findings for recovery database using the following criteria:

Contractor's Information:

Name: ,

Organization: The Ohio Teaching Family Association Date: 5/16/2023 3:50:34 PM

This search produced the following list of **14** possible matches:

Name/Organization	Address
Church of The Lion of Judah	4001 Foskett Road
Legacy Academy for Leaders and the Arts	1812 Oak Hill Avenue
Mathews, Marlon	9988 Wayne Ave
Northern Ohio Rural Water Board Members	
Praither, Randall	3626 Carroll Eastern Road
Southern Ohio Academy	522 Glenwood Ave
The Capella High School	4721 Reading Road
The Harte School	350 Columbus City Center Drive
The Haynes Group, Inc.	3897 E. 155th St.
The Haynes Group, Inc.	3897 E. 155th Street
The International Preparatory School	1301 East 9th Street, Suite 1900
The Perry Foundation	579 Valley Oak
Theodore Roosevelt Public Community School	c/o Richland Academy 75 North Walnut Street
Village of Bethel Mayor's Court	

The above list represents possible matches for the search criteria you entered. Please note that pursuant to ORC 9.24, only the person (which includes an organization) actually named in the finding for recovery is prohibited from being awarded a contract.

If the person you are searching for appears on this list, it means that the person has one or more findings for recovery and is prohibited from being awarded a contract described in ORC 9.24, unless one of the exceptions in that section apply.

If the person you are searching for does not appear on this list, an initialed copy of this page can serve as documentation of your compliance with ORC 9.24(E).

Please note that pursuant to ORC 9.24, it is the responsibility of the public office to verify that a person to whom it plans to award a contract does not appear in the Auditor of State's database. The Auditor of State's office is not responsible for inaccurate search results caused by user error or other circumstances beyond the Auditor of State's control.

(614) 466-4514 (800) 282-0370

Prosecutor's Approval Page

Resolution No.

A resolution authorizing the approval of a service agreement by and between Fairfield County Job & Family Services, Child Protective Services Division and Ohio Teaching Family Association.

(Fairfield County Job and Family Services)

Approved as to form on 6/1/2023 10:44:44 AM by Steven Darnell,

Signature Page

Resolution No. 2023-06.06.r

A resolution authorizing the approval of a service agreement by and between Fairfield County Job & Family Services, Child Protective Services Division, and the Ohio Teaching Family Association.

(Fairfield County Job and Family Services)

Upon the motion of Commissioner Jeffrey M. Fix, seconded by Commissioner David L. Levacy, this resolution has been Adopted:

Voting:

Steven A. Davis, President	Aye
David L. Levacy, Vice President	Aye
Jeffrey M. Fix	Aye

Board of County Commissioners Fairfield County, Ohio

CERTIFICATE OF CLERK

It is hereby certified that the foregoing is a true and correct transcript of a resolution acted upon by the Board of County Commissioners, Fairfield County, Ohio on the date noted above.

Rochelle Merringen

Rochelle Menningen Board of County Commissioners Fairfield County, Ohio

2023-06.06.s

A resolution authorizing the approval of a service agreement by and between Fairfield County Job & Family Services, Child Protective Services Division and Victory House LLC.

WHEREAS, Fairfield County Job & Family Services, Child Protective Services is requesting the Board of Commissioners approval of a service agreement with Victory House LLC, 4276 Linchmere Dr., Dayton, OH 45415 and

WHEREAS, the purpose of the service agreement is to provide Network Placement and Related Services for children who are in the care and custody of the Agency; and

WHEREAS, this agreement shall be effective April 1st, 2023 through March 31st, 2024; and

WHEREAS, a purchase order encumbering the funds for the services was acquired; and

WHEREAS, the Prosecuting Attorney has approved the agreement as to form.

NOW THEREFORE, BE IT RESOLVED BY THE BOARD OF COUNTY COMMISSIONERS, COUNTY OF FAIRFIELD, AND STATE OF OHIO:

Section 1. That the Fairfield County Board of Commissioners hereby approves the attached Network Placement Service Agreement for Victory House LLC

Prepared by: Brandi Downhour cc: JFS / Fiscal Supervisor

ROUTING FORM FOR CONTRACTS
The undersigned designee of the County affirms that he/she has reviewed the attached contract to ensure that it complies with County's needs and previous negotiations. The undersigned designee further affirms that the County has complied with the competitive selection process, as prescribed by the Ohio Revised Code, by selecting one of the boxes below.
A. Goods and/or Services in excess of \$50,000.00—competitively selected via an Invitation to Bid, pursuant to R.C. 307.86-307.92
B. Goods and/or Services in excess of \$50,000.00—competitively selected via a Request for Proposals, pursuant to R.C. 307.862
C. Dublic Improvement contracts—competitively selected pursuant to R.C. 153.08-153.12
D. Architect/Engineer design services for public improvements—selected through the Request for Qualifications process pursuant to R.C. 153.65-153.72
E. County Road Improvement/Construction-competitively selected pursuant to R.C. 5555.61
F. The subject matter was exempt from competitive selection for the following reason(s):
 Under \$50,000 State Term #: (copy of State Term Contract must be attached) ODOT Term #: (See R.C. 5513.01) Professional Services (See R.C. 307.86) Emergency (Follow procedure under ORC 307.86(A)) Sole Source (attach documentation as to why contract is sole source) Other: (cite to authority or explain why matter is exempt from competitive bidding)
G. Agreement not subject to Sections A-F (explain):
H. Compliance with Fairfield County Board of Commissioners Procurement Guidelines
 No County employee, employee's family member, or employee's business associate has an interest in this contract OR such interest has been disclosed and reviewed by the Prosecutor's Office No Finding for Recovery against Vendor as required under R.C. 9.24 (search via "Certified Search" on http://ffr.ohioauditor.gov/) Obtained 3 quotes for purchases under \$50,000 Purchase Order is included with Agreement
Signed this day of, 20
T_{2} (1)

* Please note that this checklist only addresses County and statutory requirements. If a contract is paid for with state and/or federal funds, please consult with the appropriate state and/or federal agency to ensure your department is complying with any additional requirements. By submitting a request for approval, you are certifying you have addressed County, statutory, and grant requirements.*

Fiscal Supervisor

4

Name and Title

01.2018 COST ANALYSIS:

For foster care placement, network providers have Title IV-E reimbursement ceilings, and Fairfield County rates have typically been below the state negotiated ceilings. Historically, the traditional, daily rate is less than \$225.

The review and evaluation of the separate cost elements and proposed profit would include an evaluation of special considerations and special needs, as there are cases which would be reviewed independently based on extraordinary factors. If the rate was more than \$225 per day, it is expected that there would be extraordinary, case specific needs, knowing what we know about the market in our area.

ORIGINAL	Carrí L. Brown, phd, mba, cgfm		Purchase Order
B JOB & FAMILY SERVICES	Fairfield County Auditor	Fiscal Year 2023	Page: 1 of 1
	210 East Main Street Lancaster, Ohio 43130		PPEAR ON ALL INVOICES, SHIPPING PAPERS.
	Revisions: 000	Purchase Order #	23004336 - 00
1239 W MAIN STREETLLANCASTER, OH 43130		Delivery must be made within	doors of specified destination.
L LANCASTER, OH 43130 Phone: 740-652-7889		Expiration Date:	12/15/2023
NICOLE VAUGHN VICTORY HOUSE LLC 4276 LINCHMERE DR DAYTON, OH 45415	S H P T O	JOB & FAMILY SERVICES 239 W MAIN STREET LANCASTER, OH 43130 Phone: 740-652-7889	5
VENDOR PHONE NUMBER VE	ENDOR FAX NUMBER REQUISITION	NUMBER DE	

VENDOR FROME N	UNDER VEN	DOR FAX NOWBER	REGUISITION NOMBER	DELIVERTREFERENCE
			4681	
DATE ORDERED	VENDOR NUMBER	DATE REQUIRED	FREIGHT METHOD/TERMS	DEPARTMENT/LOCATION
05/02/2023	17550			JOB & FAMILY SERVICES
		NOT	ES	

BOARD AND CARE

Γ

The Above Purchase Order Number Must Appear On All Correspondence - Packing Sheets And Bills Of Lading

ITEM #	DESCRIPTION / PART #	QTY	UOM	UNIT PRICE	EXTENDED PRICE
1	BOARD AND CARE	1.0	EACH	\$5,000.00	\$5,000.00

COUNTY AUDITOR'S CERTIFICATE		Total Ext. Price	\$5,000.00
t is hereby certified that the amount \$5,000.00 required to meet the contract, agreement, obligation, payment or expenditure, for the above, has been lawfully appropriated, authorized or directed for such purpose and is in the County Treasury or in process of collection to the credit of the submitted Fund(s) free from any obligation or		Total Sales Tax	\$0.00
certification now outstanding.	pliection to the credit of the submitted Fund(s) free from any obligation of	Total Freight	\$0.00
		Total Discount	\$0.00
Date: 05/02/2023	Carrie L. Brown	Total Credit	\$0.00
06/06/2023	Auditor Fairfield County, OH	Purchase Order Total	\$5,000.00

Vendor Copy

SAM,GOV®		
arch	Exact Phrase	
	e.g. 1606N020Q02	
Select Domain All Domains		+
Filter By		_
	keyword search, visit our help guide	
	keyword search, visit our help guide	
All Words (i)	keyword search, visit our help guide	
For more information on how to use our l Any Words (i) All Words (i) Exact Phrase (i)	keyword search, visit our help guide	
For more information on how to use our Any Words (i) All Words (i) Exact Phrase (i) e.g. 1606N020Q02		

Inactive	
	Reset 🔿
i	Not search did not return any results. To view Entity Registrations, you must sign in. Sign In Would you like to include inactive records in your search results?
	Yes Go Back



Our Website	
Our Partners	
Policies	
Customer Service	



General Services Administration

This is a U.S. General Services Administration Federal Government computer system that is **"FOR OFFICIAL USE ONLY."** This system is subject to monitoring. Individuals found performing 493

unauthorized activities are subject to disciplinary action including criminal prosecution.

Ohio Department of Job and Family Services

AGREEMENT FOR TITLE IV-E AGENCIES AND PROVIDERS FOR THE PROVISION OF CHILD PLACEMENT

This Agreement sets forth the terms and conditions between the parties for placement services for children who are in the care and custody of the Agency named below.

This Agreement is between Fairfield County Department of Job and Family Services, a Title IV-E Agency, hereinafter "Agency", whose address is:

Fairfield County Department of Job and Family Services 239 W Main St Lancaster, OH 43130

and

Victory House LLC, hereinafter "Provider", whose address is:

Victory House LLC 4276 Linchmere Dr Dayton, OH 45415

Collectively the "Parties".

Table of Contents

ARTICLE XXXII. ARTICLE XXXIII.	APPLICABLE LAW AND VENUE
ATTACHMENTS TO TH	

RECITALS

WHEREAS, the Agency is responsible under Ohio Revised Code (ORC) Title 51, Chapter <u>5153</u> for the provision of protective services for dependent, neglected, and abused children; and,

WHEREAS, the Agency is authorized under ORC Title 51, Chapter <u>5153.16</u> to provide care and services which it deems to be in the best interest of any child who needs or is likely to need public care and services; and,

WHEREAS, the Provider is an organization duly organized and validly existing and is qualified to do business under the laws in the State of Ohio or in the state where the Provider of services is located and has all requisite legal power and authority to execute this Agreement and to carry out its terms, conditions and provisions; and is licensed, certified or approved to provide services to children and families in accordance with Ohio law or the state where the Provider of services is located.

NOW, THEREFORE, in consideration of the mutual promises and responsibilities set forth herein, the Agency and Provider agree as follows:

Article I. SCOPE OF PLACEMENT SERVICES

In addition to the services described in Exhibit I-Scope of Work, Provider agrees to provide and shall provide the placement and related services specified in each Individual Child Care Agreement (ICCA) for children in the care and custody of the Title IV-E Agency. The ICCA shall be consistent with current federal, state and local laws, rules and regulations applicable to the Provider's license or certified functions and services. If an Agreement and ICCA both exist, the Agreement supersedes.

See Attachment 1 for additional details.

Section 1.01 FOR AGREEMENTS COMPETITIVELY PROCURED

Without limiting the services set forth herein, Provider will provide the Services pursuant to and consistent with the Requests for Proposals (RFP) and the Provider's Proposal submitted in response to the RFP, the Provider agrees to provide and shall provide the placement and related services described in Exhibit I-Scope of Work.

Section 1.02 FOR AGREEMENTS NOT COMPETITIVELY PROCURED

The Provider agrees to provide and shall provide the placement and related services described in the Exhibit I- Scope of Work.

Section 1.03 EXHIBITS

The following exhibits are deemed to be a part of this Agreement as if fully set forth herein:

- A. Exhibit I Scope of Work;
- B. Exhibit II Request for Proposals (if applicable);
- C. Exhibit III Provider's Response to the Request for Proposals (if applicable); and
- D. Exhibit IV Schedule A Rate Information.

Article II. TERM OF AGREEMENT

This Agreement is in effect from **04/01/2023** through **03/31/2024**, unless this Agreement is suspended or terminated pursuant to Article VIII prior to the termination date.

In addition to the initial term described above, this Agreement may be extended, at the option of the Agency and upon written agreement of the Provider, for ______ additional, _____ year terms not to exceed

years. Notice of Agency's intention to extend the Agreement shall be provided in writing to Provider no less than 90 calendar days before the expiration of any Agreement term then in effect. (If a previous Request for Proposal [RFP] allows, the Agreement may be extended for a period of time to ensure adequate completion of the Agency's competitive procurement process at the rates existing for the term then in effect.)

Article III. ORDER OF PRECEDENCE

This Agreement and all Exhibits are intended to supplement and complement each other and shall, where possible, be so interpreted. However, if any provision of this Agreement irreconcilably conflicts with an Exhibit, this Agreement takes precedence over the Exhibit(s).

In the event there is an inconsistency between the Exhibit(s), the inconsistency shall be resolved in the following order:

- A. Exhibit I: Scope of Work; then
- B. Exhibit II: Request for Proposals (if applicable); then
- C. Exhibit III: Provider's Proposals (if applicable); then
- D. Exhibit IV: Title IV-E Schedule A Rate Information.

Article IV. DEFINITIONS GOVERNING THIS AGREEMENT

The following definitions govern this Agreement:

- A. Agreement means this Agreement, attachments and exhibits thereto.
- B. Material Breach shall mean an act or omission that violates or contravenes an obligation required under the Agreement and which, by itself or together with one or more other breaches, has a negative effect on, or thwarts the purpose of the Agreement as stated herein. A Material Breach shall not include an act or omission, which has a trivial or negligible effect on the quality, quantity, or delivery of the goods and services to be provided under the Agreement.
- C. Child(ren) means any person under eighteen years of age or a mentally or physically handicapped person under twenty-one years of age in the Agency's custody and under the care of the Provider for the provision of placement services.
- D. All other definitions to be resolved through Federal Regulations, Ohio Administrative Code (OAC) 5101:2-1-01 and any related cross-references.

Article V. PROVIDER RESPONSIBILITIES

- A. Provider agrees to participate with Agency in the development and implementation of the Case Plan and ICCA including participation in case reviews and / or semi-annual administrative reviews, and the completion of reunification assessments for the children in placement with the Provider. Parties shall make best efforts to share information timely regarding participants and contact information involved with planning efforts related to children and families.
- B. Provider agrees to provide services agreed to in the Case Plan and ICCA (i.e.,transportation of children for routine services, including, but not limited to, court hearings, medical appointments, school therapy, recreational activities, visitations/family visits) unless otherwise negotiated in writing as an attachment to this Agreement. Any disputes involving services or placement will be resolved through mutual-agreement and modification to the ICCA. Provider agrees the Agency is the final authority in the process. The cost of providing these services is to be included in the Agency approved per diem.
- C. Provider agrees to ensure that any and all persons who may act as alternative caregivers or who have contact with the children are suitable for interaction pursuant to all applicable federal, state and local laws and regulations.
- D. Provider agrees that all caregivers must be approved by the Agency.
- E. Provider agrees to submit a progress report as negotiated by the parties for each child. The progress report will be based on the agreed upon services to be delivered to the child and/or family and will include documentation of services provided to the child and/or discharge summary. If Monthly Progress Reports are not received within 90 calendar days following the month of service provision, payment may be withheld at the Agency's discretion.

- 1. Monthly Progress Reports shall be submitted by the 20th of the month following the month of service.
- 2. The Monthly Progress Report will include the following medical related information:
 - a. Service type (i.e. medical, dental, vision, etc.);
 - b. Date(s) of service;
 - c. Reason for visit (i.e. routine, injury, etc.);
 - d. Practitioner name, address and contact number;
 - e. Name of hospital, practice, urgent care, etc.;
 - f. Prescribed medications and dosages;
 - g. Date(s) medication(s) were prescribed or changed; and
 - h. Changes to medications.
- F. Placement changes, emergency or non-emergency, shall occur only with the approval of the Agency. The following information shall be provided to the Agency for all placement changes: Name, address and phone number of the new foster home or other out-of-home care setting, the license/home study of the new care provider within 24 hours, excluding weekends and holidays.
- G. Provider agrees to notify all Agencies who have children placed in the same caregiver's home/group home/CRC when any child residing in the placement is critically injured or dies in that location. Notification will be made to the Agencies' Child Abuse/Neglect Hotline number or assigned Caseworker immediately.
- H. Notification to the Agency of Emergency Critical Incidents shall occur ASAP but no later than one hour of the Incident becoming known. Notification will be made to the Agency via the Agency's Child Abuse/Neglect Hotline or assigned Caseworker or by other established system. Critical incidents are those incidents defined in the Ohio Administrative Code that are applicable to the licensed or certified programs (ODJFS 5101:2-7-14, 5101:2-9-23 ODMHAS 5122-30-16, 5122-26-13, OAC 5123-17-02).

Emergency situations include but are not limited to the following:

- 1. Absent Without Leave (AWOL);
- 2. Child Alleging Physical or Sexual Abuse / Neglect;
- 3. Death of Child;
- 4. Illicit drug/alcohol use; Abuse of medication or toxic substance;
- 5. Sudden injury or illness requiring an unplanned medical treatment or visit to the hospital;
- 6. Perpetrator of Delinquent/Criminal Act (Assault, Dangerous Behaviors, Homicidal Behaviors);
- 7. School Expulsion / Suspension (formal action by school);
- 8. Self-Injury (Suicidal Behaviors, Self-Harm Requiring external Medical Treatment, Hospital or ER);
- 9. Victim of assault, neglect, physical or sexual abuse; and
- 10. The filing of any law enforcement report involving the child.
- I. The Provider also agrees to notify the Agency within Twenty-four (24) hours, of any non-emergency situations. Non-emergency situations include but are not limited to the following:
 - 1. When physical restraint is used/applied; and
 - 2. Medication lapses or errors.

Notification will be made to the Agency via the Agency's Child Abuse Neglect Hotline / assigned Caseworker or by other established notification system.

- J. Documentation of the emergency and non-emergency incidents as identified in "H and I" above shall be provided to the Agency via email, fax or other established notification system within 24 hours excluding weekends and holidays.
- K. The Provider agrees to submit each child's assessment and treatment plans as completed but no later than the 30th day of placement. Provider further agrees to provide treatment planning that will include, but is not limited to, education on or off site, preparation for integration into community-based school or vocational/job skills training, community service activities, independent living skills if age 14 or older, monitoring and supporting community adjustment.
- L. The Provider agrees to participate in joint planning with the Agency regarding modification to case plan services. Provider agrees that while the Provider may have input into the development of the child's case plan services and

the ICCA, any disputes involving services or placement will be resolved through mutual agreement and modification to the ICCA. Provider agrees the Agency is the final authority in the process.

- M. The Provider shall participate in a Placement Preservation meeting if requested by the Agency prior to issuing a notice of removal of a child. A placement Preservation meeting shall be held within seven (7) business days of said request. Unless otherwise mutually agreed upon a minimum of thirty (30) calendar days' notice shall be given if placement preservation is unable to be achieved. A Discharge Plan Summary shall be provided no later than fifteen (15) calendar days after the date of discharge in accordance with the applicable licensed or certified program. (OAC 5101:2-5-17, OAC 5122-30-22, OAC 5122-30-04, OAC 5123:2-30-05).
- N. The Provider shall work in cooperation and collaboration with the Agency to provide information for each child's Lifebook and will fully comply with the provision of <u>OAC 5101:2-42-67</u> as applicable to private Providers. Provider's contribution to the Agency Lifebook for a child shall be for the episode of care with the Provider.
- O. The Provider agrees to provide Independent Living Services as set forth in accordance with OAC 5101:2-42-19 for all children age 14 and above.
- P. When applicable, due to the Provider being part of a managed care agreement as defined in <u>OAC 5101:2-1-01</u>, the Provider agrees to visit with the child face-to-face in the foster home, speak privately with the child and to meet with the caregiver at least monthly in accordance with rule <u>OAC 5101:2-42-65</u> of the Ohio Administrative Code.
- Q. The Provider agrees to maintain its licenses and certifications from any source in good standing. The Provider agrees to report to Agency in writing any change in licensure or certification that negatively impacts such standing immediately if the negative action results in a temporary license, suspension of license or termination of license.
- R. Provider agrees that the reasonable and prudent parent standard training required by SEC. 471. [42 U.S.C. 671] of the Social Security Act and in accordance to <u>OAC 5101:2-5-33</u>, <u>OAC 5101:2-9-02</u> or <u>OAC 5101:2-9-03</u> has been completed.
- S. The Provider shall notify Agency of any changes in its status, such as intent to merge with another business or to close no later than forty-five (45) business days prior to the occurrence.
- T. The Provider agrees that the Agency shall have access to foster parent home studies and re-certifications for foster parents caring for children in placement, subject to confidentiality considerations. The Provider shall submit to Agency a copy of the current foster home license at the time of placement and recertification. Provider also agrees to notify Agency within twenty-four (24) hours of any change in the status of the foster home license.
- U. When there is a rule violation of a caregiver, a copy of the corrective action plan, if applicable, must be submitted to the Agency when the investigation is complete.
- V. The Provider agrees to notify the Agency of scheduling no less than fourteen (14) calendar days prior to all formal meetings (i.e. FTMs, Treatment Team Meetings, IEPs, etc.).
- W. The Provider agrees to adhere to the following Medical/Medication guidelines:
 - 1. To provide over-the-counter medications and/or supplies as part of the per diem of care;
 - 2. To comply with the medical consent process as identified by Agency;
 - 3. Only the Agency can give permission for the administering or change (addition or elimination) of psychotropic medication and its ongoing management; and
 - 4. Provide an initial placement medical screening within 72 hours of child's placement into a placement resource under the Provider's operation and/or oversight.
- X. To arrange for required health care/medical examinations within time frames required by <u>OAC 5101:2-42-66.1</u> and provide reports from the health care providers to the agency within 30 days of occurrence if the appropriate releases of information have been obtained by the Provider.
- Y. The Network Provider agrees to notify the Agency if placement resource is currently under investigation for license violations or misconduct toward children or other third-party investigation.
- Z. The Provider will immediately notify the Agency:

- 1. If the Provider is out of compliance with any licensing authority rules or the placement resource is under investigation for license violations or misconduct toward children. Immediately is defined as within one hour of knowledge of the non-compliance issue.
- 2. Child Abuse/Neglect Hotline or assigned Caseworker of any allegations of abuse or neglect made against the Caregiver within one hour of gaining knowledge of the allegation.
- 3. Of any corrective action and the result of the correction action plan. The Provider will submit a comprehensive written report to the agency within sixty (60) days of the rules violation.
- 4. Within twenty-four (24) hours any time there is an event which would impact the placement resource license.

See Attachment 2 for additional details.

Article VI. AGENCY RESPONSIBILITIES

- A. Agency certifies that it will comply with the Multiethnic Placement Act, 108 STAT. 3518, as amended by Section 1808 of the Small Business Jobs Protection Act of 1996, 110 STAT. 1755, which prohibits any Agency from denying any person the opportunity to become an adoptive or foster parent on the basis of race, color, national origin, or delaying or denying the placement of a child for adoption or into foster care on the basis of race, color, or national origin of the adoptive or foster parent or of the child involved.
- B. The Agency shall provide to the Provider within thirty (30) calendar days of placement or within a reasonable time thereafter as agreed to by the parties, a copy of each child's social history, medical history, and Medicaid card once obtained by the Agency for new cases, or at time of placement for existing cases. Agency shall make best efforts to share information timely regarding participants and contact information involved with planning efforts related to children and families.
- C. The Agency acknowledges that clinical treatment decisions must be recommended by licensed clinical professionals. Agency and Provider acknowledge that disagreement with a treatment decision may be taken through the dispute resolution process contained in Article XIV of this Agreement.
- D. Agency agrees to visit with the child in accordance with rule <u>OAC 5101:2-42-65</u> of the Ohio Administrative Code.
- E. Agency agrees to participate in periodic meetings with each child's treatment team for case treatment plan development, review, and revision. The Agency agrees to participate in the development of the treatment plan of each child placed with the Provider by the Agency.
- F. Agency certifies that it will comply with Every Student Succeeds Act (34 CFR part 200) and will work with local school districts in developing individualized plans to address the transportation needed for a child to remain in the school of origin. Agency agrees to arrange for the transfer of each child's school records to the child's new school upon placement but not later than ten (10) business days. The Agency agrees to work with the Provider for the timely enrollment of the child in the receiving school district. The Agency has the final responsibility to obtain the child's school records and to enroll the child in the receiving school district.
- G. The Agency shall provide an opportunity for the Provider to give input in the development, substantive Addendum or modification of case plans. The Agency agrees to notify the Provider of scheduling no less than seven (7) calendar days prior to of all formal meetings (e.g. SARs, court hearings, family team conferences, etc.).
- H. The Agency shall participate in a Placement Preservation meeting if requested by the Provider prior to issuing a notice of removal of a child. The Agency shall provide a minimum of thirty (30) calendar days' notice for planned removals, to the Provider for each child who is being terminated from placement with the Provider, unless so ordered by a court of competent jurisdiction.
- I. Agency agrees to provide the Provider with an emergency contact on a twenty-four (24) hour, seven (7) day per week basis.
- J. The Agency represents:
 - 1. It has adequate funds to meet its obligations under this Agreement; subject to the availability of funds as referenced in Article VIII (I);
 - 2. It intends to maintain this Agreement for the full period set forth herein and has no reason to believe that it

will not have sufficient funds to enable it to make all payments due hereunder during such period; and

- 3. It will make its best effort to obtain the appropriation of any necessary funds during the term of this Agreement.
- K. The Agency will provide information about the child being referred for placement in accordance with <u>OAC</u> <u>5101:2-42-90</u>. Prior to a child's placement in alternative care or respite, <u>OAC 5101:2-42-90 (D)</u> requires the Agency to share with care givers information that could impact the health, safety, or well-being of the child or others in the home.

Article VII. INVOICING FOR PLACEMENT SERVICES

- A. The Provider agrees to submit a monthly invoice following the end of the month in which services were provided. The invoice shall be for services delivered in accordance with Article I of this Agreement and shall include:
 - 1. Provider's name, address, telephone number, fax number, federal tax identification number, Title IV-E Provider number, if applicable and Medicaid Provider number, if applicable.
 - 2. Billing date and the billing period.
 - 3. Name of child, date of birth of child, and the child's Statewide Automated Child Welfare Information System (SACWIS) person I.D. number.
 - 4. Admission date and discharge date, if available.
 - 5. Agreed upon per diem for maintenance and the agreed per diem administration; and
 - 6. Invoicing procedures may also include the per diems associated with the following if applicable and agreeable to the Agency and Provider:
 - a. Case Management; allowable administration cost;
 - b. Transportation, allowable maintenance cost;
 - c. Transportation; allowable administration cost;
 - d. Other Direct Services; allowable maintenance cost;
 - e. Behavioral health care; non-reimbursable cost; and
 - f. Other costs (any other cost the Title IV-E Agency has agreed to participate in); non-allowable/ non-reimbursable cost.
- B. Provider warrants and represents claims made for payment for services provided are for actual services rendered and do not duplicate claims made by Provider to other sources of public funds for the same service.

Article VIII. REIMBURSEMENT FOR PLACEMENT SERVICES

- A. The maximum amount payable pursuant to this contract is **\$500,000.00**.
- B. In accordance with Schedule A of this Agreement, the per diem for maintenance and the per diem for administration will be paid for each day the child was in placement. The first day of placement will be paid regardless of the time the child was placed. The last day of placement will not be paid regardless of the time the child left the placement.
- C. In accordance with Schedule A of this Agreement and in addition to Maintenance and Administration, the Agency may agree to pay a per diem for Case Management, Other Direct Services, Transportation Administration, Transportation Maintenance, Behavioral Health Care and Other. All other services and/or fees to be paid for shall be contained in the Attachments/Exhibits of this Agreement.
- D. To the extent that the Provider maintains a foster care network, the agreed upon per diem for maintenance shall be the amount paid directly to the foster parent. Maintenance includes the provision of food, clothing, shelter, daily supervision, graduation expenses, a child's personal incidentals, and liability insurance with respect to the child, reasonable cost of travel to the child's home for visitation and reasonable cost of travel for the child to remain in the school the child was enrolled in at the time of placement. Payment for private Agency staff transporting a child to a home visit or keeping the child in their home school will be paid in accordance with Schedule A (Transportation Maintenance) of this Agreement.
- E. If the plan as determined by the Agency is to return the child to placement with the Provider, the Agency may

agree to pay for the days that a child is temporarily absent from the direct care of the Provider, as agreed to by the parties in writing.

- F. The service provider is required to utilize Medicaid-approved healthcare providers in the appropriate managed care network for the provision of mental health, dental and/or medical services (hereafter referred to collectively as "medical services") to children in the custody of Agency. The Service Provider will report applicable Medicaid/insurance information to the healthcare providers and instruct healthcare providers to seek payment from Medicaid or any other available third-party payer for medical services rendered to children in agency custody. Agency will not pay for the provision of any medical services to children in agency custody unless the agency Executive Director or authorized designee has provided specific prior written authorization for such medical services and associated costs.
- G. The Agency agrees to pay the Provider for all services agreed to on Schedule A and in the Attachments/Exhibits to this Agreement, where applicable, that have been provided and documented in the child's case file. Agency shall make best efforts to make payment of undisputed charges within thirty (30) business days of receipt.
- H. In the event of a disagreement regarding payment, Agency shall withhold payment only for that portion of the placement with which it disagrees. Agency will use best efforts to notify the Provider of any invoice discrepancies. Agency and Provider will make every effort to resolve payment discrepancies within 60 calendar days. Payment discrepancies brought to the Agency after 60 days will be reviewed on a case by case basis.
- I. This Agreement is conditioned upon the availability of federal, state, or local funds appropriated or allocated for payment for services provided under the terms and conditions of this Agreement. By sole determination of the Agency, if funds are not sufficiently allocated or available for the provision of the services performed by the Provider hereunder, the Agency reserves the right to exercise one of the following alternatives:
 - 1. Reduce the utilization of the services provided under this Agreement, without change to the terms and conditions of the Agreement; or
 - 2. Issue a notice of intent to terminate the Agreement.

The Agency will notify the Provider at the earliest possible time of such decision. No penalty shall accrue to the Agency in the event either of these provisions is exercised. The Agency shall not be obligated or liable for any future payments due or for any damages as a result of termination under this section.

Any denial of payment for service(s) rendered may be appealed in writing and will be part of the dispute resolution process contained in Article XIV.

See Attachment 3 for additional details.

Article IX. TERMINATION; BREACH AND DEFAULT

- A. This Agreement may be terminated for convenience prior to the expiration of the term then in effect by either the Agency or the Provider upon written notification given no less than sixty (60) calendar days in advance by certified mail, return receipt requested, to the last known address of the terminated party shown hereinabove or at such other address as may hereinafter be specified in writing.
- B. If Provider fails to provide the Services as provided in this Agreement for any reason other than Force Majeure, or if Provider otherwise Materially Breaches this Agreement, Agency may consider Provider in default. Agency agrees to give Provider thirty (30) days written notice specifying the nature of the default and its intention to terminate. Provider shall have seven (7) calendar days from receipt of such notice to provide a written plan of action to Agency to cure such default. Agency is required to approve or disapprove such plan within five (5) calendar days of receipt. In the event Provider fails to submit such plan or Agency disapproves such plan, Agency has the option to immediately terminate this Agreement upon written notice to Provider. If Provider fails to cure the default in accordance with an approved plan, then Agency may terminate this Agreement at the end of the thirty (30) day notice period.
- C. Upon of the effective date of the termination, the Provider agrees that it shall cease work on the terminated activities under this Agreement, take all necessary or appropriate steps to limit disbursements and minimize

costs, and furnish a report as of the date of discharge of the last child describing the status of all work under this Agreement, including without limitation, results accomplished, conclusions resulting therefrom, and such other matters as the Agency may require. The Agency agrees to remove all children in placement immediately with the Provider, consistent with the effective termination date. In all instances of termination, the Provider and Agency agree that they shall work in the best interests of children placed with the Provider to secure alternative placements for all children affected by the termination.

- D. In the event of termination, the Provider shall be entitled to reimbursement, upon submission of an invoice, for the agreed upon per diem incurred prior to the effective termination date. The reimbursement will be calculated by the Agency based on the per diem set forth in Article VIII. The Agency shall receive credit for reimbursement already made when determining the amount owed to the Provider. The Agency is not liable for costs incurred by the Provider after the effective termination date of the discharge of the last child.
- E. Notwithstanding the above, Agency may immediately terminate this Agreement upon delivery of a written notice of termination to the Provider under the following circumstances:
 - 1. Improper or inappropriate activities;
 - 2. Loss of required licenses;
 - 3. Actions, inactions or behaviors that may result in harm, injury or neglect of a child;
 - 4. Unethical business practices or procedures; and
 - 5. Any other event that Agency deems harmful to the well-being of a child; or
 - 6. Loss of funding as set forth in Article VIII.
- F. If the Agreement is terminated by Agency due to breach or default of any of the provisions, obligations, or duties embodied contained therein by the Provider, Agency may exercise any administrative, agreement, equitable, or legal remedies available, without limitation. Any extension of the time periods set forth above shall not be construed as a waiver of any rights or remedies the Agency may have under this Agreement.
- G. In the event of termination under this ARTICLE, both the Provider and the placing Agency shall make good faith efforts to minimize adverse effect on children resulting from the termination of the Agreement.

Article X. RECORDS RETENTION, CONFIDENTIALITY AND DATA SECURITY REQUIREMENTS

- A. The Provider agrees that all records, documents, writings or other information, including, but not limited to, financial records, census records, client records and documentation of legal compliance with Ohio Administrative Code rules, produced by the Provider under this Agreement, and all records, documents, writings or other information, including but not limited to financial, census and client used by the Provider in the performance of this Agreement are treated according to the following terms:
 - 1. All records relating to costs, work performed and supporting documentation for invoices submitted to the Agency by the Provider along with copies of all Deliverables, as defined in Article XXIX, submitted to the Agency pursuant to this Agreement will be retained for a minimum of three (3) years after reimbursement for services rendered under this Agreement.
 - 2. If an audit, litigation, or other action is initiated during the time period of the Agreement, the Provider shall retain such records until the action is concluded and all issues resolved or three (3) years have expired, whichever is later.
 - 3. All records referred to in Section A 1) of this Article shall be available for inspection and audit by the Agency or other relevant agents of the State of Ohio (including, but not limited to, the County Prosecutor, the Ohio Department of Job and Family Services (ODJFS), the Auditor of the State of Ohio, the Inspector General of Ohio, or any duly authorized law enforcement officials), and the United States Department of Health and Human Services within a reasonable period of time.
- B. The Provider agrees to keep all financial records in a manner consistent with Generally Accepted Accounting Principles.
- C. The Provider agrees to comply with all federal and state laws applicable to the Agency and the confidentiality of children and families. Provider understands access to the identities of any Agency's child and families shall only be as necessary for the purpose of performing its responsibilities under this Agreement. No identifying

information on child(ren) served will be released for research or other publication without the express written consent of the Agency. Provider agrees that the use or disclosure of information concerning the child for any purpose not directly related to the administration of this Agreement is prohibited. Provider shall ensure all the children's and families' documentation is protected and maintained in a secure and safe manner.

- D. The Provider agrees to comply with all applicable state and federal laws related to the confidentiality and transmission of medical records, including, but not limited to the Health Insurance Portability and Accountability Act of 1996 (HIPAA).
- E. Although information about, and generated under, this Agreement may fall within the public domain, the Provider shall not release information about, or related to, this Agreement to the general public or media verbally, in writing, or by any electronic means without prior approval from the Agency, unless the Provider is required to release requested information by law. Agency reserves the right to announce to the general public and media: award of the Agreement, Agreement terms and conditions, scope of work under the Agreement, Deliverables, as defined in Article XXIX, and results obtained under the Agreement. Except where Agency approval has been granted in advance, the Provider shall not seek to publicize and will not respond to unsolicited media queries requesting: announcement of Agreement award, Agreement terms and conditions, Agreement scope of work, government-furnished documents the Agency may provide to the Provider to fulfill the Agreement scope of work, Deliverables required under the Agreement, results obtained under the Agreement, and impact of Agreement activities.
- F. If contacted by the media about this Agreement, the Provider agrees to notify the Agency in lieu of responding immediately to media queries. Nothing in this section is meant to restrict the Provider from using Agreement information and results to market to specific business prospects.
- G. Client data must be protected and maintained in a secure and safe manner whether located in Provider's facilities, stored in the Cloud, or used on mobile devices outside Provider's facility. Security of Provider's network, data storage, and mobile devices must conform to generally recognized industry standards and best practices. Maintenance of a secure processing environment includes, but is not limited to, network firewall provisioning, intrusion detection, antivirus protection, regular third-party vulnerability assessments, and the timely application of patches, fixes and updates to operating systems and applications.
- H. Provider agrees that it has implemented and shall maintain during the term of this Agreement the highest standard of administrative, technical, and physical safeguards and controls to:
 - 1. Ensure the security and confidentiality of data;
 - 2. Protect against any anticipated security threats or hazards to the security or integrity of data; and
 - 3. Protect against unauthorized access to or use of data. Such measures shall include at a minimum:
 - a. Access controls on information systems, including controls to authenticate and permit access to data only to authorized individuals and controls to prevent Provider employees from providing data to unauthorized individuals who may seek to obtain this information (whether through fraudulent means or otherwise);
 - b. Firewall protection;
 - c. Encryption of electronic data while in transit from Provider networks to external networks;
 - d. Measures to store in a secure fashion all data which shall include multiple levels of authentication;
 - e. Measures to ensure that data shall not be altered or corrupted without the prior written consent of the Agency;
 - f. Measures to protect against destruction, loss or damage of data due to potential environmental hazards, such as fire and water damage.
- I. Immediately upon discovery of a confirmed or suspected breach involving data, Provider will notify Agency no later than twenty-four (24) hours after Provider knows or reasonably suspects a breach has or may have occurred. Provider shall promptly take all appropriate or legally required corrective actions and shall cooperate fully with the Agency in all reasonable and lawful efforts to prevent, mitigate or rectify such data breach. In the event of a suspected breach, Provider shall keep the Agency informed of the progress of its investigation until the uncertainty is resolved.
- J. In the event the Provider does not carry the appropriate cyber security insurance to cover a security breach, the

Provider shall reimburse the Agency for actual costs incurred, including, but not limited to, providing clients affected by a security breach with notice of the breach, and/or complimentary access for credit monitoring services, which the Agency deems necessary to protect such affected client.

K. In the event the Agency discontinues operation, all child records for residential or any other placement settings shall be provided to the custodial agency. If the setting is licensed by ODJFS, licensing records shall be sent to:

ODJFS ATTN: Licensing P.O. Box 183204 Columbus, OH 43218-3204

Article XI. PROVIDER ASSURANCES AND CERTIFICATIONS

- A. As applicable to the Provider's license and/or certification, the Provider certifies compliance with <u>ORC 2151.86</u>, <u>ORC 5103.0328</u>, <u>ORC 5103.0319</u> and applicable OAC Sections as defined in Article XXII of this Agreement concerning criminal record checks, arrests, convictions and guilty pleas relative to foster caregivers, employees, volunteers and interns who are involved in the care for a child. Provider is responsible for any penalties, financial or otherwise, that may accrue because of noncompliance with this provision.
- B. To the extent that the Provider maintains a residential center or group home, the Provider agrees to comply with the provisions of their licensing Agency that relates to the operation, safety and maintenance of residential facilities. Specifically, Provider agrees that no firearm or other projectile weapon and no ammunition for such weapons will be kept on the premises.
- C. Provider certifies compliance with Drug Free Work Place Requirements as outlined in 45 C.F.R. Part 76,Subpart F.
- D. Provider certifies compliance with 45 C.F.R. Part 80, Non-Discrimination under programs receiving Federal assistance through the Department of Health and Human Services effectuation of Title VI of the Civil Rights Act of 1964.
- E. Provider certifies compliance with 45 C.F.R. Part 84, Non-Discrimination on the Basis of Handicap in Programs or Activities Receiving Federal Assistance.
- F. Provider certifies compliance 45 C.F.R. Part 90, Non-Discrimination on the Basis of Age in Programs or Activities Receiving Federal Assistance.
- G. Provider certifies compliance with the American with Disabilities Act, Public Law 101-336.
- H. Provider certifies that it will:
 - 1. Provide a copy of its license(s), certification, accreditation or a letter extending an expiring license, certification, or accreditation from the issuer to the Agency prior to the signing of the Agreement.
 - 2. Maintain its license(s), certification, accreditation and that upon receipt of the renewal of its license, certification, and/or accreditation or upon receipt of a letter extending an expiring license, certification, and/or accreditation from the issuer, a copy of the license, certification and/or accreditation will be provided to the Agency within five (5) business days.
 - 3. Provider shall immediately notify the Agency of any action, modification or issue relating to said licensure, accreditation or certification.
- I. Provider certifies that it will not deny or delay services to eligible persons because of the person's race, color, religion, national origin, gender, orientation, disability, or age.
- J. The Provider shall comply with Executive Order 11246, entitled Equal Employment Opportunity, as amended by Executive Order 11375, and as supplemented in Department of Labor regulation 41 CFR part 60.
- K. Provider further agrees to comply with <u>OAC 5101:9-2-01</u> and <u>OAC 5101:9-2-05(A)(4)</u>, as applicable, which require that assure that persons with limited English proficiency (LEP) can meaningfully access services. To the

extent Provider provides assistance to an LEP Child through the use of an oral or written translator or interpretation services in compliance with this requirement, the LEP Child shall not be required to pay for such assistance.

- L. To the extent applicable, the Provider certifies compliance with all applicable standards, orders, or requirements issued under Section 306 of the Clean Air Act (42 U.S.C. 1857 (h) Section 508 of the Clean Water Act (33 U.S.C. 1368), Executive Order 11738, and Environmental Protection Agency Regulations (40 C.F.R. Part 15).
- M. The Provider certifies compliance, where applicable, with mandatory standards and policies relating to energy efficiency which are contained in the state energy conservation plan issued in compliance with the Energy Policy and Conservation Act (Pub. L. 94-163, 89 Stat. 871).
- N. The Provider certifies that all approvals, licenses, or other qualifications necessary to conduct business in Ohio have been obtained and are current.
- O. Provider shall comply with the Small Business Job Protection Act (Public Law ("P.L.") 104-188), the Multiethnic Placement Act of 1994 (P.L. 103-382), Titles IV-B (42 U.S.C. 620 et seq.) and IV-E (42 U.S.C. 670 et seq.) of the Social Security Act ("the Act"), the Personal Responsibility and Work Opportunity Reconciliation Act of 1996 (P.L. 104-193), Section 471(a) of Title IV-E of the Act (42 U.S.C. 671(a)), and 45 C.F.R. 1356, including all rules, regulations and guidelines issued by federal and state authorities, <u>OAC 5101:9-4-07</u> and <u>OAC 5101:2-47-23.1</u>.

Article XII. INDEPENDENT CONTRACTOR

- A. The Provider and the Agency agree that no employment, joint venture, or partnership has been or will be created between the parties hereto pursuant to the terms and conditions of this Agreement.
- B. The Provider and the Agency agree that the Provider is an independent contractor and assumes all responsibility for any federal, state, municipal, or other tax liabilities along with workers' compensation, unemployment compensation, and insurance premiums which may accrue as a result of compensation received for services or Deliverables rendered hereunder.
- C. The Provider and the Agency agree that no person and/or entities entering into this Agreement, nor any individual employed by any person or entity entering in to this Agreement, are public employees for purposes of contributions to Ohio Public Employees Retirement system by virtue of any work performed or services rendered in accordance with this Agreement.

Article XIII. AUDITS AND OTHER FINANCIAL MATTERS

- A. Provider agrees to submit to Agency a copy of the independent audit it receives in accordance with <u>ORC</u> <u>5103.0323</u>.
- B. Upon request from the Agency, Provider shall submit a copy of the most recent Federal income tax return and related schedules filed with the Internal Revenue Service (IRS).
- C. If Provider participates in the Title IV-E program, Provider agrees to timely file its Title IV-E cost report with all required items as outlined in <u>OAC 5101:2-47-26.2</u> to ODJFS. Provider agrees that in the event a cost report cannot be timely filed, an extension shall be requested prior to the December 31st filing deadline.
- D. If a Provider participates in the Title IV-E program, an Agreed Upon Procedures engagement must be conducted by a certified public accountant for the Provider's cost report in accordance with <u>OAC 5101:2-47-26.2.</u>The procedures are conducted to verify the accuracy of costs used to establish reimbursement ceilings for maintenance and administration costs of child in care. Any overpayments or underpayment of federal funds to the Title IV-E Agency due to adjustments of cost report reimbursement ceiling amounts as a result of an audit, shall be resolved in accordance with <u>ORC 5101.11</u>, <u>ORC 5101.14</u>, and <u>OAC 5101:2-47-01</u>.
- E. Upon request from the Agency, the Provider shall submit a copy of the JFS 02911 and Agreed Upon Procedures.
- F. For financial reporting purposes and for Title IV-E cost reporting purposes, Provider agrees to follow the cost

principles set forth in the following OAC Sections and publications:

- 1. <u>OAC 5101:2-47-11</u>: "Reimbursement for Title IV-E foster care maintenance (FCM) costs for children's residential centers (CRC), group homes, maternity homes, residential parenting facilities, private foster homes, and substance use disorder (SUD) residential facilities".
- 2. <u>OAC 5101:2-47-26.1</u>: "Public child services agencies (PCSA), private child placing agencies (PCPA), private noncustodial agencies (PNA), residential care facilities, substance use disorder (SUD) residential facilities: Title IV-E cost report filing requirements, record retention requirements, and related party disclosure requirements";
- 3. OAC 5101:2-47-26.2: "Cost Report Agreed Upon Procedures Engagement".
- 4. JFS 02911 Single Cost Report Instructions.
- 5. For Private Agencies: 2 CFR part 230, Cost Principles for Non-Profit Organizations.
- 6. For Public Agencies: 2 CFR part 225, Cost Principles for State, Local and Indian Tribal Government.
- 7. 2 CFR part 200.501, Audit Requirements.

Article XIV. GRIEVANCE/DISPUTE RESOLUTION PROCESS

In the event that a dispute arises under the provisions of this Agreement, the parties shall follow the procedures set forth below:

- 1. The party complaining of a dispute shall provide written notice of the nature of the dispute to the other party to this Agreement. A copy of the notice shall be sent to the Director or designee of the Agency and to the Executive Director or designee of the Provider. Within ten (10) business days of receiving the notice of a dispute, the parties involved in the dispute between the Agency and the Provider shall attempt to resolve the dispute.
- 2. If the parties are unable to resolve the dispute in (1 business day), the highest official or designee of the Agency shall make the final determination within twenty (20) business days, which will be non-binding.
- 3. Neither party will be deemed to have waived any other rights or remedies available to them by initiating, participating in or completing this process.

Article XV. ATTACHMENTS/ADDENDA

This Agreement, Attachments, and all Exhibits hereto constitutes the entire Agreement and may be amended only with a written Addendum signed by both parties; however, it is agreed by the parties that any Addenda to laws or regulations cited herein will result in the correlative modification of this Agreement, without the necessity for executing written Addenda. The impact of any applicable law, statute, or regulation not cited herein and enacted after the date of execution of this Agreement by written Addendum signed by both parties and effective as of the date of enactment of the law, statute, or regulation. Any other written Addendum to this Agreement is prospective in nature.

Article XVI. NOTICE

Unless otherwise set forth herein, all notices, requests, demands and other communications pertaining to this Agreement shall be in writing and shall be deemed to have been duly given if delivered or mailed by certified or registered mail, postage pre-paid:

if to Agency, to	Fairfield County Department of Job and Family Services 239 W Main St Lancaster, OH 43130
if to Provider, to	Victory House LLC 4276 Linchmere Dr Dayton, OH 45415

Article XVII. CONSTRUCTION

This Agreement shall be governed, construed, and enforced in accordance with the laws of the State of Ohio. Should

any portion of this Agreement be found to be unenforceable by operation of statute or by administrative or judicial decision, the operation of the balance of this Agreement is not affected thereby; provided, however, the absence of the illegal provision does not render the performance of the remainder of the Agreement impossible.

Article XVIII. NO ASSURANCES

- A. Provider acknowledges that, by entering into this Agreement, Agency is not making any guarantees or other assurances as to the extent, if any, that Agency shall utilize Provider's services or purchase its goods. In this same regard, this Agreement in no way precludes, prevents, or restricts Provider from obtaining and working under additional arrangement(s) with other parties, assuming the work in no way impedes Provider's ability to perform the services required under this Agreement. Provider warrants that at the time of entering into this Agreement, it has no interest in nor shall it acquire any interest, direct or indirect, in any Agreement that will impede its ability to provide the goods or perform the services under this Agreement.
- B. This Agreement, Attachments, and all Exhibits embodies the entire agreement of the Parties. There are no promises, terms, conditions or obligations other than those contained herein; and this Agreement shall supersede all previous communications, representations or Agreements, either written or oral, between the parties to this Agreement. Also, this Agreement shall not be modified in any manner except by an instrument, in writing, executed by both the parties.

Article XIX. CONFLICT OF INTEREST

- A. Provider agrees that the Provider, its officers, members and employees currently have no, nor will they acquire any interest, whether personal, professional, direct or indirect, which is incompatible, in conflict with or which would compromise the discharge and fulfillment of Provider's functions, duties and responsibilities hereunder. If the Provider, or any of its officers, members or employees acquire any incompatible, conflicting, or compromising personal or professional interest, the Provider shall immediately disclose, in writing, such interest to the Agency. If any such conflict of interest develops, the Provider agrees that the person with the incompatible, conflicting, or compromising personal or professional interest will not participate in any activities related to this Agreement.
- B. Provider agrees: (1) to refrain from promising or giving to Agency employees anything of value to manifest improper influence upon the employee; (2) to refrain from conflicts of interest; and, (3) to certify that Provider complies with <u>ORC 102.03</u>, <u>ORC 102.04</u>, <u>ORC 2921.42</u>, <u>ORC 2921.43</u>.
- C. The Provider further agrees that there is no financial interest involved on the part of the Agency or the respective county authority(ies) governing the agency. The Provider has no knowledge of any situation which would be a conflict of interest. It is understood that a conflict of interest occurs when an Agency employee or county official will gain financially or receive personal favors as a result of signing or implementation of this agreement. The Provider will report the discovery of any potential conflict of interest to the Agency. Should a conflict of interest be discovered during the term of this agreement, the Agency may exercise any right under the agreement, including termination of the agreement.

Article XX. INSURANCE

The Provider shall purchase and maintain for the term of this Agreement insurance of the types and amounts identified herein. Maintenance of the proper insurance for the duration of the Agreement is a material element of the Agreement.

Provider agrees to procure and maintain for the term of this Agreement the insurance set forth herein. The cost of all insurance shall be borne by Provider. Insurance shall be purchased from a company licensed to provide insurance in Ohio. Insurance is to be placed with an insurer provided an A.M. Best rating of no less than A-. Provider shall purchase the following coverage and minimum limits:

A. Commercial general liability insurance policy with coverage contained in the most current Insurance Services Office Occurrence Form CG 00 01 or equivalent with limits of at least One Million Dollars (\$1,000,000.00) per occurrence and One Million Dollars (\$1,000,000.00) in the aggregate and at least One Hundred Thousand Dollars (\$100,000.00) coverage in legal liability fire damage. Coverage will include:

- 1. Additional insured endorsement;
- 2. Product liability;
- 3. Blanket contractual liability;
- 4. Broad form property damage;
- 5. Severability of interests;
- 6. Personal injury; and
- 7. Joint venture as named insured (if applicable).

Endorsements for physical abuse claims and for sexual molestation claims must be a minimum of Three Hundred Thousand Dollars (\$300,000.00) per occurrence and Three Hundred Thousand Dollars (\$300,000.00) in the aggregate.

- B. Business auto liability insurance of at least One Million Dollars (\$1,000,000.00) combined single limit, on all owned, non-owned, leased and hired automobiles. If the Agreement contemplates the transportation of the users of County services (such as but not limited to Agency consumers), "Consumers" and Provider provides this service through the use of its employees' privately owned vehicles "POV", then the Provider's Business Auto Liability insurance shall sit excess to the employees "POV" insurance and provide coverage above its employee's "POV" coverage. Provider agrees the business auto liability policy will be endorsed to provide this coverage.
- C. Professional liability (errors and omission) insurance of at least One Million Dollars (\$1,000,000.00) per claim and in the aggregate.
- D. Umbrella and excess liability insurance policy with limits of at least One Million Dollars (\$1,000,000.00) per occurrence and in the aggregate, above the commercial general and business auto primary policies and containing the following coverage:
 - 1. Additional insured endorsement;
 - 2. Pay on behalf of wording;
 - 3. Concurrency of effective dates with primary;
 - 4. Blanket contractual liability;
 - 5. Punitive damages coverage (where not prohibited by law);
 - 6. Aggregates: apply where applicable in primary;
 - 7. Care, custody and control follow form primary; and
 - 8. Drop down feature.

The amounts of insurance required in this section for General Liability, Business Auto Liability and Umbrella/Excess Liability may be satisfied by Provider purchasing coverage for the limits specified or by any combination of underlying and umbrella limits, so long as the total amount of insurance is not less than the limits specified in General Liability, Business Auto Liability and Umbrella/Excess Liability when added together.

- E. Workers' Compensation insurance at the statutory limits required by ORC.
- F. The Provider further agrees with the following provisions:
 - 1. All policies, except workers' compensation and professional liability, will endorse as additional insured the Board of County Commissioners, and Agency and their respective officials, employees, agents, and volunteers, including their Board of Trustees if applicable. The additional insured endorsement shall be on an ACORD or ISO form.
 - 2. The insurance endorsement forms and the certificate of insurance forms will be sent to the Agency Director or Designee. The forms must state the following: "Board of County Commissioners, and Agency and their respective officials, employees, agents, and volunteers are endorsed as additional insured as required by agreement on the commercial general, business auto and umbrella/excess liability policies."
 - 3. Each policy required by this clause shall be endorsed to state that coverage shall not be canceled or materially changed except after thirty (30) calendar days prior written notice given to the Agency Director or Designee.
 - 4. Provider shall furnish the Agency with original certificates and amendatory endorsements effecting coverage required by this clause. All certificates and endorsements are to be received by the Agency

before the Agreement commences. The Agency reserves the right at any time to require complete, certified copies of all required insurance policies, including endorsements affecting the coverage required by these specifications.

- 5. Failure of the Agency to demand such certificate or other evidence of full compliance with these insurance requirements or failure of the Agency to identify a deficiency from evidence provided shall not be construed as a waiver of Provider's obligation to maintain such insurance.
- 6. Provider shall declare any self-insured retention to the Agency pertaining to liability insurance. Provider shall provide a financial guarantee satisfactory to the Agency guaranteeing payment of losses and related investigations, claims administration and defense expenses for any self-insured retention.
- 7. If Provider provides insurance coverage under a "claims-made" basis, Provider shall provide evidence of either of the following for each type of insurance which is provided on a claims-made basis: unlimited extended reporting period coverage, which allows for an unlimited period of time to report claims from incidents that occurred after the policy's retroactive date and before the end of the policy period (tail coverage), or; continuous coverage from the original retroactive date of coverage. The original retroactive date of coverage means original effective date of the first claim-made policy issued for a similar coverage while Provider was under Agreement with the County on behalf of the Agency.
- 8. Provider will require all insurance policies in any way related to the work and secured and maintained by Provider to include endorsements stating each underwriter will waive all rights of recovery, under subrogation or otherwise, against the County and the Agency. Provider will require of subcontractors, by appropriate written agreements, similar waivers each in favor of all parties enumerated in this section.
- 9. Provider, the County, and the Agency agree to fully cooperate, participate, and comply with all reasonable requirements and recommendations of the insurers and insurance brokers issuing or arranging for issuance of the policies required here, in all areas of safety, insurance program administration, claim reporting and investigating and audit procedures.
- 10. Provider's insurance coverage shall be primary insurance with respect to the County, the Agency, their respective officials, employees, agents, and volunteers. Any insurance maintained by the County or the Agency shall be excess of Provider's insurance and shall not contribute to it.
- 11. If any of the work or Services contemplated by this Agreement is subcontractors, Provider will ensure that any subcontractors comply with all insurance requirements contained herein.
- 12. If the Agreement provider is a government entity, insurance requirements will be fulfilled under the County Risk Sharing Authority (CORSA).

Article XXI. INDEMNIFICATION & HOLD HARMLESS

- A. To the fullest extent permitted by, and in compliance with, applicable law, Provider agrees to protect, defend, indemnify and hold harmless the Agency and the Board of County Commissioners, their respective members, officials, employees, agents, and volunteers (the "Indemnified Parties") from and against all damages, liability, losses, claims, suits, actions, administrative proceedings, regulatory proceedings/hearings, judgments and expenses, subrogation (of any party involved in the subject of this Agreement), attorneys' fees, court costs, defense costs or other injury or damage (collectively "Damages"), whether actual, alleged or threatened, resulting from injury or damages of any kind whatsoever to any business, entity or person (including death), or damage to property (including destruction, loss of, loss of use of resulting without injury damage or destruction) of whatsoever nature, arising out of or incident to in any way, the performance of the terms of this Agreement including, without limitation, by Provider, its subcontractor(s), Provider's or its subcontractor(s') employees, agents, assigns, and those designated by Provider to perform the work or services encompassed by the Agreement. Provider agrees to pay all damages, costs and expenses of the Indemnified Parties in defending any action arising out of the aforementioned acts or omissions.
- B. Each Party agrees to be responsible for any personal injury or property damage caused solely by its negligent acts or omissions as determined by a court of competent jurisdiction, or as the parties may otherwise mutually agree in writing.
- C. This Article is not applicable to Agreements between governmental entities.

Article XXII. SCREENING AND SELECTION

A. Criminal Record Check

- 1. Provider warrants and represents it will comply with Article X as it relates to criminal record checks. Provider shall insure that every individual subject to a Bureau of Criminal Investigation (BCI) criminal records check will sign a release of information to allow inspection and audit of the above criminal records transcripts or reports by the Agency or a private vendor hired by the Agency to conduct compliance reviews on their behalf.
- 2. Provider shall not assign any individual to work with or transport children until a BCI report and a criminal record transcript has been obtained.
- 3. Except as provided in Section C below, Provider shall not utilize an employee, foster caregiver or all of the above who has been convicted or plead guilty to any violations contained in <u>ORC 5153.111(B)(1)</u>, <u>ORC 2919.24</u>, and <u>ORC 2151.86</u>, and <u>OAC Chapters 5101:2-5</u>, 5101:2-7, 5101:2-9, 5101:2-48.
- 4. Provider agrees to be financially responsible for any of the following requirements in <u>OAC Chapters</u> <u>5101:2-5</u>, <u>5101:2-7</u>, <u>5101:2-9</u> and <u>5101:2-48</u> resulting in financial penalty due to lack of compliance with the criminal records checks.

B. Transportation of Child

- 1. The caregiver shall ensure the transportation of children in care will be reliable, legal and safe transportation with safety restraints, as appropriate for the child, and must be in compliance with applicable local, state and Federal transportation laws:
 - a. Maintenance of a current valid driver's license and vehicle insurance.
 - b. All children being transported by Provider must follow Ohio's Child Passenger Safety Law as defined in <u>ORC 4511.81.</u>
 - c. No child that is a passenger and is required to have a seat restraint can be transported by said provider until these requirements are met.
- 2. In addition to the requirements set forth above, Provider shall not permit any individual to transport a Child if:
 - a. The individual has a condition which would affect safe operation of a motor vehicle;
 - b. The individual has six (6) or more points on his/her driver's license; or
 - c. The individual has been convicted of, or pleaded guilty to, a violation of section <u>4511.19</u> (Operating vehicle under the influence of alcohol or drugs OVI or OVUAC) of the Revised Code if the individual previously was convicted of or plead guilty to two or more violations within the three years immediately preceding the current violation.

C. Rehabilitation

- 1. Notwithstanding the above, Provider may make a request to the Agency to utilize an individual if Provider believes the individual has met the rehabilitative standards of <u>OAC 5101:2-07-02(I)</u> as follows:
 - a. If the Provider is seeking rehabilitation for a foster caregiver, a foster care applicant or other resident of the foster caregiver's household, Provider must provide written verification that the rehabilitation standards of <u>OAC 5101:2-7-02</u> have been met.
 - b. If the Provider is seeking rehabilitation for any other individual serving Agency children, Provider must provide written verification from the individual that the rehabilitative conditions in accordance with <u>OAC 5101:2-5-09</u> have been met.
- 2. The Agency shall review the facts presented and may allow the individual to work with, volunteer with or transport Agency children on a case-by-case basis. It is the Agency's sole discretion to permit a rehabilitated individual to work with, volunteer with or transport children.
- D. Verification of Job or Volunteer Application:

Provider shall check and document each applicant's personal and employment references, general work history, relevant experience, and training information. Provider further agrees it will not employ an individual in relation to this Agreement unless it has received satisfactory employment references, work history, relevant experience, and training information.

Article XXIII. PROHIBITION OF CORPORAL & DEGRADING PUNISHMENT

Agency prohibits the use of corporal or degrading punishment against children served by Agency and must comply with requirements in <u>OAC 5101:2-7-09</u>, <u>OAC 5101:2-9-21</u>, and <u>OAC 5101:2-9-22</u>

Article XXIV. FINDINGS FOR RECOVERY

<u>ORC 9.24</u> prohibits public agencies from awarding an Agreement for goods, services, or construction paid for in whole or in part from federal, state and local funds, to an entity against whom a finding for recovery has been issued if the finding is unresolved. By entering into this Agreement, Provider warrants and represents that they do not have an unresolved finding for recovery. Provider shall notify the Agency within ten (10) business days of its notification should the Provider be issued such finding by the Auditor of the State.

Article XXV. PUBLIC RECORDS

This Agreement is a matter of public record under the Ohio public records law. By entering into this Agreement, Provider acknowledges and understands that records maintained by Provider pursuant to this Agreement may also be deemed public records and subject to disclosure under Ohio law. Upon request made pursuant to Ohio law, the Agency shall make available the Agreement and all public records generated as a result of this Agreement.

Article XXVI. CHILD SUPPORT ENFORCEMENT

Provider agrees to cooperate with ODJFS and any Ohio Child Support Enforcement Agency ("CSEA") in ensuring Provider and Provider's employees meet child support obligations established under state or federal law. Further, by executing this Agreement, Provider certifies present and future compliance with any court or valid administrative order for the withholding of support which is issued pursuant to the applicable sections in <u>ORC Chapters 3119, 3121, 3123</u>, and <u>3125</u>.

Article XXVII. DECLARATION OF PROPERTY TAX DELINQUENCY

After award of an Agreement, and prior to the time the Agreement is entered into, the successful Provider shall submit a statement in accordance with <u>ORC 5719.042</u>. Such statement shall affirm under oath that the person with whom the Agreement is to be made was not charged at the time the bid was submitted with any delinquent personal property taxes on the general tax list of personal property of any county in which the taxing district has territory, or that such person was charged with delinquent personal property taxes on any such tax list, in which case the statement shall also set forth the amount of such due and unpaid delinquent taxes any due and unpaid penalties and interest thereon. If the statement indicates that the taxpayer was charged with any such taxes, a copy of the statement shall be transmitted by the fiscal officer to the county treasurer within thirty days of the date it is submitted.

A copy of the statement shall also be incorporated into the Agreement, and no payment shall be made with respect to any contract to which this section applies unless such statement has been so incorporated as a part thereof.

Article XXVIII. SUBCONTRACTING AND DELEGATION

The performance of any duty, responsibility or function which is the obligation of the Provider under this Agreement may be delegated or subcontracted to any agent or subcontractor of Provider if Provider has obtained the prior written consent of the Agency for that delegation subcontract. Provider is responsible for ensuring that the duties, responsibilities or functions so delegated or subcontracted are performed in accordance with the provisions and standards of this Agreement, and the actions and omissions of any such agent or subcontractor shall be deemed to be the actions and omissions of Provider for purposes of this Agreement.

Article XXIX. PROPERTY OF AGENCY

The Deliverable(s) and any item(s) provided or produced pursuant to this Agreement (collectively called "Deliverables") will be considered "works made for hire" within the meaning of copyright laws of the United States of America and the State of Ohio. The Agency is the sole author of the Deliverables and the sole owner of all rights therein. If any portion of

the Deliverables are deemed not to be a "work made for hire", or if there are any rights in the Deliverables not so conveyed to the Agency, then Provider agrees to, and by executing this Agreement hereby does, assign to the Agency all worldwide rights, title, and interest in and to the Deliverables. The Agency acknowledges that its sole ownership of the Deliverables under this Agreement does not affect Provider's right to use general concepts, algorithms, programming techniques, methodologies, or technology that have been developed by Provider prior to this Agreement or that are generally known and available. Any Deliverable provided or produced by Provider under this Agreement or with funds hereunder, including any documents, data, photographs and negatives, electronic reports/records, or other media, are the property of the Agency, which has an unrestricted right to reproduce, distribute, modify, maintain, and use the Deliverables. Provider shall not obtain copyright, patent, or other proprietary protection for the Deliverables. Provider shall not include in any Deliverable any copyrighted material, unless the copyright owner gives prior written approval for the Agency and Provider to use such copyrighted material. Provider agrees that all Deliverables will be made freely available to the general public unless the Agency determines that, pursuant to state or federal law, such materials are confidential or otherwise exempt from disclosure.

Article XXX. SEVERABILITY

If any term of this Agreement or its application thereof to any person or circumstance shall to any extent be held invalid or unenforceable, the remainder of this Agreement, or the application of such term or provision to persons or circumstances other than those as to which it is held invalid or unenforceable, shall not be affected thereby. Each term and provision of this Agreement shall be valid and enforced to the fullest extent permitted by law.

Article XXXI. NO ADDITIONAL WAIVER IMPLIED

If the Agency or Provider fails to perform any obligations under this Agreement and thereafter such failure is waived by the other party, such waiver shall be limited to the particular matter waived and shall not be deemed to waive any other failure hereunder, nor a waiver of a subsequent breach of the same provision or condition. Waivers shall not be effective unless in writing.

Article XXXII. COUNTERPARTS

This Agreement may be executed as an original document only, or simultaneously in two or more counterparts, each of which shall be deemed an original, and each of these counterparts shall constitute one and the same instrument. It shall not be necessary in making proof of this Contract to produce or account for more than one such counterpart An electronic signature or a scanned or otherwise reproduced signature shall be a binding signature and carry the same legal force as the original.

Article XXXIII. APPLICABLE LAW AND VENUE

This Agreement and any modifications, Attachments, Exhibits, Addenda, or alterations, shall be governed, construed, and enforced under the laws of Ohio. Any legal action brought pursuant to this agreement will be filed in the Ohio courts, and Ohio law as well as Federal law will apply.

ATTACHMENT

Attachment One.

Reason: Article Section: Article I - Scope of Placement Services Detail: Article I SECTIONS 1.02 & 1.03, References to Exhibit I Article I, Item A ('Scope of Placement Services') will serve as Exhibit I.

Attachment Two.

Reason: Article Section: Article V - Provider Responsibilities Detail: Article V ITEM A

Provider is responsible for ensuring transportation services are in place for all case-related activities and routine needs. If extenuating circumstances exist and provider is not able to utilize its own resources to ensure coverage, Provider must notify Agency in advance of the transportation need, and coordinate with Agency to arrange needed transportation. Provider should bring any extraordinary travel needs to the attention of Agency so both parties can ensure proper coverage and explore potential compensation for needed transportation services.

ITEM B

Progress reports, demonstration completion of monthly activities as required by Ohio Administrative Code, will be submitted on a monthly basis, by no later than the 20th calendar day of the following month.

ITEMS D, E, & F

After-Hours/On-Call Process

Provider shall notify Agency of any items identified in Article V, according to the following options: For calls during business hours (Monday through Friday, from 8:00 a.m. to 4:00 p.m., excluding holidays), Provider shall call (740) 652-7854 and inform the operator of the need to urgently speak to casework staff. For calls outside of business hours, Provider shall call (740)808-0009 or (740)808-0982 in order to notify Agency's On-Call staff member of the urgent situation.

Insert new item - ITEM S

Provider will ensure access to Normalcy activities, based upon the developmental, social, and emotional functioning of each child placement.

Insert new item - ITEM T

All Provider staff, and foster caregivers when applicable, should seek and receive prior authorization from Agency for any type of non-routine medical care or medication needs. This includes, but is not limited to: major medical treatment, medical procedures, surgery, implementation of or change in psychotropic medications, and any other medical intervention that carries a high risk of side effects, impairment, or harm. Routine well visits and treatment for typical childhood illnesses will not require such prior authorization.

In the event of an emergency, the child should be taken to the nearest medical facility for prompt treatment. As soon as possible, Agency shall be contacted, according to the process outlined in the addendum to Article V.

Attachment Three.

Reason: Article Section: Article VIII - Reimbursement for Placement Services Detail: Article VIII ITEM B Agency agrees to pay Provider on the basis of a daily per diem (identified in Schedule A of this agreement) for the placement for each child, as identified by each child's current Level of Care (LOC). The LOC will be agreed upon at the time of placement. The Agency and Provider may request a re-evaluation of the child's LOC at any time in order to best meet the child's identified needs.

ITEM D

Agency agrees to provide a one-time initial clothing authorization of up to \$150.00 for children ten and under and up to \$250.00 for children over the age of ten. If the child is under ten and wearing adult sizes, the authorization will be for \$250.00. Any purchases beyond this must be due to extenuating circumstances and approved in writing, in advance of the purchase. Routine clothing needs are considered part of maintenance and will not be paid or reimbursed by Agency.

ITEM E

If a child goes on any form of unpaid leave and is reasonably expected to return to the same placement, Agency may pay for up to seven (7) consecutive nights of leave. In order for Agency to pay for such leave, Provider (including foster parents, if applicable) must agree to remain available for regular services and needed support during such leave. Leave beyond seven (7) days will not be paid unless extenuating circumstances exist, and both Agency and Provider agree to payment terms in advance, in writing.

Regardless of length, Agency will not pay Provider for any leave during which payment is being made to another provider, nor for leave where the child is in a paid or unpaid alternative placement outside of Provider's network due to a lack of placement availability with Provider. Any deviation from this must be agreed to in advance, in writing, by both Agency and Provider.

Item F

Medicaid/Insurance

Upon receipt of formal documentation, FCCPS will submit to Service Provider Medicaid/insurance numbers for children in FCCPS custody, as applicable. The service provider is required to utilize Medicaid-approved healthcare providers in the appropriate managed care network for the provision of mental health,dental and/or medical services (hereafter referred to collectively as "medical services") to children in the custody of FCCPS. The Service Provider will report applicable Medicaid/insurance information to the healthcare providers and instruct healthcare providers to seek payment from Medicaid or any other available third party payor for medical services rendered to children in FCCPS custody. FCCPS will not pay for the provision of any medical services to children in FCCPS Deputy Director or authorized designee has provided specific prior written authorization for such medical services and associated costs.

In situations where the Service Provider does not possess a Medicaid/insurance number or other information required to bill an alternative source for services provided to children in the custody of FCCPS, the Service Provider must take the following actions.

A. The Service Provider will contact the Finance Department at (740)652-7889 for assistance with resolving Medicaid/insurance number issues.

B. Within thirty (30) days if an invoice from a healthcare provider for services rendered to a child in FCCPS custody, the Service Provider should forward the invoice to the FCCPS Finance Department at: Fairfield County Job and Family Services-Protective Services, 239 West Main St., Lancaster, OH 43130 or fax such invoice to the FCCPS Finance Department at (740)-687-7070. Failure to forward this invoice to FCCPS within thirty (30)days will constitute a waiver of any claim against FCCPS for payment of the invoice. If the Service Provider receives additional notices regarding the invoice, the Service Provider must contact the FCCPS Finance Department at (740)-652-7889 to confirm that FCCPS received the initial invoice and to obtain the status of payment arrangements. The Service Provider SHALL NOT pay the invoice and expect or request reimbursement from FCCPS without the prior written approval of FCCPS.

C. If a child who is in custody of FCCPS requires pharmaceutical supplies, Service Provider must obtain the supplies from a pharmacy that accepts Medicaid/insurance payments.

ITEM G

In any instance where payment cannot be made within 30 days, Agency will make every effort to ensure Provider is

paid within 45 days, and is made aware in advance if this is not possible.

FCCPS retains the right to recoup funds from the Service Provider upon the determination that third party funds are duplicative (in the aggregate) of FCCPS payments to the Service Provider, or in the event that the Service Provider fails to properly credit any and all such third party payments. Relative to recouping funds, FCCPS may withhold from subsequent reimbursement to the Service Provider an amount equal to any un-credited or duplicate third party payments. For purposes of this paragraph, "third party" includes, but is not limited to , Medicaid and private insurance companies.

The Service Provider shall obtain and provide a written estimate for any non-routine, non-emergency, or out-ofnetwork medical and dental expenses to FCCPS along with the written recommendation of the physician or dentist. The Servide Provider is not permitted to deliver or authorize any health/dental care or treatment services (including, but not limited to, mental health services), without the prior written consent of the FCCPS Deputy Director or authorized designee (see Consent for Medical Treatment letter).

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date of the signature of the parties.

SIGNATURES OF PARTIES:

Provider: Victory House LLC

Print Name & Title	Signature	Date			
Nicole Valushin Director	A	5/1/2023			

Agency: Fairfield County Department of Job and Family Services

Print Name & Title	Signature	Date		

Additional Signatures

Print Name & Title	Signature	Date		
		100		

Contract ID. 19342482 Fairfield County Department of Job and Family Services / Victory House LLC 04/01/2023 - 03/31/2024 Page 24 of 25

Title IV-E Schedule A Rate Information

Title IV-E Schedule A Rate Information Agency: Fairfield County Department of Job and Family Services Provider / ID: Victory House LLC / 25501067

Run Date: 05/01/2023 Contract Period: 04/01/2023 - 03/31/2024

Service Description	Service ID	Person	Person ID	Maintenance Per Diem	Administration Per Diem	Case Management Per Diem	Transportation / Administration Per Diem	Transporation / Maintenance Per Diem	Other Direct Services Per Diem	Behavioral Healthcare Per Diem	Other Per Diem Cost	Total Per Diem Cost	Cost Begin Date	Cost End Date
Hillary Street Group Home (20979)	7657413			\$265.00	\$10.00						\$270.00	\$545.00	04/01/2023	03/31/2024
Hillary Street Group Home (20979)	7657413			\$265.00	\$10.00							\$275.00	04/01/2023	03/31/2024
Linchmere Drive Group Home (20919)	7475663			\$265.00	\$10.00						\$270.00	\$545.00	04/01/2023	03/31/2024
Linchmere Drive Group Home (20919)	7475663			\$265.00	\$10.00							\$275.00	04/01/2023	03/31/2024



A Contract regarding Victory House LLC between Job and Family Services and

Approved on 5/30/2023 8:00:58 AM by Sarah Fortner, Assistant Deputy Director

South Jota

Sarah Fortner Assistant Deputy Director

Approved on 5/30/2023 11:14:14 AM by Corey Clark, Director of Fairfield County Job & Family Services

GBCh_

Corey Clark, Director Fairfield County Job & Family Services

Certified Search for Unresolved Findings for Recovery



Auditor of State - Unresolved Findings for Recovery Certified Search

I have searched The Auditor of State's unresolved findings for recovery database using the following criteria:

Contractor's Information:

Name: , Organization: Victory House Group Home Date: 5/16/2023 3:59:36 PM

This search produced the following list of **13** possible matches:

Name/Organization	Address
African Refugee Educational and Cultural Services	3800 Sullivant Avenue
Anew Educational Services	11470 Euclid Ave. #170
Anew Educational Services	11470 Euclid Avenue #170
Eagle Eye Services, LLC	P.O. Box 8288
Education Recruiting Services	4601 E. 5th Avenue
Ellington Management Services	6581 Westminster Court
Exceptional Psychological Services, Phoenix Village Academy P2	1588 East 40th St.
Global Bilingual Services	5767 Vienna Drive
Greater Educational Service Center, Phoenix Village Academy P2	5455 North Marginal Road, Suite 521
Greater Educational Service Center, Phoenix Village Academy S1	5455 North Marginal Road, Suite 521
Lawrence County Educational Service Center	304 N. 2nd Street
MacWin Protocol Technology Services, Inc.	12328 Granger Road
New Opportunity Development and Management Services, Inc., c/o Mark Olds	38510 Flanders Drive

The above list represents possible matches for the search criteria you entered. Please note that pursuant to ORC 9.24, only the person (which includes an organization) actually named in the finding for recovery is prohibited from being awarded a contract.

If the person you are searching for appears on this list, it means that the person has one or more findings for recovery and is prohibited from being awarded a contract described in ORC 9.24, unless one of the exceptions in that section apply.

If the person you are searching for does not appear on this list, an initialed copy of this page can serve as documentation of your compliance with ORC 9.24(E).

Please note that pursuant to ORC 9.24, it is the responsibility of the public office to verify that a person to whom it plans to award a contract does not appear in the Auditor of State's database. The Auditor of State's office is not responsible for inaccurate search results caused by user error or other circumstances beyond the Auditor of State's control.

Office of Auditor of State 88 East Broad Street Post Office Box 1140 Columbus, OH 43216-1140

> (614) 466-4514 (800) 282-0370

Prosecutor's Approval Page

Resolution No.

A resolution authorizing the approval of a service agreement by and between Fairfield County Job & Family Services, Child Protective Services Division and Victory House LLC.

(Fairfield County Job and Family Services)

Approved as to form on 6/1/2023 11:03:45 AM by Steven Darnell,

Signature Page

Resolution No. 2023-06.06.s

A resolution authorizing the approval of a service agreement by and between Fairfield County Job & Family Services, Child Protective Services Division, and Victory House LLC.

(Fairfield County Job and Family Services)

Upon the motion of Commissioner Jeffrey M. Fix, seconded by Commissioner David L. Levacy, this resolution has been Adopted:

Voting:

Steven A. Davis, President	Aye
David L. Levacy, Vice President	Aye
Jeffrey M. Fix	Aye

Board of County Commissioners Fairfield County, Ohio

CERTIFICATE OF CLERK

It is hereby certified that the foregoing is a true and correct transcript of a resolution acted upon by the Board of County Commissioners, Fairfield County, Ohio on the date noted above.

Rochelle Merringen

Rochelle Menningen Board of County Commissioners Fairfield County, Ohio

2023-06.06.t

A resolution to appropriate from unappropriated in a major expenditure object category – Juvenile Court Fund #2408 Drug Court Program Sub Fund #8209

WHEREAS, additional appropriations are needed in the major expenditure object category for Fund 2408 Drug Court Program Sub Fund 8209; and

WHEREAS, appropriate from unappropriated will allow proper accounting in the major expenditure object category.

NOW THEREFORE, BE IT RESOLVED BY THE BOARD OF COUNTY COMMISSIONERS, COUNTY OF FAIRFIELD, STATE OF OHIO:

Section 1. The Fairfield County Board of Commissioners appropriate from unappropriated into the following category:

\$2,500.00 17820900 Materials & Supplies

For Auditor's Office Use Only:

17820900-563000 \$2,500.00

Prepared by: Alisha Mason cc: Juvenile Court Signature Page

Resolution No. 2023-06.06.t

A resolution to appropriate from unappropriated in a major expenditure object category Fund #2408, Drug Court Program Sub Fund #8209.

(Fairfield County Juvenile/Probate Court)

Upon the motion of Commissioner Jeffrey M. Fix, seconded by Commissioner David L. Levacy, this resolution has been Adopted:

Voting:

Steven A. Davis, President	Aye
David L. Levacy, Vice President	Aye
Jeffrey M. Fix	Aye

Board of County Commissioners Fairfield County, Ohio

CERTIFICATE OF CLERK

It is hereby certified that the foregoing is a true and correct transcript of a resolution acted upon by the Board of County Commissioners, Fairfield County, Ohio on the date noted above.

Pochelle Meninger

Rochelle Menningen Board of County Commissioners Fairfield County, Ohio 2023-06.06.u

A resolution to approve the contract award for the Airport Sewer Extension and Water Service Project; ARP fiscal recovery fund, #2876

WHEREAS, the Board of County Commissioners authorized use of the American Rescue Plan fiscal recovery funds for the construction of the Airport Sewer Extension and Water Service Project with resolution 2021-09.14.e;

WHEREAS, the Fairfield County Utilities Department received competitive bids for the Airport Sewer Extension and Water Service Project on May 11, 2023;

WHEREAS, the low bidder was Law General Contracting INC., with a bid price of \$495,000.00 which was less than the engineer's estimate of \$695,791.00;

WHEREAS, the Director of Utilities has reviewed the bids and recommends award of the Airport Sewer Extension and Water Service Project to Law General Contracting INC.;

WHEREAS, Assistant Prosecuting Attorney, Amy Brown-Thompson, has approved the agreement as to form;

NOW THEREFORE, BE IT RESOLVED BY THE BOARD OF COUNTY COMMISSIONERS, COUNTY OF FAIRFIELD, STATE OF OHIO, THAT:

Section 1. That the Board of Fairfield County Commissioners concurs with the recommendations of the Director of Fairfield County Utilities and authorizes itself to execute Contract Documents with Law General Contracting INC., to perform the work necessary to construct the Airport Sewer Extension and Water Service Project for the sum of \$495,000.00.

Section 2. That the Director of Fairfield County Utilities is directed to manage and administer the Contract and sign all appropriate documents up to the approved Contract Amount in an expeditious manner.

Prepared by: Joshua TC Anders cc: Utilities Department



06/06/2023

Fairfield County Utilities 6670 Lockville Road NW Carroll, Ohio 43112

Fairfield County Utilities Airport Sanitary Sewer Extension and Water Service Design

Project Manual

Bidding Requirements Contract Documents Technical Specifications

Prepared by:



8415 Pulsar Place, Suite 300 Columbus, Ohio 43240 614-839-0250

> April 6, 2023 PRIME AE Project #21513

THIS PAGE INTENTIONALLY LEFT BLANK

PROJECT TABLE OF CONTENTS

DIVISION 00 – PROCUREMENT AND CONTRACTING REQUIREMENTS

- 000101 PROJECT TITLE PAGE
- C-111 ADVERTISEMENT FOR BIDS
- C-200 INSTRUCTIONS TO BIDDERS
- C-410 BID FORM FOR CONSTRUCTION CONTRACT
- C-430 BID BOND PENAL SUM FORM
- C-510 NOTICE OF AWARD
- C-520 AGREEMENT BETWEEN OWNER AND CONTRACTOR
- C-550 NOTICE TO PROCEED
- C-610 PERFORMANCE BOND
- C-615 PAYMENT BOND
- C-625 CERTIFICATE OF SUBSTANTIAL COMPLETION
- C-626 NOTICE OF ACCEPTABILITY
- C-700 GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT
- C-800 SUPPLEMENTARY CONDITIONS OF THE CONSTRUCTION CONTRACT
- C-941 CHANGE ORDER
- C-942 FIELD ORDER
- 004536 EQUAL EMPLOYMENT OPPORTUNITY CERTIFICATION
- 004547 CERTIFICATION REGARDING DEBARMENT AND SUSPENSION
- 004548 AIS ACKNOWLEDGEMENT FORM
- 004549 BUILD AMERICA BUY AMERICA (BABA) ACKNOWLEDGEMENT
- 007343 WAGE DETERMINATION
- 007343a DAVIS-BACON ACT WAGE REQUIREMENTS FOR FAIRFIELD COUNTY

DIVISION 01 – GENERAL REQUIREMENTS

- 011000 SUMMARY
- 012000 PRICE AND PAYMENT PROCEDURES
- 012500 CONTRACT PAY ITEMS
- 013000 ADMINISTRATIVE REQUIREMENTS
- 013216 CONSTRUCTION PROGRESS SCHEDULE
- 013300 SUBMITTAL PROCEDURES
- 014000 QUALITY REQUIREMENTS
- 015000 TEMPORARY FACILITIES AND CONTROLS
- 017419 CONSTRUCTION WASTE MANAGEMENT AND DISPOSAL
- 017700 CLOSEOUT PROCEDURES

DIVISION 02 – EXISTING CONDITIONS

024119 SELECTIVE DEMOLITION

PROJECT TABLE OF CONTENTS

DIVISION 26 – ELECTRICAL

260010	BASIC ELECTRICAL REQUIREMENTS
260053	GENERAL WIRING

- 260054 CUTTING AND PATCHING
- 260060 EXCAVATING, TRENCHING, BACKFILLING, AND RESTORATION
- 260111 CONDUIT SYSTEMS
- 260120 WIRE AND CABLE
- 260140 WIRING DEVICES
- 260450 GROUNDING

DIVISION 33 – UTILITIES

333213 PACKAGED WASTEWATER PUMPING STATIONS

- END OF PROJECT TABLE OF CONTENTS -

PROJECT TABLE OF CONTENTS

2

SECTION 000101 - PROJECT TITLE PAGE

Title and Location of Work:	Airport Sanitary Sewer Extension and Water Service Design Fairfield County Airport 3430 Old Columbus Rd NW Carroll, OH 43112
Name and Address of Owner:	Fairfield County Utilities 6670 Lockville Road NW Carroll, Ohio 43112
Owner Contact:	Tony Vogel, PE Director of Utilities <u>tony.vogel@fairfieldcountyohio.gov</u> Josh Anders, PE Deputy Director of Utilities <u>josh.anders@fairfieldcountyohio.gov</u>
Engineer (Construction Services):	Mark Buchenic, PE Project Manager PRIME AE Group, Inc. <u>mbuchenic@primeeng.com</u>
Design Engineering Services:	PRIME AE Group, Inc. 8415 Pulsar Place, Suite 300 Columbus, OH 43240 614-839-0250

END OF SECTION 000101

PROJECT TITLE PAGE

000101 - 1

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

(THIS PAGE INTENTIONALLY LEFT BLANK)

PROJECT TITLE PAGE

000101 - 2

ADVERTISEMENT FOR BIDS

FAIRFIELD COUNTY UTILITIES FAIRFIELD COUNTY, OHIO AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

General Notice

Fairfield County Utilities (Owner) is requesting Bids for the construction of the following Project:

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN Project Location: 3430 Old Columbus Rd NW, Carroll, OH 43112

Bids for the construction of the Project will be received at the Fairfield County Utilities Building located at 6670 Lockville Rd NW, Carroll, Ohio 43112, until Thursday, May 11, 2023 at 10:30 AM local time. At that time the Bids received will be publicly opened and read.

The Engineer's opinion of probable construction cost is \$695,791.00 as of April 2023.

The Project includes the following Work: This project will install 613 linear feet of 8-inch gravity sewer main, 2 sanitary manholes, associated sanitary laterals, and water services for facilities at the Fairfield County Airport, Fairfield County, Ohio. The project will also install a dual-pump lift station and connect to an existing 2-inch sanitary force main.

Bids are requested for the following Contract: General Contract (all trades)

Obtaining the Bidding Documents

Information and Bidding Documents for the Project are available on the Fairfield County Utilities website at:

https://www.co.fairfield.oh.us/util/

Bidding Documents may be downloaded from the designated website. Prospective Bidders are urged to register with the designated website as a plan holder, even if Bidding Documents are obtained from a plan room or source other than the designated website in either electronic or paper format. The designated website will be updated periodically with addenda, lists of registered plan holders, reports, and other information relevant to submitting a Bid for the Project. All official notifications, addenda, and other Bidding Documents will be offered only through the designated website. Neither Owner nor Engineer will be responsible for Bidding Documents, including addenda, if any, obtained from sources other than the designated website.

Instructions to Bidders.

For all further requirements regarding bid submittal, qualifications, procedures, and contract award, refer to the Instructions to Bidders that are included in the Bidding Documents.

This Advertisement is issued by:

Owner:	Fairfield County Utilities
By:	Tony Vogel, PE
Title:	Director of Utilities
Date:	April 20, 2023

INSTRUCTIONS TO BIDDERS FOR CONSTRUCTION CONTRACT

TABLE OF CONTENTS

ge
.1
.1
.2
.3
.3
5
5
6
6
6
7
7
9
9
0
0
0
0
1
1
1
1

ARTICLE 1—DEFINED TERMS

- 1.01 Terms used in these Instructions to Bidders have the meanings indicated in the General Conditions and Supplementary Conditions. Additional terms used in these Instructions to Bidders have the meanings indicated below:
 - A. Issuing Office—The office from which the Bidding Documents are to be issued, and which registers plan holders.

ARTICLE 2—BIDDING DOCUMENTS

- 2.01 Bidder shall obtain a complete set of Bidding Requirements and proposed Contract Documents (together, the Bidding Documents). See the Agreement for a list of the Contract Documents. It is Bidder's responsibility to determine that it is using a complete set of documents in the preparation of a Bid. Bidder assumes sole responsibility for errors or misinterpretations resulting from the use of incomplete documents, by Bidder itself or by its prospective Subcontractors and Suppliers.
- 2.02 Bidding Documents are made available for the sole purpose of obtaining Bids for completion of the Project and permission to download or distribution of the Bidding Documents does not confer a license or grant permission or authorization for any other use. Authorization to download documents, or other distribution, includes the right for plan holders to print documents solely for their use, and the use of their prospective Subcontractors and Suppliers, provided the plan holder pays all costs associated with printing or reproduction. Printed documents may not be re-sold under any circumstances.
- 2.03 Owner has established a Bidding Documents Website as indicated in the Advertisement for Bids. Owner recommends that Bidder register as a plan holder with the Issuing Office at such website, and obtain a complete set of the Bidding Documents from such website. Bidders may rely that sets of Bidding Documents obtained from the Bidding Documents Website are complete, unless an omission is blatant. Registered plan holders will receive Addenda issued by Owner.
- 2.04 Deleted.
- 2.05 Plan rooms (including construction information subscription services, and electronic and virtual plan rooms) may distribute the Bidding Documents, or make them available for examination. Those prospective bidders that obtain an electronic (digital) copy of the Bidding Documents from a plan room are encouraged to register as plan holders from the Bidding Documents Website or Issuing Office. Owner is not responsible for omissions in Bidding Documents or other documents obtained from plan rooms, or for a Bidder's failure to obtain Addenda from a plan room.
- 2.06 *Electronic Documents*
 - A. When the Bidding Requirements indicate that electronic (digital) copies of the Bidding Documents are available, such documents will be made available to the Bidders as Electronic Documents in the manner specified.
 - 1. Bidding Documents will be provided in Adobe PDF (Portable Document Format) (.pdf) that is readable by Adobe Acrobat Reader Version 10 or later. It is the intent of the Engineer and Owner that such Electronic Documents are to be exactly representative of the paper copies of the documents. However, because the Owner and Engineer cannot totally control the transmission and receipt of Electronic Documents nor the Contractor's means of reproduction of such documents, the Owner and Engineer cannot

EJCDC[®] C-200, Instructions to Bidders for Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

and do not guarantee that Electronic Documents and reproductions prepared from those versions are identical in every manner to the paper copies.

- B. Unless otherwise stated in the Bidding Documents, the Bidder may use and rely upon complete sets of Electronic Documents of the Bidding Documents, described in Paragraph 2.06.A above. However, Bidder assumes all risks associated with differences arising from transmission/receipt of Electronic Documents versions of Bidding Documents and reproductions prepared from those versions and, further, assumes all risks, costs, and responsibility associated with use of the Electronic Documents versions to derive information that is not explicitly contained in printed paper versions of the documents, and for Bidder's reliance upon such derived information.
- C. After the Contract is awarded, the Owner will provide or direct the Engineer to provide for the use of the Contractor documents that were developed by Engineer as part of the Project design process, as Electronic Documents in native file formats.
 - 1. Electronic Documents that are available in native file format include:
 - a. Project Manual.
 - b. Construction Plans.
 - 2. Release of such documents will be solely for the convenience of the Contractor. No such document is a Contract Document.
 - 3. Unless the Contract Documents explicitly identify that such information will be available to the Successful Bidder (Contractor), nothing herein will create an obligation on the part of the Owner or Engineer to provide or create such information, and the Contractor is not entitled to rely on the availability of such information in the preparation of its Bid or pricing of the Work. In all cases, the Contractor shall take appropriate measures to verify that any electronic/digital information provided in Electronic Documents is appropriate and adequate for the Contractor's specific purposes.
 - 4. In no case will the Contractor be entitled to additional compensation or time for completion due to any differences between the actual Contract Documents and any related document in native file format.

ARTICLE 3—QUALIFICATIONS OF BIDDERS

- 3.01 Deleted.
- 3.02 Deleted.
- 3.03 Bidder is to submit the following information with its Bid to demonstrate Bidder's qualifications to perform the Work:
 - A. Written evidence establishing its qualifications such as financial data, previous experience, and present commitments.
 - B. A written statement that Bidder is authorized to do business in the State of Ohio, or a written certification that Bidder will obtain such authority prior to the Effective Date of the Contract.
 - C. Bidder's state or other contractor license number, if applicable.
 - D. Subcontractor and Supplier qualification information.

EJCDC[®] C-200, Instructions to Bidders for Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

- E. Other required information regarding qualifications.
- 3.04 A Bidder's failure to submit required qualification information within the times indicated may disgualify Bidder from receiving an award of the Contract.
- 3.05 No requirement in this Article 3 to submit information will prejudice the right of Owner to seek additional pertinent information regarding Bidder's qualifications.

ARTICLE 4—PRE-BID CONFERENCE

- 4.01 A pre-bid conference will not be conducted for this Project.
- 4.02 Deleted.
- 4.03 Deleted.
- 4.04 Information presented at the pre-bid conference does not alter the Contract Documents. Owner will issue Addenda to make any changes to the Contract Documents that result from discussions at the pre-bid conference. Information presented, and statements made at the pre-bid conference will not be binding or legally effective unless incorporated in an Addendum.

ARTICLE 5—SITE AND OTHER AREAS; EXISTING SITE CONDITIONS; EXAMINATION OF SITE; OWNER'S SAFETY PROGRAM; OTHER WORK AT THE SITE

- 5.01 Site and Other Areas
 - A. The Site is identified in the Bidding Documents. By definition, the Site includes rights-of-way, easements, and other lands furnished by Owner for the use of the Contractor. Any additional lands required for temporary construction facilities, construction equipment, or storage of materials and equipment, and any access needed for such additional lands, are to be obtained and paid for by Contractor.

5.02 Existing Site Conditions

- A. Subsurface and Physical Conditions; Hazardous Environmental Conditions
 - 1. The Supplementary Conditions identify the following regarding existing conditions at or adjacent to the Site:
 - a. Those reports, if any, of explorations and tests of subsurface conditions at or adjacent to the Site that contain Technical Data.
 - b. Those drawings. If any, known to Owner of existing physical conditions at or adjacent to the Site, including those drawings depicting existing surface or subsurface structures at or adjacent to the Site (except Underground Facilities), that contain Technical Data.
 - c. Reports and drawings, if any, known to Owner relating to Hazardous Environmental Conditions that have been identified at or adjacent to the Site.
 - d. Technical Data contained in such reports and drawings.
 - 2. Owner will make copies of reports and drawings referenced above available to any Bidder on request. These reports and drawings are not part of the Contract Documents, but the Technical Data contained therein upon whose accuracy Bidder is entitled to rely, as provided in the General Conditions, has been identified and established in the

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

EJCDC[®] C-200, Instructions to Bidders for Construction Contract.

Supplementary Conditions. Bidder is responsible for any interpretation or conclusion Bidder draws from any Technical Data or any other data, interpretations, opinions, or information contained in such reports or shown or indicated in such drawings.

- 3. If the Supplementary Conditions do not identify Technical Data, the default definition of Technical Data set forth in Article 1 of the General Conditions will apply.
- B. Underground Facilities: Underground Facilities are shown or indicated on the Drawings, pursuant to Paragraph 5.05 of the General Conditions, and not in the drawings referred to in Paragraph 5.02.A of these Instructions to Bidders. Information and data regarding the presence or location of Underground Facilities are not intended to be categorized, identified, or defined as Technical Data.

5.03 Other Site-related Documents

- A. No other Site-related documents are available.
- 5.04 Site Visit and Testing by Bidders
 - A. Bidder is not required to visit the Site but available to explore the Site and conduct a thorough visual examination of the Site and adjacent areas. During the visit the Bidder must not disturb any ongoing operations at the Site.
 - B. Deleted.
 - C. Deleted.
 - D. Bidders visiting the Site are required to arrange their own transportation to the Site.
 - E. All access to the Site must be coordinated through the following Owner contact for visiting the Site: Josh Anders, <u>josh.anders@fairfieldcountyohio.gov</u>. Bidder must conduct the Site visit during normal working hours. Bidder must conduct the Site visit with a date no sooner than seven (7) days before bid opening.
 - F. Bidder is not required to conduct any subsurface testing, or exhaustive investigations of Site conditions.
 - G. On request, and to the extent Owner has control over the Site, and schedule permitting, the Owner will provide Bidder general access to the Site to conduct such additional examinations, investigations, explorations, tests, and studies as Bidder deems necessary for preparing and submitting a successful Bid. Owner will not have any obligation to grant such access if doing so is not practical because of existing operations, security or safety concerns, or restraints on Owner's authority regarding the Site. Bidder is responsible for establishing access needed to reach specific selected test sites.
 - H. Bidder must comply with all applicable Laws and Regulations regarding excavation and location of utilities, obtain all permits, and comply with all terms and conditions established by Owner or by property owners or other entities controlling the Site with respect to schedule, access, existing operations, security, liability insurance, and applicable safety programs.
 - I. Bidder must fill all holes and clean up and restore the Site to its former condition upon completion of such explorations, investigations, tests, and studies.

5.05 Owner's Safety Program

- A. Site visits and work at the Site may be governed by an Owner safety program. If an Owner safety program exists, it will be noted in the Supplementary Conditions.
- 5.06 Other Work at the Site
 - A. Reference is made to Article 8 of the Supplementary Conditions for the identification of the general nature of other work of which Owner is aware (if any) that is to be performed at the Site by Owner or others (such as utilities and other prime contractors) and relates to the Work contemplated by these Bidding Documents. If Owner is party to a written contract for such other work, then on request, Owner will provide to each Bidder access to examine such contracts (other than portions thereof related to price and other confidential matters), if any.

ARTICLE 6—BIDDER'S REPRESENTATIONS AND CERTIFICATIONS

- 6.01 Express Representations and Certifications in Bid Form, Agreement
 - A. The Bid Form that each Bidder will submit contains express representations regarding the Bidder's examination of Project documentation, Site visit, and preparation of the Bid, and certifications regarding lack of collusion or fraud in connection with the Bid. Bidder should review these representations and certifications, and assure that Bidder can make the representations and certifications in good faith, before executing and submitting its Bid.
 - B. If Bidder is awarded the Contract, Bidder (as Contractor) will make similar express representations and certifications when it executes the Agreement.

ARTICLE 7—INTERPRETATIONS AND ADDENDA

- 7.01 Owner on its own initiative may issue Addenda to clarify, correct, supplement, or change the Bidding Documents.
- 7.02 Bidder shall submit all questions about the meaning or intent of the Bidding Documents to Engineer in writing. Contact information for Engineer contact and submittal procedures for such questions are as follows:

Mark Buchenic, PE Project Manager PRIME AE Group, Inc. <u>mbuchenic@primeeng.com</u>

- 7.03 Interpretations or clarifications considered necessary by Engineer in response to such questions will be issued by Addenda delivered to all registered plan holders. Questions received less than seven days prior to the date for opening of Bids may not be answered.
- 7.04 Only responses set forth in an Addendum will be binding. Oral and other interpretations or clarifications will be without legal effect. Responses to questions are not part of the Contract

Documents unless set forth in an Addendum that expressly modifies or supplements the Contract Documents.

ARTICLE 8—BID SECURITY

- 8.01 A Bid must be accompanied by Bid security made payable to Owner in an amount of ten (10) percent of Bidder's maximum Bid price (determined by adding the base bid and all alternates) and in the form of a Bid bond issued by a surety meeting the requirements of Paragraph 6.01 of the General Conditions. Such Bid bond will be issued in the form included in the Bidding Documents.
- 8.02 The Bid security of the apparent Successful Bidder will be retained until Owner awards the contract to such Bidder, and such Bidder has executed the Contract, furnished the required Contract security, and met the other conditions of the Notice of Award, whereupon the Bid security will be released. If the Successful Bidder fails to execute and deliver the Contract and furnish the required Contract security within 15 days after the Notice of Award, Owner may consider Bidder to be in default, annul the Notice of Award, and the Bid security of that Bidder will be forfeited, in whole in the case of a penal sum bid bond, and to the extent of Owner's damages in the case of a damages-form bond. Such forfeiture will be Owner's exclusive remedy if Bidder defaults.
- 8.03 The Bid security of other Bidders that Owner believes to have a reasonable chance of receiving the award may be retained by Owner until the earlier of 7 days after the Effective Date of the Contract or 61 days after the Bid opening, whereupon Bid security furnished by such Bidders will be released.
- 8.04 Bid security of other Bidders that Owner believes do not have a reasonable chance of receiving the award will be released within 7 days after the Bid opening.

ARTICLE 9—CONTRACT TIMES

- 9.01 The number of days within which, or the dates by which, the Work is to be (a) substantially completed and (b) ready for final payment, and (c) Milestones (if any) are to be achieved, are set forth in the Agreement.
- 9.02 Deleted.
- 9.03 Provisions for liquidated damages, if any, for failure to timely attain a Milestone, Substantial Completion, or completion of the Work in readiness for final payment, are set forth in the Agreement.

ARTICLE 10—SUBSTITUTE AND "OR EQUAL" ITEMS

- 10.01 Deleted.
- 10.02 The Contract for the Work, as awarded, will be on the basis of materials and equipment specified or described in the Bidding Documents, and those "or-equal" or substitute or materials and equipment subsequently approved by Engineer prior to the submittal of Bids and identified by Addendum. No item of material or equipment will be considered by Engineer as an "or-equal" or substitute unless written request for approval has been submitted by Bidder and has been received by Engineer within 10 days of the issuance of the Advertisement for Bids. Each such request must comply with the requirements of Paragraphs 7.05 and 7.06 of the General

Conditions, and the review of the request will be governed by the principles in those paragraphs. The burden of proof of the merit of the proposed item is upon Bidder. Engineer's decision of approval or disapproval of a proposed item will be final. If Engineer approves any such proposed item, such approval will be set forth in an Addendum issued to all registered Bidders. Bidders cannot rely upon approvals made in any other manner.

10.03 All prices that Bidder sets forth in its Bid will be based on the presumption that the Contractor will furnish the materials and equipment specified or described in the Bidding Documents, as supplemented by Addenda. Any assumptions regarding the possibility of post-Bid approvals of "or-equal" or substitution requests are made at Bidder's sole risk.

ARTICLE 11—SUBCONTRACTORS, SUPPLIERS, AND OTHERS

- 11.01 A Bidder must be prepared to retain specific Subcontractors and Suppliers for the performance of the Work if required to do so by the Bidding Documents or in the Specifications. If a prospective Bidder objects to retaining any such Subcontractor or Supplier and the concern is not relieved by an Addendum, then the prospective Bidder should refrain from submitting a Bid.
- 11.02 The apparent Successful Bidder, and any other Bidder so requested, must submit to Owner a list of the Subcontractors or Suppliers within five days after Bid opening.
- 11.03 If requested by Owner, such list must be accompanied by an experience statement with pertinent information regarding similar projects and other evidence of qualification for each such Subcontractor or Supplier. If Owner or Engineer, after due investigation, has reasonable objection to any proposed Subcontractor or Supplier, Owner may, before the Notice of Award is given, request apparent Successful Bidder to submit an acceptable substitute, in which case apparent Successful Bidder will submit a substitute, Bidder's Bid price will be increased (or decreased) by the difference in cost occasioned by such substitution, and Owner may consider such price adjustment in evaluating Bids and making the Contract award.
- 11.04 If apparent Successful Bidder declines to make any such substitution, Owner may award the Contract to the next lowest Bidder that proposes to use acceptable Subcontractors and Suppliers. Declining to make requested substitutions will constitute grounds for forfeiture of the Bid security of any Bidder. Any Subcontractor or Supplier, so listed and against which Owner or Engineer makes no written objection prior to the giving of the Notice of Award will be deemed acceptable to Owner and Engineer subject to subsequent revocation of such acceptance as provided in Paragraph 7.07 of the General Conditions.

ARTICLE 12—PREPARATION OF BID

- 12.01 The Bid Form is included with the Bidding Documents.
 - A. All blanks on the Bid Form must be completed in ink and the Bid Form signed in ink. Erasures or alterations must be initialed in ink by the person signing the Bid Form. A Bid price must be indicated for each section, Bid item, alternate, adjustment unit price item, and unit price item listed therein.
 - B. If the Bid Form expressly indicates that submitting pricing on a specific alternate item is optional, and Bidder elects to not furnish pricing for such optional alternate item, then Bidder may enter the words "No Bid" or "Not Applicable."

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 3/21/2023

- 12.02 If Bidder has obtained the Bidding Documents as Electronic Documents, then Bidder shall prepare its Bid on a paper copy of the Bid Form printed from the Electronic Documents version of the Bidding Documents. The printed copy of the Bid Form must be clearly legible, printed on 8½ inch by 11-inch paper and as closely identical in appearance to the Electronic Document version of the Bid Form as may be practical. The Owner reserves the right to accept Bid Forms which nominally vary in appearance from the original paper version of the Bid Form, providing that all required information and submittals are included with the Bid.
- 12.03 A Bid by a corporation must be executed in the corporate name by a corporate officer (whose title must appear under the signature), accompanied by evidence of authority to sign. The corporate address and state of incorporation must be shown.
- 12.04 A Bid by a partnership must be executed in the partnership name and signed by a partner (whose title must appear under the signature), accompanied by evidence of authority to sign. The official address of the partnership must be shown.
- 12.05 A Bid by a limited liability company must be executed in the name of the firm by a member or other authorized person and accompanied by evidence of authority to sign. The state of formation of the firm and the official address of the firm must be shown.
- 12.06 A Bid by an individual must show the Bidder's name and official address.
- 12.07 A Bid by a joint venture must be executed by an authorized representative of each joint venturer in the manner indicated on the Bid Form. The joint venture must have been formally established prior to submittal of a Bid, and the official address of the joint venture must be shown.
- 12.08 All names must be printed in ink below the signatures.
- 12.09 The Bid must contain an acknowledgment of receipt of all Addenda, the numbers of which must be filled in on the Bid Form.
- 12.10 Postal and e-mail addresses and telephone number for communications regarding the Bid must be shown.
- 12.11 The Bid must contain evidence of Bidder's authority to do business in the state where the Project is located, or Bidder must certify in writing that it will obtain such authority within the time for acceptance of Bids and attach such certification to the Bid.
- 12.12 If Bidder is required to be licensed to submit a Bid or perform the Work in the state where the Project is located, the Bid must contain evidence of Bidder's licensure, or Bidder must certify in writing that it will obtain such licensure within the time for acceptance of Bids and attach such certification to the Bid. Bidder's state contractor license number, if any, must also be shown on the Bid Form.

ARTICLE 13—BASIS OF BID

- 13.01 Lump Sum
 - A. Bidders must submit a Bid on a lump sum basis as set forth in the Bid Form.

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

- FAIRFIELD COUNTY UTILITIES 3/21/2023
- 13.02 Deleted.
- 13.03 Deleted.
- 13.04 Deleted.
- 13.05 Deleted.
- 13.06 Deleted.
- 13.07 Deleted.

ARTICLE 14—SUBMITTAL OF BID

- 14.01 The Bidding Documents include one separate unbound copy of the Bid Form, and, if required, the Bid Bond Form. The unbound copy of the Bid Form is to be completed and submitted with the Bid security and the other documents required to be submitted under the terms of Article 2 of the Bid Form.
- 14.02 A Bid must be received no later than the date and time prescribed and at the place indicated in the Advertisement for Bids and must be enclosed in a plainly marked package with the Project title, and, if applicable, the designated portion of the Project for which the Bid is submitted, the name and address of Bidder, and must be accompanied by the Bid security and other required documents. If a Bid is sent by mail or other delivery system, the sealed envelope containing the Bid must be enclosed in a separate package plainly marked on the outside with the notation "BID ENCLOSED." A mailed Bid must be addressed to the location designated in the Advertisement for Bids.
- 14.03 Bids received after the date and time prescribed for the opening of bids, or not submitted at the correct location or in the designated manner, will not be accepted and will be returned to the Bidder unopened.

ARTICLE 15-MODIFICATION AND WITHDRAWAL OF BID

- 15.01 An unopened Bid may be withdrawn by an appropriate document duly executed in the same manner that a Bid must be executed and delivered to the place where Bids are to be submitted prior to the date and time for the opening of Bids. Upon receipt of such notice, the unopened Bid will be returned to the Bidder.
- 15.02 If a Bidder wishes to modify its Bid prior to Bid opening, Bidder must withdraw its initial Bid in the manner specified in Paragraph 15.01 and submit a new Bid prior to the date and time for the opening of Bids.
- 15.03 If within 24 hours after Bids are opened any Bidder files a duly signed written notice with Owner and promptly thereafter demonstrates to the reasonable satisfaction of Owner that there was a material and substantial mistake in the preparation of its Bid, the Bidder may withdraw its Bid,

and the Bid security will be returned. Thereafter, if the Work is rebid, the Bidder will be disqualified from further bidding on the Work.

ARTICLE 16—OPENING OF BIDS

- 16.01 Bids will be opened at the time and place indicated in the Advertisement for Bids and, unless obviously non-responsive, read aloud publicly. An abstract of the amounts of the base Bids and major alternates, if any, will be made available to Bidders after the opening of Bids.
- 16.02 Deleted.

ARTICLE 17—BIDS TO REMAIN SUBJECT TO ACCEPTANCE

17.01 All Bids will remain subject to acceptance for the period of time stated in the Bid Form, but Owner may, in its sole discretion, release any Bid and return the Bid security prior to the end of this period.

ARTICLE 18—EVALUATION OF BIDS AND AWARD OF CONTRACT

- 18.01 Owner reserves the right to reject any or all Bids, including without limitation, nonconforming, nonresponsive, unbalanced, or conditional Bids. Owner also reserves the right to waive all minor Bid informalities not involving price, time, or changes in the Work.
- 18.02 Owner will reject the Bid of any Bidder that Owner finds, after reasonable inquiry and evaluation, to not be responsible.
- 18.03 If Bidder purports to add terms or conditions to its Bid, takes exception to any provision of the Bidding Documents, or attempts to alter the contents of the Contract Documents for purposes of the Bid, whether in the Bid itself or in a separate communication to Owner or Engineer, then Owner will reject the Bid as nonresponsive.
- 18.04 If Owner awards the contract for the Work, such award will be to the responsible Bidder submitting the lowest responsive Bid.
- 18.05 Evaluation of Bids
 - A. In evaluating Bids, Owner will consider whether the Bids comply with the prescribed requirements, and such alternates, unit prices, and other data, as may be requested in the Bid Form or prior to the Notice of Award.
- 18.06 In evaluating whether a Bidder is responsible, Owner will consider the qualifications of the Bidder and may consider the qualifications and experience of Subcontractors and Suppliers proposed for those portions of the Work for which the identity of Subcontractors and Suppliers must be submitted as provided in the Bidding Documents.
- 18.07 Owner may conduct such investigations as Owner deems necessary to establish the responsibility, qualifications, and financial ability of Bidders and any proposed Subcontractors or Suppliers.

ARTICLE 19—BONDS AND INSURANCE

19.01 Article 6 of the General Conditions, as may be modified by the Supplementary Conditions, sets forth Owner's requirements as to performance and payment bonds, other required bonds (if any),

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

and insurance. When the Successful Bidder delivers the executed Agreement to Owner, it must be accompanied by required bonds and insurance documentation.

19.02 Article 8, Bid Security, of these Instructions, addresses any requirements for providing bid bonds as part of the bidding process.

ARTICLE 20—SIGNING OF AGREEMENT

20.01 When Owner issues a Notice of Award to the Successful Bidder, it will be accompanied by the unexecuted counterparts of the Agreement along with the other Contract Documents as identified in the Agreement. Within 15 days thereafter, Successful Bidder must execute and deliver the required number of counterparts of the Agreement and any bonds and insurance documentation required to be delivered by the Contract Documents to Owner. Within 10 days thereafter, Owner will deliver one fully executed counterpart of the Agreement to Successful Bidder, together with printed and electronic copies of the Contract Documents as stated in Paragraph 2.02 of the General Conditions.

ARTICLE 21—SALES AND USE TAXES

21.01 Owner is exempt from Ohio state sales and use taxes on materials and equipment to be incorporated in the Work (Exemption No. 31-6400066). Said taxes must not be included in the Bid. Refer to Paragraph SC-7.10 of the Supplementary Conditions for additional information.

ARTICLE 22—CONTRACTS TO BE ASSIGNED

22.01 Deleted,

and the second se

BID FORM FOR CONSTRUCTION CONTRACT

The terms used in this Bid with initial capital letters have the meanings stated in the instructions to Bidders, the General Conditions, and the Supplementary Conditions.

ARTICLE 1—OWNER AND BIDDER

- 1.01 This Bid is submitted to: Fairfield County Utilities, 6670 Lockville Road NW, Carroll, Ohio 43112.
- 1.02 The undersigned Bidder proposes and agrees, if this Bid is accepted, to enter into an Agreement with Owner in the form included in the Bidding Documents to perform all Work as specified or indicated in the Bidding Documents for the prices and within the times indicated in this Bid and in accordance with the other terms and conditions of the Bidding Documents.

ARTICLE 2—ATTACHMENTS TO THIS BID

- 2.01 The following documents are submitted with and made a condition of this Bid:
 - A. Required Bid security;
 - B. List of Proposed Subcontractors, if applicable;
 - C. List of Proposed Suppliers;
 - D. Evidence of authority to do business in the state of the Project; or a written covenant to obtain such authority within the time for acceptance of Bids;
 - E. Contractor's license number as evidence of Bidder's State Contractor's License or a covenant by Bidder to obtain said license within the time for acceptance of Bids; and
 - F. Required Bidder Qualification Statement with supporting data.

ARTICLE 3—BASIS OF BID — LUMP SUM BID

- 3.01 Lump Sum Bids
 - A. Bidder will complete the Work in accordance with the Contract Documents for the following lump sum (stipulated) price(s), together with any Unit Prices indicated in Paragraph 3.02:
 - 1. Lump Sum Price (Single Lump Sum)

Lump Sum Bid Price	FOUR HUNDRED NEVETY FEVE THOUSAND	\$ 495,000,00
	DOLLARS AND ZERO LENTS	

3.02 Deleted.

the second s

- The second s
 - the second s
 - ¹ Let many *in the link of the link*

the second se

- - - and the second second

3.03 Total Base Bid Price (Lump Sum Price)

Total Base Bid Price (Total of Lump Sum Price Bids)	\$
	495,000.00

ARTICLE 4-DELETED

ARTICLE 5—DELETED

ARTICLE 6—TIME OF COMPLETION

- 6.01 Bidder agrees that the Work will be substantially complete and will be completed and ready for final payment in accordance with Paragraph 15.06 of the General Conditions on or before the dates or within the number of calendar days indicated in the Agreement.
- 6.02 Deleted.
- 6.03 Deleted.
- 6.04 Bidder accepts the provisions of the Agreement as to liquidated damages.

ARTICLE 7—BIDDER'S ACKNOWLEDGEMENTS: ACCEPTANCE PERIOD, INSTRUCTIONS, AND RECEIPT OF ADDENDA

- 7.01 Bid Acceptance Period
 - A. This Bid will remain subject to acceptance for 60 days after the Bid opening, or for such longer period of time that Bidder may agree to in writing upon request of Owner.
- 7.02 Instructions to Bidders
 - A. Bidder accepts all of the terms and conditions of the Instructions to Bidders, including without limitation those dealing with the disposition of Bid security.
- 7.03 Receipt of Addenda
 - A. Bidder hereby acknowledges receipt of the following Addenda:

Bidder is to complete table.

Addendum Number	Addendum Date
-1	MAU 8 2023

ARTICLE 8—BIDDER'S REPRESENTATIONS AND CERTIFICATIONS

- 8.01 Bidder's Representations
 - A. In submitting this Bid, Bidder represents the following:
 - 1. Bidder has examined and carefully studied the Bidding Documents, including Addenda.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

- 2. Bidder has visited the Site, conducted a thorough visual examination of the Site and adjacent areas, and become familiar with the general, local, and Site conditions that may affect cost, progress, and performance of the Work.
- 3. Bidder is familiar with all Laws and Regulations that may affect cost, progress, and performance of the Work.
- 4. Bidder has carefully studied the reports of explorations and tests of subsurface conditions at or adjacent to the Site and the drawings of physical conditions relating to existing surface or subsurface structures at the Site that have been identified in the Supplementary Conditions, with respect to the Technical Data in such reports and drawings.
- 5. Bidder has carefully studied the reports and drawings relating to Hazardous Environmental Conditions, if any, at or adjacent to the Site that have been identified in the Supplementary Conditions, with respect to Technical Data in such reports and drawings.
- 6. Bidder has considered the information known to Bidder itself; information commonly known to contractors doing business in the locality of the Site; information and observations obtained from visits to the Site; the Bidding Documents; and the Technical Data identified in the Supplementary Conditions or by definition, with respect to the effect of such information, observations, and Technical Data on (a) the cost, progress, and performance of the Work; (b) the means, methods, techniques, sequences, and procedures of construction to be employed by Bidder, if selected as Contractor; and (c) Bidder's (Contractor's) safety precautions and programs.
- 7. Based on the information and observations referred to in the preceding paragraph, Bidder agrees that no further examinations, investigations, explorations, tests, studies, or data are necessary for the performance of the Work at the Contract Price, within the Contract Times, and in accordance with the other terms and conditions of the Contract.
- 8. Bidder is aware of the general nature of work to be performed by Owner and others at the Site that relates to the Work as indicated in the Bidding Documents.
- 9. Bidder has given Engineer written notice of all conflicts, errors, ambiguities, or discrepancies that Bidder has discovered in the Bidding Documents, and of discrepancies between Site conditions and the Contract Documents, and the written resolution thereof by Engineer is acceptable to Contractor.
- 10. The Bidding Documents are generally sufficient to indicate and convey understanding of all terms and conditions for performance and furnishing of the Work.
- 11. The submission of this Bid constitutes an incontrovertible representation by Bidder that without exception the Bid and all prices in the Bid are premised upon performing and furnishing the Work required by the Bidding Documents.

8.02 Bidder's Certifications

- A. The Bidder certifies the following:
 - 1. This Bid is genuine and not made in the interest of or on behalf of any undisclosed individual or entity and is not submitted in conformity with any collusive agreement or rules of any group, association, organization, or corporation.

EJCDC[®] C-410, Bid Form for Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

- 2. Bidder has not directly or indirectly induced or solicited any other Bidder to submit a false or sham Bid.
- 3. Bidder has not solicited or induced any individual or entity to refrain from bidding.
- 4. Bidder has not engaged in corrupt, fraudulent, collusive, or coercive practices in competing for the Contract. For the purposes of this Paragraph 8.02.A:
 - a. Corrupt practice means the offering, giving, receiving, or soliciting of anything of value likely to influence the action of a public official in the bidding process.
 - b. Fraudulent practice means an intentional misrepresentation of facts made (a) to influence the bidding process to the detriment of Owner, (b) to establish bid prices at artificial non-competitive levels, or (c) to deprive Owner of the benefits of free and open competition.
 - c. Collusive practice means a scheme or arrangement between two or more Bidders, with or without the knowledge of Owner, a purpose of which is to establish bid prices at artificial, non-competitive levels.
 - d. Coercive practice means harming or threatening to harm, directly or indirectly, persons or their property to influence their participation in the bidding process or affect the execution of the Contract.

BIDDER hereby submits this Bid as set forth above:

D .		л.		
- B1	id	a	er	•:

bluder.	La Court Country to
	(typed or printed name of organization)
By:	1 E
	(individual's signature)
Name:	Nicholas Edwards
	(typed or printed)
Title:	Trecsurer
_	(typed or printed)
Date:	(typed or printed)
lf Bidder is	s a corporation, a partnership, or a joint venture, attach evidence of authority to sign.
Attest:	had M/Mgfri
	Al (individual's signature)
Name:	(typed or printed)
Title:	1.00
	(typed or printed)
Date:	5-9-23
	(typed or printed)
Address (or giving notices:
	9128 Mt Vernon Rd
	St Louisville, Ohio 43071
Bidder's	
Name:	(typed or printed)
Title:	Treesurer
fice.	(typed or printed)
Phone:	740 745 3420
Email:	nick@lawstreme.com
Address:	
	Low General Contracting Inc
	g128 Mt Vernon Rd
	St Louisvilk, Ohio 43071
Bidder's	Contractor License No.: (if applicable) NIM

0/6/	2022	
------	------	--

•

Ŧ .

Bidder	Surety
Name: Law General Contracting Inc.	Name: Fidelity and Deposit Company of Maryland
Address (principal place of business):	Address (principal place of business):
9128 Mt. Vernon Road	1299 Zurich Way
St. Louisville, OH 43071	Schaumburg, IL 60196
Owner	Bid
Name: Fairfield County Utilities	Project (name and location):
Address (principal place of business):	Airport Sanitary Sewer Extension and Water
Fairfield County Utilities	Service Design
6670 Lockville Road NW	3430 Old Columbus Rd NW
Carroll, Ohio 43112	Carroll, OH 43112
	Bid Due Date: May 11, 2023
Bond	
Penal Sum: \$ 1,000 Per Day	
Date of Bond: May 11, 2023	
	reby, subject to the terms set forth in this Bid Bond,
do each cause this Bid Bond to be duly executed by	
Bidder	Surety
Law General Contracting Inc.	Fidelity and Deposit Company of Maryland
(Full formal name of Bidder)	(Full formal name of Surety) (corporate seal)
By: Z E	By: Dant
(Signature)	(Signature) (Attach Power of Attorney)
Name: Nucholas Educido	Name: Peyton Janlin
(Printed or typed)	(Printed or typed)
Title: Treasurer	Title: Attorney-In-Fact
Attest: Charly up/1/2	Attest: Ityp Jamie M. Unite
Attest. (Signature)	(Signature)
Name: Chad II had and	Name: Stephanie M. White
(Printed or typed)	(Printed or typed)
Title: COC	Title: Attorney-In-Fact
	ed notice. (2) Provide execution by any additional parties, such as
joint venturers, if necessary.	

BID BOND (PENAL SUM FORM)

EJCDC[®] C-430, Bid Bond (Penal Sum Form). Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

2

- - 3.8

 - - - and the second second

- Bidder and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors, and assigns to pay to Owner upon default of Bidder the penal sum set forth on the face of this Bond. Payment of the penal sum is the extent of Bidder's and Surety's liability. Recovery of such penal sum under the terms of this Bond will be Owner's sole and exclusive remedy upon default of Bidder.
- 2. Default of Bidder occurs upon the failure of Bidder to deliver within the time required by the Bidding Documents (or any extension thereof agreed to in writing by Owner) the executed Agreement required by the Bidding Documents and any performance and payment bonds required by the Bidding Documents.
- 3. This obligation will be null and void if:
 - 3.1. Owner accepts Bidder's Bid and Bidder delivers within the time required by the Bidding Documents (or any extension thereof agreed to in writing by Owner) the executed Agreement required by the Bidding Documents and any performance and payment bonds required by the Bidding Documents, or
 - 3.2. All Bids are rejected by Owner, or
 - 3.3. Owner fails to issue a Notice of Award to Bidder within the time specified in the Bidding Documents (or any extension thereof agreed to in writing by Bidder and, if applicable, consented to by Surety when required by Paragraph 5 hereof).
- 4. Payment under this Bond will be due and payable upon default of Bidder and within 30 calendar days after receipt by Bidder and Surety of written notice of default from Owner, which notice will be given with reasonable promptness, identifying this Bond and the Project and including a statement of the amount due.
- 5. Surety waives notice of any and all defenses based on or arising out of any time extension to issue Notice of Award agreed to in writing by Owner and Bidder, provided that the total time for issuing Notice of Award including extensions does not in the aggregate exceed 120 days from the Bid due date without Surety's written consent.
- 6. No suit or action will be commenced under this Bond prior to 30 calendar days after the notice of default required in Paragraph 4 above is received by Bidder and Surety, and in no case later than one year after the Bid due date.
- 7. Any suit or action under this Bond will be commenced only in a court of competent jurisdiction located in the state in which the Project is located.
- 8. Notices required hereunder must be in writing and sent to Bidder and Surety at their respective addresses shown on the face of this Bond. Such notices may be sent by personal delivery, commercial courier, or by United States Postal Service registered or certified mail, return receipt requested, postage pre-paid, and will be deemed to be effective upon receipt by the party concerned.
- 9. Surety shall cause to be attached to this Bond a current and effective Power of Attorney evidencing the authority of the officer, agent, or representative who executed this Bond on behalf of Surety to execute, seal, and deliver such Bond and bind the Surety thereby.
- 10. This Bond is intended to conform to all applicable statutory requirements. Any applicable requirement of any applicable statute that has been omitted from this Bond will be deemed to be included herein as if set forth at length. If any provision of this Bond conflicts with any applicable statute, then the provision of said statute governs and the remainder of this Bond that is not in conflict therewith continues in full force and effect.
- 11. The term "Bid" as used herein includes a Bid, offer, or proposal as applicable.

- - the second se
- - the second s

EXTRACT FROM BY-LAWS OF THE COMPANIES

"Article V, Section 8, <u>Attorneys-in-Fact</u>. The Chief Executive Officer, the President, or any Executive Vice President or Vice President may, by written instrument under the attested corporate seal, appoint attorneys-in-fact with authority to execute bonds, policies, recognizances, stipulations, undertakings, or other like instruments on behalf of the Company, and may authorize any officer or any such attorney-in-fact to affix the corporate seal thereto; and may with or without cause modify of revoke any such appointment or authority at any time."

CERTIFICATE

I, the undersigned, Vice President of the ZURICH AMERICAN INSURANCE COMPANY, the COLONIAL AMERICAN CASUALTY AND SURETY COMPANY, and the FIDELITY AND DEPOSIT COMPANY OF MARYLAND, do hereby certify that the foregoing Power of Attorney is still in full force and effect on the date of this certificate; and I do further certify that Article V, Section 8, of the By-Laws of the Companies is still in force.

This Power of Attorney and Certificate may be signed by facsimile under and by authority of the following resolution of the Board of Directors of the ZURICH AMERICAN INSURANCE COMPANY at a meeting duly called and held on the 15th day of December 1998.

RESOLVED: "That the signature of the President or a Vice President and the attesting signature of a Secretary or an Assistant Secretary and the Seal of the Company may be affixed by facsimile on any Power of Attorney...Any such Power or any certificate thereof bearing such facsimile signature and seal shall be valid and binding on the Company_"

This Power of Attorney and Certificate may be signed by facsimile under and by authority of the following resolution of the Board of Directors of the COLONIAL AMERICAN CASUALTY AND SURETY COMPANY at a meeting duly called and held on the 5th day of May, 1994, and the following resolution of the Board of Directors of the FIDELITY AND DEPOSIT COMPANY OF MARYLAND at a meeting duly called and held on the 10th day of May, 1990.

RESOLVED: "That the facsimile or mechanically reproduced seal of the company and facsimile or mechanically reproduced signature of any Vice-President, Secretary, or Assistant Secretary of the Company, whether made heretofore or hereafter, wherever appearing upon a certified copy of any power of attorney issued by the Company, shall be valid and binding upon the Company with the same force and effect as though manually affixed.

IN TESTIMONY WHEREOF. 1	have hereunto subscribed my name and affixed the corporate seals of the said Companies
this 11th day of May	have hereunto subscribed my name and affixed the corporate seals of the said Companies.



Mary Jean Pethick By: Vice President

TO REPORT A CLAIM WITH REGARD TO A SURETY BOND, PLEASE SUBMIT A COMPLETE DESCRIPTION OF THE CLAIM INCLUDING THE PRINCIPAL ON THE BOND, THE BOND NUMBER, AND YOUR CONTACT INFORMATION TO:

Zurich Surety Claims 1299 Zurich Way Schaumburg, IL 60196-1056 Ph: 800-626-4577

If your jurisdiction allows for electronic reporting of surety claims, please submit to: reportsfclaims@zurichna.com

Authenticity of this bond can be confirmed at bondvalidator.zurichna.com or 410-559-8790

ZURICH AMERICAN INSURANCE COMPANY COLONIAL AMERICAN CASUALTY AND SURETY COMPANY FIDELITY AND DEPOSIT COMPANY OF MARYLAND POWER OF ATTORNEY

KNOW ALL MEN BY THESE PRESENTS: That the ZURICH AMERICAN INSURANCE COMPANY, a corporation of the State of New York, the COLONIAL AMERICAN CASHALTY AND SURETY COMPANY, a corporation of the State of Illinois, and the FIDELITY AND DEPOSIT COMPANY OF MARYLAND a corporation of the State of Illinois (herein collectively called the "Companics"), by Robert D. Murray, Vice President, in pursuance of authority granted by Article V, Section 8, of the By-Laws of said Companies, which are set forth on the reverse side hereof and are hereby certified to be in full force and effect on the date hereof, do hereby nominate, constitute, and appoint Gregory R. OVERMYER, Stephanie M. WHITE, Amy M. PERDUE, Jack KEHL, David N. CATANESE, Peyton JANLIN and Brian MOZENA, of Columbus, Ohio, its true and lawful agent and Attorney-in-Fact, to make, execute, seal and deliver, for, and on its behalf as surety, and as its act and deed: any and all bonds and undertakings, and the execution of such bonds or undertakings in pursuance of these presents, shall be as binding upon said Companies, as fully and amply, to all intents and purposes, as if they had been duly executed and acknowledged by the regularly elected officers of the ZURICH AMERICAN INSURANCE COMPANY at its office in York, New York, the regularly elected officers of the COLONIAL AMERICAN CASUALTY AND SURETY New COMPANY at its office in Owings Mills, Maryland., and the regularly elected officers of the FIDELITY AND DEPOSIT COMPANY OF MARYLAND at its office in Owings Mills, Maryland., in their own proper persons.

The said Vice President does hereby certify that the extract set forth on the reverse side hereof is a true copy of Article V, Section 8, of the By-Laws of said Companies, and is now in force.

IN WITNESS WHEREOF, the said Vice-President has hereunto subscribed his/her names and affixed the Corporate Seals of the said ZURICH AMERICAN INSURANCE COMPANY, COLONIAL AMERICAN CASUALTY AND SURETY COMPANY, and FIDELITY AND DEPOSIT COMPANY OF MARYLAND, this 27th day of May, A.D. 2022.

By: Robert D. Murray Vice President

By: Dawn E. Brown Secretary

State of Maryland County of Baltimore ATTEST: ZURICH AMERICAN INSURANCE COMPANY COLONIAL AMERICAN CASUALTY AND SURETY COMPANY FIDELITY AND DEPOSIT COMPANY OF MARYLAND



On this 27th day of May, A D 2022, before the subscriber, a Notary Public of the State of Maryland, duly commissioned and qualified, **Robert D. Murray, Vice President and Dawn E. Brown, Secretary** of the Companies, to me personally known to be the individuals and officers described in and who executed the preceding instrument, and acknowledged the execution of same, and being by me duly swom, deposeth and saith, that he/she is the said officer of the Company aforesaid, and that the seals affixed to the preceding instrument are the Corporate Seals of said Companies, and that the said Corporate Seals and the signature as such officer were duly affixed and subscribed to the said instrument by the authority and direction of the said Corporations

IN TESTIMONY WHEREOF, I have hereunto set my hand and affixed my Official Seal the day and year first above written



onstance a. Durn

Constance A Dunn, Notary Public My Commission Expires: July 9, 2023

Authenticity of this bond can be confirmed at bondvalidator.zurichna.com or 410-559-8790

06/06/2023

THE FIDELITY AND DEPOSIT COMPANY

OF MARYLAND

1299 Zurich Way Schaumburg, IL 60196

Statement of Financial Condition As Of December 31, 2022

ASSETS

Bonds	; 219,365,765
Stocks	17,619,752
Cash and Short-Term Investments	2,767,685
Reinsurance Recoverable	12,309,422
Federal Income Tax Recoverable	0
	32,200,768
Other Accounts Receivable	
TOTAL ADMITTED ASSETS	204,203,352

LIABILITIES, SURPLUS AND OTHER FUNDS	6	a c45.021
Reserve for Taxes and Expenses	******************************	\$ 545,031
Ceded Reinsurance Premiums Payable		42,850,834
Remittances and Items Unallocated	************************	0
Payable to parents, subs and affiliates		0
Securities Lending Collatoral Liability	•••••••••	\$ 43,395,865
Capital Stock, Paid Up \$ Surplus		
Surplus as regards Policyholders		240,867,527
TOTAL		\$ 284,263,392

Securities carried at \$78,455,274 in the above statement are deposited with various states as required by law.

Securities carried on the basis prescribed by the National Association of Insurance Commissioners. On the basis of market quotations for all bonds and stocks owned, the Company's total admitted assets at December 31, 2022 would be \$266,908,407 and surplus as regards policyholders \$223,512,542.

I, LAURA J. LAZARCZYK, Corporate Secretary of the FIDELITY AND DEPOSIT COMPANY OF MARYLAND, do hereby certify that the foregoing statement is a correct exhibit of the assets and liabilities of the said Company on the 31st day of December, 2022.

Corporate Secretary

State of Illinois City of Schaumburg SS;

Subscribed and swom to, before me, a Notary Public of the State of Illinois, in the City of Schaumburg, this 15th day of March, 2023.

Conton S RYAN HORGAN Official Seat Notary Public - State of Illinois My Commission Expires Dec 10, 2024 The second s

Notary Public

06/06/2023

Office of Risk Assessment 50 West Town Street Third Floor - Suite 300 Columbus,Ohio 43215 (614)644-2658 Fax(614)644-3256 www.insurance.ohio.gov

Ohio Department of Insurance

Mike DeWine - Governor Judith French - Director



Certificate of Compliance

Issued 03/23/2023 Effective 04/02/2023 Expires 04/01/2024

I, Judith French, hereby certify that I am the Director of Insurance in the State of Ohio and have supervision of insurance business in said State and as such I hereby certify that

FIDELITY AND DEPOSIT COMPANY OF MARYLAND, THE

of Illinois is duly organized under the laws of this State and is authorized to transact the business of insurance under the following section(s) of the Ohio Revised Code:

Section 3929.01 (A)

Allied Lines Boiler & Machinery Burglary & Theft Commercial Auto - Liability Commercial Auto - No Fault Commercial Auto - Physical Damage Credit Earthquake Fidelity Financial Guaranty Fire Glass Inland Marine Multiple Peril - Commercial Multiple Peril - Homeowners Other Liability Surety

Workers Compensation

FIDELITY AND DEPOSIT COMPANY OF MARYLAND, THE certified in its annual statement to this Department as of December 31, 2022 that it has admitted assets in the amount of \$284,263,392, liabilities in the amount of \$43,395,865, and surplus of at least \$240,867,527.

IN WITNESS WHEREOF, I have hereunto subscribed my name and caused my seal to be affixed at Columbus. Ohio, this day and date.

Sudith L. French

Judith French, Director

INS7230(Rev.6/2003)

Accredited by the National Association of Insurance Commissioners (NAIC)

9

(*)

NOTICE OF AWARD

Date of Issuance:			
Owner:	Fairfield County Utilities		
Engineer:	PRIME AE Group, Inc.	Engineer's Project No.:	21513
Project:	Airport Sanitary Sewer Extension and Wat	er Service Design	
Bidder:			
N 1 1 1 1 1			

Bidder's Address:

You are notified that Owner has accepted your Bid dated <u>5/11/2023</u> for the above Contract, and that you are the Successful Bidder and are awarded a Contract for:

Airport Sanitary Sewer Extension and Water Service Design

The Contract Price of the awarded Contract is \$_495,000.00_____. Contract Price is subject to adjustment based on the provisions of the Contract, including but not limited to those governing changes, Unit Price Work, and Work performed on a cost-plus-fee basis, as applicable.

Five unexecuted counterparts of the Agreement accompany this Notice of Award, and one copy of the Contract Documents accompanies this Notice of Award, or has been transmitted or made available to Bidder electronically.

You must comply with the following conditions precedent within 15 days of the date of receipt of this Notice of Award:

- 1. Deliver to Owner three counterparts of the Agreement, signed by Bidder (as Contractor).
- 2. Deliver with the signed Agreement(s) the Contract security (such as required performance and payment bonds) and insurance documentation, as specified in the Instructions to Bidders and in the General Conditions, Articles 2 and 6.
- 3. Other conditions precedent (if any): None

Failure to comply with these conditions within the time specified will entitle Owner to consider you in default, annul this Notice of Award, and declare your Bid security forfeited.

Within 10 days after you comply with the above conditions, Owner will return to you one fully signed counterpart of the Agreement, together with any additional copies of the Contract Documents as indicated in Paragraph 2.02 of the General Conditions.

Owner:	Fairfield County Utilities
By (signature):	
Name (printed):	Tony Vogel, PE
Title:	Director of Utilities
Copy: Engineer	

(THIS PAGE INTENTIONALLY LEFT BLANK)

EJCDC [®] C-510, Notice of Award.
Copyright [©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,
and American Society of Civil Engineers. All rights reserved.

Page 2 of 2

AGREEMENT BETWEEN OWNER AND CONTRACTOR FOR CONSTRUCTION CONTRACT (STIPULATED PRICE)

This Agreement is by and between Fairfield County Utilities ("Owner") and Law General Contracting INC ("Contractor").

Terms used in this Agreement have the meanings stated in the General Conditions and the Supplementary Conditions.

Owner and Contractor hereby agree as follows:

ARTICLE 1-WORK

1.01 Contractor shall complete all Work as specified or indicated in the Contract Documents. The Work is generally described as follows: This project will install 613 linear feet of 8-inch gravity sewer main, 2 sanitary manholes, associated sanitary laterals, and water services for facilities at the Fairfield County Airport, Fairfield County, Ohio. The project will also install a dual-pump lift station and connect to an existing 2-inch sanitary force main.

ARTICLE 2—THE PROJECT

2.01 The Project, of which the Work under the Contract Documents is a part, is generally described as follows: Airport Sanitary Sewer Extension and Water Service Design.

ARTICLE 3—ENGINEER

- 3.01 The Owner has retained PRIME AE Group, Inc. ("Engineer") to act as Owner's representative, assume all duties and responsibilities of Engineer, and have the rights and authority assigned to Engineer in the Contract.
- 3.02 The part of the Project that pertains to the Work has been designed by Engineer.

ARTICLE 4—CONTRACT TIMES

- 4.01 *Time is of the Essence*
 - A. All time limits for Milestones, if any, Substantial Completion, and completion and readiness for final payment as stated in the Contract Documents are of the essence of the Contract.
- 4.02 Deleted.
- 4.03 Contract Times: Days
 - A. The Work will be substantially complete within 330 days after the date when the Contract Times commence to run as provided in Paragraph 4.01 of the General Conditions, and completed and ready for final payment in accordance with Paragraph 15.06 of the General Conditions within 365 days after the date when the Contract Times commence to run.

4.04 Deleted.

4.05 Liquidated Damages

- A. Contractor and Owner recognize that time is of the essence as stated in Paragraph 4.01 above and that Owner will suffer financial and other losses if the Work is not completed and Milestones not achieved within the Contract Times, as duly modified. The parties also recognize the delays, expense, and difficulties involved in proving, in a legal or arbitration proceeding, the actual loss suffered by Owner if the Work is not completed on time. Accordingly, instead of requiring any such proof, Owner and Contractor agree that as liquidated damages for delay (but not as a penalty):
 - 1. Substantial Completion: Contractor shall pay Owner \$1,000 for each day that expires after the time (as duly adjusted pursuant to the Contract) specified above for Substantial Completion, until the Work is substantially complete.
 - 2. Completion of Remaining Work: After Substantial Completion, if Contractor shall neglect, refuse, or fail to complete the remaining Work within the Contract Times (as duly adjusted pursuant to the Contract) for completion and readiness for final payment, Contractor shall pay Owner \$1,000 for each day that expires after such time until the Work is completed and ready for final payment.
 - 4. Liquidated damages for failing to timely attain Substantial Completion and final completion are not additive, and will not be imposed concurrently.
- B. If Owner recovers liquidated damages for a delay in completion by Contractor, then such liquidated damages are Owner's sole and exclusive remedy for such delay, and Owner is precluded from recovering any other damages, whether actual, direct, excess, or consequential, for such delay, except for special damages (if any) specified in this Agreement.

4.06 Special Damages

- A. Contractor shall reimburse Owner (1) for any fines or penalties imposed on Owner as a direct result of the Contractor's failure to attain Substantial Completion according to the Contract Times, and (2) for the actual costs reasonably incurred by Owner for engineering, construction observation, inspection, and administrative services needed after the time specified in Paragraph 4.02 for Substantial Completion (as duly adjusted pursuant to the Contract), until the Work is substantially complete.
- B. After Contractor achieves Substantial Completion, if Contractor shall neglect, refuse, or fail to complete the remaining Work within the Contract Times, Contractor shall reimburse Owner for the actual costs reasonably incurred by Owner for engineering, construction observation, inspection, and administrative services needed after the time specified in Paragraph 4.02 for Work to be completed and ready for final payment (as duly adjusted pursuant to the Contract), until the Work is completed and ready for final payment.
- C. The special damages imposed in this paragraph are supplemental to any liquidated damages for delayed completion established in this Agreement.

ARTICLE 5—CONTRACT PRICE

- 5.01 Owner shall pay Contractor for completion of the Work in accordance with the Contract Documents, the amounts that follow, subject to adjustment under the Contract:
 - A. For all Unit Price Work, at the bid unit prices stated in the Contractor's Bid, attached hereto as an exhibit.
 - B. The extended prices for Unit Price Work set forth as of the Effective Date of the Contract are based on estimated quantities. As provided in Paragraph 13.03 of the General Conditions, estimated quantities are not guaranteed, and determinations of actual quantities and classifications are to be made by Engineer.

ARTICLE 6—PAYMENT PROCEDURES

- 6.01 Submittal and Processing of Payments
 - A. Contractor shall submit Applications for Payment in accordance with Article 15 of the General Conditions. Applications for Payment will be processed by Engineer as provided in the General Conditions.
- 6.02 Progress Payments; Retainage
 - A. Owner shall make progress payments on the basis of Contractor's Applications for Payment on or about the 21st day of each month during performance of the Work as provided in Paragraph 6.02.A.1 below, provided that such Applications for Payment have been submitted in a timely manner and otherwise meet the requirements of the Contract. All such payments will be measured by the Schedule of Values established as provided in the General Conditions (and in the case of Unit Price Work based on the number of units completed) or, in the event there is no Schedule of Values, as provided elsewhere in the Contract.
 - 1. Prior to Substantial Completion, progress payments will be made in an amount equal to the percentage indicated below but, in each case, less the aggregate of payments previously made and less such amounts as Owner may withhold, including but not limited to liquidated damages, in accordance with the Contract.
 - a. 95 percent of the value of the Work completed (with the balance being retainage).
 - If 50 percent or more of the Work has been completed, as determined by Engineer, and if the character and progress of the Work have been satisfactory to Owner and Engineer, then as long as the character and progress of the Work remain satisfactory to Owner and Engineer, there will be no additional retainage.
 - B. Upon Substantial Completion of the entire construction to be provided under the construction Contract Documents, Owner shall pay an amount sufficient to increase total payments to Contractor to 100 percent of the Work completed, less such amounts set off by Owner pursuant to Paragraph 15.01.E of the General Conditions, and less 200 percent of Engineer's estimate of the value of Work to be completed or corrected as shown on the punch list of items to be completed or corrected prior to final payment.

6.03 Final Payment

A. Upon final completion and acceptance of the Work, Owner shall pay the remainder of the Contract Price in accordance with Paragraph 15.06 of the General Conditions.

EJCDC® C-520, Agreement between Owner and Contractor for Construction Contract (Stipulated Price). Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

6.04 Consent of Surety

FAIRFIELD COUNTY UTILITIES

A. Owner will not make final payment, or return or release retainage at Substantial Completion or any other time, unless Contractor submits written consent of the surety to such payment, return, or release.

6.05 Interest

10/6/2022

A. All amounts not paid when due will bear interest at the rate of 3 percent per annum.

ARTICLE 7—CONTRACT DOCUMENTS

7.01 Contents

- A. The Contract Documents consist of all of the following:
 - 1. This Agreement.
 - 2. Bonds:
 - a. Performance bond (together with power of attorney).
 - b. Payment bond (together with power of attorney).
 - 3. General Conditions.
 - 4. Supplementary Conditions.
 - 5. Specifications as listed in the table of contents of the Project Manual.
 - 6. Drawings (not attached but incorporated by reference) consisting of 11 sheets with each sheet bearing the following general title: Fairfield County Airport Sanitary Sewer Extension and Water Service Design.
 - 7. Addenda (numbers 1 to 1, inclusive).
 - 8. Exhibits to this Agreement (enumerated as follows):
 - a. Contractor's Bid
 - b. Notice of Award
 - 9. The following which may be delivered or issued on or after the Effective Date of the Contract and are not attached hereto:
 - a. Notice to Proceed.
 - b. Work Change Directives.
 - c. Change Orders.
 - d. Field Orders.
 - e. Warranty Bond, if any.
- B. The Contract Documents listed in Paragraph 7.01.A are attached to this Agreement (except as expressly noted otherwise above).
- C. There are no Contract Documents other than those listed above in this Article 7.

D. The Contract Documents may only be amended, modified, or supplemented as provided in the Contract.

ARTICLE 8-REPRESENTATIONS, CERTIFICATIONS, AND STIPULATIONS

- 8.01 Contractor's Representations
 - A. In order to induce Owner to enter into this Contract, Contractor makes the following representations:
 - 1. Contractor has examined and carefully studied the Contract Documents, including Addenda.
 - 2. Contractor has visited the Site, conducted a thorough visual examination of the Site and adjacent areas, and become familiar with the general, local, and Site conditions that may affect cost, progress, and performance of the Work.
 - 3. Contractor is familiar with all Laws and Regulations that may affect cost, progress, and performance of the Work.
 - 4. Contractor has carefully studied the reports of explorations and tests of subsurface conditions at or adjacent to the Site and the drawings of physical conditions relating to existing surface or subsurface structures at the Site that have been identified in the Supplementary Conditions, with respect to the Technical Data in such reports and drawings.
 - 5. Contractor has carefully studied the reports and drawings relating to Hazardous Environmental Conditions, if any, at or adjacent to the Site that have been identified in the Supplementary Conditions, with respect to Technical Data in such reports and drawings.
 - 6. Contractor has considered the information known to Contractor itself; information commonly known to contractors doing business in the locality of the Site; information and observations obtained from visits to the Site; the Contract Documents; and the Technical Data identified in the Supplementary Conditions or by definition, with respect to the effect of such information, observations, and Technical Data on (a) the cost, progress, and performance of the Work; (b) the means, methods, techniques, sequences, and procedures of construction to be employed by Contractor; and (c) Contractor's safety precautions and programs.
 - 7. Based on the information and observations referred to in the preceding paragraph, Contractor agrees that no further examinations, investigations, explorations, tests, studies, or data are necessary for the performance of the Work at the Contract Price, within the Contract Times, and in accordance with the other terms and conditions of the Contract.
 - 8. Contractor is aware of the general nature of work to be performed by Owner and others at the Site that relates to the Work as indicated in the Contract Documents.
 - 9. Contractor has given Engineer written notice of all conflicts, errors, ambiguities, or discrepancies that Contractor has discovered in the Contract Documents, and of discrepancies between Site conditions and the Contract Documents, and the written resolution thereof by Engineer is acceptable to Contractor.

EJCDC® C-520, Agreement between Owner and Contractor for Construction Contract (Stipulated Price). Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

- 10. The Contract Documents are generally sufficient to indicate and convey understanding of all terms and conditions for performance and furnishing of the Work.
- 11. Contractor's entry into this Contract constitutes an incontrovertible representation by Contractor that without exception all prices in the Agreement are premised upon performing and furnishing the Work required by the Contract Documents.

8.02 Contractor's Certifications

- A. Contractor certifies that it has not engaged in corrupt, fraudulent, collusive, or coercive practices in competing for or in executing the Contract. For the purposes of this Paragraph 8.02:
 - "corrupt practice" means the offering, giving, receiving, or soliciting of anything of value likely to influence the action of a public official in the bidding process or in the Contract execution;
 - "fraudulent practice" means an intentional misrepresentation of facts made (a) to influence the bidding process or the execution of the Contract to the detriment of Owner, (b) to establish Bid or Contract prices at artificial non-competitive levels, or (c) to deprive Owner of the benefits of free and open competition;
 - 3. "collusive practice" means a scheme or arrangement between two or more Bidders, with or without the knowledge of Owner, a purpose of which is to establish Bid prices at artificial, non-competitive levels; and
 - 4. "coercive practice" means harming or threatening to harm, directly or indirectly, persons or their property to influence their participation in the bidding process or affect the execution of the Contract.

8.03 Standard General Conditions

A. Owner stipulates that if the General Conditions that are made a part of this Contract are EJCDC[®] C-700, Standard General Conditions for the Construction Contract (2018), published by the Engineers Joint Contract Documents Committee, and if Owner is the party that has furnished said General Conditions, then Owner has plainly shown all modifications to the standard wording of such published document to the Contractor, through a process such as highlighting or "track changes" (redline/strikeout), or in the Supplementary Conditions.

IN WITNESS WHEREOF, Owner and Contractor have signed this Agreement.

This Agreement will be effective on	(which is the Effective Date of the Contract).
-------------------------------------	--

Contractor:

Owner:

	(typed or printed name of organization)
(typed or printed name of organization)	(typed or printed name of organization)
Ву:	By: ZE
(individual's signature)	(individual's signature)
Date:	Date: 5-17-23
(date signed)	(date signed)
Name:	Name: Micholas Edwards
(typed or printed)	(typed or printed)
Title:	Title: Treesurer
(typed or printed)	(typed or printed)
	(If Contractor is a corporation, a partnership, or a joint venture, attach evidence of authority to sign y
	the list all is
Attest:(individual's signature)	Attest: (individual's algebrare)
	In In It all
Title:	Title: (Woed or printed)
(typed or printed) Address for giving notices:	Address for giving notices:
Address for giving holices.	201 · · · · · · · · · · · · · · · · · · ·
	Low General Contracting Inc
	9128 Mt Vernun Rd
	St Louisville, Ohio 43071
Designated Representative:	Designated Representative:
Name:	Name: Base Catho
(typed or printed)	Name: Brice Cothin (typed or printed)
Title:	Title: President
(typed or printed)	(typed or printed)
Address:	Address:
	9128 Mt Vernon Rd
	St Louisville, Ohio 43071
Phone:	Phone: <u>740 404 2576</u>
Email:	Email: Brice Qlawytrame. com
(If Owner is a corporation, attach evidence of authority	
to sign. If Owner is a public body, attach evidence of	License No.: No.: (where applicable)
authority to sign and resolution or other documents authorizing execution of this Agreement.)	
entering encourse of the reflection of	State: Ohio

EICDC[®] C-520, Agreement between Owner and Contractor for Construction Contract (Stipulated Price). Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

_

IN WITNESS WHEREOF, Owner and Contractor have signed this Agreement.

This Agreement will be effective on	(which is the Effective Date of the Contract).	
Owner:	Contractor:	
(typed or printed name of organization)	(typed or printed name of organization)	
By:	By:	
(individual's signature)	(individual's signature)	
Date:	Date:	
(date signed)	(date signed)	
Name:	Name:	
(typed or printed)	(typed or printed)	
Title:	Title:	
(typed or printed)	(typed or printed) (If Contractor is a corporation, a partnership, or a joint venture, attach evidence of authority to sign.)	
Attest:	Attest:	
(individual's signature)	(individual's signature)	
Title:	Title:	
(typed or printed)	(typed or printed)	
Address for giving notices:	Address for giving notices:	
Designated Representative: Name:	Designated Representative: Name: (typed or printed)	
Title:	Title:	
(typed or printed)	(typed or printed)	
Address:	Address:	
Phone:	Phone:	
Email:	Email:	
(If Owner is a corporation, attach evidence of authority	License No.:	
to sign. If Owner is a public body, attach evidence of authority to sign and resolution or other documents	(where applicable)	
authorizing execution of this Agreement.)	State:	
	Jiait.	

EJCDC[®] C-520, Agreement between Owner and Contractor for Construction Contract (Stipulated Price). Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

(THIS PAGE INTENTIONALLY LEFT BLANK)

EJCDC® C-520, Agreement between Owner and Contractor for Construction Contract (Stipulated Price). Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

Page 8 of 8

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

PERFORMANCE BO

Contractor	Surety	
Name:	Name:	
Address (principal place of business):	Address (principal place of business):	
Owner	Contract	
Name: Fairfield County Utilities	Description (name and location):	
Mailing address (principal place of business):	Airport Sanitary Sewer Extension and Water	
Fairfield County Utilities	Service Design	
6670 Lockville Road NW	3430 Old Columbus Rd NW	
Carroll, Ohio 43112	Carroll, OH 43112	
	Contract Price:	
	Effective Date of Contract:	
Date of Bond: (Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: None See Paragraph 16 Surety and Contractor intending to be lega		
(Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: □ None □ See Paragraph 16 Surety and Contractor, intending to be lega Performance Bond, do each cause this Performance.	ally bound hereby, subject to the terms set forth in this Formance Bond to be duly executed by an authorized office	
(Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: □ None □ See Paragraph 16 Surety and Contractor, intending to be lega Performance Bond, do each cause this Perfo	Ily bound hereby, subject to the terms set forth in this	
(Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: □ None □ See Paragraph 16 Surety and Contractor, intending to be lega Performance Bond, do each cause this Performance.	ally bound hereby, subject to the terms set forth in this Formance Bond to be duly executed by an authorized office	
(Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: None See Paragraph 16 Surety and Contractor, intending to be lega Performance Bond, do each cause this Perfor agent, or representative. Contractor as Principal (Full formal name of Contractor) By:	ally bound hereby, subject to the terms set forth in this Formance Bond to be duly executed by an authorized office Surety (Full formal name of Surety) (corporate seal) By:	
(Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: ☐ None ☐ See Paragraph 16 Surety and Contractor, intending to be lega Performance Bond, do each cause this Performance, and contractor, and con	ally bound hereby, subject to the terms set forth in this formance Bond to be duly executed by an authorized office Surety (Full formal name of Surety) (corporate seal) By: (Signature)(Attach Power of Attorney)	
(Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: None See Paragraph 16 Surety and Contractor, intending to be lega Performance Bond, do each cause this Performance, agent, or representative. Contractor as Principal (Full formal name of Contractor) By: (Signature) Name:	ally bound hereby, subject to the terms set forth in this formance Bond to be duly executed by an authorized office Surety (Full formal name of Surety) (corporate seal) By: (Signature)(Attach Power of Attorney) Name:	
(Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: None See Paragraph 16 Surety and Contractor, intending to be lega Performance Bond, do each cause this Perfor agent, or representative. Contractor as Principal (Full formal name of Contractor) By: (Signature) Name: (Printed or typed)	ally bound hereby, subject to the terms set forth in this formance Bond to be duly executed by an authorized office Surety (Full formal name of Surety) (corporate seal) By: (Signature)(Attach Power of Attorney) Name: (Printed or typed)	
(Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: None See Paragraph 16 Surety and Contractor, intending to be lega Performance Bond, do each cause this Performance, agent, or representative. Contractor as Principal (Full formal name of Contractor) By: (Signature) Name:	ally bound hereby, subject to the terms set forth in this formance Bond to be duly executed by an authorized office Surety (Full formal name of Surety) (corporate seal) By: (Signature)(Attach Power of Attorney) Name:	
(Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: None See Paragraph 16 Surety and Contractor, intending to be lega Performance Bond, do each cause this Performance, agent, or representative. Contractor as Principal (Full formal name of Contractor) By: (Signature) Name: (Printed or typed) Title: Attest:	ally bound hereby, subject to the terms set forth in this formance Bond to be duly executed by an authorized office Surety (Full formal name of Surety) (corporate seal) By: (Signature)(Attach Power of Attorney) Name: (Printed or typed) Title: Attest:	
(Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: None See Paragraph 16 Surety and Contractor, intending to be lega Performance Bond, do each cause this Performance, agent, or representative. Contractor as Principal (Full formal name of Contractor) By: (Signature) Name: (Printed or typed) Title:	ally bound hereby, subject to the terms set forth in this formance Bond to be duly executed by an authorized office Surety (Full formal name of Surety) (corporate seal) By: (Signature)(Attach Power of Attorney) Name: (Printed or typed) Title:	
(Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: None See Paragraph 16 Surety and Contractor, intending to be lega Performance Bond, do each cause this Perfor agent, or representative. Contractor as Principal (Full formal name of Contractor) By: (Signature) Name: (Signature) Name: (Signature) Name:	Ally bound hereby, subject to the terms set forth in this formance Bond to be duly executed by an authorized office Surety (Full formal name of Surety) (corporate seal) By: (Signature)(Attach Power of Attorney) Name: (Printed or typed) Title: Attest: (Signature) Name:	
(Date of Bond cannot be earlier than Effective Date of Co Modifications to this Bond form: None See Paragraph 16 Surety and Contractor, intending to be lega Performance Bond, do each cause this Perfor agent, or representative. Contractor as Principal (Full formal name of Contractor) By: (Signature) Name: (Printed or typed) Title: Attest: (Signature)	Ally bound hereby, subject to the terms set forth in this formance Bond to be duly executed by an authorized office Surety (Full formal name of Surety) (corporate seal) By: (Signature)(Attach Power of Attorney) Name: (Printed or typed) Title: Attest: (Signature)	

EJCDC[®] C-610, Performance Bond.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

and American Society of Civil Engineers. All rights reserved.

NOTICE TO PROCEED

Owner:	Fairfield County Utilities		
Engineer:	PRIME AE Group, Inc.	Engineer's Project No.:	21513
Contractor:	-	Contractor's Project No.:	
Project:	Airport Sanitary Sewer Extension	and Water Service Design	
Effective Date	of Contract:		

Owner hereby notifies Contractor that the Contract Times under the above Contract will commence to run on ______ pursuant to Paragraph 4.01 of the General Conditions.

On that date, Contractor shall start performing its obligations under the Contract Documents. No Work will be done at the Site prior to such date.

In accordance with the Agreement:

The number of days to achieve Substantial Completion is 330 days from the date stated above for the commencement of the Contract Times, resulting in a date for Substantial Completion of ______; and the number of days to achieve readiness for final payment is 365 days from the commencement date of the Contract Times, resulting in a date for readiness for final payment of

Owner:	Fairfield County Utilities	
By (signature):		
Name (printed):	Tony Vogel, PE	
Title:	Director of Utilities	-
Date Issued:		
Copy: Engineer		

- 1. The Contractor and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors, and assigns to the Owner for the performance of the Construction Contract, which is incorporated herein by reference.
- 2. If the Contractor performs the Construction Contract, the Surety and the Contractor shall have no obligation under this Bond, except when applicable to participate in a conference as provided in Paragraph 3.
- 3. If there is no Owner Default under the Construction Contract, the Surety's obligation under this Bond will arise after:
 - 3.1. The Owner first provides notice to the Contractor and the Surety that the Owner is considering declaring a Contractor Default. Such notice may indicate whether the Owner is requesting a conference among the Owner, Contractor, and Surety to discuss the Contractor's performance. If the Owner does not request a conference, the Surety may, within five (5) business days after receipt of the Owner's notice, request such a conference. If the Surety timely requests a conference, the Owner shall attend. Unless the Owner agrees otherwise, any conference requested under this Paragraph 3.1 will be held within ten (10) business days of the Surety's receipt of the Owner's notice. If the Owner, the Contractor, and the Surety agree, the Contractor shall be allowed a reasonable time to perform the Construction Contract, but such an agreement does not waive the Owner's right, if any, subsequently to declare a Contractor Default;
 - 3.2. The Owner declares a Contractor Default, terminates the Construction Contract and notifies the Surety; and
 - 3.3. The Owner has agreed to pay the Balance of the Contract Price in accordance with the terms of the Construction Contract to the Surety or to a contractor selected to perform the Construction Contract.
- 4. Failure on the part of the Owner to comply with the notice requirement in Paragraph 3.1 does not constitute a failure to comply with a condition precedent to the Surety's obligations, or release the Surety from its obligations, except to the extent the Surety demonstrates actual prejudice.
- 5. When the Owner has satisfied the conditions of Paragraph 3, the Surety shall promptly and at the Surety's expense take one of the following actions:
 - 5.1. Arrange for the Contractor, with the consent of the Owner, to perform and complete the Construction Contract;
 - 5.2. Undertake to perform and complete the Construction Contract itself, through its agents or independent contractors;
 - 5.3. Obtain bids or negotiated proposals from qualified contractors acceptable to the Owner for a contract for performance and completion of the Construction Contract, arrange for a contract to be prepared for execution by the Owner and a contractor selected with the Owners concurrence, to be secured with performance and payment bonds executed by a qualified surety equivalent to the bonds issued on the Construction Contract, and pay to the Owner the amount of damages as described in Paragraph 7 in excess of the Balance of the Contract Price incurred by the Owner as a result of the Contractor Default; or
 - 5.4. Waive its right to perform and complete, arrange for completion, or obtain a new contractor, and with reasonable promptness under the circumstances:

EJCDC[®] C-610, Performance Bond.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 10/6/2022

- 5.4.1 After investigation, determine the amount for which it may be liable to the Owner and, as soon as practicable after the amount is determined, make payment to the Owner; or
- 5.4.2 Deny liability in whole or in part and notify the Owner, citing the reasons for denial.
- 6. If the Surety does not proceed as provided in Paragraph 5 with reasonable promptness, the Surety shall be deemed to be in default on this Bond seven days after receipt of an additional written notice from the Owner to the Surety demanding that the Surety perform its obligations under this Bond, and the Owner shall be entitled to enforce any remedy available to the Owner. If the Surety proceeds as provided in Paragraph 5.4, and the Owner refuses the payment, or the Surety has denied liability, in whole or in part, without further notice, the Owner shall be entitled to enforce any remedy available to the Owner.
- 7. If the Surety elects to act under Paragraph 5.1, 5.2, or 5.3, then the responsibilities of the Surety to the Owner will not be greater than those of the Contractor under the Construction Contract, and the responsibilities of the Owner to the Surety will not be greater than those of the Owner under the Construction Contract. Subject to the commitment by the Owner to pay the Balance of the Contract Price, the Surety is obligated, without duplication for:
 - 7.1. the responsibilities of the Contractor for correction of defective work and completion of the Construction Contract;
 - 7.2. additional legal, design professional, and delay costs resulting from the Contractor's Default, and resulting from the actions or failure to act of the Surety under Paragraph 5; and
 - 7.3. liquidated damages, or if no liquidated damages are specified in the Construction Contract, actual damages caused by delayed performance or non-performance of the Contractor.
- 8. If the Surety elects to act under Paragraph 5.1, 5.3, or 5.4, the Surety's liability is limited to the amount of this Bond.
- 9. The Surety shall not be liable to the Owner or others for obligations of the Contractor that are unrelated to the Construction Contract, and the Balance of the Contract Price will not be reduced or set off on account of any such unrelated obligations. No right of action will accrue on this Bond to any person or entity other than the Owner or its heirs, executors, administrators, successors, and assigns.
- 10. The Surety hereby waives notice of any change, including changes of time, to the Construction Contract or to related subcontracts, purchase orders, and other obligations.
- 11. Any proceeding, legal or equitable, under this Bond must be instituted in any court of competent jurisdiction in the location in which the work or part of the work is located and must be instituted within two years after a declaration of Contractor Default or within two years after the Contractor ceased working or within two years after the Surety refuses or fails to perform its obligations under this Bond, whichever occurs first. If the provisions of this paragraph are void or prohibited by law, the minimum periods of limitations available to sureties as a defense in the jurisdiction of the suit will be applicable.
- 12. Notice to the Surety, the Owner, or the Contractor must be mailed or delivered to the address shown on the page on which their signature appears.
- 13. When this Bond has been furnished to comply with a statutory or other legal requirement in the location where the construction was to be performed, any provision in this Bond conflicting with said statutory or legal requirement will be deemed deleted therefrom and provisions conforming to such

EJCDC[®] C-610, Performance Bond.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. statutory or other legal requirement will be deemed incorporated herein. When so furnished, the intent is that this Bond will be construed as a statutory bond and not as a common law bond.

- 14. Definitions
 - 14.1. Balance of the Contract Price—The total amount payable by the Owner to the Contractor under the Construction Contract after all proper adjustments have been made including allowance for the Contractor for any amounts received or to be received by the Owner in settlement of insurance or other claims for damages to which the Contractor is entitled, reduced by all valid and proper payments made to or on behalf of the Contractor under the Construction Contract.
 - 14.2. Construction Contract—The agreement between the Owner and Contractor identified on the cover page, including all Contract Documents and changes made to the agreement and the Contract Documents.
 - 14.3. *Contractor Default*—Failure of the Contractor, which has not been remedied or waived, to perform or otherwise to comply with a material term of the Construction Contract.
 - 14.4. Owner Default—Failure of the Owner, which has not been remedied or waived, to pay the Contractor as required under the Construction Contract or to perform and complete or comply with the other material terms of the Construction Contract.
 - 14.5. *Contract Documents*—All the documents that comprise the agreement between the Owner and Contractor.
- 15. If this Bond is issued for an agreement between a contractor and subcontractor, the term Contractor in this Bond will be deemed to be Subcontractor and the term Owner will be deemed to be Contractor.
- 16. Modifications to this Bond are as follows: None.

PAYMENT BOND

Contract	tor	Surety	
Name:		Name:	
Address	(principal place of business):	Address (principal place of	f business):
Owner		Contract	
Fairfield 6670 Loc	Fairfield County Utilities address <i>(principal place of business)</i> : County Utilities ckville Road NW Dhio 43112	Description (name and Id Airport Sanitary Sewer Design 3430 Old Columbus Ro Carroll, OH 43112 Contract Price:	Extension and Water Service
		Effective Date of Contr	act:
(Date of Boi Modificati	nd cannot be earlier than Effective Date of Contr ions to this Bond form:	act)	
Modificati None Surety an Payment represent	nd cannot be earlier than Effective Date of Contr ions to this Bond form: See Paragraph 18 nd Contractor, intending to be legally Bond, do each cause this Payment Bo	bound hereby, subject to the and to be duly executed by an	
(Date of Boi Modificati D None Surety an Payment represent	nd cannot be earlier than Effective Date of Contr ions to this Bond form: See Paragraph 18 nd Contractor, intending to be legally Bond, do each cause this Payment Bo tative.	bound hereby, subject to the	
(Date of Boi Modificati D None D Surety an Payment represent Contracto	nd cannot be earlier than Effective Date of Contr ions to this Bond form: See Paragraph 18 nd Contractor, intending to be legally Bond, do each cause this Payment Bo tative.	bound hereby, subject to the and to be duly executed by an Surety	
(Date of Boi Modificati None Surety an Payment represent Contracto By:	nd cannot be earlier than Effective Date of Contr ions to this Bond form: See Paragraph 18 ad Contractor, intending to be legally Bond, do each cause this Payment Bo tative. or as Principal	bound hereby, subject to the and to be duly executed by an Surety (Full formal nar By:	authorized officer, agent, or
(Date of Boi Modificati None Surety an Payment represent Contracto By:	nd cannot be earlier than Effective Date of Contr ions to this Bond form: See Paragraph 18 ad Contractor, intending to be legally Bond, do each cause this Payment Bo tative. or as Principal (Full formal name of Contractor)	bound hereby, subject to the and to be duly executed by an Surety (Full formal nar By:	authorized officer, agent, or me of Surety) (corporate seal)
<i>Date of Boi</i> Modificati None Surety an Payment Payment Contracto By: Name:	nd cannot be earlier than Effective Date of Contri ions to this Bond form: See Paragraph 18 ad Contractor, intending to be legally Bond, do each cause this Payment Bo tative. or as Principal (Full formal name of Contractor) (Signature)	bound hereby, subject to the and to be duly executed by an Surety (Full formal nat By: (Signa	authorized officer, agent, or me of Surety) (corporate seal) ture)(Attach Power of Attorney)
(Date of Boi Modificati None [Surety an Payment represent Contracto By: Name: Fitle:	nd cannot be earlier than Effective Date of Contri ions to this Bond form: See Paragraph 18 ad Contractor, intending to be legally Bond, do each cause this Payment Bo tative. or as Principal (Full formal name of Contractor) (Signature) (Printed or typed)	bound hereby, subject to the and to be duly executed by an Surety (Full formal nat By: (Signa Name:	authorized officer, agent, or me of Surety) (corporate seal) ture)(Attach Power of Attorney) (Printed or typed)
(Date of Boi Modificati None Surety an Payment represent Contracto By: Name: Title: Attest:	nd cannot be earlier than Effective Date of Contri ions to this Bond form: See Paragraph 18 nd Contractor, intending to be legally Bond, do each cause this Payment Bo tative. or as Principal (Full formal name of Contractor) (Signature) (Printed or typed) (Signature)	bound hereby, subject to the and to be duly executed by an Surety (Full formal nar By: (Signa Name: Title:	authorized officer, agent, or me of Surety) (corporate seal) ture)(Attach Power of Attorney) (Printed or typed) (Signature)
(Date of Boi Modificati D None Surety an Payment represent	nd cannot be earlier than Effective Date of Contri ions to this Bond form: See Paragraph 18 ad Contractor, intending to be legally Bond, do each cause this Payment Bo tative. or as Principal (Full formal name of Contractor) (Signature) (Printed or typed)	bound hereby, subject to the and to be duly executed by an Surety (Full formal nat By: (Signa Name: Title: Attest:	authorized officer, agent, or me of Surety) (corporate seal) ture)(Attach Power of Attorney) (Printed or typed)

EJCDC® C-615, Payment Bond.

Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

and American Society of Civil Engineers. All rights reserved.

- 1. The Contractor and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors, and assigns to the Owner to pay for labor, materials, and equipment furnished for use in the performance of the Construction Contract, which is incorporated herein by reference, subject to the following terms.
- 2. If the Contractor promptly makes payment of all sums due to Claimants, and defends, indemnifies, and holds harmless the Owner from claims, demands, liens, or suits by any person or entity seeking payment for labor, materials, or equipment furnished for use in the performance of the Construction Contract, then the Surety and the Contractor shall have no obligation under this Bond.
- 3. If there is no Owner Default under the Construction Contract, the Surety's obligation to the Owner under this Bond will arise after the Owner has promptly notified the Contractor and the Surety (at the address described in Paragraph 13) of claims, demands, liens, or suits against the Owner or the Owner's property by any person or entity seeking payment for labor, materials, or equipment furnished for use in the performance of the Construction Contract, and tendered defense of such claims, demands, liens, or suits to the Contractor and the Surety.
- 4. When the Owner has satisfied the conditions in Paragraph 3, the Surety shall promptly and at the Surety's expense defend, indemnify, and hold harmless the Owner against a duly tendered claim, demand, lien, or suit.
- 5. The Surety's obligations to a Claimant under this Bond will arise after the following:
 - 5.1. Claimants who do not have a direct contract with the Contractor
 - 5.1.1. have furnished a written notice of non-payment to the Contractor, stating with substantial accuracy the amount claimed and the name of the party to whom the materials were, or equipment was, furnished or supplied or for whom the labor was done or performed, within ninety (90) days after having last performed labor or last furnished materials or equipment included in the Claim; and
 - 5.1.2. have sent a Claim to the Surety (at the address described in Paragraph 13).
 - 5.2. Claimants who are employed by or have a direct contract with the Contractor have sent a Claim to the Surety (at the address described in Paragraph 13).
- 6. If a notice of non-payment required by Paragraph 5.1.1 is given by the Owner to the Contractor, that is sufficient to satisfy a Claimant's obligation to furnish a written notice of non-payment under Paragraph 5.1.1.
- 7. When a Claimant has satisfied the conditions of Paragraph 5.1 or 5.2, whichever is applicable, the Surety shall promptly and at the Surety's expense take the following actions:
 - 7.1. Send an answer to the Claimant, with a copy to the Owner, within sixty (60) days after receipt of the Claim, stating the amounts that are undisputed and the basis for challenging any amounts that are disputed; and
 - 7.2. Pay or arrange for payment of any undisputed amounts.
 - 7.3. The Surety's failure to discharge its obligations under Paragraph 7.1 or 7.2 will not be deemed to constitute a waiver of defenses the Surety or Contractor may have or acquire as to a Claim, except as to undisputed amounts for which the Surety and Claimant have reached agreement. If, however, the Surety fails to discharge its obligations under Paragraph 7.1 or 7.2, the Surety shall indemnify the Claimant for the reasonable attorney's fees the Claimant incurs thereafter to recover any sums found to be due and owing to the Claimant.

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 10/6/2022

- 8. The Surety's total obligation will not exceed the amount of this Bond, plus the amount of reasonable attorney's fees provided under Paragraph 7.3, and the amount of this Bond will be credited for any payments made in good faith by the Surety.
- 9. Amounts owed by the Owner to the Contractor under the Construction Contract will be used for the performance of the Construction Contract and to satisfy claims, if any, under any construction performance bond. By the Contractor furnishing and the Owner accepting this Bond, they agree that all funds earned by the Contractor in the performance of the Construction Contract are dedicated to satisfying obligations of the Contractor and Surety under this Bond, subject to the Owner's priority to use the funds for the completion of the work.
- 10. The Surety shall not be liable to the Owner, Claimants, or others for obligations of the Contractor that are unrelated to the Construction Contract. The Owner shall not be liable for the payment of any costs or expenses of any Claimant under this Bond, and shall have under this Bond no obligation to make payments to or give notice on behalf of Claimants, or otherwise have any obligations to Claimants under this Bond.
- 11. The Surety hereby waives notice of any change, including changes of time, to the Construction Contract or to related subcontracts, purchase orders, and other obligations.
- 12. No suit or action will be commenced by a Claimant under this Bond other than in a court of competent jurisdiction in the state in which the project that is the subject of the Construction Contract is located or after the expiration of one year from the date (1) on which the Claimant sent a Claim to the Surety pursuant to Paragraph 5.1.2 or 5.2, or (2) on which the last labor or service was performed by anyone or the last materials or equipment were furnished by anyone under the Construction Contract, whichever of (1) or (2) first occurs. If the provisions of this paragraph are void or prohibited by law, the minimum period of limitation available to sureties as a defense in the jurisdiction of the suit will be applicable.
- 13. Notice and Claims to the Surety, the Owner, or the Contractor must be mailed or delivered to the address shown on the page on which their signature appears. Actual receipt of notice or Claims, however accomplished, will be sufficient compliance as of the date received.
- 14. When this Bond has been furnished to comply with a statutory or other legal requirement in the location where the construction was to be performed, any provision in this Bond conflicting with said statutory or legal requirement will be deemed deleted here from and provisions conforming to such statutory or other legal requirement will be deemed incorporated herein. When so furnished, the intent is that this Bond will be construed as a statutory bond and not as a common law bond.
- 15. Upon requests by any person or entity appearing to be a potential beneficiary of this Bond, the Contractor and Owner shall promptly furnish a copy of this Bond or shall permit a copy to be made.
- 16. Definitions
 - 16.1. Claim—A written statement by the Claimant including at a minimum:

16.1.1. The name of the Claimant;

- 16.1.2. The name of the person for whom the labor was done, or materials or equipment furnished;
- 16.1.3. A copy of the agreement or purchase order pursuant to which labor, materials, or equipment was furnished for use in the performance of the Construction Contract;
- 16.1.4. A brief description of the labor, materials, or equipment furnished;

EJCDC[®] C-615, Payment Bond.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

- 16.1.5. The date on which the Claimant last performed labor or last furnished materials or equipment for use in the performance of the Construction Contract;
- 16.1.6. The total amount earned by the Claimant for labor, materials, or equipment furnished as of the date of the Claim;
- 16.1.7. The total amount of previous payments received by the Claimant; and
- 16.1.8. The total amount due and unpaid to the Claimant for labor, materials, or equipment furnished as of the date of the Claim.
- 16.2. Claimant—An individual or entity having a direct contract with the Contractor or with a subcontractor of the Contractor to furnish labor, materials, or equipment for use in the performance of the Construction Contract. The term Claimant also includes any individual or entity that has rightfully asserted a claim under an applicable mechanic's lien or similar statute against the real property upon which the Project is located. The intent of this Bond is to include without limitation in the terms of "labor, materials, or equipment" that part of the water, gas, power, light, heat, oil, gasoline, telephone service, or rental equipment used in the Construction Contract, architectural and engineering services required for performance of the work of the Contractor and the Contractor's subcontractors, and all other items for which a mechanic's lien may be asserted in the jurisdiction where the labor, materials, or equipment were furnished.
- 16.3. *Construction Contract*—The agreement between the Owner and Contractor identified on the cover page, including all Contract Documents and all changes made to the agreement and the Contract Documents.
- 16.4. Owner Default—Failure of the Owner, which has not been remedied or waived, to pay the Contractor as required under the Construction Contract or to perform and complete or comply with the other material terms of the Construction Contract.
- 16.5. *Contract Documents*—All the documents that comprise the agreement between the Owner and Contractor.
- 17. If this Bond is issued for an agreement between a contractor and subcontractor, the term Contractor in this Bond will be deemed to be Subcontractor and the term Owner will be deemed to be Contractor.
- 18. Modifications to this Bond are as follows: None.

CERTIFICATE OF SUBSTANTIAL COMPLETION

Owner:	Fairfield County Utilities		
Engineer:	PRIME AE Group, Inc.	Engineer's Project No.:	21513
Contractor:		Contractor's Project No.:	
Project:	Airport Sanitary Sewer Extension	and Water Service Design	

This
Preliminary
Final Certificate of Substantial Completion applies to:

□ All Work □ The following specified portions of the Work:

Date of Substantial Completion:

The Work to which this Certificate applies has been inspected by authorized representatives of Owner, Contractor, and Engineer, and found to be substantially complete. The Date of Substantial Completion of the Work or portion thereof designated above is hereby established, subject to the provisions of the Contract pertaining to Substantial Completion. The date of Substantial Completion in the final Certificate of Substantial Completion marks the commencement of the contractual correction period and applicable warranties required by the Contract.

A punch list of items to be completed or corrected is attached to this Certificate. This list may not be allinclusive, and the failure to include any items on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents.

Amendments of contractual responsibilities recorded in this Certificate should be the product of mutual agreement of Owner and Contractor; see Paragraph 15.03.D of the General Conditions.

The responsibilities between Owner and Contractor for security, operation, safety, maintenance, heat, utilities, insurance, and warranties upon Owner's use or occupancy of the Work must be as provided in the Contract, except as amended as follows:

Amendments to Owner's Responsibilities:
None
As follows:

Amendments to Contractor's Responsibilities:
None
As follows:

The following documents are attached to and made a part of this Certificate:

This Certificate does not constitute an acceptance of Work not in accordance with the Contract Documents, nor is it a release of Contractor's obligation to complete the Work in accordance with the Contract Documents.

Engineer

By (signature):		
Name (printed):		
Title:	<u></u>	

(THIS PAGE INTENTIONALLY LEFT BLANK)

EJCDC® C-625, Certificate of Substantial Completion. Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 2 of 2

NOTICE OF ACCEPTABILITY OF WORK

Owner:	Fairfield County Utilities		
Engineer:	PRIME AE Group, Inc.	Engineer's Project No.:	21513
Contractor:		Contractor's Project No.:	
Project:	Airport Sanitary Sewer Extension an	d Water Service Design	
Notice Date:		_	
Effective Date o	f the Construction Contract:	A.)	

The Engineer hereby gives notice to the Owner and Contractor that Engineer recommends final payment to Contractor, and that the Work furnished and performed by Contractor under the Construction Contract is acceptable, expressly subject to the provisions of the Construction Contract's Contract Documents ("Contract Documents") and of the Agreement between Owner and Engineer for Professional Services dated ______ ("Owner-Engineer Agreement"). This Notice of Acceptability of Work (Notice) is made expressly subject to the following terms and conditions to which all who receive and rely on said Notice agree:

- 1. This Notice has been prepared with the skill and care ordinarily used by members of the engineering profession practicing under similar conditions at the same time and in the same locality.
- 2. This Notice reflects and is an expression of the Engineer's professional opinion.
- 3. This Notice has been prepared to the best of Engineer's knowledge, information, and belief as of the Notice Date.
- 4. This Notice is based entirely on and expressly limited by the scope of services Engineer has been employed by Owner to perform or furnish during construction of the Project (including observation of the Contractor's Work) under the Owner-Engineer Agreement, and applies only to facts that are within Engineer's knowledge or could reasonably have been ascertained by Engineer as a result of carrying out the responsibilities specifically assigned to Engineer under such Owner-Engineer Agreement.
- 5. This Notice is not a guarantee or warranty of Contractor's performance under the Construction Contract, an acceptance of Work that is not in accordance with the Contract Documents, including but not limited to defective Work discovered after final inspection, nor an assumption of responsibility for any failure of Contractor to furnish and perform the Work thereunder in accordance with the Contract Documents, or to otherwise comply with the Contract Documents or the terms of any special guarantees specified therein.
- 6. This Notice does not relieve Contractor of any surviving obligations under the Construction Contract, and is subject to Owner's reservations of rights with respect to completion and final payment.

Engineer

By (signature):	
Name (printed):	
Title:	

EJCDC® C-626, Notice of Acceptability of Work.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

and American Society of Civil Engineers. All rights reserved.

Page 1 of 2

(THIS PAGE INTENTIONALLY LEFT BLANK)

EJCDC® C-626, Notice of Acceptability of Work. Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 2 of 2

STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

TABLE OF CONTENTS

	Page			
Article 1	L—Definitions and Terminology1			
1.01	Defined Terms1			
1.02	Terminology6			
Article 2—Preliminary Matters				
2.01	Delivery of Performance and Payment Bonds; Evidence of Insurance7			
2.02	Copies of Documents			
2.03	Before Starting Construction			
2.04	Preconstruction Conference; Designation of Authorized Representatives			
2.05	Acceptance of Schedules8			
2.06	Electronic Transmittals8			
Article 3	-Contract Documents: Intent, Requirements, Reuse9			
3.01	Intent9			
3.02	Reference Standards9			
3.03	Reporting and Resolving Discrepancies10			
3.04	Requirements of the Contract Documents10			
3.05	Reuse of Documents11			
Article 4	—Commencement and Progress of the Work11			
4.01	Commencement of Contract Times; Notice to Proceed11			
4.02	Starting the Work			
4.03	Reference Points11			
4.04	Progress Schedule12			
4.05	Delays in Contractor's Progress12			
Article 5	-Site; Subsurface and Physical Conditions; Hazardous Environmental Conditions			
5.01	Availability of Lands			
5.02	Use of Site and Other Areas14			
5.03	Subsurface and Physical Conditions15			
5.04	Differing Subsurface or Physical Conditions16			

EJCDC® C-700, Standard General Conditions of the Construction Contract.

Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

5.05	Underground Facilities17
5.06	Hazardous Environmental Conditions at Site19
Article 6	-Bonds and Insurance21
6.01	Performance, Payment, and Other Bonds21
6.02	Insurance—General Provisions22
6.03	Contractor's Insurance
6.04	Builder's Risk and Other Property Insurance
6.05	Property Losses; Subrogation25
6.06	Receipt and Application of Property Insurance Proceeds27
Article 7	Contractor's Responsibilities27
7.01	Contractor's Means and Methods of Construction
7.02	Supervision and Superintendence
7.03	Labor; Working Hours27
7.04	Services, Materials, and Equipment28
7.05	"Or Equals"28
7.06	Substitutes
7.07	Concerning Subcontractors and Suppliers
7.08	Patent Fees and Royalties
7.09	Permits
7.10	Taxes
7.11	Laws and Regulations
7.12	Record Documents
7.13	Safety and Protection
7.14	Hazard Communication Programs
7.15	Emergencies
7.16	Submittals
7.17	Contractor's General Warranty and Guarantee
7.18	Indemnification
7.19	Delegation of Professional Design Services
Article 8	-Other Work at the Site40
8.01	Other Work
8.02	Coordination41
8.03	Legal Relationships41

EJCDC® C-700, Standard General Conditions of the Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

and American Society of Civil Engineers. All rights reserved.

TOC Page 2 of 5

...

Article 9	–Owner's Responsibilities42
9.01	Communications to Contractor42
9.02	Replacement of Engineer42
9.03	Furnish Data42
9.04	Pay When Due42
9.05	Lands and Easements; Reports, Tests, and Drawings43
9.06	Insurance43
9.07	Change Orders
9.08	Inspections, Tests, and Approvals43
9.09	Limitations on Owner's Responsibilities43
9.10	Undisclosed Hazardous Environmental Condition43
9.11	Evidence of Financial Arrangements43
9.12	Safety Programs43
Article 1	0—Engineer's Status During Construction44
10.01	Owner's Representative
10.02	Visits to Site
10.03	Resident Project Representative
10.04	Engineer's Authority
10.05	Determinations for Unit Price Work45
10.06	Decisions on Requirements of Contract Documents and Acceptability of Work45
10.07	Limitations on Engineer's Authority and Responsibilities45
10.08	Compliance with Safety Program45
Article 1	1—Changes to the Contract46
11.01	Amending and Supplementing the Contract46
11.02	Change Orders46
11.03	Work Change Directives46
11.04	Field Orders47
11.05	Owner-Authorized Changes in the Work47
11.06	Unauthorized Changes in the Work47
11.07	Change of Contract Price47
11.08	Change of Contract Times49
11.09	Change Proposals
11.10	Notification to Surety50

EJCDC[®] C-700, Standard General Conditions of the Construction Contract. Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

.

TOC Page 3 of 5

Article 12–	-Claims	50
12.01	Claims	50
Article 13—Cost of the Work; Allowances; Unit Price Work51		
13.01	Cost of the Work	51
13.02	Allowances	55
13.03	Unit Price Work	55
Article 14–	-Tests and Inspections; Correction, Removal, or Acceptance of Defective Work	56
14.01	Access to Work	56
14.02	Tests, Inspections, and Approvals	56
14.03	Defective Work	57
14.04	Acceptance of Defective Work	58
14.05	Uncovering Work	58
14.06	Owner May Stop the Work	58
14.07	Owner May Correct Defective Work	59
Article 15–	-Payments to Contractor; Set-Offs; Completion; Correction Period	59
15.01	Progress Payments	59
15.02	Contractor's Warranty of Title	62
15.03	Substantial Completion	62
15.04	Partial Use or Occupancy	63
15.05	Final Inspection	64
15.06	Final Payment	64
15.07	Waiver of Claims	65
15.08	Correction Period	66
Article 16-	-Suspension of Work and Termination	67
16.01	Owner May Suspend Work	67
16.02	Owner May Terminate for Cause	67
16.03	Owner May Terminate for Convenience	68
16.04	Contractor May Stop Work or Terminate	68
Article 17–	-Final Resolution of Disputes	69
17.01	Methods and Procedures	69
Article 18-	-Miscellaneous	69
18.01	Giving Notice	69
18.02	Computation of Times	69

EJCDC® C-700, Standard General Conditions of the Construction Contract. Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. TOC Page 4 of 5

 FAIRFIELD COUNTY UTILITIES
 AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

 10/6/2022
 10/6/2022

18.03	Cumulative Remedies	70
18.04	Limitation of Damages	70
18.05	No Waiver	70
18.06	Survival of Obligations	70
18.07	Controlling Law	70
18.08	Assignment of Contract	70
18.09	Successors and Assigns	70
18.10	Headings	70

STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

ARTICLE 1—DEFINITIONS AND TERMINOLOGY

1.01 Defined Terms

- A. Wherever used in the Bidding Requirements or Contract Documents, a term printed with initial capital letters, including the term's singular and plural forms, will have the meaning indicated in the definitions below. In addition to terms specifically defined, terms with initial capital letters in the Contract Documents include references to identified articles and paragraphs, and the titles of other documents or forms.
 - 1. Addenda—Written or graphic instruments issued prior to the opening of Bids which clarify, correct, or change the Bidding Requirements or the proposed Contract Documents.
 - 2. Agreement—The written instrument, executed by Owner and Contractor, that sets forth the Contract Price and Contract Times, identifies the parties and the Engineer, and designates the specific items that are Contract Documents.
 - 3. Application for Payment—The document prepared by Contractor, in a form acceptable to Engineer, to request progress or final payments, and which is to be accompanied by such supporting documentation as is required by the Contract Documents.
 - 4. *Bid*—The offer of a Bidder submitted on the prescribed form setting forth the prices for the Work to be performed.
 - 5. Bidder—An individual or entity that submits a Bid to Owner.
 - 6. *Bidding Documents*—The Bidding Requirements, the proposed Contract Documents, and all Addenda.
 - 7. *Bidding Requirements*—The Advertisement or invitation to bid, Instructions to Bidders, Bid Bond or other Bid security, if any, the Bid Form, and the Bid with any attachments.
 - 8. Change Order—A document which is signed by Contractor and Owner and authorizes an addition, deletion, or revision in the Work or an adjustment in the Contract Price or the Contract Times, or other revision to the Contract, issued on or after the Effective Date of the Contract.
 - 9. Change Proposal—A written request by Contractor, duly submitted in compliance with the procedural requirements set forth herein, seeking an adjustment in Contract Price or Contract Times; contesting an initial decision by Engineer concerning the requirements of the Contract Documents or the acceptability of Work under the Contract Documents; challenging a set-off against payments due; or seeking other relief with respect to the terms of the Contract.
 - 10. Claim
 - A demand or assertion by Owner directly to Contractor, duly submitted in compliance with the procedural requirements set forth herein, seeking an adjustment of Contract Price or Contract Times; contesting an initial decision by Engineer concerning the

EJCDC® C-700, Standard General Conditions of the Construction Contract.

Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

requirements of the Contract Documents or the acceptability of Work under the Contract Documents; contesting Engineer's decision regarding a Change Proposal; seeking resolution of a contractual issue that Engineer has declined to address; or seeking other relief with respect to the terms of the Contract.

- b. A demand or assertion by Contractor directly to Owner, duly submitted in compliance with the procedural requirements set forth herein, contesting Engineer's decision regarding a Change Proposal, or seeking resolution of a contractual issue that Engineer has declined to address.
- c. A demand or assertion by Owner or Contractor, duly submitted in compliance with the procedural requirements set forth herein, made pursuant to Paragraph 12.01.A.4, concerning disputes arising after Engineer has issued a recommendation of final payment.
- d. A demand for money or services by a third party is not a Claim.
- 11. Constituent of Concern—Asbestos, petroleum, radioactive materials, polychlorinated biphenyls (PCBs), lead-based paint (as defined by the HUD/EPA standard), hazardous waste, and any substance, product, waste, or other material of any nature whatsoever that is or becomes listed, regulated, or addressed pursuant to Laws and Regulations regulating, relating to, or imposing liability or standards of conduct concerning, any hazardous, toxic, or dangerous waste, substance, or material.
- 12. Contract—The entire and integrated written contract between Owner and Contractor concerning the Work.
- 13. Contract Documents—Those items so designated in the Agreement, and which together comprise the Contract.
- 14. *Contract Price*—The money that Owner has agreed to pay Contractor for completion of the Work in accordance with the Contract Documents.
- 15. *Contract Times*—The number of days or the dates by which Contractor shall: (a) achieve Milestones, if any; (b) achieve Substantial Completion; and (c) complete the Work.
- 16. *Contractor*—The individual or entity with which Owner has contracted for performance of the Work.
- 17. Cost of the Work—See Paragraph 13.01 for definition.
- 18. Drawings—The part of the Contract that graphically shows the scope, extent, and character of the Work to be performed by Contractor.
- 19. *Effective Date of the Contract*—The date, indicated in the Agreement, on which the Contract becomes effective.
- 20. *Electronic Document*—Any Project-related correspondence, attachments to correspondence, data, documents, drawings, information, or graphics, including but not limited to Shop Drawings and other Submittals, that are in an electronic or digital format.
- 21. Electronic Means—Electronic mail (email), upload/download from a secure Project website, or other communications methods that allow: (a) the transmission or communication of Electronic Documents; (b) the documentation of transmissions, including sending and receipt; (c) printing of the transmitted Electronic Document by the

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

EJCDC[®] C-700, Standard General Conditions of the Construction Contract.

recipient; (d) the storage and archiving of the Electronic Document by sender and recipient; and (e) the use by recipient of the Electronic Document for purposes permitted by this Contract. Electronic Means does not include the use of text messaging, or of Facebook, Twitter, Instagram, or similar social media services for transmission of Electronic Documents.

- 22. Engineer—The individual or entity named as such in the Agreement.
- 23. *Field Order*—A written order issued by Engineer which requires minor changes in the Work but does not change the Contract Price or the Contract Times.
- 24. *Hazardous Environmental Condition*—The presence at the Site of Constituents of Concern in such quantities or circumstances that may present a danger to persons or property exposed thereto.
 - a. The presence at the Site of materials that are necessary for the execution of the Work, or that are to be incorporated into the Work, and that are controlled and contained pursuant to industry practices, Laws and Regulations, and the requirements of the Contract, is not a Hazardous Environmental Condition.
 - b. The presence of Constituents of Concern that are to be removed or remediated as part of the Work is not a Hazardous Environmental Condition.
 - c. The presence of Constituents of Concern as part of the routine, anticipated, and obvious working conditions at the Site, is not a Hazardous Environmental Condition.
- 25. Laws and Regulations; Laws or Regulations—Any and all applicable laws, statutes, rules, regulations, ordinances, codes, and binding decrees, resolutions, and orders of any and all governmental bodies, agencies, authorities, and courts having jurisdiction.
- 26. *Liens*—Charges, security interests, or encumbrances upon Contract-related funds, real property, or personal property.
- 27. *Milestone*—A principal event in the performance of the Work that the Contract requires Contractor to achieve by an intermediate completion date, or by a time prior to Substantial Completion of all the Work.
- 28. *Notice of Award*—The written notice by Owner to a Bidder of Owner's acceptance of the Bid.
- 29. Notice to Proceed—A written notice by Owner to Contractor fixing the date on which the Contract Times will commence to run and on which Contractor shall start to perform the Work.
- 30. *Owner*—The individual or entity with which Contractor has contracted regarding the Work, and which has agreed to pay Contractor for the performance of the Work, pursuant to the terms of the Contract.
- 31. *Progress Schedule*—A schedule, prepared and maintained by Contractor, describing the sequence and duration of the activities comprising Contractor's plan to accomplish the Work within the Contract Times.
- 32. *Project*—The total undertaking to be accomplished for Owner by engineers, contractors, and others, including planning, study, design, construction, testing, commissioning, and start-up, and of which the Work to be performed under the Contract Documents is a part.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

EJCDC® C-700, Standard General Conditions of the Construction Contract.

- 33. Resident Project Representative—The authorized representative of Engineer assigned to assist Engineer at the Site. As used herein, the term Resident Project Representative (RPR) includes any assistants or field staff of Resident Project Representative.
- 34. Samples—Physical examples of materials, equipment, or workmanship that are representative of some portion of the Work and that establish the standards by which such portion of the Work will be judged.
- 35. Schedule of Submittals—A schedule, prepared and maintained by Contractor, of required submittals and the time requirements for Engineer's review of the submittals.
- 36. Schedule of Values—A schedule, prepared and maintained by Contractor, allocating portions of the Contract Price to various portions of the Work and used as the basis for reviewing Contractor's Applications for Payment.
- 37. Shop Drawings—All drawings, diagrams, illustrations, schedules, and other data or information that are specifically prepared or assembled by or for Contractor and submitted by Contractor to illustrate some portion of the Work. Shop Drawings, whether approved or not, are not Drawings and are not Contract Documents.
- 38. Site—Lands or areas indicated in the Contract Documents as being furnished by Owner upon which the Work is to be performed, including rights-of-way and easements, and such other lands or areas furnished by Owner which are designated for the use of Contractor.
- 39. Specifications—The part of the Contract that consists of written requirements for materials, equipment, systems, standards, and workmanship as applied to the Work, and certain administrative requirements and procedural matters applicable to the Work.
- 40. *Subcontractor*—An individual or entity having a direct contract with Contractor or with any other Subcontractor for the performance of a part of the Work.
- 41. Submittal—A written or graphic document, prepared by or for Contractor, which the Contract Documents require Contractor to submit to Engineer, or that is indicated as a Submittal in the Schedule of Submittals accepted by Engineer. Submittals may include Shop Drawings and Samples; schedules; product data; Owner-delegated designs; sustainable design information; information on special procedures; testing plans; results of tests and evaluations, source quality-control testing and inspections, and field or Site quality-control testing and inspections; warranties and certifications; Suppliers' instructions and reports; records of delivery of spare parts and tools; operations and maintenance data; Project photographic documentation; record documents; and other such documents required by the Contract Documents. Submittals, whether or not approved or accepted by Engineer, are not Contract Documents. Change Proposals, Change Orders, Claims, notices, Applications for Payment, and requests for interpretation or clarification are not Submittals.
- 42. Substantial Completion—The time at which the Work (or a specified part thereof) has progressed to the point where, in the opinion of Engineer, the Work (or a specified part thereof) is sufficiently complete, in accordance with the Contract Documents, so that the Work (or a specified part thereof) can be utilized for the purposes for which it is intended. The terms "substantially complete" and "substantially completed" as applied to all or part of the Work refer to Substantial Completion of such Work.

- 43. Successful Bidder—The Bidder to which the Owner makes an award of contract.
- 44. Supplementary Conditions—The part of the Contract that amends or supplements these General Conditions.
- 45. Supplier—A manufacturer, fabricator, supplier, distributor, or vendor having a direct contract with Contractor or with any Subcontractor to furnish materials or equipment to be incorporated in the Work by Contractor or a Subcontractor.
- 46. Technical Data
 - a. Those items expressly identified as Technical Data in the Supplementary Conditions, with respect to either (1) existing subsurface conditions at or adjacent to the Site, or existing physical conditions at or adjacent to the Site including existing surface or subsurface structures (except Underground Facilities) or (2) Hazardous Environmental Conditions at the Site.
 - b. If no such express identifications of Technical Data have been made with respect to conditions at the Site, then Technical Data is defined, with respect to conditions at the Site under Paragraphs 5.03, 5.04, and 5.06, as the data contained in boring logs, recorded measurements of subsurface water levels, assessments of the condition of subsurface facilities, laboratory test results, and other factual, objective information regarding conditions at the Site that are set forth in any geotechnical, environmental, or other Site or facilities conditions report prepared for the Project and made available to Contractor.
 - c. Information and data regarding the presence or location of Underground Facilities are not intended to be categorized, identified, or defined as Technical Data, and instead Underground Facilities are shown or indicated on the Drawings.
- 47. Underground Facilities—All active or not-in-service underground lines, pipelines, conduits, ducts, encasements, cables, wires, manholes, vaults, tanks, tunnels, or other such facilities or systems at the Site, including but not limited to those facilities or systems that produce, transmit, distribute, or convey telephone or other communications, cable television, fiber optic transmissions, power, electricity, light, heat, gases, oil, crude oil products, liquid petroleum products, water, steam, waste, wastewater, storm water, other liquids or chemicals, or traffic or other control systems. An abandoned facility or system is not an Underground Facility.
- 48. Unit Price Work—Work to be paid for on the basis of unit prices.
- 49. Work—The entire construction or the various separately identifiable parts thereof required to be provided under the Contract Documents. Work includes and is the result of performing or providing all labor, services, and documentation necessary to produce such construction; furnishing, installing, and incorporating all materials and equipment into such construction; and may include related services such as testing, start-up, and commissioning, all as required by the Contract Documents.
- 50. Work Change Directive—A written directive to Contractor issued on or after the Effective Date of the Contract, signed by Owner and recommended by Engineer, ordering an addition, deletion, or revision in the Work.

1.02 Terminology

- A. The words and terms discussed in Paragraphs 1.02.B, C, D, and E are not defined terms that require initial capital letters, but, when used in the Bidding Requirements or Contract Documents, have the indicated meaning.
- B. Intent of Certain Terms or Adjectives: The Contract Documents include the terms "as allowed," "as approved," "as ordered," "as directed" or terms of like effect or import to authorize an exercise of professional judgment by Engineer. In addition, the adjectives "reasonable," "suitable," "acceptable," "proper," "satisfactory," or adjectives of like effect or import are used to describe an action or determination of Engineer as to the Work. It is intended that such exercise of professional judgment, action, or determination will be solely to evaluate, in general, the Work for compliance with the information in the Contract Documents and with the design concept of the Project as a functioning whole as shown or indicated in the Contract Documents (unless there is a specific statement indicating otherwise). The use of any such term or adjective is not intended to and shall not be effective to assign to Engineer any duty or authority to supervise or direct the performance of the Work, or any duty or authority to undertake responsibility contrary to the provisions of Article 10 or any other provision of the Contract Documents.
- C. *Day*: The word "day" means a calendar day of 24 hours measured from midnight to the next midnight.
- D. *Defective*: The word "defective," when modifying the word "Work," refers to Work that is unsatisfactory, faulty, or deficient in that it:
 - 1. does not conform to the Contract Documents;
 - 2. does not meet the requirements of any applicable inspection, reference standard, test, or approval referred to in the Contract Documents; or
 - 3. has been damaged prior to Engineer's recommendation of final payment (unless responsibility for the protection thereof has been assumed by Owner at Substantial Completion in accordance with Paragraph 15.03 or Paragraph 15.04).
- E. Furnish, Install, Perform, Provide
 - 1. The word "furnish," when used in connection with services, materials, or equipment, means to supply and deliver said services, materials, or equipment to the Site (or some other specified location) ready for use or installation and in usable or operable condition.
 - 2. The word "install," when used in connection with services, materials, or equipment, means to put into use or place in final position said services, materials, or equipment complete and ready for intended use.
 - 3. The words "perform" or "provide," when used in connection with services, materials, or equipment, means to furnish and install said services, materials, or equipment complete and ready for intended use.
 - 4. If the Contract Documents establish an obligation of Contractor with respect to specific services, materials, or equipment, but do not expressly use any of the four words "furnish," "install," "perform," or "provide," then Contractor shall furnish and install said services, materials, or equipment complete and ready for intended use.

- F. Contract Price or Contract Times: References to a change in "Contract Price or Contract Times" or "Contract Times or Contract Price" or similar, indicate that such change applies to (1) Contract Price, (2) Contract Times, or (3) both Contract Price and Contract Times, as warranted, even if the term "or both" is not expressed.
- G. Unless stated otherwise in the Contract Documents, words or phrases that have a well-known technical or construction industry or trade meaning are used in the Contract Documents in accordance with such recognized meaning.

ARTICLE 2—PRELIMINARY MATTERS

- 2.01 Delivery of Performance and Payment Bonds; Evidence of Insurance
 - A. *Performance and Payment Bonds*: When Contractor delivers the signed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner the performance bond and payment bond (if the Contract requires Contractor to furnish such bonds).
 - B. Evidence of Contractor's Insurance: When Contractor delivers the signed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner, with copies to each additional insured (as identified in the Contract), the certificates, endorsements, and other evidence of insurance required to be provided by Contractor in accordance with Article 6, except to the extent the Supplementary Conditions expressly establish other dates for delivery of specific insurance policies.
 - C. Evidence of Owner's Insurance: After receipt of the signed counterparts of the Agreement and all required bonds and insurance documentation, Owner shall promptly deliver to Contractor, with copies to each additional insured (as identified in the Contract), the certificates and other evidence of insurance required to be provided by Owner under Article 6.
- 2.02 Copies of Documents
 - A. Owner shall furnish to Contractor four printed copies of the Contract (including one fully signed counterpart of the Agreement), and one copy in electronic portable document format (PDF). Additional printed copies will be furnished upon request at the cost of reproduction.
 - B. Owner shall maintain and safeguard at least one original printed record version of the Contract, including Drawings and Specifications signed and sealed by Engineer and other design professionals. Owner shall make such original printed record version of the Contract available to Contractor for review. Owner may delegate the responsibilities under this provision to Engineer.
- 2.03 Before Starting Construction
 - A. *Preliminary Schedules*: Within 10 days after the Effective Date of the Contract (or as otherwise required by the Contract Documents), Contractor shall submit to Engineer for timely review:
 - a preliminary Progress Schedule indicating the times (numbers of days or dates) for starting and completing the various stages of the Work, including any Milestones specified in the Contract;
 - 2. a preliminary Schedule of Submittals; and
 - 3. a preliminary Schedule of Values for all of the Work which includes quantities and prices of items which when added together equal the Contract Price and subdivides the Work

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

EJCDC® C-700, Standard General Conditions of the Construction Contract.

into component parts in sufficient detail to serve as the basis for progress payments during performance of the Work. Such prices will include an appropriate amount of overhead and profit applicable to each item of Work.

2.04 Preconstruction Conference; Designation of Authorized Representatives

- A. Before any Work at the Site is started, a conference attended by Owner, Contractor, Engineer, and others as appropriate will be held to establish a working understanding among the parties as to the Work, and to discuss the schedules referred to in Paragraph 2.03.A, procedures for handling Shop Drawings, Samples, and other Submittals, processing Applications for Payment, electronic or digital transmittals, and maintaining required records.
- B. At this conference Owner and Contractor each shall designate, in writing, a specific individual to act as its authorized representative with respect to the services and responsibilities under the Contract. Such individuals shall have the authority to transmit and receive information, render decisions relative to the Contract, and otherwise act on behalf of each respective party.

2.05 Acceptance of Schedules

- A. At least 10 days before submission of the first Application for Payment a conference, attended by Contractor, Engineer, and others as appropriate, will be held to review the schedules submitted in accordance with Paragraph 2.03.A. No progress payment will be made to Contractor until acceptable schedules are submitted to Engineer.
 - The Progress Schedule will be acceptable to Engineer if it provides an orderly progression of the Work to completion within the Contract Times. Such acceptance will not impose on Engineer responsibility for the Progress Schedule, for sequencing, scheduling, or progress of the Work, nor interfere with or relieve Contractor from Contractor's full responsibility therefor.
 - 2. Contractor's Schedule of Submittals will be acceptable to Engineer if it provides a workable arrangement for reviewing and processing the required submittals.
 - Contractor's Schedule of Values will be acceptable to Engineer as to form and substance if it provides a reasonable allocation of the Contract Price to the component parts of the Work.
 - 4. If a schedule is not acceptable, Contractor will have an additional 10 days to revise and resubmit the schedule.

2.06 *Electronic Transmittals*

- A. Except as otherwise stated elsewhere in the Contract, the Owner, Engineer, and Contractor may send, and shall accept, Electronic Documents transmitted by Electronic Means.
- B. If the Contract does not establish protocols for Electronic Means, then Owner, Engineer, and Contractor shall jointly develop such protocols.
- C. Subject to any governing protocols for Electronic Means, when transmitting Electronic Documents by Electronic Means, the transmitting party makes no representations as to long-term compatibility, usability, or readability of the Electronic Documents resulting from the recipient's use of software application packages, operating systems, or computer hardware differing from those used in the drafting or transmittal of the Electronic Documents.

ARTICLE 3—CONTRACT DOCUMENTS: INTENT, REQUIREMENTS, REUSE

- 3.01 Intent
 - A. The Contract Documents are complementary; what is required by one Contract Document is as binding as if required by all.
 - B. It is the intent of the Contract Documents to describe a functionally complete Project (or part thereof) to be constructed in accordance with the Contract Documents.
 - C. Unless otherwise stated in the Contract Documents, if there is a discrepancy between the electronic versions of the Contract Documents (including any printed copies derived from such electronic versions) and the printed record version, the printed record version will govern.
 - D. The Contract supersedes prior negotiations, representations, and agreements, whether written or oral.
 - E. Engineer will issue clarifications and interpretations of the Contract Documents as provided herein.
 - F. Any provision or part of the Contract Documents held to be void or unenforceable under any Law or Regulation will be deemed stricken, and all remaining provisions will continue to be valid and binding upon Owner and Contractor, which agree that the Contract Documents will be reformed to replace such stricken provision or part thereof with a valid and enforceable provision that comes as close as possible to expressing the intention of the stricken provision.
 - G. Nothing in the Contract Documents creates:
 - 1. any contractual relationship between Owner or Engineer and any Subcontractor, Supplier, or other individual or entity performing or furnishing any of the Work, for the benefit of such Subcontractor, Supplier, or other individual or entity; or
 - 2. any obligation on the part of Owner or Engineer to pay or to see to the payment of any money due any such Subcontractor, Supplier, or other individual or entity, except as may otherwise be required by Laws and Regulations.

3.02 Reference Standards

- A. Standards Specifications, Codes, Laws and Regulations
 - Reference in the Contract Documents to standard specifications, manuals, reference standards, or codes of any technical society, organization, or association, or to Laws or Regulations, whether such reference be specific or by implication, means the standard specification, manual, reference standard, code, or Laws or Regulations in effect at the time of opening of Bids (or on the Effective Date of the Contract if there were no Bids), except as may be otherwise specifically stated in the Contract Documents.
 - 2. No provision of any such standard specification, manual, reference standard, or code, and no instruction of a Supplier, will be effective to change the duties or responsibilities of Owner, Contractor, or Engineer from those set forth in the part of the Contract Documents prepared by or for Engineer. No such provision or instruction shall be effective to assign to Owner or Engineer any duty or authority to supervise or direct the performance of the Work, or any duty or authority to undertake responsibility

inconsistent with the provisions of the part of the Contract Documents prepared by or for Engineer.

- 3.03 *Reporting and Resolving Discrepancies*
 - A. Reporting Discrepancies
 - 1. Contractor's Verification of Figures and Field Measurements: Before undertaking each part of the Work, Contractor shall carefully study the Contract Documents, and check and verify pertinent figures and dimensions therein, particularly with respect to applicable field measurements. Contractor shall promptly report in writing to Engineer any conflict, error, ambiguity, or discrepancy that Contractor discovers, or has actual knowledge of, and shall not proceed with any Work affected thereby until the conflict, error, ambiguity, or discrepancy is resolved by a clarification or interpretation by Engineer, or by an amendment or supplement to the Contract issued pursuant to Paragraph 11.01.
 - 2. Contractor's Review of Contract Documents: If, before or during the performance of the Work, Contractor discovers any conflict, error, ambiguity, or discrepancy within the Contract Documents, or between the Contract Documents and (a) any applicable Law or Regulation, (b) actual field conditions, (c) any standard specification, manual, reference standard, or code, or (d) any instruction of any Supplier, then Contractor shall promptly report it to Engineer in writing. Contractor shall not proceed with the Work affected thereby (except in an emergency as required by Paragraph 7.15) until the conflict, error, ambiguity, or discrepancy is resolved, by a clarification or interpretation by Engineer, or by an amendment or supplement to the Contract issued pursuant to Paragraph 11.01.
 - 3. Contractor shall not be liable to Owner or Engineer for failure to report any conflict, error, ambiguity, or discrepancy in the Contract Documents unless Contractor had actual knowledge thereof.
 - B. Resolving Discrepancies
 - 1. Except as may be otherwise specifically stated in the Contract Documents, the provisions of the part of the Contract Documents prepared by or for Engineer take precedence in resolving any conflict, error, ambiguity, or discrepancy between such provisions of the Contract Documents and:
 - a. the provisions of any standard specification, manual, reference standard, or code, or the instruction of any Supplier (whether or not specifically incorporated by reference as a Contract Document); or
 - b. the provisions of any Laws or Regulations applicable to the performance of the Work (unless such an interpretation of the provisions of the Contract Documents would result in violation of such Law or Regulation).

3.04 Requirements of the Contract Documents

A. During the performance of the Work and until final payment, Contractor and Owner shall submit to the Engineer in writing all matters in question concerning the requirements of the Contract Documents (sometimes referred to as requests for information or interpretation— RFIs), or relating to the acceptability of the Work under the Contract Documents, as soon as possible after such matters arise. Engineer will be the initial interpreter of the requirements of the Contract Documents, and judge of the acceptability of the Work.

- B. Engineer will, with reasonable promptness, render a written clarification, interpretation, or decision on the issue submitted, or initiate an amendment or supplement to the Contract Documents. Engineer's written clarification, interpretation, or decision will be final and binding on Contractor, unless it appeals by submitting a Change Proposal, and on Owner, unless it appeals by filing a Claim.
- C. If a submitted matter in question concerns terms and conditions of the Contract Documents that do not involve (1) the performance or acceptability of the Work under the Contract Documents, (2) the design (as set forth in the Drawings, Specifications, or otherwise), or (3) other engineering or technical matters, then Engineer will promptly notify Owner and Contractor in writing that Engineer is unable to provide a decision or interpretation. If Owner and Contractor are unable to agree on resolution of such a matter in question, either party may pursue resolution as provided in Article 12.

3.05 *Reuse of Documents*

- A. Contractor and its Subcontractors and Suppliers shall not:
 - have or acquire any title to or ownership rights in any of the Drawings, Specifications, or other documents (or copies of any thereof) prepared by or bearing the seal of Engineer or its consultants, including electronic media versions, or reuse any such Drawings, Specifications, other documents, or copies thereof on extensions of the Project or any other project without written consent of Owner and Engineer and specific written verification or adaptation by Engineer; or
 - have or acquire any title or ownership rights in any other Contract Documents, reuse any such Contract Documents for any purpose without Owner's express written consent, or violate any copyrights pertaining to such Contract Documents.
- B. The prohibitions of this Paragraph 3.05 will survive final payment, or termination of the Contract. Nothing herein precludes Contractor from retaining copies of the Contract Documents for record purposes.

ARTICLE 4—COMMENCEMENT AND PROGRESS OF THE WORK

4.01 Commencement of Contract Times; Notice to Proceed

- A. The Contract Times will commence to run on the 30th day after the Effective Date of the Contract or, if a Notice to Proceed is given, on the day indicated in the Notice to Proceed. A Notice to Proceed may be given at any time within 30 days after the Effective Date of the Contract. In no event will the Contract Times commence to run later than the 60th day after the day of Bid opening or the 30th day after the Effective Date of the Contract, whichever date is earlier.
- 4.02 Starting the Work
 - A. Contractor shall start to perform the Work on the date when the Contract Times commence to run. No Work may be done at the Site prior to such date.
- 4.03 *Reference Points*
 - A. Owner shall provide engineering surveys to establish reference points for construction which in Engineer's judgment are necessary to enable Contractor to proceed with the Work. Contractor shall be responsible for laying out the Work, shall protect and preserve the

established reference points and property monuments, and shall make no changes or relocations without the prior written approval of Owner. Contractor shall report to Engineer whenever any reference point or property monument is lost or destroyed or requires relocation because of necessary changes in grades or locations, and shall be responsible for the accurate replacement or relocation of such reference points or property monuments by professionally qualified personnel.

4.04 Progress Schedule

- A. Contractor shall adhere to the Progress Schedule established in accordance with Paragraph 2.05 as it may be adjusted from time to time as provided below.
 - 1. Contractor shall submit to Engineer for acceptance (to the extent indicated in Paragraph 2.05) proposed adjustments in the Progress Schedule that will not result in changing the Contract Times.
 - 2. Proposed adjustments in the Progress Schedule that will change the Contract Times must be submitted in accordance with the requirements of Article 11.
- B. Contractor shall carry on the Work and adhere to the Progress Schedule during all disputes or disagreements with Owner. No Work will be delayed or postponed pending resolution of any disputes or disagreements, or during any appeal process, except as permitted by Paragraph 16.04, or as Owner and Contractor may otherwise agree in writing.

4.05 Delays in Contractor's Progress

- A. If Owner, Engineer, or anyone for whom Owner is responsible, delays, disrupts, or interferes with the performance or progress of the Work, then Contractor shall be entitled to an equitable adjustment in Contract Price or Contract Times.
- B. Contractor shall not be entitled to an adjustment in Contract Price or Contract Times for delay, disruption, or interference caused by or within the control of Contractor. Delay, disruption, and interference attributable to and within the control of a Subcontractor or Supplier shall be deemed to be within the control of Contractor.
- C. If Contractor's performance or progress is delayed, disrupted, or interfered with by unanticipated causes not the fault of and beyond the control of Owner, Contractor, and those for which they are responsible, then Contractor shall be entitled to an equitable adjustment in Contract Times. Such an adjustment will be Contractor's sole and exclusive remedy for the delays, disruption, and interference described in this paragraph. Causes of delay, disruption, or interference that may give rise to an adjustment in Contract Times under this paragraph include but are not limited to the following:
 - 1. Severe and unavoidable natural catastrophes such as fires, floods, epidemics, and earthquakes;
 - 2. Abnormal weather conditions;
 - 3. Acts or failures to act of third-party utility owners or other third-party entities (other than those third-party utility owners or other third-party entities performing other work at or adjacent to the Site as arranged by or under contract with Owner, as contemplated in Article 8); and
 - 4. Acts of war or terrorism.

- D. Contractor's entitlement to an adjustment of Contract Times or Contract Price is limited as follows:
 - 1. Contractor's entitlement to an adjustment of the Contract Times is conditioned on the delay, disruption, or interference adversely affecting an activity on the critical path to completion of the Work, as of the time of the delay, disruption, or interference.
 - 2. Contractor shall not be entitled to an adjustment in Contract Price for any delay, disruption, or interference if such delay is concurrent with a delay, disruption, or interference caused by or within the control of Contractor. Such a concurrent delay by Contractor shall not preclude an adjustment of Contract Times to which Contractor is otherwise entitled.
 - 3. Adjustments of Contract Times or Contract Price are subject to the provisions of Article 11.
- E. Each Contractor request or Change Proposal seeking an increase in Contract Times or Contract Price must be supplemented by supporting data that sets forth in detail the following:
 - 1. The circumstances that form the basis for the requested adjustment;
 - 2. The date upon which each cause of delay, disruption, or interference began to affect the progress of the Work;
 - 3. The date upon which each cause of delay, disruption, or interference ceased to affect the progress of the Work;
 - 4. The number of days' increase in Contract Times claimed as a consequence of each such cause of delay, disruption, or interference; and
 - 5. The impact on Contract Price, in accordance with the provisions of Paragraph 11.07.

Contractor shall also furnish such additional supporting documentation as Owner or Engineer may require including, where appropriate, a revised progress schedule indicating all the activities affected by the delay, disruption, or interference, and an explanation of the effect of the delay, disruption, or interference on the critical path to completion of the Work.

- F. Delays, disruption, and interference to the performance or progress of the Work resulting from the existence of a differing subsurface or physical condition, an Underground Facility that was not shown or indicated by the Contract Documents, or not shown or indicated with reasonable accuracy, and those resulting from Hazardous Environmental Conditions, are governed by Article 5, together with the provisions of Paragraphs 4.05.D and 4.05.E.
- G. Paragraph 8.03 addresses delays, disruption, and interference to the performance or progress of the Work resulting from the performance of certain other work at or adjacent to the Site.

ARTICLE 5—SITE; SUBSURFACE AND PHYSICAL CONDITIONS; HAZARDOUS ENVIRONMENTAL CONDITIONS

- 5.01 Availability of Lands
 - A. Owner shall furnish the Site. Owner shall notify Contractor in writing of any encumbrances or restrictions not of general application but specifically related to use of the Site with which Contractor must comply in performing the Work.

- B. Upon reasonable written request, Owner shall furnish Contractor with a current statement of record legal title and legal description of the lands upon which permanent improvements are to be made and Owner's interest therein as necessary for giving notice of or filing a mechanic's or construction lien against such lands in accordance with applicable Laws and Regulations.
- C. Contractor shall provide for all additional lands and access thereto that may be required for temporary construction facilities or storage of materials and equipment.

5.02 Use of Site and Other Areas

A. Limitation on Use of Site and Other Areas

- 1. Contractor shall confine construction equipment, temporary construction facilities, the storage of materials and equipment, and the operations of workers to the Site, adjacent areas that Contractor has arranged to use through construction easements or otherwise, and other adjacent areas permitted by Laws and Regulations, and shall not unreasonably encumber the Site and such other adjacent areas with construction equipment or other materials or equipment. Contractor shall assume full responsibility for (a) damage to the Site; (b) damage to any such other adjacent areas used for Contractor's operations; (c) damage to any other adjacent land or areas, or to improvements, structures, utilities, or similar facilities located at such adjacent lands or areas; and (d) for injuries and losses sustained by the owners or occupants of any such land or areas; provided that such damage or injuries result from the performance of the Work or from other actions or conduct of the Contractor or those for which Contractor is responsible.
- 2. If a damage or injury claim is made by the owner or occupant of any such land or area because of the performance of the Work, or because of other actions or conduct of the Contractor or those for which Contractor is responsible, Contractor shall (a) take immediate corrective or remedial action as required by Paragraph 7.13, or otherwise; (b) promptly attempt to settle the claim as to all parties through negotiations with such owner or occupant, or otherwise resolve the claim by arbitration or other dispute resolution proceeding, or in a court of competent jurisdiction; and (c) to the fullest extent permitted by Laws and Regulations, indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from and against any such claim, and against all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to any claim or action, legal or equitable, brought by any such owner or occupant against Owner, Engineer, or any other party indemnified hereunder to the extent caused directly or indirectly, in whole or in part by, or based upon, Contractor's performance of the Work, or because of other actions or conduct of the Contractor or those for which Contractor is responsible.
- B. Removal of Debris During Performance of the Work: During the progress of the Work the Contractor shall keep the Site and other adjacent areas free from accumulations of waste materials, rubbish, and other debris. Removal and disposal of such waste materials, rubbish, and other debris will conform to applicable Laws and Regulations.
- C. *Cleaning*: Prior to Substantial Completion of the Work Contractor shall clean the Site and the Work and make it ready for utilization by Owner. At the completion of the Work Contractor shall remove from the Site and adjacent areas all tools, appliances, construction equipment

and machinery, and surplus materials and shall restore to original condition all property not designated for alteration by the Contract Documents.

- D. Loading of Structures: Contractor shall not load nor permit any part of any structure to be loaded in any manner that will endanger the structure, nor shall Contractor subject any part of the Work or adjacent structures or land to stresses or pressures that will endanger them.
- 5.03 Subsurface and Physical Conditions
 - A. *Reports and Drawings*: The Supplementary Conditions identify:
 - 1. Those reports of explorations and tests of subsurface conditions at or adjacent to the Site that contain Technical Data;
 - 2. Those drawings of existing physical conditions at or adjacent to the Site, including those drawings depicting existing surface or subsurface structures at or adjacent to the Site (except Underground Facilities), that contain Technical Data; and
 - 3. Technical Data contained in such reports and drawings.
 - B. Underground Facilities: Underground Facilities are shown or indicated on the Drawings, pursuant to Paragraph 5.05, and not in the drawings referred to in Paragraph 5.03.A. Information and data regarding the presence or location of Underground Facilities are not intended to be categorized, identified, or defined as Technical Data.
 - C. *Reliance by Contractor on Technical Data*: Contractor may rely upon the accuracy of the Technical Data expressly identified in the Supplementary Conditions with respect to such reports and drawings, but such reports and drawings are not Contract Documents. If no such express identification has been made, then Contractor may rely upon the accuracy of the Technical Data as defined in Paragraph 1.01.A.46.b.
 - D. Limitations of Other Data and Documents: Except for such reliance on Technical Data, Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, with respect to:
 - the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences, and procedures of construction to be employed by Contractor, and safety precautions and programs incident thereto;
 - 2. other data, interpretations, opinions, and information contained in such reports or shown or indicated in such drawings;
 - 3. the contents of other Site-related documents made available to Contractor, such as record drawings from other projects at or adjacent to the Site, or Owner's archival documents concerning the Site; or
 - 4. any Contractor interpretation of or conclusion drawn from any Technical Data or any such other data, interpretations, opinions, or information.

5.04 Differing Subsurface or Physical Conditions

- A. *Notice by Contractor*: If Contractor believes that any subsurface or physical condition that is uncovered or revealed at the Site:
 - 1. is of such a nature as to establish that any Technical Data on which Contractor is entitled to rely as provided in Paragraph 5.03 is materially inaccurate;
 - is of such a nature as to require a change in the Drawings or Specifications;
 - 3. differs materially from that shown or indicated in the Contract Documents; or
 - is of an unusual nature, and differs materially from conditions ordinarily encountered and generally recognized as inherent in work of the character provided for in the Contract Documents;

then Contractor shall, promptly after becoming aware thereof and before further disturbing the subsurface or physical conditions or performing any Work in connection therewith (except in an emergency as required by Paragraph 7.15), notify Owner and Engineer in writing about such condition. Contractor shall not further disturb such condition or perform any Work in connection therewith (except with respect to an emergency) until receipt of a written statement permitting Contractor to do so.

- B. Engineer's Review: After receipt of written notice as required by the preceding paragraph, Engineer will promptly review the subsurface or physical condition in question; determine whether it is necessary for Owner to obtain additional exploration or tests with respect to the condition; conclude whether the condition falls within any one or more of the differing site condition categories in Paragraph 5.04.A; obtain any pertinent cost or schedule information from Contractor; prepare recommendations to Owner regarding the Contractor's resumption of Work in connection with the subsurface or physical condition in question and the need for any change in the Drawings or Specifications; and advise Owner in writing of Engineer's findings, conclusions, and recommendations.
- C. Owner's Statement to Contractor Regarding Site Condition: After receipt of Engineer's written findings, conclusions, and recommendations, Owner shall issue a written statement to Contractor (with a copy to Engineer) regarding the subsurface or physical condition in question, addressing the resumption of Work in connection with such condition, indicating whether any change in the Drawings or Specifications will be made, and adopting or rejecting Engineer's written findings, conclusions, and recommendations, in whole or in part.
- D. *Early Resumption of Work*: If at any time Engineer determines that Work in connection with the subsurface or physical condition in question may resume prior to completion of Engineer's review or Owner's issuance of its statement to Contractor, because the condition in question has been adequately documented, and analyzed on a preliminary basis, then the Engineer may at its discretion instruct Contractor to resume such Work.
- E. Possible Price and Times Adjustments
 - 1. Contractor shall be entitled to an equitable adjustment in Contract Price or Contract Times, to the extent that the existence of a differing subsurface or physical condition, or any related delay, disruption, or interference, causes an increase or decrease in

Contractor's cost of, or time required for, performance of the Work; subject, however, to the following:

- a. Such condition must fall within any one or more of the categories described in Paragraph 5.04.A;
- b. With respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraph 13.03; and,
- c. Contractor's entitlement to an adjustment of the Contract Times is subject to the provisions of Paragraphs 4.05.D and 4.05.E.
- 2. Contractor shall not be entitled to any adjustment in the Contract Price or Contract Times with respect to a subsurface or physical condition if:
 - a. Contractor knew of the existence of such condition at the time Contractor made a commitment to Owner with respect to Contract Price and Contract Times by the submission of a Bid or becoming bound under a negotiated contract, or otherwise;
 - b. The existence of such condition reasonably could have been discovered or revealed as a result of any examination, investigation, exploration, test, or study of the Site and contiguous areas expressly required by the Bidding Requirements or Contract Documents to be conducted by or for Contractor prior to Contractor's making such commitment; or
 - c. Contractor failed to give the written notice required by Paragraph 5.04.A.
- 3. If Owner and Contractor agree regarding Contractor's entitlement to and the amount or extent of any adjustment in the Contract Price or Contract Times, then any such adjustment will be set forth in a Change Order.
- 4. Contractor may submit a Change Proposal regarding its entitlement to or the amount or extent of any adjustment in the Contract Price or Contract Times, no later than 30 days after Owner's issuance of the Owner's written statement to Contractor regarding the subsurface or physical condition in question.
- F. Underground Facilities; Hazardous Environmental Conditions: Paragraph 5.05 governs rights and responsibilities regarding the presence or location of Underground Facilities. Paragraph 5.06 governs rights and responsibilities regarding Hazardous Environmental Conditions. The provisions of Paragraphs 5.03 and 5.04 are not applicable to the presence or location of Underground Facilities, or to Hazardous Environmental Conditions.

5.05 Underground Facilities

- A. Contractor's Responsibilities: Unless it is otherwise expressly provided in the Supplementary Conditions, the cost of all of the following are included in the Contract Price, and Contractor shall have full responsibility for:
 - 1. reviewing and checking all information and data regarding existing Underground Facilities at the Site;
 - complying with applicable state and local utility damage prevention Laws and Regulations;

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 10/6/2022

- 3. verifying the actual location of those Underground Facilities shown or indicated in the Contract Documents as being within the area affected by the Work, by exposing such Underground Facilities during the course of construction;
- 4. coordination of the Work with the owners (including Owner) of such Underground Facilities, during construction; and
- 5. the safety and protection of all existing Underground Facilities at the Site, and repairing any damage thereto resulting from the Work.
- B. Notice by Contractor: If Contractor believes that an Underground Facility that is uncovered or revealed at the Site was not shown or indicated on the Drawings, or was not shown or indicated on the Drawings with reasonable accuracy, then Contractor shall, promptly after becoming aware thereof and before further disturbing conditions affected thereby or performing any Work in connection therewith (except in an emergency as required by Paragraph 7.15), notify Owner and Engineer in writing regarding such Underground Facility.
- C. Engineer's Review: Engineer will:
 - 1. promptly review the Underground Facility and conclude whether such Underground Facility was not shown or indicated on the Drawings, or was not shown or indicated with reasonable accuracy;
 - 2. identify and communicate with the owner of the Underground Facility; prepare recommendations to Owner (and if necessary issue any preliminary instructions to Contractor) regarding the Contractor's resumption of Work in connection with the Underground Facility in question;
 - obtain any pertinent cost or schedule information from Contractor; determine the extent, if any, to which a change is required in the Drawings or Specifications to reflect and document the consequences of the existence or location of the Underground Facility; and
 - 4. advise Owner in writing of Engineer's findings, conclusions, and recommendations.

During such time, Contractor shall be responsible for the safety and protection of such Underground Facility.

- D. Owner's Statement to Contractor Regarding Underground Facility: After receipt of Engineer's written findings, conclusions, and recommendations, Owner shall issue a written statement to Contractor (with a copy to Engineer) regarding the Underground Facility in question addressing the resumption of Work in connection with such Underground Facility, indicating whether any change in the Drawings or Specifications will be made, and adopting or rejecting Engineer's written findings, conclusions, and recommendations in whole or in part.
- E. Early Resumption of Work: If at any time Engineer determines that Work in connection with the Underground Facility may resume prior to completion of Engineer's review or Owner's issuance of its statement to Contractor, because the Underground Facility in question and conditions affected by its presence have been adequately documented, and analyzed on a preliminary basis, then the Engineer may at its discretion instruct Contractor to resume such Work.
- F. Possible Price and Times Adjustments
 - 1. Contractor shall be entitled to an equitable adjustment in the Contract Price or Contract Times, to the extent that any existing Underground Facility at the Site that was not shown

EJCDC® C-700, Standard General Conditions of the Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

or indicated on the Drawings, or was not shown or indicated with reasonable accuracy, or any related delay, disruption, or interference, causes an increase or decrease in Contractor's cost of, or time required for, performance of the Work; subject, however, to the following:

- a. With respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraph 13.03;
- b. Contractor's entitlement to an adjustment of the Contract Times is subject to the provisions of Paragraphs 4.05.D and 4.05.E; and
- c. Contractor gave the notice required in Paragraph 5.05.B.
- 2. If Owner and Contractor agree regarding Contractor's entitlement to and the amount or extent of any adjustment in the Contract Price or Contract Times, then any such adjustment will be set forth in a Change Order.
- 3. Contractor may submit a Change Proposal regarding its entitlement to or the amount or extent of any adjustment in the Contract Price or Contract Times, no later than 30 days after Owner's issuance of the Owner's written statement to Contractor regarding the Underground Facility in question.
- 4. The information and data shown or indicated on the Drawings with respect to existing Underground Facilities at the Site is based on information and data (a) furnished by the owners of such Underground Facilities, or by others, (b) obtained from available records, or (c) gathered in an investigation conducted in accordance with the current edition of ASCE 38, Standard Guideline for the Collection and Depiction of Existing Subsurface Utility Data, by the American Society of Civil Engineers. If such information or data is incorrect or incomplete, Contractor's remedies are limited to those set forth in this Paragraph 5.05.F.

5.06 Hazardous Environmental Conditions at Site

- A. *Reports and Drawings*: The Supplementary Conditions identify:
 - 1. those reports known to Owner relating to Hazardous Environmental Conditions that have been identified at or adjacent to the Site;
 - 2. drawings known to Owner relating to Hazardous Environmental Conditions that have been identified at or adjacent to the Site; and
 - 3. Technical Data contained in such reports and drawings.
- B. Reliance by Contractor on Technical Data Authorized: Contractor may rely upon the accuracy of the Technical Data expressly identified in the Supplementary Conditions with respect to such reports and drawings, but such reports and drawings are not Contract Documents. If no such express identification has been made, then Contractor may rely on the accuracy of the Technical Data as defined in Paragraph 1.01.A.46.b. Except for such reliance on Technical Data, Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, with respect to:
 - 1. the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences and procedures

of construction to be employed by Contractor, and safety precautions and programs incident thereto;

- 2. other data, interpretations, opinions, and information contained in such reports or shown or indicated in such drawings; or
- 3. any Contractor interpretation of or conclusion drawn from any Technical Data or any such other data, interpretations, opinions or information.
- C. Contractor shall not be responsible for removing or remediating any Hazardous Environmental Condition encountered, uncovered, or revealed at the Site unless such removal or remediation is expressly identified in the Contract Documents to be within the scope of the Work.
- D. Contractor shall be responsible for controlling, containing, and duly removing all Constituents of Concern brought to the Site by Contractor, Subcontractors, Suppliers, or anyone else for whom Contractor is responsible, and for any associated costs; and for the costs of removing and remediating any Hazardous Environmental Condition created by the presence of any such Constituents of Concern.
- E. If Contractor encounters, uncovers, or reveals a Hazardous Environmental Condition whose removal or remediation is not expressly identified in the Contract Documents as being within the scope of the Work, or if Contractor or anyone for whom Contractor is responsible creates a Hazardous Environmental Condition, then Contractor shall immediately: (1) secure or otherwise isolate such condition; (2) stop all Work in connection with such condition and in any area affected thereby (except in an emergency as required by Paragraph 7.15); and (3) notify Owner and Engineer (and promptly thereafter confirm such notice in writing). Owner shall promptly consult with Engineer concerning the necessity for Owner to retain a qualified expert to evaluate such condition or take corrective action, if any. Promptly after consulting with Engineer, Owner shall take such actions as are necessary to permit Owner to timely obtain required permits and provide Contractor the written notice required by Paragraph 5.06.F. If Contractor or anyone for whom Contractor is responsible created the Hazardous Environmental Condition, and impose a set-off against payments to account for the associated costs.
- F. Contractor shall not resume Work in connection with such Hazardous Environmental Condition or in any affected area until after Owner has obtained any required permits related thereto, and delivered written notice to Contractor either (1) specifying that such condition and any affected area is or has been rendered safe for the resumption of Work, or (2) specifying any special conditions under which such Work may be resumed safely.
- G. If Owner and Contractor cannot agree as to entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times, as a result of such Work stoppage, such special conditions under which Work is agreed to be resumed by Contractor, or any costs or expenses incurred in response to the Hazardous Environmental Condition, then within 30 days of Owner's written notice regarding the resumption of Work, Contractor may submit a Change Proposal, or Owner may impose a set-off. Entitlement to any such adjustment is subject to the provisions of Paragraphs 4.05.D, 4.05.E, 11.07, and 11.08.
- H. If, after receipt of such written notice, Contractor does not agree to resume such Work based on a reasonable belief it is unsafe, or does not agree to resume such Work under such special

conditions, then Owner may order the portion of the Work that is in the area affected by such condition to be deleted from the Work, following the contractual change procedures in Article 11. Owner may have such deleted portion of the Work performed by Owner's own forces or others in accordance with Article 8.

- I. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, Subcontractors, and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court, arbitration, or other dispute resolution costs) arising out of or relating to a Hazardous Environmental Condition, provided that such Hazardous Environmental Condition (1) was not shown or indicated in the Drawings, Specifications, or other Contract Documents, identified as Technical Data entitled to limited reliance pursuant to Paragraph 5.06.B, or identified in the Contract Documents to be included within the scope of the Work, and (2) was not created by Contractor or by anyone for whom Contractor is responsible. Nothing in this Paragraph 5.06.I obligates Owner to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.
- J. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to the failure to control, contain, or remove a Constituent of Concern brought to the Site by Contractor or by anyone for whom Contractor is responsible, or to a Hazardous Environmental Condition created by Contractor or by anyone for whom Contractor is responsible. Nothing in this Paragraph 5.06.J obligates Contractor to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.
- K. The provisions of Paragraphs 5.03, 5.04, and 5.05 do not apply to the presence of Constituents of Concern or to a Hazardous Environmental Condition uncovered or revealed at the Site.

ARTICLE 6-BONDS AND INSURANCE

6.01 Performance, Payment, and Other Bonds

- A. Contractor shall furnish a performance bond and a payment bond, each in an amount at least equal to the Contract Price, as security for the faithful performance and payment of Contractor's obligations under the Contract. These bonds must remain in effect until one year after the date when final payment becomes due or until completion of the correction period specified in Paragraph 15.08, whichever is later, except as provided otherwise by Laws or Regulations, the terms of a prescribed bond form, the Supplementary Conditions, or other provisions of the Contract.
- B. Contractor shall also furnish such other bonds (if any) as are required by the Supplementary Conditions or other provisions of the Contract.
- C. All bonds must be in the form included in the Bidding Documents or otherwise specified by Owner prior to execution of the Contract, except as provided otherwise by Laws or

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

EJCDC® C-700, Standard General Conditions of the Construction Contract.

Regulations, and must be issued and signed by a surety named in "Companies Holding Certificates of Authority as Acceptable Sureties on Federal Bonds and as Acceptable Reinsuring Companies" as published in Department Circular 570 (as amended and supplemented) by the Bureau of the Fiscal Service, U.S. Department of the Treasury. A bond signed by an agent or attorney-in-fact must be accompanied by a certified copy of that individual's authority to bind the surety. The evidence of authority must show that it is effective on the date the agent or attorney-in-fact signed the accompanying bond.

- D. Contractor shall obtain the required bonds from surety companies that are duly licensed or authorized, in the state or jurisdiction in which the Project is located, to issue bonds in the required amounts.
- E. If the surety on a bond furnished by Contractor is declared bankrupt or becomes insolvent, or the surety ceases to meet the requirements above, then Contractor shall promptly notify Owner and Engineer in writing and shall, within 20 days after the event giving rise to such notification, provide another bond and surety, both of which must comply with the bond and surety requirements above.
- F. If Contractor has failed to obtain a required bond, Owner may exclude the Contractor from the Site and exercise Owner's termination rights under Article 16.
- G. Upon request to Owner from any Subcontractor, Supplier, or other person or entity claiming to have furnished labor, services, materials, or equipment used in the performance of the Work, Owner shall provide a copy of the payment bond to such person or entity.
- H. Upon request to Contractor from any Subcontractor, Supplier, or other person or entity claiming to have furnished labor, services, materials, or equipment used in the performance of the Work, Contractor shall provide a copy of the payment bond to such person or entity.
- 6.02 Insurance—General Provisions
 - A. Owner and Contractor shall obtain and maintain insurance as required in this article and in the Supplementary Conditions.
 - B. All insurance required by the Contract to be purchased and maintained by Owner or Contractor shall be obtained from insurance companies that are duly licensed or authorized in the state or jurisdiction in which the Project is located to issue insurance policies for the required limits and coverages. Unless a different standard is indicated in the Supplementary Conditions, all companies that provide insurance policies required under this Contract shall have an A.M. Best rating of A-VII or better.
 - C. Alternative forms of insurance coverage, including but not limited to self-insurance and "Occupational Accident and Excess Employer's Indemnity Policies," are not sufficient to meet the insurance requirements of this Contract, unless expressly allowed in the Supplementary Conditions.
 - D. Contractor shall deliver to Owner, with copies to each additional insured identified in the Contract, certificates of insurance and endorsements establishing that Contractor has obtained and is maintaining the policies and coverages required by the Contract. Upon request by Owner or any other insured, Contractor shall also furnish other evidence of such required insurance, including but not limited to copies of policies, documentation of applicable self-insured retentions (if allowed) and deductibles, full disclosure of all relevant exclusions, and evidence of insurance required to be purchased and maintained by

Subcontractors or Suppliers. In any documentation furnished under this provision, Contractor, Subcontractors, and Suppliers may block out (redact) (1) any confidential premium or pricing information and (2) any wording specific to a project or jurisdiction other than those applicable to this Contract.

- E. Owner shall deliver to Contractor, with copies to each additional insured identified in the Contract, certificates of insurance and endorsements establishing that Owner has obtained and is maintaining the policies and coverages required of Owner by the Contract (if any). Upon request by Contractor or any other insured, Owner shall also provide other evidence of such required insurance (if any), including but not limited to copies of policies, documentation of applicable self-insured retentions (if allowed) and deductibles, and full disclosure of all relevant exclusions. In any documentation furnished under this provision, Owner may block out (redact) (1) any confidential premium or pricing information and (2) any wording specific to a project or jurisdiction other than those relevant to this Contract.
- F. Failure of Owner or Contractor to demand such certificates or other evidence of the other party's full compliance with these insurance requirements, or failure of Owner or Contractor to identify a deficiency in compliance from the evidence provided, will not be construed as a waiver of the other party's obligation to obtain and maintain such insurance.
- G. In addition to the liability insurance required to be provided by Contractor, the Owner, at Owner's option, may purchase and maintain Owner's own liability insurance. Owner's liability policies, if any, operate separately and independently from policies required to be provided by Contractor, and Contractor cannot rely upon Owner's liability policies for any of Contractor's obligations to the Owner, Engineer, or third parties.
- H. Contractor shall require:
 - 1. Subcontractors to purchase and maintain worker's compensation, commercial general liability, and other insurance that is appropriate for their participation in the Project, and to name as additional insureds Owner and Engineer (and any other individuals or entities identified in the Supplementary Conditions as additional insureds on Contractor's liability policies) on each Subcontractor's commercial general liability insurance policy; and
 - 2. Suppliers to purchase and maintain insurance that is appropriate for their participation in the Project.
- I. If either party does not purchase or maintain the insurance required of such party by the Contract, such party shall notify the other party in writing of such failure to purchase prior to the start of the Work, or of such failure to maintain prior to any change in the required coverage.
- J. If Contractor has failed to obtain and maintain required insurance, Contractor's entitlement to enter or remain at the Site will end immediately, and Owner may impose an appropriate set-off against payment for any associated costs (including but not limited to the cost of purchasing necessary insurance coverage), and exercise Owner's termination rights under Article 16.
- K. Without prejudice to any other right or remedy, if a party has failed to obtain required insurance, the other party may elect (but is in no way obligated) to obtain equivalent insurance to protect such other party's interests at the expense of the party who was required to provide such coverage, and the Contract Price will be adjusted accordingly.

- L. Owner does not represent that insurance coverage and limits established in this Contract necessarily will be adequate to protect Contractor or Contractor's interests. Contractor is responsible for determining whether such coverage and limits are adequate to protect its interests, and for obtaining and maintaining any additional insurance that Contractor deems necessary.
- M. The insurance and insurance limits required herein will not be deemed as a limitation on Contractor's liability, or that of its Subcontractors or Suppliers, under the indemnities granted to Owner and other individuals and entities in the Contract or otherwise.
- N. All the policies of insurance required to be purchased and maintained under this Contract will contain a provision or endorsement that the coverage afforded will not be canceled, or renewal refused, until at least 10 days prior written notice has been given to the purchasing policyholder. Within three days of receipt of any such written notice, the purchasing policyholder shall provide a copy of the notice to each other insured and Engineer.

6.03 Contractor's Insurance

- A. *Required Insurance*: Contractor shall purchase and maintain Worker's Compensation, Commercial General Liability, and other insurance pursuant to the specific requirements of the Supplementary Conditions.
- B. *General Provisions*: The policies of insurance required by this Paragraph 6.03 as supplemented must:
 - 1. include at least the specific coverages required;
 - 2. be written for not less than the limits provided, or those required by Laws or Regulations, whichever is greater;
 - 3. remain in effect at least until the Work is complete (as set forth in Paragraph 15.06.D), and longer if expressly required elsewhere in this Contract, and at all times thereafter when Contractor may be correcting, removing, or replacing defective Work as a warranty or correction obligation, or otherwise, or returning to the Site to conduct other tasks arising from the Contract;
 - 4. apply with respect to the performance of the Work, whether such performance is by Contractor, any Subcontractor or Supplier, or by anyone directly or indirectly employed by any of them to perform any of the Work, or by anyone for whose acts any of them may be liable; and
 - 5. include all necessary endorsements to support the stated requirements.
- C. Additional Insureds: The Contractor's commercial general liability, automobile liability, employer's liability, umbrella or excess, pollution liability, and unmanned aerial vehicle liability policies, if required by this Contract, must:
 - 1. include and list as additional insureds Owner and Engineer, and any individuals or entities identified as additional insureds in the Supplementary Conditions;
 - 2. include coverage for the respective officers, directors, members, partners, employees, and consultants of all such additional insureds;
 - 3. afford primary coverage to these additional insureds for all claims covered thereby (including as applicable those arising from both ongoing and completed operations);

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

EJCDC[®] C-700, Standard General Conditions of the Construction Contract.

- 4. not seek contribution from insurance maintained by the additional insured; and
- as to commercial general liability insurance, apply to additional insureds with respect to liability caused in whole or in part by Contractor's acts or omissions, or the acts and omissions of those working on Contractor's behalf, in the performance of Contractor's operations.

6.04 Builder's Risk and Other Property Insurance

- A. Builder's Risk: Unless otherwise provided in the Supplementary Conditions, Contractor shall purchase and maintain builder's risk insurance upon the Work on a completed value basis, in the amount of the Work's full insurable replacement cost (subject to such deductible amounts as may be provided in the Supplementary Conditions or required by Laws and Regulations). The specific requirements applicable to the builder's risk insurance are set forth in the Supplementary Conditions.
- B. Property Insurance for Facilities of Owner Where Work Will Occur: Owner is responsible for obtaining and maintaining property insurance covering each existing structure, building, or facility in which any part of the Work will occur, or to which any part of the Work will attach or be adjoined. Such property insurance will be written on a special perils (all-risk) form, on a replacement cost basis, providing coverage consistent with that required for the builder's risk insurance, and will be maintained until the Work is complete, as set forth in Paragraph 15.06.D.
- C. Property Insurance for Substantially Complete Facilities: Promptly after Substantial Completion, and before actual occupancy or use of the substantially completed Work, Owner will obtain property insurance for such substantially completed Work, and maintain such property insurance at least until the Work is complete, as set forth in Paragraph 15.06.D. Such property insurance will be written on a special perils (all-risk) form, on a replacement cost basis, and provide coverage consistent with that required for the builder's risk insurance. The builder's risk insurance may terminate upon written confirmation of Owner's procurement of such property insurance.
- D. Partial Occupancy or Use by Owner: If Owner will occupy or use a portion or portions of the Work prior to Substantial Completion of all the Work, as provided in Paragraph 15.04, then Owner (directly, if it is the purchaser of the builder's risk policy, or through Contractor) will provide advance notice of such occupancy or use to the builder's risk insurer, and obtain an endorsement consenting to the continuation of coverage prior to commencing such partial occupancy or use.
- E. Insurance of Other Property; Additional Insurance: If the express insurance provisions of the Contract do not require or address the insurance of a property item or interest, then the entity or individual owning such property item will be responsible for insuring it. If Contractor elects to obtain other special insurance to be included in or supplement the builder's risk or property insurance policies provided under this Paragraph 6.04, it may do so at Contractor's expense.

6.05 Property Losses; Subrogation

A. The builder's risk insurance policy purchased and maintained in accordance with Paragraph 6.04 (or an installation floater policy if authorized by the Supplementary Conditions), will contain provisions to the effect that in the event of payment of any loss or damage the insurer will have no rights of recovery against any insureds thereunder, or against

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 10/6/2022

Engineer or its consultants, or their officers, directors, members, partners, employees, agents, consultants, or subcontractors.

- 1. Owner and Contractor waive all rights against each other and the respective officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, for all losses and damages caused by, arising out of, or resulting from any of the perils, risks, or causes of loss covered by such policies and any other property insurance applicable to the Work; and, in addition, waive all such rights against Engineer, its consultants, all individuals or entities identified in the Supplementary Conditions as builder's risk or installation floater insureds, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, under such policies for losses and damages so caused.
- 2. None of the above waivers extends to the rights that any party making such waiver may have to the proceeds of insurance held by Owner or Contractor as trustee or fiduciary, or otherwise payable under any policy so issued.
- B. Any property insurance policy maintained by Owner covering any loss, damage, or consequential loss to Owner's existing structures, buildings, or facilities in which any part of the Work will occur, or to which any part of the Work will attach or adjoin; to adjacent structures, buildings, or facilities of Owner; or to part or all of the completed or substantially completed Work, during partial occupancy or use pursuant to Paragraph 15.04, after Substantial Completion pursuant to Paragraph 15.03, or after final payment pursuant to Paragraph 15.06, will contain provisions to the effect that in the event of payment of any loss or damage the insurer will have no rights of recovery against any insureds thereunder, or against Contractor, Subcontractors, or Engineer, or the officers, directors, members, partners, employees, agents, consultants, or subcontractors of each and any of them, and that the insured is allowed to waive the insurer's rights of subrogation in a written contract executed prior to the loss, damage, or consequential loss.
 - 1. Owner waives all rights against Contractor, Subcontractors, and Engineer, and the officers, directors, members, partners, employees; agents, consultants and subcontractors of each and any of them, for all losses and damages caused by, arising out of, or resulting from fire or any of the perils, risks, or causes of loss covered by such policies.
- C. The waivers in this Paragraph 6.05 include the waiver of rights due to business interruption, loss of use, or other consequential loss extending beyond direct physical loss or damage to Owner's property or the Work caused by, arising out of, or resulting from fire or other insured peril, risk, or cause of loss.
- D. Contractor shall be responsible for assuring that each Subcontract contains provisions whereby the Subcontractor waives all rights against Owner, Contractor, all individuals or entities identified in the Supplementary Conditions as insureds, the Engineer and its consultants, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, for all losses and damages caused by, arising out of, relating to, or resulting from fire or other peril, risk, or cause of loss covered by builder's risk insurance, installation floater, and any other property insurance applicable to the Work.

6.06 Receipt and Application of Property Insurance Proceeds

- A. Any insured loss under the builder's risk and other policies of property insurance required by Paragraph 6.04 will be adjusted and settled with the named insured that purchased the policy. Such named insured shall act as fiduciary for the other insureds, and give notice to such other insureds that adjustment and settlement of a claim is in progress. Any other insured may state its position regarding a claim for insured loss in writing within 15 days after notice of such claim.
- B. Proceeds for such insured losses may be made payable by the insurer either jointly to multiple insureds, or to the named insured that purchased the policy in its own right and as fiduciary for other insureds, subject to the requirements of any applicable mortgage clause. A named insured receiving insurance proceeds under the builder's risk and other policies of insurance required by Paragraph 6.04 shall maintain such proceeds in a segregated account, and distribute such proceeds in accordance with such agreement as the parties in interest may reach, or as otherwise required under the dispute resolution provisions of this Contract or applicable Laws and Regulations.
- C. If no other special agreement is reached, Contractor shall repair or replace the damaged Work, using allocated insurance proceeds.

ARTICLE 7—CONTRACTOR'S RESPONSIBILITIES

- 7.01 Contractor's Means and Methods of Construction
 - A. Contractor shall be solely responsible for the means, methods, techniques, sequences, and procedures of construction.
 - B. If the Contract Documents note, or Contractor determines, that professional engineering or other design services are needed to carry out Contractor's responsibilities for construction means, methods, techniques, sequences, and procedures, or for Site safety, then Contractor shall cause such services to be provided by a properly licensed design professional, at Contractor's expense. Such services are not Owner-delegated professional design services under this Contract, and neither Owner nor Engineer has any responsibility with respect to (1) Contractor's determination of the need for such services, (2) the qualifications or licensing of the design professionals retained or employed by Contractor, (3) the performance of such services, or (4) any errors, omissions, or defects in such services.
- 7.02 Supervision and Superintendence
 - A. Contractor shall supervise, inspect, and direct the Work competently and efficiently, devoting such attention thereto and applying such skills and expertise as may be necessary to perform the Work in accordance with the Contract Documents.
 - B. At all times during the progress of the Work, Contractor shall assign a competent resident superintendent who will not be replaced without written notice to Owner and Engineer except under extraordinary circumstances.
- 7.03 Labor; Working Hours
 - A. Contractor shall provide competent, suitably qualified personnel to survey and lay out the Work and perform construction as required by the Contract Documents. Contractor shall maintain good discipline and order at the Site.

- B. Contractor shall be fully responsible to Owner and Engineer for all acts and omissions of Contractor's employees; of Suppliers and Subcontractors, and their employees; and of any other individuals or entities performing or furnishing any of the Work, just as Contractor is responsible for Contractor's own acts and omissions.
- C. Except as otherwise required for the safety or protection of persons or the Work or property at the Site or adjacent thereto, and except as otherwise stated in the Contract Documents, all Work at the Site will be performed during regular working hours, Monday through Friday. Contractor will not perform Work on a Saturday, Sunday, or any legal holiday. Contractor may perform Work outside regular working hours or on Saturdays, Sundays, or legal holidays only with Owner's written consent, which will not be unreasonably withheld.

7.04 Services, Materials, and Equipment

- A. Unless otherwise specified in the Contract Documents, Contractor shall provide and assume full responsibility for all services, materials, equipment, labor, transportation, construction equipment and machinery, tools, appliances, fuel, power, light, heat, telephone, water, sanitary facilities, temporary facilities, and all other facilities and incidentals necessary for the performance, testing, start up, and completion of the Work, whether or not such items are specifically called for in the Contract Documents.
- B. All materials and equipment incorporated into the Work must be new and of good quality, except as otherwise provided in the Contract Documents. All special warranties and guarantees required by the Specifications will expressly run to the benefit of Owner. If required by Engineer, Contractor shall furnish satisfactory evidence (including reports of required tests) as to the source, kind, and quality of materials and equipment.
- C. All materials and equipment must be stored, applied, installed, connected, erected, protected, used, cleaned, and conditioned in accordance with instructions of the applicable Supplier, except as otherwise may be provided in the Contract Documents.

7.05 "Or Equals"

- A. Contractor's Request; Governing Criteria: Whenever an item of equipment or material is specified or described in the Contract Documents by using the names of one or more proprietary items or specific Suppliers, the Contract Price has been based upon Contractor furnishing such item as specified. The specification or description of such an item is intended to establish the type, function, appearance, and quality required. Unless the specification or description contains or is followed by words reading that no like, equivalent, or "or equal" item is permitted, Contractor may request that Engineer authorize the use of other items of equipment or material, or items from other proposed Suppliers, under the circumstances described below.
 - 1. If Engineer in its sole discretion determines that an item of equipment or material proposed by Contractor is functionally equal to that named and sufficiently similar so that no change in related Work will be required, Engineer will deem it an "or equal" item. For the purposes of this paragraph, a proposed item of equipment or material will be considered functionally equal to an item so named if:
 - a. in the exercise of reasonable judgment Engineer determines that the proposed item:
 - 1) is at least equal in materials of construction, quality, durability, appearance, strength, and design characteristics;

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

EJCDC® C-700, Standard General Conditions of the Construction Contract.

- will reliably perform at least equally well the function and achieve the results imposed by the design concept of the completed Project as a functioning whole;
- 3) has a proven record of performance and availability of responsive service; and
- 4) is not objectionable to Owner.
- b. Contractor certifies that, if the proposed item is approved and incorporated into the Work:
 - 1) there will be no increase in cost to the Owner or increase in Contract Times; and
 - 2) the item will conform substantially to the detailed requirements of the item named in the Contract Documents.
- B. *Contractor's Expense*: Contractor shall provide all data in support of any proposed "or equal" item at Contractor's expense.
- C. Engineer's Evaluation and Determination: Engineer will be allowed a reasonable time to evaluate each "or-equal" request. Engineer may require Contractor to furnish additional data about the proposed "or-equal" item. Engineer will be the sole judge of acceptability. No "or-equal" item will be ordered, furnished, installed, or utilized until Engineer's review is complete and Engineer determines that the proposed item is an "or-equal," which will be evidenced by an approved Shop Drawing or other written communication. Engineer will advise Contractor in writing of any negative determination.
- D. *Effect of Engineer's Determination*: Neither approval nor denial of an "or-equal" request will result in any change in Contract Price. The Engineer's denial of an "or-equal" request will be final and binding, and may not be reversed through an appeal under any provision of the Contract.
- E. Treatment as a Substitution Request: If Engineer determines that an item of equipment or material proposed by Contractor does not qualify as an "or-equal" item, Contractor may request that Engineer consider the item a proposed substitute pursuant to Paragraph 7.06.

7.06 Substitutes

- A. Contractor's Request; Governing Criteria: Unless the specification or description of an item of equipment or material required to be furnished under the Contract Documents contains or is followed by words reading that no substitution is permitted, Contractor may request that Engineer authorize the use of other items of equipment or material under the circumstances described below. To the extent possible such requests must be made before commencement of related construction at the Site.
 - Contractor shall submit sufficient information as provided below to allow Engineer to determine if the item of material or equipment proposed is functionally equivalent to that named and an acceptable substitute therefor. Engineer will not accept requests for review of proposed substitute items of equipment or material from anyone other than Contractor.
 - 2. The requirements for review by Engineer will be as set forth in Paragraph 7.06.B, as supplemented by the Specifications, and as Engineer may decide is appropriate under the circumstances.

- 3. Contractor shall make written application to Engineer for review of a proposed substitute item of equipment or material that Contractor seeks to furnish or use. The application:
 - a. will certify that the proposed substitute item will:
 - 1) perform adequately the functions and achieve the results called for by the general design;
 - 2) be similar in substance to the item specified; and
 - 3) be suited to the same use as the item specified.
 - b. will state:
 - 1) the extent, if any, to which the use of the proposed substitute item will necessitate a change in Contract Times;
 - 2) whether use of the proposed substitute item in the Work will require a change in any of the Contract Documents (or in the provisions of any other direct contract with Owner for other work on the Project) to adapt the design to the proposed substitute item; and
 - whether incorporation or use of the proposed substitute item in connection with the Work is subject to payment of any license fee or royalty.
 - c. will identify:
 - 1) all variations of the proposed substitute item from the item specified; and
 - 2) available engineering, sales, maintenance, repair, and replacement services.
 - d. will contain an itemized estimate of all costs or credits that will result directly or indirectly from use of such substitute item, including but not limited to changes in Contract Price, shared savings, costs of redesign, and claims of other contractors affected by any resulting change.
- B. Engineer's Evaluation and Determination: Engineer will be allowed a reasonable time to evaluate each substitute request, and to obtain comments and direction from Owner. Engineer may require Contractor to furnish additional data about the proposed substitute item. Engineer will be the sole judge of acceptability. No substitute will be ordered, furnished, installed, or utilized until Engineer's review is complete and Engineer determines that the proposed item is an acceptable substitute. Engineer's determination will be evidenced by a Field Order or a proposed Change Order accounting for the substitution itself and all related impacts, including changes in Contract Price or Contract Times. Engineer will advise Contractor in writing of any negative determination.
- C. *Special Guarantee*: Owner may require Contractor to furnish at Contractor's expense a special performance guarantee or other surety with respect to any substitute.
- D. Reimbursement of Engineer's Cost: Engineer will record Engineer's costs in evaluating a substitute proposed or submitted by Contractor. Whether or not Engineer approves a substitute so proposed or submitted by Contractor, Contractor shall reimburse Owner for the reasonable charges of Engineer for evaluating each such proposed substitute. Contractor shall also reimburse Owner for the reasonable charges of Engineer for the reasonable charges in the Contract Documents (or in the provisions of any other direct contract with Owner) resulting from the acceptance of each proposed substitute.

- E. *Contractor's Expense*: Contractor shall provide all data in support of any proposed substitute at Contractor's expense.
- F. Effect of Engineer's Determination: If Engineer approves the substitution request, Contractor shall execute the proposed Change Order and proceed with the substitution. The Engineer's denial of a substitution request will be final and binding, and may not be reversed through an appeal under any provision of the Contract. Contractor may challenge the scope of reimbursement costs imposed under Paragraph 7.06.D, by timely submittal of a Change Proposal.
- 7.07 Concerning Subcontractors and Suppliers
 - A. Contractor may retain Subcontractors and Suppliers for the performance of parts of the Work. Such Subcontractors and Suppliers must be acceptable to Owner. The Contractor's retention of a Subcontractor or Supplier for the performance of parts of the Work will not relieve Contractor's obligation to Owner to perform and complete the Work in accordance with the Contract Documents.
 - B. Contractor shall retain specific Subcontractors and Suppliers for the performance of designated parts of the Work if required by the Contract to do so.
 - C. Subsequent to the submittal of Contractor's Bid or final negotiation of the terms of the Contract, Owner may not require Contractor to retain any Subcontractor or Supplier to furnish or perform any of the Work against which Contractor has reasonable objection.
 - D. Prior to entry into any binding subcontract or purchase order, Contractor shall submit to Owner the identity of the proposed Subcontractor or Supplier (unless Owner has already deemed such proposed Subcontractor or Supplier acceptable during the bidding process or otherwise). Such proposed Subcontractor or Supplier shall be deemed acceptable to Owner unless Owner raises a substantive, reasonable objection within 5 days.
 - E. Owner may require the replacement of any Subcontractor or Supplier. Owner also may require Contractor to retain specific replacements; provided, however, that Owner may not require a replacement to which Contractor has a reasonable objection. If Contractor has submitted the identity of certain Subcontractors or Suppliers for acceptance by Owner, and Owner has accepted it (either in writing or by failing to make written objection thereto), then Owner may subsequently revoke the acceptance of any such Subcontractor or Supplier so identified solely on the basis of substantive, reasonable objection after due investigation. Contractor shall submit an acceptable replacement for the rejected Subcontractor or Supplier.
 - F. If Owner requires the replacement of any Subcontractor or Supplier retained by Contractor to perform any part of the Work, then Contractor shall be entitled to an adjustment in Contract Price or Contract Times, with respect to the replacement; and Contractor shall initiate a Change Proposal for such adjustment within 30 days of Owner's requirement of replacement.
 - G. No acceptance by Owner of any such Subcontractor or Supplier, whether initially or as a replacement, will constitute a waiver of the right of Owner to the completion of the Work in accordance with the Contract Documents.

- H. On a monthly basis, Contractor shall submit to Engineer a complete list of all Subcontractors and Suppliers having a direct contract with Contractor, and of all other Subcontractors and Suppliers known to Contractor at the time of submittal.
- I. Contractor shall be solely responsible for scheduling and coordinating the work of Subcontractors and Suppliers.
- J. The divisions and sections of the Specifications and the identifications of any Drawings do not control Contractor in dividing the Work among Subcontractors or Suppliers, or in delineating the Work to be performed by any specific trade.
- K. All Work performed for Contractor by a Subcontractor or Supplier must be pursuant to an appropriate contractual agreement that specifically binds the Subcontractor or Supplier to the applicable terms and conditions of the Contract for the benefit of Owner and Engineer.
- L. Owner may furnish to any Subcontractor or Supplier, to the extent practicable, information about amounts paid to Contractor for Work performed for Contractor by the Subcontractor or Supplier.
- M. Contractor shall restrict all Subcontractors and Suppliers from communicating with Engineer or Owner, except through Contractor or in case of an emergency, or as otherwise expressly allowed in this Contract.

7.08 Patent Fees and Royalties

- A. Contractor shall pay all license fees and royalties and assume all costs incident to the use in the performance of the Work or the incorporation in the Work of any invention, design, process, product, or device which is the subject of patent rights or copyrights held by others. If an invention, design, process, product, or device is specified in the Contract Documents for use in the performance of the Work and if, to the actual knowledge of Owner or Engineer, its use is subject to patent rights or copyrights calling for the payment of any license fee or royalty to others, the existence of such rights will be disclosed in the Contract Documents.
- B. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, and its officers, directors, members, partners, employees, agents, consultants, and subcontractors, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court or arbitration or other dispute resolution costs) arising out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device specified in the Contract Documents, but not identified as being subject to payment of any license fee or royalty to others required by patent rights or copyrights.
- C. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device not specified in the Contract Documents.

7.09 Permits

A. Unless otherwise provided in the Contract Documents, Contractor shall obtain and pay for all construction permits, licenses, and certificates of occupancy. Owner shall assist Contractor, when necessary, in obtaining such permits and licenses. Contractor shall pay all governmental charges and inspection fees necessary for the prosecution of the Work which are applicable at the time of the submission of Contractor's Bid (or when Contractor became bound under a negotiated contract). Owner shall pay all charges of utility owners for connections for providing permanent service to the Work.

7.10 Taxes

A. Contractor shall pay all sales, consumer, use, and other similar taxes required to be paid by Contractor in accordance with the Laws and Regulations of the place of the Project which are applicable during the performance of the Work.

7.11 Laws and Regulations

- A. Contractor shall give all notices required by and shall comply with all Laws and Regulations applicable to the performance of the Work. Neither Owner nor Engineer shall be responsible for monitoring Contractor's compliance with any Laws or Regulations.
- B. If Contractor performs any Work or takes any other action knowing or having reason to know that it is contrary to Laws or Regulations, Contractor shall bear all resulting costs and losses, and shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such Work or other action. It is not Contractor's responsibility to make certain that the Work described in the Contract Documents is in accordance with Laws and Regulations, but this does not relieve Contractor of its obligations under Paragraph 3.03.
- C. Owner or Contractor may give written notice to the other party of any changes after the submission of Contractor's Bid (or after the date when Contractor became bound under a negotiated contract) in Laws or Regulations having an effect on the cost or time of performance of the Work, including but not limited to changes in Laws or Regulations having an effect on procuring permits and on sales, use, value-added, consumption, and other similar taxes. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times resulting from such changes, then within 30 days of such written notice Contractor may submit a Change Proposal, or Owner may initiate a Claim.

7.12 Record Documents

A. Contractor shall maintain in a safe place at the Site one printed record copy of all Drawings, Specifications, Addenda, Change Orders, Work Change Directives, Field Orders, written interpretations and clarifications, and approved Shop Drawings. Contractor shall keep such record documents in good order and annotate them to show changes made during construction. These record documents, together with all approved Samples, will be available to Engineer for reference. Upon completion of the Work, Contractor shall deliver these record documents to Engineer.

7.13 Safety and Protection

- A. Contractor shall be solely responsible for initiating, maintaining, and supervising all safety precautions and programs in connection with the Work. Such responsibility does not relieve Subcontractors of their responsibility for the safety of persons or property in the performance of their work, nor for compliance with applicable safety Laws and Regulations.
- B. Contractor shall designate a qualified and experienced safety representative whose duties and responsibilities are the prevention of Work-related accidents and the maintenance and supervision of safety precautions and programs.
- C. Contractor shall take all necessary precautions for the safety of, and shall provide the necessary protection to prevent damage, injury, or loss to:
 - 1. all persons on the Site or who may be affected by the Work;
 - 2. all the Work and materials and equipment to be incorporated therein, whether in storage on or off the Site; and
 - other property at the Site or adjacent thereto, including trees, shrubs, lawns, walks, pavements, roadways, structures, other work in progress, utilities, and Underground Facilities not designated for removal, relocation, or replacement in the course of construction.
- D. All damage, injury, or loss to any property referred to in Paragraph 7.13.C.2 or 7.13.C.3 caused, directly or indirectly, in whole or in part, by Contractor, any Subcontractor, Supplier, or any other individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable, shall be remedied by Contractor at its expense (except damage or loss attributable to the fault of Drawings or Specifications or to the acts or omissions of Owner or Engineer or anyone employed by any of them, or anyone for whose acts any of them may be liable, and not attributable, directly or indirectly, in whole or in part, to the fault or negligence of Contractor or any Subcontractor, Supplier, or other individual or entity directly or indirectly employed by any of them).
- E. Contractor shall comply with all applicable Laws and Regulations relating to the safety of persons or property, or to the protection of persons or property from damage, injury, or loss; and shall erect and maintain all necessary safeguards for such safety and protection.
- F. Contractor shall notify Owner; the owners of adjacent property; the owners of Underground Facilities and other utilities (if the identity of such owners is known to Contractor); and other contractors and utility owners performing work at or adjacent to the Site, in writing, when Contractor knows that prosecution of the Work may affect them, and shall cooperate with them in the protection, removal, relocation, and replacement of their property or work in progress.
- G. Contractor shall comply with the applicable requirements of Owner's safety programs, if any. Any Owner's safety programs that are applicable to the Work are identified or included in the Supplementary Conditions or Specifications.
- H. Contractor shall inform Owner and Engineer of the specific requirements of Contractor's safety program with which Owner's and Engineer's employees and representatives must comply while at the Site.

- I. Contractor's duties and responsibilities for safety and protection will continue until all the Work is completed, Engineer has issued a written notice to Owner and Contractor in accordance with Paragraph 15.06.C that the Work is acceptable, and Contractor has left the Site (except as otherwise expressly provided in connection with Substantial Completion).
- J. Contractor's duties and responsibilities for safety and protection will resume whenever Contractor or any Subcontractor or Supplier returns to the Site to fulfill warranty or correction obligations, or to conduct other tasks arising from the Contract Documents.

7.14 *Hazard Communication Programs*

A. Contractor shall be responsible for coordinating any exchange of safety data sheets (formerly known as material safety data sheets) or other hazard communication information required to be made available to or exchanged between or among employers at the Site in accordance with Laws or Regulations.

7.15 Emergencies

A. In emergencies affecting the safety or protection of persons or the Work or property at the Site or adjacent thereto, Contractor is obligated to act to prevent damage, injury, or loss. Contractor shall give Engineer prompt written notice if Contractor believes that any significant changes in the Work or variations from the Contract Documents have been caused by an emergency, or are required as a result of Contractor's response to an emergency. If Engineer determines that a change in the Contract Documents is required because of an emergency or Contractor's response, a Work Change Directive or Change Order will be issued.

7.16 Submittals

- A. Shop Drawing and Sample Requirements
 - 1. Before submitting a Shop Drawing or Sample, Contractor shall:
 - a. review and coordinate the Shop Drawing or Sample with other Shop Drawings and Samples and with the requirements of the Work and the Contract Documents;
 - b. determine and verify:
 - all field measurements, quantities, dimensions, specified performance and design criteria, installation requirements, materials, catalog numbers, and similar information with respect to the Submittal;
 - the suitability of all materials and equipment offered with respect to the indicated application, fabrication, shipping, handling, storage, assembly, and installation pertaining to the performance of the Work; and
 - all information relative to Contractor's responsibilities for means, methods, techniques, sequences, and procedures of construction, and safety precautions and programs incident thereto;
 - c. confirm that the Submittal is complete with respect to all related data included in the Submittal.
 - 2. Each Shop Drawing or Sample must bear a stamp or specific written certification that Contractor has satisfied Contractor's obligations under the Contract Documents with respect to Contractor's review of that Submittal, and that Contractor approves the Submittal.

EJCDC® C-700, Standard General Conditions of the Construction Contract.

Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

- 3. With each Shop Drawing or Sample, Contractor shall give Engineer specific written notice of any variations that the Submittal may have from the requirements of the Contract Documents. This notice must be set forth in a written communication separate from the Submittal; and, in addition, in the case of a Shop Drawing by a specific notation made on the Shop Drawing itself.
- B. Submittal Procedures for Shop Drawings and Samples: Contractor shall label and submit Shop Drawings and Samples to Engineer for review and approval in accordance with the accepted Schedule of Submittals.
 - 1. Shop Drawings
 - a. Contractor shall submit the number of copies required in the Specifications.
 - b. Data shown on the Shop Drawings must be complete with respect to quantities, dimensions, specified performance and design criteria, materials, and similar data to show Engineer the services, materials, and equipment Contractor proposes to provide, and to enable Engineer to review the information for the limited purposes required by Paragraph 7.16.C.
 - 2. Samples
 - a. Contractor shall submit the number of Samples required in the Specifications.
 - b. Contractor shall clearly identify each Sample as to material, Supplier, pertinent data such as catalog numbers, the use for which intended and other data as Engineer may require to enable Engineer to review the Submittal for the limited purposes required by Paragraph 7.16.C.
 - 3. Where a Shop Drawing or Sample is required by the Contract Documents or the Schedule of Submittals, any related Work performed prior to Engineer's review and approval of the pertinent submittal will be at the sole expense and responsibility of Contractor.
- C. Engineer's Review of Shop Drawings and Samples
 - Engineer will provide timely review of Shop Drawings and Samples in accordance with the accepted Schedule of Submittals. Engineer's review and approval will be only to determine if the items covered by the Submittals will, after installation or incorporation in the Work, comply with the requirements of the Contract Documents, and be compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents.
 - 2. Engineer's review and approval will not extend to means, methods, techniques, sequences, or procedures of construction, or to safety precautions or programs incident thereto.
 - 3. Engineer's review and approval of a separate item as such will not indicate approval of the assembly in which the item functions.
 - 4. Engineer's review and approval of a Shop Drawing or Sample will not relieve Contractor from responsibility for any variation from the requirements of the Contract Documents unless Contractor has complied with the requirements of Paragraph 7.16.A.3 and Engineer has given written approval of each such variation by specific written notation thereof incorporated in or accompanying the Shop Drawing or Sample. Engineer will

document any such approved variation from the requirements of the Contract Documents in a Field Order or other appropriate Contract modification.

- 5. Engineer's review and approval of a Shop Drawing or Sample will not relieve Contractor from responsibility for complying with the requirements of Paragraphs 7.16.A and B.
- 6. Engineer's review and approval of a Shop Drawing or Sample, or of a variation from the requirements of the Contract Documents, will not, under any circumstances, change the Contract Times or Contract Price, unless such changes are included in a Change Order.
- 7. Neither Engineer's receipt, review, acceptance, or approval of a Shop Drawing or Sample will result in such item becoming a Contract Document.
- 8. Contractor shall perform the Work in compliance with the requirements and commitments set forth in approved Shop Drawings and Samples, subject to the provisions of Paragraph 7.16.C.4.
- D. Resubmittal Procedures for Shop Drawings and Samples
 - 1. Contractor shall make corrections required by Engineer and shall return the required number of corrected copies of Shop Drawings and submit, as required, new Samples for review and approval. Contractor shall direct specific attention in writing to revisions other than the corrections called for by Engineer on previous Submittals.
 - 2. Contractor shall furnish required Shop Drawing and Sample submittals with sufficient information and accuracy to obtain required approval of an item with no more than two resubmittals. Engineer will record Engineer's time for reviewing a third or subsequent resubmittal of a Shop Drawing or Sample, and Contractor shall be responsible for Engineer's charges to Owner for such time. Owner may impose a set-off against payments due Contractor to secure reimbursement for such charges.
 - 3. If Contractor requests a change of a previously approved Shop Drawing or Sample, Contractor shall be responsible for Engineer's charges to Owner for its review time, and Owner may impose a set-off against payments due Contractor to secure reimbursement for such charges, unless the need for such change is beyond the control of Contractor.
- E. Submittals Other than Shop Drawings, Samples, and Owner-Delegated Designs
 - 1. The following provisions apply to all Submittals other than Shop Drawings, Samples, and Owner-delegated designs:
 - a. Contractor shall submit all such Submittals to the Engineer in accordance with the Schedule of Submittals and pursuant to the applicable terms of the Contract Documents.
 - b. Engineer will provide timely review of all such Submittals in accordance with the Schedule of Submittals and return such Submittals with a notation of either Accepted or Not Accepted. Any such Submittal that is not returned within the time established in the Schedule of Submittals will be deemed accepted.
 - c. Engineer's review will be only to determine if the Submittal is acceptable under the requirements of the Contract Documents as to general form and content of the Submittal.

- d. If any such Submittal is not accepted, Contractor shall confer with Engineer regarding the reason for the non-acceptance, and resubmit an acceptable document.
- 2. Procedures for the submittal and acceptance of the Progress Schedule, the Schedule of Submittals, and the Schedule of Values are set forth in Paragraphs 2.03. 2.04, and 2.05.
- F. Owner-delegated Designs: Submittals pursuant to Owner-delegated designs are governed by the provisions of Paragraph 7.19.
- 7.17 Contractor's General Warranty and Guarantee
 - A. Contractor warrants and guarantees to Owner that all Work will be in accordance with the Contract Documents and will not be defective. Engineer is entitled to rely on Contractor's warranty and guarantee.
 - B. Owner's rights under this warranty and guarantee are in addition to, and are not limited by, Owner's rights under the correction period provisions of Paragraph 15.08. The time in which Owner may enforce its warranty and guarantee rights under this Paragraph 7.17 is limited only by applicable Laws and Regulations restricting actions to enforce such rights; provided, however, that after the end of the correction period under Paragraph 15.08:
 - 1. Owner shall give Contractor written notice of any defective Work within 60 days of the discovery that such Work is defective; and
 - 2. Such notice will be deemed the start of an event giving rise to a Claim under Paragraph 12.01.B, such that any related Claim must be brought within 30 days of the notice.
 - C. Contractor's warranty and guarantee hereunder excludes defects or damage caused by:
 - abuse, or improper modification, maintenance, or operation, by persons other than Contractor, Subcontractors, Suppliers, or any other individual or entity for whom Contractor is responsible; or
 - 2. normal wear and tear under normal usage.
 - D. Contractor's obligation to perform and complete the Work in accordance with the Contract Documents is absolute. None of the following will constitute an acceptance of Work that is not in accordance with the Contract Documents, a release of Contractor's obligation to perform the Work in accordance with the Contract Documents, or a release of Owner's warranty and guarantee rights under this Paragraph 7.17:
 - 1. Observations by Engineer;
 - 2. Recommendation by Engineer or payment by Owner of any progress or final payment;
 - 3. The issuance of a certificate of Substantial Completion by Engineer or any payment related thereto by Owner;
 - 4. Use or occupancy of the Work or any part thereof by Owner;
 - 5. Any review and approval of a Shop Drawing or Sample submittal;
 - 6. The issuance of a notice of acceptability by Engineer;
 - 7. The end of the correction period established in Paragraph 15.08;
 - 8. Any inspection, test, or approval by others; or

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

EJCDC® C-700, Standard General Conditions of the Construction Contract.

- 9. Any correction of defective Work by Owner.
- E. If the Contract requires the Contractor to accept the assignment of a contract entered into by Owner, then the specific warranties, guarantees, and correction obligations contained in the assigned contract will govern with respect to Contractor's performance obligations to Owner for the Work described in the assigned contract.
- 7.18 Indemnification
 - A. To the fullest extent permitted by Laws and Regulations, and in addition to any other obligations of Contractor under the Contract or otherwise, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from losses, damages, costs, and judgments (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court or arbitration or other dispute resolution costs) arising from third-party claims or actions relating to or resulting from the performance or furnishing of the Work, provided that any such claim, action, loss, cost, judgment or damage is attributable to bodily injury, sickness, disease, or death, or to damage to or destruction of tangible property (other than the Work itself), including the loss of use resulting therefrom, but only to the extent caused by any negligent act or omission of Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable.
 - B. In any and all claims against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, by any employee (or the survivor or personal representative of such employee) of Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable, the indemnification obligation under Paragraph 7.18.A will not be limited in any way by any limitation on the amount or type of damages, compensation, or benefits payable by or for Contractor or any such Subcontractor, Supplier, or other individual or entity under workers' compensation acts, disability benefit acts, or other employee benefit acts.
- 7.19 Delegation of Professional Design Services
 - A. Owner may require Contractor to provide professional design services for a portion of the Work by express delegation in the Contract Documents. Such delegation will specify the performance and design criteria that such services must satisfy, and the Submittals that Contractor must furnish to Engineer with respect to the Owner-delegated design.
 - B. Contractor shall cause such Owner-delegated professional design services to be provided pursuant to the professional standard of care by a properly licensed design professional, whose signature and seal must appear on all drawings, calculations, specifications, certifications, and Submittals prepared by such design professional. Such design professional must issue all certifications of design required by Laws and Regulations.
 - C. If a Shop Drawing or other Submittal related to the Owner-delegated design is prepared by Contractor, a Subcontractor, or others for submittal to Engineer, then such Shop Drawing or other Submittal must bear the written approval of Contractor's design professional when submitted by Contractor to Engineer.

- D. Owner and Engineer shall be entitled to rely upon the adequacy, accuracy, and completeness of the services, certifications, and approvals performed or provided by the design professionals retained or employed by Contractor under an Owner-delegated design, subject to the professional standard of care and the performance and design criteria stated in the Contract Documents.
- E. Pursuant to this Paragraph 7.19, Engineer's review, approval, and other determinations regarding design drawings, calculations, specifications, certifications, and other Submittals furnished by Contractor pursuant to an Owner-delegated design will be only for the following limited purposes:
 - 1. Checking for conformance with the requirements of this Paragraph 7.19;
 - 2. Confirming that Contractor (through its design professionals) has used the performance and design criteria specified in the Contract Documents; and
 - 3. Establishing that the design furnished by Contractor is consistent with the design concept expressed in the Contract Documents.
- F. Contractor shall not be responsible for the adequacy of performance or design criteria specified by Owner or Engineer.
- G. Contractor is not required to provide professional services in violation of applicable Laws and Regulations.

ARTICLE 8—OTHER WORK AT THE SITE

8.01 Other Work

- A. In addition to and apart from the Work under the Contract Documents, the Owner may perform other work at or adjacent to the Site. Such other work may be performed by Owner's employees, or through contracts between the Owner and third parties. Owner may also arrange to have third-party utility owners perform work on their utilities and facilities at or adjacent to the Site.
- B. If Owner performs other work at or adjacent to the Site with Owner's employees, or through contracts for such other work, then Owner shall give Contractor written notice thereof prior to starting any such other work. If Owner has advance information regarding the start of any third-party utility work that Owner has arranged to take place at or adjacent to the Site, Owner shall provide such information to Contractor.
- C. Contractor shall afford proper and safe access to the Site to each contractor that performs such other work, each utility owner performing other work, and Owner, if Owner is performing other work with Owner's employees, and provide a reasonable opportunity for the introduction and storage of materials and equipment and the execution of such other work.
- D. Contractor shall do all cutting, fitting, and patching of the Work that may be required to properly connect or otherwise make its several parts come together and properly integrate with such other work. Contractor shall not endanger any work of others by cutting, excavating, or otherwise altering such work; provided, however, that Contractor may cut or alter others' work with the written consent of Engineer and the others whose work will be affected.

- E. If the proper execution or results of any part of Contractor's Work depends upon work performed by others, Contractor shall inspect such other work and promptly report to Engineer in writing any delays, defects, or deficiencies in such other work that render it unavailable or unsuitable for the proper execution and results of Contractor's Work. Contractor's failure to so report will constitute an acceptance of such other work as fit and proper for integration with Contractor's Work except for latent defects and deficiencies in such other work.
- F. The provisions of this article are not applicable to work that is performed by third-party utilities or other third-party entities without a contract with Owner, or that is performed without having been arranged by Owner. If such work occurs, then any related delay, disruption, or interference incurred by Contractor is governed by the provisions of Paragraph 4.05.C.3.

8.02 Coordination

- A. If Owner intends to contract with others for the performance of other work at or adjacent to the Site, to perform other work at or adjacent to the Site with Owner's employees, or to arrange to have utility owners perform work at or adjacent to the Site, the following will be set forth in the Supplementary Conditions or provided to Contractor prior to the start of any such other work:
 - 1. The identity of the individual or entity that will have authority and responsibility for coordination of the activities among the various contractors;
 - 2. An itemization of the specific matters to be covered by such authority and responsibility; and
 - 3. The extent of such authority and responsibilities.
- B. Unless otherwise provided in the Supplementary Conditions, Owner shall have sole authority and responsibility for such coordination.

8.03 Legal Relationships

A. If, in the course of performing other work for Owner at or adjacent to the Site, the Owner's employees, any other contractor working for Owner, or any utility owner that Owner has arranged to perform work, causes damage to the Work or to the property of Contractor or its Subcontractors, or delays, disrupts, interferes with, or increases the scope or cost of the performance of the Work, through actions or inaction, then Contractor shall be entitled to an equitable adjustment in the Contract Price or the Contract Times. Contractor must submit any Change Proposal seeking an equitable adjustment in the Contract Price or the Contract Times under this paragraph within 30 days of the damaging, delaying, disrupting, or interfering event. The entitlement to, and extent of, any such equitable adjustment will take into account information (if any) regarding such other work that was provided to Contractor in the Contract Documents prior to the submittal of the Bid or the final negotiation of the terms of the Contract, and any remedies available to Contractor under Laws or Regulations concerning utility action or inaction. When applicable, any such equitable adjustment in Contract Price will be conditioned on Contractor assigning to Owner all Contractor's rights against such other contractor or utility owner with respect to the damage, delay, disruption, or interference that is the subject of the adjustment. Contractor's entitlement to an adjustment of the Contract Times or Contract Price is subject to the provisions of Paragraphs 4.05.D and 4.05.E.

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 10/6/2022

- B. Contractor shall take reasonable and customary measures to avoid damaging, delaying, disrupting, or interfering with the work of Owner, any other contractor, or any utility owner performing other work at or adjacent to the Site.
 - If Contractor fails to take such measures and as a result damages, delays, disrupts, or interferes with the work of any such other contractor or utility owner, then Owner may impose a set-off against payments due Contractor, and assign to such other contractor or utility owner the Owner's contractual rights against Contractor with respect to the breach of the obligations set forth in this Paragraph 8.03.B.
 - 2. When Owner is performing other work at or adjacent to the Site with Owner's employees, Contractor shall be liable to Owner for damage to such other work, and for the reasonable direct delay, disruption, and interference costs incurred by Owner as a result of Contractor's failure to take reasonable and customary measures with respect to Owner's other work. In response to such damage, delay, disruption, or interference, Owner may impose a set-off against payments due Contractor.
- C. If Contractor damages, delays, disrupts, or interferes with the work of any other contractor, or any utility owner performing other work at or adjacent to the Site, through Contractor's failure to take reasonable and customary measures to avoid such impacts, or if any claim arising out of Contractor's actions, inactions, or negligence in performance of the Work at or adjacent to the Site is made by any such other contractor or utility owner against Contractor, Owner, or Engineer, then Contractor shall (1) promptly attempt to settle the claim as to all parties through negotiations with such other contractor or utility owner, or otherwise resolve the claim by arbitration or other dispute resolution proceeding or at law, and (2) indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them from and against any such claims, and against all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such damage, delay, disruption, or interference.

ARTICLE 9—OWNER'S RESPONSIBILITIES

9.01 Communications to Contractor

A. Except as otherwise provided in these General Conditions, Owner shall issue all communications to Contractor through Engineer.

9.02 Replacement of Engineer

A. Owner may at its discretion appoint an engineer to replace Engineer, provided Contractor makes no reasonable objection to the replacement engineer. The replacement engineer's status under the Contract Documents will be that of the former Engineer.

9.03 Furnish Data

- A. Owner shall promptly furnish the data required of Owner under the Contract Documents.
- 9.04 Pay When Due
 - A. Owner shall make payments to Contractor when they are due as provided in the Agreement.

- 9.05 Lands and Easements; Reports, Tests, and Drawings
 - A. Owner's duties with respect to providing lands and easements are set forth in Paragraph 5.01.
 - B. Owner's duties with respect to providing engineering surveys to establish reference points are set forth in Paragraph 4.03.
 - C. Article 5 refers to Owner's identifying and making available to Contractor copies of reports of explorations and tests of conditions at the Site, and drawings of physical conditions relating to existing surface or subsurface structures at the Site.
- 9.06 Insurance
 - A. Owner's responsibilities, if any, with respect to purchasing and maintaining liability and property insurance are set forth in Article 6.

9.07 Change Orders

- A. Owner's responsibilities with respect to Change Orders are set forth in Article 11.
- 9.08 Inspections, Tests, and Approvals
 - A. Owner's responsibility with respect to certain inspections, tests, and approvals is set forth in Paragraph 14.02.B.
- 9.09 Limitations on Owner's Responsibilities
 - A. The Owner shall not supervise, direct, or have control or authority over, nor be responsible for, Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Owner will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
- 9.10 Undisclosed Hazardous Environmental Condition
 - A. Owner's responsibility in respect to an undisclosed Hazardous Environmental Condition is set forth in Paragraph 5.06.
- 9.11 Evidence of Financial Arrangements
 - A. Upon request of Contractor, Owner shall furnish Contractor reasonable evidence that financial arrangements have been made to satisfy Owner's obligations under the Contract (including obligations under proposed changes in the Work).
- 9.12 Safety Programs
 - A. While at the Site, Owner's employees and representatives shall comply with the specific applicable requirements of Contractor's safety programs of which Owner has been informed.
 - B. Owner shall furnish copies of any applicable Owner safety programs to Contractor.

ARTICLE 10—ENGINEER'S STATUS DURING CONSTRUCTION

- 10.01 Owner's Representative
 - A. Engineer will be Owner's representative during the construction period. The duties and responsibilities and the limitations of authority of Engineer as Owner's representative during construction are set forth in the Contract.
- 10.02 Visits to Site
 - A. Engineer will make visits to the Site at intervals appropriate to the various stages of construction as Engineer deems necessary in order to observe, as an experienced and qualified design professional, the progress that has been made and the quality of the various aspects of Contractor's executed Work. Based on information obtained during such visits and observations, Engineer, for the benefit of Owner, will determine, in general, if the Work is proceeding in accordance with the Contract Documents. Engineer will not be required to make exhaustive or continuous inspections on the Site to check the quality or quantity of the Work. Engineer's efforts will be directed toward providing for Owner a greater degree of confidence that the completed Work will conform generally to the Contract Documents. On the basis of such visits and observations, Engineer will keep Owner informed of the progress of the Work and will endeavor to guard Owner against defective Work.
 - B. Engineer's visits and observations are subject to all the limitations on Engineer's authority and responsibility set forth in Paragraph 10.07. Particularly, but without limitation, during or as a result of Engineer's visits or observations of Contractor's Work, Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work.

10.03 Resident Project Representative

- A. If Owner and Engineer have agreed that Engineer will furnish a Resident Project Representative to represent Engineer at the Site and assist Engineer in observing the progress and quality of the Work, then the authority and responsibilities of any such Resident Project Representative will be as provided in the Supplementary Conditions, and limitations on the responsibilities thereof will be as provided in the Supplementary Conditions and in Paragraph 10.07.
- B. If Owner designates an individual or entity who is not Engineer's consultant, agent, or employee to represent Owner at the Site, then the responsibilities and authority of such individual or entity will be as provided in the Supplementary Conditions.

10.04 Engineer's Authority

- A. Engineer has the authority to reject Work in accordance with Article 14.
- B. Engineer's authority as to Submittals is set forth in Paragraph 7.16.
- C. Engineer's authority as to design drawings, calculations, specifications, certifications and other Submittals from Contractor in response to Owner's delegation (if any) to Contractor of professional design services, is set forth in Paragraph 7.19.
- D. Engineer's authority as to changes in the Work is set forth in Article 11.

- E. Engineer's authority as to Applications for Payment is set forth in Article 15.
- 10.05 Determinations for Unit Price Work
 - A. Engineer will determine the actual quantities and classifications of Unit Price Work performed by Contractor as set forth in Paragraph 13.03.
- 10.06 Decisions on Requirements of Contract Documents and Acceptability of Work
 - A. Engineer will render decisions regarding the requirements of the Contract Documents, and judge the acceptability of the Work, pursuant to the specific procedures set forth herein for initial interpretations, Change Proposals, and acceptance of the Work. In rendering such decisions and judgments, Engineer will not show partiality to Owner or Contractor, and will not be liable to Owner, Contractor, or others in connection with any proceedings, interpretations, decisions, or judgments conducted or rendered in good faith.
- 10.07 Limitations on Engineer's Authority and Responsibilities
 - A. Neither Engineer's authority or responsibility under this Article 10 or under any other provision of the Contract, nor any decision made by Engineer in good faith either to exercise or not exercise such authority or responsibility or the undertaking, exercise, or performance of any authority or responsibility by Engineer, will create, impose, or give rise to any duty in contract, tort, or otherwise owed by Engineer to Contractor, any Subcontractor, any Supplier, any other individual or entity, or to any surety for or employee or agent of any of them.
 - B. Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Engineer will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
 - C. Engineer will not be responsible for the acts or omissions of Contractor or of any Subcontractor, any Supplier, or of any other individual or entity performing any of the Work.
 - D. Engineer's review of the final Application for Payment and accompanying documentation, and all maintenance and operating instructions, schedules, guarantees, bonds, certificates of inspection, tests and approvals, and other documentation required to be delivered by Contractor under Paragraph 15.06.A, will only be to determine generally that their content complies with the requirements of, and in the case of certificates of inspections, tests, and approvals, that the results certified indicate compliance with the Contract Documents.
 - E. The limitations upon authority and responsibility set forth in this Paragraph 10.07 also apply to the Resident Project Representative, if any.
- 10.08 Compliance with Safety Program
 - A. While at the Site, Engineer's employees and representatives will comply with the specific applicable requirements of Owner's and Contractor's safety programs of which Engineer has been informed.

ARTICLE 11—CHANGES TO THE CONTRACT

11.01 Amending and Supplementing the Contract

- A. The Contract may be amended or supplemented by a Change Order, a Work Change Directive, or a Field Order.
- B. If an amendment or supplement to the Contract includes a change in the Contract Price or the Contract Times, such amendment or supplement must be set forth in a Change Order.
- C. All changes to the Contract that involve (1) the performance or acceptability of the Work, (2) the design (as set forth in the Drawings, Specifications, or otherwise), or (3) other engineering or technical matters, must be supported by Engineer's recommendation. Owner and Contractor may amend other terms and conditions of the Contract without the recommendation of the Engineer.

11.02 Change Orders

- A. Owner and Contractor shall execute appropriate Change Orders covering:
 - Changes in Contract Price or Contract Times which are agreed to by the parties, including any undisputed sum or amount of time for Work actually performed in accordance with a Work Change Directive;
 - 2. Changes in Contract Price resulting from an Owner set-off, unless Contractor has duly contested such set-off;
 - 3. Changes in the Work which are: (a) ordered by Owner pursuant to Paragraph 11.05, (b) required because of Owner's acceptance of defective Work under Paragraph 14.04 or Owner's correction of defective Work under Paragraph 14.07, or (c) agreed to by the parties, subject to the need for Engineer's recommendation if the change in the Work involves the design (as set forth in the Drawings, Specifications, or otherwise) or other engineering or technical matters; and
 - 4. Changes that embody the substance of any final and binding results under: Paragraph 11.03.B, resolving the impact of a Work Change Directive; Paragraph 11.09, concerning Change Proposals; Article 12, Claims; Paragraph 13.02.D, final adjustments resulting from allowances; Paragraph 13.03.D, final adjustments relating to determination of quantities for Unit Price Work; and similar provisions.
- B. If Owner or Contractor refuses to execute a Change Order that is required to be executed under the terms of Paragraph 11.02.A, it will be deemed to be of full force and effect, as if fully executed.

11.03 Work Change Directives

A. A Work Change Directive will not change the Contract Price or the Contract Times but is evidence that the parties expect that the modification ordered or documented by a Work Change Directive will be incorporated in a subsequently issued Change Order, following negotiations by the parties as to the Work Change Directive's effect, if any, on the Contract Price and Contract Times; or, if negotiations are unsuccessful, by a determination under the terms of the Contract Documents governing adjustments, expressly including Paragraph 11.07 regarding change of Contract Price.

- B. If Owner has issued a Work Change Directive and:
 - 1. Contractor believes that an adjustment in Contract Times or Contract Price is necessary, then Contractor shall submit any Change Proposal seeking such an adjustment no later than 30 days after the completion of the Work set out in the Work Change Directive.
 - 2. Owner believes that an adjustment in Contract Times or Contract Price is necessary, then Owner shall submit any Claim seeking such an adjustment no later than 60 days after issuance of the Work Change Directive.
- 11.04 Field Orders
 - A. Engineer may authorize minor changes in the Work if the changes do not involve an adjustment in the Contract Price or the Contract Times and are compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents. Such changes will be accomplished by a Field Order and will be binding on Owner and also on Contractor, which shall perform the Work involved promptly.
 - B. If Contractor believes that a Field Order justifies an adjustment in the Contract Price or Contract Times, then before proceeding with the Work at issue, Contractor shall submit a Change Proposal as provided herein.
- 11.05 Owner-Authorized Changes in the Work
 - A. Without invalidating the Contract and without notice to any surety, Owner may, at any time or from time to time, order additions, deletions, or revisions in the Work. Changes involving the design (as set forth in the Drawings, Specifications, or otherwise) or other engineering or technical matters will be supported by Engineer's recommendation.
 - B. Such changes in the Work may be accomplished by a Change Order, if Owner and Contractor have agreed as to the effect, if any, of the changes on Contract Times or Contract Price; or by a Work Change Directive. Upon receipt of any such document, Contractor shall promptly proceed with the Work involved; or, in the case of a deletion in the Work, promptly cease construction activities with respect to such deleted Work. Added or revised Work must be performed under the applicable conditions of the Contract Documents.
 - C. Nothing in this Paragraph 11.05 obligates Contractor to undertake work that Contractor reasonably concludes cannot be performed in a manner consistent with Contractor's safety obligations under the Contract Documents or Laws and Regulations.
- 11.06 Unauthorized Changes in the Work
 - A. Contractor shall not be entitled to an increase in the Contract Price or an extension of the Contract Times with respect to any work performed that is not required by the Contract Documents, as amended, modified, or supplemented, except in the case of an emergency as provided in Paragraph 7.15 or in the case of uncovering Work as provided in Paragraph 14.05.C.2.
- 11.07 Change of Contract Price
 - A. The Contract Price may only be changed by a Change Order. Any Change Proposal for an adjustment in the Contract Price must comply with the provisions of Paragraph 11.09. Any Claim for an adjustment of Contract Price must comply with the provisions of Article 12.
 - B. An adjustment in the Contract Price will be determined as follows:

- 1. Where the Work involved is covered by unit prices contained in the Contract Documents, then by application of such unit prices to the quantities of the items involved (subject to the provisions of Paragraph 13.03);
- 2. Where the Work involved is not covered by unit prices contained in the Contract Documents, then by a mutually agreed lump sum (which may include an allowance for overhead and profit not necessarily in accordance with Paragraph 11.07.C.2); or
- 3. Where the Work involved is not covered by unit prices contained in the Contract Documents and the parties do not reach mutual agreement to a lump sum, then on the basis of the Cost of the Work (determined as provided in Paragraph 13.01) plus a Contractor's fee for overhead and profit (determined as provided in Paragraph 11.07.C).
- C. *Contractor's Fee*: When applicable, the Contractor's fee for overhead and profit will be determined as follows:
 - 1. A mutually acceptable fixed fee; or
 - 2. If a fixed fee is not agreed upon, then a fee based on the following percentages of the various portions of the Cost of the Work:
 - a. For costs incurred under Paragraphs 13.01.B.1 and 13.01.B.2, the Contractor's fee will be 15 percent;
 - b. For costs incurred under Paragraph 13.01.B.3, the Contractor's fee will be 5 percent;
 - c. Where one or more tiers of subcontracts are on the basis of Cost of the Work plus a fee and no fixed fee is agreed upon, the intent of Paragraphs 11.07.C.2.a and 11.07.C.2.b is that the Contractor's fee will be based on: (1) a fee of 15 percent of the costs incurred under Paragraphs 13.01.B.1 and 13.01.B.2 by the Subcontractor that actually performs the Work, at whatever tier, and (2) with respect to Contractor itself and to any Subcontractors of a tier higher than that of the Subcontractor that actually performs the Work, a fee of 5 percent of the amount (fee plus underlying costs incurred) attributable to the next lower tier Subcontractor; provided, however, that for any such subcontracted Work the maximum total fee to be paid by Owner will be no greater than 27 percent of the costs incurred by the Subcontractor that actually performs the Work;
 - d. No fee will be payable on the basis of costs itemized under Paragraphs 13.01.B.4, 13.01.B.5, and 13.01.C;
 - e. The amount of credit to be allowed by Contractor to Owner for any change which results in a net decrease in Cost of the Work will be the amount of the actual net decrease in Cost of the Work and a deduction of an additional amount equal to 5 percent of such actual net decrease in Cost of the Work; and
 - f. When both additions and credits are involved in any one change or Change Proposal, the adjustment in Contractor's fee will be computed by determining the sum of the costs in each of the cost categories in Paragraph 13.01.B (specifically, payroll costs, Paragraph 13.01.B.1; incorporated materials and equipment costs, Paragraph 13.01.B.2; Subcontract costs, Paragraph 13.01.B.3; special consultants costs, Paragraph 13.01.B.4; and other costs, Paragraph 13.01.B.5) and applying to each such cost category sum the appropriate fee from Paragraphs 11.07.C.2.a through 11.07.C.2.e, inclusive.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

and American Society of Civil Engineers. All rights reserved.

- 11.08 Change of Contract Times
 - A. The Contract Times may only be changed by a Change Order. Any Change Proposal for an adjustment in the Contract Times must comply with the provisions of Paragraph 11.09. Any Claim for an adjustment in the Contract Times must comply with the provisions of Article 12.
 - B. Delay, disruption, and interference in the Work, and any related changes in Contract Times, are addressed in and governed by Paragraph 4.05.
- 11.09 Change Proposals
 - A. Purpose and Content: Contractor shall submit a Change Proposal to Engineer to request an adjustment in the Contract Times or Contract Price; contest an initial decision by Engineer concerning the requirements of the Contract Documents or relating to the acceptability of the Work under the Contract Documents; challenge a set-off against payment due; or seek other relief under the Contract. The Change Proposal will specify any proposed change in Contract Times or Contract Price, or other proposed relief, and explain the reason for the proposed change, with citations to any governing or applicable provisions of the Contract Documents. Each Change Proposal will address only one issue, or a set of closely related issues.
 - B. Change Proposal Procedures
 - 1. *Submittal*: Contractor shall submit each Change Proposal to Engineer within 30 days after the start of the event giving rise thereto, or after such initial decision.
 - 2. Supporting Data: The Contractor shall submit supporting data, including the proposed change in Contract Price or Contract Time (if any), to the Engineer and Owner within 15 days after the submittal of the Change Proposal.
 - a. Change Proposals based on or related to delay, interruption, or interference must comply with the provisions of Paragraphs 4.05.D and 4.05.E.
 - b. Change proposals related to a change of Contract Price must include full and detailed accounts of materials incorporated into the Work and labor and equipment used for the subject Work.

The supporting data must be accompanied by a written statement that the supporting data are accurate and complete, and that any requested time or price adjustment is the entire adjustment to which Contractor believes it is entitled as a result of said event.

- 3. Engineer's Initial Review: Engineer will advise Owner regarding the Change Proposal, and consider any comments or response from Owner regarding the Change Proposal. If in its discretion Engineer concludes that additional supporting data is needed before conducting a full review and making a decision regarding the Change Proposal, then Engineer may request that Contractor submit such additional supporting data by a date specified by Engineer, prior to Engineer beginning its full review of the Change Proposal.
- 4. Engineer's Full Review and Action on the Change Proposal: Upon receipt of Contractor's supporting data (including any additional data requested by Engineer), Engineer will conduct a full review of each Change Proposal and, within 30 days after such receipt of the Contractor's supporting data, either approve the Change Proposal in whole, deny it in whole, or approve it in part and deny it in part. Such actions must be in writing, with a copy provided to Owner and Contractor. If Engineer does not take action on the Change

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 10/6/2022

Proposal within 30 days, then either Owner or Contractor may at any time thereafter submit a letter to the other party indicating that as a result of Engineer's inaction the Change Proposal is deemed denied, thereby commencing the time for appeal of the denial under Article 12.

- 5. *Binding Decision*: Engineer's decision is final and binding upon Owner and Contractor, unless Owner or Contractor appeals the decision by filing a Claim under Article 12.
- C. Resolution of Certain Change Proposals: If the Change Proposal does not involve the design (as set forth in the Drawings, Specifications, or otherwise), the acceptability of the Work, or other engineering or technical matters, then Engineer will notify the parties in writing that the Engineer is unable to resolve the Change Proposal. For purposes of further resolution of such a Change Proposal, such notice will be deemed a denial, and Contractor may choose to seek resolution under the terms of Article 12.
- D. *Post-Completion*: Contractor shall not submit any Change Proposals after Engineer issues a written recommendation of final payment pursuant to Paragraph 15.06.B.

11.10 Notification to Surety

A. If the provisions of any bond require notice to be given to a surety of any change affecting the general scope of the Work or the provisions of the Contract Documents (including, but not limited to, Contract Price or Contract Times), the giving of any such notice will be Contractor's responsibility. The amount of each applicable bond will be adjusted to reflect the effect of any such change.

ARTICLE 12—CLAIMS

12.01 Claims

- A. *Claims Process*: The following disputes between Owner and Contractor are subject to the Claims process set forth in this article:
 - 1. Appeals by Owner or Contractor of Engineer's decisions regarding Change Proposals;
 - 2. Owner demands for adjustments in the Contract Price or Contract Times, or other relief under the Contract Documents;
 - Disputes that Engineer has been unable to address because they do not involve the design (as set forth in the Drawings, Specifications, or otherwise), the acceptability of the Work, or other engineering or technical matters; and
 - 4. Subject to the waiver provisions of Paragraph 15.07, any dispute arising after Engineer has issued a written recommendation of final payment pursuant to Paragraph 15.06.B.
- B. Submittal of Claim: The party submitting a Claim shall deliver it directly to the other party to the Contract promptly (but in no event later than 30 days) after the start of the event giving rise thereto; in the case of appeals regarding Change Proposals within 30 days of the decision under appeal. The party submitting the Claim shall also furnish a copy to the Engineer, for its information only. The responsibility to substantiate a Claim rests with the party making the Claim. In the case of a Claim by Contractor seeking an increase in the Contract Times or Contract Price, Contractor shall certify that the Claim is made in good faith, that the supporting data are accurate and complete, and that to the best of Contractor's knowledge

and belief the amount of time or money requested accurately reflects the full amount to which Contractor is entitled.

- C. *Review and Resolution*: The party receiving a Claim shall review it thoroughly, giving full consideration to its merits. The two parties shall seek to resolve the Claim through the exchange of information and direct negotiations. The parties may extend the time for resolving the Claim by mutual agreement. All actions taken on a Claim will be stated in writing and submitted to the other party, with a copy to Engineer.
- D. Mediation
 - 1. At any time after initiation of a Claim, Owner and Contractor may mutually agree to mediation of the underlying dispute. The agreement to mediate will stay the Claim submittal and response process.
 - 2. If Owner and Contractor agree to mediation, then after 60 days from such agreement, either Owner or Contractor may unilaterally terminate the mediation process, and the Claim submittal and decision process will resume as of the date of the termination. If the mediation proceeds but is unsuccessful in resolving the dispute, the Claim submittal and decision process will resume as of the date of the mediation, as determined by the mediator.
 - 3. Owner and Contractor shall each pay one-half of the mediator's fees and costs.
- E. *Partial Approval*: If the party receiving a Claim approves the Claim in part and denies it in part, such action will be final and binding unless within 30 days of such action the other party invokes the procedure set forth in Article 17 for final resolution of disputes.
- F. Denial of Claim: If efforts to resolve a Claim are not successful, the party receiving the Claim may deny it by giving written notice of denial to the other party. If the receiving party does not take action on the Claim within 90 days, then either Owner or Contractor may at any time thereafter submit a letter to the other party indicating that as a result of the inaction, the Claim is deemed denied, thereby commencing the time for appeal of the denial. A denial of the Claim will be final and binding unless within 30 days of the denial the other party invokes the procedure set forth in Article 17 for the final resolution of disputes.
- G. Final and Binding Results: If the parties reach a mutual agreement regarding a Claim, whether through approval of the Claim, direct negotiations, mediation, or otherwise; or if a Claim is approved in part and denied in part, or denied in full, and such actions become final and binding; then the results of the agreement or action on the Claim will be incorporated in a Change Order or other written document to the extent they affect the Contract, including the Work, the Contract Times, or the Contract Price.

ARTICLE 13—COST OF THE WORK; ALLOWANCES; UNIT PRICE WORK

- 13.01 Cost of the Work
 - A. *Purposes for Determination of Cost of the Work*: The term Cost of the Work means the sum of all costs necessary for the proper performance of the Work at issue, as further defined below. The provisions of this Paragraph 13.01 are used for two distinct purposes:
 - 1. To determine Cost of the Work when Cost of the Work is a component of the Contract Price, under cost-plus-fee, time-and-materials, or other cost-based terms; or

- 2. When needed to determine the value of a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price. When the value of any such adjustment is determined on the basis of Cost of the Work, Contractor is entitled only to those additional or incremental costs required because of the change in the Work or because of the event giving rise to the adjustment.
- B. Costs Included: Except as otherwise may be agreed to in writing by Owner, costs included in the Cost of the Work will be in amounts no higher than those commonly incurred in the locality of the Project, will not include any of the costs itemized in Paragraph 13.01.C, and will include only the following items:
 - 1. Payroll costs for employees in the direct employ of Contractor in the performance of the Work under schedules of job classifications agreed upon by Owner and Contractor in advance of the subject Work. Such employees include, without limitation, superintendents, foremen, safety managers, safety representatives, and other personnel employed full time on the Work. Payroll costs for employees not employed full time on the Work will be apportioned on the basis of their time spent on the Work. Payroll costs include, but are not limited to, salaries and wages plus the cost of fringe benefits, which include social security contributions, unemployment, excise, and payroll taxes, workers' compensation, health and retirement benefits, sick leave, and vacation and holiday pay applicable thereto. The expenses of performing Work outside of regular working hours, on Saturday, Sunday, or legal holidays, will be included in the above to the extent authorized by Owner.
 - 2. Cost of all materials and equipment furnished and incorporated in the Work, including costs of transportation and storage thereof, and Suppliers' field services required in connection therewith. All cash discounts accrue to Contractor unless Owner deposits funds with Contractor with which to make payments, in which case the cash discounts will accrue to Owner. All trade discounts, rebates, and refunds and returns from sale of surplus materials and equipment will accrue to Owner, and Contractor shall make provisions so that they may be obtained.
 - 3. Payments made by Contractor to Subcontractors for Work performed by Subcontractors. If required by Owner, Contractor shall obtain competitive bids from subcontractors acceptable to Owner and Contractor and shall deliver such bids to Owner, which will then determine, with the advice of Engineer, which bids, if any, will be acceptable. If any subcontract provides that the Subcontractor is to be paid on the basis of Cost of the Work plus a fee, the Subcontractor's Cost of the Work and fee will be determined in the same manner as Contractor's Cost of the Work and fee as provided in this Paragraph 13.01.
 - 4. Costs of special consultants (including but not limited to engineers, architects, testing laboratories, surveyors, attorneys, and accountants) employed or retained for services specifically related to the Work.
 - 5. Other costs consisting of the following:
 - a. The proportion of necessary transportation, travel, and subsistence expenses of Contractor's employees incurred in discharge of duties connected with the Work.
 - b. Cost, including transportation and maintenance, of all materials, supplies, equipment, machinery, appliances, office, and temporary facilities at the Site, which are

consumed in the performance of the Work, and cost, less market value, of such items used but not consumed which remain the property of Contractor.

- In establishing included costs for materials such as scaffolding, plating, or sheeting, consideration will be given to the actual or the estimated life of the material for use on other projects; or rental rates may be established on the basis of purchase or salvage value of such items, whichever is less. Contractor will not be eligible for compensation for such items in an amount that exceeds the purchase cost of such item.
- c. Construction Equipment Rental
 - 1) Rentals of all construction equipment and machinery, and the parts thereof, in accordance with rental agreements approved by Owner as to price (including any surcharge or special rates applicable to overtime use of the construction equipment or machinery), and the costs of transportation, loading, unloading, assembly, dismantling, and removal thereof. All such costs will be in accordance with the terms of said rental agreements. The rental of any such equipment, machinery, or parts must cease when the use thereof is no longer necessary for the Work.
 - 2) Costs for equipment and machinery owned by Contractor or a Contractor-related entity will be paid at a rate shown for such equipment in the equipment rental rate book specified in the Supplementary Conditions. An hourly rate will be computed by dividing the monthly rates by 176. These computed rates will include all operating costs.
 - 3) With respect to Work that is the result of a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price ("changed Work"), included costs will be based on the time the equipment or machinery is in use on the changed Work and the costs of transportation, loading, unloading, assembly, dismantling, and removal when directly attributable to the changed Work. The cost of any such equipment or machinery, or parts thereof, must cease to accrue when the use thereof is no longer necessary for the changed Work.
- d. Sales, consumer, use, and other similar taxes related to the Work, and for which Contractor is liable, as imposed by Laws and Regulations.
- e. Deposits lost for causes other than negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, and royalty payments and fees for permits and licenses.
- f. Losses and damages (and related expenses) caused by damage to the Work, not compensated by insurance or otherwise, sustained by Contractor in connection with the performance of the Work (except losses and damages within the deductible amounts of builder's risk or other property insurance established in accordance with Paragraph 6.04), provided such losses and damages have resulted from causes other than the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable. Such losses include settlements made with the written consent and approval of Owner. No such losses, damages, and expenses will be included in the Cost of the Work for the purpose of determining Contractor's fee.

EJCDC[®] C-700, Standard General Conditions of the Construction Contract. Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

- g. The cost of utilities, fuel, and sanitary facilities at the Site.
- h. Minor expenses such as communication service at the Site, express and courier services, and similar petty cash items in connection with the Work.
- i. The costs of premiums for all bonds and insurance that Contractor is required by the Contract Documents to purchase and maintain.
- C. Costs Excluded: The term Cost of the Work does not include any of the following items:
 - Payroll costs and other compensation of Contractor's officers, executives, principals, general managers, engineers, architects, estimators, attorneys, auditors, accountants, purchasing and contracting agents, expediters, timekeepers, clerks, and other personnel employed by Contractor, whether at the Site or in Contractor's principal or branch office for general administration of the Work and not specifically included in the agreed upon schedule of job classifications referred to in Paragraph 13.01.B.1 or specifically covered by Paragraph 13.01.B.4. The payroll costs and other compensation excluded here are to be considered administrative costs covered by the Contractor's fee.
 - 2. The cost of purchasing, renting, or furnishing small tools and hand tools.
 - 3. Expenses of Contractor's principal and branch offices other than Contractor's office at the Site.
 - 4. Any part of Contractor's capital expenses, including interest on Contractor's capital employed for the Work and charges against Contractor for delinquent payments.
 - 5. Costs due to the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, including but not limited to, the correction of defective Work, disposal of materials or equipment wrongly supplied, and making good any damage to property.
 - 6. Expenses incurred in preparing and advancing Claims.
 - 7. Other overhead or general expense costs of any kind and the costs of any item not specifically and expressly included in Paragraph 13.01.B.
- D. Contractor's Fee
 - 1. When the Work as a whole is performed on the basis of cost-plus-a-fee, then:
 - a. Contractor's fee for the Work set forth in the Contract Documents as of the Effective Date of the Contract will be determined as set forth in the Agreement.
 - b. for any Work covered by a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price on the basis of Cost of the Work, Contractor's fee will be determined as follows:
 - 1) When the fee for the Work as a whole is a percentage of the Cost of the Work, the fee will automatically adjust as the Cost of the Work changes.
 - 2) When the fee for the Work as a whole is a fixed fee, the fee for any additions or deletions will be determined in accordance with Paragraph 11.07.C.2.
 - 2. When the Work as a whole is performed on the basis of a stipulated sum, or any other basis other than cost-plus-a-fee, then Contractor's fee for any Work covered by a Change

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 10/6/2022

Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price on the basis of Cost of the Work will be determined in accordance with Paragraph 11.07.C.2.

E. Documentation and Audit: Whenever the Cost of the Work for any purpose is to be determined pursuant to this Article 13, Contractor and pertinent Subcontractors will establish and maintain records of the costs in accordance with generally accepted accounting practices. Subject to prior written notice, Owner will be afforded reasonable access, during normal business hours, to all Contractor's accounts, records, books, correspondence, instructions, drawings, receipts, vouchers, memoranda, and similar data relating to the Cost of the Work and Contractor's fee. Contractor shall preserve all such documents for a period of three years after the final payment by Owner. Pertinent Subcontractors will afford such access to Owner, and preserve such documents, to the same extent required of Contractor.

13.02 Allowances

- A. It is understood that Contractor has included in the Contract Price all allowances so named in the Contract Documents and shall cause the Work so covered to be performed for such sums and by such persons or entities as may be acceptable to Owner and Engineer.
- B. Cash Allowances: Contractor agrees that:
 - the cash allowances include the cost to Contractor (less any applicable trade discounts) of materials and equipment required by the allowances to be delivered at the Site, and all applicable taxes; and
 - Contractor's costs for unloading and handling on the Site, labor, installation, overhead, profit, and other expenses contemplated for the cash allowances have been included in the Contract Price and not in the allowances, and no demand for additional payment for any of the foregoing will be valid.
- C. Owner's Contingency Allowance: Contractor agrees that an Owner's contingency allowance, if any, is for the sole use of Owner to cover unanticipated costs.
- D. Prior to final payment, an appropriate Change Order will be issued as recommended by Engineer to reflect actual amounts due Contractor for Work covered by allowances, and the Contract Price will be correspondingly adjusted.

13.03 Unit Price Work

- A. Where the Contract Documents provide that all or part of the Work is to be Unit Price Work, initially the Contract Price will be deemed to include for all Unit Price Work an amount equal to the sum of the unit price for each separately identified item of Unit Price Work times the estimated quantity of each item as indicated in the Agreement.
- B. The estimated quantities of items of Unit Price Work are not guaranteed and are solely for the purpose of comparison of Bids and determining an initial Contract Price. Payments to Contractor for Unit Price Work will be based on actual quantities.
- C. Each unit price will be deemed to include an amount considered by Contractor to be adequate to cover Contractor's overhead and profit for each separately identified item.
- D. Engineer will determine the actual quantities and classifications of Unit Price Work performed by Contractor. Engineer will review with Contractor the Engineer's preliminary determinations on such matters before rendering a written decision thereon (by recommendation of an Application for Payment or otherwise). Engineer's written decision

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 10/6/2022

thereon will be final and binding (except as modified by Engineer to reflect changed factual conditions or more accurate data) upon Owner and Contractor, and the final adjustment of Contract Price will be set forth in a Change Order, subject to the provisions of the following paragraph.

- E. Adjustments in Unit Price
 - 1. Contractor or Owner shall be entitled to an adjustment in the unit price with respect to an item of Unit Price Work if:
 - a. the quantity of the item of Unit Price Work performed by Contractor differs materially and significantly from the estimated quantity of such item indicated in the Agreement; and
 - b. Contractor's unit costs to perform the item of Unit Price Work have changed materially and significantly as a result of the quantity change.
 - 2. The adjustment in unit price will account for and be coordinated with any related changes in quantities of other items of Work, and in Contractor's costs to perform such other Work, such that the resulting overall change in Contract Price is equitable to Owner and Contractor.
 - 3. Adjusted unit prices will apply to all units of that item.

ARTICLE 14—TESTS AND INSPECTIONS; CORRECTION, REMOVAL, OR ACCEPTANCE OF DEFECTIVE WORK

- 14.01 Access to Work
 - A. Owner, Engineer, their consultants and other representatives and personnel of Owner, independent testing laboratories, and authorities having jurisdiction have access to the Site and the Work at reasonable times for their observation, inspection, and testing. Contractor shall provide them proper and safe conditions for such access and advise them of Contractor's safety procedures and programs so that they may comply with such procedures and programs as applicable.

14.02 Tests, Inspections, and Approvals

- A. Contractor shall give Engineer timely notice of readiness of the Work (or specific parts thereof) for all required inspections and tests, and shall cooperate with inspection and testing personnel to facilitate required inspections and tests.
- B. Owner shall retain and pay for the services of an independent inspector, testing laboratory, or other qualified individual or entity to perform all inspections and tests expressly required by the Contract Documents to be furnished and paid for by Owner, except that costs incurred in connection with tests or inspections of covered Work will be governed by the provisions of Paragraph 14.05.
- C. If Laws or Regulations of any public body having jurisdiction require any Work (or part thereof) specifically to be inspected, tested, or approved by an employee or other representative of such public body, Contractor shall assume full responsibility for arranging and obtaining such inspections, tests, or approvals, pay all costs in connection therewith, and furnish Engineer the required certificates of inspection or approval.

- D. Contractor shall be responsible for arranging, obtaining, and paying for all inspections and tests required:
 - 1. by the Contract Documents, unless the Contract Documents expressly allocate responsibility for a specific inspection or test to Owner;
 - 2. to attain Owner's and Engineer's acceptance of materials or equipment to be incorporated in the Work;
 - 3. by manufacturers of equipment furnished under the Contract Documents;
 - 4. for testing, adjusting, and balancing of mechanical, electrical, and other equipment to be incorporated into the Work; and
 - 5. for acceptance of materials, mix designs, or equipment submitted for approval prior to Contractor's purchase thereof for incorporation in the Work.

Such inspections and tests will be performed by independent inspectors, testing laboratories, or other qualified individuals or entities acceptable to Owner and Engineer.

- E. If the Contract Documents require the Work (or part thereof) to be approved by Owner, Engineer, or another designated individual or entity, then Contractor shall assume full responsibility for arranging and obtaining such approvals.
- F. If any Work (or the work of others) that is to be inspected, tested, or approved is covered by Contractor without written concurrence of Engineer, Contractor shall, if requested by Engineer, uncover such Work for observation. Such uncovering will be at Contractor's expense unless Contractor had given Engineer timely notice of Contractor's intention to cover the same and Engineer had not acted with reasonable promptness in response to such notice.

14.03 Defective Work

- A. *Contractor's Obligation*: It is Contractor's obligation to assure that the Work is not defective.
- B. *Engineer's Authority*: Engineer has the authority to determine whether Work is defective, and to reject defective Work.
- C. *Notice of Defects*: Prompt written notice of all defective Work of which Owner or Engineer has actual knowledge will be given to Contractor.
- D. *Correction, or Removal and Replacement*: Promptly after receipt of written notice of defective Work, Contractor shall correct all such defective Work, whether or not fabricated, installed, or completed, or, if Engineer has rejected the defective Work, remove it from the Project and replace it with Work that is not defective.
- E. *Preservation of Warranties*: When correcting defective Work, Contractor shall take no action that would void or otherwise impair Owner's special warranty and guarantee, if any, on said Work.
- F. Costs and Damages: In addition to its correction, removal, and replacement obligations with respect to defective Work, Contractor shall pay all claims, costs, losses, and damages arising out of or relating to defective Work, including but not limited to the cost of the inspection, testing, correction, removal, replacement, or reconstruction of such defective Work, fines levied against Owner by governmental authorities because the Work is defective, and the costs of repair or replacement of work of others resulting from defective Work. Prior to final payment, if Owner and Contractor are unable to agree as to the measure of such claims, costs,

losses, and damages resulting from defective Work, then Owner may impose a reasonable set-off against payments due under Article 15.

- 14.04 Acceptance of Defective Work
 - A. If, instead of requiring correction or removal and replacement of defective Work, Owner prefers to accept it, Owner may do so (subject, if such acceptance occurs prior to final payment, to Engineer's confirmation that such acceptance is in general accord with the design intent and applicable engineering principles, and will not endanger public safety). Contractor shall pay all claims, costs, losses, and damages attributable to Owner's evaluation of and determination to accept such defective Work (such costs to be approved by Engineer as to reasonableness), and for the diminished value of the Work to the extent not otherwise paid by Contractor. If any such acceptance occurs prior to final payment, the necessary revisions in the Contract Documents with respect to the Work will be incorporated in a Change Order. If the parties are unable to agree as to the decrease in the Contract Price, reflecting the diminished value of Work so accepted, then Owner may impose a reasonable set-off against payments due under Article 15. If the acceptance of defective Work occurs after final payment, Contractor shall pay an appropriate amount to Owner.

14.05 Uncovering Work

- A. Engineer has the authority to require additional inspection or testing of the Work, whether or not the Work is fabricated, installed, or completed.
- B. If any Work is covered contrary to the written request of Engineer, then Contractor shall, if requested by Engineer, uncover such Work for Engineer's observation, and then replace the covering, all at Contractor's expense.
- C. If Engineer considers it necessary or advisable that covered Work be observed by Engineer or inspected or tested by others, then Contractor, at Engineer's request, shall uncover, expose, or otherwise make available for observation, inspection, or testing as Engineer may require, that portion of the Work in question, and provide all necessary labor, material, and equipment.
 - If it is found that the uncovered Work is defective, Contractor shall be responsible for all claims, costs, losses, and damages arising out of or relating to such uncovering, exposure, observation, inspection, and testing, and of satisfactory replacement or reconstruction (including but not limited to all costs of repair or replacement of work of others); and pending Contractor's full discharge of this responsibility the Owner shall be entitled to impose a reasonable set-off against payments due under Article 15.
 - 2. If the uncovered Work is not found to be defective, Contractor shall be allowed an increase in the Contract Price or an extension of the Contract Times, directly attributable to such uncovering, exposure, observation, inspection, testing, replacement, and reconstruction. If the parties are unable to agree as to the amount or extent thereof, then Contractor may submit a Change Proposal within 30 days of the determination that the Work is not defective.

14.06 Owner May Stop the Work

A. If the Work is defective, or Contractor fails to supply sufficient skilled workers or suitable materials or equipment, or fails to perform the Work in such a way that the completed Work will conform to the Contract Documents, then Owner may order Contractor to stop the Work,

EJCDC[®] C-700, Standard General Conditions of the Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

or any portion thereof, until the cause for such order has been eliminated; however, this right of Owner to stop the Work will not give rise to any duty on the part of Owner to exercise this right for the benefit of Contractor, any Subcontractor, any Supplier, any other individual or entity, or any surety for, or employee or agent of any of them.

14.07 Owner May Correct Defective Work

- A. If Contractor fails within a reasonable time after written notice from Engineer to correct defective Work, or to remove and replace defective Work as required by Engineer, then Owner may, after 7 days' written notice to Contractor, correct or remedy any such deficiency.
- B. In exercising the rights and remedies under this Paragraph 14.07, Owner shall proceed expeditiously. In connection with such corrective or remedial action, Owner may exclude Contractor from all or part of the Site, take possession of all or part of the Work and suspend Contractor's services related thereto, and incorporate in the Work all materials and equipment stored at the Site or for which Owner has paid Contractor but which are stored elsewhere. Contractor shall allow Owner, Owner's representatives, agents and employees, Owner's other contractors, and Engineer and Engineer's consultants access to the Site to enable Owner to exercise the rights and remedies under this paragraph.
- C. All claims, costs, losses, and damages incurred or sustained by Owner in exercising the rights and remedies under this Paragraph 14.07 will be charged against Contractor as set-offs against payments due under Article 15. Such claims, costs, losses and damages will include but not be limited to all costs of repair, or replacement of work of others destroyed or damaged by correction, removal, or replacement of Contractor's defective Work.
- D. Contractor shall not be allowed an extension of the Contract Times because of any delay in the performance of the Work attributable to the exercise by Owner of Owner's rights and remedies under this Paragraph 14.07.

ARTICLE 15—PAYMENTS TO CONTRACTOR; SET-OFFS; COMPLETION; CORRECTION PERIOD

- 15.01 Progress Payments
 - A. Basis for Progress Payments: The Schedule of Values established as provided in Article 2 will serve as the basis for progress payments and will be incorporated into a form of Application for Payment acceptable to Engineer. Progress payments for Unit Price Work will be based on the number of units completed during the pay period, as determined under the provisions of Paragraph 13.03. Progress payments for cost-based Work will be based on Cost of the Work completed by Contractor during the pay period.
 - B. Applications for Payments
 - At least 20 days before the date established in the Agreement for each progress payment (but not more often than once a month), Contractor shall submit to Engineer for review an Application for Payment filled out and signed by Contractor covering the Work completed as of the date of the Application and accompanied by such supporting documentation as is required by the Contract Documents.
 - 2. If payment is requested on the basis of materials and equipment not incorporated in the Work but delivered and suitably stored at the Site or at another location agreed to in writing, the Application for Payment must also be accompanied by: (a) a bill of sale, invoice, copies of subcontract or purchase order payments, or other documentation

establishing full payment by Contractor for the materials and equipment; (b) at Owner's request, documentation warranting that Owner has received the materials and equipment free and clear of all Liens; and (c) evidence that the materials and equipment are covered by appropriate property insurance, a warehouse bond, or other arrangements to protect Owner's interest therein, all of which must be satisfactory to Owner.

- 3. Beginning with the second Application for Payment, each Application must include an affidavit of Contractor stating that all previous progress payments received by Contractor have been applied to discharge Contractor's legitimate obligations associated with prior Applications for Payment.
- 4. The amount of retainage with respect to progress payments will be as stipulated in the Agreement.
- C. Review of Applications
 - Engineer will, within 10 days after receipt of each Application for Payment, including each resubmittal, either indicate in writing a recommendation of payment and present the Application to Owner, or return the Application to Contractor indicating in writing Engineer's reasons for refusing to recommend payment. In the latter case, Contractor may make the necessary corrections and resubmit the Application.
 - 2. Engineer's recommendation of any payment requested in an Application for Payment will constitute a representation by Engineer to Owner, based on Engineer's observations of the executed Work as an experienced and qualified design professional, and on Engineer's review of the Application for Payment and the accompanying data and schedules, that to the best of Engineer's knowledge, information and belief:
 - a. the Work has progressed to the point indicated;
 - b. the quality of the Work is generally in accordance with the Contract Documents (subject to an evaluation of the Work as a functioning whole prior to or upon Substantial Completion, the results of any subsequent tests called for in the Contract Documents, a final determination of quantities and classifications for Unit Price Work under Paragraph 13.03, and any other qualifications stated in the recommendation); and
 - c. the conditions precedent to Contractor's being entitled to such payment appear to have been fulfilled in so far as it is Engineer's responsibility to observe the Work.
 - 3. By recommending any such payment Engineer will not thereby be deemed to have represented that:
 - a. inspections made to check the quality or the quantity of the Work as it has been performed have been exhaustive, extended to every aspect of the Work in progress, or involved detailed inspections of the Work beyond the responsibilities specifically assigned to Engineer in the Contract; or
 - b. there may not be other matters or issues between the parties that might entitle Contractor to be paid additionally by Owner or entitle Owner to withhold payment to Contractor.

- 4. Neither Engineer's review of Contractor's Work for the purposes of recommending payments nor Engineer's recommendation of any payment, including final payment, will impose responsibility on Engineer:
 - a. to supervise, direct, or control the Work;
 - b. for the means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto;
 - c. for Contractor's failure to comply with Laws and Regulations applicable to Contractor's performance of the Work;
 - d. to make any examination to ascertain how or for what purposes Contractor has used the money paid by Owner; or
 - e. to determine that title to any of the Work, materials, or equipment has passed to Owner free and clear of any Liens.
- 5. Engineer may refuse to recommend the whole or any part of any payment if, in Engineer's opinion, it would be incorrect to make the representations to Owner stated in Paragraph 15.01.C.2.
- 6. Engineer will recommend reductions in payment (set-offs) necessary in Engineer's opinion to protect Owner from loss because:
 - a. the Work is defective, requiring correction or replacement;
 - b. the Contract Price has been reduced by Change Orders;
 - c. Owner has been required to correct defective Work in accordance with Paragraph 14.07, or has accepted defective Work pursuant to Paragraph 14.04;
 - d. Owner has been required to remove or remediate a Hazardous Environmental Condition for which Contractor is responsible; or
 - e. Engineer has actual knowledge of the occurrence of any of the events that would constitute a default by Contractor and therefore justify termination for cause under the Contract Documents.
- D. Payment Becomes Due
 - 1. Ten days after presentation of the Application for Payment to Owner with Engineer's recommendation, the amount recommended (subject to any Owner set-offs) will become due, and when due will be paid by Owner to Contractor.
- E. Reductions in Payment by Owner
 - 1. In addition to any reductions in payment (set-offs) recommended by Engineer, Owner is entitled to impose a set-off against payment based on any of the following:
 - a. Claims have been made against Owner based on Contractor's conduct in the performance or furnishing of the Work, or Owner has incurred costs, losses, or damages resulting from Contractor's conduct in the performance or furnishing of the Work, including but not limited to claims, costs, losses, or damages from workplace injuries, adjacent property damage, non-compliance with Laws and Regulations, and patent infringement;

- b. Contractor has failed to take reasonable and customary measures to avoid damage, delay, disruption, and interference with other work at or adjacent to the Site;
- c. Contractor has failed to provide and maintain required bonds or insurance;
- d. Owner has been required to remove or remediate a Hazardous Environmental Condition for which Contractor is responsible;
- e. Owner has incurred extra charges or engineering costs related to submittal reviews, evaluations of proposed substitutes, tests and inspections, or return visits to manufacturing or assembly facilities;
- f. The Work is defective, requiring correction or replacement;
- g. Owner has been required to correct defective Work in accordance with Paragraph 14.07, or has accepted defective Work pursuant to Paragraph 14.04;
- h. The Contract Price has been reduced by Change Orders;
- i. An event has occurred that would constitute a default by Contractor and therefore justify a termination for cause;
- j. Liquidated or other damages have accrued as a result of Contractor's failure to achieve Milestones, Substantial Completion, or final completion of the Work;
- k. Liens have been filed in connection with the Work, except where Contractor has delivered a specific bond satisfactory to Owner to secure the satisfaction and discharge of such Liens; or
- I. Other items entitle Owner to a set-off against the amount recommended.
- 2. If Owner imposes any set-off against payment, whether based on its own knowledge or on the written recommendations of Engineer, Owner will give Contractor immediate written notice (with a copy to Engineer) stating the reasons for such action and the specific amount of the reduction, and promptly pay Contractor any amount remaining after deduction of the amount so withheld. Owner shall promptly pay Contractor the amount so withheld, or any adjustment thereto agreed to by Owner and Contractor, if Contractor remedies the reasons for such action. The reduction imposed will be binding on Contractor unless it duly submits a Change Proposal contesting the reduction.
- 3. Upon a subsequent determination that Owner's refusal of payment was not justified, the amount wrongfully withheld will be treated as an amount due as determined by Paragraph 15.01.D.1 and subject to interest as provided in the Agreement.

15.02 Contractor's Warranty of Title

A. Contractor warrants and guarantees that title to all Work, materials, and equipment furnished under the Contract will pass to Owner free and clear of (1) all Liens and other title defects, and (2) all patent, licensing, copyright, or royalty obligations, no later than 7 days after the time of payment by Owner.

15.03 Substantial Completion

A. When Contractor considers the entire Work ready for its intended use Contractor shall notify Owner and Engineer in writing that the entire Work is substantially complete and request that Engineer issue a certificate of Substantial Completion. Contractor shall at the same time submit to Owner and Engineer an initial draft of punch list items to be completed or corrected before final payment.

- B. Promptly after Contractor's notification, Owner, Contractor, and Engineer shall make an inspection of the Work to determine the status of completion. If Engineer does not consider the Work substantially complete, Engineer will notify Contractor in writing giving the reasons therefor.
- C. If Engineer considers the Work substantially complete, Engineer will deliver to Owner a preliminary certificate of Substantial Completion which will fix the date of Substantial Completion. Engineer shall attach to the certificate a punch list of items to be completed or corrected before final payment. Owner shall have 7 days after receipt of the preliminary certificate during which to make written objection to Engineer as to any provisions of the certificate or attached punch list. If, after considering the objections to the provisions of the preliminary certificate, Engineer concludes that the Work is not substantially complete, Engineer will, within 14 days after submission of the preliminary certificate to Owner, notify Contractor in writing that the Work is not substantially complete, stating the reasons therefor. If Owner does not object to the provisions of the certificate, or if despite consideration of Owner's objections Engineer concludes that the Work is substantially complete, then Engineer will, within said 14 days, execute and deliver to Owner and Contractor a final certificate of Substantial Completion (with a revised punch list of items to be completed or corrected) reflecting such changes from the preliminary certificate as Engineer believes justified after consideration of any objections from Owner.
- D. At the time of receipt of the preliminary certificate of Substantial Completion, Owner and Contractor will confer regarding Owner's use or occupancy of the Work following Substantial Completion, review the builder's risk insurance policy with respect to the end of the builder's risk coverage, and confirm the transition to coverage of the Work under a permanent property insurance policy held by Owner. Unless Owner and Contractor agree otherwise in writing, Owner shall bear responsibility for security, operation, protection of the Work, property insurance, maintenance, heat, and utilities upon Owner's use or occupancy of the Work.
- E. After Substantial Completion the Contractor shall promptly begin work on the punch list of items to be completed or corrected prior to final payment. In appropriate cases Contractor may submit monthly Applications for Payment for completed punch list items, following the progress payment procedures set forth above.
- F. Owner shall have the right to exclude Contractor from the Site after the date of Substantial Completion subject to allowing Contractor reasonable access to remove its property and complete or correct items on the punch list.

15.04 Partial Use or Occupancy

A. Prior to Substantial Completion of all the Work, Owner may use or occupy any substantially completed part of the Work which has specifically been identified in the Contract Documents, or which Owner, Engineer, and Contractor agree constitutes a separately functioning and usable part of the Work that can be used by Owner for its intended purpose without

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 10/6/2022

significant interference with Contractor's performance of the remainder of the Work, subject to the following conditions:

- 1. At any time, Owner may request in writing that Contractor permit Owner to use or occupy any such part of the Work that Owner believes to be substantially complete. If and when Contractor agrees that such part of the Work is substantially complete, Contractor, Owner, and Engineer will follow the procedures of Paragraph 15.03.A through 15.03.E for that part of the Work.
- 2. At any time, Contractor may notify Owner and Engineer in writing that Contractor considers any such part of the Work substantially complete and request Engineer to issue a certificate of Substantial Completion for that part of the Work.
- 3. Within a reasonable time after either such request, Owner, Contractor, and Engineer shall make an inspection of that part of the Work to determine its status of completion. If Engineer does not consider that part of the Work to be substantially complete, Engineer will notify Owner and Contractor in writing giving the reasons therefor. If Engineer considers that part of the Work to be substantially complete, the provisions of Paragraph 15.03 will apply with respect to certification of Substantial Completion of that part of the Work and the division of responsibility in respect thereof and access thereto.
- 4. No use or occupancy or separate operation of part of the Work may occur prior to compliance with the requirements of Paragraph 6.04 regarding builder's risk or other property insurance.

15.05 Final Inspection

A. Upon written notice from Contractor that the entire Work or an agreed portion thereof is complete, Engineer will promptly make a final inspection with Owner and Contractor and will notify Contractor in writing of all particulars in which this inspection reveals that the Work, or agreed portion thereof, is incomplete or defective. Contractor shall immediately take such measures as are necessary to complete such Work or remedy such deficiencies.

15.06 Final Payment

A. Application for Payment

- After Contractor has, in the opinion of Engineer, satisfactorily completed all corrections identified during the final inspection and has delivered, in accordance with the Contract Documents, all maintenance and operating instructions, schedules, guarantees, bonds, certificates or other evidence of insurance, certificates of inspection, annotated record documents (as provided in Paragraph 7.12), and other documents, Contractor may make application for final payment.
- 2. The final Application for Payment must be accompanied (except as previously delivered) by:
 - a. all documentation called for in the Contract Documents;
 - b. consent of the surety, if any, to final payment;
 - c. satisfactory evidence that all title issues have been resolved such that title to all Work, materials, and equipment has passed to Owner free and clear of any Liens or other title defects, or will so pass upon final payment.

- d. a list of all duly pending Change Proposals and Claims; and
- e. complete and legally effective releases or waivers (satisfactory to Owner) of all Lien rights arising out of the Work, and of Liens filed in connection with the Work.
- 3. In lieu of the releases or waivers of Liens specified in Paragraph 15.06.A.2 and as approved by Owner, Contractor may furnish receipts or releases in full and an affidavit of Contractor that: (a) the releases and receipts include all labor, services, material, and equipment for which a Lien could be filed; and (b) all payrolls, material and equipment bills, and other indebtedness connected with the Work for which Owner might in any way be responsible, or which might in any way result in liens or other burdens on Owner's property, have been paid or otherwise satisfied. If any Subcontractor or Supplier fails to furnish such a release or receipt in full, Contractor may furnish a bond or other collateral satisfactory to Owner to indemnify Owner against any Lien, or Owner at its option may issue joint checks payable to Contractor and specified Subcontractors and Suppliers.
- B. Engineer's Review of Final Application and Recommendation of Payment: If, on the basis of Engineer's observation of the Work during construction and final inspection, and Engineer's review of the final Application for Payment and accompanying documentation as required by the Contract Documents, Engineer is satisfied that the Work has been completed and Contractor's other obligations under the Contract have been fulfilled, Engineer will, within 10 days after receipt of the final Application for Payment, indicate in writing Engineer's recommendation of final payment and present the final Application for Payment to Owner for payment. Such recommendation will account for any set-offs against payment that are necessary in Engineer's opinion to protect Owner from loss for the reasons stated above with respect to progress payments. Otherwise, Engineer will return the Application for Payment to Contractor, indicating in writing the reasons for refusing to recommend final payment, in which case Contractor shall make the necessary corrections and resubmit the Application for Payment.
- C. Notice of Acceptability: In support of its recommendation of payment of the final Application for Payment, Engineer will also give written notice to Owner and Contractor that the Work is acceptable, subject to stated limitations in the notice and to the provisions of Paragraph 15.07.
- D. Completion of Work: The Work is complete (subject to surviving obligations) when it is ready for final payment as established by the Engineer's written recommendation of final payment and issuance of notice of the acceptability of the Work.
- E. Final Payment Becomes Due: Upon receipt from Engineer of the final Application for Payment and accompanying documentation, Owner shall set off against the amount recommended by Engineer for final payment any further sum to which Owner is entitled, including but not limited to set-offs for liquidated damages and set-offs allowed under the provisions of this Contract with respect to progress payments. Owner shall pay the resulting balance due to Contractor within 30 days of Owner's receipt of the final Application for Payment from Engineer.
- 15.07 Waiver of Claims
 - A. By making final payment, Owner waives its claim or right to liquidated damages or other damages for late completion by Contractor, except as set forth in an outstanding Claim,

appeal under the provisions of Article 17, set-off, or express reservation of rights by Owner. Owner reserves all other claims or rights after final payment.

B. The acceptance of final payment by Contractor will constitute a waiver by Contractor of all claims and rights against Owner other than those pending matters that have been duly submitted as a Claim, or appealed under the provisions of Article 17.

15.08 Correction Period

- A. If within one year after the date of Substantial Completion (or such longer period of time as may be prescribed by the Supplementary Conditions or the terms of any applicable special guarantee required by the Contract Documents), Owner gives Contractor written notice that any Work has been found to be defective, or that Contractor's repair of any damages to the Site or adjacent areas has been found to be defective, then after receipt of such notice of defect Contractor shall promptly, without cost to Owner and in accordance with Owner's written instructions:
 - 1. correct the defective repairs to the Site or such adjacent areas;
 - 2. correct such defective Work;
 - 3. remove the defective Work from the Project and replace it with Work that is not defective, if the defective Work has been rejected by Owner, and
 - 4. satisfactorily correct or repair or remove and replace any damage to other Work, to the work of others, or to other land or areas resulting from the corrective measures.
- B. Owner shall give any such notice of defect within 60 days of the discovery that such Work or repairs is defective. If such notice is given within such 60 days but after the end of the correction period, the notice will be deemed a notice of defective Work under Paragraph 7.17.B.
- C. If, after receipt of a notice of defect within 60 days and within the correction period, Contractor does not promptly comply with the terms of Owner's written instructions, or in an emergency where delay would cause serious risk of loss or damage, Owner may have the defective Work corrected or repaired or may have the rejected Work removed and replaced. Contractor shall pay all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such correction or repair or such removal and replacement (including but not limited to all costs of repair or replacement of work of others). Contractor's failure to pay such costs, losses, and damages within 10 days of invoice from Owner will be deemed the start of an event giving rise to a Claim under Paragraph 12.01.B, such that any related Claim must be brought within 30 days of the failure to pay.
- D. In special circumstances where a particular item of equipment is placed in continuous service before Substantial Completion of all the Work, the correction period for that item may start to run from an earlier date if so provided in the Specifications.
- E. Where defective Work (and damage to other Work resulting therefrom) has been corrected or removed and replaced under this paragraph, the correction period hereunder with respect to such Work will be extended for an additional period of one year after such correction or removal and replacement has been satisfactorily completed.

F. Contractor's obligations under this paragraph are in addition to all other obligations and warranties. The provisions of this paragraph are not to be construed as a substitute for, or a waiver of, the provisions of any applicable statute of limitation or repose.

ARTICLE 16—SUSPENSION OF WORK AND TERMINATION

- 16.01 Owner May Suspend Work
 - A. At any time and without cause, Owner may suspend the Work or any portion thereof for a period of not more than 90 consecutive days by written notice to Contractor and Engineer. Such notice will fix the date on which Work will be resumed. Contractor shall resume the Work on the date so fixed. Contractor shall be entitled to an adjustment in the Contract Price or an extension of the Contract Times directly attributable to any such suspension. Any Change Proposal seeking such adjustments must be submitted no later than 30 days after the date fixed for resumption of Work.

16.02 Owner May Terminate for Cause

- A. The occurrence of any one or more of the following events will constitute a default by Contractor and justify termination for cause:
 - Contractor's persistent failure to perform the Work in accordance with the Contract Documents (including, but not limited to, failure to supply sufficient skilled workers or suitable materials or equipment, or failure to adhere to the Progress Schedule);
 - 2. Failure of Contractor to perform or otherwise to comply with a material term of the Contract Documents;
 - 3. Contractor's disregard of Laws or Regulations of any public body having jurisdiction; or
 - 4. Contractor's repeated disregard of the authority of Owner or Engineer.
- B. If one or more of the events identified in Paragraph 16.02.A occurs, then after giving Contractor (and any surety) 10 days' written notice that Owner is considering a declaration that Contractor is in default and termination of the Contract, Owner may proceed to:
 - 1. declare Contractor to be in default, and give Contractor (and any surety) written notice that the Contract is terminated; and
 - 2. enforce the rights available to Owner under any applicable performance bond.
- C. Subject to the terms and operation of any applicable performance bond, if Owner has terminated the Contract for cause, Owner may exclude Contractor from the Site, take possession of the Work, incorporate in the Work all materials and equipment stored at the Site or for which Owner has paid Contractor but which are stored elsewhere, and complete the Work as Owner may deem expedient.
- D. Owner may not proceed with termination of the Contract under Paragraph 16.02.B if Contractor within 7 days of receipt of notice of intent to terminate begins to correct its failure to perform and proceeds diligently to cure such failure.
- E. If Owner proceeds as provided in Paragraph 16.02.B, Contractor shall not be entitled to receive any further payment until the Work is completed. If the unpaid balance of the Contract Price exceeds the cost to complete the Work, including all related claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects,

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

attorneys, and other professionals) sustained by Owner, such excess will be paid to Contractor. If the cost to complete the Work including such related claims, costs, losses, and damages exceeds such unpaid balance, Contractor shall pay the difference to Owner. Such claims, costs, losses, and damages incurred by Owner will be reviewed by Engineer as to their reasonableness and, when so approved by Engineer, incorporated in a Change Order. When exercising any rights or remedies under this paragraph, Owner shall not be required to obtain the lowest price for the Work performed.

- F. Where Contractor's services have been so terminated by Owner, the termination will not affect any rights or remedies of Owner against Contractor then existing or which may thereafter accrue, or any rights or remedies of Owner against Contractor or any surety under any payment bond or performance bond. Any retention or payment of money due Contractor by Owner will not release Contractor from liability.
- G. If and to the extent that Contractor has provided a performance bond under the provisions of Paragraph 6.01.A, the provisions of that bond will govern over any inconsistent provisions of Paragraphs 16.02.B and 16.02.D.

16.03 Owner May Terminate for Convenience

- A. Upon 7 days' written notice to Contractor and Engineer, Owner may, without cause and without prejudice to any other right or remedy of Owner, terminate the Contract. In such case, Contractor shall be paid for (without duplication of any items):
 - 1. completed and acceptable Work executed in accordance with the Contract Documents prior to the effective date of termination, including fair and reasonable sums for overhead and profit on such Work;
 - expenses sustained prior to the effective date of termination in performing services and furnishing labor, materials, or equipment as required by the Contract Documents in connection with uncompleted Work, plus fair and reasonable sums for overhead and profit on such expenses; and
 - 3. other reasonable expenses directly attributable to termination, including costs incurred to prepare a termination for convenience cost proposal.
- B. Contractor shall not be paid for any loss of anticipated profits or revenue, post-termination overhead costs, or other economic loss arising out of or resulting from such termination.

16.04 Contractor May Stop Work or Terminate

- A. If, through no act or fault of Contractor, (1) the Work is suspended for more than 90 consecutive days by Owner or under an order of court or other public authority, or (2) Engineer fails to act on any Application for Payment within 30 days after it is submitted, or (3) Owner fails for 30 days to pay Contractor any sum finally determined to be due, then Contractor may, upon 7 days' written notice to Owner and Engineer, and provided Owner or Engineer do not remedy such suspension or failure within that time, terminate the contract and recover from Owner payment on the same terms as provided in Paragraph 16.03.
- B. In lieu of terminating the Contract and without prejudice to any other right or remedy, if Engineer has failed to act on an Application for Payment within 30 days after it is submitted, or Owner has failed for 30 days to pay Contractor any sum finally determined to be due, Contractor may, 7 days after written notice to Owner and Engineer, stop the Work until payment is made of all such amounts due Contractor, including interest thereon. The

EJCDC[®] C-700, Standard General Conditions of the Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

provisions of this paragraph are not intended to preclude Contractor from submitting a Change Proposal for an adjustment in Contract Price or Contract Times or otherwise for expenses or damage directly attributable to Contractor's stopping the Work as permitted by this paragraph.

ARTICLE 17—FINAL RESOLUTION OF DISPUTES

- 17.01 Methods and Procedures
 - A. *Disputes Subject to Final Resolution*: The following disputed matters are subject to final resolution under the provisions of this article:
 - 1. A timely appeal of an approval in part and denial in part of a Claim, or of a denial in full, pursuant to Article 12; and
 - 2. Disputes between Owner and Contractor concerning the Work, or obligations under the Contract Documents, that arise after final payment has been made.
 - B. *Final Resolution of Disputes*: For any dispute subject to resolution under this article, Owner or Contractor may:
 - 1. elect in writing to invoke the dispute resolution process provided for in the Supplementary Conditions;
 - 2. agree with the other party to submit the dispute to another dispute resolution process; or
 - 3. if no dispute resolution process is provided for in the Supplementary Conditions or mutually agreed to, give written notice to the other party of the intent to submit the dispute to a court of competent jurisdiction.

ARTICLE 18—MISCELLANEOUS

- 18.01 Giving Notice
 - A. Whenever any provision of the Contract requires the giving of written notice to Owner, Engineer, or Contractor, it will be deemed to have been validly given only if delivered:
 - 1. in person, by a commercial courier service or otherwise, to the recipient's place of business;
 - 2. by registered or certified mail, postage prepaid, to the recipient's place of business; or
 - 3. by e-mail to the recipient, with the words "Formal Notice" or similar in the e-mail's subject line.
- 18.02 *Computation of Times*
 - A. When any period of time is referred to in the Contract by days, it will be computed to exclude the first and include the last day of such period. If the last day of any such period falls on a Saturday or Sunday or on a day made a legal holiday by the law of the applicable jurisdiction, such day will be omitted from the computation.

18.03 Cumulative Remedies

- A. The duties and obligations imposed by these General Conditions and the rights and remedies available hereunder to the parties hereto are in addition to, and are not to be construed in any way as a limitation of, any rights and remedies available to any or all of them which are otherwise imposed or available by Laws or Regulations, by special warranty or guarantee, or by other provisions of the Contract. The provisions of this paragraph will be as effective as if repeated specifically in the Contract Documents in connection with each particular duty, obligation, right, and remedy to which they apply.
- 18.04 Limitation of Damages
 - A. With respect to any and all Change Proposals, Claims, disputes subject to final resolution, and other matters at issue, neither Owner nor Engineer, nor any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, shall be liable to Contractor for any claims, costs, losses, or damages sustained by Contractor on or in connection with any other project or anticipated project.

18.05 No Waiver

- A. A party's non-enforcement of any provision will not constitute a waiver of that provision, nor will it affect the enforceability of that provision or of the remainder of this Contract.
- 18.06 Survival of Obligations
 - A. All representations, indemnifications, warranties, and guarantees made in, required by, or given in accordance with the Contract, as well as all continuing obligations indicated in the Contract, will survive final payment, completion, and acceptance of the Work or termination of the Contract or of the services of Contractor.
- 18.07 Controlling Law
 - A. This Contract is to be governed by the law of the state in which the Project is located.

18.08 Assignment of Contract

A. Unless expressly agreed to elsewhere in the Contract, no assignment by a party to this Contract of any rights under or interests in the Contract will be binding on the other party without the written consent of the party sought to be bound; and, specifically but without limitation, money that may become due and money that is due may not be assigned without such consent (except to the extent that the effect of this restriction may be limited by law), and unless specifically stated to the contrary in any written consent to an assignment, no assignment will release or discharge the assignor from any duty or responsibility under the Contract.

18.09 Successors and Assigns

A. Owner and Contractor each binds itself, its successors, assigns, and legal representatives to the other party hereto, its successors, assigns, and legal representatives in respect to all covenants, agreements, and obligations contained in the Contract Documents.

18.10 Headings

A. Article and paragraph headings are inserted for convenience only and do not constitute parts of these General Conditions.

SUPPLEMENTARY CONDITIONS OF THE CONSTRUCTION CONTRACT

TABLE OF CONTENTS

Article 1— Definitions and Terminology1
Article 2— Preliminary Matters1
Article 3— Contract Documents: Intent, Requirements, Reuse5
Article 4— Commencement and Progress of the Work6
Article 5— Site, Subsurface and Physical Conditions, Hazardoud Environmental Conditions
Article 6— Bonds and Insurance
Article 7— Contractor's Responsibilities
Article 8— Other Work at the Site
Article 9— Owner's Responsibilities19
Article 10— Engineer's Status During Construction19
Article 11— Changes to the Contract19
Article 12— Claims20
Article 13— Cost of Work; Allowances, Unit Price Work21
Article 14— Tests and Inspections; Correction, Removal, or Accceptance of Defective Work22
Article 15— Payments to Contractor, Set Offs; Completions; Correction Period22
Article 16— Suspension of Work and Termination25
Article 17— Final Resolutions of Disputes
Article 18— Miscellaneous
Exhibit A— Software Requirements for Electronic Document Exchange1
Exhibit B— Foreseeable Bad Weather Days1
Exhibit C— Geotechnical Baseline Report Supplement to the Supplementary Conditions1

SUPPLEMENTARY CONDITIONS OF THE CONSTRUCTION CONTRACT

These Supplementary Conditions amend or supplement EJCDC[®] C-700, Standard General Conditions of the Construction Contract (2018). The General Conditions remain in full force and effect except as amended.

The terms used in these Supplementary Conditions have the meanings stated in the General Conditions. Additional terms used in these Supplementary Conditions have the meanings stated below, which are applicable to both the singular and plural thereof.

The address system used in these Supplementary Conditions is the same as the address system used in the General Conditions, with the prefix "SC" added—for example, "Paragraph SC-4.05."

ARTICLE 1— DEFINITIONS AND TERMINOLOGY

1.01 Defined Terms

- SC-1.01 Amend Paragraph 1.01.A.9 by striking out "challenging a set off against payments due; or seeking other relief with respect to the terms of the Contract".
- SC-1.01 Amend Paragraph 1.01.A.50 to read as follows:
 - 50. Work Change Directive—A written directive to Contractor issued on or after the Effective Date of the Contract, signed by Owner and recommended by Engineer, ordering an addition, deletion, or revision in the Work. A Work Change Directive will not change the Contract Price or the Contract Times but is evidence that the parties expect that the change ordered or documented by a Work Change Directive will be incorporated in a subsequently issued Change Order following negotiations by the parties as to its effect, if any, on the Contract Price or Contract Times.

ARTICLE 2— PRELIMINARY MATTERS

- 2.01 Delivery of Bonds and Evidence of Insurance
- SC-2.01 Delete Paragraphs 2.01.B. and C. in their entirety and insert the following in their place:
 - B. Evidence of Contractor's Insurance: When Contractor delivers the signed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner copies of the policies (including all endorsements, and identification of applicable self-insured retentions and deductibles) of insurance required to be provided by Contractor in this Contract.
 - C. *Evidence of Owner's Insurance:* After receipt from Contractor of the signed counterparts of the Agreement and all required bonds and insurance documentation, Owner shall promptly deliver to Contractor copies of the policies of insurance to be provided by Owner in this Contract (if any).
- 2.02 *Copies of Documents*

- SC-2.02 Deleted.
- SC-2.02 Delete Paragraph 2.02.A in its entirety and insert the following new paragraph in its place:
 - A. Owner shall furnish to Contractor five (5) printed copies of conformed Contract Documents incorporating and integrating all Addenda and any amendments negotiated prior to the Effective Date of the Contract (including one fully signed counterpart of the Agreement), and one copy in electronic portable document format (PDF). Additional printed copies of the conformed Contract Documents will be furnished upon request at the cost of reproduction.
- 2.06 Electronic Transmittals
- SC-2.06 Delete Paragraphs 2.06.B and 2.06.C in their entirety and insert the following in their place:
 - B. *Electronic Documents Protocol:* The parties shall conform to the following provisions in Paragraphs 2.06.B and 2.06.C, together referred to as the Electronic Documents Protocol ("EDP" or "Protocol") for exchange of electronic transmittals.
 - 1. Basic Requirements
 - a. To the fullest extent practical, the parties agree to and will transmit and accept Electronic Documents in an electronic or digital format using the procedures described in this Protocol. Use of the Electronic Documents and any information contained therein is subject to the requirements of this Protocol and other provisions of the Contract.
 - b. The contents of the information in any Electronic Document will be the responsibility of the transmitting party.
 - c. Electronic Documents as exchanged by this Protocol may be used in the same manner as the printed versions of the same documents that are exchanged using non-electronic format and methods, subject to the same governing requirements, limitations, and restrictions, set forth in the Contract Documents.
 - d. Except as otherwise explicitly stated herein, the terms of this Protocol will be incorporated into any other agreement or subcontract between a party and any third party for any portion of the Work on the Project, or any Project-related services, where that third party is, either directly or indirectly, required to exchange Electronic Documents with a party or with Engineer. Nothing herein will modify the requirements of the Contract regarding communications between and among the parties and their subcontractors and consultants.
 - e. When transmitting Electronic Documents, the transmitting party makes no representations as to long term compatibility, usability, or readability of the items resulting from the receiving party's use of software application packages, operating systems, or computer hardware differing from those established in this Protocol.
 - f. Nothing herein negates any obligation 1) in the Contract to create, provide, or maintain an original printed record version of Drawings and Specifications, signed and sealed according to applicable Laws and Regulations; 2) to comply with any applicable Law or Regulation governing the signing and sealing of design documents or the signing and electronic transmission of any other documents; or 3) to comply with the notice requirements of Paragraph 18.01 of the General Conditions.

2. System Infrastructure for Electronic Document Exchange

- a. Each party will provide hardware, operating system(s) software, internet, e-mail, and large file transfer functions ("System Infrastructure") at its own cost and sufficient for complying with the EDP requirements. With the exception of minimum standards set forth in this EDP, and any explicit system requirements specified by attachment to this EDP, it is the obligation of each party to determine, for itself, its own System Infrastructure.
 - 1) The maximum size of an email attachment for exchange of Electronic Documents under this EDP is10 MB. Attachments larger than that may be exchanged using large file transfer functions or physical media.
 - 2) Each Party assumes full and complete responsibility for any and all of its own costs, delays, deficiencies, and errors associated with converting, translating, updating, verifying, licensing, or otherwise enabling its System Infrastructure, including operating systems and software, for use with respect to this EDP.
- b. Each party is responsible for its own system operations, security, back-up, archiving, audits, printing resources, and other Information Technology ("IT") for maintaining operations of its System Infrastructure during the Project, including coordination with the party's individual(s) or entity responsible for managing its System Infrastructure and capable of addressing routine communications and other IT issues affecting the exchange of Electronic Documents.
- c. Each party will operate and maintain industry-standard, industry-accepted, ISO-standard, commercial-grade security software and systems that are intended to protect the other party from: software viruses and other malicious software like worms, trojans, adware; data breaches; loss of confidentiality; and other threats in the transmission to or storage of information from the other parties, including transmission of Electronic Documents by physical media such as CD/DVD/flash drive/hard drive. To the extent that a party maintains and operates such security software and systems, it shall not be liable to the other party for any breach of system security.
- d. In the case of disputes, conflicts, or modifications to the EDP required to address issues affecting System Infrastructure, the parties shall cooperatively resolve the issues; but, failing resolution, the Owner is authorized to make and require reasonable and necessary changes to the EDP to effectuate its original intent. If the changes cause additional cost or time to Contractor, not reasonably anticipated under the original EDP, Contractor may seek an adjustment in price or time under the appropriate process in the Contract.
- e. Each party is responsible for its own back-up and archive of documents sent and received during the term of the contract under this EDP, unless this EDP establishes a Project document archive, either as part of a mandatory Project website or other communications protocol, upon which the parties may rely for document archiving during the specified term of operation of such Project document archive. Further, each party remains solely responsible for its own post-Project back-up and archive of Project documents after the term of the Contract, or after termination of the

Project document archive, if one is established, for as long as required by the Contract and as each party deems necessary for its own purposes.

- f. If a receiving party receives an obviously corrupted, damaged, or unreadable Electronic Document, the receiving party will advise the sending party of the incomplete transmission.
- g. The parties will bring any non-conforming Electronic Documents into compliance with the EDP. The parties will attempt to complete a successful transmission of the Electronic Document or use an alternative delivery method to complete the communication.
- C. Software Requirements for Electronic Document Exchange; Limitations
 - 1. Each party will acquire the software and software licenses necessary to create and transmit Electronic Documents and to read and to use any Electronic Documents received from the other party (and if relevant from third parties), using the software formats required in this section of the EDP.
 - a. Prior to using any updated version of the software required in this section for sending Electronic Documents to the other party, the originating party will first notify and receive concurrence from the other party for use of the updated version or adjust its transmission to comply with this EDP.
 - 2. The parties agree not to intentionally edit, reverse engineer, decrypt, remove security or encryption features, or convert to another format for modification purposes any Electronic Document or information contained therein that was transmitted in a software data format, including Portable Document Format (PDF), intended by sender not to be modified, unless the receiving party obtains the permission of the sending party or is citing or quoting excerpts of the Electronic Document for Project purposes.
 - 3. Software and data formats for exchange of Electronic Documents will conform to the requirements set forth in Exhibit A to this EDP, including software versions, if listed.
- SC-2.06 Supplement Paragraph 2.06 of the General Conditions by adding the following paragraph:
 - D. Requests by Contractor for Electronic Documents in Other Formats
 - 1. Release of any Electronic Document versions of the Project documents in formats other than those identified in the Electronic Documents Protocol (if any) or elsewhere in the Contract will be at the sole discretion of the Owner.
 - 2. To extent determined by Owner, in its sole discretion, to be prudent and necessary, release of Electronic Documents versions of Project documents and other Project information requested by Contractor ("Request") in formats other than those identified in the Electronic Documents Protocol (if any) or elsewhere in the Contract will be subject to the provisions of the Owner's response to the Request, and to the following conditions to which Contractor agrees:
 - a. The content included in the Electronic Documents created by Engineer and covered by the Request was prepared by Engineer as an internal working document for Engineer's purposes solely, and is being provided to Contractor on an "AS IS" basis without any warranties of any kind, including, but not limited to any implied warranties of fitness for any purpose. As such, Contractor is advised and

acknowledges that the content may not be suitable for Contractor's application, or may require substantial modification and independent verification by Contractor. The content may include limited resolution of models, not-to-scale schematic representations and symbols, use of notes to convey design concepts in lieu of accurate graphics, approximations, graphical simplifications, undocumented intermediate revisions, and other devices that may affect subsequent reuse.

- b. Electronic Documents containing text, graphics, metadata, or other types of data that are provided by Engineer to Contractor under the request are only for convenience of Contractor. Any conclusion or information obtained or derived from such data will be at the Contractor's sole risk and the Contractor waives any claims against Engineer or Owner arising from use of data in Electronic Documents covered by the Request.
- c. Contractor shall indemnify and hold harmless Owner and Engineer and their subconsultants from all claims, damages, losses, and expenses, including attorneys' fees and defense costs arising out of or resulting from Contractor's use, adaptation, or distribution of any Electronic Documents provided under the Request.
- d. Contractor agrees not to sell, copy, transfer, forward, give away or otherwise distribute this information (in source or modified file format) to any third party without the direct written authorization of Engineer, unless such distribution is specifically identified in the Request and is limited to Contractor's subcontractors. Contractor warrants that subsequent use by Contractor's subcontractors complies with all terms of the Contract Documents and Owner's response to Request.
- 3. In the event that Owner elects to provide or directs the Engineer to provide to Contractor any Contractor-requested Electronic Document versions of Project information that is not explicitly identified in the Contract Documents as being available to Contractor, the Owner shall be reimbursed by Contractor on an hourly basis (at \$100 [number] per hour) for any engineering costs necessary to create or otherwise prepare the data in a manner deemed appropriate by Engineer.

ARTICLE 3— CONTRACT DOCUMENTS: INTENT, REQUIREMENTS, REUSE

- 3.01 Intent
- SC-3.01 Delete Paragraph 3.01.C in its entirety.
- 3.03 Reporting and Resolving Discrepencies
- SC-3.03 Amend Paragraph 3.03.A.3 to read as follows:
 - 3. Contractor shall not be liable to Owner or Engineer for failure to report any conflict, error, ambiguity, or discrepancy in the Contract Documents unless Contractor knew or reasonably should have known there of, or Contractor failed to perform its obligations under the Instructions to Bidders.
- SC-3.03 Amend Paragraph 3.03.A by adding the following subparagraph.
 - 4. In addition to its obligations under the Instructions to Bidders, if Contractor proceeds with work that Contractor had actual knowledge or should have known that a conflict, error, ambiguity, or discrepancy existed as indicated above, correction of work constructed

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

EJCDC® C-800, Supplementary Conditions of the Construction Contract.

without such notification to Engineer shall be at Contractor's expense (except in an emergency as authorized by Paragraph 7.15).

- SC-3.03 Amend Paragraph 3.03.B.1 by striking out "prepared by or for Engineer".
- 3.04 Requirements of the Contract Documents
- SC-3.04 Amend the last sentence of Paragraph 3.04.A to read as "Owner will be the initial interpreter of the requirements of the Contract Documents, and judge of the acceptability of the Work."
- SC-3.04 Amend Paragraph 3.04.B to read as follows:
 - B. Engineer will, with reasonable promptness, render a written clarification, interpretation, or decision on the issue submitted, or initiate an amendment or supplement to the Contract Documents. Engineer's written clarification, interpretation, or decision will be final and binding on Contractor upon the Owner's acceptance of the Engineer's ruling, unless it appeals by submitting a Change Proposal and that Change Proposal is accepted by both the Engineer and Owner.

ARTICLE 4— COMMENCEMENT AND PROGRESS OF THE WORK

4.05 Delays in Contractor's Progress

SC-4.05 Paragraph is mandatory for WWD projects.

- SC-4.05 Amend Paragraph 4.05.C by striking out "Contractor shall" and replacing with "Contractor may".
- SC-4.05 Amend Paragraph 4.05.C by adding the following subparagraphs:
 - 5. Weather-Related Delays
 - a. If "abnormal weather conditions" as set forth in Paragraph 4.05.C.2 of the General Conditions are the basis for a request for an equitable adjustment in the Contract Times, such request must be documented by data substantiating each of the following: 1) that weather conditions were abnormal for the period of time in which the delay occurred, 2) that such weather conditions could not have been reasonably anticipated, and 3) that such weather conditions had an adverse effect on the Work as scheduled. Abnormal weather conditions are hereby defined as a condition in which the Contractor's workday production is reduced by more than 50 percent of items on the Critical Path due to weather or season conditions. Extreme or unusual weather that is typical for a given region, elevation, or season should not be considered abnormal weather conditions. Requests for time extensions due to abnormal weather conditions will be submitted to the Engineer within five days of the end of the abnormal weather condition event. It is the responsibility of the Contractor to provide the information listed in SC 4.05.C.5.b.
 - b. The existence of abnormal weather conditions will be determined on a month-bymonth basis in accordance with the following:
 - 1) Every workday on which one or more of the following conditions exist will be considered a "bad weather day":
 - i) Total precipitation (as rain equivalent) occurring between 7:00 p.m. on the preceding day (regardless of whether such preceding day is a

Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

EJCDC[®] C-800, Supplementary Conditions of the Construction Contract.

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

workday) through 7:00 p.m. on the workday in question equals or exceeds [threshold precipitation quantity].5-inch of precipitation (as rain equivalent, based on the snow/rain conversion indicated in the table entitled Foreseeable Bad Weather Days; such table is hereby incorporated in this SC-4.05.C by reference.

- Ambient outdoor air temperature at 11:00 a.m. is equal to or less than the following low temperature threshold: 20 degrees Fahrenheit; or, at 3:00 p.m. the ambient outdoor temperature is equal to or greater than the following high temperature threshold: 100 degrees Fahrenheit.
- Determination of actual bad weather days during performance of the Work will be based on the weather records measured and recorded by National Oceanic and Atmospheric Administration (NOAA) weather monitoring station at Wilmington Air Park (ILN).
- 3) Contractor shall anticipate the number of foreseeable bad weather days per month indicated in the table in Exhibit B—Foreseeable Bad Weather Days.
- 4) In each month, every bad weather day exceeding the number of foreseeable bad weather days established in the table in Exhibit B —Foreseeable Bad Weather Days will be considered as "abnormal weather conditions." The existence of abnormal weather conditions will not relieve Contractor of the obligation to demonstrate and document that delays caused by abnormal weather are specific to the planned work activities or that such activities thus delayed were on Contractor's then-current Progress Schedule's critical path for the Project.
- c. Extreme or unusual weather that is typical for a given region, elevation, or season should not be considered abnormal weather conditions. Requests for time extensions due to abnormal weather conditions will be submitted to the Engineer within five (5) days or the end of the abnormal weather event. It is the Contractor's responsibility to provide the information listed in SC-4.05.C.b.
- d. The Contractor's accepted progress Schedule must reflect the anticipated weather delays as presented in the table in Exhibit B. The table applies to the duration between contract execution and original completion date. Extensions for abnormal weather condition delays beyond the original completion date will be the actual workdays lost due to abnormal weather conditions.
- e. The Owner/Engineer will not consider weekends and holidays as lost workdays unless the Owner directs the Contractor to work those days.
- SC-4.05 Amend Paragraph 4.05.D.2 to read as follows:
 - 2. Contractor shall not be entitled to an adjustment in Contract Price for any delay, disruption, or interference if such delay is concurrent with a delay, disruption, or interference caused by or within the control of Contractor.

ARTICLE 5— SITE, SUBSURFACE AND PHYSICAL CONDITIONS, HAZARDOUS ENVIRONMENTAL CONDITIONS

- 5.01 Availability of Lands
- SC-5.01 Amend Paragraph 5.01.B to read as follows:
 - B. Upon reasonable request, Owner shall furnish Contractor with a Notice of Commencement prepared for the Project, conforming to the provisions of Ohio Revised Code Section 1311.252.
- 5.02 Use of Site and Other Areas
- SC-5.02 Amend Paragraph 5.02.A.2 to read as follows:
 - 2. If a damage or injury claim is made by the owner or occupant of any such land or area because of the performance of the Work, or because of other actions or conduct of the Contractor or those for which Contractor is responsible, Contractor shall (a) take immediate corrective or remedial action as required by Paragraph 7.13, or otherwise; (b) promptly attempt to settle the claim as to all parties through negotiations with such owner or occupant, or otherwise resolve the claim by dispute resolution proceeding, or in a court of competent jurisdiction located in Fairfield County, Ohio; and (c) to the fullest extent permitted by Laws and Regulations, indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from and against any such claim, and against all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or dispute resolution costs) arising out of or relating to any claim or action, legal or equitable, brought by any such owner or occupant against Owner, Engineer, or any other party indemnified hereunder to the extent caused directly or indirectly, in whole or in part by, or based upon, Contractor's performance of the Work, or because of other actions or conduct of the Contractor or those for which Contractor is responsible.
- 5.03 Subsurface and Physical Conditions
- SC-5.03 Add the following new paragraphs immediately after Paragraph 5.03.D:
 - E. The following table lists the reports of explorations and tests of subsurface conditions at or adjacent to the Site that contain Technical Data, and specifically identifies the Technical Data in the report upon which Contractor may rely:

Report Title	Date of Report	Technical Data
lone		

F. The following table lists the drawings of existing physical conditions at or adjacent to the Site, including those drawings depicting existing surface or subsurface structures at or adjacent to

EJCDC® C-800, Supplementary Conditions of the Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

the Site (except Underground Facilities), that contain Technical Data, and specifically identifies the Technical Data upon which Contractor may rely:

Drawings Title	Date of Drawings	Technical Data
TUSSING ROAD WATER TREATMENT FACILITY IMPROVEMENTS	07/07/2003	Record Drawings

- G. Contractor may examine copies of reports and drawings identified in SC-5.03.E and SC-5.03.F that were not included with the Bidding Documents at Fairfield County Utilities building during regular business hours, or may request copies from Owner.
- 5.04 Differing Subsurface or Physical Conditions
- SC-5.04 Amend Paragraph 5.04.A by striking out "promptly" and replacing with "within 48 hours".

SC-5.04 Amend Paragraph 5.04.C by striking out "Owner shall" and replacing with "Owner may".

SC-5.04 Amend Paragraph 5.04.E.4 by striking out "30 days" and replacing with "10 days".

- 5.05 Underground Facilities
- SC-5.05 Amend Paragraph 5.05.D by striking out "Owner shall" and replacing with "Owner may".
- SC-5.05 Amend Paragraph 5.05.F.1 to read as follows:
 - Contractor may be entitled to an equitable adjustment in the Contract Price or Contract Times, to the extent that any existing Underground Facility at the Site that was not shown or indicated on the Drawings, or was not shown or indicated with reasonable accuracy, or any related delay, disruption, or interference, causes an increase or decrease in Contractor's cost of, or time required for, performance of the Work; subject, however, to all of the following:
- 5.06 Hazardous Environmental Conditions
- SC-5.06 Modify Paragraph 5.06.A to read:
 - A. Reports and Drawings: The Supplementary Conditions/Agreement identifies:

1, 0, 2020

- SC-5.06 Add the following new paragraphs immediately after Paragraph 5.06.A.3:
 - 4. The following table lists the reports known to Owner relating to Hazardous Environmental Conditions at or adjacent to the Site, and the Technical Data (if any) upon which Contractor may rely:

Report Title	Date of Report	Technical Data
None		

5. The following table lists the drawings known to Owner relating to Hazardous Environmental Conditions at or adjacent to the Site, and Technical Data (if any) contained in such Drawings upon which Contractor may rely:

Drawings Title	Date of Drawings	Technical Data
lone		

- SC-5.06 Amend Paragraph 5.06.G to read as follows:
 - G. If Owner and Contractor cannot agree as to entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times, as a result of such Work stoppage, such special conditions under which Work is agreed to be resumed by Contractor, or any costs or expenses incurred in response to the Hazardous Environmental Condition, then within 10 days of Owner's written notice regarding the resumption of Work, Contractor may submit a Change Proposal for the Owner's consideration, or the Owner may impose a set-off. Entitlement to any such adjustment is subject to the provisions of Paragraphs 4.05.D, 4.05.E, 11.07, and 11.08.
- SC-5.06 Delete Paragraph 5.06.1 in its entirety.
- SC-5.06 Amend Paragraph 5.06.J by striking out "arbitration or other" within the subparagraph.

ARTICLE 6— BONDS AND INSURANCE

- 6.01 *Performance, Payment, and Other Bonds*
- SC-6.01 Add the following paragraphs immediately after Paragraph 6.01.A:
 - 1. *Required Performance Bond Form:* The performance bond that Contractor furnishes will be in the form of EJCDC[®] C-610, Performance Bond (2010, 2013, or 2018 edition).
 - 2. *Required Payment Bond Form:* The payment bond that Contractor furnishes will be in the form of EJCDC[®] C-615, Payment Bond (2010, 2013, or 2018 edition).
- SC-6.01 Amend Paragraph 6.01.F by striking out "Owner shall" and replacing with "Owner may".
- SC-6.01 Add the following paragraph immediately after Paragraph 6.01.H:
 - 1. Material Default or Termination. If the Owner notifies the Contractor's surety that the Contractor is in material default, the surety will complete its investigation of the claimed

EJCDC[®] C-800, Supplementary Conditions of the Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

material default within 21 days. The surety is advised to start looking for a replacement contractor upon notice of material default. As part of its investigation, the surety shall promptly visit the offices of the Contractor, Engineer, and Owner to inspect and copy the available Project records. The Owner, Engineer, and Contractor, upon written request by the surety, shall make such records available during regular business hours for such inspection and copying. The Owner and Engineer's making such records available as provided herein shall satisfy the Owner's obligation to the surety to furnish documents for the investigation. The surety will provide the Owner with the results of its investigation, including any written report or documents. If the Owner terminates the Contract and the surety proposes to takeover the Work, the surety shall do so no later than the later of the expiration of the 21day investigation period or 10 days after the date the Owner terminates the Contract, whichever is later. If the Owner terminates the Contract, and the surety proposes to provide a replacement contractor, the replacement contractor shall be fully capable of performing the Work in accordance with the Contract Documents. If the Contractor is terminated for cause, the replacement contractor shall not be the Contractor or a contractor comprised of mostly Contractor's employees, unless the Owner agrees in writing. In the event the Surety takes over the Project, the surety's obligation shall not be limited to the penal sum of the Bond. If the surety does not propose an acceptable contractor as required by this Paragraph 5.01.D, the Owner may complete the Work by such means as it deems appropriate. In the event the Owner agrees to accept a replacement contractor, the replacement contractor shall furnish its own bond for the replacement contractor's scope of work, and neither the Contractor nor the surety shall be relieved of their obligations under the Contract Documents. This Paragraph 5.01.D is in addition to any other rights of the Owner under the Contract Documents and is not intended to create any rights of the surety, including but not limited to the right to take over the Contractor's obligations. In the event of the Contractor's termination and if the surety does not takeover the Work as provided in this Paragraph 5.01.D., the Owner may take possession of and use all materials, facilities and equipment at the Project Site or stored off-site for which Owner has paid in whole or in part.

SC-6.01 Deleted.

6.02 Insurance—General Provisions

- SC-6.02 Amend Paragraph 6.02.E by striking out "Owner shall" and replacing with "Owner may" in both locations.
- SC-6.02 Amend Paragraph 6.02.J by striking out "Owner may" and replacing with "Owner shall".
- 6.03 Contractor's Insurance
- SC-6.03 Supplement Paragraph 6.03 with the following provisions after Paragraph 6.03.C:
 - D. Workers' Compensation and Employer's Liability: Contractor shall purchase and maintain workers' compensation and employer's liability insurance, including, as applicable, United States Longshoreman and Harbor Workers' Compensation Act, Jones Act, stop-gap employer's liability coverage for monopolistic states, and foreign voluntary workers' compensation (from available sources, notwithstanding the jurisdictional requirement of Paragraph 6.02.B of the General Conditions).

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

4/	6/	20	23	
4/	6/	20	23	

Workers' Compensation and Related Policies	Policy limits of not less than:
Workers' Compensation	
State	Statutory
Applicable Federal (e.g., Longshoreman's)	Statutory
Foreign voluntary workers' compensation (employer's responsibility coverage), if applicable	Statutory
Employer's Liability	
Each accident	\$500,000
Each employee	\$500,000
Policy limit	\$500,000
Stop-gap Liability Coverage	
For work performed in monopolistic states, stop-gap liability coverage must be endorsed to either the worker's compensation or commercial general liability policy with a minimum limit of:	\$1,000,000

- Commercial General Liability-Claims Covered: Contractor shall purchase and maintain F. commercial general liability insurance, covering all operations by or on behalf of Contractor, on an occurrence basis, against claims for:
 - 1. damages because of bodily injury, sickness or disease, or death of any person other than Contractor's employees,
 - 2. damages insured by reasonably available personal injury liability coverage, and
 - damages because of injury to or destruction of tangible property wherever located, 3. including loss of use resulting therefrom.
- G. Commercial General Liability-Form and Content: Contractor's commercial liability policy must be written on a 1996 (or later) Insurance Services Organization, Inc. (ISO) commercial general liability form (occurrence form) and include the following coverages and endorsements:
 - 1. Products and completed operations coverage.
 - Such insurance must be maintained for three years after final payment. a.
 - b. Contractor shall furnish Owner and each other additional insured (as identified in the Supplementary Conditions or elsewhere in the Contract) evidence of continuation of such insurance at final payment and three years thereafter.
 - Blanket contractual liability coverage, including but not limited to coverage of 2. Contractor's contractual indemnity obligations in Paragraph 7.18.
 - Severability of interests and no insured-versus-insured or cross-liability exclusions. 3.
 - Underground, explosion, and collapse coverage. 4.
 - 5. Personal injury coverage.
 - Additional insured endorsements that include both ongoing operations and products 6. and completed operations coverage through ISO Endorsements CG 20 10 10 01 and CG 20 37 10 01 (together). If Contractor demonstrates to Owner that the specified ISO

Ι.

endorsements are not commercially available, then Contractor may satisfy this requirement by providing equivalent endorsements.

- For design professional additional insureds, ISO Endorsement CG 20 32 07 04 "Additional Insured—Engineers, Architects or Surveyors Not Engaged by the Named Insured" or its equivalent.
- H. *Commercial General Liability—Excluded Content:* The commercial general liability insurance policy, including its coverages, endorsements, and incorporated provisions, must not include any of the following:
 - 1. Any modification of the standard definition of "insured contract" (except to delete the railroad protective liability exclusion if Contractor is required to indemnify a railroad or others with respect to Work within 50 feet of railroad property).
 - 2. Any exclusion for water intrusion or water damage.

Commercial General Liability—Minimum Policy Limits

- 3. Any provisions resulting in the erosion of insurance limits by defense costs other than those already incorporated in ISO form CG 00 01.
- 4. Any exclusion of coverage relating to earth subsidence or movement.
- 5. Any exclusion for the insured's vicarious liability, strict liability, or statutory liability (other than worker's compensation).
- 6. Any limitation or exclusion based on the nature of Contractor's work.
- 7. Any professional liability exclusion broader in effect than the most recent edition of ISO form CG 22 79.
- Commercial General LiabilityPolicy limits of not
less than:General Aggregate\$2,000,000Products—Completed Operations Aggregate\$1,000,000Personal and Advertising Injury\$1,000,000Bodily Injury and Property Damage—Each Occurrence\$2,000,000
- J. Automobile Liability: Contractor shall purchase and maintain automobile liability insurance for damages because of bodily injury or death of any person or property damage arising out of the ownership, maintenance, or use of any motor vehicle. The automobile liability policy must be written on an occurrence basis.

Automobile Liability	Policy limits of not less than:	
Combined Single Limit		
Combined Single Limit (Bodily Injury and Property Damage)	\$1,000,000	

K. Umbrella or Excess Liability: Contractor shall purchase and maintain umbrella or excess liability insurance written over the underlying employer's liability, commercial general liability, and automobile liability insurance described in the Paragraphs above. The coverage afforded must be at least as broad as that of each and every one of the underlying policies.

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

4/6/2023	
----------	--

Excess or Umbrella Liability	Policy limits of not less than:	
Each Occurrence	\$5,000,000	
General Aggregate	\$5,000,000	

- Using Umbrella or Excess Liability Insurance to Meet CGL and Other Policy Limit L. Requirements: Contractor may meet the policy limits specified for employer's liability, commercial general liability, and automobile liability through the primary policies alone, or through combinations of the primary insurance policy's policy limits and partial attribution of the policy limits of an umbrella or excess liability policy that is at least as broad in coverage as that of the underlying policy, as specified herein. If such umbrella or excess liability policy was required under this Contract, at a specified minimum policy limit, such umbrella or excess policy must retain a minimum limit of \$5,000,000 after accounting for partial attribution of its limits to underlying policies, as allowed above.
- M. The following provisions shall also apply to the insurance provided by the Contractor:
 - 1. Contractor's insurance shall be primary and non-contributory.
 - Insurance policies shall be written on an occurrence basis only. 2.
 - The Contractor shall require all Subcontractors to provide Workers' Compensation, CGL, 3. and Automobile Liability Insurance with the same minimum limits specified herein, unless the Owner agrees to a lesser amount.
 - Owner shall be named as a certificate holder on the policies of insurance maintained by 4. Contractor. The Contractor shall provide each additional insured with a certificate of insurance.
 - The additional insured endorsement shall be ISO 20101001 and CG 20371001 or their 5. equivalents so that Completed Operations liability extends to the additional insureds after the completion of the Project.
- Builder's Risk and Other Property Insurance 6.04
- Amend Paragraph 6.04.C by striking out "Owner shall" and replacing with "Owner may". SC-6.04
- Supplement Paragraph 6.04 of the General Conditions with the following provisions: SC-6.04
 - Builder's Risk Requirements: The builder's risk insurance must: F.
 - be written on a builder's risk "all risk" policy form that at a minimum includes insurance 1. for physical loss or damage to the Work, temporary buildings, falsework, and materials and equipment stored and in transit, and must not exclude the coverage of the following risks: fire; windstorm; hail; flood; earthquake, volcanic activity, and other earth movement; lightning; riot; civil commotion; terrorism; vehicle impact; aircraft; smoke; theft; vandalism and malicious mischief; mechanical breakdown, boiler explosion, and artificially generated electric current; collapse; explosion; debris removal; demolition occasioned by enforcement of Laws and Regulations; and water damage (other than that caused by flood).
 - Such policy will include an exception that results in coverage for ensuing losses а. from physical damage or loss with respect to any defective workmanship, methods, design, or materials exclusions.

EJCDC® C-800, Supplementary Conditions of the Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

- b. If insurance against mechanical breakdown, boiler explosion, and artificially generated electric current; earthquake, volcanic activity, and other earth movement; or flood, are not commercially available under builder's risk policies, by endorsement or otherwise, such insurance will be provided through other insurance policies acceptable to Owner and Contractor.
- 2. cover, as insured property, at least the following: (a) the Work and all materials, supplies, machinery, apparatus, equipment, fixtures, and other property of a similar nature that are to be incorporated into or used in the preparation, fabrication, construction, erection, or completion of the Work, including Owner-furnished or assigned property; (b) spare parts inventory required within the scope of the Contract; and (c) temporary works which are not intended to form part of the permanent constructed Work but which are intended to provide working access to the Site, or to the Work under construction, or which are intended to provide temporary support for the Work under construction, including scaffolding, form work, fences, shoring, falsework, and temporary structures.
- 3. cover expenses incurred in the repair or replacement of any insured property (including but not limited to fees and charges of contractors, engineers, and architects).
- 4. extend to cover damage or loss to insured property while in temporary storage at the Site or in a storage location outside the Site (but not including property stored at the premises of a manufacturer or Supplier). If this coverage is subject to a sublimit, such sublimit will be a minimum of \$500,000.
- 5. extend to cover damage or loss to insured property while in transit. If this coverage is subject to a sublimit, such sublimit will be a minimum of \$250,000.
- 6. allow for the waiver of the insurer's subrogation rights, as set forth in this Contract.
- 7. allow for partial occupancy or use by Owner by endorsement, and without cancellation or lapse of coverage.
- 8. include performance/hot testing and start-up, if applicable.
- 9. be maintained in effect until the Work is complete, as set forth in Paragraph 15.06.D of the General Conditions, or until written confirmation of Owner's procurement of property insurance following Substantial Completion, whichever occurs first.
- 10 include as named insureds the Owner, Contractor, Subcontractors (of every tier), and any other individuals or entities required by this Contract to be insured under such builder's risk policy. For purposes of Paragraphs 6.04, 6.05, and 6.06 of the General Conditions, and this and all other corresponding Supplementary Conditions, the parties required to be insured will be referred to collectively as "insureds."
- SC-6.04 Supplement Paragraph 6.04 of the General Conditions with the following provision:
 - G. Coverage for Completion Delays: The builder's risk policy will include, for the benefit of Owner, loss of revenue and soft cost coverage for losses arising from delays in completion that result from covered physical losses or damage. Such coverage will include, without limitation, fixed expenses and debt service for a minimum of 12 months with a maximum deductible of 30 days, compensation for loss of net revenues, rental costs, and attorneys' fees and engineering or other consultants' fees, if not otherwise covered.

- SC-6.04 Supplement Paragraph 6.04 of the General Conditions with the following provisions:
 - H. Builder's Risk and Other Property Insurance Deductibles: The purchaser of any required builder's risk, installation floater, or other property insurance will be responsible for costs not covered because of the application of a policy deductible.
 - 1. The builder's risk policy (or if applicable the installation floater) will be subject to a deductible amount of no more than \$5,000 for direct physical loss in any one occurrence.
- SC-6.04 Deleted.

ARTICLE 7— CONTRACTOR'S RESPONSIBILITIES

- 7.03 Labor; Working Hours
- SC-7.03 Add the following new subparagraphs immediately after Paragraph 7.03.D:
 - 1. Regular working hours will be 7:00 AM to 4:30 PM EST, Monday through Friday.
 - 2. Observedlegal holidays are New Year's Day, Martin Luther King Day, Presidents' Day, Good Friday (Close at noon), Memorial Day, Juneteenth, Independence Day, Labor Day, Fairfield County Fair Day, Veteran's Day, Thanksgiving, Day After Thanksgiving, Christmas Eve (close at noon), Christmas Day.
- SC-7.03 Add the following new subparagraph immediately after Paragraph SC-7.03.D:

For purposes of administering the foregoing requirement, additional overtime costs are defined as 1.5 times the hourly rate.

- 7.04 Services, Materials, and Equipment
- SC-7.04 Supplement Paragraph 7.04.B by adding the following at the end of this paragraph:

Contractor warrants that all materials and equipment for which Contractor has primary responsibility for choosing are suitable and fit for the intended use of such materials and equipment and are free from defects in material, workmanship or design. The foregoing applies whether the materials or equipment are specified in the Contract Documents.

- 7.06 Substitutes
- SC-7.06 Amend Paragraph 7.06.F by striking out the last sentence: "Contractor may challenge the scope of reimbursement costs imposed under Paragraph 7.06.D, by timely submittals of a Change Proposal."
- 7.07 Concerning Subcontractors and Suppliers
- SC-7.07 Delete Paragraph 7.07.A. in its entirety and replace with the following:
 - A. Contractor shall not employ any Subcontractor, Supplier, or other individual or entity, whether initially or as a replacement, against whom Owner may have reasonable objection. Contractor shall not be required to employ any Subcontractor, Supplier, or other individual or entity to furnish or perform any of the Work against whom Contractor has reasonable objection. If Owner or Engineer after due investigations has reasonable objections to any proposed Subcontractor, Supplier, or other individual or entity, either

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

EJCDC® C-800, Supplementary Conditions of the Construction Contract.

may request Contractor submit an acceptable substitute without an increase in Contract Price.

- SC-7.07 Delete Paragraph 7.07.F. in its entirety and replace with the following:
 - F. If Owner requires the replacement of any Subcontractor or Supplier retained by Contractor to perform any part of the Work, then Contractor may be entitled to an adjustment in Contract Price or Contract Times, with respect to the replacement.
- 7.08 Patent Fees and Royalties
- SC-7.08 Delete Paragraph 7.08.B in its entirety.
- SC-7.08 Amend Paragraph 7.08.C by striking out "arbitration or other" within the subparagraph.
- 7.10 Taxes
- SC-7.10 Add a new paragraph immediately after Paragraph 7.10.A:
 - A. Owner is exempt from payment of sales and compensating use taxes of the State of Ohio and of cities and counties thereof on all materials to be incorporated into the Work.
 - 1. Owner will furnish the required certificates of tax exemption to Contractor for use in the purchase of supplies and materials to be incorporated into the Work.
 - 2. Owner's exemption does not apply to construction tools, machinery, equipment, or other property purchased by or leased by Contractor, or to supplies or materials not incorporated into the Work.
- 7.11 Laws and Regulations
- SC-7.11 Amend Paragraph 7.11.B by striking out "arbitration or other" within the subparagraph.
- SC-7.11 Delete Paragraph 7.11.C. in its entirety and replace with the following:
 - C. Owner or Contractor may give written notice to the other party of any changes after the submission of Contractor's Bid (or after the date when Contractor became bound under a negotiated contract) in Laws or Regulations having an effect on the cost or time of performance of the Work, including but not limited to changes in Laws or Regulations having an effect on procuring permits and on sales, use, value-added, consumption, and other similar taxes. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times resulting from such changes, then within 10 days of such written notice Contractor are unable to agree on entitlement to or on the amount or extent, if or on the amount or extent, if any, of any initiate a Claim. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any submit a Change Proposal, or Owner may initiate a Claim. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any such adjustment, a Claim may be made therefor as provided in Paragraph 12.05.
- 7.13 Safety and Protection
- 7.13 Safety and Protection
- SC-7.13 Amend Paragraph 7.13.D to read as follows:
 - D. All damage, injury, or loss to any property referred to in Paragraph 7.13.C.2 or 7.13.C.3 caused, directly or indirectly, in whole or in part, by Contractor, any Subcontractor, Supplier, or any other individual or entity directly or indirectly employed by any of them

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

EJCDC® C-800, Supplementary Conditions of the Construction Contract.

to perform any of the Work, or anyone for whose acts any of them may be liable, shall be remedied by Contractor at its expense.

7.17 Contractor's General Warranty and Guarantee

- SC-7.17 Amend Paragraph 7.13.B and Subparagraphs 7.17.B.1-2 to read as follows:
 - B. Owner's rights under this warranty and guarantee are in addition to, and are not limited by, Owner's rights under the correction period provisions of Paragraph 15.08.
 - 1. Owner shall give Contractor written notice of any defective Work within 1 year of the discovery that such Work is defective; and
 - 2. Such notice will be deemed the start of an event giving rise to a Claim under Paragraph 12.01.B, such that any related Claim must be brought within 30 days of the notice.

7.18 Indemnification

- SC-7.18 Delete Paragraph 7.18.A. in its entirety and replace with the following:
 - B. To the fullest extent permitted by Laws and Regulations, and in addition to any other obligations of Contractor under the Contract or otherwise, Contractor shall indemnify, defend, and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from losses, damages, costs, and judgments (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court or other dispute resolution costs) arising from third-party claims or actions relating to or resulting from the performance or furnishing of the Work, provided that any such claim, action, loss, cost, judgment or damage is attributable to bodily injury, sickness, disease, or death, to damage to or destruction of tangible property (other than the Work itself), or from a breach of the Contractor's obligations under the Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable.

ARTICLE 8— OTHER WORK AT THE SITE

- 8.02 Coordination
- SC-8.02 Deleted.
- 8.03 Legal Relationships
- SC-8.03 Amend Paragraph 8.03.A by striking out "Contractor shall" and replacing with "Contractor may" and striking out "30 days" and replacing with "10 days".
- SC-8.03 Amend Paragraph 8.03.B.1 by striking out "Owner shall" and replacing with "Owner may".

SC-8.03 Amend Paragraph 8.03.C by striking out "arbitration or other" in both locations within the subparagraph.

ARTICLE 9— OWNER'S RESPONSIBILITIES

- 9.02 *Replacement of Engineer*
- SC-9.02 Delete Paragraph 9.02.A. in its entirety and replace with the following:
 - A. Owner may at its discretion appoint an engineer to replace Engineer. The replacement engineer's status under the Contract Documents will be that of the former Engineer.
- SC-9.13 Add the following new paragraph immediately after Paragraph 9.12 of the General Conditions:
 - 9.13 Owner's Site Representative
 - A. Owner will furnish an "Owner's Site Representative" to represent Owner at the Site and assist Owner in observing the progress and quality of the Work. The Owner's Site Representative is not Engineer's consultant, agent, or employee. Owner's Site Representative will be an employee of Owner.

ARTICLE 10— ENGINEER'S STATUS DURING CONSTRUCTION

- 10.03 Resident Project Representative
- SC-10.03 Add the following new subparagraph immediately after Paragraph 10.03.A:
 - 1. On this Project, by agreement with the Owner, the Engineer will not furnish a Resident Project Representative to represent Engineer at the Site or assist Engineer in observing the progress and quality of the Work.
- SC-10.03 Deleted.

ARTICLE 11— CHANGES TO THE CONTRACT

- 11.02 Change Orders
- SC-11.02 Amend Paragraph 11.02.A by striking out "Owner shall" and replacing with "Owner may".
- SC-11.02 Delete Paragraph 11.02.A.2 in its entirety and replace with the following:
 - 2. Changes in Contract Price where the Owner and Contractor have mutually agreed to the change in price;
- 11.03 Work Change Directives
- SC-11.03 Amend Paragraph 11.03.B.1 by striking out "30 days" and replacing with "10 days".
- 11.04 Field Orders
- SC-11.04 Delete Paragraph 11.04.B in its entirety and replace with the following:
 - B. If Contractor believes that a Field Order justifies an adjustment in the Contract Price or Contract Times, the Contractor shall submit a Change Proposal no later than 7 days after receiving notice of the Field Order.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

EJCDC® C-800, Supplementary Conditions of the Construction Contract.

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 4/6/2023

11.07 Change of Contract Price

SC-11.07 Amend Paragraph 11.07.A to read as follows:

- A. The Contract Price may only be changed by a mutually agreed upon Change Order. Any Change Proposal for an adjustment in the Contract Price must comply with the provisions of Paragraph 11.09. Any Claim for an adjustment of Contract Price must comply with the provisions of Article 12.
- SC-11.07 Delete Paragraph 11.07.C. in its entirety and replace with the following:
 - C. *Contractor's Fee*: When applicable, the Contractor's fee for overhead and profit may be determined as follows:
 - 1. A mutually acceptable fixed fee.

ARTICLE 12— CLAIMS

12.01 Claims

SC-12.01 Delete Subparagraphs 12.01.A-G and add the following new subparagraphs:

A. Engineer's Decision Required: All Claims, except those waived pursuant to Paragraph _____, shall be referred to the Engineer for decision. A decision by Engineer shall be required as a condition precedent to any exercise by Owner or Contractor of any rights or remedies either may otherwise have under the Contract Documents or by Laws and Regulations in respect of such Claims.

B. *Notice*: As a condition precedent to a change in the Contract Price or the Contract Times, for each Claim the Contractor shall deliver a fully completed Statement of Claim Form, a copy of which form is a Contract Document, to the Engineer and the Owner, within 21 days of the start of the event giving rise to the Claim. The Contractor shall be responsible for substantiating its Claim. The Contractor's failure to deliver a fully completed Statement of Claim form of additional compensation, be it in time or money, arising out of the Claim or the circumstances underlying the Claim. Further, the Contractor's obligation to deliver a fully completed Statement of Claim form within such 21 day period is a material term of the Contract Documents and provides the Owner with the opportunity to mitigate its damages.

C. *Engineer's Action*: Engineer will review each Claim and, within 30 days after receipt of the last submittal of the claimant or the last submittal of the opposing party, if any, take one of the following actions in writing:

1. deny the Claim in whole or in part;

2. approve the Claim; or

3. notify the parties that the Engineer is unable to resolve the Claim if, in the Engineer's sole discretion, it would be inappropriate for the Engineer to do so. For purposes of further resolution of the Claim, such notice shall be deemed a denial.

D. In the event that Engineer does not take action on a Claim within said 30 days, the Claim shall be deemed denied.

E. Engineer's written action under Paragraph 12.01.C or denial pursuant to Paragraphs 12.01.C.3 or 12.01.D will be final and binding upon Owner and Contractor, unless Owner or Contractor commences an action in a court of exclusive jurisdiction as set forth in Article 17 within 30 days of substantial completion of the Work.

F. No Claim for an adjustment in Contract Price or Contract Times will be valid if not submitted in accordance with this Paragraph 12.01.

G. False or Fraudulent Claim: The Contractor shall not knowingly present or cause to be presented to the Owner a false or fraudulent Claim. Knowingly shall have the same meaning as in Section 3729(b) USC of the Federal False Claims Act. If the Contractor knowingly presents or causes to be presented a false or fraudulent Claim, then the Contractor shall be liable to the Owner for the same civil penalty and damages as the United States Government would be entitled to recover under such Section 3729(a) USC and shall also indemnify and hold the Owner harmless from all costs and expenses, including Owner's attorneys' and consultants' fees and expenses incurred in investigating and defending against such Claim and in pursuing the collection of such penalty, damages and fees and expenses.

H. Claim Documentation: Within ten (10) days of written request from the Owner, Contractor shall make available to Owner or its representative any books, records, or other documents in its possession or to which it has access, including but not limited to Contractor's daily logs/reports, original estimates of Work and applicable agreements, correspondence with subcontractors and suppliers, internal correspondence (including e-mail), accounting records, and other information from which the Contractor's costs may be derived. To the extent permitted by law, the Owner shall keep the Project accounting records and estimate for the Project confidential. As requested by the Owner, the Contractor shall provide such documents and information in paper copies and/or computer format (including the format of the Contractor's accounting software and/or ASCII format). The Contractor's provision of the requested documents and information shall be a condition precedent to any further proceeding under the Contract Documents or to payment of an Application for Payment. Failure to provide the requested documents shall be a material breach of the Contract, and Contractor shall indemnify Owner for all of Owner's costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to Contractor's failure to comply with this provision. If the Contractor fails to provide the requested documents, the Contractor shall be precluded from presenting such documents in any subsequent dispute resolution proceedings, if the data was reasonably available at the time of the request.

ARTICLE 13- COST OF WORK; ALLOWANCES, UNIT PRICE WORK

13.01 Cost of the Work

SC-13.01 Supplement Paragraph 13.01.B.5.c.(2) by adding the following sentence:

The equipment rental rate book that governs the included costs for the rental of machinery and equipment owned by Contractor (or a related entity) under the Cost of the Work provisions of this Contract is the most current edition of Rental Rate Blue Book for Construction Equipment.

- SC-13.01 Supplement Paragraph 13.01.C.2 by adding the following definition of small tools and hand tools:
 - **a.** For purposes of this paragraph, "small tools and hand tools" means any tool or equipment whose current price if it were purchased new at retail would be less than \$500.
- 13.03 Unit Price Work
- SC-13.03 Supplement Paragraph 13.03.A to read as follows:
 - A. Where the Contract Documents provide that all or part of the Work is to be Unit Price Work, initially the Contract Price will be deemed to include for all Unit Price Work an amount equal to the sum of the unit price for each separately identified item of Unit Price Work times the estimated quantity of each item as indicated in the Agreement. Contractor shall not be paid for any Unit Price Work that represents an actual quantity greater than 110% of the estimated quantity, without a Change Order. The Contractor shall maintain such records as required to track the quantities of Unit Price Work in anticipation of exceeding the 110% threshold.

ARTICLE 14— TESTS AND INSPECTIONS; CORRECTION, REMOVAL, OR ACCCEPTANCE OF DEFECTIVE WORK

- 14.07 Owner May Correct Defective Work
- SC-14.07 Delete Paragraph 14.07.A. in its entirety and replace with the following:
 - A. If Contractor fails within two (2) business days of written notice from the Owner or Engineer, or such longer time as may be stated in such notice, to correct, or take reasonable steps to commence to correct defective Work, or to remove and replace, or take reasonable steps to remove and replace, defective Work as required by Engineer, then Owner may correct or remedy any such deficiency. In such case an appropriate Change Order shall be issued deducting from payments then or thereafter due the Contractor all the costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) incurred or sustained by Owner in exercising the rights and remedies under this Paragraph 14.07. If payments then or thereafter due the Contractor irrevocably designates the Owner as the Contractor's attorney-in-fact to execute the Change Orders provided for in this Paragraph 14.07.

ARTICLE 15— PAYMENTS TO CONTRACTOR, SET OFFS; COMPLETIONS; CORRECTION PERIOD

- 15.01 Progress Payments
- SC-15.01 Delete subparagraph 15.01.B.4 in its entirety and add the following new subparagraphs:
 - 4. Retainage: Partial payments to Contractor for labor performed shall be made at the rate of 92 percent of the amount invoiced through the Application for Payment that shows

the total Contract Completion at 50 percent or greater, pursuant to Ohio Revised Code Section 153.14. After the Contract is 50 percent complete as evidenced by payments in the amount of at least 50 percent of the Contract Price to Contractor, no additional funds shall be retained from payments for labor.

- 5. Contractor shall submit one original on 8-1/2 by 11-inch paper of each lien waiver submitted.
- 6. Contractor shall submit one copy of each pay request for approval.
- 7. No advanced payment for shop drawing preparation will be made. Shop drawing costs will be paid when equipment and materials are delivered and suitably stored on the site.
- 8. All stored equipment and materials for which payment is requested shall have six copies (unless a different quantity is otherwise agreed upon) of invoices included with the pay request. Equipment shall be identified thoroughly on the invoices, including serial numbers.
- 9. Payment for the stored equipment and material which are on the site shall not exceed the invoiced amount for each item, less the Contract retainage. The overhead and profit for the stored items shall not be invoiced until the item is installed.
- 10. Payment for off-site storage is normally reserved for sensitive or very large pieces of equipment that in Engineer's opinion would not be practical to have stored on the site. Payment for off-site stored items shall be limited to 75% of the invoiced value of the item, less Contract retainage. Contractor shall reimburse Owner the Cost of inspecting off-site stored items. When off-site storage is approved, Contractor shall provide Insurance Certificates and Document of Ownership to Owner.
- SC-15.01 Add the following new Paragraph 15.01.F:
 - F. For contracts in which the Contract Price is based on the Cost of Work, if Owner determines that progress payments made to date substantially exceed the actual progress of the Work (as measured by reference to the Schedule of Values), or present a potential conflict with the Guaranteed Maximum Price, then Owner may require that Contractor prepare and submit a plan for the remaining anticipated Applications for Payment that will bring payments and progress into closer alignment and take into account the Guaranteed Maximum Price (if any), through reductions in billings, increases in retainage, or other equitable measures. Owner will review the plan, discuss any necessary modifications, and implement the plan as modified for all remaining Applications for Payment.

15.03 Substantial Completion

- SC-15.03 Add the following new subparagraph to Paragraph 15.03.B:
 - If some or all of the Work has been determined not to be at a point of Substantial Completion and will require re-inspection or re-testing by Engineer, the cost of such reinspection or re-testing, including the cost of time, travel and living expenses, will be paid by Contractor to Owner. If Contractor does not pay, or the parties are unable to agree as to the amount owed, then Owner may impose a reasonable set-off against payments due under this Article 15.

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

4/6/2023

FAIRFIELD COUNTY UTILITIES

- SC-15.03 Add the following new subparagraph to Paragraph 15.03:
 - G. Time for Completion of Items on Tentative List and Remedies. The time fixed by the Engineer for the completion of all items on the list accompanying the tentative certificate of Substantial Completion shall not be greater than forty-five (45) days. The Contractor shall complete all items on the list within such 45-day period. If the Contractor fails to do so, the Owner in its discretion may perform the Work by itself or others and the cost thereof shall be charged to the Contractor. The Contractor irrevocably designates the Owner as the Contractor's attorney-in-fact to execute a Change Order deducting such cost from the balance of the Contract Price and also any additional costs or expenses incurred by the Owner arising out of or related to the failure of the Contractor to complete such items, including but not limited to attorneys', consultants', and Engineer's fees. The Contractor's warranties under the Contract Documents shall remain in full force and effect and cover any remedial Work, even if performed by others. If more than one inspection by the Engineer for purposes of evaluating corrected Work is required, it will be performed at the Contractor's expense.

15.04 Partial Use or Occupancy

- SC-15.04 Add the following new Paragraph 15.04.A.4 immediately following to Paragraph 15.04.A.3:
 - 4. Owner may at any time request Contractor in writing to permit Owner to take over operation of any part of the Work although it is not substantially complete. A copy of such request will be sent to Engineer, and within a reasonable time thereafter, Owner, Contractor, and Engineer shall inspect that part of the Work to determine its status of completion and will prepare a list of the items remaining to be completed or corrected thereon before final payment. If Contractor does not object in writing to Owner and Engineer that such part of the Work is not ready for separate operation by Owner, Engineer will finalize the list of items to be completed or corrected and will deliver such lists to Owner and Contractor together with a written recommendation as to the division of responsibilities pending final payment between Owner and Contractor with respect to security, operation, safety, maintenance, utilities, insurance, warranties, and guarantees for that part of the Work which will become binding upon Owner and Contractor at the time when Owner takes over such operation (unless they shall have otherwise agreed in writing and so informed Engineer). During such operation and prior to Substantial Completion of such part of the Work, Owner shall allow Contractor reasonable access to complete or correct items on said list and to complete other related Work.

15.06 Final Payment

- SC-15.06 Amend Paragraph 15.06.A.3 by striking out "Contractor may" and replacing with "Contractor shall" in both locations.
- SC-15.06 Delete "set offs for liquidated damages and" from Paragraph 15.06.E.

15.08 Correction Period

- SC-15.08 Amend Paragraph 15.08.B by striking out "60 days" and replacing with "180 days" within both locations of the subparagraph.
- SC-15.08 Amend Paragraph 15.08.C by striking out "60 days" and replacing with "180 days" within the subparagraph.
- SC-15.08 Amend Paragraph 15.08.C by striking out "arbitration or other" within the subparagraph.

ARTICLE 16— SUSPENSION OF WORK AND TERMINATION

16.02 Owner May Terminate for Cause

- SC-16.02 Amend Paragraph 16.02.A.1 by striking out "persistent" within the subparagraph.
- SC-16.02 Amend Paragraph 16.02.B by striking out "10 days' written notice" and replacing with "three (3) business days' written notice".
- SC-16.02 Amend Paragraph 16.02.B by adding the following subparagraphs:
 - 3. Complete the Work as Owner may deem expedient
 - 4. Such termination shall be effective as of the date stated in the termination notice provided to Contractor
- SC-16.02 Amend Paragraph 16.02.D by striking out "7 days" and replacing with "three (3) business days".
- 16.03 Owner May Terminate for Convenience
- SC-16.03 Delete Paragraphs 16.03.A and 16.03.A.1-3 in their entirety and replace with the following:
 - A. Upon three (3) business days' written notice to Contractor and Engineer, Owner may, without cause and without prejudice to any other right or remedy of Owner, terminate the Contract. Such termination shall be effective as of the date stated in the written notice. In such case, Contractor shall be paid for (without duplication of any items):
 - 1. completed and acceptable Work executed in accordance with the Contract Documents prior to the effective date of termination; and
 - 2. expenses sustained prior to the effective date of termination in performing services and furnishing labor, materials, or equipment as required by the Contract Documents in connection with uncompleted Work.
- SC-16.03 Add the following new subparagraph to Paragraph 16.03:
 - C. Contractor shall require similar provisions contained in Paragraph 16.03 in each of its subcontracts to protect Contractor from claims by Subcontractors arising from the Owner's termination for convenience, or to minimize claims by such subcontractors. The remedy provided to Contractor under this Paragraph 16.03 shall be the Contractor's sole remedy in the event of termination for convenience by Owner.

16.04 Contractor May Stop Work or Terminate

SC-16.04 Amend paragraph 16.04.B by striking out the following: ", including interest thereon".

ARTICLE 17— FINAL RESOLUTIONS OF DISPUTES

- 17.01 Methods and Procedures
- SC-17.01 Delete Paragraphs 17.01.A-B and replace with the following:
 - A. Settlement, Methods and Procedures

1. In addition to Owner's entitlement to attorney's fees set forth elsewhere in the Contract Documents, in the event that Contractor files a Claim or files an action against Owner, Owner shall be entitled to make an offer of settlement of the Claim to Contractor at any time up to the date of trial. Such offer of settlement shall not be admissible into evidence at the litigation except on the issue of entitlement to recovery of attorneys' fees, costs and expenses. If at any stage of the litigation, including any appeals, Contractor's claim is dismissed or found to be without merit, or if the damages awarded to Contractor on its claim do not exceed Owner's offer of settlement, Contractor shall be liable to Owner and shall reimburse Owner for all attorney's fees, costs and expenses incurred by Owner from the date of the offer of settlement until the date of the final adjudication and resolution of Contractor's claim.

2. Any dispute, claim or other matter not settled by negotiation or other means as mutually agreed upon by Owner, Contractor, and surety where applicable, shall be determined by the Court of Common Pleas for Fairfield County, Ohio, which shall have exclusive venue and jurisdiction over such matters and claims, to the exclusion of any other court, including any U.S. District Court

ARTICLE 18— MISCELLANEOUS

- 18.07 Controlling Law
- SC-18.07 Amend Paragraph 18.07.A by striking out "state in which Project is located" and replacing with "State of Ohio".
- 18.08 Assignment of Contract
- SC-18.08 Deleted.
- SC-18.11 Add the following new Section 18.11 immediately after Section 18.10:
- 18.11 Equal Employment Opportunity and Non Discrimination
 - A. The Contractor shall comply with, and shall require all Subcontractors of any tier to comply with, the applicable equal employment opportunity and nondiscrimination statutes and regulations of the State of Ohio.

- SC-18.12 Add the following new Section 18.12 to the General Conditions:
- 18.12 Contract Work Hours and Safety Standard Act
 - A. All Contractors and Subcontractors, of any tier, shall comply with the applicable federal regulations for contract work hours and safety standards.
- SC-18.13 Add the following new Section 18.13 to the General Conditions:
- 18.13 Clean Air Act
 - A. All Contractors and Subcontractors, of any tier, shall comply with the applicable standards, orders, or regulations issued pursuant to the Clean Air Act of 1970 (42 U.S.C. 1857, et seq.) and the Federal Water Pollution Control Act (33 U.S. C. 1251, et seq.), as amended.

Item	Electronic Documents	Transmittal Means	Data Format	Note (1)
a.1	General communications, transmittal covers, meeting notices and responses to general information requests for which there is no specific prescribed form.	Email	Email	
a.2	Meeting agendas, meeting minutes, RFI's and responses to RFI's, and Contract forms.	Email w/ Attachment	PDF	(2)
a.3	Contactors Submittals (Shop Drawings, "or equal" requests, substitution requests, documentation accompanying Sample submittals and other submittals) to Owner and Engineer, and Owner's and Engineer's responses to Contractor's Submittals, Shop Drawings, correspondence, and Applications for Payment.	Email w/ Attachment	PDF	
a.4	Correspondence; milestone and final version Submittals of reports, layouts, Drawings, maps, calculations and spreadsheets, Specifications, Drawings and other Submittals from Contractor to Owner or Engineer and for responses from Engineer and Owner to Contractor regarding Submittals.	Email w/ Attachment or LFE	PDF	
a.5	Layouts and drawings to be submitted to Owner for future use and modification.	Email w/ Attachment or LFE	DWG	
a.6	Correspondence, reports and Specifications to be submitted to Owner for future word processing use and modification.	Email w/ Attachment or LFE	DOC	
a.7	Spreadsheets and data to be submitted to Owner for future data processing use and modification.	Email w/ Attachment or LFE	EXC	
a.8	Database files and data to be submitted to Owner for future data processing use and modification.	Email w/ Attachment or LFE	DB	
Notes				
(1)	All exchanges and uses of transmitted data are subject to the appropriate process of the appropriate of the approprese of the appropriate of the a	opriate provisions of C	ontract	
(2)	Transmittal of written notices is governed by Paragraph 18.01 of th	e General Conditions.		
Кеу				
Email	Standard Email formats (.htm, .rtf, or .txt). Do not use stationery impair legibility of content on screen or in printed copies	formatting or other fe	atures tha	t
LFE	Agreed upon Large File Exchange method (FTP, CD, DVD, hard dri	ve)		
PDF	Portable Document Format readable by Adobe® Acrobat Reader	Version 10 or later		
DWG	Autodesk [®] AutoCAD .dwg format Version 2013			
DOC	Microsoft [®] Word .docx format Version Office 2019			
EXC	Microsoft [®] Excel .xls or .xml format Version Office 2019			

EXHIBIT A—SOFTWARE REQUIREMENTS FOR ELECTRONIC DOCUMENT EXCHANGE

Microsoft® Access .mdb format Version Office 2019

DB

EXHIBIT B-FORESEEABLE BAD WEATHER DAYS

Month	Number of Foreseeable Bad Weather Days in Month
January	8
February	8
March	7
April	6
May	5
June	5
July	4
August	4
September	5
October	6
November	6
December	6

Exhibit B—Foreseeable Bad Weather Days. EJCDC® C-800, Supplementary Conditions of the Construction Contract. Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 1 of 1

EXHIBIT C-GEOTECHNICAL BASELINE REPORT SUPPLEMENT TO THE SUPPLEMENTARY CONDITIONS

Not Used.

Exhibit C—Geotechnical Baseline Report Supplement to the Supplementary Conditions. EJCDC® C-800, Supplementary Conditions of the Construction Contract. Copyright[®] 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

CHANGE ORDER NO.

Owner:	Fairfield County Utilities		
Engineer:	PRIME AE Group, Inc.	Engineer's Project No.:	21513
Contractor:		Contractor's Project No.:	
Project:	Airport Sanitary Sewer Extension and Water Service Design		
Date Issued:		-	
Effective Date of	Change Order:		

The Contract is modified as follows upon execution of this Change Order:

Description:

Attachments:

Change in Contract Times [State Contract Times as either a specific date or a

Change in Contract Price	number of days]
Original Contract Price:	Original Contract Times: Substantial Completion: Ready for final payment:
[Increase] [Decrease] from previously approved Change Orders No. 1 to No. [Number of previous Change Order]: \$	[Increase] [Decrease] from previously approved Change Orders No.1 to No. [Number of previous Change Order]: Substantial Completion: Ready for final payment:
Contract Price prior to this Change Order:	Contract Times prior to this Change Order: Substantial Completion: Ready for final payment:
[Increase] [Decrease] this Change Order:	[Increase] [Decrease] this Change Order: Substantial Completion: Ready for final payment:
Contract Price incorporating this Change Order:	Contract Times with all approved Change Orders: Substantial Completion: Ready for final payment:

Recommended by Engineer (if required)	Authorized by Owner
Authorized by Owner	Approved by Funding Agency (if applicable)

EJCDC[®] C-941, Change Order.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 10/6/2022

FIELD ORDER NO.: ____

Contractor is hereby directed to promptly perform the Work described in this Field Order, issued in accordance with Paragraph 11.04 of the General Conditions, for minor changes in the Work without changes in Contract Price or Contract Times. If Contractor considers that a change in Contract Price or Contract Times is required, submit a Change Proposal before proceeding with this Work.

Reference:

Specification Section(s):

Drawing(s) / Details (s):

Description:

Attachments:

Issued by Engineer

By:	
Title:	
Date:	

EJCDC[®] C-942, Field Order. Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies and American Society of Civil Engineers. All rights reserved. Page 1 of 1

DOCUMENT 004536 - EQUAL EMPLOYMENT OPPORTUNITY CERTIFICATION

During the performance of this contract, the undersigned agrees as follows:

- 1. The undersigned will not discriminate against any employee or applicant for employment because of race, color, religion, sex, or national origin. The undersigned will take affirmative action to ensure that applicants are employed, and that employees are treated during employment without regard to their race, color, religion or national origin. Such action shall include, but not be limited to the following: Employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The undersigned agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided setting forth the provisions of this equal opportunity (federally assisted construction) clause.
- The undersigned will, in all solicitations or advertisements for employees placed by or on behalf of the undersigned, state the all qualified applicants will receive consideration for employment without regard to race, color, religion, sex or national origin.
- 3. The undersigned will send to each labor union or representative of workers, with which he has a collective bargaining agreement or other contract or understanding, a notice to be provided advising the said labor union or workers' representative of the undersigned's commitment under this section, and shall post copies of the notice in conspicuous places available to employees and applicants for employment.
- 4. The undersigned will comply with all provisions of Executive Order No. 11246 of September 24, 1965, and of the rules, regulations, and relevant orders of the Secretary of Labor.
- 5. The undersigned will furnish all information and reports required by Executive Order No. 11246 of September 24, 1965, and by the rules, regulations, and relevant orders of the Secretary of Labor, or pursuant thereto, and will permit access to his books, records and accounts by the administering agency of the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations, and orders.
- 6. In the event of the undersigned's non-compliance with the equal opportunity (federally assisted construction) clause of this contract of with any of the said rules, regulations, or orders, this contract may be canceled, terminated or suspended in whole or in part, and the undersigned may be declared ineligible for further Government contracts of federally assisted construction contracts in accordance with procedures authorized in Executive Order No. 11246 of September 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order No 11246 of September 24, 1965, or by rules, regulations, or order of the Secretary of Labor, or as provided by law.
- 7. The undersigned will include this equal opportunity (federally assisted construction) clause in every subcontract or purchase order unless exempted by the rules, regulations, or orders of the Secretary of Labor issued pursuant to section 204 of Executive Order No 11246 of September 24, 1965, so that such provision will be binding upon each subcontract or vender. The undersigned will take such action with respect to any subcontract or purchase order as the administering agency may direct as a means of enforcing such provisions, including sanctions for non compliance: Provided, however, that in the event a contractor becomes involved in, or is threatened with, litigation with a subcontractor or vendor, as a result of such direction by the administering agency the undersigned may request the United States to enter into such litigation to protect the interest of the United States.

004536 - 1

1.1

Signature

Date

Name and Title of Authorized Signatory, Please Print or Type

Bidder's Firm

END OF SECTION 004536

(THIS PAGE INTENTIONALLY LEFT BLANK)

EQUAL EMPLOYMENT OPPORTUNITY CERTIFICATION

DOCUMENT 004547 – CERTIFICATION REGARDING DEBARMENT AND SUSPENSION

The prospective participant certifies to the best of its knowledge and belief that it and its principals:

- (a) Are not presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from covered transactions by any Federal department or agency;
- (b) Have not within a three year period preceding this proposal been convicted of or had a civil judgement rendered against them for commission of fraud or a criminal offense in connection with obtaining, attempting to obtain, or performing a public (Federal, State, or local) transaction or contract under a public transaction; violation of Federal of State antitrust statues or commission if embezzlement, theft, forgery, bribery, falsification or destruction of records, making false statements, or receiving stolen property;
- (c) Are not presently indicted for or otherwise criminally or civilly charged by a government entity (Federal, State or local) with commission of any of the offenses enumerated in paragraph (b) of this certification;
- (d) Have not within a three year period preceding this application / proposal had one or more public transactions (Federal, State, or local) terminated for cause or default; and
- (e) Will not utilize a subcontractor or supplier who is unable to certify (a) through (d) above.

I understand that a false statement on this certification may be grounds for rejection of this proposal or termination of the award. In addition, under 18 USC Sec. 1001, a false statement may result in a fine of up to \$10,000 or imprisonment for up to 5 years, or both.

Type Name & Title of Authorized Representative

Signature of Authorized Representative

Date

 \Box I am unable to certify to the above statements. My explanation is attached.

END OF SECTION 004547

CERTIFICATION REGARDING DEBARMENT AND SUSPENSION

(THIS PAGE INTENTIONALLY LEFT BLANK)

CERTIFICATION REGARDING DEBARMENT AND SUSPENSION

DOCUMENT 004548 - AIS ACKNOWLEDGEMENT FORM

The Contractor acknowledges to and for the benefit of FAIFIELD COUNTY ("Purchaser") and the State of Ohio (the "State") that it understands the goods and services under this Agreement are being funded with monies made available by the Clean Water State Revolving Fund and/or Drinking Water State Revolving Fund that have statutory requirements commonly known as "American Iron and Steel;" that requires all of the iron and steel products used in the project to be produced in the United States ("American Iron and Steel Requirement") including iron and steel products provided by the Contactor pursuant to this Agreement. The Contractor hereby represents and warrants to and for the benefit of the Purchaser and the State that (a) the Contractor has reviewed and understands the American Iron and Steel Requirement, (b) all of the iron and steel products used in the project will be and/or have been produced in the United States in a manner that complies with the American Iron and Steel Requirement, unless a waiver of the requirement is approved, and (c) the Contractor will provide any further verified information, certification or assurance of compliance with this paragraph, or information necessary to support a waiver of the American Iron and Steel Requirement, as may be requested by the Purchaser or the State. Notwithstanding any other provision of this Agreement, any failure to comply with this paragraph by the Contractor shall permit the Purchaser or State to recover as damages against the Contractor any loss, expense, or cost (including without limitation attorney's fees) incurred by the Purchaser or State resulting from any such failure (including without limitation any impairment or loss of funding, whether in whole or in part, from the State or any damages owed to the State by the Purchaser). While the Contractor has no direct contractual privity with the State, as a lender to the Purchaser for the funding of its project, the Purchaser and the Contractor agree that the State is a third-party beneficiary and neither this paragraph (nor any other provision of this Agreement necessary to give this paragraph force or effect) shall be amended or waived without the prior written consent of the State.

Signature

Date

Name and Title of Authorized Signatory, Please Print or Type

Bidder's Firm

END OF SECTION 004548

(THIS PAGE INTENTIONALLY LEFT BLANK)

AIS ACKNOWLEDGEMENT FORM

DOCUMENT 004549 - BUILD AMERICA, BUY AMERICA (BABA) ACKNOWLEDGEMENT

The Contractor acknowledges to and for the benefit of Fairfield County ("Owner") and the State of Ohio (State) that it understands the goods and services under this Agreement are being funded with federal monies and have statutory requirements commonly known as "Build America, Buy America;" that requires all of the iron and steel, manufactured products, and construction materials used in the project to be produced in the United States ("Build America, Buy America Requirements") including iron and steel, manufactured products, and construction materials provided by the Contactor pursuant to this Agreement. The Contractor hereby represents and warrants to and for the benefit of the Owner and Funding Authority (a) the Contractor has reviewed and understands the Build America, Buy America Requirements, (b) all of the iron and steel, manufactured products, and construction materials used in the project will be and/or have been produced in the United States in a manner that complies with the Build America, Buy America Requirements, unless a waiver of the requirements is approved, and (c) the Contractor will provide any further verified information, certification or assurance of compliance with this paragraph, or information necessary to support a waiver of the Build America, Buy America Requirements, as may be requested by the Owner or the Funding Authority. Notwithstanding any other provision of this Agreement, any failure to comply with this paragraph by the Contractor shall permit the Owner or Funding Authority to recover as damages against the Contractor any loss, expense, or cost (including without limitation attorney's fees) incurred by the Owner or Funding Authority resulting from any such failure (including without limitation any impairment or loss of funding, whether in whole or in part, from the Funding Authority or any damages owed to the Funding Authority by the Owner). If the Contractor has no direct contractual privity with the Funding Authority, as a lender or awardee to the Owner for the funding of its project, the Owner and the Contractor agree that the Funding Authority is a third-party beneficiary and neither this paragraph (nor any other provision of this Agreement necessary to give this paragraph force or effect) shall be amended or waived without the prior written consent of the Funding Authority

Signature

Date

Name and Title of Authorized Signatory, Please Print or Type

Bidder's Firm

END OF SECTION 004549

(THIS PAGE INTENTIONALLY LEFT BLANK)

BUILD AMERICA, BUY AMERICA (BABA) ACKNOWLEDGEMENT

1

SECTION 007343 - WAGE DETERMINATION

As used in these provisions "subrecipient" means Fairfield County.

(a) The following applies to any contract in excess of \$2,000 which is entered into for the actual construction, alteration and/or repair, including painting and decorating, of a public building or public work, or building or work financed in whole or in part from Federal funds or in accordance with guarantees of a Federal agency or financed from funds obtained by pledge of any contract of a Federal agency to make a loan, grant or annual contribution (except where a different meaning is expressly indicated), and which is subject to the labor standards provisions of any of the acts listed in § 5.1.

(1) Minimum wages.

(i) All laborers and mechanics employed or working upon the site of the work will be paid unconditionally and not less often than once a week, and without subsequent deduction or rebate on any account (except such payroll deductions as are permitted by regulations issued by the Secretary of Labor under the Copeland Act (29 CFR part 3)), the full amount of wages and bona fide fringe benefits (or cash equivalents thereof) due at time of payment computed at rates not less than those contained in the wage determination of the Secretary of Labor which is attached hereto and made a part hereof, regardless of any contractual relationship which may be alleged to exist between the contractor and such laborers and mechanics.

Contributions made or costs reasonably anticipated for bona fide fringe benefits under section 1(b)(2) of the Davis-Bacon Act on behalf of laborers or mechanics are considered wages paid to such laborers or mechanics, subject to the provisions of paragraph (a)(1)(iv) of this section; also, regular contributions made or costs incurred for more than a weekly period (but not less often than quarterly) under plans, funds, or programs which cover the particular weekly period, are deemed to be constructively made or incurred during such weekly period. Such laborers and mechanics shall be paid the appropriate wage rate and fringe benefits on the wage determination for the classification of work actually performed, without regard to skill, except as provided in § 5.5(a)(4). Laborers or mechanics performing work in more than one classification may be compensated at the rate specified for each classification for the time actually worked therein: Provided, that the employer's payroll records accurately set forth the time spent in each classification in which work is performed. The wage determination (including any additional classification and wage rates conformed under paragraph (a)(1)(ii) of this section) and the Davis-Bacon poster (WH- 1321) shall be posted at all times by the contractor and its subcontractors at the site of the work in a prominent and accessible place where it can be easily seen by the workers.

Subrecipients may obtain wage determinations from the U.S. Department of Labor's web site, www.wdol.gov.

(ii)(A) The subrecipient(s), on behalf of EPA, shall require that any class of laborers or mechanics, including helpers, which is not listed in the wage determination and which is to be employed under the contract shall be classified in conformance with the wage determination. The EPA award official shall approve an additional classification and wage rate and fringe benefits therefore only when the following criteria have been met:

(1) The work to be performed by the classification requested is not performed by a classification in the wage determination; and

(2) The classification is utilized in the area by the construction industry; and

(3) The proposed wage rate, including any bona fide fringe benefits, bears a reasonable relationship to the

WAGE DETERMINATION

007343 - 1

wage rates contained in the wage determination.

(B) If the contractor and the laborers and mechanics to be employed in the classification (if known), or their representatives, and the subrecipient(s) agree on the classification and wage rate (including the amount designated for fringe benefits where appropriate), a report of the action taken shall be sent by the subrecipient(s) to the State award official. The State award official will transmit the report, to the Administrator of the Wage and Hour Division, Employment Standards Administration, U.S. Department of Labor, Washington, DC 20210. The Administrator, or an authorized representative, will approve, modify, or disapprove every additional classification action within 30 days of receipt and so advise the State award official or will notify the State award official within the 30-day period that additional time is necessary.

(C) In the event the contractor, the laborers or mechanics to be employed in the classification or their representatives, and the and the subrecipient(s) do not agree on the proposed classification and wage rate (including the amount designated for fringe benefits, where appropriate), the award official shall refer the questions, including the views of all interested parties and the recommendation of the State award official, to the Administrator for determination. The Administrator, or an authorized representative, will issue a determination within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.

(D) The wage rate (including fringe benefits where appropriate) determined pursuant to paragraphs (a)(1)(ii)(B) or (C) of this section, shall be paid to all workers performing work in the classification under this contract from the first day on which work is performed in the classification.

(iii) Whenever the minimum wage rate prescribed in the contract for a class of laborers or mechanics includes a fringe benefit which is not expressed as an hourly rate, the contractor shall either pay the benefit as stated in the wage determination or shall pay another bona fide fringe benefit or an hourly cash equivalent thereof.

(*iv*) If the contractor does not make payments to a trustee or other third person, the contractor may consider as part of the wages of any laborer or mechanic the amount of any costs reasonably anticipated in providing bona fide fringe benefits under a plan or program, Provided, That the Secretary of Labor has found, upon the written request of the contractor, that the applicable standards of the Davis-Bacon Act have been met. The Secretary of Labor may require the contractor to set aside in a separate account assets for the meeting of obligations under the plan or program.

(2) Withholding. The subrecipient(s), shall upon written request of the EPA Award Official or an authorized representative of the Department of Labor, withhold or cause to be withheld from the contractor under this contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to Davis-Bacon prevailing wage requirements, which is held by the same prime contractor, so much of the accrued payments or advances as may be considered necessary to pay laborers and mechanics, including apprentices, trainees, and helpers, employed by the contractor or any subcontractor the full amount of wages required by the contract. In the event of failure to pay any laborer or mechanic, including any apprentice, trainee, or helper, employed or working on the site of the work, all or part of the wages required by the contract, the (Agency) may, after written notice to the contractor, sponsor, applicant, or owner, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds until such violations have ceased.

WAGE DETERMINATION

Payrolls and basic records.

(*i*) Payrolls and basic records relating thereto shall be maintained by the contractor during the course of the work and preserved for a period of three years thereafter for all laborers and mechanics working at the site of the work. Such records shall contain the name, address, and social security number of each such worker, his or her correct classification, hourly rates of wages paid (including rates of contributions or costs anticipated for bona fide fringe benefits or cash equivalents thereof of the types described in section 1(b)(2)(B) of the Davis-Bacon Act), daily and weekly number of hours worked, deductions made and actual wages paid. Whenever the Secretary of Labor has found under 29 CFR 5.5(a)(1)(iv) that the wages of any laborer or mechanic include the amount of any costs reasonably anticipated in providing benefits under a plan or program described in section 1(b)(2)(B) of the Davis-Bacon Act, the contractor shall maintain records which show that the commitment to provide such benefits is enforceable, that theplan or program is financially responsible, and that the plan or program has been communicated in writing to the laborers or mechanics affected, and records which show the costs anticipated or the actual cost incurred in providing such benefits. Contractors employing apprentices or trainees under approved programs shall maintain written evidence of the registration of apprenticeship programs and certification of trainee programs, the registration of the apprentices and trainees, and the ratios and wage rates prescribed in the applicable programs.

(ii)(A) The contractor shall submit weekly, for each week in which any contract work is performed, a copy of all payrolls to the subrecipient, that is, the entity that receives the subgrant or loan from the State capitalization grant recipient. Such documentation shall be available on request of the State recipient or EPA. As to each payroll copy received, the subrecipient shall provide written confirmation in a form satisfactory to the State indicating whether or not the project is in compliance with the requirements of 29 CFR 5.5(a)(1) based on the most recent payroll copies for the specified week. The payrolls shall set out accurately and completely all of the information required to be maintained under 29 CFR 5.5(a)(3)(i), except that full social security numbers and home addresses shall not be included on the weekly payrolls. Instead the payrolls shall only need to include an individually identifying number for each employee (e.g., the last four digits of the employee's social security number). The required weekly payroll information may be submitted in any form desired. Optional Form WH-347 is available for this purpose from the Wage and Hour Division Web site at http://www.dol.gov/esa/whd/forms/wh347instr.htm or its successor site. The prime contractor is responsible for the submission of copies of payrolls by all subcontractors. Contractors and subcontractors shall maintain the full social security number and current address of each covered worker, and shall provide them upon request to the subrecipient(s) for transmission to the State or EPA if requested by EPA, the State, the contractor, or the Wage and Hour Division of the Department of Labor for purposes of an investigation or audit of compliance with prevailing wage requirements. It is not a violation of this section for a prime contractor to require a subcontractor to provide addresses and social security numbers to the prime contractor for its own records, without weekly submission to the subrecipient(s).

(B) Each payroll submitted shall be accompanied by a "Statement of Compliance," signed by the contractor or subcontractor or his or her agent who pays or supervises the payment of the persons employed under the contract and shall certify the following:

(1) That the payroll for the payroll period contains the information required to be provided under § 5.5 (a)(3)(ii) of Regulations, 29 CFR part 5, the appropriate information is being maintained under § 5.5 (a)(3)(i) of Regulations, 29 CFR part 5, and that such information is correct and complete;

(2) That each laborer or mechanic (including each helper, apprentice, and trainee) employed on the contract during the payroll period has been paid the full weekly wages earned, without rebate, either directly or indirectly, and that no deductions have been made either directly or indirectly from the full wages earned, other than permissible deductions as set forth in Regulations, 29 CFR part 3;

(3) That each laborer or mechanic has been paid not less than the applicable wage rates and fringe benefits

WAGE DETERMINATION

007343 - 3

or cash equivalents for the classification of work performed, as specified in the applicable wage determination incorporated into the contract.

(C) The weekly submission of a properly executed certification set forth on the reverse side of Optional Form WH-347 shall satisfy the requirement for submission of the "Statement of Compliance" required by paragraph (a)(3)(ii)(B) of this section.

(D) The falsification of any of the above certifications may subject the contractor or subcontractor to civil or criminal prosecution under section 1001 of title 18 and section 231 of title 31 of the United States Code.

(iii) The contractor or subcontractor shall make the records required under paragraph (a)(3)(i) of this section available for inspection, copying, or transcription by authorized representatives of the State, EPA or the Department of Labor, and shall permit such representatives to interview employees during working hours on the job. If the contractor or subcontractor fails to submit the required records or to make them available, the Federal agency or State may, after written notice to the contractor, sponsor, applicant, or owner, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds. Furthermore, failure to submit the required records upon request or to make such records available may be grounds for debarment action pursuant to 29 CFR 5.12.

(3) Apprentices and trainees --

(i) Apprentices. Apprentices will be permitted to work at less than the predetermined rate for the work they performed when they are employed pursuant to and individually registered in a bona fide apprenticeship program registered with the U.S. Department of Labor, Employment and Training Administration, Office of Apprenticeship Training, Employer and Labor Services, or with a State Apprenticeship Agency recognized by the Office, or if a person is employed in his or her first 90 days of probationary employment as an apprentice in such an apprenticeship program, who is not individually registered in the program, but who has been certified by the Office of Apprenticeship Training, Employer and Labor Services or a State Apprenticeship Agency (where appropriate) to be eligible for probationary employment as an apprentice. The allowable ratio of apprentices to journeymen on the job site in any craft classification shall not be greater than the ratio permitted to the contractor as to the entire work force under the registered program. Any worker listed on a payroll at an apprentice wage rate, who is not registered or otherwise employed as stated above, shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any apprentice performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. Where a contractor is performing construction on a project in a locality other than that in which its program is registered, the ratios and wage rates (expressed in percentages of the journeyman's hourly rate) specified in the contractor's or subcontractor's registered program shall be observed. Every apprentice must be paid at not less than the rate specified in the registered program for the apprentice's level of progress, expressed as a percentage of the journeymen hourly rate specified in the applicable wage determination. Apprentices shall be paid fringe benefits in accordance with the provisions of the apprenticeship program. If the apprenticeship program does not specify fringe benefits, apprentices must be paid the full amount of fringe benefits listed on the wage determination for the applicable classification. If the Administrator determines that a different practice prevails for the applicable apprentice classification, fringes shall be paid in accordance with that determination. In the event the Office of Apprenticeship Training, Employer and Labor Services, or a State Apprenticeship Agency recognized by the Office, withdraws approval of an apprenticeship program, the contractor will no longer be permitted to utilize apprentices at less than the applicable predetermined rate for the work performed until an acceptable program is approved.

(*ii*) Trainees. Except as provided in 29 CFR 5.16, trainees will not be permitted to work at less than the predetermined rate for the work performed unless they are employed pursuant to and individually registered in a program which has received prior approval, evidenced by formal certification by the U.S. Department of Labor,

WAGE DETERMINATION

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 4/6/2023

Employment and Training Administration. The ratio of trainees to journeymen on the job site shall not be greater than permitted under the plan approved by the Employment and Training Administration. Every trainee must be paid at not less than the rate specified in the approved program for the trainee's level of progress, expressed as a percentage of the journeyman hourly rate specified in the applicable wage determination. Trainees shall be paid fringe benefits in accordance with the provisions of the trainee program. If the trainee program does not mention fringe benefits, trainees shall be paid the full amount of fringe benefits listed on the wage determination unless the Administrator of the Wage and Hour Division determines that there is an apprenticeship program associated with the corresponding journeyman wage rate on the wage determination which provides for less than full fringe benefits for apprentices. Any employee listed on the payroll at a trainee rate who is not registered and participating in a training plan approved by the Employment and Training Administration shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any trainee performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. In the event the Employment and Training Administration withdraws approval of a training program, the contractor will no longer be permitted to utilize trainees at less than the applicable predetermined rate for the work performed until an acceptable program is approved.

(iii) Equal employment opportunity. The utilization of apprentices, trainees and journeymen under this part shall be in conformity with the equal employment opportunity requirements of Executive Order 11246, as amended, and 29 CFR part 30.

(4) Compliance with Copeland Act requirements. The contractor shall comply with the requirements of 29 CFR part 3, which are incorporated by reference in this contract.

(5) Subcontracts. The contractor or subcontractor shall insert in any subcontracts the clauses contained in 29 CFR 5.5(a)(1) through (10) and such other clauses as the EPA determines may by appropriate, and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for the compliance by any subcontractor or lower tier subcontractor with all the contract clauses in 29 CFR 5.5.

(6) Contract termination: debarment. A breach of the contract clauses in 29 CFR 5.5 may be grounds for termination of the contract, and for debarment as a contractor and a subcontractor as provided in 29 CFR 5.12.

(7) Compliance with Davis-Bacon and Related Act requirements. All rulings and interpretations of the Davis-Bacon and Related Acts contained in 29 CFR parts 1, 3, and 5 are herein incorporated by reference in this contract.

(8) Disputes concerning labor standards. Disputes arising out of the labor standards provisions of this contract shall not be subject to the general disputes clause of this contract. Such disputes shall be resolved in accordance with the procedures of the Department of Labor set forth in 29 CFR parts 5, 6, and

7. Disputes within the meaning of this clause include disputes between the contractor (or any of its subcontractors) and subrecipient(s), State, EPA, the U.S. Department of Labor, or the employees or their representatives.

(9) Certification of eligibility.

(*i*) By entering into this contract, the contractor certifies that neither it (nor he or she) nor any person or firm who has an interest in the contractor's firm is a person or firm ineligible to be awarded Government contracts by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).

WAGE DETERMINATION

(*ii*) No part of this contract shall be subcontracted to any person or firm ineligible for award of a Government contract by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).

(iii) The penalty for making false statements is prescribed in the U.S. Criminal Code, 18 U.S.C. 1001.

WAGE DETERMINATION

007343 - 6

Contract Provision For Contracts In Excess Of \$100,000 And Subject To The Overtime Provisions Of The Contract Work Hours And Safety Standards Act

(b) Contract Work Hours and Safety Standards Act. The following applies to any contract in an amount in excess of \$100,000 and subject to the overtime provisions of the Contract Work Hours and Safety Standards Act. As used in these paragraphs, the terms laborers and mechanics include watchmen and guards.

(1) Overtime requirements. No contractor or subcontractor contracting for any part of the contract work which may require or involve the employment of laborers or mechanics shall require or permit any such laborer or mechanic in any workweek in which he or she is employed on such work to work in excess of forty hours in such workweek unless such laborer or mechanic receives compensation at a rate not less than one and one-half times the basic rate of pay for all hours worked in excess of forty hours in such workweek.

(2) Violation; liability for unpaid wages; liquidated damages. In the event of any violation of the clause set forth in paragraph (b)(1) of this section the contractor and any subcontractor responsible therefore shall be liable for the unpaid wages. In addition, such contractor and subcontractor shall be liable to the United States (in the case of work done under contract for the District of Columbia or a territory, to such District or to such territory), for liquidated damages. Such liquidated damages shall be computed with respect to each individual laborer or mechanic, including watchmen and guards, employed in violation of the clause set forth in paragraph (a)(1) of this section, in the sum of \$10 for each calendar day on which such individual was required or permitted to work in excess of the standard workweek of forty hours without payment of the overtime wages required by the clause set forth in paragraph (b)(1) of this section.

(3) Withholding for unpaid wages and liquidated damages. The subrecipient, upon written request of the EPA Award Official or an authorized representative of the Department of Labor, shall withhold or cause to be withheld, from any moneys payable on account of work performed by the contractor or subcontractor under any such contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to the Contract Work Hours and Safety Standards Act, which is held by the same prime contractor, such sums as may be determined to be necessary to satisfy any liabilities of such contractor or subcontractor for unpaid wages and liquidated damages as provided in the clause set forth in paragraph (b)(2) of this section.

(4) Subcontracts. The contractor or subcontractor shall insert in any subcontracts the clauses set forth in paragraph (b)(1) through (4) of this section and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for compliance by any subcontractor or lower tier subcontractor with the clauses set forth in paragraphs (b)(1) through (4) of this section.

Contract Provision For Contracts In Excess Of \$100,000 Subject ONLY To The Contract Work Hours And Safety Standards Act

(c) The following applies to any contract subject only to the Contract Work Hours and Safety Standards Act and not to any of the other statutes cited in 29 CFR 5.1.

WAGE DETERMINATION

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 10/6/2022

The contractor or subcontractor shall maintain payrolls and basic payroll records during the course of the work and shall preserve them for a period of three years from the completion of the contract for all laborers and mechanics, including guards and watchmen, working on the contract. Such records shall contain the name and address of each such employee, social security number, correct classifications, hourly rates of wages paid, daily and weekly number of hours worked, deductions made, and actual wages paid.

The records shall be maintained under this paragraph shall be made available by the contractor or subcontractor for inspection, copying, or transcription by authorized representatives of the Ohio EPA, EPA and the Department of Labor, and the contractor or subcontractor will permit such representatives to interview employees during working hours on the job.

END OF SECTION 007343

WAGE DETERMINATION

"General Decision Number: OH20230001 03/03/2023

Superseded General Decision Number: OH20220001

State: Ohio

Construction Types: Heavy and Highway

Counties: Ohio Statewide.

Heavy and Highway Construction Projects

Note: Contracts subject to the Davis-Bacon Act are generally required to pay at least the applicable minimum wage rate required under Executive Order 14026 or Executive Order 13658. Please note that these Executive Orders apply to covered contracts entered into by the federal government that are subject to the Davis-Bacon Act itself, but do not apply to contracts subject only to the Davis-Bacon Related Acts, including those set forth at 29 CFR 5.1(a)(2)-(60).

If the contract is entered into on or after January 30, 2022, or the contract is renewed or extended (e.g., an option is exercised) on or after January 30, 2022:	Executive Order 14026 generally applies to the contract. The contractor must pay all covered workers at least \$16.20 per hour (or the applicable wage rate listed on this wage determination, if it is higher) for all hours spent performing on the contract in 2023.
If the contract was awarded on or between January 1, 2015 and January 29, 2022, and the contract is not renewed or extended on or after January 30, 2022:	

The applicable Executive Order minimum wage rate will be adjusted annually. If this contract is covered by one of the Executive Orders and a classification considered necessary for performance of work on the contract does not appear on this wage determination, the contractor must still submit a conformance request.

Additional information on contractor requirements and worker protections under the Executive Orders is available at http://www.dol.gov/whd/govcontracts.

Modification Number	Publication Date
0	01/06/2023
1	02/03/2023
2	03/03/2023

BROH0001-001 06/01/2021

DEFIANCE, FULTON (Excluding Fulton, Amboy & Swan Creek Townships), HENRY (Excluding Monroe, Bartlow, Liberty, Washington, Richfield, Marion, Damascus & Townships & that part of Harrison Township outside corporate limits of city of Napoleon), PAULDING, PUTNAM and WILLIAMS COUNTIES

	Rates	Fringes
Bricklayer, Stonemason		17.55
BR0H0001-004 06/01/2021		
	Rates	Fringes
CEMENT MASON/CONCRETE FINISHER	\$ 30.40	17.55

BROH0003-002 06/01/2021

FULTON (Townships of Amboy, Swan Creek & Fulton), HENRY (Townships of Washington, Damascus, Richfield, Bartlow, Liberty, Harrison, Monroe, & Marion), LUCAS and WOOD (Townships of Perrysburg, Ross, Lake, Troy, Freedom, Montgomery, Webster, Center, Portage, Middleton, Plain, Liberty, Henry, Washington, Weston, Milton, Jackson & Grand Rapids) COUNTIES

	Rates	Fringes
Bricklayer, Stonemason		17.55
BROH0005-003 06/01/2020		
CUYAHOGA, LORAIN & MEDINA (Hincd Liverpool, Montville, York, Hom Litchfield & Spencer Townships a	er, Harrisville,	Chatham,
	Rates	Fringes
BRICKLAYER BRICKLAYERS; CAULKERS; CLEANERS; POINTERS; & STONEMASONS SANDBLASTERS SEWER BRICKLAYERS & STACK BUILDERS SWING SCAFFOLDS	\$ 36.64 \$ 37.14	17.13 17.13 17.13 17.13 17.13
BROH0006-005 05/01/2022		
CARROLL, COLUMBIANA (Knox, Butl STARK & TUSCARAWAS	er, West & Hanov	ver Townships),
	Rates	Fringes
Bricklayer, Stonemason	.\$ 30.76	19.07
BR0H0007-002 06/01/2021		
LAWRENCE		
	Rates	Fringes
Bricklayer, Stonemason		17.55
BR0H0007-005 06/01/2021		
PORTAGE & SUMMIT		
	Rates	Fringes
BRICKLAYER		17.55
BR0H0007-010 06/01/2017		
PORTAGE & SUMMIT		
	Rates	Fringes
MASON - STONE		14.55
BR0H0008-001 06/01/2021		
COLUMBIANA (Salem, Perry, Fairf: Middleton, & Unity Townships and MAHONING & TRUMBULL		
	Rates	Fringes
BRICKLAYER		17.55
BROH0009-002 06/01/2021		
BELMONT & MONROE COUNTIES and the Pleasant and the Village of Di		
	Rates	Fringes
Bricklayer, Stonemason Refractory	.\$ 31.45	17.55 19.01

	Rates	Fringes
Bricklayer, Stonemason		19.07
BROH0007-002 06/01/2021		
LAWRENCE		
	Rates	Fringes
Bricklayer, Stonemason	\$ 30.40	17.55
BROH0007-005 06/01/2021		
PORTAGE & SUMMIT		
	Rates	Fringes
BRICKLAYER		17.55
BROH0007-010 06/01/2017		
PORTAGE & SUMMIT		
	Rates	Fringes
MASON - STONE		14.55
BROH0008-001 06/01/2021		
COLUMBIANA (Salem, Perry, Fairfie Middleton, & Unity Townships and MAHONING & TRUMBULL		
	Rates	Fringes
BRICKLAYER BROH0009-002 06/01/2021		17.55
	• Townships of W	arren & Mt.

COLUMBIANA (St. Clair, Madison, Wayne, Franklin, Washington, Yellow Creek & Liverpool Townships) & JEFFERSON (Brush Creek & Saline Townships)

	Rates	Fringes
Bricklayer, Stonemason		
		17.55
BROH0014-002 06/01/2021		
HARRISON & JEFFERSON (Except Mt. Saline & Salineville Townships &	Pleasant, the Villag	Warren, Brush Creek, e of Dillonvale)
	Rates	Fringes
Bricklayer, Stonemason		17.55
BROH0016-002 06/01/2021		
ASHTABULA, GEAUGA, and LAKE COUN	TIES	
	Rates	Fringes
Bricklayer, Stonemason		17.55
BROH0018-002 06/01/2021		
BROWN, BUTLER, CLERMONT, HAMILTO Israel, Lanier, Somers & Gratis	N, PREBLE (Townships)	Gasper, Dixon, & WARREN COUNTIES:
	Rates	Fringes
Bricklayer, Stonemason		17.55
BROH0022-004 06/01/2021		*******
CHAMPAIGN, CLARK, CLINTON, DARKE MIAMI, MONTGOMERY, PREBLE (Jacks Jefferson & Washington Townships	on, Monroe,	Harrison, Twin,
	Rates	Fringes
Bricklayer, Stonemason	.\$ 30.40	17.55
BR0H0032-001 06/01/2021		9
GALLIA & MEIGS		
	Rates	Fringes
Bricklayer, Stonemason		17.55
BROH0035-002 06/01/2021		•••••••
ALLEN, AUGLAIZE, MERCER and VAN	WERT COUNTI	ES
	Rates	Fringes
Bricklayer, Stonemason	.\$ 30.40	17.55
BROH0039-002 06/01/2021		
ADAMS & SCIOTO		
	Rates	Fringes
Bricklayer, Stonemason	\$ 30.40	17.55
BROH0040-003 06/01/2021		
ASHLAND, CRAWFORD, HARDIN, HOLME: WAYNE and WYANDOT (Except Crawfor	S, MARION, M	MORROW, RICHLAND,

Rates Fringes Bricklayer, Stonemason.....\$ 31.93 22.54 FOOTNOTE: Layout Man and Sawman rate: \$1.00 per hour above journeyman rate. Free standing stack work ground level to top of stack;

Sandblasting and laying of carbon masonry material in swing stage and/or scaffold; Ramming and spading of plastics and gunniting: \$1.50 per hour above journeyman rate.
""Hot"" work: \$2.50 above journeyman rate.

BROH0044-002	06/01/2021

Rates

Fringes

Bricklayer, Stonemason COSHOCTON, FAIRFIELD,	
GUERNSEY, HOCKING, KNOX,	
KICKING, MORGAN,	
MUSKINGUM, NOBLE (Beaver,	
Buffalo, Seneca & Wayne	
Townships) & PERRY	
COUNTIES:\$ 30.40	17.55
BROH0045-002 06/01/2021	

FAYETTE, JACKSON, PIKE, ROSS and VINTON COUNTIES

Rates	Fringes
Naces	

Bricklayer, Stonemason\$ 30.40	17.66
BR0H0046-002 06/01/2021	

ERIE, HANCOCK, HURON, OTTAWA, SANDUSKY, SENECA, WOOD (Perry & Bloom Townships) and WYANDOT (Tymochtee, Crawford, Ridge & Richland Townships) COUNTIES & the Islands of Lake Erie north of Sandusky

	I	Rates	Fringes
Bricklayer,	Stonemason\$	30.40	17.55

FOOTNOTE: Layout Man and Sawman rate: \$1.00 per hour above journeyman rate.

Free standing stack work ground level to top of stack; Sandblasting and laying of carbon masonry material in swing stage and/or scaffold; Ramming and spading of plastics and gunniting: \$1.50 per hour above journeyman rate. ""Hot"" work: \$2.50 above journeyman rate.

_____ BROH0052-001 06/01/2021

ATHENS COUNTY

	Rates	Fringes
Bricklayer, Stonemason		17.55
BR0H0052-003 06/01/2021		

NOBLE (Brookfield, Noble, Center, Sharon, Olive, Enoch, Stock, Jackson, Jefferson & Elk Townships) and WASHINGTON COUNTIES

	Rates	Fringes
Bricklayer, Stonemason	\$ 30.40	17.55
DELAWARE, FRANKLIN, MADISON, PIC	KAWAY and l	INION COUNTIES
	Rates	Fringes
Bricklayer, Stonemason	\$ 30.40	17.55
MAHONING & TRUMBULL		
	Rates	Fringes
CARPENTER CARP0069-003 05/01/2017	\$ 26.20	17.42
CARROLL, STARK, TUSCARAWAS & WAY	/NE	
	Rates	Fringes

CARPENTER\$	15.98
CARP0069-006 05/01/2017	

COSHOCTON, HOLMES, KNOX & MORROW

	Rates	Fringes
CARPENTER	•	15.29
CARP0171-002 05/01/2019		
BELMONT, COLUMBIANA, HARRISON, J	EFFERSON & MONRO	DE
	Rates	Fringes

CARPENTER	\$ 27.37	20.02
CARP0200-002 05/01/2021		

ADAMS, ATHENS, DELAWARE, FAIRFIELD, FAYETTE, FRANKLIN, GALLIA, GUERNSEY, HIGHLAND, HOCKING, JACKSON, LAWRENCE, LICKING, MADISON, MARION, MEIGS, MORGAN, MUSKINGUM, NOBLE, PERRY, PICKAWAY, PIKE, ROSS, SCIOTO, UNION, VINTON and WASHINGTON COUNTIES

1	Rates	Fringes
CARPENTER\$		20.08
Diver\$		10.40
PILEDRIVERMAN\$		20.08

CARP0248-005 07/01/2008

LUCAS & WOOD

	Rates	Fringes
CARPENTER	•	
CARP0248-008 07/01/2008		
	Rates	Fringes
CARPENTER DEFIANCE, FULTON, HANCOCK, HENRY, PAULDING & WILLIAMS COUNTIES	.\$ 23.71	13.28
CARP0254-002 05/01/2017		
ASHTABULA, CUYAHOGA, GEAUGA & LA	KE	
	Rates	Fringes
CARPENTER		16.97
CARP0372-002 05/01/2016		
ALLEN, AUGLAIZE, HARDIN, MERCER,	PUTNAM &	VAN WERT
	Rates	Fringes
CARPENTER		18.21
CARP0639-003 05/01/2017		
MEDINA, PORTAGE & SUMMIT		
	Rates	Fringes
CARPENTER		16.99
CARP0735-002 05/01/2019		
ASHLAND, ERIE, HURON, LORAIN & R	ICHLAND	
	Rates	Fringes
CARPENTER	\$ 26.30	17.91
CARP1311-001 05/01/2017		

BROWN, BUTLER, CHAMPAIGN, CLARK, CLERMONT, CLINTON, DARKE, GREENE, HAMILTON, LOGAN, MIAMI, MONTGOMERY, PREBLE, SHELBY &

724

WARREN

	Rates	Fringes
Carpenter & Piledrivermen	¢ 20 34	15.95
Diver	\$ 40.58	9.69
CARP1393-002 07/01/2008		
CRAWFORD, DEFIANCE, FULTON, HANCO PAULDING, SANDUSKY, SENECA, WILLI		
	Rates	Fringes
Piledrivermen & Diver's Tender.	\$ 27.30	16.05
DIVERS - \$250.00 per day		
CARP1393-003 07/01/2008		
ALLEN, AUGLAIZE, HARDIN, MERCER,	PUTNAM.	VAN WERT & WYANDOT
ALLEN, AUGENIZE, IMADIN, MENCEN,		Fringes
Piledrivermen & Diver's Tender	.\$ 25.15	15.92
DIVERS - \$250.00 per day		
CARP1871-006 05/01/2017		
BELMONT, HARRISON, & MONROE		
	Rates	Fringes
Diver, Wet Piledrivermen; Diver, Dry	.\$ 48.11 .\$ 32.07	17.33 17.33
CARP1871-008 05/01/2017		
ASHLAND, ASHTABULA, CUYAHOGA, ER LORAIN, MEDINA, PORTAGE, RICHLAN		
	Rates	Fringes
Diver, Wet Piledrivermen; Diver, Dry	.\$ 30.53	18.84
CARP1871-014 05/01/2017	*******	
CARROLL, STARK, TUSCARAWAS & WAY	NE	
	Rates	Fringes
Diver, Wet Piledrivermen; Diver, Dry	\$ 25.56	16.95
CARP1871-015 05/01/2017		
COSHOCTON, HOLMES, KNOX & MORROW		
	Rates	Fringes
Diver, Wet	.\$ 37.34	16.07
Piledrivermen; Diver, Dry CARP1871-017 05/01/2017	.\$ 24.89	16.07
MAHONING & TRUMBULL		
I FUIGHANG & INVISULA	Rator	Fringes
		_
Diver, Wet Piledrivermen; Diver, Dry	.\$ 27.10	17.62
CARP2235-012 01/01/2014		
COLUMBIANA & JEFFERSON		
	Rates	Fringes
PILEDRIVERMAN	.\$ 31.74	16.41
CARP2239-001 07/01/2008		

CRAWFORD, OTTAWA, SANDUSKY, SENECA & WYANDOT

Rates	Fringes
	13.28
RY, LUCAS, OT TAMS & WOOD	TAWA, PAULDING,
Rates	Fringes
\$ 44.79	18.96 4.5%+21.61
MERCER, SHE	LBY, VAN WERT & flin, Ridgeland,
Rates	Fringes
\$ 34.67	21,48
hester & Russ	sell Townships) &
Rates	Fringes
\$ 40.88	22.75
& Christmas E 1 year's ser service	vice; 2 weeks' paid
hester & Russ	ell Townships) &
Rates	Fringes
	13.29 13.25
& Christmas D 1 vear's ser	al Day; July 4th; ay vice; 2 weeks' paid
erlin, Boardm Green, Jackso	& Unity Townships) an, Canfield, n, Poland, ULL (Hubbard &
Rates	Fringes
\$ 36.10	Fringes 18.91
	IAMS & WOOD Rates Rates \$ 38.98 \$ 44.79 MERCER, SHE rseilles, Mif Rates \$ 34.67 Chester & Russ Rates \$ 40.88 s Day; Memori & Christmas D 1 year's ser service hester & Russ Rates \$ 29.30 \$ 28.05 s Day; Memori & Christmas D 1 year's ser service

FAIRFIELD, FAYETTE, FRANKLIN, GUERNSEY, HIGHLAND, HOCKING, JACKSON (Coal, Jackson, Liberty, Milton, Washington & Wellston Townships), KNOX, LICKING, MADISON, MARION, MONROE, MORGAN, MORROW, MUSKINGUM, NOBLE, PERRY, PICKAWAY, PIKE (Beaver, Benton, Jackson, Mifflin, Pebble, Peepee, Perry & Seal Townships), RICHLAND, ROSS, TUSCARAWAS (Auburn, Bucks, Clay, Jefferson, Oxford, Perry, Salem, Rush, Washington & York Townships), UNION, VINTON (Clinton, Eagle, Elk, Harrison, Jackson, Richland & Swan Townships), and WASHINGTON COUNTIES

Equipment Operators\$ 33.62 13.40 Groundmen & Cable Splicers\$ 38.27 14.42 ELECG071-004 01/01/2019 UGLAIZE, CLINTON, DARKE, GREENE, LOGAN, MERCER, MIAMI, ONTGOMERY, PREBLE, and SHELBY COUNTIES Rates Fringes ine Construction \$ 33.62 13.40 Equipment Operator\$ 33.62 13.40 Groundman		Rates	Fringes
ELEC0071-004 01/01/2019 UGLAIZE, CLINTON, DARKE, GREENE, LOGAN, MERCER, MIAMI, ONTGOMERY, PREBLE, and SHELBY COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman	Line Construction Equipment Operators	\$ 33.62	13.40
ELEC0071-004 01/01/2019 UGLAIZE, CLINTON, DARKE, GREENE, LOGAN, MERCER, MIAMI, ONTGOMERY, PREBLE, and SHELBY COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman	Groundmen	. \$ 24.17	11.32
ELEC0071-004 01/01/2019 UGLAIZE, CLINTON, DARKE, GREENE, LOGAN, MERCER, MIAMI, ONTGOMERY, PREBLE, and SHELBY COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman	Linemen & Cable Splicers.	\$ 38.27	14.42
Rates Fringes ine Construction \$ 33.62 13.40 Groundman	ELEC0071-004 01/01/2019		
RatesFringesine Construction\$ 33.6213.40Groundman	AUGLAIZE, CLINTON, DARKE, GREE		ERCER, MIAMI,
ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-005 12/31/2018 SHTABULA, CUYAHOGA, GEAUGA, LAKE & LORAIN Rates Fringes INE CONSTRUCTION: Equipment perator DOT/Traffic Signal & Highway Lighting Projects\$ 32.44 14.10 Municipal Power/Transit Projects\$ 40.10 16.42 INE CONSTRUCTION: Groundman DOT/Traffic Signal & Highway Lighting Projects\$ 25.06 12.26 Municipal Power/Transit Projects	MONIGOMERY, PREBLE, and SHELBY	COUNTIES	
Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-005 12/31/2018 SHTABULA, CUYAHOGA, GEAUGA, LAKE & LORAIN Rates Fringes INE CONSTRUCTION: Equipment perator DOT/Traffic Signal & Highway Lighting Projects\$ 32.44 14.10 Municipal Power/Transit Projects\$ 40.10 16.42 INE CONSTRUCTION: Groundman DOT/Traffic Signal & Highway Lighting Projects\$ 25.06 12.26 Municipal Power/Transit Projects\$ 31.19 14.11 INE CONSTRUCTION: Inemer/Cable Splicer DOT/Traffic Signal & Highway Lighting Projects\$ 36.13 15.03 Municipal Power/Transit Projects		Rates	Fringes
Groundman	Line Construction		
ELEC0071-005 12/31/2018 SHTABULA, CUYAHOGA, GEAUGA, LAKE & LORAIN Rates Fringes INE CONSTRUCTION: Equipment perator DOT/Traffic Signal & Highway Lighting Projects\$ 32.44 14.10 Municipal Power/Transit Projects	Equipment Operator	\$ 33.62	13.40
ELEC0071-005 12/31/2018 SHTABULA, CUYAHOGA, GEAUGA, LAKE & LORAIN Rates Fringes INE CONSTRUCTION: Equipment perator DOT/Traffic Signal & Highway Lighting Projects\$ 32.44 14.10 Municipal Power/Transit Projects	Groundman		14 42
ELEC0071-005 12/31/2018 SHTABULA, CUYAHOGA, GEAUGA, LAKE & LORAIN Rates Fringes INE CONSTRUCTION: Equipment perator DOT/Traffic Signal & Highway Lighting Projects\$ 32.44 14.10 Municipal Power/Transit Projects	Lineman & Cable Splicers.		14.42
RatesFringesINE CONSTRUCTION: Equipment peratorDOT/Traffic Signal & Highway Lighting Projects\$ 32.4414.10 Municipal Power/Transit Projects	ELEC0071-005 12/31/2018		
INE CONSTRUCTION: Equipment perator DOT/Traffic Signal & Highway Lighting Projects\$ 32.44 14.10 Municipal Power/Transit Projects	ASHTABULA, CUYAHOGA, GEAUGA, L	AKE & LORAIN	
perator DOT/Traffic Signal & Highway Lighting Projects\$ 32.44 14.10 Municipal Power/Transit Projects		Rates	Fringes
Highway Lighting Projects\$ 32.4414.10Municipal Power/TransitProjects	LINE CONSTRUCTION: Equipment Operator		
Projects	Highway Lighting Projects	\$ 32.44	14.10
DOT/Traffic Signal & Highway Lighting Projects\$ 25.06 12.26 Municipal Power/Transit Projects	Projects	\$ 40.10	16.42
Municipal Power/Transit Projects			
INE CONSTRUCTION: inemen/Cable Splicer DOT/Traffic Signal & Highway Lighting Projects\$ 36.13 15.03 Municipal Power/Transit Projects\$ 44.56 17.58 ELEC0071-008 01/01/2019 OLUMBIANA, MAHONING, and TRUMBULL COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-010 01/01/2019 ELMONT, CARROLL, HARRISON, HOLMES, JEFFERSON, MEDINA, PORTAGE, TARK, SUMMIT, and WAYNE COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 33.62 13.40 Groundman\$ 33.62 13.40 Groundman	Municipal Power/Transit		12.26
<pre>inemen/Cable Splicer DOT/Traffic Signal & Highway Lighting Projects\$ 36.13 15.03 Municipal Power/Transit Projects\$ 44.56 17.58 ELEC0071-008 01/01/2019 OLUMBIANA, MAHONING, and TRUMBULL COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 33.62 13.40 Groundman\$ 38.27 14.42 ELEC0071-010 01/01/2019 ELMONT, CARROLL, HARRISON, HOLMES, JEFFERSON, MEDINA, PORTAGE, TARK, SUMMIT, and WAYNE COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 33.62 13.40 Groundman\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman</pre>	Projects	\$ 31.19	14.11
Highway Lighting Projects\$ 36.1315.03Municipal Power/TransitProjects	Linemen/Cable Splicer		
Municipal Power/Transit Projects\$44.56 17.58 ELEC0071-008 01/01/2019 OLUMBIANA, MAHONING, and TRUMBULL COUNTIES Rates Fringes ine Construction Equipment Operator\$33.62 13.40 Groundman\$24.17 11.32 Lineman & Cable Splicers\$38.27 14.42 ELEC0071-010 01/01/2019 ELMONT, CARROLL, HARRISON, HOLMES, JEFFERSON, MEDINA, PORTAGE, TARK, SUMMIT, and WAYNE COUNTIES Rates Fringes ine Construction Equipment Operator\$33.62 13.40 Groundman\$38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$33.62 13.40 Groundman\$33.62 13.40 Groundman		e 26 13	15 07
ELEC0071-008 01/01/2019 OLUMBIANA, MAHONING, and TRUMBULL COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-010 01/01/2019 ELMONT, CARROLL, HARRISON, HOLMES, JEFFERSON, MEDINA, PORTAGE, TARK, SUMMIT, and WAYNE COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman	Municipal Power/Transit		
ELEC0071-008 01/01/2019 OLUMBIANA, MAHONING, and TRUMBULL COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-010 01/01/2019 ELMONT, CARROLL, HARRISON, HOLMES, JEFFERSON, MEDINA, PORTAGE, TARK, SUMMIT, and WAYNE COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman	Projects	\$ 44.56	17.58
RatesFringesine Construction33.6213.40Groundman	ELEC0071-008 01/01/2019		
ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-010 01/01/2019 ELMONT, CARROLL, HARRISON, HOLMES, JEFFERSON, MEDINA, PORTAGE, TARK, SUMMIT, and WAYNE COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32	COLUMBIANA, MAHONING, and TRU	IMBULL COUNTIN	ES
Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-010 01/01/2019 ELMONT, CARROLL, HARRISON, HOLMES, JEFFERSON, MEDINA, PORTAGE, TARK, SUMMIT, and WAYNE COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 ELEC0071-013 01/01/2019		Rates	Fringes
Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-010 01/01/2019 ELMONT, CARROLL, HARRISON, HOLMES, JEFFERSON, MEDINA, PORTAGE, TARK, SUMMIT, and WAYNE COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32	ine Construction	£ 33 C3	12 40
ELINEMAIN & CADLE Spitters, 38.27 14.42 ELEC0071-010 01/01/2019 ELMONT, CARROLL, HARRISON, HOLMES, JEFFERSON, MEDINA, PORTAGE, TARK, SUMMIT, and WAYNE COUNTIES Rates Fringes ine Construction Equipment Operator	Equipment Operator		11 37
ELEC0071-010 01/01/2019 ELMONT, CARROLL, HARRISON, HOLMES, JEFFERSON, MEDINA, PORTAGE, TARK, SUMMIT, and WAYNE COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32	LTHEMPHI & CADLE SUTTCELS		74.47
ELMONT, CARROLL, HARRISON, HOLMES, JEFFERSON, MEDINA, PORTAGE, TARK, SUMMIT, and WAYNE COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32			
Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction \$ 33.62 13.40 Groundman	BELMONT, CARROLL, HARRISON, HO		SON, MEDINA, PORTAGE,
ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32	STARK, SUMMIT, and WAYNE COUNT	IES	
Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32		Rates	Fringes
Groundman\$ 24.17 11.32 Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32	Line Construction		
Lineman & Cable Splicers\$ 38.27 14.42 ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32			
ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32			
ELEC0071-013 01/01/2019 ROWN, BUTLER, CLERMONT, HAMILTON, and WARREN COUNTIES Rates Fringes ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32			
RatesFringesine Construction33.6213.40Equipment Operator\$24.1711.32	ELEC0071-013 01/01/2019		~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
ine Construction Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32	BROWN, BUTLER, CLERMONT, HAMIL	TON, and WAR	REN COUNTIES
Equipment Operator\$ 33.62 13.40 Groundman\$ 24.17 11.32		Rates	Fringes
Groundman\$ 24.17 11.32	Line Construction	¢ 33 43	13 /0
Linoman & Cable Splicens (2007) 14.43			
	Lineman & Cable Splicers	\$ 38.27	14.42

ELEC0071-014 01/01/2019

ADAMS, ATHENS, GALLIA, JACKSON (Bloomfield, Franklin, Hamilton, Lick, Jefferson, Scioto & Madison Townships), LAWRENCE, MEIGS, PIKE (Camp Creek, Marion, Newton, Scioto, Sunfish & Union Townships), SCIOTO & VINTON (Brown, Knox, Madison, Vinton & Wilkesville Townships)

		Rates	Fringes
Line	Construction		
	Equipment Operator\$	33.62	13.40
	Groundman\$	24.17	11.32
	Lineman & Cable Splicers\$		14.42

ELEC0082-002 12/05/2022

CLINTON, DARKE, GREENE, MIAMI, MONTGOMERY, PREBLE & WARREN (Wayne, Clear Creek & Franklin Townships)

	Rates	Fringes
ELECTRICIAN		21.26
* ELEC0082-006 11/29/2021		

CLINTON, DARKE, GREENE, MIAMI, MONTGOMERY, PREBLE & WARREN (Wayne, Clear Creek & Franklin Townships)

	Rates	Fringes
Sound & Communication Technician		
Cable Puller		3.89
Installer/Technician		12.27
* ELEC0129-003 02/27/2023		*************************

LORAIN (Except Columbia Township) & MEDINA (Litchfield & Liverpool Townships)

	Rates	Fringes
ELECTRICIAN	\$ 39.30	18.30

* ELEC0129-004 02/27/2023

ERIE & HURON (Lyme, Ridgefield, Norwalk, Townsend, Wakeman, Sherman, Peru, Bronson, Hartland, Clarksfield, Norwich, Greenfield, Fairfield, Fitchville & New London Townships)

	Rates	Fringes
ELECTRICIAN	\$ 39,30	18.30
ELEC0141-003 09/01/2019		
BELMONT COUNTY		
	Rates	Fringes
CABLE SPLICER	30.38	25.87 25.87
BROWN, CLERMONT & HAMILTON		
	Rates	Fringes
Sound & Communication Technician	24.35	10.99
BROWN, CLERMONT, and HAMILTON COUN	TIES	
	Rates	Fringes
ELECTRICIAN\$		21.15

ELEC0245-001 08/29/2022

ALLEN, HARDIN, VAN WERT & WYANDOT (Crawford, Jackson, Marseilles, Mifflin, Richland, Ridge & Salem Townships)

Rates	Fringes
Line Construction	
Equipment Operator\$ 32.37	26.5%+7.25
Groundman Truck Driver\$ 19.35	7.00+27.25%
Lineman\$ 44.22	7.00+27.25%

FOOTNOTE: a. Half day's Paid Holiday: The last 4 hours of the workday prior to Christmas or New Year's Day

ELEC0245-003 08/29/2022

DEFIANCE, FULTON, HANCOCK, HENRY, HURON, LUCAS, OTTAWA, PAULDING, PUTNAM, SANDUSKY, SENECA, WILLIAMS, and WOOD COUNTIES

	Rates	Fringes
Line Construction Cable Splicer Groundman/Truck Driver Heli-arc Welding Lineman Operator - Class 1 Operator - Class 2 Traffic Signal & Lighting Technician.	.\$ 19.35 .\$ 40.76 .\$ 44.22 .\$ 35.38 .\$ 28.32	7.00+27.25% 7.00+27.25% 7.00+27.25% 7.00+27.25% 7.00+27.25% 7.00+27.25% 7.00+27.25%

FOOTNOTE: a. 6 Observed Holidays: New Year's Day; Memorial Day; Independence Day; Labor Day; Thanksgiving Day; & Christmas Day. Employees who work on a holiday shall be paid at a rate of double their applicable classified straight-time rates for the work performed on such holiday.

ELEC0245-004 08/29/2022

ERIE COUNTY

Rates	Fringes
Line Construction	
Cable Splicer\$ 49.14	26.75%+6.75
Cablesplicer 50.85	7.00+27.25%
Groundman/Truck Driver\$ 19.35	7.00+27.25%
Lineman\$ 44.22	7.00+27.25%
Operator - Class 1\$ 35.38	7.00+27.25%
Operator - Class 2\$ 28.32	7.00+27.25%

FOOTNOTE: a. 6 Observed Holidays: New Year's Day; Memorial Day; Independence Day; Labor Day; Thanksgiving Day; & Christmas Day. Employees who work on a holiday shall be paid at a rate of double their applicable classified straight-time rates for the work performed on such holiday.

* ELEC0246-001 10/31/2022	

	Rates	Fringes
ELECTRICIAN	\$ 40.50	84%+36.47

FOOTNOTE: a. 1 1/2 Paid Holidays: The last scheduled workday prior to Christmas & 4 hours on Good Friday.

.....

ELEC0306-005 05/28/2018

MEDINA (Brunswick, Chatham, Granger, Guilford, Harrisville, Hinckley, Homer, Lafayette, Medina, Montville, Sharon, Spencer, Wadsworth, Westfield & York Townships), PORTAGE (Atwater, Aurora, Brimfield, Deerfield, Franklin, Mantua, Randolph, Ravenna, Rootstown, Shalersville, Streetsboro & Suffield Townships), SUMMIT & WAYNE (Baughman, Canaan, Chester, Chippewa, Congress, Green, Milton, & Wayne Townships)

I	Rates	Fringes
CABLE SPLICER\$	36.87	16.56
ELECTRICIAN\$	34.54	5%+18.06

ELEC0317-002 05/30/2022

GALLIA & LAWRENCE

	Rates	Fringes
CABLE SPLICER		18.13
ELECTRICIAN	35.85	28.25

ELEC0540-005 12/27/2021

CARROLL (Northern half, including Fox, Harrison, Rose & Washington Townhships), COLUMBIANA (Knox Township), HOLMES, MAHONING (Smith Township), STARK, TUSCARAWAS (North of Auburn, Clay, Rush & York Townships), and WAYNE (South of Baughman, Chester, Green & Wayne Townships) COUNTIES

	Rates	Fringes
ELECTRICIAN	\$ 35.28	22.63
ELEC0573-003 11/28/2022		******

ASHTABULA (Colebrook, Wayne, Williamsfield, Orwell & Windsor Townships), GEAUGA (Auburn, Middlefield, Parkman & Troy Townships), MAHONING (Milton Township), PORTAGE (Charlestown, Edinburg, Freedom, Hiram, Nelson, Palmyra, Paris & Windham Townships), and TRUMBULL (Except Liberty & Hubbard Townships)

	Rates	Fringes	
ELECTRICIAN	\$ 38.70	20.94	

* ELEC0575-001 11/21/2022

ADAMS, FAYETTE, HIGHLAND, HOCKING, JACKSON (Bloomfield, Franklin, Hamilton, Jefferson, Lick, Madison, Scioto, Coal, Jackson, Liberty, Milton & Washington Townships), PICKAWAY (Deer Creek, Perry, Pickaway, Salt Creek & Wayne Townships), PIKE (Beaver, Benton, Jackson, Mifflin, Pebble, PeePee, Perry, Seal, Camp Creek, Newton, Scioto, Sunfish, Union & Marion Townships), ROSS, SCIOTO & VINTON (Clinton, Eagle, Elk, Harrison, Jackson, Richland & Swan Townships)

	Rates	Fringes
ELECTRICIAN	.\$ 36.00	21.14

ELEC0648-001 08/29/2022

BUTLER and WARREN COUNTIES (Deerfield, Hamilton, Harlan, Massie, Salem, Turtle Creek, Union & Washington Townships)

	Rates	Fringes	
CABLE SPLICER		18.23	
ELECTRICIAN	\$ 33.00	21.44	

* ELEC0673-004 01/01/2023

ASHTABULA (Excluding Orwell, Colebrook, Williamsfield, Wayne & Windsor Townships), GEAUGA (Burton, Chardon, Claridon, Hambden, Huntsburg, Montville, Munson, Newbury & Thompson Townships) and LAKE COUNTIES

	Rates	Fringes
CABLE SPLICER		21.47
ELECTRICIAN	\$ 35.15	23.41

ELEC0683-002 05/30/2022

CHAMPAIGN, CLARK, DELAWARE, FAIRFIELD, FRANKLIN, MADISON, PICKAWAY (Circleville, Darby, Harrison, Jackson, Madison, Monroe, Muhlenberg, Scioto, Walnut & Washington Townships), and UNION COUNTIES

	Rates	Fringes
CABLE SPLICER	37.50	23,15
ELECTRICIAN	36.50	23.15

ELEC0688-003 05/30/2022

ASHLAND, CRAWFORD, HURON (Richmond, New Haven, Ripley & Greenwich Townships), KNOX (Liberty, Clinton, Union, Howard, Monroe, Middleberry, Morris, Wayne, Berlin, Pike, Brown & Jefferson Townships), MARION, MORROW, RICHLAND and WYANDOT (Sycamore, Crane, Eden, Pitt, Antrim & Tymochtee Townships) COUNTIES

	Rates	Fringes
ELECTRICIAN	.\$ 32.30	21.83

ELEC0972-002 06/01/2021

ATHENS, MEIGS, MONROE, MORGAN, NOBLE, VINTON (Brown, Knox, Madison, Vinton & Wilkesville Townships), and WASHINGTON COUNITES

	Rates	Fringes
CABLE SPLICER\$		27.81
ELECTRICIAN\$	34.30	27.62

ELEC1105-001 05/30/2022

COSHOCTON, GUERNSEY, KNOX (Jackson, Clay, Morgan, Miller, Milford, Hilliar, Butler, Harrison, Pleasant & College Townships), LICKING, MUSKINGUM, PERRY, and TUSCARAWAS (Auburn, York, Clay, Jefferson, Rush, Oxford, Washington, Salem, Perry & Bucks Townships) COUNTIES

	Rates	Fringes
ELECTRICIAN	\$ 35.25	22.18
ENGI0018-003 05/01/2019	•••••	

ASHTABULA, CUYAHOGA, ERIE, GEAUGA, LAKE, LORAIN, MEDINA, PORTAGE, and SUMMIT COUNTIES

	Rates	Fringes
POWER EQUIPMENT OPERATOR		
GROUP 1	.\$ 38.63	15.20
GROUP 2	.\$ 38.53	15.20
GROUP 3	.\$ 37,49	15.20
GROUP 4	.\$ 36.27	15.20
GROUP 5	.\$ 30.98	15.20
GROUP 6	.\$ 38.88	15.20
GROUP 7	.\$ 39.13	15.20

OPERATING ENGINEER CLASSIFICATIONS

GROUP 1 - Air Compressor on Steel Erection; Barrier Moving Machine; Boiler Operator on Compressor or Generator when mounted on a Rig; Cableway; Combination Concrete Mixer & Tower; Concrete Plant (over 4 yd. Capacity); Concrete Pump; Crane (All Types, Including Boom Truck, Cherry Picker); Crane-Compact, Track or Rubber over 4,000 lbs. capacity; Cranes-Self Erecting, Stationary, Track or Truck (All Configurations); Derrick; Dragline; Dredge (Dipper, Clam or Suction); Elevating Grader or Euclid Loader; Floating Equipment (All Types); Gradall; Helicopter Crew (Operator-Hoist or Winch); Hoe (all types); Hoisting Engine on Shaft or Tunnel Work; Hydraulic Gantry (Lifting System); Industrial-Type Tractor; Jet Engine Dryer (D8 or D9) Diesel Tractor; Locomotive (Standard Gauge); Maintenance Operator Class A; Mixer, Paving (Single or Double Drum); Mucking Machine; Multiple Scraper; Piledriving Machine (All Types); Power Shovel; Prentice Loader; Quad 9 (Double Pusher); Rail Tamper (with auto lifting & aligning device); Refrigerating Machine (Freezer Operation); Rotary Drill, on Caisson work; Rough Terrain Fork Lift with Winch/Hoist; Side-Boom; Slip-Form Paver; Tower Derrick; Tree Shredder; Trench Machine (Over 24"" wide); Truck Mounted Concrete Pump; Tug Boat; Tunnel Machine and/or Mining Machine; Wheel Excavator; and Asphalt Plant Engineer (Cleveland District Only).

GROUP 2 - Asphalt Paver; Automatic Subgrader Machine, Self-Propelled (CMI Type); Bobcat Type and/or Skid Steer Loader with Hoe Attachment Greater than 7,000 lbs.; Boring Machine More than 48"; Bulldozer; Endloader; Horizontal Directional Drill (Over 50,000 ft lbs thrust); Hydro Milling Machine; Kolman-type Loader (production type-Dirt); Lead Greaseman; Lighting & Traffic Signal Installation Equipment (includes all groups or classifications); Material Transfer Equipment (Shuttle Buggy) Asphalt; Pettibone-Rail Equipment; Power Grader; Power Scraper; Push Cat; Rotomill (all), Grinders & Planers of All types; Trench Machine (24" wide & under); Vermeer type Concrete Saw; and Maintenance Operators (Portage and Summit Counties Only).

GROUP 3 - A-Frame; Air Compressor on Tunnel Work (low pressure); Asphalt Plant Engineer (Portage and Summit Counties Only); Bobcat-type and/or Skid Steer Loader with or without Attachments; Highway Drills (all types); Locomotive (narrow gauge); Material Hoist/Elevator; Mixer, Concrete (more than one bag capacity); Mixer, one bag capacity (Side Loader); Power Boiler (Over 15 lbs. Pressure) Pump Operator installing & operating Well Points; Pump (4^{***} & over discharge); Roller, Asphalt; Rotovator (lime soil stabilizer); Switch & Tie Tampers (without lifting & aligning device); Utility Operator (Small equipment); Welding Machines; and Railroad Tie Inserter/Remover; Articulating/straight bed end dumps if assigned (minus \$4.00 per hour.

GROUP 4 - Backfiller; Ballast Re-locator; Bars, Joint & Mesh Installing Machine; Batch Plant; Boring Machine Operator (48"" or less); Bull Floats; Burlap & Curing Machine; Concrete Plant (capacity 4 yd. & under); Concrete Saw (Multiple); Conveyor (Highway); Crusher; Deckhand; Farm-type Tractor with attachments (highway); Finishing Machine; Fireperson, Floating Equipment (all types); Forklift; Form Trencher; Hydro Hammer expect masonary; Hydro Seeder; Pavement Breaker; Plant Mixer; Post Driver; Post Hole Digger (Power Auger); Power Brush Burner; Power Form Handling Equipment; Road Widening Trencher; Roller (Brick, Grade & Macadam); Self-Propelled Power Spreader; Self-Propelled Power Subgrader; Steam Fireperson; Tractor (Pulling Sheepfoot, Roller or Grader); and Vibratory Compactor with Integral Power.

GROUP 5 - Compressor (Portable, Sewer, Heavy & Highway); Drum Fireperson (Asphalt Plant); Generator; Masonry Fork Lift; Inboard-Outboard Motor Boat Launch; Oil Heater (asphalt plant); Oiler/Helper; Power Driven Heater; Power Sweeper & Scrubber; Pump (under 4"" discharge); Signalperson; Tire Repairperson; VAC/ALLS; Cranes - Compact, track or rubber under 4,000 pound capacity; fueling and greasing; and Chainmen.

GROUP 6 - Master Mechanic & Boom from 150 to 180.

GROUP 7 - Boom from 180 and over.

ENGI0018-004 05/01/2019

ADAMS, ALLEN, ASHLAND, ATHENS, AUGLAIZE, BELMONT, BROWN, BUTLER, CARROLL, CHAMPAIGN, CLARK, CLERMONT, CLINTON, COSHOCTON, CRAWFORD, DARKE, DEFIANCE, DELAWARE, FAIRFIELD, FAYETTE, FRANKLIN, FULTON, GALLIA, GREENE, GUERNSEY, HAMILTON, HANCOCK, HARDIN, HARRISON, HENRY, HIGHLAND, HOCKING, HOLMES, HURON, JACKSON, JEFFERSON, KNOX, LAWRENCE, LICKING, LOGAN, LUCAS, MADISON, MARION, MEIGS, MERCER, MIAMI, MONROE, MONTGOMERY, MORGAN, MORROW, MUSKINGUM, NOBLE, OTTAWA, PAULDING, PERRY, PICKAWAY, PIKE, PREBLE, PUTNAM, RICHLAND, ROSS, SANDUSKY, SCIOTO, SENECA, SHELBY, STARK, TUSCARAWAS, UNION, VAN WERT, VINTON, WARREN, WASHINGTON, WAYNE, WILLIAMS, WOOD, and YANDOT COUNTIES

	Rates	Fringes
OWER EQUIPMENT OPERATOR		
GROUP 1	\$ 37.14	15.20
GROUP 2	\$ 37.02	15.20
GROUP 3	\$ 35.98	15.20
GROUP 4	\$ 34.80	15.20
GROUP 5		15.20
GROUP 6	\$ 37.39	15.20

15.20

OPERATING ENGINEER CLASSIFICATIONS

GROUP 7.....\$ 37.64

P

GROUP 1 - Air Compressor on Steel Erection; Barrier Moving Machine; Boiler Operator on Compressor or Generator when mounted on a Rig; Cableway; Combination Concrete Mixer & Tower; Concrete Plant (over 4 yd. Capacity); Concrete Pump; Crane (All Types, Including Boom Truck, Cherry Picker); Crane-Compact, Track or Rubber over 4,000 lbs. capacity; Cranes-Self Erecting, Stationary, Track or Truck (All Configurations); Derrick; Dragline; Dredge (Dipper, Clam or Suction); Elevating Grader or Euclid Loader; Floating Equipment (All Types); Gradall; Helicopter Crew (Operator-Hoist or Winch); Hoe (all types); Hoisting Engine on Shaft or Tunnel Work; Hydraulic Gantry (Lifting System); Industrial-Type Tractor; Jet Engine Dryer (DB or D9) Diesel Tractor; Locomotive (Standard Gauge); Maintenance Operator Class A; Mixer, Paving (Single or Double Drum); Mucking Machine; Multiple Scraper; Piledriving Machine (All Types); Power Shovel; Prentice Loader; Quad 9 (Double Pusher); Rail Tamper (with auto lifting & aligning device); Refrigerating Machine (Freezer Operation); Rotary Drill, on Caisson work; Rough Terrain Fork Lift with Winch/Hoist; Side-Boom; Slip-Form Paver; Tower Derrick; Tree Shredder; Trench Machine (Over 24" wide); Truck Mounted Concrete Pump; Tug Boat; Tunnel Machine and/or Mining Machine; and Wheel Excavator.

GROUP 2 - Asphalt Paver; Automatic Subgrader Machine, Self-Propelled (CMI Type); Bobcat Type and/or Skid Steer Loader with Hoe Attachment Greater than 7,000 lbs.; Boring Machine More than 48""; Bulldozer; Endloader; Hydro Milling Machine; Horizontal Directional Drill (over 50,000 ft. lbs. thrust);Kolman-type Loader (production type-Dirt); Lead Greaseman; Lighting & Traffic Signal Installation Equipment (includes all groups or classifications); Material Transfer Equipment (Shuttle Buggy) Asphalt; Pettibone-Rail Equipment; Power Grader; Power Scraper; Push Cat; Rotomill (all), Grinders & Planers of All types; Trench Machine (24"" wide & under); and Vermeer type Concrete Saw.

GROUP 3 - A-Frame; Air Compressor on Tunnel Work (low pressure); Asphalt Plant Engineer; Bobcat-type and/or Skid Steer Loader with or without Attachments; Highway Drills (all types); Locomotive (narrow gauge); Material Hoist/Elevator; Mixer, Concrete (more than one bag capacity); Mixer, one bag capacity (Side Loader); Power Boiler (Over 15 lbs. Pressure) Pump Operator installing & operating Well Points; Pump (4"" & over discharge); Railroad Tie Inserter/Remover; Roller, Asphalt; Rotovator (lime soil stabilizer); Switch & Tie Tampers (without lifting & aligning device); Utility Operator (Small equipment); and Welding Machines; Artiaculating/straight bed end dumps if assigned (minus \$4.00 per hour.

GROUP 4 - Backfiller; Ballast Re-locator; Bars, Joint & Mesh Installing Machine; Batch Plant; Boring Machine Operator (48" or less); Bull Floats; Burlap & Curing Machine; Concrete Plant (capacity 4 yd. & under); Concrete Saw (Multiple); Conveyor (Highway); Crusher; Deckhand; Farm-type Tractor with attachments (highway); Finishing Machine; Fireperson, Floating Equipment (all types); Fork Lift; Form Trencher; Hydro Hammer expect masonary; Hydro Seeder; Pavement Breaker; Plant Mixer; Post Driver; Post Hole Digger (Power Auger); Power Brush Burner; Power Form Handling Equipment; Road Widening Trencher; Roller (Brick, Grade & Macadam); Self-Propelled Power Spreader; Self-Propelled Power Subgrader; Steam Fireperson; Tractor (Pulling Sheepfoot, Roller or Grader); and Vibratory Compactor with Integral Power.

GROUP 5 - Compressor (Portable, Sewer, Heavy & Highway); Drum Fireperson (Asphalt Plant); Generator; Masonary Forklift; Inboard-Outboard Motor Boat Launch; Oil Heater (asphalt plant); Oiler/Helper; Power Driven Heater; Power Sweeper & Scrubber; Pump (under 4"" discharge); Signalperson; Tire Repairperson; VAC/ALLS; Cranes - Compact, track or rubber under 4,000 pound capacity; fueling and greasing; and Chainmen.

GROUP 6 - Master Mechanic & Boom from 150 to 180.

GROUP 7 - Boom from 180 and over.

ENGI0066-023 06/01/2017

Rates

COLUMBIANA, MAHONING & TRUMBULL COUNTIES

Fringes

POWER EQUIPMENT OPERATOR	
ASBESTOS; HAZARDOUS/TOXIC	
WASTE PROJECTS	
GROUP 1 - A & B\$ 39.23 ASBESTOS; HAZARDOUS/TOXIC	19.66
WASTE PROJECTS	
GROUP 2 - A & B\$ 38.90	19,66
ASBESTOS; HAZARDOUS/TOXIC	19.00
WASTE PROJECTS	
GROUP 3 - A & B\$ 34.64	19.66
ASBESTOS; HAZARDOUS/TOXIC	19.00
WASTE PROJECTS	
GROUP 4 - A & B\$ 30.70	19.66
ASBESTOS; HAZARDOUS/TOXIC	
WASTE PROJECTS	
GROUP 5 - A & B\$ 27.30	19.66
HAZARDOUS/TOXIC WASTE	
PROJECTS	
GROUP 1 - C & D\$ 35.96	19.66
HAZARDOUS/TOXIC WASTE	
PROJECTS	
GROUP 2 - C & D\$ 35.66	19.66
HAZARDOUS/TOXIC WASTE	
PROJECTS GROUP 3 - C & D\$ 31.76	
HAZARDOUS/TOXIC WASTE	19.66
PROJECTS	
GROUP 4 - C & D\$ 28.14	19.66
HAZARDOUS/TOXIC WASTE	19.00
PROJECTS	
GROUP 5 - C & D\$ 25.03	19.66
ALL OTHER WORK	10.00
GROUP 1\$ 32.69	19.66
ALL OTHER WORK	
GROUP 2\$ 32.42	19.66
ALL OTHER WORK	
GROUP 3\$ 28.87	19.66
ALL OTHER WORK	
GROUP 4\$ 25.58	19.66
ALL OTHER WORK	
GROUP 5\$ 22.75	19.66

GROUP 1 - Rig, Pile Driver or Caisson Type; & Rig, Pile Hydraulic Unit Attached

GROUP 2 - Asphalt Heater Planer; Backfiller with Drag Attachment; Backhoe; Backhoe with Shear attached; Backhoe-Rear Pivotal Swing; Batch Plant-Central Mix Concrete; Batch Plant, Portable concrete; Berm Builder-Automatic; Boat Derrick; Boat-Tug; Boring Machine Attached to Tractor; Bullclam; Bulldozer; C.M.I. Road Builder & Similar Type; Cable Placer & Layer; Carrier-Straddle; Carryall-Scraper or Scoop; Chicago Boom; Compactor with Blade Attached; Concrete Saw (Vermeer or similar type); Concrete Spreader Finisher; Combination, Bidwell Machine; Crane; Crane-Electric Overhead; Crane-Rough Terrain; Crane-Side Boom; Crane-Truck; Crane-Tower; Derrick-Boom; Derrick-Car; Digger-Wheel (Not trencher or road widener); Double Nine; Drag Line; Dredge; Drill-Kenny or Similar Type; Easy Pour Median Barrier Machine (or similar type); Electromatic; Frankie Pile; Gradall; Grader; Gurry; Self-Propelled; Heavy Equipment Robotics Operator/Mechanic; Hoist-Monorail; Hoist-Stationary & Mobile Tractor; Hoist, 2 or 3 drum; Horizontal Directional Drill Operator; Jackall; Jumbo Machine; Kocal & Kuhlman; Land-Seagoing Vehicle; Loader, Elevating; Loader, Front End; Loader, Skid Steer; Locomotive; Mechanic/Welder; Metro Chip Harvester with Boom; Mucking Machine; Paver-Asphalt Finishing Machine; Paver-Road Concrete; Paver-Slip Form (C.M.I. or similar); Place Crete Machine with Boom; Post Driver (Carrier mounted); Power Driven Hydraulic Pump & Jack (When used in Slip Form or Lift Slab Construction); Pump Crete Machine; Regulator-Ballast; Hydraulic Power Unit not attached to Rig for Pile Drillings; Rigs-Drilling; Roto Mill or similar Full Lane (8' Wide & Over); Roto Mill or similar type (Under 8'); Shovel; Slip Form Curb Machine; Speedwing; Spikemaster; Stonecrusher; Tie Puller & Loader; Tie Tamper; Tractor-Double Boom; Tractor with Attachments; Truck-Boom; Truck-Tire; Trench Machine; Tunnel Machine (Mark 21 Java or similar); & Whirley (or similar type)

GROUP 3 - Asphalt Plant; Bending Machine (Pipeline or similar type); Boring machine, Motor Driven; Chip Harvester without Boom; Cleaning Machine, Pipeline Type; Coating Machine, Pipeline Type; Compactor; Concrete Belt Placer;

734

Concrete Finisher; Concrete Planer or Asphalt; Concrete Spreader; Elevator; Fork Lift (Home building only); Fork lift & Lulls; Fork Lift Walk Behind (Hoisting over 1 buck high); Form Line Machine; Grease Truck operator; Grout Pump; Gunnite Machine; Horizontal Directional Drill Locator; Single Drum Hoist with or without Tower; Huck Bolting Machine; Hydraulic Scaffold (Hoisting building materials); Paving Breaker (Self-propelled or Ridden); Pipe Dream; Pot Fireperson (Power Agitated); Refrigeration Plant; Road Widener; Roller; Sasgen Derrick; Seeding Machine; Soil Stabilizer (Pump type); Spray Cure Machine, Self-Propelled; Straw Blower Machine; Sub-Grader; Tube Finisher or Broom C.M.I. or similar type; & Tugger Hoist

GROUP 4 - Air Curtain Destructor & Similar Type; Batch Plant-Job Related; Boiler Operator; Compressor; Conveyor; Curb Builder, self-propelled; Drill Wagon; Generator Set; Generator-Steam; Heater-Portable Power; Hydraulic Manipulator Crane; Jack-Hydraulic Power driven; Jack-Hydraulic (Railroad); Ladavator; Minor Machine Operator; Mixer-Concrete; Mulching Machine; Pin Puller; Power Broom; Pulverizer; Pump; Road Finishing Machine (Pull Type); Saw-Concrete-Self-Propelled (Highway Work); Signal Person; Spray Cure Machine-Motor Powered; Stump Cutter; Tractor; Trencher Form; Water Blaster; Steam Jenny; Syphon; Vibrator-Gasoline; & Welding Machine

GROUP 5 - Brakeperson; Fireperson; & Oiler

IRON0017-002 05/01/2022

ASHTABULA (North of Route 6, starting at the Geauga County Line, proceeding east to State Route 45), CUYAHOGA, ERIE (Eastern 2/3), GEAUGA, HURON (East of a line drawn from the north border through Monroeville & Willard), LAKE, LORAIN, MEDINA (North of Old Rte. #224), PORTAGE (West of a line from Middlefield to Shalersville to Deerfield), and SUMMIT (North of Old Rte. #224, including city limits of Barberton) COUNTIES

Rates

Fringes

IRONWORKER Ornamental, Reinforcing, & Structural.....\$ 34.33 27.51 -----IRON0017-010 05/01/2022

ASHTABULA (Eastern part from Lake Erie on the north to route #322 on the south to include Conneaut, Kingsville, Sheffield, Denmark, Dorset, Cherry Valley, Wayne, Monroe, Pierpont, Richmond, Andover & Williamsfield Townships)

	Rates	Fringes
IRONWORKER		1
Structural, including		
metal building erection &		
Reinforcing	\$ 34.33	27.51

IRON0044-001 06/01/2022

ADAMS (Western Part), BROWN, BUTLER (Southern Part), CLERMONT, CLINTON (South of a line drawn from Blanchester to Lynchburg), HAMILTON, HIGHLAND (Excluding eastern one-fifth & portion of county inside lines drawn from Marshall to Lynchburg from the northern county line through E. Monroe to Marshall) and WARREN (South of a line drawn from Blanchester through Morrow to the west county line) COUNTIES

	Rates	Fringes
IRONWORKER, REINFORCING Beyond 30-mile radius of	.\$ 32.37	22.30
Hamilton County Courthouse. Up to & including 30-mile radius of Hamilton County	\$ 28.67	21.20
Courthouse	\$ 27.60	20.70

IRON0044-002 06/01/2022

CLINTON (South of a line drawn from Blanchester to Lynchburg), HAMILTON, HIGHLAND (Excluding eastern one-fifth & portion of county inside lines drawn from Marshall to Lynchburg from the

northern county line through E. Monroe to Marshall) & WARREN (South of a line drawn from Blanchester through Morrow to the west county line)

	Rates	Fringes
IRONWORKER		
Fence Erector	\$ 30.28	22,30
Ornamental; Structural	\$ 31.87	22.30

IRON0055-003 07/01/2021

CRAWFORD (Area Between lines drawn from where Hwy #598 & #30 meet through N. Liberty to the northern border & from said Hwy junction point due west to the border), DEFIANCE (S. of a line drawn from where Rte. #66 meets the northern line through Independence to the eastern county border), ERIE (Western 1/3), FULTON, HANCOCK, HARDIN (North of a line drawn from Maysville to a point 4 miles south of the northern line on the eastern line), HENRY, HURON (West of a line drawn from the northern border through Monroeville & Willard), LUCAS, OTTAWA, PUTNAM (East of a line drawn from the northern border down through Miller City to where #696 meets the southern border), SANDUSKY, SENECA, WILLIAMS (East of a line drawn from Pioneer through Stryker to the southern border), WOOD & WYANDOT (North of Rte. #30)

	Rates	Fringes
IRONWORKER		
Fence Erector	.\$ 21.30	20.92
Flat Road Mesh	.\$ 29.77	21,30
Tunnels & Caissons Under		
Pressure	.\$ 29.77	21.30
All Other Work	\$ 31.25	26.90
IRON0147-002 06/01/2022	•••••	

ALLEN (Northern half), DEFIANCE (Northern part, excluding south of a line drawn from where Rte. #66 meets the northern line through Independence to the eastern county border) MERCER

through Independence to the eastern county border), MERCER (Northern half), PAULDING, PUTNAM (Western part, excluding east of a line drawn from the northern border down through Miller City to where #696 meets the southern border), VAN WERT, and WILLIAMS (Western part, excluding east of a line drawn from Pioneer through Stryker to the southern border) COUNTIES

	Rates	Fringes
IRONWORKER	.\$ 31.20	28.47
IRON0172-002 06/01/2022		

CHAMPAIGN (Eastern one-third), CLARK (Eastern one-fourth), COSMOCTON (West of a line beginning at the northwestern county line going through Walhonding & Tunnel Hill to the southern county line), CRAWFORD (South of Rte. #30), DELAWARE, FAIRFIELD, FAYETTE, FRANKLIN, MARDIN (Excluding a line drawn from Roundhead to Maysville), HIGHLAND (Eastern one-fifth), HOCKING, JACKSON (Northern half), KNOX, LICKING, LOGAN (Eastern one-third), MADISON, MARION, MORROW, MUSKINSUM (West of a line starting at Adams Mill going to Adamsville & going from Adamsville through Blue Rock to the southern border), PERRY, PICKAWAY, PIKE (Northern half), ROSS, UNION, VINTON and WYANDOT (South of Rte. #30) COUNTIES

	Rates	Fringes
IRONWORKER	\$ 33.27	21.20
IRON0207-004 06/01/2022		••••••

ASHTABULA (Southern part starting at the Geauga County line), COLUMBIANA (E. of a line from Damascus to Highlandtown), MAHONING (N. of Old Route #224), PORTAGE (E. of a line from Middlefield to Shalersville to Deerfield) & TRUMBULL

Rates	Fringes
IRONWORKER	
Layout; Sheeter\$ 32.92	26.26
Ornamental; Reinforcing;	

Structural\$	31.92	26.26
Ornamental; Reinforcing\$	28.92	25.61

IRON0290-002 06/01/2022

ALLEN (Southern half), AUGLAIZE, BUTLER (North of a line drawn from east to the west county line going through Oxford, Darrtown & Woodsdale), CHAMPAIGN (Excluding east of a line drawn from Catawla to the point where #68 intersects the northern county line), CLARK (Western two-thirds), CLINTON (Excluding south of a line drawn from Blanchester to Lynchburg), DARKE, GREENE, HIGHLAND (Inside lines drawn from Marshall to Lynchburg & from the northern county line through East Monroe to Marshall), LOGAN (West of a line drawn from West Liberty to where the northern county line meets the western county line of Hardin), MERCER (Southern half), MIAMI, MONTGOMERY, PREBLE, SHELBY & WARREN (Excluding south of a line drawn from Blanchester through Morrow to the western county line) COUNTIES

	Rates	Fringes
IRONWORKER	\$ 31.59	23.85

IRON0549-003 12/01/2022

BELMONT, GUERNSEY, HARRISON, JEFFERSON, MONROE & MUSKINGUM (Excluding portion west of a line starting at Adams Mill going to Adamsville and going from Adamsville through Blue Rock to the south border)

	Rates	Fringes
IRONWORKER		25.66
IRON0550-004 05/01/2022		

ASHLAND, CARROLL, COLUMBIANA (W. of a line from Damascus to Highlandtown), COSHOCTON (E. of a line beginning at NW Co. line going through Walhonding & Tunnel Hill to the South Co. line), HOLMES, HURON (S. of Old Rte. #224), MAHONING (S. of Old Rte. #224), MEDINA (S. of Old Rte. #224), PORTAGE (S. of Old Rte. #224), RICHLAND, STARK, SUMMIT (S. of Old Rte. #224, Excluding city limits of Barberton), TUSCARAWAS, & WAYNE

	Rates	Fringes
Ironworkers:Structural, Ornamental and Reinforcing	\$ 30.97	21.69

IRON0769-004 06/01/2022

ADAMS (Eastern Half), GALLIA, JACKSON (Southern Half), LAWRENCE & SCIOTO

	Rates	Fringes
IRONWORKER	\$ 33.71	27.69
IRON0787-003 06/01/2022		
ATHENS, MEIGS, MORGAN, NOBLE	, and WASHINGTON	COUNTIES
	Rates	Fringes
IRONWORKER	\$ 31.50	23.75
LAB00265-008 05/01/2022		
	Rates	Fringes
LABORER ASHTABULA, ERIE, HURON, LORAIN, LUCAS, MAHONING MEDINA, OTTAWA, PORTA SANDUSKY, STARK, SUMMIT TRUMBULL & WOOD COUNTIE GROUP 1 GROUP 2 GROUP 3 GROUP 4 CUYAHOGA AND GEAUGA COUNTIES ONLY: SEWAGE	GE, S \$ 34.95 \$ 35.12 \$ 35.45	12.10 12.10 12.10 12.10 12.10

PLANTS, WASTE PLANTS, WATER TREATMENT FACILITIES, PUMPING STATIONS, & ETHANOL PLANTS	
CONSTRUCTION\$ 37.56	12.10
CUYAHOGA, GEAUGA & LAKE	
COUNTIES	
GROUP 1\$ 36.18	12.10
GROUP 2\$ 36.35	12.10
GROUP 3\$ 36.68	12.10
GROUP 4\$ 37.13	12.10
REMAINING COUNTIES OF OHIO	
GROUP 1\$ 34.52	12.10
GROUP 2\$ 34.69	12.10
GROUP 3\$ 35.02	12.10
GROUP 4\$ 35.47	12.10

LABORER CLASSIFICATIONS

GROUP 1 - Asphalt Laborer; Carpenter Tender; Concrete Curing Applicator; Dump Man (Batch Truck); Guardrail and Fence Installer; Joint Setter; Laborer (Construction); Landscape Laborer; Mesh Handlers & Placer; Right-of-way Laborer; Riprap Laborer & Grouter; Scaffold Erector; Seal Coating; Surface Treatment or Road Mix Laborer; Sign Installer; Slurry Seal; Utility Man; Bridge Man; Handyman; Waterproofing Laborer; Flagperson; Hazardous Waste (level D); Diver Tender; Zone Person & Traffic Control

GROUP 2 - Asphalt Raker; Concrete Puddler; Kettle Man Pipeline); Machine Driven Tools (Gas, Electric, Air); Mason Tender; Brick Paver; Mortar Mixer; Power Buggy or Power Wheelbarrow; Paint Striper; Sheeting & Shoring Man; Surface Grinder Man; Plastic Fusing Machine Operator; Pug Mill Operator; & Vacuum Devices (wet or dry); Rodding Machine Operator; Diver; Screwman or Paver; Screed Person; Water Blast, Hand Held Wand; Pumps 4"" & Under (Gas, Air or Electric) & Hazardous Waste (level C); Air Track and Wagon Drill; Bottom Person; Cofferdam (below 25 ft. deep); Concrete Saw Person; Cutting with Burning Torch; Form Setter; Hand Spiker (Railroad); Pipelayer; Tunnel Laborer (without air) & Caisson; Underground Person (working in Sewer and Waterline, Cleaning, Repairing & Reconditioning); Sandblaster Nozzle Person; & Hazardous Waste (level B)

GROUP 3 - Blaster; Mucker; Powder Person; Top Lander; Wrencher (Mechanical Joints & Utility Pipeline); Yarner; Hazardous Waste (level A); Concrete Specialist; Concrete Crew in Tunnels (With Air-pressurized - \$1.00 premium); Curb Setter & Cutter; Grade Checker; Utility Pipeline Tapper; Waterline; and Caulker

GROUP 4 - Miner (With Air-pressurized - \$1.00 premium); & Gunite Nozzle Person

TUNNEL LABORER WITH AIR-PRESSURIZED ADD \$1.00 TO BASE RATE

SIGNAL PERSON WILL RECEIVE THE RATE EQUAL TO THE RATE PAID THE LABORER CLASSIFICATION FOR WHICH HE OR SHE IS SIGNALING.

PAIN0006-002 05/01/2018

ASHTABULA, CUYAHOGA, GEAUGA, LAKE, LORAIN, PORTAGE (N. of the East-West Turnpike) & SUMMIT (N. of the East-West Turnpike)

Datas

F-1----

	Kates	Fringes
PAINTER		
COMMERCIAL NEW WORK;		
REMODELING; & RENOVATIONS		
GROUP 1\$	27.90	16.16
GROUP 2\$	28.30	16.16
GROUP 3\$	28.60	16.16
GROUP 4\$	34.16	16.16
COMMERCIAL REPAINT		
GROUP 1\$	26.40	16.16
GROUP 2\$		16.16
GROUP 3\$	27.10	16.16

PAINTER CLASSIFICATIONS ~ COMMERCIAL NEW WORK; REMODELING; & RENOVATIONS

GROUP 1 - Brush; & Roller

GROUP 2 - Sandblasting & Buffing

GROUP 3 - Spray Painting; Closed Steel Above 55 feet; Bridges & Open Structural Steel; Tanks - Water Towers; Bridge Painters; Bridge Riggers; Containment Builders

GROUP 4 - Bridge Blaster

PAINTER CLASSIFICATIONS ~ COMMERCIAL REPAINT

GROUP 1 - Brush; & Roller

GROUP 2 - Sandblasting & Buffing

GROUP 3 - Spray Painting

PAIN0007-002 07/01/2021

FULTON, HENRY, LUCAS, OTTAWA (Excluding Allen, Bay, Bono, Catawba Island, Clay Center, Curtice, Danbury, Eagle Beach, Elliston, Elmore, Erie, Fishback, Gem Beach & Genova) & WOOD

	F	Rates	Fringes
PAINTER			
NEW CON	MERCIAL WORK		
GROUP	1\$	28.74	18.77
	2\$		18.77
GROUP	3\$	28.74	18.77
GROUP	4\$	28.74	18.77
GROUP	5\$	28.74	18.77
GROUP	6\$	28.74	18.77
GROUP	7\$	28.74	18.77
GROUP	8\$	28.74	18.77
GROUP	9\$	28.74	18,77

REPAINT IS 90% OF JR

PAINTER CLASSIFICATIONS

GROUP 1 - Brush; Spray & Sandblasting Pot Tender

GROUP 2 - Refineries & Refinery Tanks; Surfaces 30 ft. or over where material is applied to or labor performed on above ground level (exterior), floor level (interior)

GROUP 3 - Swing Stage & Chair

GROUP 4 - Lead Abatement

GROUP 5 - All Methods of Spray

GROUP 6 - Solvent-Based Catalized Epoxy Materials of 2 or More Component Materials, to include Solvent-Based Conversion Varnish (excluding water based)

GROUP 7 - Spray Solvent Based Material; Sand & Abrasive Blasting

GROUP 8 - Towers; Tanks; Bridges; Stacks Over 30 Feet

GROUP 9 - Epoxy Spray (excluding water based)

PAIN0012-008 05/01/2019

BUTLER COUNTY

PA

	I	Rates	Fringes
AINTER			
GROUP	1\$	21.95	10.20
GROUP	2\$	25.30	10.20
GROUP	3\$	25.80	10.20
GROUP	4\$	26.05	10.20
GROUP	5\$	26.30	10.20

PAINTER CLASSIFICATIONS

GROUP 1: Bridge Equipment Tender; Bridge/Containment Builder

GROUP 2: Brush & Roller

GROUP 3: Spray

GROUP 4: Sandblasting; & Waterblasting

GROUP 5: Elevated Tanks; Steeplejack Work; Bridge; & Lead Abatement

PAIN0012-010 05/01/2019

BROWN, CLERMONT, CLINTON, HAMILTON & WARREN

	Rates	Fringes
PAINTER		
HEAVY & HIGHWAY BRIDGES-		
GUARDRAILS-LIGHTPOLES-		
STRIPING		
Bridge Equipment Tender		
and Containment Builder	\$ 21.95	10.20
Bridges when highest		
point of clearance is 60		
feet or more; & Lead		
Abatement Projects	\$ 26.30	10.20
Brush & Roller		10.20
Sandblasting & Hopper		
Tender; Water Blasting	26.05	10.20
Spray		10.20

* PAIN0093-001 12/01/2022

ATHENS, GUERNSEY, HOCKING, MONROE, MORGAN, NOBLE and WASHINGTON COUNTIES

	Rates	Fringes
PAINTER		
Bridges; Locks; Dams;		
Tension Towers; &		
Energized Substations	\$ 34.81	22.47
Power Generating Facilitie	s.\$ 31.66	22.47
PAIN0249-002 06/01/2020		

CLARK, DARKE, GREENE, MIAMI, MONTGOMERY & PREBLE

Rate	es Fr	inges
PAINTER		
GROUP 1 - Brush & Roller\$ 24 GROUP 2 - Swing, Scaffold Bridges; Structural Steel; Open Acid Tank; High Tension Electrical	.17	11.22
Equipment; & Hot Pipes\$ 24. GROUP 3 - Spray; Sandblast; Steamclean;	.17	11.22
Lead Abatement\$ 24.	.92	11.22
GROUP 4 - Steeplejack Work\$ 25.	.12	11.22
GROUP 5 - Coal Tar\$ 25. GROUP 6 - Bridge Equipment Tender & or Containment	. 67	11.22
Builder\$ 32. GROUP 7 - Tanks, Stacks &	. 88	11.22
Towers\$ 27. GROUP 8 - Bridge Blaster,	.81	11.22
Rigger\$ 35.		11.22

PAIN0356-002 09/01/2009

KNOX, LICKING, MUSKINGUM, and PERRY

I	Rates	Fringes
PAINTER		
Bridge Equipment Tenders		
and Containment Builders\$ Bridges; Blasters;	27.93	7.25
andRiggers\$	34.60	7.25
Brush and Roller\$ Sandblasting; Steam	20.93	7.25
Cleaning; Waterblasting;		
and Hazardous Work\$	25.82	7.25
Spray\$ Structural Steel and Swing	21.40	7.25
Stage\$	25.42	7.25
Tanks; Stacks; and Towers\$	28.63	7.25

the second se

and the second sec

* PAIN0438-002 12/01/2021

BELMONT, HARRISON and JEFFERSON COUNTIES

	Rates	Fringes
PAINTER		
Bridges, Locks, Dams, Tension Towers & Energized		
Substations Power Generating Facilities	.\$ 34.44	18.19 18.19
	, þ 32.29	10,19
PAIN0476-001 06/01/2021		
COLUMBIANA, MAHONING, and TRUMBU	LL COUNITES	
	Rates	Fringes
PAINTER		
GROUP 1		15.81
GROUP 2 GROUP 3		15.81 15.81
GROUP 4		15.81
GROUP 5	.\$ 27.79	15.81
GROUP 6		15.81
GROUP 7	.\$ 27.79	15.81
PAINTER CLASSIFICATIONS:		
GROUP 1: Painters, Brush & Roll	er	
GROUP 2: Bridges		
GROUP 3: Structural Steel		
GROUP 4: Spray, Except Bar Jois	t/Deck	
GROUP 5: Epoxy/Mastic; Spray- 50 Feet; and Swingstages	Bar Joist/Deck	; Working Above
GROUP 6: Tanks; Sandblasting		
GROUP 7: Towers; Stacks		
PAIN0555-002 05/01/2021		
ADAMS, HIGHLAND, JACKSON, PIKE &	SCIOTO	
	Rates	Fringes
	Naces	TT Inges
PAINTER GROUP 1	¢ 31 0F	17.05
GROUP 2		17.05
GROUP 3		17.05
GROUP 4		17.05
PAINTER CLASSIFICATIONS		
GROUP 1 - Containment Builder		
GROUP 2 - Brush; Roller; Power T	ools, Under 40	feet
GROUP 3 - Sand Blasting; Spray Washing; Epoxy & Two Component Hazardous Waste; Toxic Materia 25,000 Gallon Capacity or More	Materials; Lead ls; Bulk & Stora	d Abatement; age Tanks of
GROUP 4 - Stacks; Bridges		
PAIN0639-001 05/01/2011		

Rates Fringes

Sign Painter & Erector.....\$ 20.61 3.50+a+b+c

FOOTNOTES: a. 7 Paid Holidays: New Year's Day; Memorial Day; July 4th; Labor Day; Thanksgiving Day; Christmas Day & 1

July 4th; Labor Day; Thanksgiving Day; Christmas Day & 1 Floating Day b. Vacation Pay: After 1 year's service - 5 days' paid vacation; After 2, but less than 10 years' service - 10 days' paid vacation; After 10, but less than 20 years' service - 15 days' paid vacation; After 20 years' service -20 days' paid vacation c. Superal leave up to 2 days maximum paid leave for death (

c. Funeral leave up to 3 days maximum paid leave for death of mother, father, brother, sister, spouse, child,

mother-in-law, father-in-law, grandparent and inlaw provided employee attends funeral

..... PAIN0788-002 06/01/2022

ASHLAND, CRAWFORD, ERIE, HANCOCK, HURON, MARION, MORROW, OTTAWA (Allen, Bay, Bono, Catawba Island, Clay Center, Curtice, Danbury, Eagle Beach, Elliston, Elmore, Erie, Fishback, Gem Beach & Genoa), RICHLAND, SANDUSKY, SENECA & WYANDOT

	Rates	Fringes
PAINTER		
Brush & Roller\$	25.08	16.72
Structural Steel\$	26.68	16.72

WINTER REPAINT: Between December 1 to March 31 - 90%JR

\$.50 PER HOUR SHALL BE ADDED TO THE RATE OF PAY FOR THE CLASSIFICATION OF WORK:

While working swingstage, boatswain chair, needle beam and horizontal cable. While operating sprayguns, sandblasting, cobblasting and high pressure waterblasting (4000psi).

\$1.00 PER HOUR SHALL BE ADDED TO THE RATE OF PAY FOR THE CLASSIFICATION OF WORK:

For the application of catalized epoxy, including latex epoxy that is deemed hazardous, lead abatement, or for work or material where special precautions beyond normal work duties must be taken. For working on stacks, tanks, and towers over 40 feet in height.

..... PAIN0813-005 12/01/2008

GALLIA, LAWRENCE, MEIGS & VINTON

	Rates	Fringes
PAINTER		
Base Rate	\$ 24.83	10.00
Bridges, Locks, Dams &		
Tension Towers	\$ 27.83	10.00
PAIN0841-001 06/01/2018		

MEDINA, PORTAGE (South of and including Ohio Turnpike), and SUMMIT (South of and including Ohio Turnpike) COUNTIES

	Rates	Fringes
Painters:		
GROUP	1\$ 25.75	14.35
GROUP	2\$ 26.40	14.35
GROUP	3\$ 26.50	14.35
GROUP	4\$ 26.60	14.35
GROUP	5\$ 27.00	14.35
	6\$ 39.20	11.75
GROUP	7\$ 27.00	14.35

PAINTER CLASSIFICATIONS:

GROUP 1 - Brush, Roller & Paperhanger

- GROUP 2 Epoxy Application
- GROUP 3 Swing Scaffold, Bosum Chair, & Window Jack
- GROUP 4 Spray Gun Operator of Any & All Coatings

GROUP 5 - Sandblast, Painting of Standpipes, etc. from Scaffolds, Bridge Work and/or Open Structural Steel, Standpipes and/or Water Towers

GROUP 6 - Public & Commerce Transportation, Steel or Galvanized, Bridges, Tunnels & Related Support Items (concrete)

GROUP 7 - Synthetic Exterior, Drywall Finisher and/or Taper, Drywall Finisher and Follow-up Man Using Automatic Tools

PAIN0841-002 06/01/2022

CARROLL, COSHOCTON, HOLMES, STARK, TUSCARAWAS & WAYNE

	Rates	Fringes
PAINTER		
Bridges; Towers, Poles & Stacks; Sandblasting		
Steel: Structural Steel &		
Metalizing\$	23.50	15.45
Brush & Roller\$		15.45
Spray; Tank Interior &		
Exterior\$	23.50	15.45

PAIN1020-002 06/01/2022

ALLEN, AUGLAIZE, CHAMPAIGN, DEFIANCE, HARDIN, LOGAN, MERCER, PAULDING, PUTNAM, SHELBY, VAN WERT, and WILLIAMS COUNTIES

	Rates	Fringes
PAINTER		
Brush & Roller\$	26.20	15.00
Drywall Finishing & Taping\$	24.90	15.00
Lead Abatement\$	27.95	15.00
Spray, Sandblasting		
Pressure Cleaning, &		
Refinery\$	26.95	15.00
Swing Stage, Chair,		
Spiders, & Cherry Pickers\$	25.47	15.00
Wallcoverings\$	23.80	15.00

All surfaces 40 ft. or over where material is applied to or labor performed on, above ground level (exterior), floor level (interior) - \$.50 premium

Applying Coal Tar Products - \$1.00 premium

PAIN1275-002 06/01/2020

DELAWARE, FAIRFIELD, FAYETTE, FRANKLIN, MADISON, PICKAWAY, ROSS & UNION

ſ	Rates	Fringes
PAINTER		
Bridges\$	34.64	14.40
Brush; Roller\$ Sandblasting;	25.16	14.40
Steamcleaning; Waterblasting (3500 PSI or		
Over)& Hazardous Work\$	25.86	14.40
Spray\$	25.66	14.40
Stacks; Tanks; & Towers\$ Structural Steel & Swing	28.67	14.40
Stage\$	25.46	14.40

PLAS0109-001 05/01/2018

MEDINA, PORTAGE, STARK, and SUMMIT COUNTIES

	Rates	Fringes
PLASTERER PLAS0109-003 05/01/2018	\$ 28.86	17.11
CARROLL, HOLMES, TUSCARAWAS, an	d wayne coun	TIES
	Rates	Fringes
PLASTERER PLAS0132-002 06/01/2022	\$ 28.21	17.11
BROWN, BUTLER, CLERMONT, HAMIL	TON, HIGHLAN	D, WARREN COUNTIES
	Rates	Fringes
PLASTERER	\$ 29.25	14.69

PLAS0404-002 05/01/2018

ASHTABULA, CUYAHOGA, GEAUGA, AND LAKE COUNTIES

	Rates	Fringes
PLASTERER		
PLA50404-003 05/01/2018		
LORAIN COUNTY		
	Rates	Fringes
PLASTERER		17.11
PLAS0526-022 05/01/2018		
COLUMBIANA, MAHONING, and TRUMB	ULL COUNTI	ES
	Rates	Fringes
PLASTERER PLAS0526-023 05/01/2018		17.11
BELMONT, HARRISON, and JEFFERSON	COUNTIES	
	Rates	Fringes
PLASTERER	.\$ 28.21	17.11
PLAS0886-001 05/01/2018	•••••	
FULTON, HANCOCK, HENRY, LUCAS, PI	UTNAM. and	WOOD COUNTIES
		Fringes
		_
PLASTERER PLAS0886-003 05/01/2018		17.11
DEFIANCE, ERIE, HURON, OTTAWA, P/ COUNTIES	AULDING, S	ANDUSKY, and SENECA
	Rates	Fringes
PLASTERER	\$ 28.86	17.11
PLAS0886-004 05/01/2018		
ALLEN, AUGLAIZE, HARDIN, LOGAN, M	MERCER, an	d VAN WERT COUNTIES
	Rates	Fringes
PLASTERER		17.11
PLUM0042-002 07/01/2022		
ASHLAND, CRAWFORD, ERIE, HURON, M & WYANDOT	KNOX, LORA	IN, MORROW, RICHLAND
	Rates	Fringes
Plumber, Pipefitter, Steamfitter		25.47
PLUM0050-002 07/04/2022		
DEFIANCE, FULTON, HANCOCK, HENRY, PUTNAM, SANDUSKY, SENECA, WILLIAM		TTAWA, PAULDING,
	Rates	Fringes
Plumber, Pipefitter, Steamfitter		28.51
PLUM0055-003 05/01/2022		
ASHTABULA, CUYAHOGA, GEAUGA, LAKE Smith Road) & SUMMIT (N. of Rte.		

Smith Road) & SUMMIT (N. of Rte. #303, including the corporate limits of the city of Hudson)

Rates Fringes

06/06/2023

.

PLUMBER	\$ 40.00	28.43
BELMONT & MONROE (North of Rte. 4	¥78)	
	Rates	Fringes
Plumber and Steamfitter		31.51
CARROLL (Northen Half), STARK, an	d WAYNE COUNTI	ES
	Rates	Fringes
PLUMBER/PIPEFITTER		22.99
ASHTABULA, CUYAHOGA, GEAUGA, LAKI House in Avon Lake), MEDINA (N. d #303)	E, LORAIN (the C of Rte. #18) & S	C.E.I. Power SUMMIT (N. of
	Rates	Fringes
PIPEFITTER		28.34
PLUM0162-002 06/01/2022		
CHAMPAIGN, CLARK, CLINTON, DARKE, MONTGOMERY & PREBLE	, FAYETTE, GREEN	NE, MIAMI,
	Rates	Fringes
Plumber, Pipefitter, Steamfitter		26.80
PLUM0168-002 06/01/2022		
MEIGS, MONROE (South of Rte. #78) & WASHINGTON), MORGAN (South	n of Rte. #78)
	Rates	Fringes
PLUMBER/PIPEFITTER		
* PLUM0189-002 06/01/2022		
DELAWARE, FAIRFIELD, FRANKLIN, HU MARION, PERRY, PICKAWAY, ROSS &	OCKING, LICKING, UNION	, MADISON,
	Rates	Fringes
Plumber, Pipefitter, Steamfitter		26.94
PLUM0219-002 06/01/2022		
MEDINA (Rte. #18 from eastern edge of Medina Co., west to eastern corporate limits of the city of Medina, & on the county road from the west corporate limits of Medina running due west to and through community of Risley to the western edge of Medina County - All territory south of this line), PORTAGE, and SUMMIT (S. of Rte. #303) COUNTIES		
	Rates	Fringes
Plumber and Steamfitter		26.64
PLUM0392-002 06/01/2022		
BROWN, BUTLER, CLERMONT, HAMILTOP	N & WARREN	
	Rates	Fringes

Rates Fringes
PLUMBER/PIPEFITTER......\$ 36.71 24.89

PLUM0396-001 06/01/2022

COLUMBIANA (Excluding Washington & Yellow Creek Townships & Liverpool Twp. - Secs. 35 & 36 - West of County Road #427),

MAHONING and TRUMBULL COUNTIES

	Rates	Fringes
PLUMBER/PIPEFITTER	\$ 36.00	27.91
PLUM0495-002 06/01/2022 CARROLL (Rose, Monroe, Union, Le Townships), COLUMBIANA (Washingt Liverpool Township, Secs. 35 & 3 COSHOCTON, GUERNSEY, HARRISON, H to State Rte. #78 & from McConne to the Perry County line), MUSKI COUNTIES	on & Yell 6, West o OLMES, JE Sville w	ow Creek Townships & f County Rd. #427), FFERSON, MORGAN (South est on State Rte. #37
	Rates	Fringes
Plumber, Pipefitter, Steamfitter PLUM0577-002 06/01/2022		
ADAMS, ATHENS, GALLIA, HIGHLAND, SCIOTO & VINTON	JACKSON,	LAWRENCE, PIKE,
	Rates	Fringes
Plumber, Pipefitter, Steamfitter	\$ 37.56	25.73
PLUM0776-002 07/01/2022		
ALLEN, AUGLAIZE, HARDIN, LOGAN, COUNTIES	MERCER, SH	ELBY and VAN WERT
	Rates	Fringes
Plumber, Pipefitter, Steamfitter		
TEAM0377-003 05/01/2021		
STATEWIDE, EXCEPT CUYAHOGA, GEAU	GA & LAKE	
	Rates	Fringes
TRUCK DRIVER GROUP 1 GROUP 2		
TRUCK DRIVER CLASSIFICATIONS		
GROUP 1 - Asphalt Distributor; 4-Wheel Dump; Oil Distributor (Batch; 4- & Tandem	Wheel Service;
GROUP 2 - Tractor-Trailer Comb: Ready Mix; Semi-Tractor; & Asp Operated From Cab; 5 Axles & O Articulated Dump; Heavy Duty Ed Mechanic	halt Oil S ver; Belly	praybar Man When Dump; End Dump;
TEAM0436-002 05/01/2021		
CUYAHOGA, GEAUGA & LAKE		
	Rates	Fringes
TRUCK DRIVER GROUP 1 GROUP 2		16.95 16.95
GROUP 1: Straight & Dump, Straig	ght Fuel	

GROUP 2: Semi Fuel, Semi Tractor, Euclids, Darts, Tank, Asphalt Spreaders, Low Boys, Carry-All, Tourna-Rockers, Hi-Lifts, Extra Long Trailers, Semi-Pole Trailers, Double Hook-Up Tractor Trailers including Team Track & Railroad Siding, Semi-Tractor & Tri-Axle Trailer, Tandem Tractor & Tandem Trailer, Tag Along Trailer, Expandable Trailer or Towing Requiring Road Permits, Ready-Mix (Agitator or Non-Agitator), Bulk Concrete Driver, Dry Batch Truck,

Articulated End Dump

.....

WELDERS - Receive rate prescribed for craft performing operation to which welding is incidental.

** Workers in this classification may be entitled to a higher minimum wage under Executive Order 14026 (\$16.20) or 13658 (\$12.15). Please see the Note at the top of the wage determination for more information.

Note: Executive Order (EO) 13706, Establishing Paid Sick Leave for Federal Contractors applies to all contracts subject to the Davis-Bacon Act for which the contract is awarded (and any solicitation was issued) on or after January 1, 2017. If this contract is covered by the EO, the contractor must provide employees with 1 hour of paid sick leave for every 30 hours they work, up to 56 hours of paid sick leave each year. Employees must be permitted to use paid sick leave for their own illness, injury or other health-related needs, including preventive care; to assist a family member (or person who is like family to the employee) who is ill, injured, or has other health-related needs, including preventive care; or for reasons resulting from, or to assist a family member (or person who is like family to the employee) who is a victim of, domestic violence, sexual assault, or stalking. Additional information on contractor requirements and worker protections under the EO is available at

https://www.dol.gov/agencies/whd/government-contracts.

Unlisted classifications needed for work not included within the scope of the classifications listed may be added after award only as provided in the labor standards contract clauses (29CFR 5.5 (a) (1) (ii)).

.....

The body of each wage determination lists the classification and wage rates that have been found to be prevailing for the cited type(s) of construction in the area covered by the wage determination. The classifications are listed in alphabetical order of ""identifiers"" that indicate whether the particular rate is a union rate (current union negotiated rate for local), a survey rate (weighted average rate) or a union average rate (weighted union average rate).

Union Rate Identifiers

A four letter classification abbreviation identifier enclosed in dotted lines beginning with characters other than "SU"" or ""UAVG"" denotes that the union classification and rate were prevailing for that classification in the survey. Example: PLUM0198-005 07/01/2014. PLUM is an abbreviation identifier of the union which prevailed in the survey for this classification, which in this example would be Plumbers. 0198 indicates the local union number or district council number where applicable, i.e., Plumbers Local 0198. The next number, 005 in the example, is an internal number used in processing the wage determination. 07/01/2014 is the effective date of the most current negotiated rate, which in this example is July 1, 2014.

Union prevailing wage rates are updated to reflect all rate changes in the collective bargaining agreement (CBA) governing this classification and rate.

Survey Rate Identifiers

Classifications listed under the ""SU"" identifier indicate that no one rate prevailed for this classification in the survey and the published rate is derived by computing a weighted average rate based on all the rates reported in the survey for that classification. As this weighted average rate includes all rates reported in the survey, it may include both union and non-union rates. Example: SULA2012-007 5/13/2014. SU indicates the rates are survey rates based on a weighted average calculation of rates and are not majority rates. LA indicates the State of Louisiana. 2012 is the year of survey on which these classifications and rates are based. The next number, 007 in the example, is an internal number used in producing the wage determination. 5/13/2014 indicates under that identifier. Survey wage rates are not updated and remain in effect until a new survey is conducted.

Union Average Rate Identifiers

Classification(s) listed under the UAVG identifier indicate that no single majority rate prevailed for those classifications; however, 100% of the data reported for the classifications was union data. EXAMPLE: UAVG-OH-0010 08/29/2014. UAVG indicates that the rate is a weighted union average rate. OH indicates the state. The next number, 0010 in the example, is an internal number used in producing the wage determination. 08/29/2014 indicates the survey completion date for the classifications and rates under that identifier.

A UAVG rate will be updated once a year, usually in January of each year, to reflect a weighted average of the current negotiated/CBA rate of the union locals from which the rate is based.

WAGE DETERMINATION APPEALS PROCESS

1.) Has there been an initial decision in the matter? This can be:

* an existing published wage determination

- * a survey underlying a wage determination
- * a Wage and Hour Division letter setting forth a position on a wage determination matter
- * a conformance (additional classification and rate) ruling

On survey related matters, initial contact, including requests for summaries of surveys, should be with the Wage and Hour National Office because National Office has responsibility for the Davis-Bacon survey program. If the response from this initial contact is not satisfactory, then the process described in 2.) and 3.) should be followed.

With regard to any other matter not yet ripe for the formal process described here, initial contact should be with the Branch of Construction Wage Determinations. Write to:

Branch of Construction Wage Determinations Wage and Hour Division U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

2.) If the answer to the question in 1.) is yes, then an interested party (those affected by the action) can request review and reconsideration from the Wage and Hour Administrator (See 29 CFR Part 1.8 and 29 CFR Part 7). Write to:

Wage and Hour Administrator U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

The request should be accompanied by a full statement of the interested party's position and by any information (wage payment data, project description, area practice material, etc.) that the requestor considers relevant to the issue.

3.) If the decision of the Administrator is not favorable, an interested party may appeal directly to the Administrative Review Board (formerly the Wage Appeals Board). Write to:

Administrative Review Board U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

4.) All decisions by the Administrative Review Board are final.

END OF GENERAL DECISIO"

FAIRFIELD COUNTY UTILITIES 10/6/2022

SECTION 011000 - SUMMARY

PART 1 - GENERAL

1.1 SUMMARY

- A. Section Includes:
 - 1. Contract description.
 - 2. Work by Owner or other Work at the Site.

1.2 CONTRACT DESCRIPTION

- A. This project will install 613 linear feet of 8-inch gravity sewer main, 2 sanitary manholes, associated sanitary laterals, water services, and 815 linear feet of chain link fence for facilities at the Fairfield County Airport, Fairfield County, Ohio. The project will also install a duplex lift station and connection to an existing 2-inch sanitary force main.
- B. Use Drawings as necessary to define extent of Contract; include affected temporary facilities, site work, and utilities.

1.3 WORK BY OWNER OR OTHERS

A. Coordinate Work with utilities of Owner and public or private agencies.

PART 2 - PRODUCTS - Not Used

PART 3 - EXECUTION - Not Used

END OF SECTION 011000

(THIS PAGE INTENTIONALLY LEFT BLANK)

the second s

011000 - 2

SUMMARY

SECTION 012000 - PRICE AND PAYMENT PROCEDURES

PART 1 - GENERAL

1.1 SECTION INCLUDES

- A. Application for Payment.
- B. Change procedures.
- C. Defect assessment.

1.2 APPLICATION FOR PAYMENT

- A. Submit one copy of Application for Payment on EJCDC Document C-620 Contractor's Application for Payment and submit updated construction schedule with each Application for Payment.
- B. Payment Period: Submit at monthly intervals.
- C. Submit submittals with transmittal letter as specified in Section 013300 Submittal Procedures
- D. Substantiating Data: When Engineer requires substantiating information, submit data justifying dollar amounts in question. Include the following with Application for Payment:
 - 1. Current construction photographs specified in Section 013300 Submittal Procedures.
 - 2. Partial release of liens from major Subcontractors and vendors.
 - 3. Affidavits attesting to off-Site stored products.
 - Construction Progress Schedule, revised and current as specified in Section 013300 -Submittal Procedures.

1.3 CHANGE PROCEDURES

- A. Submittals: Submit name of individual who is authorized to receive change documents and is responsible for informing others in Contractor's employ or Subcontractors of changes to the Work.
- B. Carefully study and compare Contract Documents before proceeding with fabrication and installation of Work. Promptly advise Engineer of any error, inconsistency, omission, or apparent discrepancy.
- C. Requests for Interpretation (RFI) and Clarifications: Allot time in construction scheduling for liaison with Engineer; establish procedures for handling queries and clarifications.

- D. Engineer will advise of minor changes in the Work not involving adjustment to Contract Sum/Price or Contract Time by issuing supplemental instructions on EJCDC C-942.
- E. Engineer may issue Notice of Change including a detailed description of proposed change with supplementary or revised Drawings and Specifications, a change in Contract Time for executing the change.
- F. Contractor may propose changes by submitting a request for change to Engineer, describing proposed change and its full effect on the Work. Include a statement describing reason for the change and the effect on Contract Sum/Price and Contract Time with full documentation and a statement describing effect on the Work by separate or other Contractors.

1.4 DEFECT ASSESSMENT

- A. Replace the Work, or portions of the Work, not conforming to specified requirements.
- B. If, in the opinion of Engineer it is not practical to remove and replace the Work, Engineer will direct appropriate remedy or adjust payment.
- C. Individual Specification Sections may modify these options or may identify specific formula or percentage sum/price reduction.
- D. Authority of Engineer to assess defects and identify payment adjustments is final.
- E. Nonpayment for Rejected Products: Payment will not be made for rejected products for any of the following reasons:
 - 1. Products wasted or disposed of in a manner that is not acceptable.
 - 2. Products determined as unacceptable before or after placement.
 - 3. Products not completely unloaded from transporting vehicle.
 - 4. Products placed beyond lines and levels of the required Work.
 - 5. Products remaining on hand after completion of the Work.
 - 6. Loading, hauling, and disposing of rejected products.

PART 2 – PRODUCTS - Not Used

PART 3 – EXECUTION - Not Used

END OF SECTION 012000

PRICE AND PAYMENT PROCEDURES

SECTION 012500 - CONTARCT PAY ITEMS

PART 1 - GENERAL

1.1 SECTION INCLUDES

A. Pay item descriptions.

1.2 PAY ITEM DESCRIPTIONS

- A. **Pay Item 1: Construction Layout Stakes & Surveying.** Pay item includes work that consists of furnishing, placing, and maintaining construction layout stakes necessary for the proper prosecution of the work under the contract and removing all stakes at the completion of the project. Basis of payment shall be lump sum, completed and accepted by Engineer.
- B. **Pay Item 2:** Clearing and Grubbing. Pay item includes all equipment and labor. This work consists of clearing, grubbing, scalping, removing trees and stumps, and removing all vegetation and construction debris from the limits shown on the plans, except such objects that are to remain or are to be removed according to other items of work. Basis of payment shall be lump sum, completed and accepted by Engineer.
- C. **Pay Item 3:** 3' Fence Removed Type CLT. Pay item includes all material and equipment. This work consists of removing existing 3-foot-high chain link fencing, posts, posts set in concrete, tension wire, and any other material pertaining to the fencing. Basis of payment shall be per linear foot of 3-foot-high fencing removed and accepted by Engineer.
- D. Pay Item 4: 8-Inch PVC, SDR-35 Sanitary Main, Open Cut, w/ Class B Bedding, w/ SDR 26 Fittings. Pay item includes all pipe, fittings, existing lateral reconnections, connections to proposed manholes, excavation, material disposal, granular bedding, compaction, testing, and televising. Basis of payment shall be linear foot of 8-inch PVC Conduit SDR-35 pipe and SDR 26 fittings installed and accepted by Engineer.
- E. Pay Item 5: 6-Inch PVC, SDR-35 Sanitary Lateral, Open Cut, w/ Class B Bedding, w/ SDR 26 Fittings. Pay item includes all pipe, fittings, existing lateral reconnections, connections to existing manholes, excavation, material disposal, granular bedding, compaction, testing, and televising. Basis of payment shall be linear foot of 6-inch PVC Conduit SDR-35 pipe and SDR 26 fittings installed and accepted by Engineer.
- F. Pay Item 6: Sanitary Manhole. Pay item includes precast manhole, casting and lid, connecting of proposed sewers and pipe, excavation, material disposal, granular bedding, compaction, and testing. Basis of payment shall be per manhole installed and accepted by Engineer.
- G. Pay Item 7: Sanitary Wye Connections. Pay item includes work that consists of furnishing, and installing sanitary wye connections for sanitary laterals, any excavation, material disposal,

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 4/6/2023

granular bedding, and compaction. Basis of payment shall be per wye fitting installed and accepted by Engineer.

- H. Pay Item 8: 2-Inch PVC, SCH. 40 Sanitary Force Main, Open Cut, w/ Class B Bedding, w/ DI fittings and Concrete Backing. Pay item includes all pipe, fittings, excavation, existing force main connection, pump station connection, material disposal, granular bedding, compaction, and testing. Basis of payment shall be linear foot of 2-Inch PVC, SCH. 40 sanitary force main pipe and fittings installed and accepted by Engineer.
- I. Pay Item 9: Permanent Pavement Replacement. Pay item includes all material and equipment. This work consists of removing existing asphalt concrete, brick, portland cement concrete, or aggregate pavement courses; shaping and compacting the exposed material; and placing new asphalt concrete pavement or aggregate and asphalt concrete pavement courses. Basis of payment shall be square yard completed and accepted by Engineer.
- J. Pay Item 10: Restoration Seeding and Mulching, Seed Mix 2. Pay item includes all material, placing, grading, raking, and compacting of topsoil, seeding, and planting to restore unpaved surfaces to pre-construction conditions. Basis of payment shall be lump sum, completed and accepted by Engineer.
- K. Pay Item 11: Install Fencing, 6' Type CLT. Pay item includes all material and equipment. This work consists of installing new 6-foot-high chain link fencing, posts, posts set in concrete, tension wire, and any other material pertaining to the fencing. Basis of payment shall be per linear foot of 6-foot-high fencing installed and accepted by Engineer.
- L. Pay Item 12: Installation of 3/4-Inch Water Service. Pay item includes all tapping saddles, corporation stops, curb stops, service valve boxes with lids, service valves, necessary fittings, excavation, dewatering, material disposal, backfilling, and compaction. Basis of payment shall be per each 3/4-inch water service installed and accepted by Engineer.
- M. Pay Item 13: 3/4-Inch CTS DR9 Water Service. Pay item includes all pipe, necessary fittings, excavation, dewatering, material disposal, backfilling, and compaction. Basis of payment shall be per linear foot of 3/4-inch water service installed and accepted by Engineer.
- N. Pay Item 14: 2-Inch CTS DR9 Water Service. Pay item includes all pipe, necessary fittings, excavation, dewatering, material disposal, backfilling, and compaction. Basis of payment shall be per linear foot of 2-inch water service installed and accepted by Engineer.
- O. **Pay Item 15: 2-Inch Long Service.** Pay item includes all tapping saddles, corporation stops, curb stops, service valve boxes with lids, service valves, necessary fittings, excavation, dewatering, material disposal, backfilling, and compaction. Basis of payment shall be per each 2-inch water service installed and accepted by Engineer.
- P. Pay Item 16: Mobilization. Pay item includes the preparatory work and operations including, but not limited to, those necessary for the movement of personnel, equipment, supplies, and incidentals to the project site; for the establishment of all facilities necessary for work on the project; for all other work and operations that must be performed, or costs incurred before beginning the Work on the other contract items; and for demobilization. Basis of payment shall be lump sum, completed and accepted by Engineer.

012500 - 2

- Q. Pay Item 17: Water Meter Pits. Pay item includes water meter, meter box, valves, necessary fittings, excavation, dewatering, material disposal, backfilling, and compaction. Basis of payment shall be per each water meter box installed and accepted by Engineer.
- R. Pay Item 18: Temporary Access Drive & Erosion Control. Pay item includes all material and equipment for the construction of a temporary access drive and installation of temporary erosion control devices. Basis of payment shall be lump sum, completed and accepted by Engineer.
- S. **Pay Item 19:** Abandon Existing Lift Station. Pay item includes all material and equipment. This work consists of removal of existing equipment, demolition of sides and backfilling with gravel. Basis of payment shall be lump sum, completed and accepted by Engineer.
- T. Pay Item 20: New Lift Station. Pay item includes all material and equipment. This work consists of excavation, material disposal, granular bedding, compaction, testing, furnishing of pumps, rails, chains, piping, valves, fiberglass basin, cover, control panel, devices, and necessary equipment for the installation of the proposed lift station. Basis of payment shall be lump sum, completed and accepted by Engineer.
- U. **Pay Item 21:** Lift Station Electrical. Pay item includes all material, equipment, and labor. This work consists of electrical work for the operation of the lift station. Any conduit, wiring, and associated materials required for the lift station electrical shall be included in this pay item. Basis of payment shall be lump sum, completed and accepted by Engineer.
- V. **Pay Item 22:** As-Builts. Pay item includes all material, equipment, and labor. This work consists of completing final documents for the "As-Built" condition of the project. Contractor redline mark-ups shall submitted with survey coordinates points for all water meter pits, curb boxes, manholes, inverts of manholes, lift station, valves and sanitary lateral stubs. Basis of payment shall be lump sum, completed and accepted by Engineer.

END OF SECTION 012500

THIS PAGE INTENTIONALLY LEFT BLANK

756

.

SECTION 013000 - ADMINISTRATIVE REQUIREMENTS

PART 1 - GENERAL

1.1 SECTION INCLUDES

- A. Coordination and Project conditions.
- B. Preconstruction meeting.
- C. Progress meetings.
- D. Closeout meeting.

1.2 COORDINATION AND PROJECT CONDITIONS

- A. Coordinate scheduling, submittals, and Work of various Sections of Project Manual to ensure efficient and orderly sequence of installation of interdependent construction elements
- B. Coordinate space requirements, supports, and installation of mechanical and electrical Work indicated diagrammatically on Drawings.
 - 1. Coordination Drawings: Prepare as required to coordinate all portions of Work. Show relationship and integration of different construction.
- C. Coordination Meetings: In addition to other meetings specified in this Section, hold coordination meetings with personnel and Subcontractors to ensure coordination of Work.
- D. Coordinate completion and clean-up of Work of separate Sections in preparation for Substantial Completion. After Owner's occupancy of premises, coordinate access to Site for correction of defective Work and Work not complying with Contract Documents, to minimize disruption of Owner's activities.

1.3 PRECONSTRUCTION MEETING

- A. Engineer will schedule and preside over meeting after Notice of Award.
- B. Attendance Required: Engineer, Owner, Resident Project Representative, appropriate governmental agency representatives, Construction Manager, major Subcontractors, and Contractor.
- C. Minimum Agenda:
 - 1. Execution of Owner-Contractor Agreement.
 - 2. Submission of executed bonds and insurance certificates.

- FAIRFIELD COUNTY UTILITIES 10/6/2022
 - 3. Distribution of Contract Documents.
 - 4. Submission of list of Subcontractors, list of products, schedule of values, and Progress Schedule.
 - 5. Designation of personnel representing parties in Contract, and Engineer.
 - 6. Communication procedures.
 - 7. Procedures and processing of requests for interpretations, field decisions, field orders, submittals, substitutions, Applications for Payments, proposal request, Change Orders, and Contract closeout procedures.
 - 8. Scheduling.
 - 9. Critical Work sequencing.
 - D. Contractor: Record minutes and distribute copies to participants within two days after meeting, with two copies each to Engineer, Owner, and those affected by decisions made.

1.4 PROGRESS MEETINGS

- A. Schedule and administer meetings throughout progress of the Work at maximum monthly intervals.
- B. Engineer will make arrangements for meetings, prepare agenda with copies for participants, and preside over meetings.
- C. Attendance Required: Job superintendent, major Subcontractors, Contractors and suppliers, and Engineer, Owner, as appropriate to agenda topics for each meeting.
- D. Minimum Agenda:
 - 1. Review minutes of previous meetings.
 - 2. Review of Work progress.
 - 3. Field observations, problems, and decisions.
 - 4. Identification of problems impeding planned progress.
 - 5. Review of submittal schedule and status of submittals.
 - 6. Review of off-Site fabrication and delivery schedules.
 - 7. Maintenance of Progress Schedule.
 - 8. Corrective measures to regain projected schedules.
 - 9. Planned progress during succeeding work period.
 - 10. Coordination of projected progress.
 - 11. Maintenance of quality and work standards.
 - 12. Effect of proposed changes on Progress Schedule and coordination.
 - 13. Other business relating to Work.
- E. Contractor: Record minutes and distribute copies to participants within two days after meeting, to Engineer, Owner, and those affected by decisions made.

1.5 CLOSEOUT MEETING

- A. Schedule Project closeout meeting with sufficient time to prepare for requesting Substantial Completion. Preside over meeting and be responsible for minutes.
- B. Attendance Required: Contractor Construction Manager, Notify Engineer seven days in advance of meeting date.
- C. Minimum Agenda:
 - 1. Start-up of facilities and systems.
 - 2. Operations and maintenance manuals.
 - 3. Testing, adjusting, and balancing.
 - 4. System demonstration and observation.
 - 5. Operation and maintenance instructions for Owner's personnel.
 - 6. Contractor's inspection of Work.
 - 7. Contractor's preparation of an initial "punch list."
 - 8. Procedure to request Engineer inspection to determine date of Substantial Completion.
 - 9. Completion time for correcting deficiencies.
 - 10. Final cleaning.
 - 11. Preparation for final inspection.
 - 12. Closeout Submittals:
 - a. Project record documents.
 - b. Operating and maintenance documents.
 - c. Operating and maintenance materials.
 - d. Affidavits.
 - 13. Final Application for Payment.
 - 14. Contractor's demobilization of Site.
 - 15. Maintenance.
- D. Record minutes and distribute copies to participants within two days after meeting, with two copies each to Engineer, Owner, and those affected by decisions made.

END OF SECTION 013000

FAIRFIELD COUNTY UTILITIES 10/6/2022

(THIS PAGE INTENTIONALLY LEFT BLANK)

ADMINISTRATIVE REQUIREMENTS

SECTION 013216 - CONSTRUCTION PROGRESS SCHEDULE

PART 1 - GENERAL

1.1 SECTION INCLUDES

- A. Submittals.
- B. Quality assurance.
- C. Bar chart schedules.
- D. Review and evaluation.
- E. Updating schedules.
- F. Distribution.

1.2 SUBMITTALS

- A. Schedule Updates:
 - 1. Overall percent complete, projected and actual.
 - 2. Completion progress by listed activity and subactivity, to within five working days prior to submittal.
 - 3. Changes in Work scope and activities modified since submittal.
 - 4. Delays in submittals or resubmittals, deliveries, or Work.
 - 5. Adjusted or modified sequences of Work.
 - 6. Other identifiable changes.
 - 7. Revised projections of progress and completion.
- B. Narrative Progress Report:
 - 1. Submit with each monthly submission of Progress Schedule.
 - 2. Summary of Work completed during the past period between reports.
 - 3. Work planned during the next period.
 - 4. Explanation of differences between summary of Work completed and Work planned in previously submitted report.
 - 5. Current and anticipated delaying factors and estimated impact on other activities and completion milestones.
 - 6. Corrective action taken or proposed.

1.3 QUALITY ASSURANCE

- A. Scheduler: Contractor's personnel specializing in scheduling with two years' minimum experience in scheduling construction work of complexity comparable to the Project, and having use of computer facilities capable of delivering detailed graphic printout within 48 hours of request.
- B. Contractor's Administrative Personnel: 2 years' minimum experience in using and monitoring schedules on comparable Projects.

1.4 BAR CHART SCHEDULES

- A. Format: Bar chart Schedule, to include at least:
 - 1. Identification and listing in chronological order of those activities reasonably required to complete the Work, including:
 - a. Subcontract Work.
 - b. Major equipment design, fabrication, factory testing, and delivery dates including required lead times.
 - c. Move-in and other preliminary activities.
 - d. Equipment and equipment system test and startup activities.
 - e. Project closeout and cleanup.
 - f. Work sequences, constraints, and milestones.
 - 2. Listings identified by Specification Section number.
 - 3. Identification of the following:
 - a. Horizontal time frame by year, month, and week.
 - b. Duration, early start, and completion for each activity and subactivity.
 - c. Critical activities and Project float.
 - d. Subschedules to further define critical portions of Work.

1.5 REVIEW AND EVALUATION

- A. Participate in joint review and evaluation of schedules with Engineer at each submittal.
- B. Evaluate Project status to determine Work behind schedule and Work ahead of schedule.
- C. After review, revise schedules incorporating results of review, and resubmit within 10 days.

1.6 UPDATING SCHEDULES

A. Maintain schedules to record actual start and finish dates of completed activities.

CONSTRUCTION PROGRESS SCHEDULE

- B. Indicate progress of each activity to date of revision, with projected completion date of each activity. Annotate schedules to depict current status of Work.
- C. Identify activities modified since previous submittal, major changes in Work, and other identifiable changes.
- D. Upon approval of a Change Order, include the change in the next schedule submittal.
- E. Indicate changes required to maintain Date of Substantial Completion.
- F. Submit sorts as required to support recommended changes.
- G. Prepare narrative report to define problem areas, anticipated delays, and impact on schedule. Report corrective action taken or proposed and its effect

1.7 DISTRIBUTION

- A. Following joint review, distribute copies of updated schedules to Contractor's Project site file, to Subcontractors, suppliers, Engineer, Owner, and other concerned parties.
- B. Instruct recipients to promptly report, in writing, problems anticipated by projections shown in schedules.

PART 2 - PRODUCTS - Not Used

PART 3 - EXECUTION - Not Used

END OF SECTION 013216

FAIRFIELD COUNTY UTILITIES 10/6/2022

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

(THIS PAGE INTENTIONALLY LEFT BLANK)

013216 - 4

CONSTRUCTION PROGRESS SCHEDULE

SECTION 013300 - SUBMITTAL PROCEDURES

PART 1 - GENERAL

- 1.1 SECTION INCLUDES
 - A. Definitions.
 - B. Submittal procedures.
 - C. Construction progress schedules.
 - D. Proposed product list.
 - E. Product data.
 - F. Electronic CAD files of Project Drawings.
 - G. Samples.
 - H. Test reports.
 - I. Certificates.
 - J. Contractor review.
 - K. Engineer review.

1.2 DEFINITIONS

A. Action Submittals: Written and graphic information and physical samples that require Engineer's responsive action.

1.3 SUBMITTAL PROCEDURES

- A. Identify: Project, Date, Contractor, Subcontractor and supplier, pertinent Drawing and detail number, and Specification Section number appropriate to submittal.
- B. Apply Contractor's stamp, signed or initialed, certifying that review, approval, verification of products required, field dimensions, adjacent construction Work, and coordination of information is according to requirements of the Work and Contract Documents.
- C. For each submittal for review, allow 15 calendar days excluding delivery time to and from Contractor.

- D. Identify variations in Contract Documents and product or system limitations that may be detrimental to successful performance of completed Work.
- E. Allow space on submittals for Contractor and Engineer review stamps.
- F. When revised for resubmission, identify changes made since previous submission.
- G. Distribute copies of reviewed submittals as appropriate. Instruct parties to promptly report inability to comply with requirements.
- H. Submittals not requested will not be recognized nor processed.
- I. Incomplete Submittals: Engineer will not review. Complete submittals for each item are required. Delays resulting from incomplete submittals are not the responsibility of Engineer.

1.4 CONSTRUCTION PROGRESS SCHEDULES

A. Comply with Section 013216 - Construction Progress Schedule

1.5 PROPOSED PRODUCT LIST

- A. Within 15 calendar days after date of Owner-Contractor Agreement, submit list of major products proposed for use, with name of manufacturer, trade name, and model number of each product.
- B. For products specified only by reference standards, indicate manufacturer, trade name, model or catalog designation, and reference standards.

1.6 PRODUCT DATA

- A. Product Data: Action Submittal: Submit to Engineer for review for assessing conformance with information given and design concept expressed in Contract Documents.
- B. Submit number of copies Contractor requires, plus two copies Engineer will retain.
- C. Mark each copy to identify applicable products, models, options, and other data. Supplement manufacturers' standard data to provide information specific to this Project.
- D. Indicate product utility and electrical characteristics, utility connection requirements, and location of utility outlets for service for functional equipment and appliances.
- E. After review, produce copies and distribute according to "Submittal Procedures" Article and for record documents.

1.7 ELECTRONIC CAD FILES OF PROJECT DRAWINGS

- A. Electronic CAD Files of Project Drawings: May only be used to expedite production of Shop Drawings for the Project. Use for other Projects or purposes is not allowed.
- B. Electronic CAD Files of Project Drawings: Distributed only under the following conditions:
 - 1. Use of files is solely at receiver's risk. Engineer does not warrant accuracy of files. Receiving files in electronic form does not relieve receiver of responsibilities for measurements, dimensions, and quantities set forth in Contract Documents. In the event of ambiguity, discrepancy, or conflict between information on electronic media and that in Contract Documents, notify Engineer of discrepancy and use information in hard-copy Drawings and Specifications.
 - 2. CAD files do not necessarily represent the latest Contract Documents, existing conditions, and as-built conditions. Receiver is responsible for determining and complying with these conditions and for incorporating addenda and modifications.
 - 3. User is responsible for removing information not normally provided on Shop Drawings and removing references to Contract Documents. Shop Drawings submitted with information associated with other trades or with references to Contract Documents will not be reviewed and will be immediately returned.
 - 4. Receiver shall not hold Engineer responsible for data or file clean-up required to make files usable, nor for error or malfunction in translation, interpretation, or use of this electronic information.
 - 5. Receiver shall understand that even though Engineer has computer virus scanning software to detect presence of computer viruses, there is no guarantee that computer viruses are not present in files or in electronic media.
 - 6. Receiver shall not hold Engineer responsible for such viruses or their consequences, and shall hold Engineer harmless against costs, losses, or damage caused by presence of computer virus in files or media.
- C. Shop Drawings: Action Submittal: Submit to Engineer for assessing conformance with information given and design concept expressed in Contract Documents.
- D. Indicate special utility and electrical characteristics, utility connection requirements, and location of utility outlets for service for functional equipment and appliances.
- E. When required by individual Specification Sections, provide Shop Drawings signed and sealed by a professional Engineer responsible for designing components shown on Shop Drawings.
 - 1. Include signed and sealed calculations to support design.
 - 2. Submit Shop Drawings and calculations in form suitable for submission to and approval by authorities having jurisdiction.
 - 3. Make revisions and provide additional information when required by authorities having jurisdiction.

1.8 SAMPLES

- A. Samples: Action Submittal: Submit to Engineer for assessing conformance with information given and design concept expressed in Contract Documents.
- B. Samples for Selection as Specified in Product Sections:
 - 1. Submit to Engineer for aesthetic, color, and finish selection.
 - 2. Submit Samples of finishes, textures, and patterns for Engineer selection.
- C. Submit Samples to illustrate functional and aesthetic characteristics of products, with integral parts and attachment devices. Coordinate Sample submittals for interfacing work.
- D. Include identification on each Sample, with full Project information.
- E. Submit number of Samples specified in individual Specification Sections; Engineer will retain one Sample.
- F. Reviewed Samples that may be used in the Work are indicated in individual Specification Sections.
- G. Samples will not be used for testing purposes unless specifically stated in Specification Section.
- H. After review, produce copies and distribute according to "Submittal Procedures" Article and for record documents
- I. Informational Submittal: Design data are typically submitted for Engineer's knowledge as Contract administrator or for Owner.
- J. Informational Submittal: Submit data for Engineer's knowledge as Contract administrator or for Owner.
- K. Submit information for assessing conformance with information given and design concept expressed in Contract Documents.

1.9 TEST REPORTS

- A. Informational Submittal: Submit reports for Engineer's knowledge as Contract administrator or for Owner.
- B. Submit test reports for information for assessing conformance with information given and design concept expressed in Contract Documents.

1.10 CERTIFICATES

A. Informational Submittal: Submit certification by manufacturer, installation/application Subcontractor, or Contractor to Engineer, in quantities specified for Product Data.

- B. Indicate material or product conforms to or exceeds specified requirements. Submit supporting reference data, affidavits, and certifications as appropriate.
- C. Certificates may be recent or previous test results on material or product but must be acceptable to Engineer.
- D. Identify each print on front. Identify name of Project orientation of view, date and time of view, name and address of photographer, and photographer's numbered identification of exposure.
- E. Digital Images: Deliver complete set of digital image electronic files to Owner with Project record documents. Identify electronic media with date photographs were taken. Submit images that have same aspect ratio as sensor, uncropped.
 - 1. Digital Images: Uncompressed TIFF format, produced by digital camera with minimum sensor size of 4.0 megapixels, and image resolution of not less than 1024 by 768
 - 2. Date and Time: Include date and time in filename for each image.

1.11 CONTRACTOR REVIEW

- A. Review for compliance with Contract Documents and approve submittals before transmitting to Engineer
- B. Contractor: Responsible for:
 - 1. Determination and verification of materials including manufacturer's catalog numbers.
 - 2. Determination and verification of field measurements and field construction criteria.
 - 3. Checking and coordinating information in submittal with requirements of Work and of Contract Documents.
 - 4. Determination of accuracy and completeness of dimensions and quantities.
 - 5. Confirmation and coordination of dimensions and field conditions at Site.
 - 6. Construction means, techniques, sequences, and procedures.
 - 7. Safety precautions.
 - 8. Coordination and performance of Work of all trades.
- C. Stamp, sign or initial, and date each submittal to certify compliance with requirements of Contract Documents.
- D. Do not fabricate products or begin Work for which submittals are required until approved submittals have been received from Engineer.

1.12 ENGINEER REVIEW

A. Do not make "mass submittals" to Engineer. "Mass submittals" are defined as six or more submittals or items in one day or 20 or more submittals or items in one week. If "mass submittals" are received, Engineer's review time stated above will be extended as necessary to

perform proper review. Engineer will review "mass submittals" based on priority determined by Engineer after consultation with Owner and Contractor Informational submittals and other similar data are for Engineer's information, do not require Engineer's responsive action, and will not be reviewed or returned with comment.

- B. Submittals made by Contractor that are not required by Contract Documents may be returned without action.
- C. Submittal approval does not authorize changes to Contract requirements unless accompanied by Change Order, or Work Change Directive.

PART 2 - PRODUCTS - Not Used

PART 3 - EXECUTION - Not Used

END OF SECTION 013300

FAIRFIELD COUNTY UTILITIES 10/6/2022

SECTION 014000 - QUALITY REQUIREMENTS

PART 1 - GENERAL

1.1 SECTION INCLUDES

- A. Quality control.
- B. Testing and inspection services.

1.2 QUALITY CONTROL

- A. Monitor quality control over suppliers, manufacturers, products, services, Site conditions, and workmanship, to produce Work of specified quality.
- B. Comply with specified standards as the minimum quality for the Work except where more stringent tolerances, codes, or specified requirements indicate higher standards or more precise workmanship.
- C. Perform Work using persons qualified to produce required and specified quality.
- D. Products, materials, and equipment may be subject to inspection by Engineer at place of manufacture or fabrication. Such inspections shall not relieve Contractor of complying with requirements of Contract Documents.
- E. Supervise performance of Work in such manner and by such means to ensure that Work, whether completed or in progress, will not be subjected to harmful, dangerous, damaging, or otherwise deleterious exposure during construction period.

1.3 TESTING AND INSPECTION SERVICES

- A. Employ and pay for services of an independent testing agency or laboratory acceptable to Owner to perform specified testing.
 - 1. Before starting Work, submit testing laboratory name, address, and telephone number, and names of full-time.
 - Submit copy of report of laboratory facilities' inspection made by Materials Reference Laboratory of National Bureau of Standards during most recent inspection, with memorandum of remedies of deficiencies reported by inspection.
- B. Testing, inspections, and source quality control may occur on or off Project Site. Perform off-Site testing as required by Engineer or Owner.

- C. Reports shall be submitted by independent firm to Engineer, Contractor, and authorities having jurisdiction, indicating observations and results of tests and compliance or noncompliance with Contract Documents.
 - 1. Submit final report indicating correction of Work previously reported as noncompliant.
- D. Cooperate with independent firm; furnish samples of materials, design mix, equipment, tools, storage, safe access, and assistance by incidental labor as requested.
 - 1. Notify Engineer and independent firm **24** hours before expected time for operations requiring services.
 - 2. Make arrangements with independent firm and pay for additional Samples and tests required for Contractor's use.
- E. Employment of testing agency or laboratory shall not relieve Contractor of obligation to perform Work according to requirements of Contract Documents.
- F. Retesting or re-inspection required because of nonconformance with specified or indicated requirements shall be performed by same independent firm on instructions from Engineer. Payment for retesting or re-inspection will be charged to Contractor by deducting testing charges from Contract Sum/Price.
- G. Agency Responsibilities:
 - 1. Test Samples of mixes submitted by Contractor.
 - 2. Provide qualified personnel at Site. Cooperate with Engineer and Contractor in performance of services.
 - 3. Perform indicated sampling and testing of products according to specified standards.
 - 4. Ascertain compliance of materials and mixes with requirements of Contract Documents.
 - 5. Promptly notify Engineer and Contractor of observed irregularities or nonconformance of Work or products.
 - 6. Perform additional tests required by Engineer.
 - 7. Attend preconstruction meetings and progress meetings.
- H. Agency Reports: After each test, promptly submit **two** copies of report to Engineer, Contractor, and authorities having jurisdiction. When requested by Engineer, provide interpretation of test results. Include the following:
 - 1. Date issued.
 - 2. Project title and number.
 - 3. Name of inspector.
 - 4. Date and time of sampling or inspection.
 - 5. Identification of product and Specification Section.
 - 6. Location in Project.
 - 7. Type of inspection or test.
 - 8. Date of test.
 - 9. Results of tests.
 - 10. Conformance with Contract Documents.

QUALITY REQUIREMENTS

- I. Limits on Testing Authority:
 - 1. Agency or laboratory may not release, revoke, alter, or enlarge on requirements of Contract Documents.
 - 2. Agency or laboratory may not approve or accept any portion of the Work.
 - 3. Agency or laboratory may not assume duties of Contractor.
 - 4. Agency or laboratory has no authority to stop the Work.

PART 2 - PRODUCTS - Not Used

PART 3 - EXECUTION - Not Used

END OF SECTION 014000

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

(THIS PAGE INTENTIONALLY LEFT BLANK)

QUALITY REQUIREMENTS

Q.

SECTION 015000 - TEMPORARY FACILITIES AND CONTROLS

PART 1 - GENERAL

1.1 REFERENCE

A. The requirements of Division 01 apply to the work of all other sections.

1.2 GENERAL

- A. Provide and maintain as a minimum the temporary facilities described herein. Locate all facilities where approved by Engineer and remove same at completion of the work or when otherwise directed.
- B. Comply with all laws, regulations, and safe practices.
- C. Provide temporary utilities throughout construction period as required to facilitate progress of work, to protect work, to provide safe and adequate working conditions throughout Project, to provide for public safety, and to meet all construction needs.
- D. Temporary utilities include, but are not limited to, temporary electricity, lighting, telephone, water, toilets, enclosures, signs, and traffic control.
- E. Remove all temporary utilities, repair all damage caused in installation and restore to existing condition.
- F. Codes and Standards: National Electric Code (ANSI C1); National Electric Safety Code; Federal and State requirements; Utility company regulations; Ohio Department of Transportation.

1.3 TEMPORARY UTILITIES

- A. Monitor Temporary Utilities
 - 1. Any party designated to provide a temporary utility shall be responsible for all damage to his work or to that of other Contractors caused by a defect in such utility.
 - a. Enforce compliance with applicable codes and standards.
 - b. Enforce safe practices.
 - c. Prevent abuse of services and utilities.
 - d. Prevent damage to finishes.
 - 2. Do not allow wasteful use of consumables.
- B. The General Contractor is to pay for utility services including electricity and natural gas consumption until final acceptance by the Owner.

PART 2 - PRODUCTS

- 2.1 MATERIALS
 - A. May be new or used, but must be adequate for the intended purpose.

- B. Must not create unsafe or unsanitary conditions nor violate requirements of applicable Codes.
- C. Comply with applicable Federal and State regulations.
- 2.2 FACILITIES
 - A. Temporary Toilets
 - 1. Equipment: Standard products, meeting code requirements.
 - 2. Facilities: Self ventilated portable toilets, either:
 - a. Privies.
 - b. Chemical toilets.
 - c. Recirculating toilets, or
 - d. Combustion toilets.
 - 3. Toilet Tissue: Provide at each toilet, on suitable dispenser, with adequate reserve supply. Monitor daily.
 - B. Traffic Control Devices: Comply with the Ohio Manual of Uniform Traffic Control Devices.

PART 3 - EXECUTION

- 3.1 GENERAL
 - A. Comply with applicable Federal and State regulations.
 - B. Install work in neat and orderly manner.
 - C. Make structurally, mechanically, and electrically sound throughout.
 - D. Maintain to give safe, continuous services and to provide safe working conditions.
 - E. Modify and extend systems as work progress requires.

3.2 FACILITIES

- A. Telephones
 - 1. Provide cell phone numbers for field personnel.
- B. Temporary Toilets
 - 1. Erect securely, anchor to prevent dislocation.
 - 2. Service as often as necessary to prevent accumulation of wastes and creation of unsanitary conditions.
 - 3. Provide the following minimum number of approved enclosed combination toilet and urinal units for construction personnel:
 - a. For less than 20 workers: 1.
 - b. For 20 or more workers: 2 per 40 workers.
 - c. Provide 2 private facilities: one for men and one for women.
 - 4. Location
 - a. Within the project site.
 - b. Secluded from public observation.

TEMPORARY FACILITIES AND CONTROLS

- c. Obtain acceptance of locations by the Engineer.
- 5. Enclosures for Toilet Facilities
 - a. Weatherproof, sightproof, sturdy temporary enclosures.
 - b. Ventilated to meet applicable Federal and State requirements.
 - c. For enclosures accommodating two or more persons, provide privacy screens for each toilet fixture.
- C. Project Sign: Locate sign as directed by the Engineer.
- D. Traffic Control Devices
 - 1. Contractor shall develop and submit for approval, a general traffic control plan for the project.
 - 2. Specifics of the traffic control plan shall conform to the Ohio Manual of Uniform Traffic Control Devices.
 - 3. It may be required to modify the specifics of the traffic control in order to safely protect the public.
- E. Removal: Each installing Contractor shall remove his temporary utility, repair all damage caused in installation and restore to original conditions.
- F. Cost of Installation, Operation and Maintenance: The Contractor will provide and maintain specified temporary utilities until Date of Substantial Completion unless otherwise indicated. Pay all costs of installation, operation and maintenance of temporary utilities.
- G. Cost of Consumables: The Contractor will pay all costs of consumables for temporary utilities unless otherwise indicated.
- 3.3 OTHER TEMPORARY FACILITIES
 - A. General: Contractor shall provide all other facilities necessary for the proper execution of the Project.
 - B. Water: Water used for pressure and bacteria testing of the completed Plant and lines is to be provided by the Owner, but facilitated by the Contractor.
- 3.4 PAYMENT
 - A. The Contractor shall pay for all temporary facilities except as listed below.

END OF SECTION 015000

TEMPORARY FACILITIES AND CONTROLS

SECTION 017419 - CONSTRUCTION WASTE MANAGEMENT AND DISPOSAL

PART 1 - GENERAL

1.1 SUMMARY

- A. Section Includes:
 - 1. Construction waste management plan.
 - 2. Construction waste recycling.

1.2 PLAN REQUIREMENTS

- A. Develop and implement construction waste management plan as approved by Engineer.
- B. Intent:
 - 1. Divert construction, demolition, and land-clearing debris from landfill disposal.
 - 2. Redirect recyclable material back to manufacturing process.
 - 3. Generate cost savings or increase minimal additional cost to Project for waste disposal.

1.3 SUBMITTALS

- A. Section 013300 Submittal Procedures contains requirements for submittals.
- B. Construction Waste Management Plan: Submit construction waste management plan describing methods and procedures for implementation and monitoring compliance including the following:
 - 1. Transportation company hauling construction waste to waste processing facilities.
 - Recycling and adaptive reuse processing facilities and waste type each facility will accept.
 - 3. Construction waste materials anticipated for recycling and adaptive reuse.
 - 4. On-Site sorting and Site storage methods.
- C. Submit documentation prior to Substantial Completion substantiating construction waste management plan was maintained and goals were achieved.
 - 1. Trash: Quantity by weight deposited in landfills. Include associated fees, transportation costs, container rentals, and taxes for total cost of disposal.
 - 2. Salvaged Material: Quantity by weight with destination for each type of material salvaged for resale, recycling, or adaptive reuse. Include associated fees, transportation costs, container rentals, taxes for total cost of disposal, and reimbursements due to salvage resale.

3. Total Cost: Indicate total cost or savings for implementation of construction waste management plan.

1.4 CONSTRUCTION WASTE MANAGEMENT PLAN

- A. Implement construction waste management plan at start of construction.
- B. Review construction waste management plan at preconstruction meeting and progress meetings specified in Section 013000 Administrative Requirements.
- C. Distribute approved construction waste management plan to Subcontractors and others affected by plan requirements.
- D. Oversee plan implementation, instruct construction personnel for plan compliance, and document plan results.
- E. Purchase products to prevent waste by:
 - 1. Ensuring correct quantity of each material is delivered to Site.
 - 2. Choosing products with minimal or no packaging.
 - 3. Requiring suppliers to use returnable pallets or containers.
 - 4. Requiring suppliers to take or buy back rejected or unused items.

1.5 CONSTRUCTION WASTE RECYCLING

- A. Use source separation method or comingling method suitable to sorting and processing method of selected recycling center. Dispose nonrecyclable trash separately into landfill.
- B. Source Separation Method: Recyclable materials separated from trash and sorted into separate bins or containers, identified by waste type, prior to transportation to recycling center.
- C. Comingling Method: Recyclable materials separated from trash and placed in unsorted bins or container for sorting at recycling center.
- D. Materials suggested for recycling include:
 - 1. Packing materials including paper, cardboard, foam plastic, and sheeting,
 - 2. Recyclable plastics.
 - 3. Organic plant debris.
 - 4. Earth materials.
 - 5. Native stone and granular fill.
 - 6. Asphalt and concrete paving.
 - 7. Wood with and without embedded nails and staples.
 - 8. Glass, clear and colored types.
 - 9. Metals.
 - 10. Equipment oil.

CONSTRUCTION WASTE MANAGEMENT AND DISPOSAL

PART 2 - PRODUCTS - Not Used

PART 3 - EXECUTION

3.1 CONSTRUCTION WASTE COLLECTION

- A. Collect construction waste materials in marked bins or containers and arrange for transportation to recycling centers or adaptive salvage and reuse processing facilities.
- B. Maintain recycling and adaptive reuse storage and collection area in orderly arrangement with materials separated to eliminate co-mingling of materials required to be delivered separately to waste processing facility.
- C. Store construction waste materials to prevent environmental pollution, fire hazards, hazards to persons and property, and contamination of stored materials.
- D. Cover construction waste materials subject to disintegration, evaporation, settling, or runoff to prevent polluting air, water, and soil.

3.2 CONSTRUCTION WASTE DISPOSAL

- A. Deliver construction waste to waste processing facilities. Obtain receipt for deliveries.
- B. Dispose of construction waste not capable of being recycled or adaptively reused by delivery to landfill, incinerator, or other legal disposal facility. Obtain receipt for deliveries.

END OF SECTION 017419

CONSTRUCTION WASTE MANAGEMENT AND DISPOSAL

(THIS PAGE INTENTIONALLY LEFT BLANK)

CONSTRUCTION WASTE MANAGEMENT AND DISPOSAL

SECTION 017700 - CLOSEOUT PROCEDURES

PART 1 - GENERAL

1.1 SUMMARY

- A. Section includes administrative and procedural requirements for Contract closeout, including, but not limited to, the following:
 - 1. Substantial Completion procedures.
 - 2. Final completion procedures.
 - 3. Warranties.
 - 4. Final cleaning.
- B. Related Requirements:
 - 1. C-700 Standard General Conditions of the Construction Contract.
 - 2. C-800 Supplementary Conditions of the Construction Contract.
 - 3. Section 013000 Administrative Requirements.
 - 4. Section 013300 Submittal Requirements.

1.2 ACTION SUBMITTALS

- A. Contractor's List of Incomplete Items: Initial submittal at Substantial Completion.
- B. Certified List of Incomplete Items: Final submittal at Final Completion.

1.3 CLOSEOUT SUBMITTALS

- A. Certificates of Release: From authorities having jurisdiction.
- B. Certificate of Insurance: For continuing coverage.

1.4 SUBSTANTIAL COMPLETION PROCEDURES

A. Contractor's List of Incomplete Items: Prepare and submit a list of items to be completed and corrected (Contractor's "punch list"), indicating the value of each item on the list and reasons why the Work is incomplete.

- B. Submittals Prior to Substantial Completion: Complete the following a minimum of 10 days prior to requesting inspection for determining date of Substantial Completion. List items below that are incomplete at time of request.
 - 1. Certificates of Release: Obtain and submit releases from authorities having jurisdiction, permitting Owner unrestricted use of the Work and access to services and utilities. Include occupancy permits, operating certificates, and similar releases.
 - 2. Submit closeout submittals specified in other Division 01 Sections, including Project Record Documents, operation and maintenance manuals, damage or settlement surveys, property surveys, and similar final record information.
 - 3. Submit closeout submittals specified in individual Sections, including specific warranties, workmanship bonds, maintenance service agreements, final certifications, and similar documents.
 - 4. Submit maintenance material submittals specified in individual Sections, including tools, spare parts, extra materials, and similar items, and deliver to location designated by Engineer. Label with manufacturer's name and model number.
 - 5. Submit testing, adjusting, and balancing records.
 - 6. Submit changeover information related to Owner's occupancy, use, operation, and maintenance.
- C. Procedures Prior to Substantial Completion: Complete the following a minimum of 10 days prior to requesting inspection for determining date of Substantial Completion. List items below that are incomplete at time of request.
 - 1. Advise Owner of pending insurance changeover requirements.
 - 2. Make final changeover of permanent locks and deliver keys to Owner. Advise Owner's personnel of changeover in security provisions.
 - 3. Complete startup and testing of systems and equipment.
 - 4. Perform preventive maintenance on equipment used prior to Substantial Completion.
 - 5. Instruct Owner's personnel in operation, adjustment, and maintenance of products, equipment, and systems.
 - 6. Advise Owner of changeover in utility services.
 - 7. Participate with Owner in conducting inspection and walkthrough with local emergency responders.
 - 8. Terminate and remove temporary facilities from Project site, along with mockups, construction tools, and similar elements.
 - 9. Complete final cleaning requirements.
 - 10. Touch up paint and otherwise repair and restore marred exposed finishes to eliminate visual defects.
- D. Inspection: Submit a written request for inspection to determine Substantial Completion a minimum of 10 days prior to date the Work will be completed and ready for final inspection and tests. On receipt of request, Engineer will either proceed with inspection or notify Contractor of unfulfilled requirements. Architect will prepare the Certificate of Substantial Completion after inspection or will notify Contractor of items, either on Contractor's list or additional items identified by Engineer, that must be completed or corrected before certificate will be issued.

1.5 FINAL COMPLETION PROCEDURES

- A. Submittals Prior to Final Completion: Before requesting final inspection for determining Final Completion, complete the following:
 - 1. Submit a final Application for Payment in accordance with Section 012900 Price and Payment Procedures.
 - 2. Certified List of Incomplete Items: Submit certified copy of Engineer's Substantial Completion inspection list of items to be completed or corrected (punch list), endorsed and dated by Engineer. Certified copy of the list shall state that each item has been completed or otherwise resolved for acceptance.
 - 3. Certificate of Insurance: Submit evidence of final, continuing insurance coverage complying with insurance requirements.
 - 4. Submit pest-control final inspection report.
- B. Inspection: Submit a written request for final inspection to determine acceptance a minimum of 10 days prior to date the Work will be completed and ready for final inspection and tests. On receipt of request, Engineer will either proceed with inspection or notify Contractor of unfulfilled requirements. Engineer will prepare a final Certificate for Payment after inspection or will notify Contractor of construction that must be completed or corrected before certificate will be issued.

1.6 LIST OF INCOMPLETE ITEMS

2.

- A. Organization of List: Include name and identification of each space and area affected by construction operations for incomplete items and items needing correction including, if necessary, areas disturbed by Contractor that are outside the limits of construction.
 - 1. Organize list of spaces in sequential order,
 - Submit list of incomplete items in the following format:
 - a. MS Excel electronic file. Engineer will return annotated file.
 - b. Three paper copies. Engineer will return two copies.

1.7 SUBMITTAL OF PROJECT WARRANTIES

- A. Time of Submittal: Submit written warranties on request of Architect for designated portions of the Work where warranties are indicated to commence on dates other than date of Substantial Completion, or when delay in submittal of warranties might limit Owner's rights under warranty.
- B. Organize warranty documents into an orderly sequence based on the table of contents of Project Manual.
- C. Warranty Electronic File: Provide warranties and bonds in PDF format. Assemble complete warranty and bond submittal package into a single electronic PDF file with bookmarks enabling navigation to each item. Provide bookmarked table of contents at beginning of document.

(THIS PAGE INTENTIONALLY LEFT BLANK)

PACKAGED WASTEWATER PUMPING STATIONS

F. Furnish installation certificate from equipment manufacturer's representative attesting equipment has been properly installed and is ready for service.

END OF SECTION 333213

PACKAGED WASTEWATER PUMPING STATIONS

B. Access door units shall be in alignment to permit unobstructed removal of pumps and valves. Complete the pump station piping and fittings in accordance with the Drawings and Specifications relating thereto. Remove all lifting eyes from units, rub all joints, and grade site.

3.2 INSTALLATION

- A. The Contractor shall install equipment as required by the manufacturer's written installation instructions and approved shop drawings unless otherwise directed by the Engineer.
- B. The floor of the pump station shall be installed level. Pump piping shall be completely secured to prevent any excess movement. Provide permanent supports, and wedges as indicated to restrain movement.
- C. Excess motor and control wire shall be carefully coiled and hung inside the wet well. These wires shall not be cut, and all identification tags shall be in place. Cables shall be supported with stainless steel woven strain relief grips hung in the wet well and be routed in a manner that will not interfere with access to any equipment or terminals in the control panels.

3.3 FIELD QUALITY CONTROL

- A. Section 014000 Quality Requirements: Requirements for inspecting and testing.
- B. Inspection:
 - 1. Check pump and motor for high bearing temperature and excessive vibration.
 - 2. Check for motor overload by taking ampere readings.
- C. Preoperational Inspection:
 - 1. Check alignment of pumps and motors.
 - 2. Check pumps for proper rotation.
 - 3. Check pump and drive units for proper lubrication.
- D. Pump Test:
 - 1. Perform a pump test. Submit the test results to Engineer. Test all panels to the power requirements as shown on the plans to assure proper component operation. Activate each control function to check for proper operation and indication.
- E. Equipment Acceptance:
 - 1. Adjust, repair, modify, or replace system components failing to perform as specified and rerun tests.
 - 2. Make final adjustments to equipment under direction of manufacturer's representative.
 - 3. Document adjustments, repairs, and replacements in manufacturer's field services certification.

- H. Vent Assembly:
 - 1. Diameter: 2 in.
 - 2. Material: Schedule 80 PVC.
 - 3. Type: Mushroom vent with bug screen.
- I. Anti-Flotation bottom flange.

2.5 CONTROL PANEL

- A. Description.
 - 1. Duplex alternating control panel.
 - 2. UL 508A listed in the United States.
 - 3. Convenience receptacle (110 V).
 - 4. Type: NEMA 4X enclosure, stainless steel.
 - 5. Factory fabricated, self-contained.

B. Devices:

- 1. Float Switches: Mercury-free float switch (one) including cord and weight.
- 2. Lights:
 - a. Pump motor run lights (1 per pump).
 - b. Pump elapsed run time meters (1 per pump).
 - c. Moisture sensor and motor high temperature alarm lights (1 each per pump), remote alarm contact.
 - d. Pump shutdown alarm lights (1 per pump).
 - e. Indicator lights may be red LED, or similar.
- 3. Alarm: 360-degree red high water alarm light with audible buzzer and test/silence buttons.
- 4. Switches: Hand/Off/Auto heavy-duty toggle switches.
- 5. Pressure transducer with high level float
- 6. Precision Digital Controller: PD6000-6R7

PART 3 - EXECUTION

3.1 PUMP STATION (GENERAL)

A. Compact clean subgrade under wet well to 100% of maximum density at optimum moisture content as determined by AASHTO T-99, Method A. Place 12" minimum of Type B backfill compacted to 12" outside of the wet well base.

- 4. Anti-floatation flange.
- B. Cover:
 - 1. Material: Aluminum cover plate.
 - 2. Access Door:
 - a. Materials: Aluminum with 316 stainless steel hardware.
 - b. Load Rating: H-20.
 - c. Accessories: Locking cover, spring assist, hold open latch, hinged safety grate.
- C. Discharge Pipe & Fittings:
 - 1. Material: Schedule 80 PVC.
 - 2. Size: 2 in.
 - 3. The sealing of the pumping unit to the discharge connection shall be accomplished by a machined metal to metal contact. Sealing of the discharge interface with a diaphragm, O-ring or profile gasket will not be accepted. The entire weight of the pump/motor unit shall be borne by the pump discharge elbow. No portion of the pump/motor unit shall bear on the sump floor directly or on a sump floor mounted stand.
- D. Rail System:
 - 1. Description: Slide rail assembly, consisting of Type 316 stainless-steel upper and lower rail brackets.
 - 2. Guide Rails:
 - a. Quantity: 2 (each pump)
 - b. Diameter: 1 in.
- E. Ball Valves:
 - 1. Description: Stainless Steel Ball Valve, Non-Rising Stem, Threaded, with Stainless Steel Handle
 - 2. Quantity: 3
 - 3. Diameter: 2 in.
 - 4. Material: Stainless Steel
- F. Check Valves:
 - 1. Description: 521 Brass Check Valve, Swing Type, Threaded
 - 2. Quantity: 2
 - 3. Diameter: 2 in.
 - 4. Material: Stainless Steel
- G. Lifting Chains/Cables:
 - 1. Description: Chain of sufficient thickness to lift specified pump(s)
 - 2. Quantity: 1 per pump
 - 3. Material: 316 SST

PACKAGED WASTEWATER PUMPING STATIONS

- 1. Design point: 22 gpm at 36 feet of head.
- 2. Speed: 3,450 rpm.
- 3. Pump Casing: Cast Iron ASTM A48 Class 40.
- C. Motors
 - 1. Type: Submersible.
 - 2. Motor Housing & Cover: Cast Iron ASTM A48 Class 40.
 - 3. Impeller: Cast Iron ASTM A48 Class 40.
 - 4. Motor Pump Shaft: AISI 430 F stainless steel.
 - 5. Lower Bearing: Double row angular ball.
 - 6. Upper Bearing: Deep groove ball.
 - 7. Horsepower: 2.0, continuous duty.
 - 8. Voltage: 208 V, 3 phase.
- D. Operation:
 - 1. Duplex Control Sequence:
 - a. When basin liquid level increases to LEAD PUMP START setting, LEAD PUMP START switch energizes lead pump.
 - b. When basin liquid level decreases to PUMP STOP setting, PUMP STOP switch deenergizes lead pump.
 - c. When lead pump is de-energized, alternating relay indexes such that lag pump starts on next rise in basin liquid level.
 - d. If basin liquid level continues to rise to LAG PUMP START setting, LAG PUMP START switch energizes lag pump.
 - e. When basin liquid level decreases to PUMP STOP setting, PUMP STOP switch deenergizes both pumps.
 - f. If basin liquid level continues to rise, HIGH LEVEL alarm switch energizes alarm signal when liquid level reaches HIGH-LEVEL setting.
 - g. Refer to Drawings for control elevations.
 - h. Alternation: Set pumps to automatically switch lead and lag designations after each pumping cycle.
- E. Sound, Vibration, and Thermal Control:
 - 1. Dampen or suppress noise, and absorb vibration.
 - 2. Accommodate thermal expansion and stresses.
 - 3. Adjust or correct for misalignment in piping systems.
- 2.4 BASIN
 - A. Description:
 - 1. Submersible duplex basin system, including cover with vent and door, rail assemblies, discharge and fittings, union ball valves, junction boxes, and level controls.
 - 2. Material: Fiberglass
 - 3. Dimensions: 72" Diameter x Depth (shown on plans).

PACKAGED WASTEWATER PUMPING STATIONS

- - C. The control panel shall be UL 508A listed in the United States and the manufacturer shall provide a two-year warranty on the control panel.

PART 2 - PRODUCTS

2.1 SYSTEM DESCRIPTION

- Α. **Pumping Station:**
 - 1. **Configuration: Duplex**
 - 2. Assembly: Shop
 - 3. **Basin Material: Fiberglass**
 - Pump Type: Submersible Grinder Pump 4.
 - 5. Controls:
 - Pressure transducer with high level float. a.
 - Operation: Hand/Off/Auto for each pump. Automatically alternating lead pumps. b.
 - Pump run lights. c.
 - Alarms: red beacon, audible alarm, alarm test, and alarm silence d.
 - Precision Digital Controller: PD6000-6R7 e.

2.2 MANUFACTURER

- Α. Manufacturers include:
 - 1. Flygt
 - 2. Barnes
 - **Excel Fluid Group** 3.
 - 4. Or engineer approved equal
- Contractors proposing alternate manufacturers will be responsible for all costs associated with Β. redesign including all process, electrical, structural, mechanical, and civil aspects of the installation.
- Proposed alternates to the specified equipment will be considered if the proposed alternate is C. superior or equal in construction and/or efficiency to that described in the specifications and the high quality has been demonstrated by at least five years in continuous service installations.

2.3 PERFORMANCE AND DESIGN REQUIREMENTS

- General Α.
 - Number of Pumps: 2 1.
 - 2. Fluid to be Pumped: Municipal wastewater.
 - Discharge Size: Per manufacturer standards. Piping will connect to an existing 2" force 3. main.
- Β. Pump Performance Characteristics (each pump)

FAIRFIELD COUNTY UTILITIES AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN 4/6/2023

- 3. The Contractor shall submit to the Owner the following set of spare parts as listed of the same Manufacturer's type installed on the Project. The spare part kit shall consist of the following:
 - a. O-rings, mechanical seals, and bearings.

1.7 QUALITY ASSURANCE

A. Section 014000 – Quality Requirements: Requirements for quality control, testing, and inspections.

1.8 QUALIFICATIONS

- A. Manufacturer: Company specializing in manufacturing packaged pumping station products specified in this Section with minimum of five years of experience.
- B. Section C-451-2018 Qualifications Statement.

1.9 DELIVERY, STORAGE, AND HANDLING

- A. Delivery, storage, and handling of the packaged pumping station equipment shall be in accordance with the manufacturer's recommendations. No extra cost shall be charged to the Owner for the handling, delivery, or storage.
- B. Inspection: Accept materials on Site in manufacturer's original packaging and inspect for damage. Any damaged materials shall not be accepted and shall not be installed.
- C. Protection:
 - 1. Protect materials, equipment, and components from moisture and dust by storing in clean, dry location remote from construction work areas.
 - 2. Provide temporary end caps and closures on piping and fittings and maintain in place until installation.
 - 3. Provide additional protection according to manufacturer instructions.

1.10 EXISTING CONDITIONS

- A. Field Measurements:
 - 1. Verify field measurements prior to fabrication.
 - 2. Indicate field measurements on Shop Drawings where needed.

1.11 WARRANTY

- A. Section 017700 Closeout Procedures: Requirements for warranties.
- B. Contractor shall provide a one-year warranty on the entire packaged pump station system.

- 1. Indicate station layout as designed by station manufacturer.
- 2. Indicate size, materials of construction, and components of system.
- 3. Indicate basin size, inlet and discharge locations, cover dimensions, vent location, lifting cable/chain location, check valve locations, ball valve locations, pump locations, discharge piping location, wiring diagrams, junction box locations, guide rail assembly location, level control locations, and ballast support flange dimensions.
- D. Test and Evaluation Reports: Submit written report showing that factory and field pump inspections, tests, and startup have been successfully performed.
- E. Manufacturer Instructions: Submit manufacturer's installation instructions and instructions for basin, pump, and panel systems procedures.
- F. Source Quality-Control Submittals: Indicate results of shop tests and inspections.
- G. Field Quality-Control Submittals: Indicate results of Contractor-furnished tests and inspections.
- H. Manufacturer Reports:
 - 1. Submit report of each visit of manufacturer's personnel to provide technical assistance during installation.
 - 2. Submit startup report before final acceptance of pumps to document that pumping station operation meets performance requirements.
- I. Qualifications Statements:
 - 1. Submit gualifications for manufacturer, installer, and licensed professional.
 - 2. Submit manufacturer's approval of installer.

1.5 CLOSEOUT SUBMITTALS

- A. Section 017700 Closeout Procedures: Requirements for closeout submittals.
- B. Project Record Documents: Record actual locations of packaged pumping stations, including basins and control panel.
- C. Submit certification of pumping station after performance testing.

1.6 MAINTENANCE MATERIAL SUBMITTALS

- A. Section 017700 Closeout Procedures: Requirements for maintenance materials.
- B. Spare Parts:
 - 1. Spare parts used during startup and prior to acceptance of the equipment shall be replaced by the Contractor at no additional cost to the Owner.
 - 2. Spare parts shall be shipped with the equipment and stored by the Contractor until turned over to the Engineer during project closeout.

PACKAGED WASTEWATER PUMPING STATIONS

- 3. ASTM A153/ (A153M Standard Specification for Zinc Coating (Hot-Dip) on Iron and Steel Hardware).
- 4. ASTM A307 Standard Specification for Carbon Steel Bolts, Studs, and Threaded Rod 60 000 PSI Tensile Strength.
- 5. ASTM A709/ (A709M Standard Specification for Structural Steel for Bridges).
- 6. ASTM D698 Standard Test Methods for Laboratory Compaction Characteristics of Soil Using Standard Effort (12,400 ft-lbf/ft3 (600 kN-m/m3).
- ASTM D1557 Standard Test Methods for Laboratory Compaction Characteristics of Soil Using Modified Effort (56,000 ft-lbf/ft3 (2,700 kN-m/m3).
- ASTM D1785 Standard Specification for Poly(Vinyl Chloride) (PVC) Plastic Pipe, Schedules 40, 80, and 120.
- ASTM D2241 Standard Specification for Poly(Vinyl Chloride) (PVC) Pressure-Rated Pipe (SDR Series).
- 10. ASTM D2466 Standard Specification for Poly(Vinyl Chloride) (PVC) Plastic Pipe Fittings, Schedule 40.
- 11. ASTM D2467 Standard Specification for Poly(Vinyl Chloride) (PVC) Plastic Pipe Fittings, Schedule 80.
- 12. ASTM D6938 Standard Test Method for In-Place Density and Water Content of Soil and Soil-Aggregate by Nuclear Methods (Shallow Depth).
- D. National Electrical Manufacturers Association:
 - 1. NEMA 250 Enclosures for Electrical Equipment (1,000 Volts Maximum).
- E. UL Inc.:
 - 1. UL 83 Thermoplastic-Insulated Wires and Cables.

1.3 PRECONSTRUCTION MEETINGS

- A. Section 013000 Administrative Requirements: Requirements for coordination meeting.
- B. Contractor to schedule a coordination meeting prior to commencing field work in this Section.

1.4 SUBMITTALS

- A. Section 013300 Submittal Procedures: Requirements for submittals.
- B. Product Data:
 - 1. Manufacturer information for basin, cover, hinged door, slide rail assembly, discharge piping, and valves.
 - 2. Manufacturer information for pumps, including performance curve, base, breakaway fittings data, lifting mechanism, and access frame data.
 - 3. Control panel data, level controls, junction box, panel wiring schematic.
- C. Shop Drawings:

PACKAGED WASTEWATER PUMPING STATIONS

SECTION 333213 - PACKAGED WASTEWATER PUMPING STATIONS

PART 1 - GENERAL

1.1 UNIT PRICE - MEASUREMENT AND PAYMENT

- A. Section 012000 Price and Payment Procedures: Contract Sum/Price modification procedures.
- B. Submersible Pump Basin:
 - 1. Basis of Measurement: By each.
 - 2. Basis of Payment: Includes excavation, base, basin with cover, vent and access door, inlet and discharge, rail assemblies, autocoupling, control panel and level controls, electrical terminations, valves, junction box, discharge piping, tie-in, backfilling, and startup testing.
- C. Pump(s):
 - 1. Basis of Measurement: By each.
 - 2. Basis of Payment: Includes pump, pump brackets and check valve, discharge piping, and pipe fittings in basin from pump to discharge.
- D. Control Panel:
 - 1. Basis of measurement: By each.
 - 2. Basis of Payment: Includes enclosure, internal components, and wiring within, as well as to and from power source and basin.

1.2 REFERENCES

- A. American Association of State Highway and Transportation Officials:
 - 1. AASHTO T 180 Standard Method of Test for Moisture-Density Relations of Soils Using a 4.54-kg (10-lb) Rammer and a 457-mm (18-in.) Drop.
- B. American Bearing Manufacturers Association:
 - 1. ABMA 9 Load Ratings and Fatigue Life for Ball Bearings.
 - 2. ABMA 25.2 Rolling Bearings, Linear Motion Recirculating Ball, Sleeve Type Inch Series.
 - 3. ABMA 10285 Rolling Bearings Sleeve type linear ball bearings Boundary dimensions and tolerances (Metric).
- C. ASTM International:
 - 1. ASTM A48/ (A48M Standard Specification for Gray Iron Castings).
 - 2. ASTM A126 Standard Specification for Gray Iron Castings for Valves, Flanges, and Pipe Fittings.

- E. Each feeder and branch circuit associated with a two-pole or three-pole protective device shall be provided with a separate green insulated equipment grounding conductor. The required equipment grounding conductor shall be sized as shown on the drawings and shall not be smaller than shown in N.E.C. Table 250.122 and shall be installed in a common conduit with the related phase and/or neutral conductors. In the case of parallel feeders, each raceway shall have a full size green insulated equipment ground conductor.
- F. Single phase branch circuits required for 120 volt and 277 volt lighting, receptacles, and motors shall consist of phase, neutral and grounding conductors installed in a common metallic conduit.
 - 1. Flexible metallic conduit equipment connections utilized in conjunction with the above single-phase branch circuits shall be provided with suitable green insulated equipment grounding conductors connected to approved grounding terminals at each end of the flexible conduit.
 - 2. Single phase branch circuits installed in non-metallic conduits shall be provided with a separate green insulated grounding conductor as hereinbefore specified.
- G. Furnish and install driven ground rods and grounding conductors as indicated on the Drawings.
- H. All concealed ground grid and ground rod connections shall be welded with "Cadweld" (or equal by "Thermoweld") fittings. Connections to electrical distribution equipment busses shall be made by use compression type lugs.
- I. Furnish and install all required grounding conductors and fittings as delineated in National Electric Code Article 250.

END OF SECTION 260450

SECTION 260450 - GROUNDING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. The other Contract Documents complement the requirements of this Section. The General Requirements apply to the work of this Section.

1.2 SCOPE

A. Provide material, equipment, labor and supervision as required for the complete installation of a grounding system.

1.3 CODES AND STANDARDS

- A. Except where otherwise required, the following codes shall govern:
 - 1. NEC, Article 250
 - 2. NFPA 101
 - 3. UL Listed

PART 2 - PRODUCTS

- 2.1 GENERAL
 - A. Ground grid conductors installed below slab on grade shall be bare stranded copper.
 - B. All other grounding conductors shall be copper with Type TW, THW or THWN green insulation.
 - C. Ground rods shall be "Copperweld" as manufactured by Joslyn and shall be of the sizes indicated on the drawings.

PART 3 - EXECUTION

3.1 INSTALLATION

- A. Metallic conduit system shall be electrically continuous throughout.
- B. All motors and metal frames of all electrical equipment shall be grounded.
- C. System neutral conductor shall be identified throughout the building.
- D. All cord connected electrical equipment frames shall be grounded to the conduit system through a grounding conductor in the cord.

GROUNDING 1

(THIS PAGE INTENTIONALLY LEFT BLANK)

WIRING DEVICES

260140 - 4

06/06/2023

- F. Prior to roughing-in outlet boxes, Contractor shall verify from general construction drawings door swings, type of wall finishes and locations for counters and work benches.
- G. Devices installed on opposite sides of a common wall shall not be installed back-to-back. Outlet boxes on opposite sides of a common wall shall be separated 6" (minimum). Where separation is less than 12", install sound absorption insulation material between boxes and seal any conduits between boxes with duct seal at both ends.
- H. This section Contractor shall provide material as required to completely install work indicated on the Drawings which is located in mill and casework.

END OF SECTION 260140

- A. Receptacles shall be as described on the Drawings.
- B. Receptacles shall contain a 1-piece, die stamped (not riveted), brass mounting strap with highimpact, chemical resistant face.
- C. Receptacles shall be "Industrial Grade", and shall be UL498 Listed.
- D. Ground fault circuit interrupting (GFCI) receptacles shall be a U.L. 943 and U.L. 498 Listed duplex feed-through type with power/failure indicating LED, and test and reset buttons.
- E. Acceptable manufacturers and catalog numbers series are as follows:

1.	Device 20A Duplex Receptacles	Hubbell HBL5362	Pass & Seymour PS5362A
2.	20A GFCI Duplex Receptacles	GF20L	2095

2.4 COVERPLATES

A. Covers for receptacles in wet locations shall be cast aluminum type with self-locking vertical flip covers and UL Listed "suitable for wet locations while in use". Hubbell #WP26M or Thomas Betts #CKMGV.

PART 3 - EXECUTION

3.1 INSTALLATION

- A. Install wiring devices as indicated on the Drawings, and as described below.
- B. Switches and receptacles shall be installed and located as follows, unless noted otherwise on Drawings:
 - 1. Switches: 48" above finished floors.
 - 2. Receptacles: 18" above finished floors generally; 36" above unfinished floors or 8" above counters and work benches in kitchens, shops, mechanical equipment rooms and similar areas.
- C. In masonry walls, switches and receptacle heights shall be adjusted as required so outlets are at nearest mortar joint to specified height.
- D. Where light switches are located adjacent to doors, they shall be installed on "knob" side of door opening, unless indicated otherwise.
- E. Where walls have wainscot or chair moldings, switch height shall be adjusted as required so switch is either all in wainscot or all in wall above wainscot.

SECTION 260140 – WIRING DEVICES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. The other Contract Documents complement the requirements of this Section. The General Requirements apply to the work of this Section.

1.2 SCOPE

A. Provide materials, equipment, labor and supervision necessary to install wiring devices as required by the Drawings and this section.

1.3 STANDARDS AND CODES

- A. Except where otherwise required by this section, the following Standards and Codes shall govern:
 - 1. Receptacles; NEC Article 406
 - 2. Wall Switches; NEC Article 404
 - 3. UL Listed
 - 4. NEMA Standards

1.4 QUALIFICATIONS

A. Products by Hubbell HBL series or Pass & Seymour - Industrial Spec Grade Series.

1.5 SUBMITTALS

A. Prior to the purchase of wiring devices and plates, submit shop drawings to the Architect for approval.

PART 2 - PRODUCTS

- 2.1 GENERAL
 - A. All wiring devices shall minimally be "Industrial Specification Grade" except where higher grade is required by the Drawings.
 - B. Wiring devices color shall be as follows:
 - 1. Devices connected to normal power: gray
 - 2. Devices connected to emergency power: red

2.2 RECEPTACLES

WIRING DEVICES

- C. Power feeder conductors shall be pulled with the use of an approved pulling compound or powder.
- D. Conductor splices shall be made only in readily accessible junction boxes.
- E. Cable supports and boxes shall be installed in all vertical feeders required by Article 300.19 of the National Electrical Code. Cables shall be supported at the top of the vertical raceway plus an additional support for each interval of spacing as specified in table 300.19 (A) of the NEC. For cables without a metallic sheath, the cable support shall be of the split wedge type which clamps each individual conductor firmly and tightens due to weight of cables.
- F. Conductors installed in raceways that are installed outdoors and exposed to direct sunlight shall have 30 degree Fahrenheit added to the outdoor ambient temperature used to calculate the conductor derating factor in compliance with NEC Article 310.10.

END OF SECTION 260120

WIRE AND CABLE

D. Control wiring shall be color coded such that red is used exclusively for all 120 volt conductors and white for all neutral conductors. All control wiring to be identified at both ends with permanent wire markers.

2.4 CONNECTORS

- A. In-line splices and taps for conductors # 8 AWG and smaller shall be 600V rated with "live spring" and insulated rigid nylon wing style body, 3M "Ranger", Buchanan "B-Cap", or equal.
- B. In-line connectors for 600V copper conductors # 6 AWG thru # 3 AWG shall be ILSCO type "CT" circumferential compression sleeves or equal by T&B or Buchanan.
- C. In-line connectors for 600V copper conductors # 2 AWG and larger shall be extra long barrel dual-crimp ILSCO type "CTL" compression sleeves (or equal by T&B or Buchanan), with 3M Brand "PST" silicone cold shrink insulators.
- D. Taps for copper conductor 600V or less, sizes No. # 6 AWG and larger shall be ILSCO "AH" series or equal by T & B or Buchanan.
- E. Cable terminations to bus bars, switch studs and terminal blocks, shall be Buchanan "Cytolok" two-hole mechanical connectors, or two-hole long-barrel dual-crimp compression type ILSCO or approved equivalent compression termination.
- F. Insulate splices and taps to thickness of conductor insulation with half-lapped layers of 3M "Scotch" branch No. 33 vinyl electrical tape. Connectors having irregular surfaces; fill voids and smooth contours with 3M "Scotchfil" electrical putty prior to taping.

PART 3 - EXECUTION

3.1 PREPARATION

- A. For new construction, wiring shall not be installed in the conduit system until the building is enclosed and masonry work is completed.
- B. Conduit shall be swabbed free of moisture and debris prior to pulling in the conductors.

3.2 INSTALLATION

- A. All multi-wire branch circuits that feed more than one device or equipment mounted on, or wired from, a common yoke, shall be powered from a branch circuit overcurrent device that simultaneously disconnects all ungrounded supply conductors in full compliance with NEC Article 210.
- B. Feeder conductors shall be routed continuous from origin to destination, without splicing, unless specifically noted otherwise on the drawings.

WIRE AND CABLE

- A. Wire and cable for power, lighting, control and signal circuits shall have copper conductors of not less than 98% conductivity and shall be insulated to 600V. Conductors shall be stranded except where specifically noted otherwise.
- B. Wire and cable type for the various applications shall be as follows:
 - 1. Type THHN, THWN or XHHW (75°C): Use for branch circuits, panel and equipment feeders in wet and dry locations.
 - 2. Type THHN or XHHW (90°C): Use for branch circuits, panel and equipment feeders in dry locations only. Use where lighting branch circuit conductors are routed through fluorescent fixture channels.
 - 3. Type UF: Use where permitted by other Sections or by the drawings for underground direct burial branch circuits.
 - 4. Type AF or SF-2 silicone rubber with heat-resistant glass braid (rated minimum 150°C) shall be used for branch circuit conductors connecting to fixture conductors within the fixture housing or to a connection box attached to the fixture and subject to temperatures equal to the temperatures within the fixture housing.

2.2 APPROVED WIRING SYSTEMS

A. All raceways, conductors, and wiring systems furnished and installed under this project shall be restricted to that specifically described on the electrical construction drawings and/or in the electrical specifications.

2.3 CONDUCTOR COLOR CODING AND IDENTIFICATION

A. Wiring systems shall be color coded. Conductor insulation shall be factory colored in sizes up through No. 8 AWG. Conductors No. 6 AWG and larger shall have black insulation and shall be phase color coded with one half inch band of colored tape at all junctions and terminations. Colors shall be assigned to each conductor as described below and carried throughout all main and branch circuit distribution.

		208/120 Volt	480/277 Volt	
1.	Phase 'A' conductor	Black	Brown	
2.	Phase 'B' conductor	Red	Orange	
3.	Phase 'C' conductor	Blue	Yellow	
4.	Neutral conductor	White	Gray	
5.	Equipment grounding conductor	Green	Green	
6.	Isolated equipment grounding conductor	Green with	Green with Yellow Stripe	

- B. Where more than one nominal voltage system is present within a premise, the identification used for each phase, neutral, and ground conductor shall be permanently posted at each distribution panel and at each branch panel.
- C. Contractor shall take extraordinary care to ensure that phase and bus orientation in each and every panel is identical.

SECTION 260120 - WIRE AND CABLE

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. The other Contract Documents complement the requirements of this Section. The General Requirements apply to the work of this Section.

1.2 SCOPE

A. Furnish material, tools, labor and supervision necessary to install wire and cable.

1.3 STANDARDS AND CODES

- A. Methods of installation shall comply with the provisions of applicable sections of NEC, Article 300.
- B. Materials shall be in accordance with NEC, Article 310 and shall be UL listed for application intended.

1.4 DESCRIPTION

- A. This section describes the basic materials and methods of installation for general wiring systems of 600 volts and less. Wiring for a higher voltage rating, if required, shall be as specified in other sections or as called for on the drawings.
- B. Minimum size conductors shall be No. 12 AWG for power and lighting. No. 14 AWG and smaller conductors shall only be used for signal and control.
- C. Refer to Specification Section 260053 for specific instructions with respect to sizing and installation of feeder and branch circuit conductors.

1.5 QUALIFICATIONS

A. The material used for the wiring systems shall be the products of a manufacturer regularly engaged in the manufacturing of the specified material. Where a manufacturer is named for a particular material, the material of other manufacturers will be acceptable provided the material meets requirements of the Specifications.

PART 2 - PRODUCTS

2.1 WIRE AND CABLE

WIRE AND CABLE

- B. For concealed conduit, install boxes flush with ceiling or wall, with covers accessible and easily removable. Where flush boxes are installed in finished ceilings or walls, provide cover which shall exceed the box face dimensions by a sufficient amount to allow no gap between box and finished material.
- C. Boxes shall not be exposed in finished, occupied rooms, without prior approval of Architect.
- 3.4 HANGER AND SUPPORT INSTALLATION
 - A. Hangers and supports shall be installed for all conduit and boxes.
 - B. Conduit and boxes shall not be attached to or supported from mechanical pipes, plumbing pipes or sheet metal ducts.
 - C. Conduits routed in lay-in grid ceiling plenum shall not be supported from the ceiling hanger iron or ceiling tees.

END OF SECTION 260111

CONDUIT SYSTEMS

- H. Where conduits cross building expansion joints, O-Z Gedney Company type "DX" conduit expansion fittings complete with bonding jumpers shall be used.
- I. Metal conduit buried in earth fill shall be protected with an approved corrosion resistant material.
- J. Conduits shall be capped during construction to prevent the entrance of foreign materials and moisture.
- K. Conduits installed on exteriors of buildings or other structures shall be arranged to drain (not trap water), and shall be rain-tight in wet locations.
- L. Liquid-tight flexible metal conduit and wiring methods containing non-metallic outer jacket shall not be installed in environmental air plenums.

3.2 OUTLET BOX INSTALLATION

- A. Outlet boxes shall be installed for fixtures, switches, receptacles and other devices.
- B. Approximate location of outlets are shown on the plans, but each outlet location shall be verified by the Contractor before installing the outlet box.
- C. Openings for electrical boxes in fire-rated walls that do not exceed 16 square inches in area are permitted in fire-rated construction provided that the aggregate area of such openings does not exceed 100 square inches for any 100 square feet of wall area.
- D. Where service utility boxes greater than 16 square inches exist in fire-rated wall construction, if the opening is not otherwise detailed to maintain the fire-rated integrity of the wall, provide firestopping wrap to the back side of each utility box.
- E. Outlet boxes on opposite sides of fire-rated walls shall be separated by a horizontal distance of not less than 24 inches.
- F. Outlet boxes on opposite sides of a common wall shall not be installed back-to-back. Outlet boxes on opposite sides of a common wall shall be separated 6" (minimum). Where separation is less than 12", install sound absorption insulation material between boxes and seal any conduits between boxes with duct seal at both ends.
- G. Outlet boxes shall be installed plumb and square with wall face and with front of box or cover located within 1/8" of face of finish wall. Boxes in masonry shall be set with bottom of the box tight to the masonry unit.

3.3 PULL AND JUNCTION BOX INSTALLATION

A. Install pull boxes and junction boxes where indicated on Drawings and where required to facilitate installation of the wiring.

CONDUIT SYSTEMS

PART 3 - EXECUTION

3.1 CONDUIT INSTALLATION

- A. In general, horizontal runs of conduit shall be installed in ceiling plenum. Conduit for convenience outlets, wall mounted fixtures and other wall outlets shall be routed overhead and concealed in wall to the outlet. Conduit shall not be installed in concrete floor slabs except where conditions will not permit the conduit to be routed overhead.
 - 1. Liquid-tight flexible metallic conduit, PVC conduit, and all other wiring methods containing nonmetallic outer jackets shall not be installed in environmental air plenums.
- B. Generally, conduit shall be concealed, except in crawl spaces, tunnels, shafts, mechanical equipment rooms, and at connections to surface panels and free-standing equipment, and as otherwise noted on Drawings. Surface raceways shall not be installed on floor surfaces.
- C. Exposed conduit shall be routed in lines parallel to building construction lines. Exposed conduit locations shall be approved by the Architect prior to installation.
- D. No conduit shall be installed less than 6" from piping installed by other trades. Conduits shall be located to avoid ductwork, piping and other obstructions.
- E. Certain conduits are permitted to be embedded in structural concrete work. Contractor shall cooperate with other Contractors of their respective trades to affect the following:
 - 1. Reinforcing steel shall be securely anchored in place before installing conduit.
 - 2. No steel reinforcing shall be displaced from plan dimensions without approval of Architect.
 - 3. Conduit shall not be placed over top of reinforcing or under bottom of reinforcing, where crossing beams.
 - 4. Conduit and fittings shall not displace concrete in columns in excess of 4% of total cross-section area of column without approval of Architect.
 - 5. Conduit shall not be placed closer than 3 diameters on center.
 - 6. Maximum size of embedded conduit or pipe shall not exceed 1/3 thickness of structural slab; 2/3 thickness of topping slab.
- F. Minimum size conduit shall be 1/2" trade size. Where specific size is not called for on Drawings or in the specification, Contractor shall select size required from Chapter 9 of NEC. Where specific sizes required by Drawings or Specifications are larger than Code requires, the larger size shall be installed.
- G. Install the conduit system mechanically and electrically continuous from outlet to outlet and to cabinets, junction or pull boxes. Conduit shall enter and be secured to cabinets and boxes in such a manner that all parts of the system will have electrical continuity. Feeder raceways shall terminate in cabinets and pull boxes with double locknuts and insulating bushings.

FAIRFIELD COUNTY UTILITIES 10/6/2022

F. Grade level junction boxes shall be manufactured by Synertech, Quazite, or CDR Systems Corporation with open flared bottom and cover. Logo on cover to read "ELECTRIC", etc. Enclosures and covers shall be concrete gray color and rated for no less than 5,000 pounds over a 10" x 10" area and be designed and tested to temperatures of -50 degrees F. Material compressive strength should be no less than 11,000 psi. Covers shall be secured with two pentahead stainless steel bolts. Bolts shall be retained in lid when unscrewed. Bolts shall be secured to replaceable threaded inserts with openings at base to allow debris to fall through and thereby avoiding clogged threaded inserts.

2.5 HANGERS AND SUPPORTS

- A. Provide conduit hanger and support devices of approved type for required methods of support to include: structural steel members, suspension rods, conduit clamps, concrete inserts, expansion shields, beam clamps and welding pins. All devices shall have galvanized finish or other approved corrosion resistive finish. In general, hangers and supports shall be as follows:
 - 1. Where a multiple run of conduit is routed on surface of structure, use conduit clamps mounted on Unistrut or equal channel so as to maintain clearance between conduit and structure.
 - 2. Where single run of conduit is suspended from overhead; use split ring conduit clamp suspended by steel drop rod.
 - 3. Where multiple parallel runs of conduit are suspended from overhead; use split ring conduit clamps uniformly spaced and supported on trapeze hangers fabricated of Unistrut Channels, suspended by not less than 1/2" continuously threaded steel drop rods.
 - Where conduit is buried in concrete floor slabs; anchor conduit with conduit clamps, at 10'-0" (maximum) intervals.
 - 5. Any form of strap iron or wire hangers will not be accepted.
 - 6. Maximum hanger and support spacing shall be in accordance with NEC Sections 342.30 (IMC), 344.30 (GRC), and 358.30 (EMT). Regardless of listed spacing provide additional hangers or supports at not more than 2'-0" from each change of direction and at each side of any box or fitting.
- B. Hangers and supports shall be anchored to structure as follows:
 - 1. Hangers and supports anchored to poured concrete: Use malleable iron or steel concrete inserts attached to concrete forms.
 - 2. Hangers or supports anchored to precast concrete: Use self-drilling expansion shields. Expansion shields may also be used where concrete inserts have been missed or additional support is required in poured concrete.
 - 3. Hangers or supports anchored to structural steel: Use beam clamps and/or steel channels as required by structural system.
 - 4. Hangers or supports anchored to metal deck: Use spring clips or approved welding pins. Maximum permissible load on each hanger shall not exceed 50 pounds.
 - 5. The use of explosive force hammer actuated, booster assist or similar anchoring device will not be permitted without prior approval of the Architect.

CONDUIT SYSTEMS

through No. 958. Boxes to be provided with raised adapters of depth as required for thickness of wall materials.

- 2. In masonry and poured concrete walls: For single outlets requiring two conduit connections in top and/or bottom of box use 4" square by 2-1/8" deep box No. 232 or 233 with raised square cut adapter. For ganged outlets use 3-3/4" high by 2-1/2" deep multiple gang masonry boxes No. 691 through No. 694 and No. 960 through No. 964.
- 3. Surface mounted wall outlets: For single outlet use 2-1/8" deep handy box No. 674, for two outlets use 4" square by 2-1/8" deep box No. 232 or 233. For more than two ganged outlets use 3-3/4" by 2-1/2" deep multiple gang masonry boxes No. 692 through No. 694 and No. 960 through No. 964. Boxes to be provided with 1/2" raised cover as required for device.
- 4. In suspended ceilings: Use 3-1/2" deep octagon box No. 280 or No. 281 with fixture studs and steel mounting bars.
- 5. In poured concrete ceiling slabs: Use octagon concrete rings with back plates.
- 6. Where outlet boxes are free standing on conduit stubs in kitchens, laundries, shops and other areas indicated, use Crouse-Hinds Type FS or FD malleable iron cadmium finish boxes with appropriate gasketed cover plate to suit device.
- 7. Outlets installed outdoors or in wet locations: Use Crouse-Hinds Type FS or FD box with NEMA 3R coverplates listed for "raintight while in use" for receptacles. Covers for switches shall be Crouse Hinds No. DS185. Diecast "bell" type boxes are not acceptable.
- 8. Floor outlets in above grade concrete slabs: Use concrete tight stamped steel galvanized box with fully adjustable top, Hubbell No. B-2527 for greater than 3" fill, No. B-2529 for 2" to 3" fill. Floor outlets in concrete slabs on grade: Use watertight cast iron box with fully adjustable top, Hubbell No. B-2536 for greater than 3" fill, No. B-2537 for 2" to 3" fill. Furnish for each outlet a No. S-2525 cover. Service fittings shall be as described on the Drawings. Furnish for each outlet in carpeted floor a No. S-3082 carpet flange.

2.4 PULL AND JUNCTION BOXES

- A. Construction, size and installation of pull and junction boxes shall comply with NEC, Article 314.
- B. Pull and junction boxes shall be fabricated of heavy gauge galvanized steel with screw covers, brass screws and hardware with enamel finish.
- C. Junction boxes shall be bonded to the equipment ground conductors contained within, whenever said conductors are "spliced" or "terminated" within the box, in full compliance with NEC Article 250.148.
- D. Pull and junction boxes for installation in poured concrete floors shall be flush type, cast iron, with watertight gasketed covers. Boxes for installation in floors with tile or carpet floor covering shall have recessed covers to accommodate the floor covering.
- E. Pull and junction boxes for above grade outdoor installations shall be rain-tight.

CONDUIT SYSTEMS

- 2. Use where installed exposed to outdoor conditions and where specifically required by the drawings.
- H. Metal Clad Cable (NEC Type MC)
 - 1. At the Contractor's option, Metal Clad cable assembly (NEC type MC) may be used only where permitted by national and local codes, and its use shall be further limited as follows:
 - For receptacle and equipment branch circuit wiring only where concealed in drywall partition walls or in casework. MC cable is not permitted to be installed above finished ceilings or exposed on walls.
 - b. MC Cable is not permitted for feeder circuits.
 - c. MC Cable is not permitted for emergency power circuits.
 - 2. All MC Cable shall by UL listed for its use. Where MC Cable is installed in Healthcare Facilities (as defined by NEC), the cable assembly shall be rated for the application.
 - 3. Conductors contained in MC Cable assembly shall comply with Specification 260120.

2.2 CONDUIT FITTINGS

- A. Rigid Conduit Fittings: Threaded, galvanized malleable iron or heavy steel, water and concrete tight.
- B. Metallic Tubing Fittings: Set screw type galvanized steel, concrete tight. Die cast type indentor type fittings will not be allowed.
- C. Flexible Metal Conduit Fittings: Squeeze or screw type galvanized malleable iron or steel with nylon insulated throats, or steel with set screws.
- D. Liquid-tight Flexible Metal Conduit: Galvanized malleable iron or steel, with watertight gaskets, "O" ring and retainer, and nylon insulated throats.
- E. Condulet Fittings: Exposed conduit fittings shall be condulet type for sharp turns, tees, etc. Condulet covers shall be gasketed where exposed to moisture.
- F. Threaded conduit terminations for weatherproof applications shall be made by use of Myers Hubs.

2.3 OUTLET BOXES

- A. Material, size and installation for outlet boxes shall comply with NEC Article 314.
- B. Boxes shall be Raco, Steel City, Appleton or equivalent, catalog numbers listed below are based on Raco, unless otherwise indicated. In general the type of boxes shall be as follows:
 - 1. In stud walls: For single outlet use 4" square by 2-1/8" deep box No. 232 or 233. For ganged outlets use 4-1/2" high by 1-13/16" deep multiple gang boxes No. 951

CONDUIT SYSTEMS

FAIRFIELD COUNTY UTILITIES 10/6/2022

- 2. Use where installed in poured concrete walls, columns, floors or other concrete slabs and where specifically required by the drawings.
- B. Electrical Metallic Tubing (EMT)
 - 1. Thinwall, electrically welded cold rolled steel conduit, galvanized inside and out by electro galvanized process and with baked clear elastic enamel coating.
 - 2. Use where installed concealed in walls, above finished ceilings, and where exposed in dry locations.
- C. Flexible Metallic Conduit
 - 1. Formed of one continuous length of spirally wound electro galvanized steel strip.
 - 2. Use for final connections to mechanical equipment, transformers, and for wiring within casework and millwork. Aluminum flexible conduit may be used for connections from junction boxes to lighting fixtures.
- D. Liquid-tight Flexible Metallic Conduit
 - 1. Formed of one continuous length of spirally wound steel strip, with water and oil tight neoprene jacket.
 - 2. Use for final connections to equipment listed in paragraph C above when located in wet or damp areas.
 - 3. Liquid-tight conduit shall not be used in environmental air plenums.
- E. Non-Metallic PVC Conduit
 - 1. Heavy wall rigid, rated for 90° cable, composed of polyvinyl chloride and shall conform to NEMA Standards. Conduit, fittings, and pipe-joining materials shall be produced by the same manufacturer.
 - 2. PVC conduit shall be Schedule 40 or greater unless specifically indicated otherwise on the drawings.
 - 3. Use where permitted by NEC and local jurisdiction and where buried outside building, encased in concrete, or in/below floor slabs. PVC conduit shall not be used in environmental air plenums.
- F. Intermediate Metallic Conduit (IMC)
 - 1. May be used as approved by the NEC for 2" and larger where rigid conduit is required or specified. IMC shall not be used for conduit buried in earth fill.
- G. Coated Rigid Metallic Conduit (CGRC)
 - Full weight, threaded, rigid steel, 40 mil PVC coated on outside, 2 mil urethane coated on inside conduit. Approved sources for this raceway are Robroy "Plastibond-Red" or Ocal, Inc "Ocal-Blue". All associated raceway fittings, sweeps, etc., shall be coated. Field cut raceways shall be touched up with matching finish.

FAIRFIELD COUNTY UTILITIES 10/6/2022

SECTION 260111 - CONDUIT SYSTEMS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. The other Contract Documents complement the requirements of this Section. The General Requirements apply to the work of this Section.

1.2 SCOPE

- A. Furnish materials, tools, labor and supervision necessary to fabricate and install complete conduit systems.
- B. Conduit systems shall be provided for all wiring, except where the Drawings or other specification sections indicate that wiring is permitted to be installed without conduit.

1.3 STANDARDS AND CODES

- A. Methods of fabrication and installation shall comply with the provisions of applicable Section NEC, Article 300.
- B. Materials shall be UL and NEC approved for the application intended.
- C. Materials shall meet or exceed ANSI and NEMA standards.

1.4 DESCRIPTION

A. This Section describes the basic materials and methods of installation for circular cross section conduit systems. Other types of conduit or raceways when required shall be as specified in other Sections, or as called for on the Drawings.

1.5 QUALIFICATIONS

A. The materials used in the fabrication of the conduit system shall be products of a manufacturer regularly engaged in the manufacturing of the specified material. Where a manufacturer is named for a particular material, the material of other manufacturers shall be acceptable provided the material meets requirements of the specification.

PART 2 - PRODUCTS

- 2.1 CONDUIT REQUIREMENTS
 - A. Rigid Metallic Conduit (GRC)
 - 1. Full weight, threaded, rigid steel conduit, galvanized inside and out by hot dip or electro galvanizing process with electrostatically applied baked coating.

CONDUIT SYSTEMS

260111 - 1

- B. The Electrical Contractor shall perform all backfilling associated with the above described excavation work.
 - 1. Backfill about the structures shall be placed, when practical, as the work of construction progresses. Backfilling on or against concrete work shall be done only when directed.
 - 2. Backfilling of duct lines shall progress as rapidly as the testing and acceptance of the finished sections of the work will permit and shall be carried to a crown approximately six (6) inches above the existing grades. In backfilling around duct lines, selected material shall be compacted firmly around and to a depth of not less than six (6) inches over the top of the duct. Rough grading shall be compacted thoroughly in layers and shall be brought up to within six (6) inches of finished grades. Fill and backfill shall be clean and free from vegetable matter, sticks, rocks and refuse.
 - 3. Backfill under roadways, drives and parking areas shall be bank run gravel or approved granular material.
 - 4. Backfill under building walls, and/or footers shall be concrete of the same strength as walls of footers.

END OF SECTION 260060

SECTION 260060 - EXCAVATING, TRENCHING, BACKFILLING AND RESTORATION

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. The other Contract Documents complement the requirements of this Section. The General Requirements apply to the work of this Section.

1.2 SCOPE

- A. Furnish materials, tools, labor and supervision necessary to provide all excavating, trenching and backfilling required for the proper installation of electrical equipment and wiring.
- B. Exact routing of trenching shall be determined by the Electrical Contractor and approved, in advance, by the Construction Manager and the Engineer.
- PART 2 PRODUCTS Not Used

PART 3 - EXECUTION

3.1 INSTALLATION

- A. The Electrical Contractor shall perform all excavating and trenching required for light pole bases, underground wiring, conduit and duct banks and shall backfill trenches and excavations after work has been inspected.
 - 1. Trenches shall be opened in straight lines and bottomed out at least 4" below conduits or ducts. Minimum depth as indicated shall be maintained between top of largest conduit or duct and finish grade.
 - 2. Care shall be taken in excavating that walls and footings and adjacent load bearing soils are not disturbed in any way, except where lines must cross under a wall footing. Where a line must pass under a footing, the crossing shall be made by the smallest possible trench to accommodate the duct or conduit.
 - 3. Excavations shall be kept free from water by pumping if necessary. No greater length of trench shall be left open, in advance of conduit laying or duct installation, than that which is authorized or directed by the Construction Manager and/or the Architect.
 - 4. Contractor shall cut any interfering trees, remove all stumps, rocks, etc. in the line of the excavation. Approval of the Engineer must be obtained before any tree is removed.
 - 5. Any shrubbery in line of excavation shall be removed with a ball of dirt and replaced at completion of excavation.
 - 6. Roots shall be removed to a level of eighteen (18) inches below finished grades and deeper as required for duct runs, manholes, and light pole bases. No roots shall be allowed to remain under the work.

FAIRFIELD COUNTY UTILITIES 10/6/2022

(THIS PAGE INTENTIONALLY LEFT BLANK)

CUTTING AND PATCHING

260054 - 2

06/06/2023

SECTION 260054 - CUTTING AND PATCHING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. The other Contract Documents complement the requirements of this Section. The General Requirements apply to the work of this Section.

1.2 SCOPE

A. Furnish materials, tools, labor and supervision necessary to cut and patch walls, floors, ceilings, etc as required for the installation of the electrical work.

1.3 DESCRIPTION

A. This Section describes the cutting and patching required by the installation of the electrical work and for the removal of existing electrical devices and wiring.

PART 2 - PRODUCTS

2.1 MATERIALS

- A. Provide proper materials to properly cut and patch the construction.
- B. Provide firestopping to maintain wall and floor ratings, as described in subsequent specification section.

PART 3 - EXECUTION

- A. This Contractor shall provide all holes and channels required for concealed installation of electrical wiring and equipment.
- B. Holes and channels shall be cut as small as practical and in a manner satisfactory to the Owner.
- C. This Contractor shall patch and finish sand all holes and channels cut for the installation of electrical wiring and electrical equipment and shall patch all damage caused by the installation and/or removal of electrical wiring and equipment.
- D. Finishing (paint, wall covering, etc.) shall not be included under this Section unless indicated otherwise on the Drawings.

END OF SECTION 260054

CUTTING AND PATCHING

- I. Cables shall not be bent, either permanently or temporarily during installation, to radii less than 10 times the outer diameters, except where shorter radii are approved for conditions making the specified radius impractical.
- J. No. 14 AWG wire and smaller shall be permitted only for systems control and alarm circuits.
- K. All wiring for the individual specified systems (fire alarm, telephone/data, intercom, paging, etc.) shall be as scheduled on the drawings and/or as described within the appropriate spec Sections.
- L. Refer to Section 260120 for conductor insulation, color coding, connectors, and support requirements.

END OF SECTION 260053

260053 - 4

- C. Feeder conductors shall be routed continuous from origin to destination, without splicing, unless specifically noted otherwise on plans.
- D. Refer to Section 260120 for feeder conductor insulation, color coding, connectors, and support requirements.

3.3 BRANCH AND SYSTEM CIRCUITS

- A. A general layout of branch circuit wiring is indicated on the Drawings. Generally, receptacles and appliances shall be on separate circuits from lighting.
- B. Branch panel circuits are numbered to match NEMA pole numbering system; poles 1 and 2 Phase A, poles 3 and 4 Phase B, poles 5 and 6 Phase C, etc.
- C. Where homerun circuit numbers are shown on drawings, such numbers shall be followed in connecting circuits to panelboards.
- D. Multi-wire branch circuits (as defined by NEC) are not permitted to be utilized. It is therefore required that a dedicated neutral conductor be installed for each single-phase branch circuit (as shown on plans). This installation requirement avoids NEC Article 210.4 necessity for multi-pole disconnecting means of multi-wire branch circuits. Multi-pole disconnecting means of individual single-pole branch circuits is not permitted.
- E. Where panelboard cabinets are recessed, conduits with sufficient capacity to carry the required number and size of future conductors for all spare branch circuit protective devices and spaces in the panelboard shall be stubbed out above accessible ceilings. In no case shall there be less than three 3/4" conduit stubbed out.
- F. Branch circuit conductor sizes shall minimally be #12 AWG. Where the length of a homerun, from panel to first outlet, exceeds 75 feet for a 120 volt circuit or 175 feet for a 277 volt circuit, the minimum conductor size shall be #10 AWG.
- G. In general, conductor sizes for larger branch circuits, such as motor and equipment branch circuits, are noted on Drawings. Where conductor sizes for such circuits are not noted, Contractor shall provide branch circuit conductors sized as follows:
 - 1. Conductors for individual motor branch circuits shall have ampere capacity of not less than 125% of the running current of the motors (Article 430.22, NEC).
 - Conductors for multiple motor branch circuits shall have ampere capacity of not less than 125% of the running current of the largest motor plus 100% of the running current for each additional motor connected to the circuit (Article 430.24, NEC).
 - 3. Conductors for individual or multiple equipment branch circuits shall have an ampere capacity of not less than 125% of the total connected ampere load served by the branch circuit.
- H. Where specific conductor sizes required by the Drawings are larger than the NEC requires, the larger sizes shall be installed.

GENERAL WIRING

B. Unless specifically allowed in Specification Section 260111 or in the contract electrical drawings, the Electrical Contractor shall not install Type AC armored cable, Type FC flat cable, Type FCC flat conductor cable, Type MC metal-clad cable, Types NM / NMC / NMS nonmetallic sheathed cable, Types SE / USE service cable or Type UF underground feed cable.

PART 3 - INSTALLATION

3.1 GENERAL

- A. All wiring shall be furnished and installed complete from point of service connection to all receptacles, lighting fixtures, clocks, fans, power outlets and outlets for future extensions, etc., as indicated on the drawings. Ample slack wire shall be provided for motor loops, service connections, etc. Unless otherwise specified, all branch circuit conductors shall be # 12 AWG or larger. In outlet boxes for future installation of devices, ends of wires shall be taped and tagged for identification at both ends and outlets supplied with blank covers.
- B. All conductors not larger than #10 located in branch circuit panelboards, signal cabinets, signal control boards, switchboards and motor control centers shall be neatly and securely bundled. All conductors larger than #10 located in switchboards, motor control centers and pull boxes shall be neatly and securely cabled in individual circuits. Bundling and cabling shall be done with either (1) marlin twine or 3 ply lacing or (2) nylon straps made of self-extinguishing nylon having temperature range of 65°F to 350°F. Each strap shall be constructed with a locking hub or head on one end and a taper on the other. Arcproofing, where required on feeders shall be applied after cabling.
- C. Branch circuit conductors to internal wiring connections of an individual surface or recessed fluorescent lighting fixture housing or the first unit of a surface or recessed multiple unit fluorescent lighting fixture housing shall be Type XHHW or THHN.
- D. Branch circuit wiring within fixture wireways of multiple unit sections or surface, recessed or pendant type fluorescent lighting fixtures shall be minimum #12 AWG stranded XHHW or THHN.

3.2 FEEDER CIRCUITS

- A. A riser diagram, distribution diagram, and/or a general layout of feeder circuits are indicated on the Drawings. Where feeders are located on the drawings, the Contractor shall install the feeders generally as indicated, but shall determine the exact location and routing of feeders to best fit the field conditions.
- B. In general, conductor sizes for feeder circuits are noted on the Drawings. Where conductor sizes for feeder circuits are not shown, the Contractor shall immediately notify the Engineer, who in-turn, will provide the missing information and further directives.

FAIRFIELD COUNTY UTILITIES 10/6/2022

SECTION 260053 - GENERAL WIRING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. The other Contract Documents complement the requirements of this Section. The General Requirements apply to the work of this Section.

1.2 SCOPE

- A. Provide materials, equipment, labor and supervision necessary to install feeder, branch, control, and system circuits as required by the Drawings and this Section, to include:
 - 1. Conductors
 - 2. Conduit Fittings and Boxes
 - 3. Overcurrent Protection
 - 4. Panelboards
 - 5. Conduit Hangers and Supports
 - 6. Wiring Devices
 - 7. Motor and Equipment Connections

1.3 STANDARDS AND CODES

- A. Methods of fabrication and installation shall comply with the provisions of applicable articles in the NEC.
- B. Materials shall be UL and NEC approved for the application intended.

PART 2 - PRODUCTS

2.1 RELATED EQUIPMENT AND MATERIAL

- A. The equipment and material related to feeder and branch circuit systems as called for on the Drawings and specified in the electrical specifications.
- B. The materials used in the installation of general wiring shall be products of manufacturers regularly engaged in the manufacturing of the specified material. Where a manufacturer is named for a particular material, the material of other manufacturers will be accepted provided the material meets requirements of the Specification.

2.2 APPROVED WIRING SYSTEMS

A. All raceways, conductors, and wiring systems furnished and installed under this project shall be restricted to that specifically described on the electrical construction drawings and/or in the electrical specifications.

FAIRFIELD COUNTY UTILITIES 4/6/2023

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

(THIS PAGE INTENTIONALLY LEFT BLANK)

BASIC ELECTRICAL REQUIREMENTS

260010 - 8

FAIRFIELD COUNTY UTILITIES 4/6/2023

E. Contractor shall coordinate dates and times for performing all demonstrations with the Owner.

1.16 OPERATION AND MAINTENANCE MANUALS

- A. Electrical Contractor shall furnish to the Owner operation/maintenance manuals as described in the Division 1 Specifications.
- B. Manuals shall meet or exceed all Specification requirements and shall minimally include three (3) individually bound and indexed (thumb tabbed) manuals. Each manual shall provide operating instructions, maintenance manuals, spare parts listing, copies of warranties, wiring diagrams, inspection procedures and shop drawings on all equipment and systems.
- C. Unless otherwise directed by the Architect, each manual shall be bound in a heavy-duty, 3 inch, three-ring vinyl covered binder with pocket folders for drawings and folded sheet information. Each binder shall be identified on both the front and the spine.

1.17 AS-BUILT DRAWINGS

- A. As work progresses during the construction period, the Electrical Contractor shall record (on a dedicated set of bid drawings) any deviations from the design drawings. The completed record set of as-built drawings shall be delivered to the Architect prior to the Electrical Contractor's request for final payment.
- B. As-built documentation shall meet or exceed all Specification requirements.

1.18 PROJECT CLOSE-OUT

- A. The installing Contractor shall contact the Engineers' office upon completion of the installation to request final inspection. At that time the following documents shall be assembled and provided for review at the job site:
 - 1. Photocopies of all branch panel schedules.
 - 2. Photocopies of all signed electrical inspection permits.
 - 3. O & M Manuals (as described above).
 - 4. Photocopies of certified test results, as required by all specification sections.
 - 5. "As-Built" print set.
 - 6. Photocopy of Printout from Fire Alarm System listing device addresses and custom labels.

PART 2 – PRODUCTS – Not Used

PART 3 – EXECUTION – Not Used

END OF SECTION 260010

BASIC ELECTRICAL REQUIREMENTS

260010 - 7

1.14 TESTS

- A. Systems shall be tested by the Electrical Contractor and placed in proper working order prior to demonstrating systems to Owner.
- B. After work is completed a load balance test shall be made by the Electrical Contractor to demonstrate that with full lighting and mechanical load the balance between phases is within 5%. Unbalance beyond this limit shall be corrected. Special care shall be taken during load balance adjustments to assure that reverse rotation of motors does not occur.
- C. System ground shall be tested to demonstrate that the ground resistance does not exceed the requirements of NEC.
- D. Perform such tests as required by authorities having jurisdiction over the site.
- E. Perform tests as described in all subsequent sections of the Electrical Specifications.

1.15 DEMONSTRATIONS

- A. Prior to acceptance of the work, the Contractor shall demonstrate to the Owner or his designated representative all features and functions of all systems and shall instruct the Owner in the proper operation of the systems. Each system shall be demonstrated once.
- B. The demonstrations shall consist of not less than the following:
 - 1. Point out the actual location of each component of a system and demonstrate its function and its relationship to other components within the system.
 - 2. Demonstrate the electrical systems by actual "start-stop" operation showing how to work controls, how to reset protective devices, how to replace fuses, and what to do in an emergency.
 - 3. Demonstrate communication, signal, alarm and detection systems by actual operation of the systems and show how to reset signal, alarm and detection devices.
- C. Systems to be demonstrated shall include but not be limited to the following:
 - 1. Service and Power Distribution Systems
 - 2. Lighting and Lighting Controls Systems
 - 3. Emergency Lighting Systems
 - 4. Motor and Equipment Control
 - 5. Alarm Detection and Signal Systems
 - 6. Communication Systems
 - 7. Standby Power System
- D. Contractor shall furnish the necessary trained personnel to perform the demonstrations and instruction, and shall arrange to have the manufacturer's representatives present to assist with the demonstrations.

1.12 NAMEPLATES AND LABELS

- A. The Electrical Contractor shall furnish and install a system of nameplates designed to identify each piece of equipment, control unit thereon, and major distribution points. The following color scheme shall be used as a guide:
 - 1. For switchboards, panelboards, control centers, panels, switches, and devices served by "normal" power, use black plastic laminated nameplates with white engraved letters. For equipment served by "emergency" power, use red plastic laminated nameplates with white engraved letters. Unless noted otherwise on the drawings, nameplates shall identify equipment name/designation.
 - 2. For fire alarm system cabinet and panels, use red laminated plastic nameplates with white engraved letters.
 - 3. Size of nameplates shall be made to readily differentiate between, and identify, equipment and usage. Nameplate identifying items that are transferred to emergency power shall carry a nameplate saying "EMERGENCY".
 - 4. Exposed feeder conduits shall be identified as to load fed and voltage (Normal or Emergency) with 1" high black stenciled letters and numerals; conduit shall be marked every 50 feet and at the supply end of the feeder. This shall include all "spare" conduits.
- B. A label reading "contains emergency circuits" shall be installed on all boxes and enclosures that contain emergency powered circuits to comply with NEC Article 700. Labels shall be installed on front covers of all pullboxes, junction boxes and control enclosures.
- C. Fasten nameplates to all enclosures by use of self-adhesive material.
- D. The Electrical Contractor shall furnish and install Arc Flash Warning labels in a clearly visible location on the front trim of all switchboards, panelboards (inside the hinged panel cover), industrial control panels, meter socket enclosures, and motor control centers to comply with NEC Article 110.

1.13 CLEANING AND PAINTING

- A. Touch up and repair any damaged factory finishes on equipment and materials furnished. Other painting will be done under the Painting Division of the Specifications.
- B. Remove any rust spots and prime with rust inhibitive paint any metal surfaces of electrical devices not provided with rust inhibitive coatings. Then apply one coat of paint in color as directed by Architect.
- C. Swab interiors of conduits clean and dry before pulling wire. Clean interiors of boxes and cabinets before installing trims and covers.

1.9 GUARANTEE

- A. The Electrical Contractor shall guarantee for a period of one year that all work and equipment will remain free from all defects in workmanship and materials, and that it will comply with all the specific requirements of the Specifications and other Contract Documents governing the work.
- B. All work found to be defective will be replaced with new work meeting all the requirements of the Contract. The Electrical Contractor will bear all costs of supplying such new work, and installing and finishing same, and will assume all costs for replacing other work damaged by the removal and replacement of any of the work. The Electrical Contractor will bear all costs for freight, drayage and demurrage, and all labor in connection therewith.

1.10 SUBMITTALS

- A. This Contractor shall prepare or obtain from the manufacturer certified shop drawings for all equipment and material to be furnished as indicated in each of the individual Specification Sections.
- B. Submit hard copies of each submittal to the Architect for approval before proceeding with installation or construction. Refer to Division 1 specifications for further instructions. Electronic submittals are not acceptable.
- C. AutoCAD floor plans are available to Vendors and Contractors to assist in generation of shop drawings. A nominal fee will be charged to prepare the plans for transmission. Refer to Attachment "A" of this specification section for the required faxable "Electronic File Transfer Agreement" form.
- D. Prior to the signing of the contract, the successful bidder shall submit to the Architect a list of manufacturers of the major items of equipment he proposes to furnish and the names of any subcontractors he proposes to employ.

1.11 SUBSTITUTION

- A. Bidders desiring to make a substitution for the specified brand or method shall list such proposed substitution. In each case state the difference in price where substitution is offered. If there is no difference in price, so state.
- B. It shall be understood that the proposal submitted shall be based on the different branches of work and materials specified, and that the Owner is entitled to the use of the materials so specified. Substitution sheet shall be signed and dated by the Electrical Contractor and shall be formatted as follows:

Brand or make specified

Proposed Substitution

Add Deduct

- G. Architect reserves the right to make reasonable changes in indicated locations without extra cost to the Owner.
- H. Drawings other than electrical drawings, and other sections of this Specification, may show or specify electrically operated equipment, wiring diagrams, etc. The Contractor shall examine all such drawings and specification sections and become familiar with the characteristics and required connections for all equipment.
- I. Conduits, wiring and equipment shall be arranged substantially as indicated. Any change resulting in a savings in labor or material shall be made only in accordance with a contract change order. Deviations shall be made only where necessary to avoid interferences and only after drawings showing the proposed deviations have been submitted to and approved by the Architect.

1.8 MATERIALS

- A. All equipment and devices shall be new and shall conform to NEMA and Underwriters' Laboratories Standards. Where Specifications describe, or plans show, materials or equipment of higher quality than required by code and local ruling, the Drawings and Specifications shall govern the quality of the material or equipment.
- B. Materials and equipment used as extensions to existing special systems shall be of matching electrical characteristics for satisfactory operation of the complete system and shall be of the same manufacture and design unless otherwise approved.
- C. The Contractor shall submit proof, if requested by the Architect, that the materials, appliances, equipment or devices that he furnishes and installs under this contract, meet the requirements of the Underwriters' Laboratories, Inc. and its publications will be referred to hereinafter by the abbreviation UL, with or without additional identifying symbols.
- D. The National Electrical Code (NEC) of the National Fire Protection Association, and Publications and Standards of the organizations listed below are referenced herein by the abbreviations noted in parentheses, with or without additional identifying symbols. Unless otherwise specified, all work shall be manufactured, tested and installed in accordance with such reference standards.
 - 1. American Society for Testing and Materials (ASTM)
 - 2. Underwriters' Laboratories, Inc. (UL)
 - 3. Insulated Power Cable Engineers Association (IPCEA)
 - 4. National Electrical Manufacturers Association (NEMA)
 - 5. Institute of Electrical and Electronic Engineers (IEEE)
 - 6. American National Standards Institute, Inc. (ANSI)
 - 7. National Fire Protection Association (NFPA)

1.5 VISIT TO THE SITE

A. The Electrical Contractor is not required to visit the site of the work but is available to visit the site and familiarize himself with all such conditions affecting the work. The submission of his bid proposal shall presuppose his knowledge of all such conditions.

1.6 WORKMANSHIP

- A. Employ only experienced craftsmen under direct supervision of a full time competent foreman.
- B. Keep fully informed as to progress of work, so that work of this Division may be built into place in sufficient time to insure against delay to other trades, and to prevent misalignments or damage to electrical work.
- C. All work shall be completed in a neat and workmanlike manner as described and illustrated in the ANSI standards publication "NECA 1-2000 Standard Practices for Good Workmanship in Electrical Contracting".

1.7 COORDINATION, CONDUCT AND SCHEDULING OF WORK

- A. Drawings are diagrammatic, indicating general arrangement, approximate sizes, general locations of equipment and outlets. Verify dimensions in field; adjust to manufacturer's shop drawings. Do not scale drawings.
- B. Architectural and structural drawings supersede electrical drawings. Determine that work of this Division can be accommodated within spaces provided. Notify Construction Manager and/or Architect of any interferences before starting installation.
- C. Determine sizes, locations for chases, openings necessary for installation of electrical work; cooperate with other trades in setting of sleeves, inserts and hangers.
- D. Coordinate and schedule all proposed power or utility cutovers and shutdowns in advance with the Owner. Where required by the Owner, perform cutover work during non-business hours or on weekends. Existing services shall be removed only after cutover to new services.
- E. Coordinate this work with all trades, serving utilities and equipment suppliers. Arrange operation, submittal approvals and equipment delivery, so as not to delay installation or completion of any parts of interrelated work so that construction may proceed on schedule.
- F. Cooperate with Mechanical trades in preparing interference drawings for points where there is possible conflict between trades. Exact locations of pipes, ducts, conduit based on field measurements with final arrangement to be determined by intra-trade agreements subject to Construction Manager's and/or Architect's review.

SECTION 260010 - BASIC ELECTRICAL REQUIREMENTS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Related and general provisions of Contract, including General and Supplementary Conditions and Division 1 Specification Sections, apply to this Section.

1.2 SCOPE

- A. The work shall include the furnishing of systems, equipment and materials specified in this Division and as called for on the Drawings, to include: supervision, operations, methods and labor for the fabrication, installation, start-up and tests for the complete electrical installation.
- B. Drawings for the work are diagrammatic, intended to convey the scope of the work and to indicate the general arrangement and locations of the work. Because of the scale of the Drawings, certain basic items such as conduit fittings, access panels, sleeves, pull and junction boxes may not be shown. Where such items are required by Code or by other Sections, or where they are required for proper installation of the work, such items shall be included.
- C. Equipment Specification may not deal individually with minute items such as components, parts, controls and devices which may be required to produce the equipment performance specified or as required to meet the equipment warranties. Where such items are required, they shall be included by the supplier of the equipment, whether or not specifically called for.

1.3 ELECTRICAL REFERENCE SYMBOLS

- A. Symbols used on the floor plans are defined in the Electrical Symbols Schedule on the Drawings. Not necessarily will all symbols scheduled be required for the project.
- B. The symbols used for schematic or one line power and control wiring diagrams are American Standard Graphical Electrical Symbols and are published in American Standard Chart Z32.3.

1.4 PERMITS, INSPECTIONS AND CODES

- A. The Contractor shall secure and pay for all permits and inspections required by the governing authorities for the prosecution of the electrical work. All permits and certificates of inspection and approval signed by the controlling building department shall become the property of the Owner.
- B. All wiring shall be in compliance with the current edition of the National Electric Code, applicable State and City regulations and OSHA. In cases of conflict between Code and Specifications, the more restrictive requirements shall govern.

FAIRFIELD COUNTY UTILITIES AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN 10/6/2022

3.7 CLEANING

A. Clean adjacent structures and improvements of dust, dirt, and debris caused by selective demolition operations. Return adjacent areas to condition existing before selective demolition operations began.

END OF SECTION 024119

SELECTIVE DEMOLITION

5

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

FAIRFIELD COUNTY UTILITIES 10/6/2022

- C. Removed and Salvaged Items:
 - 1. Clean salvaged items.
 - 2. Pack or crate items after cleaning. Identify contents of containers.
 - 3. Store items in a secure area until delivery to Owner.
 - 4. Transport items to Owner's storage area off-site.
 - 5. Protect items from damage during transport and storage.
- D. Removed and Reinstalled Items:
 - 1. Clean and repair items to functional condition adequate for intended reuse.
 - 2. Pack or crate items after cleaning and repairing. Identify contents of containers.
 - 3. Protect items from damage during transport and storage.
 - 4. Reinstall items in locations indicated. Comply with installation requirements for new materials and equipment. Provide connections, supports, and miscellaneous materials necessary to make item functional for use indicated.
- E. Existing Items to Remain: Protect construction indicated to remain against damage and soiling during selective demolition. When permitted by Architect, items may be removed to a suitable, protected storage location during selective demolition and cleaned and reinstalled in their original locations after selective demolition operations are complete.

3.5 SELECTIVE DEMOLITION PROCEDURES FOR SPECIFIC MATERIALS

- A. Concrete: Demolish in small sections. Using power-driven saw, cut concrete to a depth of at least 3/4 inch at junctures with construction to remain. Dislodge concrete from reinforcement at perimeter of areas being demolished, cut reinforcement, and then remove remainder of concrete. Neatly trim openings to dimensions indicated.
- B. Concrete: Demolish in sections. Cut concrete full depth at junctures with construction to remain and at regular intervals using power-driven saw, and then remove concrete between saw cuts.
- C. Masonry: Demolish in small sections. Cut masonry at junctures with construction to remain, using power-driven saw, and then remove masonry between saw cuts.
- D. Concrete Slabs-on-Grade: Saw-cut perimeter of area to be demolished, and then break up and remove.

3.6 DISPOSAL OF DEMOLISHED MATERIALS

- A. Remove demolition waste materials from Project site and dispose of them in an EPAapproved construction and demolition waste landfill acceptable to authorities having jurisdiction.
 - 1. Do not allow demolished materials to accumulate on-site.
 - 2. Remove and transport debris in a manner that will prevent spillage on adjacent surfaces and areas.
 - 3. Remove debris from elevated portions of building by chute, hoist, or other device that will convey debris to grade level in a controlled descent.
- B. Burning: Do not burn demolished materials,

cooling specified in Section 015000 "Temporary Facilities and Controls."

- B. Temporary Shoring: Design, provide, and maintain shoring, bracing, and structural supports as required to preserve stability and prevent movement, settlement, or collapse of construction and finishes to remain, and to prevent unexpected or uncontrolled movement or collapse of construction being demolished.
 - 1. Strengthen or add new supports when required during progress of selective demolition.
- C. Remove temporary barricades and protections where hazards no longer exist.

3.4 SELECTIVE DEMOLITION, GENERAL

- A. General: Demolish and remove existing construction only to the extent required by new construction and as indicated. Use methods required to complete the Work within limitations of governing regulations and as follows:
 - 1. Proceed with selective demolition systematically, from higher to lower level. Complete selective demolition operations above each floor or tier before disturbing supporting members on the next lower level.
 - 2. Neatly cut openings and holes plumb, square, and true to dimensions required. Use cutting methods least likely to damage construction to remain or adjoining construction. Use hand tools or small power tools designed for sawing or grinding, not hammering and chopping. Temporarily cover openings to remain.
 - 3. Cut or drill from the exposed or finished side into concealed surfaces to avoid marring existing finished surfaces.
 - 4. Do not use cutting torches until work area is cleared of flammable materials. At concealed spaces, such as duct and pipe interiors, verify condition and contents of hidden space before starting flame-cutting operations. Maintain portable fire-suppression devices during flame-cutting operations.
 - 5. Maintain fire watch during and for at least two hours after flame-cutting operations.
 - 6. Maintain adequate ventilation when using cutting torches.
 - 7. Remove decayed, vermin-infested, or otherwise dangerous or unsuitable materials and promptly dispose of off-site.
 - 8. Remove structural framing members and lower to ground by method suitable to avoid free fall and to prevent ground impact or dust generation.
 - 9. Locate selective demolition equipment and remove debris and materials so as not to impose excessive loads on supporting walls, floors, or framing.
- B. Site Access and Temporary Controls: Conduct selective demolition and debris-removal operations to ensure minimum interference with roads, streets, walks, walkways, and other adjacent occupied and used facilities.
 - 1. Install temporary 20'-wide gravel driveway west of driveway to airport terminal parking lot. Restore to grass after construction is complete.
 - 2. Conduct pavement demolition in both parking lots in halves to allow for traffic thoroughfare.
 - a. See Demolition Plan Drawings (Sheet 6) for additional reference.

B. Verify that hazardous materials have been remediated before proceeding with building demolition operations.

3.2 UTILITY SERVICES AND MECHANICAL/ELECTRICAL SYSTEMS

- A. Existing Services/Systems to Remain: Maintain services/systems indicated to remain and protect them against damage.
- B. Existing Services/Systems to Be Removed, Relocated, or Abandoned: Locate, identify, disconnect, and seal or cap off utility services and mechanical/electrical systems serving areas to be selectively demolished.
 - 1. Owner will arrange to shut off indicated services/systems when requested by Contractor.
 - 2. Arrange to shut off utilities with utility companies.
 - 3. If services/systems are required to be removed, relocated, or abandoned, provide temporary services/systems that bypass area of selective demolition and that maintain continuity of services/systems to other parts of building.
 - 4. Disconnect, demolish, and remove fire-suppression systems, plumbing, and HVAC systems, equipment, and components indicated on Drawings to be removed.
 - a. Piping to Be Removed: Remove portion of piping indicated to be removed and cap or plug remaining piping with same or compatible piping material.
 - b. Piping to Be Abandoned in Place: Drain piping and cap or plug piping with same or compatible piping material and leave in place.
 - c. Equipment to Be Removed: Disconnect and cap services and remove equipment.
 - d. Equipment to Be Removed and Reinstalled: Disconnect and cap services and remove, clean, and store equipment; when appropriate, reinstall, reconnect, and make equipment operational.
 - e. Equipment to Be Removed and Salvaged: Disconnect and cap services and remove equipment and deliver to Owner.
 - f. Ducts to Be Removed: Remove portion of ducts indicated to be removed and plug remaining ducts with same or compatible ductwork material.
 - g. Ducts to Be Abandoned in Place: Cap or plug ducts with same or compatible ductwork material and leave in place.

3.3 PROTECTION

- A. Temporary Protection: Provide temporary barricades and other protection required to prevent injury to people and damage to adjacent buildings and facilities to remain.
 - 1. Provide protection to ensure safe passage of people around selective demolition area and to and from occupied portions of building.
 - Provide temporary weather protection, during interval between selective demolition of existing construction on exterior surfaces and new construction, to prevent water leakage and damage to structure and interior areas.
 - 3. Protect walls, ceilings, floors, and other existing finish work that are to remain or that are exposed during selective demolition operations.
 - 4. Cover and protect furniture, furnishings, and equipment that have not been removed.
 - 5. Comply with requirements for temporary enclosures, dust control, heating, and

1.5 CLOSEOUT SUBMITTALS

FAIRFIELD COUNTY UTILITIES

10/6/2022

A. Inventory: Submit a list of items that have been removed and salvaged.

1.6 FIELD CONDITIONS

- A. Owner will occupy portions of building immediately adjacent to selective demolition area. Conduct selective demolition so Owner's operations will not be disrupted.
- B. Conditions existing at time of inspection for bidding purpose will be maintained by Owner as far as practical.
- C. Notify Architect of discrepancies between existing conditions and Drawings before proceeding with selective demolition.

D. Hazardous Materials: Present in buildings and structures to be selectively demolished. A report on the presence of hazardous materials is on file for review and use. Examine report to become aware of locations where hazardous materials are present.

- 1. Hazardous material remediation is specified elsewhere in the Contract Documents.
- 2. Do not disturb hazardous materials or items suspected of containing hazardous materials except under procedures specified elsewhere in the Contract Documents.

1.7 SOLID AND HAZARDOUS WASTE MANAGEMENT

A. The contractor shall manage all solid and Hazardous wastes generated in carrying out the work in compliance with Ohio's solid and hazardous waste laws, ORC Chapter 3734 and the regulations promulgated thereunder.

1.8 COORDINATION

A. Arrange selective demolition schedule so as not to interfere with Owner's operations.

PART 2 - PRODUCTS

2.1 PERFORMANCE REQUIREMENTS

- A. Regulatory Requirements: Comply with governing EPA notification regulations before beginning selective demolition. Comply with hauling and disposal regulations of authorities having jurisdiction.
- B. Standards: Comply with ASSE A10.6 and NFPA 241.

PART 3 - EXECUTION

- 3.1 EXAMINATION
 - A. Review Project Record Documents of existing construction or other existing condition and hazardous material information provided by Owner. Owner does not guarantee that existing conditions are same as those indicated in Project Record Documents.

SECTION 024119 - SELECTIVE DEMOLITION

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

A. Section Includes:

- 1. Demolition and removal of selected portions of building or structure.
- 2. Demolition and removal of selected site elements.
- 3. Salvage of existing items to be reused or recycled.

1.3 DEFINITIONS

- A. Remove: Detach items from existing construction and dispose of them off-site unless indicated to be salvaged or reinstalled.
- B. Remove and Salvage: Detach items from existing construction, in a manner to prevent damage, and deliver to Owner ready for reuse.
- C. Remove and Reinstall: Detach items from existing construction, in a manner to prevent damage, prepare for reuse, and reinstall where indicated.
- D. Existing to Remain: Leave existing items that are not to be removed and that are not otherwise indicated to be salvaged or reinstalled.
- E. Dismantle: To remove by disassembling or detaching an item from a surface, using gentle methods and equipment to prevent damage to the item and surfaces; disposing of items unless indicated to be salvaged or reinstalled.

1.4 MATERIALS OWNERSHIP

- A. Unless otherwise indicated, demolition waste becomes property of Contractor.
- B. Historic items, relics, antiques, and similar objects including, but not limited to, cornerstones and their contents, commemorative plaques and tablets, and other items of interest or value to Owner that may be uncovered during demolition remain the property of Owner.
 - 1. Carefully salvage in a manner to prevent damage and promptly return to Owner.

FAIRFIELD COUNTY UTILITIES 10/6/2022

(THIS PAGE INTENTIONALLY LEFT BLANK)

- - the second se

d and a second

CLOSEOUT PROCEDURES

FAIRFIELD COUNTY UTILITIES 10/6/2022

B. Repair, or remove and replace, defective construction. Repairing includes replacing defective parts, refinishing damaged surfaces, touching up with matching materials, and properly adjusting operating equipment. Where damaged or worn items cannot be repaired or restored, provide replacements. Remove and replace operating components that cannot be repaired. Restore damaged construction and permanent facilities used during construction to specified condition.

END OF SECTION 017700

CLOSEOUT PROCEDURES

AIRPORT SANITARY SEWER EXTENSION AND WATER SERVICE DESIGN

- FAIRFIELD COUNTY UTILITIES 10/6/2022
 - 1. Submit on digital media acceptable to Owner.
 - D. Warranties in Paper Form:
 - 1. Bind warranties and bonds in heavy-duty, three-ring, vinyl-covered, loose-leaf binders, thickness as necessary to accommodate contents, and sized to receive 8-1/2-by-11-inch paper.

PART 2 - PRODUCTS - Not Used

PART 3 - EXECUTION

3.1 FINAL CLEANING

- A. Perform final cleaning. Conduct cleaning and waste-removal operations to comply with local laws and ordinances and Federal and local environmental and antipollution regulations.
 - 1. Complete the following cleaning operations before requesting inspection for certification of Substantial Completion for entire Project or for a designated portion of Project:
 - a. Clean Project site of rubbish, waste material, litter, and other foreign substances.
 - b. Clean exposed exterior and interior hard-surfaced finishes to a dirt-free condition, free of stains, films, and similar foreign substances. Avoid disturbing natural weathering of exterior surfaces. Restore reflective surfaces to their original condition.
 - c. Remove debris and surface dust from limited-access spaces, including roofs, plenums, shafts, trenches, equipment vaults, manholes, attics, and similar spaces.
 - d. Clean flooring, removing debris, dirt, and staining; clean according to manufacturer's recommendations.
 - e. Vacuum and mop concrete.
 - f. Wipe surfaces of mechanical and electrical equipment and similar equipment. Remove excess lubrication, paint and mortar droppings, and other foreign substances.
 - g. Clean ducts, blowers, and coils if units were operated without filters during construction or that display contamination with particulate matter on inspection.
- B. Construction Waste Disposal: Remove all construction waste from the site specified in Section 017419 Construction Waste Management and Disposal. Construction materials classified as construction and demolition debris shall be disposed of in a certified landfill.

3.2 REPAIR OF THE WORK

A. Complete repair and restoration operations, before requesting inspection for determination of Substantial Completion.

FAIRFIELD COUNTY UTILITIES 10/6/2022

6-6-23 (which is the Effective Date of the Contract). This Agreement will be effective on Owner: Contractor: Commissioners Board of tairtield County aw General Contraction 100 (typed or printed name of organization) ted name of organization By: By: (individual's signature) (individual's signature) Date: Date: 5-17-23 (date signed) (date signed) Name: Davis Name: Edwards (typed or printed) (typed or printed) esider Title: Title: Treasurer (typed or printed) vped or printed) (If Contractor is a corporation, a partnership, or a joint venture, attach evidence of authority to sign Attes Attest: individual's signature offic P Title: Title (typed or pi (typed or printed) Address for giving notices: Address for giving notices: 6670 Lock ville Rd Lew General Contracting Inc Corroll, 04:0 43112 Mt Vernun Rd 9128 St Louisville, Ohio 4307 **Designated Representative: Designated Representative:** Name: lony VoseL Name: Brice Colbin (typed or printed) (typed or printed) Director of UTiliTies (typed or printed) Title: Title: President (typed or printed) Address: Address: 6670 Lock ville Rd 9128 Mt Vernon Rd Curroll, ohio 47112 St Louisville, Ohio 4307 740-652-7120 Phone: Phone: 740 404 2574 Email: Email: Brice Q lawytrame. com (If Owner is a corporation, attach evidence of authority License No.: NIA to sign. If Owner is a public body, attach evidence of (where applicable) authority to sign and resolution or other documents authorizing execution of this Agreement.) State: Ohiu

IN WITNESS WHEREOF, Owner and Contractor have signed this Agreement.

EICDC® C-520, Agreement between Owner and Contractor for Construction Contract (Stipulated Price). Copyright® 2018 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

Page 7 of 8

ORIGINAL		Carrí L. Brown, phd, MBA, CGFM	Purchase Order			
		Fairfield County Auditor	Fiscal Year 2023	Page: 1 of 1		
		210 East Main Street Lancaster, Ohio 43130	THIS NUMBER MUST APPEAR ON ALL INVOICES, PACKAGES AND SHIPPING PAPERS.			
Γ	B COUNTY COMMISSIONERS	Revisions: 000	Purchase Order #	23004771 - 00		
	210 E MAIN ST 3RD FLOOR LANCASTER, OH 43130		Delivery must be made within doors of specified destination.			
	T O		Expiration Date	: 12/15/2024		
[V LAW GENERAL CONTRACTTING 9128 MT VERNON ROAD		COUNTY COMMISSIONE			
	9128 MT VERNON ROAD ST LOUISVILLE, OH 43071	H I P	210 E MAIN ST 3RD FLO LANCASTER, OH 43130	JUK		
	D ST LOUISVILLE, OH 43071 O R	Ŏ				

VENDOR PHONE NUMBER		DOR FAX NUMBER	REQUISITION NUMBER	DELIVERY REFERENCE	
			5162		
DATE ORDERED	VENDOR NUMBER	DATE REQUIRED	FREIGHT METHOD/TERMS	DEPARTMENT/LOCATION	
05/26/2023	12841			COMMISSIONERS ADMIN	
NOTES					

PO Requisitioner Name : Staci Knisley

E mail Address : staci.knisley@fairfieldcountyohio.gov

ITEM #	DESCRIPTION / PART #	QTY	UOM	UNIT PRICE	EXTENDED PRICE	
1	Airport water ARP project GL Account: 12287600 - 573700 - R511b	\$38,950.00	1.0	EACH	\$38,950.00	\$38,950.00
2	Airport Sewer ARP project GL Account: 12287600 - 573725 - R52a	1.0	EACH	\$456,050.00	\$456,050.00	
	GL SUMMARY					
-	12287600 - 573700 - R511b					
	12287600 - 573725 - R52a					

Invoice Date / / /	_ Invoice Amount \$	To Be paid	_//	Warrant #	
expenditure, for the above, has been	495,000.00 required to meet the contract, a lawfully appropriated, authorized or directed llection to the credit of the submitted Fur	ed for such purpose and is in th	е		
Date: 05/26/2023	- $ -$	own			
00/00/2022	Auditor Fairfield Count	y, OH	Purchase Orde	r Total	\$495,000.00
06/06/2023	For Dep	parment Use ONLY			840

OR	G	IINA

Prosecutor's Approval Page

Resolution No.

A resolution to approve the contract award for the Airport Sanitary Sewer Extension and Water Service; ARP fiscal recovery fund, #2876

(Fairfield County Utilities Department)

Approved as to form on 6/1/2023 5:30:33 PM by Amy Brown-Thompson,

Any Brown Thempson

Amy Brown-Thompson Prosecutor's Office Fairfield County, Ohio

Signature Page

Resolution No. 2023-06.06.u

A resolution to approve the contract award for the airport sanitary sewer extension and water service, ARP fiscal recovery Fund #2876.

(Fairfield County Utilities Department)

Upon the motion of Commissioner Jeffrey M. Fix, seconded by Commissioner David L. Levacy, this resolution has been Adopted:

Voting:

Steven A. Davis, President	Aye
David L. Levacy, Vice President	Aye
Jeffrey M. Fix	Aye

Board of County Commissioners Fairfield County, Ohio

CERTIFICATE OF CLERK

It is hereby certified that the foregoing is a true and correct transcript of a resolution acted upon by the Board of County Commissioners, Fairfield County, Ohio on the date noted above.

Pochelle Meninger

Rochelle Menningen Board of County Commissioners Fairfield County, Ohio 2023-06.06.v

A resolution authorizing the approval of payment of invoices for departments that need Board of Commissioners' approval.

WHEREAS, departments that need the Board of Commissioners' approval for payment of their invoices have submitted their invoices to the County Auditor; and

WHEREAS, the County Auditor has submitted the cash disbursement journal for payment of invoices for the check date of June 8, 2023.

NOW THEREFORE, BE IT RESOLVED, BY THE BOARD OF COUNTY COMMISSIONERS, FAIRFIELD COUNTY, STATE OF OHIO:

Section 1. That the Fairfield County Board of Commissioners approves the attached cash disbursement journal.

Prepared by: Auditor/Finance cc: Finance Office

INVOICES BY DEPARTMENT Department									o 06/08/2023
Check #		Vendor #	Vendor Name	Invoice #	Invoice Date	PO#	Warrant	Line Item Description	Amount
1200	COMMISSIONERS ADMIN Fund: 1001 - GENERAL FUND								
1579887	06/08/2023	80132	AUNDREA N CORDLE	5/17/23	05/17/2023	23000875	C0606	TRAVEL & EXPENSE REIMB	81.51
5376788	06/08/2023	3720	LANC FF CO CHAMBER OF COMMERCE	48390	05/26/2023	23000796	C0606	2023 CHAMBER ANNUAL AWARDS DAVIS/CORDLE/PORTER	165.00
							TOTAL	: COMMISSIONERS ADMIN	246.51

Summary Total for this report: \$246.51

Commissioner Steven A. Davis

Commissioner Jeffrey M. Fix

Commissioner David L. Levacy

Date

Signature Page

Resolution No. 2023-06.06.v

A resolution authorizing the approval of payment of invoices for departments that need Board of Commissioners' approval.

(Fairfield County Commissioners)

Upon the motion of Commissioner Jeffrey M. Fix, seconded by Commissioner David L. Levacy, this resolution has been Adopted:

Voting:

Steven A. Davis, President	Aye
David L. Levacy, Vice President	Aye
Jeffrey M. Fix	Aye

Board of County Commissioners Fairfield County, Ohio

CERTIFICATE OF CLERK

It is hereby certified that the foregoing is a true and correct transcript of a resolution acted upon by the Board of County Commissioners, Fairfield County, Ohio on the date noted above.

Rochelle Meninger

Rochelle Menningen Board of County Commissioners Fairfield County, Ohio







